Diet, Exercise, and Chronic Disease

The Biological Basis of Prevention



Diet, Exercise, and Chronic Disease

The Biological Basis of Prevention This page intentionally left blank

Diet, Exercise, and Chronic Disease

The Biological Basis of Prevention

Edited by C. Murray Ardies



CRC Press is an imprint of the Taylor & Francis Group, an **informa** business CRC Press Taylor & Francis Group 6000 Broken Sound Parkway NW, Suite 300 Boca Raton, FL 33487-2742

© 2014 by Taylor & Francis Group, LLC CRC Press is an imprint of Taylor & Francis Group, an Informa business

No claim to original U.S. Government works Version Date: 20140220

International Standard Book Number-13: 978-1-4398-5029-9 (eBook - PDF)

This book contains information obtained from authentic and highly regarded sources. Reasonable efforts have been made to publish reliable data and information, but the author and publisher cannot assume responsibility for the validity of all materials or the consequences of their use. The authors and publishers have attempted to trace the copyright holders of all material reproduced in this publication and apologize to copyright holders if permission to publish in this form has not been obtained. If any copyright material has not been acknowledged please write and let us know so we may rectify in any future reprint.

Except as permitted under U.S. Copyright Law, no part of this book may be reprinted, reproduced, transmitted, or utilized in any form by any electronic, mechanical, or other means, now known or hereafter invented, including photocopying, microfilming, and recording, or in any information storage or retrieval system, without written permission from the publishers.

For permission to photocopy or use material electronically from this work, please access www.copyright.com (http://www.copyright.com/) or contact the Copyright Clearance Center, Inc. (CCC), 222 Rosewood Drive, Danvers, MA 01923, 978-750-8400. CCC is a not-for-profit organization that provides licenses and registration for a variety of users. For organizations that have been granted a photocopy license by the CCC, a separate system of payment has been arranged.

Trademark Notice: Product or corporate names may be trademarks or registered trademarks, and are used only for identification and explanation without intent to infringe.

Visit the Taylor & Francis Web site at http://www.taylorandfrancis.com

and the CRC Press Web site at http://www.crcpress.com

Contents

Preface	vii
Acknowledg	gments1x
Contributor	s xiii
Chapter 1	Introduction
	C. Murray Ardies
Chapter 2	Inflammation
	C. Murray Ardies
Chapter 3	Diabetes
	Raymond E. Bourey, Meenakshi K. Kaw, Sumona Ghosh Lester, Simona S. Ghanem, and Sonia M. Najjar
Chapter 4	Atherosclerosis
	C. Murray Ardies and Christian K. Roberts
Chapter 5	Osteoporosis
	David W. Dodington and Wendy E. Ward
Chapter 6	Cancer
	Jonathan Cannizzo and C. Murray Ardies
Chapter 7	Neurodegenerative Disease
	Aparna Raghavan and Zahoor A. Shah
Chapter 8	Hunger and Satiety Signaling
	Denovan P. Begg and Stephen C. Woods
Chapter 9	Summary and Recommendations
	C. Murray Ardies

This page intentionally left blank

Preface

The idea of producing a book on the prevention of chronic diseases through exercise and diet was intriguing for me when I was first approached with the idea. I had been teaching nutrition, exercise, and health science courses with a focus on cellular aspects of prevention for many years and had to develop all of my own materials because very few books were available. And of those that were, they had a decided clinical approach with very little discussion about the actual biochemistry or molecular mechanisms involved in prevention. When I broached the concept with several potential coauthors, the prevailing opinion was that books that covered biochemical and molecular aspects of disease etiology also were in very short supply. Thus this book was born: an attempt to collate the latest cellular- and molecularbased research on the etiology of chronic diseases with how these mechanisms of cause are modified by various aspects of diet and exercise. Essentially, we have tried to produce a text that translates molecular-based data on etiology and prevention into a clinical prescription for the prevention of chronic disease.

The focus on diabetes, atherosclerosis, osteoporosis, cancer, and degenerative neurological disease is because they are the major causes of morbidity and mortality by chronic disease and also because there is sufficient molecular evidence for a strong dietary and activity (or rather, an insufficiency of both) component to their etiology. The inclusion of a separate chapter on "Inflammation" became necessary when it was clear that inflammatory signaling is a fundamental component of each of these diseases and that reducing inflammation is key to reducing risk for all of these diseases. At the time we started, obesity had not yet been declared a disease, so it did not get its own chapter. It is discussed as a major contributor to other diseases rather than a disease in and of itself. The chapter on "Hunger and Satiety Signaling" provides a very interesting and probably the most realistic take on the regulation of eating behavior available. As it becomes more obvious from reading this and the other chapters, many things may not really be what "everyone" seems to believe.

Ultimately, we hope that this book can help readers to develop a broader understanding of how chronic disease "works"; to better integrate molecular, biochemical, and cellular mechanisms of cause and prevention into the study of chronic disease; and to get excited about developing new research into the etiology of prevention and into clinical methodologies that translate that research into successful prevention strategies. With recent advances in techniques to determine various aspects of proteomics, metabolomics, epigenetics, and functional genomics, we can anticipate great things in the near future for prevention research. This page intentionally left blank

Acknowledgments

In the end, I am very grateful to all for the tremendous amount of work necessary to complete this text. From Wendy Ward and David Dodington, Sonia Najjar and Raymond Bourey, and Christian Roberts and "my" graduate student Jonathan Cannizzo who all signed on to this project at the very beginning to Stephen Woods and Denovan Begg and Zahoor Shah who helped rescue it (and then made it even better) when other authors had to leave the project for other commitments; I thank you all for your insight, dedication, and hard work in this very difficult undertaking in the face of never-ending grant deadlines, site visits, innumerable meetings with administrators and graduate students, and the constant laboratory work that never seems to end before midnight.

I also thank Tony Treston, director of manufacturing at Unither Virology (past research scientist at National Cancer Institute (NCI) and VP product development at EntreMed), for his many insightful editorial comments that helped make this text an easier "read"; Milie Fang, Biology Major at University of Illinois at Chicago, for her expert help in converting figure drawings into Adobe Illustrator; and Randy Brehm, senior editor, Chemical and Life Sciences Group at Taylor & Francis Group/CRC Press, for putting up with our ever lengthening schedule and ensuring that the project made it to completion.

This page intentionally left blank

Editor

C. Murray Ardies earned a bachelor of physical education from the University of Manitoba in Winnipeg, Canada, in 1975 following which he worked for the Government of Manitoba as a counselor and then as codirector of Operation ReNu, a public health education program that emphasized proper diet and activity as preventive medicine. In 1978, he earned a master's degree with a major in health education and a major in physical education along with a minor in nutrition from Northern Michigan University, Marquette. He then earned a multidisciplinary doctoral degree at The University of Texas in 1985 with majors in pharmacology, nutrition, and exercise physiology. His dissertation concerned the interactive effects of exercise and chronic ethanol consumption on mitochondrial metabolism in liver, where for the first time exercise was demonstrated to prevent alcohol-induced toxic reactions in nonmuscle tissues.

Dr. Ardies subsequently worked at the Icahn School of Medicine at Mount Sinai in New York City researching biochemical mechanisms of ethanol toxicity, where he developed new methodologies for CYP and CYP reductase purification and the determination of oxygen radical production; then in the Department of Endocrinology at Stanford University Medical School to purify steroid-binding proteins and clone their genes; and next in the Department of Anesthesia with Dr. James Trudell, where he helped characterize molecular aspects of alcohol toxicity involving direct autoimmune attack on liver cells.

In 1989, he joined Northeastern Illinois University, where he worked on defining mechanisms through which repeated endurance exercise reduces risk for chemical toxicities and cancer. As part of this work, he was among the first to demonstrate that beneficial alterations in cellular function were tied to a generalized stress response mediated by the activation of the AP-1 response element in nuclear DNA.

This page intentionally left blank

Contributors

C. Murray Ardies

Northeastern Illinois University Chicago, Illinois

Denovan P. Begg

Department of Psychiatry and Behavioral Neuroscience University of Cincinnati Cincinnati, Ohio

Raymond E. Bourey

Department of Medicine Center for Diabetes and Endocrine Research The University of Toledo Toledo, Ohio

Jonathan Cannizzo Cenegenics Chicago Chicago, Illinois

David W. Dodington

Center for Bone and Muscle Health Faculty of Applied Health Sciences Brock University St. Catharines, Ontario, Canada

Simona S. Ghanem

Department of Physiology and Pharmacology Center for Diabetes and Endocrine Research College of Medicine and Life Sciences The University of Toledo Toledo, Ohio

Meenakshi K. Kaw

Department of Physiology and Pharmacology Center for Diabetes and Endocrine Research College of Medicine and Life Sciences The University of Toledo Toledo, Ohio

Sumona Ghosh Lester

Department of Physiology and Pharmacology Center for Diabetes and Endocrine Research College of Medicine and Life Sciences The University of Toledo Toledo, Ohio

Sonia M. Najjar

Department of Physiology and Pharmacology Center for Diabetes and Endocrine Research College of Medicine and Life Sciences The University of Toledo Toledo, Ohio

Aparna Raghavan

Department of Medicinal and Biological Chemistry The University of Toledo Toledo, Ohio

Christian K. Roberts

Translational Sciences Section School of Nursing University of California, Los Angeles Los Angeles, California

Zahoor A. Shah

Department of Medicinal and Biological Chemistry The University of Toledo Toledo, Ohio Wendy E. Ward Center for Bone and Muscle Health Faculty of Applied Health Sciences Brock University St. Catharines, Ontario, Canada

Stephen C. Woods

Department of Psychiatry and Behavioral Neuroscience University of Cincinnati Cincinnati, Ohio

1 Introduction

C. Murray Ardies

CONTENTS

1.1	Introduction		
	1.1.1	Definition of Disease	1
	1.1.2	Disease Etiology and Prevention	2
Acknowledgment			9
Refe	References		
1010	i en e e o		1

1.1 INTRODUCTION

The focus of this chapter is on prevention of chronic diseases through diet and exercise. Chronic diseases comprise a major source of mortality from disease for the world and are by far the greatest source of mortality among the high-income countries.^{1,2} In the United States, chronic diseases account for about 70% of all deaths and 75% of health-care costs³ and their prevention is extremely important in terms of alleviating personal suffering and reducing economic impact, for individuals and for the nation. In addition, and possibly even more importantly, various aspects of a poor diet and physical inactivity account for the majority of risk factors among the top 15 sources of risk for both all-cause mortality and disease burden. These risks factors include overweight and obesity, high cholesterol, zinc deficiency, iron deficiency, alcohol use, child underweight, physical inactivity, vitamin A deficiency, and low fruit and vegetable intake.¹

Because issues related to diet and physical activity are so important for disease risk, it is certainly possible that developing preventive methods that focus solely on these parameters also could have a huge impact on morbidity and mortality from disease and, of course, on the economic issues as well. This is not to say that pharmacological- and nutraceutical-based management of disease risks has no place in prevention. They can be very important in reducing genetic- or environmentalbased risks as well as in correcting serious dietary deficiencies, regardless of the cause. They are, however, outside the scope of this chapter.

1.1.1 DEFINITION OF DISEASE

In order to create an emphasis on the biological (cellular and molecular) basis of prevention, the concept of disease used in this chapter is a slight variation of that commonly used elsewhere in a variety of books on prevention,^{4–8} medical textbooks,^{9–11} and medicine-related Websites.^{12–16} For our purposes, disease is a complex process that is initiated by detrimental chemical, biochemical, or cellular processes (endogenous or exogenous), either alone or in some combination thereof,

that if allowed to continue for long enough or are severe enough will result in some form of cellular damage, cellular dysfunction, or cell death. An array of symptoms that are dependent on the specific type of cellular dysfunction and the organ/tissue in which the dysfunction (or cell death) exists will occur only if the cellular dysfunction is prolonged and extensive enough (or if a sufficient number of cells die).

The major difference between the aforementioned definition and other definitions is that in many of the references listed both clinical symptoms and (cellular or organ) dysfunction coexist. Clinically observable symptoms are obviously a component of disease and are especially useful for the purposes of diagnosis. However, if symptoms are a necessity for disease to exist, then there can be no disease if there are no symptoms. For most experts in the biological and medical sciences this distinction may not be important, but for those who are less well versed in human biology it may make a large difference in terms of how they view prevention of disease. For the purposes of prevention in this chapter, the processes of disease that lead to symptomatic disease are the focus.

For a layperson who might be familiar only with those concepts promoted by the experts who write for Web-based media (or, as is far more likely, who has watched innumerable television advertisements that feature middle- to late-middle-aged people newly concerned about preventing coronary atherosclerosis, stroke, osteoporosis, or prostate cancer), it may appear that many diseases just do not *happen* until around 40 years of age or later and that preventing diseases does not start until then either. This incorrect interpretation creates a dilemma for those trying to elucidate and correct the root causes of preventable diseases. On the other hand, many people simply may not be interested in prevention until it is much too late just because they do not feel sick in spite of the fact that they know that various disease processes may have been going on for many years.

In order to create a timely focus on prevention, symptomatic disease needs to be separated from the processes of disease. To use a well-understood example, in atherosclerosis the build-up of plaque in the walls of coronary arteries occurs over a period of several decades before clinical symptoms actually appear. If the concept of disease requires symptoms, then this particular disease does not really start for the *average* person until about the fourth decade of life.¹⁷ The reality, however, is that the processes that produce symptoms of coronary artery disease are ongoing continuously for decades prior to the appearance of the symptoms with asymptomatic lesions already evident in early childhood.^{18–21}

Therefore, prevention also should be considered to be a continuous process, one that starts long before symptoms appear. From the aforementioned atherosclerosis example, because the processes of disease start before the appearance of fatty lesions in early childhood prevention must start before early childhood as well. From this standpoint, prevention is a *lifelong* issue and not a middle-age issue. Further, because poor diet and a lack of exercise are integral to high risk for chronic diseases, prevention is clearly a *lifestyle* issue.

1.1.2 DISEASE ETIOLOGY AND PREVENTION

This returns us to the definition of disease. By broadening the concept of disease to include the processes that produce cellular dysfunction, which can then progress to produce symptoms of disease, a focus on cellular and molecular mechanisms of disease etiology (or cause) is forced on us. Attenuating these mechanisms, which ultimately lead to enough dysfunction to cause symptoms, then becomes the focus of prevention. To clarify, the term *disease* refers to symptomatic disease, whereas the term *disease processes* refers to various mechanisms of disease etiology that occur prior to the appearance of symptoms (Figure 1.1).

A molecular approach to studying the etiology of diseases has revealed that there are cellular mechanisms that are common to many different diseases, which leads us to our approach to prevention. Rather than looking for specific treatments to prevent specific diseases, we look at integrating common mechanisms of disease etiology into a (single) cell-function model that can then be applied to a variety of different cell types and different disease outcomes. This cell-function model of disease should then allow us to develop a (single) lifestyle model of prevention that is applicable to most, if not all, diseases.

To continue with the example of atherosclerosis, the physical perturbation of membrane and cellular proteins of the endothelial cells of coronary arteries caused by turbulent flow, along with chemical-mediated and oxygen radical-mediated trauma, leads to the production of proinflammatory signaling molecules: various eicosanoids and cytokines. These, in turn, lead to the attraction and activation of monocytes to the intima at the site of initial trauma. These monocytes develop into (lipid-filled) foam cells that also produce additional inflammatory molecules as well as more damage-causing radical species. The continuing (low-level) production of inflammatory cytokines leads to the gradual accumulation of more foam cells, smooth muscle cell proliferation, and synthesis of fibrotic tissue, a vicious cycle of inflammatory responses progressing into the development of atherosclerotic plaque.



FIGURE 1.1 Effects of positive and negative effectors on morbidity and death: a variety of factors produce positive or negative effects on cell function with some being able to produce both, depending on the amounts produced and the duration of effects. Those factors with predominantly negative effects on cell function can enhance the rate of our inevitable progression toward death while increasing morbidity due to disease. Those factors with predominantly positive effects can optimize function and reduce risk for morbidity to promote healthy aging to senescence.

Eventually, the plaque grows sufficiently large to impede or block blood flow through the coronary artery or to break off to produce a thrombus, thus producing symptoms of the disease.^{18–29}

Inflammation-related cellular signaling molecules (and their effects on cell function) and reactive molecules (a variety of reactive oxygen and other molecular species) produced by activated inflammatory cells are therefore integral to the development of atherosclerosis.^{19,22,24–29} They are also involved in many cancers,^{30–35} and in degenerative neurological disorders such as Parkinson's^{36–39} and Alzheimer's.^{40–42} In addition, they also contribute to osteoporosis^{43–46} and type 1 and 2 diabetes,^{47–50} with the inflammatory mechanisms in type 1 diabetes most likely being a result of autoimmune attack on B cells of the pancreas.^{51–53} With such global effects on disease etiology mechanisms associated with inflammation should be common primary targets for prevention of chronic diseases.

The details of inflammation are discussed in Chapter 2, whereas the inflammatory mechanisms associated with diabetes, atherosclerosis, osteoporosis, cancer, and neurodegenerative disease are discussed in Chapters 3 through 7. As described in these chapters, cell functions that are affected by various inflammation-related signaling molecules play a contributory role in all of these diseases, as does the molecular damage caused by radical species produced by activated inflammatory cells. In addition to inflammatory factors, noninflammatory factors related to poor diet and inactivity also are important components of many disease processes.

Although the general concepts of poor diet and inactivity might seem straightforward, they actually are fairly complex. Regarding dietary issues, *poor diet* might refer to the consumption of insufficient amounts of specific nutrients leading to nutritional imbalances. It might also refer to the consumption of large amounts of refined carbohydrates, or to excessive alcohol consumption, too little fiber, too many calories, or any one of a myriad number of excesses or specific nutrient insufficiencies that may occur in spite of consuming macronutrients in adequate amounts. From our prevention standpoint, it also refers to consuming too few phytochemical-rich foods while maintaining both caloric balance and proper nutrition.

In general, detrimental alterations in cell function due to a combination of poor diet, excessive calorie intake, and lack of physical activity are the fundamental causes of obesity and metabolic syndrome.^{54–56} Although not usually considered a disease per se, metabolic syndrome is a collection of coexisting risk factors associated with atherosclerosis, type 2 diabetes, and stroke that includes abdominal obesity, hypertension, insulin resistance, and dyslipidemia.^{54,56} Metabolic syndrome and obesity, and by default poor diet, excess calories, and inactivity, are also causally associated with many cancers.^{55,57–61} Metabolic syndrome, along with other dietary factors, is also causally associated with neurological disease^{62–65} and osteoporosis.^{66–68} Further, when studied in isolation of other risk factors a lack of physical activity seems to be associated with an increased risk for just about any chronic disease,^{69–74} so much so that some of the leading experts in our field have suggested that inactivity should be considered a disease in itself.⁷⁴

The overall consumption of calories and the amount of calories expended through various activities are fundamental issues in weight control. As detailed in Chapters 2

through 8, it is how these variables affect disease etiology through altering cellular metabolism and activities of signal transduction pathways that, in part, reveals how prevention through weight control works. In addition to the roles of proper diet and caloric balance in the prevention of diseases, specific components of foods called phytochemicals are also very important for prevention.

Phytochemicals are nonnutritive molecules from plants that produce biological effects that are beneficial to health.^{75–77} The roles that phytochemicals play in reducing risk for disease are very complex. For some phytochemicals the molecular mechanisms of prevention may be related to their antioxidant or prooxidant properties, whereas for others the ability to modify enzyme activity or the cellular expression of various cytokines, enzymes, and transcription factors appears to be integral to their protective effects.^{77–85} In terms of overall prevention of disease, it is most likely through the additive and synergistic effects of many *different* phytochemicals that their preventive effects are realized.^{77,79,82,83,85} This implies, of course, that incorporating a variety of phytochemical-rich foods into an otherwise nutritionally adequate and calorie-balanced diet may enhance the preventive effects of the said diet, a concept that is directly in line with one of the major objectives of this chapter: the development of a (single) lifestyle model of optimal prevention that is applicable to most, if not all, chronic diseases.

When it comes to exercise and prevention, it is important to recognize that exercise is not just a *calorie burner*. Exercise can also produce changes to cell function that in some respects are similar to those observed with many phytochemicals. Activation of signal transduction pathways and transcription factors along with altered expression of various functional proteins have all been observed in skeletal muscle cells⁸⁶⁻⁹⁰ and in nonmuscle cells such as adipose, liver, lung, and brain,⁹¹⁻⁹⁶ to reference just a few of the available studies. Just to clarify, for our purposes, exercise is much more than physical activity. Physical activity refers to any bodily movement produced by skeletal muscles that results in an energy expenditure that is substantially greater than that at rest⁹⁷ and refers to a single session of activity. The term *exercise* or *physical* exercise as defined by Courneya and Friedenreich⁹⁸ is any physical activity that is performed repeatedly over prolonged periods of time with the intent of improving fitness or health. Because improvement of health is central to the concept of prevention, it is the incorporation of exercise into our daily lifestyle that is relevant for prevention. How exercise affects cell function is the means through which exercise alters health.

A simplified general model of some of the molecular aspects of cellular function is illustrated in Figure 1.2. Changes in the activity of various signal transduction pathways due to responses to extracellular signaling molecules (hormones, cytokines, etc.), to damage from reactive oxygen species and other chemical oxidants, and to changes in the intracellular redox environment lead to alterations in the expression of various proteins that are important in maintaining cellular integrity and function. Deleterious cellular responses can lead to disruptions in cell function that may then lead to tissue dysfunction and ultimately to symptoms of disease. Cellular responses to different forms of exercise (or lack thereof) and to various compounds obtained from the diet and the environment can modify these responses to subsequently alter risks for cellular dysfunction and disease.



FIGURE 1.2 A model for the molecular aspects of cell function: alterations in activity of various signaling pathways that regulate the activation of transcription factors are fundamental to the mechanisms through which disease processes result in cellular dysfunction as well as how physical activity and dietary components alter risk for chronic disease. Filled small arrows indicate signal transduction pathways that lead to activation of various transcription factors. Open arrows indicate some of the mechanisms through which the activity of these pathways is modified by both intra- and extracellular factors.

Although such a model may have utility for aiding in understanding a complex topic, it can also be used to assist in testing hypotheses related to disease prevention by highlighting potential biomarkers of risk and effect. Biomarkers are simply any molecular marker obtained from tissues or biological fluids that can be used to indicate a biological event has occurred, and they can be used for a variety of purposes including indicating some sort of cellular damage has occurred or some risk exists for future events.⁹⁹ For example, changes over time in the level of urinary 8-hydroxy-deoxyguanosine (8-OH-dG), a hydroxyl radical–caused DNA adduct, can be used to indicate that genotoxic exposures of different degrees have occurred, that there have been changes in the amount of repair of oxidative damage to DNA, as well as a marker for potential future risk for cancer.¹⁰⁰ On a cellular or tissue basis, alterations in the activity of redox-control enzymes and antioxidant enzymes may indicate changes in risk for cellular and tissue damage and dysfunction due to exposures to cellular oxidants from either endogenous or exogenous sources. Acute changes in redox and antioxidant status may result in altered serum levels of certain proteins if the changes alter the expression of proteins that are exported from the cell. Stress-response signaling molecules such as certain cytokine antagonists, for example, appear in the blood following long-term metabolic stress due to exercise, and they initiate a variety of cellular responses that are anti-inflammatory in nature and may then influence many disease processes.^{101,102}

The use of such biomarkers in prevention research would require detailed knowledge of how they specifically relate to both cell function and the etiology of disease. Fortunately, as discussed in detail in this book, the various roles that many potential biomarkers play in the etiology of chronic diseases is known. One advantage of utilizing biomarkers in research is that hypothesized paradigms for prevention can be tested in a relatively short time based on the following premise: if there are no changes in mechanisms, there are no changes in risks. For example, changes in activation patterns of transcription factors or in the activity of antioxidant enzymes can be observed within minutes to hours after the start of exercise.86-88,90-92,103 Conversely, looking for an actual change in incidence of a particular disease in order to experimentally determine the clinical effectiveness of a specific form of exercise may take many years or even decades before any indication of an actual preventive effect might be observable. This does not mean that random assignment, placebocontrolled clinical trials should not be performed; they are essential and are the gold standard for definitive evidence of effectiveness. Judicious use of biomarkers in experiments can, however, provide valuable information that can lead to the development of more refined prevention paradigms that are based on observed alterations in mechanisms that are part of the etiology of disease rather than on associations observed in epidemiology surveys.

The actual amount and intensity of exercise required to reduce risk for various cancers, for example, is not well understood.¹⁰⁴ There have been conflicting recommendations for exercise, depending on which professional body is making the recommendations.^{105–108} The recommendations range from a minimum of 60 minutes of continuous moderate-intensity activity each day for children and adolescents to a minimum total of 20 minutes of moderate-intensity activity spread throughout the day on most days, a very wide range. There is also a general recognition by

these same groups that additional benefits are gained by following the longer, more intense guidelines, raising the issue of why one set of recommendations would be made when a different one will result in greater health gains.

The aforementioned recommendations for physical activity are based on epidemiological assessments of reported patterns of physical activity and incidence of disease,^{105,106,108} and they all suffer the limitations of such efforts: inaccurate memories of just how much activity was performed, lack of precise control over a large array of specific variables (including diet), and no actual measurement of cellular function variables directly affected by exercise and associated with cause of disease, to name a few. In fact, one of the major factors in producing some of the guidelines was the type of exercise that increases maximum oxygen consumption,¹⁰⁵ which is hardly a cellular mechanism associated with the etiology of any disease. It is even plausible to interpret many of the observed preventive benefits of exercise as happening simply because the detrimental effects of obesity/overweight were avoided. Only through understanding the molecular mechanisms of prevention obtained through weight management and those obtained exclusively from exercise is it even possible to determine whether it is the exercise, weight management, or a combination of both that promotes optimal prevention. For the sake of argument, if it is weight management that does the job of prevention, and there are no additional beneficial effects from exercise per se, then exercise would not even be necessary for prevention. Maintaining a healthy body weight through diet alone is certainly possible; it just may not be as easy without exercise.

Of course, the bulk of research evidence certainly does suggest that both exercise and diet can have a profoundly positive effect on health. Further, as described in Chapters 2 through 8, the molecular mechanisms of how those effects are realized are now being delineated. These molecular events not only help to explain how prevention works but also form the basis for the development of specific biomarkers of effect that provide more meaningful measurements of the preventive effects of dietary components and of various types, amounts, and intensities of exercise. As suggested by Rundle,¹⁰⁴ the concept of using molecular markers (or biomarkers) of risk is well established in the field of cancer epidemiology,^{109–111} and a similar approach should be followed by investigators in the exercise–health field as well.

Only through a comprehensive analysis of the molecular etiology of chronic diseases can targets for effective prevention be determined. Further, effective prevention strategies can be devised only through an understanding of how various dietary components and forms of exercise affect these molecular mechanisms. In addition, it is only through molecular approaches to epidemiology and experimental clinical trials that effective prevention protocols can be validated and refined without waiting for decades for symptomatic disease to (not) appear.

Ultimately, we have to start somewhere in our quest for a definitive model for a prevention-based lifestyle. Where known, each chapter will delineate mechanisms through which specific dietary components and different forms of exercise may reduce risk for the specific chronic disease or syndrome. Because inflammatory mechanisms underlie many chronic diseases, a recommendation for an anti-inflammatory lifestyle may present a reasonable model from which to start. In the end, all of the recommendations will be consolidated into a comprehensive prevention diet complete with

recommendations for specific foods and numbers of servings, as well as for specific forms of exercise based on the preventive benefits of each. Because these recommendations are made on the basis of molecular-based evidence on how etiology of disease is affected by dietary components or specific forms of exercise, they will differ in both quality and quantity from the current recommendations that are predominantly based on epidemiological associations between diet, exercise, and disease risk. As such, the recommendations might be considered to comprise a set of research hypotheses that are designed to translate existing evidence of how dietary components and exercise stresses alter molecular mechanisms of cell function to alter the etiology of chronic disease.

ACKNOWLEDGMENT

This work was supported in part by a partial salary grant from NEIU to C. Murray Ardies.

REFERENCES

- Lopez AD, Mathers CD, Ezzati M, Jamison DT, Murray CJ. Global and regional burden of disease and risk factors, 2001: Systematic analysis of population health data. *Lancet*. 2006;367:1747–1757.
- Mathers CD, Loncar D. Projections of global mortality and burden of disease from 2002 to 2030. *PLoS Medicine*. 2006;3:e442.
- Henry J. Health Care and the 2008 Election. Menlo Park, CA: Kaiser Family Foundation; 2008.
- 4. Simopoulos AP. Evolutionary Aspects of Nutrition and Health: Diet, Exercise, Genetics, and Chronic Disease. Basel, Switzerland: Karger; 1999.
- 5. McTiernan A. Cancer Prevention and Management Through Exercise and Weight Control. Boca Raton, FL: Taylor & Francis; 2006.
- 6. Leutholtz BC, Ripoll I. *Exercise and Disease Management*. Boca Raton, FL: CRC Press; 2011.
- 7. Watson RR, Eisinger M. Exercise and Disease. Boca Raton, FL: CRC Press; 1992.
- 8. Rippe JM. Lifestyle Medicine. London: Blackwell Science; 1999.
- 9. McPhee SJ, Papadakis M, Rabow MW. *Current Medical Diagnosis and Treatment 2011*. New York, NY: McGraw-Hill Companies; 2011.
- 10. Fauci AS, Braunwald E, Kasper D, Hauser S, Longo DL. *Harrison's Manual of Medicine*. New York, NY: McGraw-Hill Professional; 2009.
- 11. Goldman L, Schafer AJ. *Goldman's Cecil Medicine*. Philadelphia, PA: Saunders; 2012.
- 12. Mayoclinic. 2011. http://www.mayoclinic.com/health/heart-disease/DS01120. Accessed 14 February 2011.
- WebMD. 2011. http://www.webmd.com/heart-disease/guide/heart-disease-symptomstypes. Accessed 14 February 2011.
- 14. Merriam Webster. 2011. http://www.merriam-webster.com/medlineplus/disease. Accessed 14 February 2011.
- 15. Wikipedia. 2011. http://en.wikipedia.org/wiki/Disease. Accessed 14 February 2011.
- 16. MedicineNet. 2011. http://www.medterms.com/script/main/art.asp?articlekey=3011. Accessed 14 February 2011
- 17. Lloyd-Jones D, Adams R, Carnethon M, De Simone G, Ferguson TB, Flegal K, Ford E, Furie K, Go A, Greenlund K. Heart disease and stroke statistics—2009 update: A report

from the American Heart Association Statistics Committee and Stroke Statistics Subcommittee. *Circulation*. 2009;119:e21–e181.

- Stary H, Blankenhorn D, Chandler AB, Glagov S, Insull Jr W, Richardson M, Rosenfeld M, Schaffer S, Schwartz C, Wagner W. A definition of the intima of human arteries and of its atherosclerosis-prone regions. A report from the committee on vascular lesions of the council on arteriosclerosis, American Heart Association. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1992;12:120.
- Stary HC, Chandler AB, Dinsmore RE, Fuster V, Glagov S, Insull Jr W, Rosenfeld ME, Schwartz CJ, Wagner WD, Wissler RW. A definition of advanced types of atherosclerotic lesions and a histological classification of atherosclerosis: A report from the committee on vascular lesions of the council on arteriosclerosis, American Heart Association. *Circulation*. 1995;92:1355.
- 20. Stary HC, Chandler AB, Glagov S, Guyton JR, Insull Jr W, Rosenfeld ME, Schaffer SA, Schwartz CJ, Wagner WD, Wissler RW. A definition of initial, fatty streak, and intermediate lesions of atherosclerosis. A report from the committee on vascular lesions of the council on arteriosclerosis, American Heart Association. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1994;14:840.
- 21. Ross R. The pathogenesis of atherosclerosis: A perspective for the 1990s. *Nature*. 1993;362:801–809.
- 22. Munro J, Cotran R. The pathogenesis of atherosclerosis: Atherogenesis and inflammation. *Laboratory Investigation: A Journal of Technical Methods and Pathology*. 1988;58:249.
- 23. Wissler RW. Update on the pathogenesis of atherosclerosis 1. American Journal of Medicine. 1991;91:S3–S9.
- Schwartz C, Valente A, Sprague EA, Kelley J, Nerem R. The pathogenesis of atherosclerosis: An overview. *Clinical Cardiology*. 1991;14:1–16.
- DiCorleto P. Cellular mechanisms of atherogenesis. American Journal of Hypertension. 1993;6:314S.
- Berliner JA, Navab M, Fogelman AM, Frank JS, Demer LL, Edwards PA, Watson AD, Lusis AJ. Atherosclerosis: Basic mechanisms: Oxidation, inflammation, and genetics. *Circulation*. 1995;91:2488.
- 27. Rosenfeld ME. Cellular mechanisms in the development of atherosclerosis 1. *Diabetes Research and Clinical Practice*. 1996;30:S1–S11.
- 28. Blum A, Miller H. The role of inflammation in atherosclerosis. *Israel Journal of Medical Sciences*. 1996;32:1059.
- 29. Kiechl S, Willeit J. The natural course of atherosclerosis: Part II: Vascular remodeling. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1999;19:1491.
- Williams CS, Mann M, DuBois RN. The role of cyclooxygenases in inflammation, cancer, and development. *Oncogene*. 1999;18:7908.
- 31. Coussens LM, Werb Z. Inflammation and cancer. Nature. 2002;420:860-867.
- 32. Ardies CM. Inflammation as cause for scar cancers of the lung. *Integrative Cancer Therapies*. 2003;2:238–246.
- Marx J. Inflammation and cancer: The link grows stronger. Science(Washington). 2004;306:966–968.
- 34. Pikarsky E, Porat RM, Stein I, Abramovitch R, Amit S, Kasem S, Gutkovich-Pyest E, Urieli-Shoval S, Galun E, Ben-Neriah Y. NF-κβ functions as a tumour promoter in inflammation-associated cancer. *Nature*. 2004;431:461–465.
- 35. Aggarwal BB, Shishodia S, Sandur SK, Pandey MK, Sethi G. Inflammation and cancer: How hot is the link? *Biochemical Pharmacology*. 2006;72:1605–1621.
- 36. McGeer PL, McGeer EG. Inflammation and neurodegeneration in Parkinson's disease. *Parkinsonism and Related Disorders*. 2004;10:S3–S7.

- Perry VH. The influence of systemic inflammation on inflammation in the brain: Implications for chronic neurodegenerative disease. *Brain, Behavior, and Immunity*. 2004;18:407–413.
- Hald A, Lotharius J. Oxidative stress and inflammation in Parkinson's disease: Is there
 a causal link? *Experimental Neurology*. 2005;193:279–290.
- 39. Whitton P. Inflammation as a causative factor in the aetiology of Parkinson's disease. *British Journal of Pharmacology*. 2007;150:963–976.
- Rogers J, Webster S, Lue LF, Brachova L, Harold Civin W. Inflammation and Alzheimer's disease pathogenesis. *Neurobiology of Aging*. 1996;17:681–686.
- 41. Akiyama H, Barger S, Barnum S, Bradt B, Bauer J, Cole GM, Cooper NR, Eikelenboom P, Emmerling M, Fiebich BL. Inflammation and Alzheimer's disease. *Neurobiology of Aging*. 2000;21:383–421.
- 42. Lim G, Yang F, Chu T, Chen P, Beech W, Teter B, Tran T, Ubeda O, Ashe KH, Frautschy S. Ibuprofen suppresses plaque pathology and inflammation in a mouse model for Alzheimer's disease. *Journal of Neuroscience*. 2000;20:5709.
- 43. Sugiyama T. Involvement of interleukin-6 and prostaglandin e 2 in periarticular osteoporosis of postmenopausal women with rheumatoid arthritis. *Journal of Bone and Mineral Metabolism.* 2001;19:89–96.
- Ginaldi L, Di Benedetto MC, De Martinis M. Osteoporosis, inflammation and ageing. *Immunity and Ageing*. 2005;2:14–18.
- 45. Mundy GR. Osteoporosis and inflammation. Nutrition Reviews. 2007;65:S147-S151.
- 46. McCormick RK. Osteoporosis: Integrating biomarkers and other diagnostic correlates into the management of bone fragility. *Alternative Medicine Review*. 2007;12:113.
- 47. Ford ES, Giles WH, Dietz WH. Prevalence of the metabolic syndrome among US adults. *Journal of the American Medical Association*. 2002;287:356.
- Brunner E, Hemingway H, Walker B, Page M, Clarke P, Juneja M, Shipley M, Kumari M, Andrew R, Seckl J. Adrenocortical, autonomic, and inflammatory causes of the metabolic syndrome: Nested case-control study. *Circulation*. 2002;106:2659.
- 49. Furukawa S, Fujita T, Shimabukuro M, Iwaki M, Yamada Y, Nakajima Y, Nakayama O, Makishima M, Matsuda M, Shimomura I. Increased oxidative stress in obesity and its impact on metabolic syndrome. *Journal of Clinical Investigation*. 2004;114:1752–1761.
- 50. Pickup JC. Inflammation and activated innate immunity in the pathogenesis of type 2 diabetes. *Diabetes Care*. 2004;27:813.
- 51. Wellen KE, Hotamisligil GS. Inflammation, stress, and diabetes. *Journal of Clinical Investigation*. 2005;115:1111–1119.
- 52. Eizirik DL, Colli ML, Ortis F. The role of inflammation in insulitis and cell loss in type 1 diabetes. *Nature Reviews Endocrinology*. 2009;5:219–226.
- 53. Koulmanda M, Strom TB. T-cell-directed treatment strategies for type 1 diabetes and the confounding role of inflammation. *Immunotherapy*. 2010;2:431.
- 54. Dandona P, Aljada A, Chaudhuri A, Mohanty P, Garg R. Metabolic syndrome: A comprehensive perspective based on interactions between obesity, diabetes, and inflammation. *Circulation*. 2005;111:1448.
- 55. Lund Håheim L, Wisløff T, Holme I, Nafstad P. Metabolic syndrome predicts prostate cancer in a cohort of middle-aged Norwegian men followed for 27 years. *American Journal of Epidemiology*. 2006;164:769.
- Rana J, Nieuwdorp M, Jukema J, Kastelein J. Cardiovascular metabolic syndrome— An interplay of obesity, inflammation, diabetes and coronary heart disease. *Diabetes, Obesity and Metabolism*. 2007;9:218–232.
- 57. Vona Davis L, Howard McNatt M, Rose D. Adiposity, type 2 diabetes and the metabolic syndrome in breast cancer. *Obesity Reviews*. 2007;8:395–408.
- 58. Lee J, Cho SI, Park H. Metabolic syndrome and cancer-related mortality among Korean men and women. *Annals of Oncology*. 2010;21:640.

- 59. McCann J. Obesity, cancer links prompt new recommendations. *Journal of the National Cancer Institute*. 2001;93:901.
- 60. Pan SY, Johnson KC, Ugnat AM, Wen SW, Mao Y. Association of obesity and cancer risk in Canada. *American Journal of Epidemiology*. 2004;159:259.
- Calle EE, Kaaks R. Overweight, obesity and cancer: Epidemiological evidence and proposed mechanisms. *Nature Reviews Cancer*. 2004;4:579–591.
- 62. Profenno LA, Porsteinsson AP, Faraone SV. Meta-analysis of Alzheimer's disease risk with obesity, diabetes, and related disorders. *Biological Psychiatry*. 2010;67: 505–512.
- 63. Yaffe K. Metabolic syndrome and cognitive disorders: Is the sum greater than its parts? *Alzheimer Disease and Associated Disorders*. 2007;21:167.
- Keller JN, Zhang L, Bruce-Keller AJ, Dasuri K, Nguyen AT, Liu Y. Diet-induced metabolic disturbances as modulators of brain homeostasis. *Biochimica Et Biophysica Acta*. 2009;1792(5):417–422.
- 65. Luchsinger JA, Noble JM, Scarmeas N. Diet and Alzheimer's disease. *Current Neurology* and Neuroscience Reports. 2007;7:366–372.
- 66. Von Muhlen D, Safii S, Jassal S, Svartberg J, Barrett-Connor E. Associations between the metabolic syndrome and bone health in older men and women: The Rancho Bernardo study. *Osteoporosis International*. 2007;18:1337–1344.
- Yaturu S, Humphrey S, Landry C, Jain S. Decreased bone mineral density in men with metabolic syndrome alone and with type 2 diabetes. *Medical Science Monitor: International Medical Journal of Experimental and Clinical Research*. 2009;15:CR5.
- Lee DM, Rutter MK, O'Neill TW, Boonen S, Vanderschueren D, Bouillon R, Bartfai G, Casanueva FF, Finn JD, Forti G. Vitamin D, parathyroid hormone and the metabolic syndrome in middle-aged and older European men. *European Journal of Endocrinology*. 2009;161:947.
- 69. Fletcher GF, Balady G, Blair SN, Blumenthal J, Caspersen C, Chaitman B, Epstein S, Froelicher ESS, Froelicher VF, Pina IL. Statement on exercise: Benefits and recommendations for physical activity programs for all Americans: A statement for health professionals by the committee on exercise and cardiac rehabilitation of the council on clinical cardiology, American Heart Association. *Circulation*. 1996;94:857.
- Strong K, Mathers C, Leeder S, Beaglehole R. Preventing chronic diseases: How many lives can we save? *Lancet*. 2005;366:1578–1582.
- 71. Colditz GA. Economic costs of obesity and inactivity. *Medicine and Science in Sports and Exercise*. 1999;31:S663.
- 72. Katzmarzyk PT, Gledhill N, Shephard RJ. The economic burden of physical inactivity in Canada. *Canadian Medical Association Journal*. 2000;163:1435.
- Slentz CA, Houmard JA, Kraus WE. Modest exercise prevents the progressive disease associated with physical inactivity. *Exercise and Sport Sciences Reviews*. 2007;35:18.
- 74. Leesa SJ, Booth FW. Physical inactivity is a disease. *World Review of Nutrition and Dietetics*. 2005;95:73–79.
- Biesalski HK, Dragsted LO, Elmadfa I, Grossklaus R, Muller M, Schrenk D, Walter P, Weber P. Bioactive compounds: Definition and assessment of activity. *Nutrition*. 2009;25:1202–1205.
- Wahlqvist ML, Wattanapenpaiboon N. Phytochemical malabsorption: Clinical significance. World Journal of Gastroenterology. 1998;4:469–470.
- 77. Liu RH. Potential synergy of phytochemicals in cancer prevention: Mechanism of action. *Journal of Nutrition*. 2004;134:3479S.
- Ames BN, Shigenaga MK, Hagen TM. Oxidants, antioxidants, and the degenerative diseases of aging. *Proceedings of the National Academy of Sciences*. 1993;90:7915.
- Dembinska-Kiec A, Mykkänen O, Kiec-Wilk B, Mykkänen H. Antioxidant phytochemicals against type 2 diabetes. *British Journal of Nutrition*. 2008;99:109–117.

- Scalbert A, Johnson IT, Saltmarsh M. Polyphenols: Antioxidants and beyond. *American Journal of Clinical Nutrition*. 2005;81:215S.
- 81. Byers T, Perry G. Dietary carotenes, vitamin C, and vitamin E as protective antioxidants in human cancers. *Annual Review of Nutrition*. 1992;12:139–159.
- Liu RH. Health benefits of fruit and vegetables are from additive and synergistic combinations of phytochemicals. *American Journal of Clinical Nutrition*. 2003;78:517S.
- Carluccio MA, Siculella L, Ancora MA, Massaro M, Scoditti E, Storelli C, Visioli F, Distante A, De Caterina R. Olive oil and red wine antioxidant polyphenols inhibit endothelial activation. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2003;23:622–629.
- 84. Martin D, Rojo AI, Salinas M, Diaz R, Gallardo G, Alam J, de Galarreta CMR, Cuadrado A. Regulation of heme oxygenase-1 expression through the phosphatidylinositol 3-kinase/akt pathway and the Nrf2 transcription factor in response to the antioxidant phytochemical carnosol. *Journal of Biological Chemistry*. 2004;279:8919.
- Kundu JK, Surh YJ. Molecular basis of chemoprevention with dietary phytochemicals: Redox-regulated transcription factors as relevant targets. *Phytochemistry Reviews*. 2009;8:333–347.
- 86. Akimoto T, Pohnert SC, Li P, Zhang M, Gumbs C, Rosenberg PB, Williams RS, Yan Z. Exercise stimulates Pgc-1α transcription in skeletal muscle through activation of the p38 MAPK pathway. *Journal of Biological Chemistry*. 2005;280:19587.
- MacLean PS, Zheng D, Jones JP, Olson AL, Dohm GL. Exercise-induced transcription of the muscle glucose transporter (GLUT 4) gene. *Biochemical and Biophysical Research Communications*. 2002;292:409–414.
- Russell AP, Hesselink MKC, Lo SK, Schrauwen P. Regulation of metabolic transcriptional co-activators and transcription factors with acute exercise. *FASEB Journal*. 2005;19:986.
- Tunstall RJ, Mehan KA, Wadley GD, Collier GR, Bonen A, Hargreaves M, Cameron-Smith D. Exercise training increases lipid metabolism gene expression in human skeletal muscle. *American Journal of Physiology-Endocrinology and Metabolism*. 2002;283:E66.
- McGee SL, Howlett KF, Starkie RL, Cameron-Smith D, Kemp BE, Hargreaves M. Exercise increases nuclear AMPK 2 in human skeletal muscle. *Diabetes*. 2003;52:926.
- 91. Ruderman N, Park H, Kaushik V, Dean D, Constant S, Prentki M, Saha A. AMPK as a metabolic switch in rat muscle, liver and adipose tissue after exercise. *Acta Physiologica Scandinavica*. 2003;178:435–442.
- 92. Salo DC, Donovan CM, Davies KJA. Hsp70 and other possible heat shock or oxidative stress proteins are induced in skeletal muscle, heart, and liver during exercise 1. *Free Radical Biology and Medicine*. 1991;11:239–246.
- 93. Hatao H, Oh-ishi S, Itoh M, Leeuwenburgh C, Ohno H, Ookawara T, Kishi K, Yagyu H, Nakamura H, Matsuoka T. Effects of acute exercise on lung antioxidant enzymes in young and old rats. *Mechanisms of Ageing and Development*. 2006;127:384–390.
- Cotman CW, Berchtold NC. Exercise: A behavioral intervention to enhance brain health and plasticity. *TRENDS in Neurosciences*. 2002;25:295–301.
- Gómez-Pinilla F, Ying Z, Roy RR, Molteni R, Edgerton VR. Voluntary exercise induces a BDNF-mediated mechanism that promotes neuroplasticity. *Journal of Neurophysiology*. 2002;88:2187.
- Molteni R, Ying Z, Gómez Pinilla F. Differential effects of acute and chronic exercise on plasticity related genes in the rat hippocampus revealed by microarray. *European Journal of Neuroscience*. 2002;16:1107–1116.
- Bouchard C. Physical activity, fitness, and health: The model and key concepts. In. C. Bouchard, RJ Shepard, and T. Stephens. *Physical Activity, Fitness and Health International Proceedings and Consensus Statement*. 1994:11–23.

- Courneya KS, Friedenreich CM. Framework peace: An organizational model for examining physical exercise across the cancer experience. *Annals of Behavioral Medicine*. 2001;23:263–272.
- 99. Goldstein B, Gibson J, Henderson R, Hobbie J, Landrigan P, Mattison D, Perera E, Sübergeld E, Wogan G. Biological markers in environmental health research. *Environmental Health Perspectives*. 1987;74:3–9.
- 100. Valavanidis A, Vlachogianni T, Fiotakis C. 8-Hydroxy-2'-deoxyguanosine(8-OHdG): A critical biomarker of oxidative stress and carcinogenesis. *Journal of Environmental Science and Health, Part C: Environmental Carcinogenesis and Ecotoxicology Reviews*. 2009;27:120–139.
- Helge JW, Stallknecht B, Pedersen BK, Galbo H, Kiens B, Richter EA. The effect of graded exercise on IL-6 release and glucose uptake in human skeletal muscle. *Journal* of *Physiology*. 2003;546:299–305.
- 102. Petersen AMW, Pedersen BK. The anti-inflammatory effect of exercise. *Journal of Applied Physiology*. 2005;98:1154.
- 103. Hollander J, Fiebig R, Gore M, Ookawara T, Ohno H, Ji L. Superoxide dismutase gene expression is activated by a single bout of exercise in rat skeletal muscle. *Pflügers Archiv European Journal of Physiology*. 2001;442:426–434.
- 104. Rundle A. Molecular epidemiology of physical activity and cancer. *Cancer Epidemiology Biomarkers and Prevention*. 2005;14:227.
- 105. Blair SN, LaMonte MJ, Nichaman MZ. The evolution of physical activity recommendations: How much is enough? *American Journal of Clinical Nutrition*. 2004;79:913S.
- 106. Brooks GA, Butte NF, Rand WM, Flatt JP, Caballero B. Chronicle of the Institute of Medicine physical activity recommendation: How a physical activity recommendation came to be among dietary recommendations. *American Journal of Clinical Nutrition*. 2004;79:921S.
- 107. Physical Activity Guidelines Committee. Physical activity guidelines advisory committee report, 2008. US DHHS, 2008.
- 108. Pate RR, Yancey AK, Kraus WE. The 2008 physical activity guidelines for Americans: Implications for clinical and public health practice. *American Journal of Lifestyle Medicine*. 2010;4:209.
- 109. Harris CC. P53 tumor suppressor gene: At the crossroads of molecular carcinogenesis, molecular epidemiology, and cancer risk assessment. *Environmental Health Perspectives*. 1996;104:435.
- 110. Hussain SP, Harris CC. Molecular epidemiology of human cancer: Contribution of mutation spectra studies of tumor suppressor genes. *Cancer Research*. 1998;58:4023.
- 111. Perera FP. Molecular epidemiology: Insights into cancer susceptibility, risk assessment, and prevention. *Journal of the National Cancer Institute*. 1996;88:496.

2 Inflammation

C. Murray Ardies

CONTENTS

2.1	Introd	luction		
2.2	Synthesis of Proinflammatory Molecules		17	
	2.2.1	Pathogen-Associated Molecular Patterns, Damage-Associated		
		Molecular Patterns, and Cytokines	18	
	2.2.2	Prostaglandins, Leukotrienes, and Thromboxanes	23	
2.3	Cellular Responses to Proinflammatory Molecules		27	
	2.3.1	Vascular Endothelial Cells, Vascular Smooth Muscle Cells,		
		Neutrophils, and Monocytes	27	
	2.3.2	Platelets, Fibroblasts, and Macrophages: Blood Clots	33	
	2.3.3	Wound Healing	35	
	2.3.4	Resolution of Inflammation	36	
	2.3.5	Adaptive Immunity and Inflammation	41	
2.4	Contri	butions of Inflammatory Processes to Chronic Disease	43	
	2.4.1	Reactive Molecular Species	44	
	2.4.2	Reactive Molecular Species and Cellular Signaling	45	
	2.4.3	Inflammation-Associated Cytokines and Growth Factors and		
		Cellular Signaling	48	
2.5	Attenu	ation of Inflammatory Processes through Diet	49	
	2.5.1	General Nutrition	49	
	2.5.2	Cellular Signaling	51	
		2.5.2.1 Linoleic Acid, Linolenic Acid, Arachidonic Acid,		
		Eicosapentaenoic Acid, and Docosahexaenoic Acid	52	
		2.5.2.2 Proinflammatory Signal Transduction	55	
2.6	Attenu	ation of Inflammatory Processes through Exercise	65	
2.7	Summary and Recommendations		70	
	2.7.1	Inflammation	70	
	2.7.2	Prevention of Inflammation	71	
	2.7.3	Dietary Recommendations	73	
	2.7.4	Exercise Recommendations	75	
Ack	nowledg	gment	75	
Refe	rences.	-	75	

2.1 INTRODUCTION

From the standpoint of health and chronic diseases, inflammation certainly appears to be the scourge of humankind. Inflammation is not, however, considered to be a disease. This is in spite of the fact that a variety of processes that are associated with inflammation are fundamental to the cause of many chronic diseases,¹⁻¹⁵ including those detailed in this book. Traditional immunology views inflammation as a component of the innate immune system and a local response to the entry of infectious agents into a tissue and to cellular damage resulting from the infection.¹⁶⁻¹⁹ Inflammation comprises a well-defined set of cellular responses that are designed to eliminate the damaging pathogen and initiate repair of the damage caused by the pathogen. The responses are (usually) localized to the specific tissue involved in the infection/damage, and while it typically occurs as a result of some type of infection the same array of inflammatory processes can be initiated by cellular damage that is independent of infection, which is called sterile inflammation. We put this in context with the initiation component of our definition of disease: a disease is a complex process that is initiated by detrimental chemical, biochemical, or cellular (endogenous or exogenous) processes, either alone or in some combination thereof, that if prolonged for long enough or are severe enough, will result in some form of cellular damage, cellular dysfunction, or cell death. Inflammation is a response to those detrimental processes that initiate disease, and it is started at the same time as those initiating events. In this context, inflammatory responses comprise what might be termed a precautionary response, a set of responses designed to limit the cause of impending damage and to "ramp up" repair processes on first appearance of that cause.

As an immune function issue, the inflammatory response is necessary for initiating a series of events that leads to removal of the infectious agent and repair of the damage. Removal of the infectious agent means killing the infectious agent, and the killing processes are a component of innate as well as adapted immune responses. Although immune responses are necessary to kill off infectious agents to avoid cellular death, the killing function can also have negative effects on unaffected cells near the site of infection. In the same context, repair processes that are initiated by an inflammatory response, processes that are designed to promote the repair of cells damaged (or replacement of those killed) by the infection, can also have negative effects on other cells.^{16–19} Thus, the inflammatory response is a double-edged sword: if an appropriate inflammatory response occurs, then the damaging agent is removed and the damage is repaired. On the other hand, if the inflammatory response is too aggressive, or insufficiently aggressive and continues too long, then additional problems are created, problems that include increased risks for a variety of chronic diseases.

Inflammatory responses to infectious pathogens and to damage from noninfectious sources contribute to the development of chronic diseases through a variety of complex interactions. The contribution of inflammatory responses to atherosclerosis illustrates this nicely and might be summarized briefly as follows: inflammationassociated responses to vascular stresses due to the physical characteristics of blood flow in the coronary arteries are mechanistically involved in the initiation of atherosclerosis, and a variety of molecules have been implicated in enhancing the responses, molecules including oxidized lipoproteins, oxidized cholesterol, homocysteine, and damaging reactive oxygen species (ROSs) from various exogenous and endogenous sources.^{20–26} In some cases, inflammation-related processes that are produced in conjunction with various infections (herpesvirus, *Chlamydia pneumonia, Porphyromonas gingivalis, Helicobacter pylori*, and hepatitis A) have also been implicated as contributing factors to atherosclerosis with both innate and adaptive responses to the infectious agents being involved in the production of atherosclerotic plaque.^{20,21,27–34} As a result of a plethora of research studies in this particular area, the proximate cause of atherosclerosis is now known to be a complex interplay of local inflammation-associated responses (see Chapter 4 for details) and not disordered lipids in the blood or cholesterol (although they are among the many factors that can influence the disease process, they are not the cause of the disease process).

From the standpoint of prevention, inflammatory processes are responsive to a variety of dietary components as well as to exercise. Further, because inflammatory processes are a component of cause for a large number of chronic diseases, attenuating inflammatory processes should reduce risk for these diseases. The ω -3 fatty acids from fish oils, for example, predominantly eicosapentaenoic acid (EPA) and docosahexaenoic acid (DHA), comprise one well-known dietary component that has antiinflammatory properties. These ω-3 fatty acids are commonly known to compete with arachidonic acid (AA) for incorporation into membrane phospholipids and, as a substrate for cyclooxygenase (COX)-1 and COX-2, lead to a reduced production of prostaglandin E₂ (PGE₂) and leukotriene B₄ (LTB₄) and an enhanced production of LTB₅.^{35,36} These alterations in inflammation signaling reduce the adhesion of peripheral blood leukocytes to endothelial cells and diminish inflammatory responses. In a variety of epidemiology-based studies, consumption of these ω -3 fatty acids from marine sources appears to be strongly associated with lowered markers of inflammation in serum³⁷⁻⁴² as well as lowered risks for a variety of chronic diseases, including atherosclerosis, 38,43-49 Alzheimer's, 50-52 various cancers, 53-56 and osteoporosis. 57-59 Other dietary components that appear to have anti-inflammatory effects include olive oil; red wine; green and black teas; and a variety of whole grains, fruits, nuts, and vegetables.^{38,60–67} In addition to these dietary components, physical exercise is also well known for both promoting anti-inflammatory effects⁶⁸⁻⁷⁵ and reducing risk for a variety of inflammation-associated diseases, including various cancers, atherosclerosis, type 2 diabetes, obesity, Alzheimer's, and osteoporosis.^{76–92}

In order to see how inflammatory responses can contribute to chronic diseases and how dietary components and exercise can contribute to prevention, a detailed understanding of the molecular aspects of inflammatory signaling is required. For the purposes of this discussion, the innate immune responses of inflammation are emphasized; aspects of adaptive immunity are discussed only briefly to provide context. For additional details relating to the aspects of immune function not covered here, many excellent texts and reviews are available.^{16–19,93–102}

2.2 SYNTHESIS OF PROINFLAMMATORY MOLECULES

As previously mentioned, the overall purpose of inflammatory response appears to be to initiate a variety of cellular responses that lead to removal of the damaging agent and repair of any damage. As described in detail here, a variety of proinflammatory signaling molecules are produced by an array of cells in order to initiate and regulate the different processes that comprise an inflammatory response. Once the damaging agent is removed and the tissue is repaired, the production of proinflammatory signaling molecules stops and the inflammatory response disappears.^{16–18}

In general, an acute inflammatory response to infectious agents typically results in an array of responses that are localized to the area of infection, including the following:

- Vasodilation (and increased permeability of capillaries) to increase the flow of neutrophils, monocytes, and plasma proteins (antibodies, complement, and various acute-phase proteins) into the tissue vasculature; the activation and attraction of any local tissue macrophages and dendritic cells in the immediate area to kill the infectious agents as well as the attraction (and activation) of circulating neutrophils and monocytes to infiltrate the tissue and kill the pathogens
- 2. Activation of platelets to enhance clotting in order to close damaged areas of blood vessels and the attraction of fibroblasts into the area to enhance the healing process and synthesize scar tissue to permanently close wounds
- 3. The release of various growth factors from a variety of cells to increase rates of synthesis of a variety of cellular components for repair and rates of cell division and cell growth in order to replace cells that were killed by the infectious agent
- 4. Initiation of various cellular functions that lead to an adaptive immune response resulting in far more efficient killing of the pathogen

2.2.1 PATHOGEN-ASSOCIATED MOLECULAR PATTERNS, DAMAGE-ASSOCIATED MOLECULAR PATTERNS, AND CYTOKINES

Each of the four responses that are localized to the site of infection is initiated by an array of signaling molecules produced by cells that have receptors for various components of infectious agents and by various cells that have receptors for one or another of the signaling molecules that are produced, molecules that include a variety of cyto-kines, predominantly tumor necrosis factor (TNF)- α , interferon (IFN)- β , interleukin (IL)-1 β (IL-1 β), and IL-6; prostaglandins (PGs), leukotrienes (LTs), and thromboxanes (TXs); histamines; chemokines, predominantly IL-8, monocyte chemoattractant molecule (MCP)-2, and macrophage inflammatory protein (MIP)-2; and adhesion molecules, predominantly P-selectin, E-selectin, intracellular adhesion molecules (ICAMs), and vascular cell adhesion molecules (VCAMs). In addition to the infectious agents themselves, cellular components from damaged or necrotic cells can also initiate the same array of inflammatory signaling through similar receptor-mediated processes by essentially the same cells involved in the pathogen-stimulated inflammatory response.^{99,101,103} These damage-associated inflammatory responses are also important to understand how inflammation can play a role in a variety of chronic diseases.

The major cells that are capable of directly responding to components of infectious agents or cellular components released by damaged or dying cells include dendritic cells (found in skin, lymphoid tissue, mucosal epithelium, and organ parenchyma), natural killer (NK) cells (found mainly in blood, spleen, and liver), neutrophils, tissue macrophages, mast cells (in most tissues, with higher amounts in skin, airway, and intestinal

mucosa), as well as endothelial cells and fibroblasts.^{93,99,103–105} More specifically, these cells respond when specific cellular receptors bind to various components from pathogens, components known as pathogen-associated molecular patterns (PAMPs). Toll-like receptors (TLRs: TLR1, TLR2, TLRs4-6) on the plasma membrane bind bacterial lipopolysaccharides (LPSs), bacterial peptidoglycans, bacterial flagellin, envelope proteins from viruses, and hemagglutinin protein from viruses, whereas TLR3, and TLRs 7–9, in the intracellular endosome membranes and endoplasmic reticulum, bind a variety of pathogen-associated nucleotides (such as ssRNA, dsRNA, and CpG DNA) as well as endogenous nucleotides. TLRs 2 and 4 also recognize mammalian HSPs and fibrinogen, TLR9 binds mammalian DNA, and TLR3 binds mammalian RNA. These latter compounds are commonly released from cells that have been severely damaged and/ or are dying and are therefore called damage-associated molecular patterns (DAMPs). Figure 2.1 shows a summary of PAMP- and DAMP-mediated cellular responses.

Other pattern recognition receptors (PRRs) include a variety of cytoplasmic nucleotide oligomerization domain (NOD)-like receptors (NLRs)^{103,106} and the cytoplasmic retinoic acid–inducible gene I (RIG-I)-like receptor family, which includes the RIG-I-like receptor, melanoma differentiation–associated gene 5 receptor, and the laboratory of genetics and physiology 2 receptor, collectively referred to as RLRs.^{93,107–109} Similar to TLRs, NLRs and RLRs also bind various PAMPs. The major PAMPs for NLRs include peptidoglycan (PGN) (components of cell walls of all bacteria), bacterial flagellin, bacterial LPS, bacterial RNA, viral RNA, and viral DNA.^{103,106,107} RLRs, on the other hand, appear to be restricted to recognizing viral nucleotides such as dsRNA and uncapped 5'-triphosphate ssRNA.^{108,109} TLRs, NLRs, and RLRs and RLRs that are known, and they are discussed further, although there are others, including the membrane-bound C-type lectin receptors and absence in melanoma 2-like receptors.¹⁰³

When various pathogen or damage-associated particles bind to PRRs, signal transduction pathways are activated, which then lead to the synthesis of proinflammatory cytokines. The major cytokines produced following the activation of TLRs are IL-6, IL-1 β , TNF- α , IFN- β , IL-8, and IP-10, with the individual cytokines that are produced being dependent on the specific TLRs that are activated. Once activated, TLRs will dimerize and then form (various) complexes with adapter molecules. These TLR complexes then activate one or another of four major signaling pathways: nuclear factor κ - β (NF $\kappa\beta$), p38–mitogen-activated protein kinase (p38–MAPK) pathway, JNK–MAPK, and/or IFN regulatory factors. Heterodimers of TLRs, TLR2/1 and TLR2/6, and homodimers of TLRs 4, 5, and 7–9 form complexes with the adapter protein MyD88, whereas both heterodimers and homodimers of TLR4 can bind with the additional adapter protein Mal. TLR4 can also form complexes with the adapter proteins TRAM and TRIF to activate a MyD88-independent pathway, and TLR3 is unique in that it only seems to form a complex with the (MyD88-independent) adapter protein TRIF.^{108,110,111}

Although every detail of the signal transduction pathways is not discussed, TLR– adapter protein complexes mediate the activation of the NF $\kappa\beta$ and MAPK signaling pathways through the activation of TAK1 kinase, which occurs through both the MyD88-dependent and the MyD88-independent pathways.^{103,108,112,113} TAK1 subsequently activates NF $\kappa\beta$ (comprised of the heterodimer p65, p50 complexed to the inhibitory unit I κ B in the cytosol) by phosphorylating I $\kappa\kappa\gamma$, which subsequently



FIGURE 2.1 Activation of proinflammatory signaling by pathogen-associated molecular patterns (PAMPs) and damage-associated molecular patterns (DAMPs): binding of DAMPs and PAMPs to toll-like receptors (TLRs), RLRs, and nucleotide oligomerization domain (NOD)-like receptors (NLRs) triggers the formation of protein complexes (dark, filled arrows), which then activate signal transduction pathways to activate the transcription factors p50 and p65 (nuclear factor κ - β [NF $\kappa\beta$]), IRF3 and IRF7, and others (light, filled arrows). Subsequent to binding to various promoters, the proinflammatory cytokines, chemokines, and adhesion molecules are expressed (open arrows). In addition to the activation of transcription, the p38–MAPK and the JNK–MAPK also activate PLA₂ to initiate the synthesis of proinflammatory eicosanoids.

phosphorylates IκB and frees NFκβ to enter the nucleus. TAK1 can also activate the JNK–MAPK and p38–MAPK pathways by phosphorylating MEK7 and MEKK3. The activated p38–MAPK and JNK–MAPK pathways then activate a variety of transcription factors including AP-1 factors as well as NFκβ, enhancing TAK1-mediated NFκβ activation. NFκβ and the various AP-1 units then enter the nucleus and coactivate the expression of IL-6, pro-IL-1β, TNF-α, IL-8, and IP-10 with the specific cytokines produced being dependent on the array of AP-1 subunits that are coactivated along with NFκβ. The various transcription factors activated by p38–JNK include cJun, cFos, ATF2, JunB, JunD, CREB, cEBPβ, Elk1, and Iκβζ.

The TLR3–TRIF adapter protein complex activates IFN factors 3 and 7 (IRF3, IRF7), which then, along with NF $\kappa\beta$ coactivation, stimulates the expression of IFN- β and IP-10.^{108,110,111} IP-10 synthesis is included here as a component of the proinflammatory signals produced on first contact with pathogens because, even though IP-10 is normally considered to be an IFN-inducible protein, it is instructive that it is actually induced by AP-1 binding.

Because TRL4 binds both MyD88 and TRIF proteins it is capable of inducing the synthesis of IFN as well as the other cytokines, whereas TLR3 appears to be restricted to activating the synthesis of IFN and IP-10. Thus, IFN expression is a relatively modest consequence of the activation of plasma membrane–bound TLRs 1, 2, and 4–6 (by a variety of PAMPs and DAMPs) and a major response following the activation of endosomal TLRs 3 and 7–9.^{101,108,110,111} Of the different inflammatory cells, dendritic cells and macrophages (which happen to be present in almost all tissues) express the greatest array of PRRs and therefore have the ability respond to almost any PAMP or DAMP. The plasmacytoid class of dendritic cells also expresses the endosomal TLRs (TLRs 3 and 7–9) to a far greater degree compared to other inflammatory cells, therefore making this class of dendritic cells a major source of IFN (outside NK and Th cell production of IFN- γ , as discussed in Section 2.3.3) resulting from a viral infection.

NLRs in the cytosol of cells that initiate an inflammatory response activate the expression of the same major proinflammatory cytokines in a manner that is similar to that observed through TLR activation and has one important additional process: the NLRP family of NLRs forms oligomer complexes along with other proteins to form an inflammasome complex. The NLRs NOD1 and NOD2, when activated, will recruit the adapter proteins RICK, TAB1, and caspase recruitment domain 9, which then complexes with TAK1 (among others) to activate NF $\kappa\beta$ and the MAPK pathways. Again, the activation of p38 and JNK leads to the activation of an array of AP-1 subunits and the expression of IL-6, pro-IL-1 β , TNF- α , and IP-10. The inflammasome then attracts and activates the enzyme caspase-1 to cleave an inactivating peptide from pro-IL-1 β in order to produce active IL-1 β , which is then secreted from the cell^{101,106,111} (as are the other signaling molecules).

The third major family of PRRs is the cytosolic RIG-I-like receptor family, the RLRs, which bind dsRNA and uncapped 5'-triphosphate ssRNA (predominantly from RNA viruses). Some of the activated RLRs interact with IPS-1 on the outer membrane of mitochondria to form complexes with TNF receptor–associated death domain (TRADD) proteins, which then attract others to activate two separate pathways. The TRADD/TRAF3/TANK complex activates TBK1/IKKζ to phosphorylate
IRF3 and IRF7 for the subsequent coactivation of IFN expression. Other activated RLRs form complexes with IPS-1 and then with TRADD, Fas-associated death domain, and caspase-8/-10 to activate NF $\kappa\beta$ for coactivation of IFN- β expression.

As described earlier, caspase-mediated activation of NF $\kappa\beta$ via RLRs and activation of IRFs through TLRs also occur, indicating that additive effects are most likely to occur when more than one of the various classes of PRRs is activated. There are other points where the different signal transduction pathways for the various classes of PRRs intersect (e.g., different PRRs activate MAPKs via multiple/overlapping pathways), and this is of course very important as well. This *cross talk* between activation pathways allows for a highly graded inflammatory response that depends on the number of PAMPs associating with plasma membrane TLRs and the number of PAMPs that interact with endosomal TLRs, cytosolic NLRs, and RLRs following phagocytosis of the pathogens (or invasion of the cells by viruses).

The addition of DAMP-mediated activation of PRRs also produces an important source of possible interactions. These latter compounds are a sensor for the amount of damage caused by the pathogens and allows for an even greater inflammatory response, analogous to the concept that a weak pathogen that causes no damage requires a less intense response compared to a highly toxic pathogen that causes extensive damage. It also implies that with sufficient damage to tissues in one location the DAMPs may enter the circulation and activate some PRRs in inflammatory cells far away from the actual damage. Of course, this also holds true for infectious pathogens; there is no reason to expect that a few organisms will not be able to enter the circulation (or lymph) and then activate PRRs at sites far away from the original site of infection. This is especially true with exposures due to inhaling pathogens (such as aflatoxin or influenza virus, or many others) or ingesting them (food poisoning by *Escherichia coli* or *Vibrio cholera*, or many others) where the pathogen load may be much greater than that from a small scratch or a casual exposure and where the pathogens have easy access to a large surface area with a high rate of blood flow in the exposed tissue. The implication of this is that a minimal and highly localized inflammatory response to a very small amount of damage may be exacerbated by the appearance in that location of PAMPs and/or DAMPs that arose from an infection or a larger scale area of damage far away from that site. Interestingly, NF $\kappa\beta$ can also be activated by hydrogen peroxide (H₂O₂)-induced phosphorylation of IkB via C-Src as well as by ataxia telangiectasia mutated (ATM) (ATM is activated in response to DNA damage).103,108,113 The importance of these is that ROS and ROS-mediated DNA damage can also activate NFkB, indicating that factors associated with ROS stress may also be additive to the TLR-, RLR-mediated activation process.

In addition to the synthesis and release of various cytokines in response to PRR activation, as described earlier in this section, cells such as mast cells (as well as the granulocytes basophils and eosinophils) can degranulate and release a variety of preformed compounds from intracellular granules in response to injury as well as through the binding of proteins and other small molecules to the IgE that is bound to the Fc receptors on their cell membranes. The compounds released by mast cells include histamine, several protease enzymes, TNF, heparin, vascular endothelial growth factor (VEGF), and basic fibroblast growth factor (bFGF). In response to PRRs, damage, and allergens, activated mast cells also synthesize an array of cytokines (including IL-3, IL-4, IL-5, IL-6, IL-9, and IL-13). In addition to these

released compounds, several of the AA-derived proinflammatory signaling molecules are produced as well: platelet-activating factor (PAF), prostaglandin D_2 (PGD₂), PGE₂, leukotriene C₄ (LTC₄), and LTB₄.^{8,105,114–116}

2.2.2 **PROSTAGLANDINS, LEUKOTRIENES, AND THROMBOXANES**

The production of the AA-derived LTs, PGs, and TXs (as well as PAF) is not limited to mast cells. Most cells of the body can produce one or another of these important proinflammatory compounds. In general, the production of these is determined predominantly by the amount of AA released into the cell from phospholipids that form membranes of the endoplasmic reticulum, nucleus, and Golgi apparatus. The AA is cleaved from the phospholipids by phospholipase A_2 (cPLA₂), and the production of various lipid mediators is then dependent on the amount of the constitutive COX-1 and inducible COX-2 enzymes; amount of the 5-lipoxygenase (5-LOX) enzyme; and array of specific PG, LT, and TX synthases in the cell,^{117–120} as outlined here and illustrated in Figure 2.2.

Once AA molecules are released from membrane phospholipids (predominantly from phosphatidylcholine), the AA is metabolized into prostaglandin G_2 (PGG₂) by COX-1 and COX-2. The PGG₂ is converted into PGH₂ via a peroxidase reaction (also by COX-1 and COX-2), and the PGH₂ is then converted into a variety of proinflammatory PGs by specific PG synthases. PGE_2 is synthesized by three different PG synthases that are expressed in almost all cells, the cytosolic form cPGES-1 and the two microsomal forms mPGES-1 and mPGES-2, with mPGES-1 being inducible in concert with COX-2.^{119,121} PGI₂ (prostacyclin) is synthesized by an isozyme of cytochrome P450 known as prostacyclin synthase, which is predominantly expressed in vascular endothelial cells.^{119,122} PGG₂ is also converted to PGD₂ by either lipocalin-type PGD₂ synthase (l-PGDS), found mainly in brain, or hematopoietic-type synthase (h-PGDS), found predominantly in mast cells and activated macrophages, platelets, and dendritic cells.^{119,123,124} Because PGD₂ spontaneously dehydrates to produce PGJ₂,^{119,123} a highaffinity ligand for peroxisome proliferator-activated receptor-y (PPARy), this PG may have important implications as an anti-inflammatory PG, as discussed in Chapter 3, Section 3.4. Thromboxane A₂ (TXA₂) synthase, a CYP5 enzyme,¹²⁵ utilizes PGG₂ as a substrate to produce TXA₂ and is mainly found in platelets and macrophages.¹¹⁷

AA is also made into 5-hydroperoxyeicosatetraenoic acid and then into leukotriene A_4 (LTA₄) by 5-LOX. LTA₄ is then converted into a variety of other LTs including LTB₄ by LTA₄ hydrolase (LTA₄H), which is expressed in activated neutrophils, mast cells, and macrophages.^{117,119,126} LTC₄ is synthesized by LTC synthase, which is also expressed by neutrophils, macrophages, and mast cells.^{117,119} The cellular responses to these lipid mediators are discussed in Section 2.3, whereas other anti-inflammatory lipid mediators that are produced from AA, EPA, and DHA by 12-lipoxygenase (12-LOX) and 15-lipoxygenase (15-LOX) are discussed in Section 2.3.4.

In terms of regulating the synthesis of PGs, LTs, and TXs it is the availability of AA for the various synthase enzymes that governs the production of these products, and this is determined through the activation of PLA₂, predominantly its cytosolic form (cPLA₂).¹¹⁹ cPLA₂ can be activated by calcium, p38–MAPK and extracellular signal-related kinase (ERK)–MAPK, protein kinase C (PKC), or ROSs^{117,127–130}





and, following activation, $cPLA_2$ binds to the endoplasmic, nuclear, and Golgi membranes. The resulting $cPLA_2$ activity usually peaks within 1 minute and declines 2 to 3 minutes after binding.^{117,127–130}

Overall, PGs, LTs, and TXs are, by themselves, unstable, and they readily inactivate, usually within a few minutes. Therefore, unlike cytokine signaling, signaling by these compounds is a relatively transient and highly localized function and only with continuing activation of cPLA₂ will a sustained production and presence of these lipid mediators occur to initiate various inflammatory responses, conditions that certainly occur with cellular damage. It is interesting to note, however, that both AA and the unstable intermediates of AA metabolism are very stable within membrane environments and that they can be readily transferred from one cell to another during cell–cell contact,¹³¹ indicating that activated cells can transfer both AA and AA metabolites to cells with inactive cPLA₂. The implication of this is that lipid signaling initiated by one activated cell (e.g., endothelial cells) could *spread* the production of these lipid mediators to other cells through direct contact (possibly to a monocyte, neutrophil, or platelet), thus bypassing the normal signal transduction pathways that activate cPLA₂ and enhancing and possibly prolonging the production of PGs, LTs, and TXs in the local region.

The final lipid-derived signaling molecule to be mentioned is PAF, a product that is synthesized predominantly by vascular endothelial cells and activated macrophages and mast cells. PAF is synthesized from lysophosphatidylcholine (lysoPC) through acetylation via acetyl Co-A by the enzyme lysoPC acetyltransferase.^{132,133} LysoPC is the product remaining after cPLA₂ cleaves AA from phosphatidylcholine and, therefore, PAF synthesis is regulated through cPLA₂ activation (see earlier). Functions of PAF are briefly described in Section 2.3.2.

From these discussions, synthesis of the various proinflammatory cytokines and lipid mediators appears to be a highly regulated process. However, when considering that activation of cPLA₂ occurs via calcium-, ROS-, and MAPK-mediated events,^{127–130,134,135} it seems that almost any change in cell function will result in the production of LTs, PGs, and TXs. Therefore, production of these eicosanoid compounds in cells may actually provide an exquisitely sensitive response to any form of cellular stress and inflammation-associated signaling mediated through PAMPs and DAMPs is only part of the reason it occurs. Production of these lipid mediators following normal cellular stresses in otherwise unaffected cells could therefore contribute to a preexisting local production of these same lipid mediators by cells that are responding to actual damage and/or infection.

In addition to the activation mechanisms for cPLA₂, the signal transduction pathways for cytokine production through both TLRs and NLRs also involve MAPK pathway activation. Because ROSs, PKC, and a wide variety of normal cell functions that are activated and/or regulated by G-protein activation processes¹³⁴ can activate MAPK pathways as well, the production of various cytokines may also be sensitive to any alterations in cell function beyond what might be called *resting homeostasis*. In a traditional view of inflammation signaling, PAMPs and DAMPs initiate the process. However, because of a significant amount of cross talk among the various activation pathways for a wide array of cellular processes that are regulated through the activation of the stress-response MAPK pathways,^{134,136} the synthesis of proinflammatory cytokines also may be a component of a cellular response to stress, a response that can occur independent of any actual damage or infection (Figure 2.3). With similar



FIGURE 2.3 Stress-activated proinflammatory signaling: entry of calcium into the cytosol via normal signaling events mediated by diacyl glycerol (DAG) or through reactive oxygen species (ROS)-damaged membranes activates protein kinase C (PKC), which then may activate PLA₂, and the ERK–MAPK and p38–MAPK pathways (dark, filled arrows). At the same time, calcium-activated increases in cellular metabolism subsequently increase the production of ROSs, which may also activate PLA₂ as well as p38–MAPK and JNK–MAPK via Trx–Ask1. ROS-caused single- and double-strand breaks in DNA activate repair by ATM, which also activates nuclear factor κ - β (NF $\kappa\beta$), and ROS-mediated PI3K activity also enhances mTORC1 via Akt to enhance overall rates of protein synthesis (dark, filled arrows). The result of all these effects is the activation of a variety of transcription factors and the subsequent synthesis of a variety of proinflammatory molecules (open arrows) in response to cellular stress.

mechanisms of activation, one would expect LTs, PGs, TXs, and cytokines to be expressed following the same activation events, which, of course, is the case.^{137,138} The production of various lipid mediators and of cytokines by human skeletal muscle as a direct result of repeated muscle contractions and the enhanced rate of metabolism necessary to support the muscle contractions,^{72,73,139,140} and their production by adipocytes,^{141–144} provide excellent examples of this concept. Perhaps, inflammatory signaling should be viewed as a component of a continuum of graded responses to cellular stress and not as a component of the immune system, with a full-blown inflammatory response being simply the result of the additive and synergistic effects of multiple sources of cellular stress that includes the presence of PAMPs and DAMPs.

In addition to the aforementioned proinflammatory products, there are many more cytokines and lipid-derived signaling molecules that play a role in immune functions, products such as IL-4, IL-10, IL-13, IL-15, lipoxins, resolvins, protectins, and maresins, among many others. Many of the cytokine products play different roles in regulating growth and differentiation of various immune cells or have anti-inflammatory effects and have not been discussed due to the proinflammatory focus of this section. IL-10, resolvins, lipoxins, protectins, and possibly maresins actually are anti-inflammatory molecules that play a very important role in mediating the resolution of inflammation.^{145–147} These are discussed in more detail in Section 2.3.

2.3 CELLULAR RESPONSES TO PROINFLAMMATORY MOLECULES

The possible implications of proinflammatory signaling to a variety of disease issues quickly become evident when the cellular responses to proinflammatory cytokines and lipid-derived molecules are considered. A variety of cells (and tissues) are capable of responding to the proinflammatory molecules, and the specific response is governed by the expression in those cells of cellular receptors that bind the various proinflammatory molecules. The cells that initiate the production of proinflammatory molecules, in turn, also express various receptors for some of these same molecules, indicating that proinflammatory signaling molecules can have autocrine as well as paracrine and endocrine functions. Although cells of blood vessels, bone marrow, brain, liver, heart, skeletal muscle, adipose tissue, and lymphocytes all respond to inflammatory signals, the focus of this discussion is on those cells immediately involved in local inflammatory response.

2.3.1 VASCULAR ENDOTHELIAL CELLS, VASCULAR SMOOTH MUSCLE CELLS, NEUTROPHILS, AND MONOCYTES

Vasodilation usually occurs within minutes of the start of a proinflammatory stress response. Of the signaling molecules discussed Section 2.2, some of the lipid-derived mediators are the predominant molecules responsible for this early response because they are synthesized immediately on stimulus, whereas the synthesis of various cyto-kines takes much longer; increased mRNA appears at about 30 minutes and peaks, as does expression, at about 1 hour.¹⁴⁸ Vasodilation is predominantly caused by the nitric oxide (NO[•]), PGI₂, and PGE₂ that are released by vascular endothelial and other inflammatory cells and by histamine that is released mainly by mast cells.

Of these, NO[•] activates soluble guanylyl cyclase to synthesize cGMP while PGI₂ and PGE₂ increase cAMP production in vascular smooth muscle cells through the activation of the G-coupled IP and EP2/EP4 receptors, respectively. The result is the direct activation of protein kinase A by cAMP, activation of protein kinase G (PKG) by cGMP, and cross-activation of PKG by cAMP, resulting in smooth muscle relaxation and vasodilation through the activation of myosin light chain phosphorylase and the opening of large high-conductance calcium-activated potassium channels.^{117,149,150}

Histamine causes increased vasopermeability as well as vasodilation through activating the H1 histamine receptors on the vascular endothelial cell membrane. H1 activation leads to a G-protein-coupled increase in phospholipase C activity, producing inositol triphosphate (IP₃), which subsequently activates calcium channels in the endoplasmic reticulum and cell membrane. The increased cytosolic calcium subsequently activates PKC through calcium-stimulated NO[•] production via eNOS:PKG. The elevated NO[•] contributes to vasodilation,^{150,151} and the elevated calcium also triggers a calmodulin-mediated activation of myosin light chain kinase to induce the constriction of vascular epithelial cells.^{152,153} Enhanced PKC and PKG activities also activate Rho-GTPases, which induce the phosphorylation of VE-cadherin, leading to the disassembly of adherens junctions and tight junctions between endothelial cells and an enhanced permeability of the blood vessels.¹⁵³ Both PAF and VEGF also stimulate enhanced vasopermeability through similar signal transduction pathways.^{154,155} The end result of these vasopermeability and vasodilation effects is an increase in the entry of plasma proteins and circulating inflammatory cells (predominantly neutrophils and monocytes) into the area. The increased vasopermeability ensures that plasma proteins, neutrophils, and monocytes migrate into the tissue much more readily.

Activation of phagocytic cells that are already present in the tissue (as well as neutrophils, mast cells, and dendritic cells that are recruited into the tissue) occurs in order to enhance the destruction of pathogens. The activation process includes a complex interaction of PAMPs and/or DAMPs binding to PRRs as well as cellular responses to proinflammatory molecules and/or plasma proteins (complement)105,156-158 in the immediate vicinity. Macrophages that are already resident in the tissues provide a good model to illustrate this process. Contact with pathogens stimulates an endocytosis response where pathogens or cellular debris are engulfed by a small section of the plasma membrane and internalized with the resulting vesicle fusing with lysosomes to create a phagolysosome. During the phagocytic response, components of necrotic or apoptotic cells and/or pathogens are destroyed by various hydrolytic lysosomal enzymes. The nicotinamide adenine dinucleotide phosphate (NADPH) oxidase enzyme that is associated with the resulting phagolysosome generates a variety of ROSs, superoxide anions that then dismutate to form H_2O_2 and interact with the peroxide to form hydroxyl radicals. The NO' that is produced due to eNOS activity reacts with the superoxide anions and with H₂O₂ to produce hypochlorous acid (HOCl) and a variety of reactive nitrogen species (RNSs); all of these contribute to the destruction process.

Unlike with pathogens, phagocytosis of apoptotic cells does not result in a robust activation of macrophages; this is in part because the cell membranes of apoptotic cells remain intact during the process, including when the cells split apart into smaller apoptotic bodies. This not only minimizes the release of DAMPs into the surrounding areas but also ensures that they are destroyed in the phagolysosome before they can interact with the PRRs.¹⁵⁹ When a few cells die through necrotic processes, the activation and subsequent production of proinflammatory cytokines by local macrophages is minimal simply because of the presence of a limited number of necrotic cells and, therefore, a limited activation of DAMP receptors. The presence of greater numbers of necrotic cells due to a localized pathogenic infection will activate both PAMP and DAMP PRRs, and as a result the production of TNF- α , IFN- β , IL-1 β , and IL-6 will be more robust. IFN- β production by activated macrophages in a small localized inflammatory response is relatively low and probably does not contribute very much to enhanced levels of IFN- β in serum; however, the presence

of IFN receptors on macrophages ensures that any IFN- β production will have autocrine effects. Moderate production of IFN- β by local dendritic cells can also contribute to the activation of IFN receptors on the macrophages without significantly increasing serum levels of IFN- β . In response to IFN, iNOS is expressed at high levels in macrophages and the combination of enhanced ROS, RNS, and hypochlorite production greatly increases the efficiency of the destruction process. The increased production of highly reactive molecular species by activated macrophages (and other phagocytic cells) also has important implications for the development of chronic diseases (discussed in Section 2.4) through damaging and affecting the function of otherwise healthy cells in the immediate vicinity.

In response to proinflammatory signals produced by the PAMP/DAMP, activated tissue cells circulating neutrophils and monocytes can also be activated and stimulated to migrate into the tissues. As already mentioned, the vasodilation and increased vasopermeability response to PGI₂, NO[•], PGE₂, and histamine can facilitate this process. Neutrophils and monocytes are activated to exit the circulation through the coordinated function of a variety of molecules on the vascular endothelial cells as well as the neutrophils and monocytes. In response to TNF- α , either alone or in combination with IL-6 and IL-1, vascular endothelial cells synthesize a variety of chemoattractant molecules including ICAMs, VCAMs, E-selectin and P-selection, MCP-1 (chemokine: CCL2), IL-8 (chemokine: CXCL8), and RANTES (chemokine: CCL5) through NF $\kappa\beta$ and JNK-dependent, as well as JNK-independent, pathways.^{160–170} The binding of TNF- α to its receptor initiates the formation of TRADD–TRAF complexes, which mediate the activation of NF $\kappa\beta$ and both JNK–MAPK and p38–MAPK via TAK1 and apoptosis signal-regulating kinase 1 (ASK1), respectively. Activated IL-1 receptors form complexes with MyD88 and subsequently with IRAK and TRAF to initiate the same TAK1- and ASK1-mediated responses. Activated IL-6 receptors, on the other hand, form complexes with SOS-GRB and with Jak-STAT to activate the transcription factor STAT3 as well as the ERK–MAPK pathway through Ras. In addition, PI3K activity is also activated, which subsequently activates PDK1 and Akt, leading to the activation of mTORC1. These responses lead to the increased activation of various factors (AP-1, STAT3, and others) that are necessary for the coactivation of transcription of various proinflammatory molecules via MAPK and STAT. In addition, through the activation of mTORC1 rates of protein synthesis are enhanced, which leads to an accelerated synthesis of all proinflammatory molecules. Thus, rather than being proinflammatory in and of itself, IL-6 complements the proinflammatory effects of other cytokines (Figure 2.3).

The enhanced expression of the various adhesion molecules, cytokines, and chemokines by vascular endothelial cells then enhances the recruitment of neutrophils and monocytes into the tissue. Following the binding of E-selectin and P-selectin to sialyl-glycoprotein ligands, as well as ICAM and VCAM to the integrin ligands on neutrophil and monocyte plasma membranes, these cells start to roll slowly along the endothelial cell layer of the blood vessel. The MIP-1 and CCR2 receptors of monocytes bind with RANTES and MCP-1, and the MIP-1, CXCR1, and CXCR2 receptors of neutrophils interact with RANTES, MCP-1, and IL-8; both are *guided* to migrate through the endothelial layer at the site of the greatest concentration of these chemokines in order to enter the tissue.

LTB₄, produced by activated macrophages, mast cells, and neutrophils, is a potent chemoattractant molecule for neutrophils.^{117,119} Because AA-derived signaling molecules are produced immediately on the initiation of cellular stress or damage, LTB₄ acts as the first chemoattractant that recruits neutrophils into the area through binding to G-coupled BLT1 receptors on neutrophils, initiating the activation process through increasing cytosolic calcium and the subsequent activation of MAPKs. These first-recruited neutrophils are then activated to produce various cytokines locally, which in turn enhance the recruitment and activation of additional neutrophils and monocytes through the synthesis of various chemokines. LTB₄ thus acts as an initial lipid mediator of a lipid–cytokine–chemokine cascade resulting in what might be called a full-blown inflammatory response.^{119,171,172} With LTB₄ being able to initiate an inflammatory response, it is possible that any prolonged cellular stress that activates PLA₂, even without the involvement of pathogens, can promote a localized inflammatory response as well.

The activated neutrophils that migrate into the tissues at the site of the original inflammatory response phagocytize any unwanted particles and contribute to the killing of pathogens via ROS and RNS production, processes that are enhanced by the IFN- β produced by the (already) activated tissue macrophages and dendritic cells in the immediate area. Activated neutrophils also synthesize and release several cytokines and chemokines, including TNF-α and IL-1β. Macrophages that are activated through the inflammatory process also produce proinflammatory cytokines (TNF- α , IL-1 α/β , and IL-6), as well as performing enhanced phagocytic functions. The end result of these vascular and immune cell responses is an increase in the number of active phagocytic cells in the local area as well as an increase in the local production of proinflammatory cytokines by the newly recruited cells. This has the effect of enhancing the local production of proinflammatory cytokines, prolonging the local inflammatory response, enhancing even more the recruitment of circulating immune cells, as well as greatly increasing the risk for ROS- and RNS-mediated damage to noninvolved cells by increasing the localized production of these reactive species.

The cytokine-mediated activation of selectin and adhesion molecule synthesis in endothelial cells is important because the activation is in part through NF $\kappa\beta$ and JNK–MAPK-dependent as well as JNK-independent pathways. From the previous discussion, there are several possibilities for additive or synergistic cross talk between the various PRR activation pathways, cross talk that may be enhanced through stress-mediated activation of different components of these same pathways.

Because vascular endothelial cells express a variety of TLRs, NLRs, and RLRs,¹⁷³ significant interactions among those activating pathways triggered by direct interaction with PAMPs and DAMPs as well as those triggered by proinflammatory cytokines from both paracrine and autocrine sources can significantly enhance the recruitment and activation of circulating inflammatory cells compared to the effects of PAMPs and DAMPs alone (Figure 2.4). While all cells express TNF-α receptors, IL-6 receptors are expressed predominantly by hepatocytes, monocytes, T cells, and B cells. All cells, however, express gp-130, which complexes with IL-6R to form the active receptor when IL-6 binds. Interestingly, soluble IL-6R binds to gp-130 in any cell to form active IL-6R complexes, indicating that IL-6 (through transactivation events) and TNF- α (through autocrine and paracrine effects) can have effects in vascular endothelial cells.¹⁷⁴ These recruitment effects can be enhanced through the addition of any one of a number of non-pathogen-related stressors as well. Thus, a variety of events that activate various stress-response pathways, starting from an initial activation by PAMPs and DAMPs, can lead to an accelerating production of proinflammatory signals that leads to the recruitment of additional proinflammatory cells that also participate in proinflammatory signaling.

The final molecule to be discussed in the context of inflammation is C-reactive protein (CRP). CRP synthesis in hepatocytes is activated in response to IL-6, with an enhanced effect with coactivation by IL-1B.175,176 Activation of transcription occurs through the binding of a variety of factors to the CRP promoter region, including STAT3, c-rel, and rel p50 (indicating the involvement of NF $\kappa\beta$), although there is negligible transcription activity until binding of CCAAT/enhancer binding protein (C/EBP) occurs, following activation of C/EBP through IL-6 signaling.¹⁷⁷ IL-6 signaling involves direct receptor-mediated activation of the Jak-STAT pathways as well as Ras-mediated activation of MAPK pathways.¹⁷⁸ Although details of the complex interactions of these pathways are not discussed, it is important to recognize that there are a wide variety of stress-related processes and growth factors (as discussed in Section 2.2 for the production of proinflammatory cytokines and PGs, LTs, and TXs) that also activate these same pathways.^{179–181} As a result, average nonstimulated levels of CRP of approximately 0.8 mg/mL (to as high as 10 mg/mL) are observed in healthy volunteers, with acute-phase responses following synthesis of IL-6 and IL-1 by local inflammatory cells being as high as 500 mg/mL.¹⁷⁶

Typically, CRP has been associated with complement activation. Complement refers to a family of proteins that are produced by hepatocytes and circulate freely in the blood. Although details of complement activation and function are not discussed in detail, in the classic activation process the C1q complement protein is activated by binding to the Fc fragment of antibodies, precipitating a cascade of protease-mediated reactions leading to the production of active complement protein complexes and protein products. These activated proteins then greatly enhance a local inflammatory response by acting as a chemoattractant for neutrophils and monocytes and by activating mast cells, endothelial cells, macrophages, and neutrophils (among many other cell types).^{182,183} These effects are especially important in adapted immune function where the binding of antibodies to antigens can lead to a swift activation of a protective inflammatory response that destroys the infectious pathogens.



FIGURE 2.4 Acceleration of proinflammatory signaling through autocrine and paracrine signaling: the initial activation of toll-like receptors (TLRs), RLRs, and nucleotide oligomerization domain (NOD)-like receptors (NLRs) by pathogen-associated molecular patterns (PAMPs) and damage-associated molecular patterns (DAMPs) leads to the synthesis of interleukin (IL)-6, tumor necrosis factor (TNF)- α , IL-1- β , and others (filled arrows). Autocrine binding of TNF- α and IL-1- β to their receptors mediates additional activation of MAPKs as well as p50/p65 (open arrows) via TAK1 and ASK1 to greatly enhance the expression of proinflammatory molecules, which then increases activation and chemotaxis of inflammatory cells into the region. Activation of PI3K/Akt by IL-6 (open arrows) enhances the protein synthesis in general, leading to the activation of p70s6k.

CRP can also initiate complement activation by binding to phosphocholine residues on membranes of damaged and apoptotic cells^{184,185} coupled with cobinding to the C1q complement protein. But in this case where antibodies are not involved, CRP also activates the expression of several factors that inhibit further activation of the complement; therefore, it actually functions to limit the production of a highly inflammatory response caused by the later stages of the processes involved in complement-mediated acceleration of inflammatory responses when it is only DAMPs originating from *normal* necrotic processes that are involved in activating macrophages and dendritic cells. This benefit is important considering the fact that an initial inflammatory response includes vasodilation/vasopermeability responses that increase the entry of circulating complement and antibodies (as well as CRP) into the local area.

From this discussion, CRP may function to ensure that a complement-mediated acceleration of inflammatory responses occurs only as a component of an adapted immune response. The initial activation of C1q, however, leads to the generation of active C3 fragments that are then involved in activating phagocytosis of the CRP-bound cells as well as in enhancing oxidative bursts following phagocytosis.^{175,183} CRP also mediates an increase in the synthesis of ICAMs in endothelial cells as well as the synthesis and release of IL-1, IL-6, and TNF- α ,^{186–188} indicating that early-stage inflammatory responses, including ROS production, can be enhanced by circulating CRP. Thus, as with the other proinflammatory factors already discussed, an inflammatory response in one location that enhances IL-6 and IL-1 levels in serum can mediate and enhance proinflammatory processes in other locations by inducing CRP synthesis in the liver. This is why levels of CRP in the blood can be used as a biomarker of risk for inflammation-associated disease.

2.3.2 PLATELETS, FIBROBLASTS, AND MACROPHAGES: BLOOD CLOTS

In the event of vascular and tissue damage, a process to stop bleeding and permanently close and repair any wounds is necessary to prevent excessive blood loss as well as to return the damaged tissue to normal function. The processes of blood clot formation and wound healing are, of course, initiated by the cells that are involved in the inflammatory process. With respect to the contribution of inflammation to chronic diseases in general, the formation of blood clots and successful healing of damaged tissue may have minimal significance; therefore, all the details of these processes are not described here. Some of the signaling processes associated with platelet activation, blood clot formation, and wound healing are important contributing factors to various aspects of certain chronic diseases and so are discussed here.

The formation of blood clots involves the activation of platelets and the coordinated activation of various plasma proteins called clotting factors via *intrinsic* (common) and *extrinsic* (tissue factor) pathways. Platelets usually activate and adhere to disrupted endothelium by binding to exposed von Willebrand factor and/or collagen via their integrin receptors (GP1b, IX, V, IIb, and IIIa). They can also adhere to endothelial cells of nondisrupted endothelium through the binding of their GP1b integrin receptors to ICAM that is expressed on the endothelial cell membrane in response to inflammatory cytokines.^{189,190} Once activated via activation of phosphatidylinositol 3-kinase and PKC activities coupled with increased cytosolic calcium,^{191,192} platelets degranulate and release a variety of proinflammatory molecules (IL-1 β and TNF- α) and platelet aggregation factors (including fibrinogen, fibronectin, P-selectin, and TXA₂; growth factors (platelet-derived growth factor [PDGF], transforming growth factor β [TGF β], epidermal growth factor, and bFGF); chemokines (RANTES, platelet factor 4, and IL-8, among others); and the coagulation factors V, XI, plasminogen activator inhibitor, plasminogen, and protein S.^{189,190,193} Although the step-by-step formation of clots is not described here, the ability of platelets to bind to endothelial cells and to be activated independently of endothelial disruption is important. The cytokines and chemokines released by activated platelets provide an additional source of signaling molecules for both neutrophil and monocyte recruitment and for the synthesis of adhesive molecules by endothelial cells. To compound this process, platelets can also adhere to activated neutrophils and monocytes that are already rolling on the endothelium, further enhancing the recruitment of additional circulating inflammatory cells¹⁸⁹ without attaching directly to endothelial cells. Platelets attach to both neutrophils and monocytes through the binding of P-selectin (activated platelets) and ICAM-2 (resting platelets) expressed on platelet membranes to P-selectin glycoprotein ligand-1 and lymphocyte function-associated antigen-1 that is expressed on neutrophils and monocytes,¹⁹⁴ providing a vital signal amplification function for the inflammatory process, one that is independent of clot formation. Thus, interactions between platelets, macrophages, and neutrophils can amplify localized inflammatory signaling to enhance the recruitment and activation of more inflammatory cells, resulting in further amplification of the inflammatory signaling, a platelet connection to the lipid-cytokine-chemokine cascade that was described earlier.

Several issues relating to TXA₂ release by both macrophages and activated platelets are important in this platelet-mediated amplification process and deserve further discussion. As previously mentioned, activated macrophages release IL-1 β , IFN- β , and TNF-a. These cytokines, through autocrine stimulation, activate COX-2 synthesis in these same macrophages (and, through paracrine stimulation, in any neutrophils that are rolling along the adjacent endothelial cells).^{195–199} The increase in COX-2 activity in macrophages initially increases their production of TXA₂, which then increases the activation of platelets that may be rolling on the endothelium in the inflamed region. Platelets are activated by TXA₂ through the activation of the G-protein-coupled TP receptor, resulting in the stimulation of phosphatidylinositol 3-kinase and PKC activities and an increase in cytosolic calcium.¹⁹⁴ Following activation platelets degranulate and release a variety of signaling molecules, including TXA₂. The TXA₂ released by the platelets then acts in an autocrine fashion to accelerate the activation process as well as enhance the activation of other platelets in the region through paracrine effects. The activated platelets then provide additional stimuli to the already activated neutrophils and monocytes (as well as endothelial cells) via the platelet-derived cytokines and chemokines. Thus, the induction of COX-2 in activated neutrophils (as well as in macrophages) enhances the production of LTB₄, contributing further to a platelet-enhanced lipid–cytokine–chemokine cascade. However, as described in Section 2.3.4, induction of COX-2 is not as bad as is generally thought to be.

Inflammation

Another aspect of platelet activation that occurs independently from both endothelial disruption and TXA₂-mediated events is the activation of platelets by PAF. As described in Section 2.2.2, PAF is produced by vascular endothelial cells as well as activated macrophages, and PAF receptors are common in macrophages, neutrophils, vascular endothelial cells, and platelets. PAF activates platelets to aggregate as well as release their bioactive compounds through the activation of G-coupled PAF receptors (PAFRs) and the subsequent activation of phospholipase C, which leads to the production of IP₃ and diacyl glycerol (DAG), increasing cytosolic Ca²⁺ and activity of PKC.^{200,201} PI3K is also activated by PAFRs; this can lead to SOS– GRB-mediated activation of Ras, which results in the activation of ERK–MAPK, an activity that can be enhanced by the activated PKC and results in phosphorylation of PLA₂ to further enhance the synthesis of proinflammatory eicosanoids. Thus, the ability of platelets to contribute to both the production and the amplification of inflammatory signaling is clear, and their contribution to the pathological consequences of chronic inflammation should not be underestimated.

From this discussion, it is certainly possible that a minimal activation of tissue macrophages by DAMPs released as a result of a small amount of local cellular damage could be accelerated through the autocrine induction of COX-2, a cascade aided by platelets through a variety of cellular and cytokine interactions described in this section. With an induction of COX-2 occurring within 4–6 hours of stimulation in macrophages, neutrophils, and endothelial cells,202-204 such progressions could occur relatively quickly. In normal circumstances where responses to the various clotting and growth factors lead to complete healing, a localized inflammatory response to damage does not necessarily result in an accelerating cascade of increased signaling complete with subsequent invasion of the tissue with many additional activated neutrophils, monocytes, and macrophages. In cases where healing is incomplete, or where continuing low-level and nonnecrotic damage to cells occurs, the processes of the platelet-enhanced lipid-chemokine-cytokine cascade ensure that there will be a sustained localized production of inflammatory cytokines and chemokines, as well as growth factors and coagulation factors. The production of all these factors will last for as long as the initiating PAMPs, DAMPs, or cell stressors exist. Further, it is the prolonged production of these signaling molecules that, even at very low levels, has implications for the development of chronic diseases.

2.3.3 WOUND HEALING

Wound healing is also a very important consequence of an inflammatory response, when there is an open wound or a necrotic tissue damage that needs healing. However, in instances where there is only a low-level inflammatory response due to nonnecrotic sterile cellular damage, such processes can contribute to chronic disease. As with clot formation, all the details of wound healing are not discussed here; however, the important signaling processes that lead to healing are discussed.

In general, wound healing is started by the release of various signaling molecules by activated platelets, which are then responsible for stimulating the healing processes. These signaling molecules include IL-1 β ; TNF- α ; PDGF; TGF β ; epidermal growth factor; bFGF; and the chemokines RANTES, platelet factor 4, and IL-8 (among others). PDGF and TGF β , as well as the various chemokines, initiate the chemotaxis of tissue macrophages, smooth muscle cells, and fibroblasts into the damaged area. PDGF also has the effect of stimulating mitogenesis of smooth muscle cells and fibroblasts, whereas TGF β stimulates macrophages to release (among others) fibroblast growth factor and PDGF, as well as enhancing the ability of fibroblasts to deposit collagen. All of these effects result in the efficient formation of new scar and connective tissue to close the wound. Continual remodeling and maturation of the newly formed scar tissue by fibroblast activity is maintained through further production of growth factors by the macrophages and fibroblasts, as well as by vascular endothelial cells (and epidermal cells if it is a skin wound) and growth factors that include VEGF and bFGF (also produced by platelets). The latter two are very important for stimulating angiogenesis in the newly formed tissues.^{205–207}

If everything goes well, a relatively strong mature scar with fully repaired and functional tissue in the formerly damaged region will result. Unfortunately, these same platelet-initiated processes occur even when the inflammation is initiated by infectious agents or toxic chemicals and there are no open wounds to close. In the case of chronic or repeated infections in the lung and liver, for example, the (in these cases, aberrant) fibrotic processes can result in the pulmonary fibrosis observed with tuberculosis infections or the cirrhosis in infectious hepatitis. Similar fibrotic changes are also observed with chronic exposures to chemical toxins such as those resulting in chronic obstructive pulmonary disease in smokers or cirrhosis of the liver in alcoholics. Although these kinds of chronic inflammatory conditions that result in pathologic fibrosis are not the focus of this chapter, these same plateletinitiated fibrotic processes are very much involved in the etiology of atherosclerosis, as described in Chapter 4, and contributory to the neurodegenerative diseases discussed in Chapter 7. In addition, the mitogenic and angiogenic effects of some of the signaling molecules readily contribute to processes associated with cancer, as do many other processes of inflammation, as described in Chapter 6.

2.3.4 **R**ESOLUTION OF INFLAMMATION

One final issue that needs to be dealt with is the transition from proinflammatory signaling to resolution of the initial inflammatory response in order for healing to occur. Although components of this topic might ordinarily be placed in the preceding wound healing section (Section 2.3.3), they are placed here in order to emphasize the importance of the resolution process. Obviously, without resolution inflammation can become a chronic process; the lipoxin A_4 (LXA₄) and lipoxin B_4 (LXB₄) made from AA, D-series resolvins made from DHA, and E-series resolvins made from EPA play an important role in the resolution process (Figure 2.5).

Details of the resolution phase of inflammation and the roles played by AA, EPA, and DHA metabolites have only recently come to light with the first descriptions of lipoxins in 1984²⁰⁸ and, more recently, those of various resolvins, protectins, and maresins.^{119,145–147,208–213} LXA₄ and LXB₄ can be synthesized from AA via two main dual cell processes. 15-HETE, which is synthesized by mucosal epithelial cells, vascular endothelial cells, or macrophages by 12-LOX/15-LOX, can be transferred through direct cell–cell contact to neutrophils and converted



FIGURE 2.5 Resolution of inflammation: proinflammatory signaling leads to an induction of cyclooxygenase (COX)-2 and mPGES, increasing prostaglandin (PG) E-2 synthesis and subsequently resulting in an enhanced expression of 12-lipoxygenase (12-LOX) and 15-lipoxygenase (15-LOX) (upper box). Increased LOX in conjunction with COX-2 greatly enhances the synthesis of lipoxins, resolvins, and PGJ-2. Lipoxins and resolvins suppress (rounded, shaded boxes) proinflammatory signaling via inhibiting the activation of MAPKs and nuclear factor κ - β (NF $\kappa\beta$) through enhanced expression of suppressor of cytokine synthesis (SOCS) in all inflammatory cells while PGJ-2 (rounded, shaded box) activates apoptosis in neutrophils as well as suppresses mitogen-activated protein kinases (MAPKs) and NF $\kappa\beta$ via activation of peroxisome proliferator–activated receptor- γ (PPAR γ) and Nrf2. At the same time, macrophages phagocytize apoptotic neutrophils (without production of reactive oxygen species [ROSs]) and are stimulated to egress from the tissue, whereas chemotaxis of other inflammatory cells is inhibited, effectively ending an inflammatory response.

to 15*S*-hydroxy-5(6)-epoxy-ETE by 5-LOX in the neutrophils. The epoxide is then reduced by epoxide hydrolase to produce either LXA₄ or LXB₄. Neutrophil– platelet interactions in blood vessels can also produce lipoxins through the production of LTA₄ in neutrophils via 5-LOX activity and then LXA₄ and LXB₄ via 12-LOX/15-LOX in platelets.^{147,214–216} One can easily discern that lipoxins also could be produced by neutrophil-produced LTA₄ that is made into LXA₄ by vascular endothelial cell interactions while rolling. Because of the phenomenon of the platelet activation cascade (described in Section 2.3.2), the predominant source of lipoxins during initial stages of inflammation would most likely be neutrophil–platelet interactions with the contribution of neutrophil–endothelial cell contact also. However, because of very low 12-LOX/15-LOX activity in neutrophils, the lipoxin production in these cells would be comparatively small during early phases of the inflammatory response relative to that of the proinflammatory PGs, LTs, and TXs.

Synthesis of resolvins, protectins, and maresins, of which the resolvin E1 (RvE1) and the protectin D1 (PD1) have been well characterized, also is mediated by lipoxygenase (LOX) activities. The 18*R*-hydroperoxy EPE (18*R*-HPEPE) produced by aspirin-acetylated COX-2 from EPA is converted to 5*S*-hydroperoxy 18*R*-hydroxy EPE (18*R*-HEPE) by 5-LOX, and this product can be reduced to resolvin E2 (RvE2) or further metabolized by 5-LOX to RvE1. While the initial characterization of resolvin synthesis centered on aspirin-acetylated COX-2 studies, bioactive amounts of RvEs are produced in humans in conjunction with COX-2 activity without the presence of aspirin.^{147,211,214,217} The D-series resolvins (RvD1-4) are produced from DHA by a series of reactions initiated by 5-LOX, 12-LOX, and/or 15-LOX, whereas PD1 synthesis from DHA is initiated through 15-LOX activity and maresin (MaR1) synthesis is initiated by 12-LOX.^{147,209,214,217,218} Thus, activity of 5-LOX/12-LOX/15-LOX is central to the production of lipoxins, resolvins, protectins, and maresins.

In order for the proresolving effects of various lipoxins and resolvins to be realized, a mechanism to enhance their production is needed; in fact, multiple mechanisms exist for this. In addition to its well-known vasodilation effects, PGE₂ will also induce the synthesis of 12-LOX/15-LOX^{145,219} in neutrophils and macrophages, an effect that greatly enhances the production of lipoxins through cell–cell interactions as well as by solitary neutrophils.^{145,146,211,220,221} Because the production of RvDs depends on LOX activities, the induction of 12-LOX/15-LOX resulting from PGE₂ will also increase their synthesis while synthesis of E-series resolvins are enhanced because of the induction of COX-2 by the proinflammatory cytokines IL-1 β , IFN- β , and TNF- α . As previously mentioned in Section 2.2.2, the synthesis of mPGES-1 is coinduced along with COX-2, resulting in a concomitant increase in PGE₂ synthesis along with its inducing effects, further enhancing the ability to synthesize these proresolving mediators.

In order to produce their effects, the various proresolving/anti-inflammatory mediators bind to their appropriate receptors and alter the function of target cells. LXA_4 (LXB₄ receptor has not yet been characterized) binds to the G-protein-coupled ALX/FPR2 receptors of neutrophils, endothelial cells, macrophages, and monocytes^{214,215} and produces a variety of proresolution effects by attenuating p38–MAPK, p42/44–MAPK (ERK1/2), and PI3K activities as well as attenuating the activation and expression of NF $\kappa\beta$ and AP1.^{119,214,215,222,223} From previous discussions

on the activation of neutrophils, macrophages, endothelial cells, and dendritic cells, the implications of these effects on the proinflammatory signal transduction pathways should be profound. As expected, neutrophil chemotaxis and infiltration into the region is inhibited; apoptosis of neutrophils is enhanced; and production of ROS, RNS, and IL-8 is reduced in both neutrophils and macrophages. In macrophages, TNF- α release is blocked and TGF β (inhibits toll-like signaling to reduce the activation of inflammatory cells) and IL-10 release is enhanced.

IL-10, through autocrine activation of the Jak–STAT pathway coupled with a prolonged STAT3 activation (activation of this pathway by IL-6 leads to a transient activation of STAT3), results in the synthesis of suppressor of cytokine synthesis (SOCS) proteins, which then inhibit the synthesis and release of IL-1 α , IL-1 β , IL-6, IL-8, and TNF- α .^{224–226} IL-10 is also associated with the suppression of MAPK pathway and NF α activities, which, again, inhibits the synthesis of the acute-phase cytokines.^{224–227} In addition to these IL-10 effects, macrophages are also stimulated to phagocytize apoptotic neutrophils without producing ROSs/RNSs (recall that stimulation of phagocytic activity in macrophages through the activation of PRRs is accompanied by stimulated ROS/RNS production) and to egress from the tissue.^{145,214,228} This stimulation of nonphlogistic phagocytosis along with enhanced TGF β release is an important resolution event because it helps to clear apoptotic neutrophils from the area without causing ROS-/RNS-mediated damage at the same time severely inhibiting the proinflammatory actions of macrophages and other cells in the region.

As a result of the aforementioned inducing effects of the proinflammatory mediators on LOX and COX activities, there is a progressive change in the production of proinflammatory lipid mediators over time in the inflamed region to one in which proresolution and anti-inflammatory lipid mediators predominate in the later stages of an inflammatory response. Another effect of the induced COX-2 synthesis is an increase in the production of PGD₂. The importance of this is that it spontaneously dehydrates to PGJ₂, which is a potent activator of PPARy.^{119,123,229} PPAR activation promotes neutrophil apoptosis as well as suppressing NO' synthase and IL-1ß and TNF- α release by macrophages and dendritic cells.^{124,229,230} Similar to the effects described for LXA₄ binding to the ALX/FPR2 receptor, PGJ₂-mediated responses result from inhibiting TLR actions by attenuating ERK1 and p38–MAPK; by blocking AP1 activation; and through inhibiting NFκβ activation by promoting the formation of p50 homodimers, which inhibit the normal p60/p50 transactivation dimers from binding.^{124,229–231} The end result of all these effects is to greatly reduce the production of both proinflammatory cytokines and lipid mediators, enhance the production of proresolution and anti-inflammatory mediators, reduce the production of damaging ROSs and RNSs, and remove neutrophils and macrophages from the area.

PGJ₂ also activates the transcription factor Nrf-2 (via modification of thiols in Keap1), resulting in the upregulation of heme oxygenase (HO)-1 as well as other electrophile-response element (ERE) (a.k.a. antioxidant-response element [ARE]) responsive genes such as glutathione peroxidase (GPX), superoxide dismutase (SOD), catalase (CAT), glutathione-*S*-transferase (GST), and UDP-glucuronosyl transferase (UDPGT).^{124,232–236} The significance of these antioxidant enzymes in protecting cells from oxidative damage (including neutrophil and macrophage-generated ROSs and

RNSs) cannot be overestimated, and these responses are discussed in more detail here. As an aside, with many of the aforementioned benefits being dependent on the induction of COX-2, the clinical use of non-aspirin COX-2 inhibitors might be brought into question.

The induction of COX-2 also increases the synthesis of E-series resolvins from EPA, whereas increases in 12-LOX/15-LOX enhance the production of D-series resolvins as well as protectins and maresins. These also have important receptormediated effects, with those of RvE1 and resolvin D1 (RvD1) being characterized.^{147,215} RvE1 binds to the G-protein-coupled ChemR23 (a.k.a. CMKLR1) receptor as well as the BLT1 (of LTB₄ fame) receptor. Thus, RvE1 competes with LTB₄ for binding and at physiological concentrations (1 nM) RvE1 attenuates the LTB₄dependent activation of NF $\kappa\beta$ by approximately 40%–50% by blocking increases in Ca^{2+,147,215,237} The interaction of RvE1 with the ChemR23 receptor is interesting because chemerin, the peptide ligand for the same receptor, is an adipokine produced by differentiating adipocytes that has proinflammatory properties by acting as a chemoattractant (chemokine) for monocytes, macrophages, and dendritic cells. Binding of RvE1 to this receptor produces a completely different response: phosphorylation of the ribosomal S6 protein, apparently via Akt (a.k.a. PKB), is enhanced to significantly enhance nonphlogistic phagocytosis of apoptotic neutrophils by macrophages.^{147,238} RvE1 can also bind to components of neutrophil membranes and through non-receptor-mediated actions to greatly reduce neutrophil chemotaxis and transmigration.^{147,210,215} From these results, RvE1-mediated activities significantly enhance the resolution of inflammation in a manner similar to that of LXA₄-ALX binding. Although the RvE2 receptor has not yet been identified, there is evidence of additive effects of RvE1 and RvE2, indicating either multiple-binding effects mediated through the same receptor or the presence of a specific RvE2 receptor.²¹⁵ In any event, both RvE1 and RvE2 clearly contribute to the resolution process.

RvD1, synthesized following the induction of 12-LOX/15-LOX activities, also binds to the ALX receptor and to the GPR32 receptor. While GPR32 receptor– mediated events are not known, RvD1 binding to ALX activates phagocytic and clearance functions of macrophages very similar to those seen following the binding of LXA₄ to ALX.^{147,239} The receptor-mediated events of PD1, also very poorly understood, result in reduced neutrophil infiltration and stimulated nonphlogistic phagocytosis of apoptotic neutrophils by macrophages.^{147,210,215,240}

From all of the aforementioned factors, the binding of lipoxins, resolvins, protectins, and maresins to specific receptors that are present on neutrophils, macrophages, monocytes, and dendritic cells in the immediate region leads to the resolution of an inflammatory response. The generic sequence of events—pathogen-/damageinduced proinflammatory signaling \rightarrow infiltration and activation of inflammatory cells \rightarrow induction of various LOX/COX enzymes \rightarrow anti-inflammatory/proresolution signaling \rightarrow removal of inflammatory cells—has been elucidated in a variety of in vitro, in vivo, and in situ experiments that mostly relate to processes occurring in exudates or air-pouch models of inflammation. These models are highly relevant to a localized infection where the ordered process of inflammation events leading to the removal of the damaging agent followed by healing occurs. However, the significance of these resolution events to chronic disease is a little more difficult to ascertain. In many cases of chronic disease, there is a constant production of sterile trauma that results in constant initiation (of inflammation) events followed by the constant presence of inflammatory cells and their associated signaling molecules, ROS and RNS, and on the surface resolution does not seem to be much of an option in these circumstances. When proinflammatory signaling molecules enter an area from some distal infectious (or sterile) inflammatory response, resolution also does not seem to be a local option. Even with a properly resolved inflammatory response that was initiated by an infectious event, adapted immune responses to reinfection can modify or produce inflammation-mediated events in otherwise unaffected cells, even if they are far away from the reinfection event.

2.3.5 Adaptive Immunity and Inflammation

As is well known, an adapted immune response to a particular pathogen gives us the ability to respond swiftly to a reinfection event in order to remove the pathogen before extensive damage (or death) can occur. Further, it is the innate inflammatory response that gets the ball rolling for the development of the adapted response. Once pathogens and cellular debris have been phagocytized by activated dendritic cells and macrophages, these professional antigen-presenting cells can migrate to the lymph where they interact with naive T lymphocytes (T cells) and B lymphocytes (B cells) to initiate an adaptive immune response. Once produced, the various T cell subpopulations, as well as the B cell subpopulations, circulate throughout the secondary lymphoid organs (predominantly spleen, lymph nodes, and the Peyer's patches of intestinal epithelium) via the blood with much of their time spent in the lymphoid tissues, while the long-lived plasma B cells tend to remain in the bone marrow.^{241–244} Although the details of the adaptive immune processes the result in the generation of the various cell types are not a topic for this chapter, the resulting effector functions of the T and B cells that are associated with proinflammatory responses will be discussed.

An array of CD4⁺ and CD8⁺ T cells resulting from adaptive immune responses can respond to chemokines produced during a local inflammatory response and migrate into the area. CD4⁺ effector T cells include the Th1, Th2, Th17, and Treg subsets.^{241,245-247} Once in the area, if the effector T cells bind with the same (or structurally similar) antigens to which they were originally selected for, then they are activated to produce a variety of cytokines that are necessary for enhancing the inflammatory response as well as to initiate their effector functions. Th1 (CD4⁺) cells express IL-2 and IFN- γ , whereas Th2 (CD4⁺) cells express a variety of ILs, including IL-4, IL-5, IL-6, and IL-10. The different expression of cytokines relates to the different effector functions of the activated Th cells. Th1 cells are involved, in part, in activating both inflammation and cytotoxic activities through the expression of IL-2 and IFN- γ (activates macrophages) as well as lymphotoxin- α , whereas Th2 cells are potent activators of B cells (including stimulation of Ig class switching to IgE) and can downregulate macrophages through the production of IL-4 and IL-10. Thus, Th1 responses are predominantly proinflammatory and cytotoxic, whereas Th2 responses are predominantly anti-inflammatory. Not coincidently, a variety of autoimmune diseases are associated with Th1 overactivation without appropriate activation of anti-inflammatory Th2 responses.^{246–249} The

more recently described Th17 and Treg cells also are associated with autoimmune diseases and inflammation. Th17 cells express the chemokine receptor CCR6, which binds the CCL20 produced by cells associated with Peyer's patches and lymphoid follicles as well as by inflamed intestinal epithelium, joints, and central nervous system. The Th17 cells produce the proinflammatory cytokines IL-17, TNF- α , and lymphotoxin- β . IL-17 is proinflammatory because it stimulates the synthesis and release of a variety of proinflammatory cytokines (IL-1, IL-6, and TNF- α) through the activation of NF $\kappa\beta$ in a variety of cells, including endothelial cells, macrophages, and fibroblasts.^{250,251} On the other hand, Treg cells are associated with regulating and inhibiting autoimmune responses and autoimmune conditions have recently been associated with enhanced Th17 and reduced Treg cell function.^{245,246,252-256}

Cytotoxic T cells (Tc cells—CD8⁺) also respond to various chemokines produced during an inflammatory response and home in to the inflamed tissue. Here, they are activated when they come in contact with and bind to the antigen. Activated Tc cells secrete perforins to create pores in the membranes of antigen-containing cells through which they release granules that contain enzymes, which then induce apoptosis in the target cells by activating caspases. They can also activate apoptosis by directly binding to surface-bound Fas molecules on the target cells. In addition to contact-mediated mechanisms of killing, Tc cells also release the proinflammatory cytokines IFN- γ , TNF- α , and TNF- β .^{241,257–259} Finally, similar to the Th1/Th2 dichotomy, CCR7⁻ and CCR7⁺ memory T cells have differing functions. The CCR7⁻ memory cells circulate and migrate into inflamed areas, whereas the CCR7⁺ memory T cells home in to the lymph because of the differing expression of chemokine receptors. It is the CCR7⁻ memory T cells that produce IFN- γ , resulting in a proinflammatory effect by further activating macrophages.^{246,260,261}

The proinflammatory effects of cytokine release by the various subsets of Th cells (as part of their effector functions) may be relevant as a component of risk for inflammation-associated chronic diseases, especially in the event of recurrent infections. In addition, there is a subset of memory T cells that tends to circulate through the tissues in which the original antigen exposure occurred.^{262–264} This enhances the likelihood of proinflammatory effects in those tissues susceptible to multiple exposures over time, for example, tissues such as skin, oral and intestinal mucosa, and lung. Although the effector functions resulting from an adapted immune response limit the pathological consequences of a second infection through efficient removal of the pathogen, they can also increase the probability that a local inflammation response due to non-infection-related trauma can be exacerbated through circulating IFN- γ , or even locally produced cytokines in the event *stray* pathogens enter an already inflamed area.

Unlike the various subsets of T cells, B cells do not actually appear to have much to do with acute inflammation, most likely because if they are not already in the secondary lymphoid tissues they will migrate to those locations in response to the chemokines SDF-1, SLC, CK β -11, and BLC.^{241,247,265} B cells do, however, appear to play a role in the development of autoimmune disease as a consequence of their ability to act as antigen-presenting cells for T lymphocytes (B cells are able to activate T cells even at very low titers of antigen,²⁶⁵ as would be expected for self-antigens),

as well as the ability of some subsets of memory B cells to regulate the activation of T cells. 246,265,266

The last lymphocyte family to be considered is that of NK cells. NK cells are typically identified with innate immunity, although recent evidence indicates that NK cells not only interact reciprocally with dendritic cells through (paracrine) cytokine release during the initiation of adaptive responses but some subsets also appear to have memory functions.^{267–271} NK cells migrate into sites of inflammation in response to various chemokines, where they interact with a variety of ligands on target cells (such as CD16, NK cell protein 30 [NKp30], NKp44, and NKp46, among others), ligands commonly seen on cells that are infected with viruses or damaged. The cells that do not express sufficient class I MHC molecules to inhibit the killing function of NK cells are then lysed through secreted perforins and *granzymes* (similar to Tc cells). Cells that do not produce sufficient class I MHC also tend to be those that are transformed or have undergone some form of damaging stress. Thus, NK cells provide the vital function of eliminating what essentially are unhealthy and poorly functioning cells.^{16,18,241}

Although details of the roles that NK cells play in developing adaptive immunity and (NK cell) memory functions are beyond the scope of this chapter, NK cells can and do play a role in inflammation. In addition to their cytolytic activity, activated NK cells synthesize and release the proinflammatory cytokines IFN- γ and TNF- α as well as various chemokines, including CCL2, CCL3, CCL4, and CCL5.^{267,271–273} The release of chemokines by activated NK cells attracts a variety of inflammatory cells to the area, including macrophages and dendritic cells. The end result of activating NK cells is therefore an increased inflammatory response with an increase in killing of invading pathogens via activated macrophage phagocytosis and ROS production as well as NK-mediated destruction of infected cells. Although all this results in a more efficient removal of pathogens, it also means an increased likelihood of damage to the otherwise healthy cells in the immediate area of the infection as well as a possible increase in the localized production of various growth factors. The implications of peripheral damage may be minimal in terms of a single infection-mediated activation of NK cell activity. The ramifications may be much more severe, however, when repeat viral infections are considered, especially as NK cells retain some memory functions and will more readily home into areas of previous infections.270,273

2.4 CONTRIBUTIONS OF INFLAMMATORY PROCESSES TO CHRONIC DISEASE

The mechanisms through which inflammatory responses may contribute to chronic disease have been mentioned as part of the summary discussions throughout the various sections in this chapter. In this section, the general risk-enhancing effects of inflammation are discussed, whereas the application of these mechanisms to specific diseases is discussed in detail in Chapters 3 through 8. At its simplest, inflammatory responses contribute to disease through two main mechanisms: the first is through the effects of various ROSs, RNSs, and other active molecular species produced by activated neutrophils, mast cells, and macrophages as a means to kill or destroy

pathogens. The second is through the generation of signaling molecules that can alter the function of noninflammatory cells as well as those specifically involved in inflammatory response.

2.4.1 REACTIVE MOLECULAR SPECIES

As already described in Section 2.3.1, the generation of various ROSs and RNSs occurs in the phagolysosomes of phagocytes as a result of the activities of NADPH oxidase and nitric oxide synthase. It is these reactive molecules produced by enzymes that contribute to disease, as first described by McCord, Halliwell, Beckman, and others.²⁷⁴⁻²⁸¹ Single electrons released by NADPH oxidase get picked up by molecular oxygen to create the superoxide anion radical: $O_2 \overline{\bullet}$. The superoxide anions cannot cross cellular membranes due to their charge, but they do react with one another to produce peroxide and oxygen: $O_2 \cdot + O_2 \cdot + 2H_2O \rightarrow H_2O_2 + O_2 + 2OH^-$. The peroxide is very stable and readily diffuses through cellular membranes to enter the extracellular space as well as into and through the membranes of surrounding cells. The nitric oxide (an uncharged radical) that is produced by activated phagocytes also readily diffuses through cellular membranes. Most NO' produced within the phagolysosome will react with O_2 to produce peroxynitrite: ONOO⁻ (a highly reactive nonradical oxidant); some of the NO' will diffuse through membranes, out of the cell, and into adjacent cells. Although H₂O₂ and NO[•] are not highly toxic alone, it is their reactions with $O_2 \overline{\bullet}$ and carbon dioxide (CO₂) that are highly relevant because of the resulting production of extremely reactive products.

The ONOO⁻ produced from the NO+ O_2 reaction is highly reactive with CO₂, producing carbonate radicals (CO₃ $\overline{\bullet}$) and nitrogen dioxide radicals (NO₂ \cdot). There is certainly no shortage of CO_3 because of the high rate of production of NO[•] during inflammatory responses and the easy availability of CO₂ and O₂ - in tissues because of normal cellular respiration. Thus, DNA, proteins, and phospholipids in cells within the immediate region of the activated phagocytes are all susceptible to attack by both the reactive carbonate radicals and the moderately reactive nitrogen dioxide radicals.^{281–283} The ONOO⁻ can also react with transition metals in proteins such as SOD, myeloperoxidase, aconitase, zinc-containing transcription factors, heme, and others to produce a nitrating species, which results in the production of a variety of nitration reactions with almost any nearby amino acids, 274, 276, 281 most of which result in inactivation of the proteins. As might be expected, nitration products are highly associated with chronic inflammation and a variety of chronic diseases, including atherosclerosis, Alzheimer's, cancer, and Parkinson's.^{276,281,284-288} Thus, inflammation-associated nitric oxide production is a significant source of RNS that leads to chemical damage in cells that otherwise would not have occurred. While the affected cells have an antioxidant capacity due to their own SOD, glutathione reductase (GSR), GPX, and CAT enzymes, the additional H₂O₂ and NO[•] from extracellular sources simply overwhelms the endogenous antioxidant capacity to produce additional damage and an enhanced risk for chronic disease.

HOCl is another oxidant produced by activated neutrophils that can diffuse through cell membranes and cause damage to cells. Neutrophils (and macrophages) secrete the enzyme myeloperoxidase, which uses the extracellular H_2O_2 and chloride

ion to produce HOCl.^{289–293} Because HOCl is produced extracellularly prior to diffusing into adjacent cells, this oxidizer can react with a variety of compounds in extracellular spaces as well, including lipids. This reaction is relatively slow, however, rendering it relevant mainly in cases of a prolonged exposure of extracellular lipids to HOCl, such as in atherosclerotic plaque. Once inside a cell, HOCl can oxidize a variety of cellular components, including DNA, RNA, proteins, and lipids.^{289–293} However, HOCl reacts most strongly with thiol residues of proteins,²⁹⁰ indicating that this oxidant might function predominantly as a modifier of the cellular redox state (discussed here).

The H_2O_2 produced by the phagolysosomes and by the plasma membrane-bound NADPH oxidases is a significant source of ROSs for cells within the inflammatory region. Hydroxyl radicals, probably the most aggressive oxidizing ROS known, are produced from the reaction between O_2^{-1} and $H_2O_2(O_2^{-1} + H_2O_2) \rightarrow OH + HO^{-1} + O_2)$. As with NO', H_2O_2 is sufficiently stable for it to diffuse across membranes and throughout the entire cell where it reacts with locally produced O_2^{-1} ; the resulting •OH will then oxidize any molecule within the immediate vicinity. Because of the continuous production of O_2^{-1} from normal respiration and the dismutation of O_2^{-1} to H_2O_2 , the production of •OH and subsequent damage to proteins, DNA, and phospholipids is inevitable. Even without inflammation, this damage is common and considered normal. However, when additional H_2O_2 enters cells from inflammation-related events and ultimately overwhelms their antioxidant capacity the additional damage caused by the ROS can significantly alter cell functions. It is the alterations in cell function due to excessive inflammation-associated ROS damage that contributes (in part) to greater risks for chronic diseases.^{274–281,294,295}

2.4.2 REACTIVE MOLECULAR SPECIES AND CELLULAR SIGNALING

Inactivation of proteins, formation of DNA adducts, and peroxidation of lipids are well-known consequences of endogenous ROS- and RNS-mediated reactions and are a major reason for the constant synthesis activities in all cells; the damaged structures simply must either be replaced or be repaired in order to minimize the inefficiencies produced. Transcription is a highly regulated activity, and increased rates of protein synthesis in response to enhanced rates of damage can only occur following alterations in the signal transduction pathways that regulate transcription. In the case of increasing amounts of oxidative damage due to additional inflammation-produced ROSs/RNSs, however, increased synthesis and repair activities are not necessarily a direct response to the increased damage. In fact, cells have evolved the interesting ability to use the damaging agent itself as a signaling device to activate repair and synthesis. As is now well known, ROSs, RNSs, HOCl, and other oxidizing agents indirectly modify the activity of signal transduction pathways by oxidizing the *redox sensor* proteins thioredoxin (Trx), peroxiredoxin (Prx), and glutaredoxin. These redox sensor proteins affect the function of specific enzymes of the pathways or factors that then activate the pathways, depending on their oxidation state.²⁹⁵⁻³⁰² Although all the details of these forms of transcription regulation are not the major focus of this chapter, some general issues are discussed here.

A reduction in cellular glutathione (GSH) concentration through a sustained-GPX-mediated reduction of H₂O₂ leads to decreased GPX activity and increased cellular H₂O₂ and ultimately to increased oxidation of the different redox sensor proteins. H_2O_2 directly oxidizes the susceptible cysteines of reduced Prx and Trx, resulting in the formation of disulfides on the proteins and H₂O from the peroxide. For this reason, both Trx and Prx have been considered to be antioxidant enzymes. One role for Trx is to form an inhibitory complex with the ASK1 protein, which disassociates on Trx oxidation, allowing ASK1 to then activate the JNK and p38-MAPK pathways with the resulting pattern of protein expression depending on the degree of Trx oxidation by H₂O₂. The antioxidant function of Trx can also be an important component of its regulatory function. For example, cytosolic NFκβ can be activated by H_2O_2 and the peroxidase activity of Trx reduces the H_2O_2 to water, thereby reducing the activation of NFkß. Because of the heterogeneous distribution of these redox sensors throughout a cell, a given redox state in one location of a cell produces a differing pattern of protein expression than the same redox state in another location of the cell. At this point, it is instructive to mention that the activation of NADPH oxidases at the plasma membrane (NOX-1) is one of the immediate effects when cytokines bind to their G-protein receptors (a component of activating the formation of phagolysosomes in phagocytes, as previously discussed in Section 2.3.1). This creates a transient and highly localized production of H_2O_2 , which then alters the oxidation state of Trx in the immediate vicinity and, subsequently, the activity of signal transduction proteins (again, only in the immediate vicinity). The result is a pattern of transcription activation that is highly specific to the degree of cellular oxidation at one specific location. One can imagine that as the concentration of a specific cytokine increases, there will be an increase in the duration, amount, and spatial distribution of H₂O₂ production, which then slightly changes the pattern of transcription activation. Add to this the additional localized productions of H_2O_2 by different cytokine receptors (at different physical locations on the membrane) and the complexity of the various possible interactions provides an explanation for the differing responses to different cytokines in spite of the fact that they utilize common NFκβ and MAPK pathways.302-304

Interestingly, Prx can form dimers with specific regulator proteins such as JNK, c-Abl, and c-Myc to attenuate their activation. The activation of certain transcription factors also appears to be dependent, in part, on the redox state of Trx and Prx, including Jun/Fos (AP-1) and heat shock factor 1.^{298,302,303,305} All of these activation events at the transcription factor and signal transduction level are important when one considers that a highly localized and transient production of ROSs is a common component of receptor-mediated regulation of signal transduction pathways. Because global changes in the redox state of a cell can occur as a result of excess ROS entry into cells due to phagocytic responses, the ability of inflammatory responses to alter a highly regulated process to cause large changes in signal transduction activity is profound. Of course, many of the cellular responses to global changes in ROS-mediated activation of signal transduction pathways provide important protection effects through induction of a variety of antioxidant enzymes and an increased capacity for repair and replacement of damaged proteins. On the other hand, with prolonged and nontransient activation of some of these pathways, JNK,

AP-1, and c-Abl, for example, cell division can be stimulated and tumorigenesis starts to become a distinct risk.

One additional issue related to the redox-based regulation of these pathways is the activity and cellular distribution of antioxidant enzymes, which serve to quench the various ROSs. As with redox sensors and oxidants, antioxidant enzymes such as SOD, CAT, and GPX alter the local concentrations of ROSs and therefore contribute to the optimal regulation of signal transduction pathways in the cell.^{295–303,305} As with redox sensors, the cellular distribution of antioxidant enzymes is heterogeneous, resulting in a very complex, yet highly regulated (through both redox-sensing and antioxidant enzyme systems) system of control. This does raise the interesting possibility that nutritional insufficiencies and the resulting inability to synthesize optimal amounts of these antioxidant enzymes might interfere with what would otherwise be normal regulation of cell signaling.

The inflammation-associated peroxidation of phospholipids resulting from the additional supply of oxidants also may lead to alterations in cellular function through mechanisms that are independent of protein and DNA damage and alterations in transcription caused by these oxidants. The unsaturated fatty acids on phospholipids are highly susceptible to attack by hydroxyl radicals, and the resulting lipid peroxidation makes the membrane leakier to various charged molecules such as K⁺ and Ca²⁺ that normally diffuse across the membrane very poorly.³⁰⁵ The importance of calcium in intracellular signaling process cannot be overemphasized.

Highly localized transient increases in calcium throughout the cell regulate a large variety of cellular functions, with the duration, concentration, and spatial distribution of the calcium spikes being the key determining factors in the resulting cellular effects.³⁰⁶⁻³¹³ The entry of calcium into cells is a highly regulated process with a variety of endoplasmic membrane- and plasma membrane-bound calcium channels being involved, including voltage-gated calcium channels; DAG-regulated calcium channels; inositol-1,4,5-3P receptor (I3PR); and ryanodine receptors. Opening and closing of the various calcium channels is normally controlled by a wide array of cytokine, paracrine, and autocrine factors, as well as being modified by ROSs, NOSs, and calcium. Once in the cytosol, calcium exerts its regulatory effects by binding to a large variety of proteins, including calcium-dependent protein kinases (PKC), calmodulin, PLA₂, calcineurin, adenylate cyclase, phosphorylase kinase, and a variety of metabolic enzymes in the cytosol and mitochondria. Through the activation of PKC and calmodulin, a variety of Rho-GTPases and Ras are activated, leading to phosphorylation and activation of p38–MAPK, JNK–MAPK, ERK1/2–MAPK, and Jak-STAT pathways. If sufficient calcium enters the cell, then mitochondrial function will be compromised, lysosomal activity will be activated, and the cell will die through either necrosis or apoptosis depending on the degree of mitochondrial disruption. If nonfatal amounts of calcium enter through damaged membranes, then extensive activation of the MAPK and STAT pathways can lead to the protective effects of increased synthesis, repair, and antioxidant function; if chronic, a risk for hypertrophy/mitosis and tumorigenesis due to the proliferative effects of prolonged activation of MAPKs is the result. With respect to the last comment, lipid peroxidation due to excessive ROS production results in the production of ketones, aldehydes, malondialdehyde, acrolein, 4-hydroxy-2-nonenal, and crotonaldehyde from the peroxidized fatty acids. All of these peroxidation products can damage other lipids, proteins, as well as DNA,^{314–316} significantly disrupting cellular function and enhancing mutagenesis, thereby increasing the risk for cancers.

2.4.3 INFLAMMATION-ASSOCIATED CYTOKINES AND GROWTH FACTORS AND CELLULAR SIGNALING

The last major mechanism through which inflammatory processes contribute to disease that we discuss is through the signaling process itself. The various cytokines and growth factors produced by activated inflammatory cells initiate processes that are necessary for repair of wounds and replacement of the cells that die as a result of the pathogenic or endogenous trauma. The variety of growth factors released by activated platelets, macrophages, fibroblasts, and endothelial cells, including VEGF, PDGF, bFGF, EGF, and TGF β , are necessary for normal scar formation and wound healing. In essence, the effects of these growth factors are to activate resident fibroblasts to produce scar tissue to close the wound and to proliferate and differentiate into functional tissue cells to replace dead cells. Unfortunately, these same processes can result in the pathological scar formation that occurs in the liver as a result of the chronic inflammation associated with alcohol abuse or infection with hepatitis C. Chronic inflammation also increases the risk for tumorigenesis, which helps explain the increased incidence of hepatocellular cancer in drinking alcoholics and those infected with hepatitis B and C.³¹⁷

Following extensive tissue damage, endothelial cells produce VEGF and bFGF (along with a variety of other factors) for a long period. One effect is angiogenesis in the newly regenerating tissue through activating a growth and proliferative response in fibroblasts and smooth muscle cells. With unresolved inflammation, the production of growth factors also provides exactly the right mix of signaling molecules that can lead to tumorigenesis because of their effects on any adult stem cells that might be in the region. Although the presence of fibroblasts in tissue (as well as the recruitment of more through inflammation-mediated chemotaxis) is more than sufficient for the repair of relatively small regions of tissue damage, more extensive damage (and the resulting prolonged signaling) will include the recruitment and activation of adult stem cells in the area to proliferate and differentiate into functional tissue cells. Although this is necessary for closing and healing extensive wounds, adult stem cells can also be activated to proliferate with prolonged inflammatory signaling and without any overt wounding present,^{318,319} again, leading to a greater risk for tumorigenesis.

The proinflammatory cytokines also bind to receptors on nonimmune/ inflammatory cells in the vicinity and affect their function. Of the proinflammatory cytokines, IL-1 β , TNF- α , and IL-6 are of the greatest interest because of the near ubiquitous expression of their receptors and the detrimental effects of chronic II-1, IL-6, and TNF- α exposure on cellular function. The concept is best illustrated by the syndrome cachexia. Cachexia is mostly recognized as a wasting syndrome associated with end-stage cancer, but it also occurs with other chronic immune disorders. It is characterized by abnormal cellular metabolism, increased metabolic production of ROSs, insulin resistance, muscle and adipose tissue wasting, and extreme fatigue and is known to be a chronic inflammatory syndrome associated with the severe metabolic and immune stresses of the underlying disease conditions.^{320–324} Cachexia is characterized by moderate to high levels of the proinflammatory cytokines IL-1 β , IL-6, TNF- α , and IFN in serum as well as the proteolysis inducing factor, with each of the cytokines being directly involved in the pathogenesis of the syndrome. With the recognition that inflammatory signaling is of paramount importance to the pathology of this syndrome, and that metabolic disturbances, insulin resistance, and fatigue are common components of chronic inflammation, diabetes, and metabolic syndrome, it is certainly tempting to consider that cachexia is nothing more than the extreme end of a continuum of systemic inflammatory disorders: chronic metabolic disturbances \rightarrow metabolic syndrome \rightarrow diabetes \rightarrow cachexia.

To briefly summarize, proinflammatory signaling processes can lead to enhanced phagocytic activities, which can result in excessive ROS- and oxidant-mediated damage to membrane lipids, proteins, and DNA in adjacent cells. These types of damage can affect the already ROS-/calcium-altered signal transduction and regulatory pathways in detrimental ways. Add to these interactive effects the effects of proinflammatory cytokines and it is clear that unresolved inflammation can have profound effects on cellular function and subsequent risk for chronic disease. As discussed in detail throughout this book, these risks for chronic disease by inflammatory processes can be enhanced through nutritional insufficiencies, weight gain, and inactivity.

2.5 ATTENUATION OF INFLAMMATORY PROCESSES THROUGH DIET

2.5.1 GENERAL NUTRITION

The possible array of cellular dysfunctions caused by excess damage to individual cellular components, or to alterations in signaling pathways due to ROSs-calcium interactions caused by inflammatory responses, is nearly infinite; however, in terms of reducing inflammation-associated risk for chronic disease there appear to be only a few relevant issues. The first is that entry of a variety of inflammation-produced oxidants into adjacent cells provides a constant source of damaging agents. These are primarily H₂O₂, HOCl, and NO[•], of which H₂O₂ and NO[•] react with locally produced CO₂ and O₂ $\overline{\bullet}$ to produce the damaging oxidants CO₃ $\overline{\bullet}$, NO₂, and \bullet OH. Damage through oxidant-mediated reactions is a major source of cellular dysfunctions; therefore, optimizing the antioxidant function of cells should be a major focus for prevention. This concept needs to be tempered with the fact that it is the normal, metabolically produced O_2 (and CO_2) in the tissue cells that reacts with the inflammatory cell-produced oxidants and that metabolic regulation of ROSs through the expression of antioxidant enzymes does play a vital role in a wide array of normal cellular functions. Severely reducing endogenous ROSs through supplementation with antioxidant compounds should therefore produce cellular dysfunctions at some level and would be contraindicated. Exercise studies provide interesting examples of this. The well-known increase in mitochondrial biogenesis and insulin sensitivity in skeletal muscle caused by exercise in humans can, in some cases, be prevented by

supplementation with vitamin C (1 g) and vitamin E (400 IU).³²⁵ In a variety of other studies, antioxidant supplementation has been observed to reduce aerobic performance, reduce training-induced increase in VO2 max, and reduce maximum muscle force.^{326–329} Thus, optimizing antioxidant function should really mean maintaining optimal cellular antioxidant enzyme function and redox sensing function to ensure appropriate regulation of transcription activities. This should be achieved through eating a sufficient array of foods to provide the recommended dietary allowance (RDA) or adequate intake (AI) amounts of all nutrients. This may be especially relevant in light of the fact that although epidemiological evidence clearly indicates a decreasing risk for chronic diseases with an increasing intake of antioxidant nutrients and compounds in foods, clinical trials testing the effects of antioxidant supplementation on chronic diseases such as atherosclerosis and a variety of cancers rarely provide evidence of a positive benefit.³³⁰ In fact, negative effects such as increased deaths due to lung cancer with β -carotene or with vitamin C or E supplements and increased stomach cancer with α -tocopherol and/or β -carotene have been observed.331-337 These results might have been expected had the importance of regulating cellular redox states at the microenvironment level been considered; these concepts were, of course, not known at the time that any of the clinical trials were designed.

The concept of using supplements for prevention in a general population deserves a little discussion at this point. The concept of using supplements was developed on the basis of a large array of epidemiology-based research studies on diet and risk for disease that revealed an inverse association between antioxidant intake and risk for a variety of chronic diseases. A common conclusion of these studies was that antioxidant supplements should therefore reduce risk for the diseases as well, a conclusion that, on the basis of negative results from several recent large clinical trials, was probably inaccurate. Possibly, the observed associations between individual dietary antioxidants and preventive effects in epidemiological studies were more likely due to the wide range of nutritional and dietary statuses of the subjects. For instance, intakes of fruits and vegetables observed in American diets are traditionally low with less than 10% of Americans actually eating the recommended minimum number of servings of fruits and vegetables.³³⁸ This likely means that a majority of Americans consume insufficient amounts of many nutrients and beneficial phytochemicals because fruits and vegetables are by far the main food sources for these compounds. In fact, only about 50% of Americans meet their RDA intake recommendations for vitamin C^{339} Actually, the estimated average requirement (EAR) is the proper reference value for groups (obviously producing a lower percentage of potential insufficiencies), and according to the latest National Health and Nutrition Examination Survey (NHANES) data (2003-2006) intakes for vitamins C, D, and E are below the EAR for 25%, 70%, and 60% of Americans, respectively, and below the EAR for zinc (~8%), iron, copper, and selenium (~5% to 6%), as well as calcium (38%) and magnesium (45%).³⁴⁰ These specific vitamins and minerals were chosen to demonstrate the extremely poor dietary choices of Americans (insufficiencies of other nutrients also are common). They were chosen because they are antioxidants, components of antioxidant enzymes, or directly involved in calcium homeostasis, illustrating a potentially very important point. The point is as follows: while highly

useful statistical techniques are used to control for a variety of possible confounders in epidemiological studies, such as exercise history, ethanol intake, financial status, ethnicity, age, gender, and so on (and certainly not all relevant lifestyle factors are controlled for in each study), it is nearly impossible to test for and then control for each of the wide variety of existing nutritional insufficiencies. Because of the wide array of nutritional insufficiencies found in many of the populations studied, it is much more likely that the general epidemiological associations between antioxidant nutrient intake and prevention of disease are largely a reflection of a decreasing degree of nutritional/dietary insufficiency with increasing fruit/vegetable intake (correcting many different nutritional insufficiencies) and not necessarily a specific preventive effect of any individual antioxidant nutrient or compound. Although this does not mean that supplements are not useful to correct dietary insufficiencies and deficiencies, it does raise doubts whether supplementing beyond normal cellular requirements has any additional beneficial effects beyond that of normal nutrient function. In fact, the effect could be detrimental; the previously mentioned negative results of clinical trials with antioxidant supplements certainly attest to this. In a similar vein, nutritional supplements, other than correcting nutritional insufficiencies, have little to no positive effects on athletic performance either.^{341,342} This and the previously mentioned prevention of an exercise-induced mitochondrial biogenesis by the antioxidant vitamins C and E should certainly raise serious doubts about any potential use of antioxidant supplements beyond that of ensuring normal cellular nutritional requirements. For these reasons, the starting point of a dietary approach to reducing risk for disease through attenuating inflammation-associated events is a diet that provides RDA/AI amounts of all the necessary nutrients and not one that includes supplements. Considering the fact that a majority of Americans consume insufficient quantities of many different nutrients through less-than-ideal food selections, moderately large changes in diet are more than likely needed just to accomplish this.

2.5.2 CELLULAR SIGNALING

The second major area for reducing risk through attenuating inflammation is cellular signaling, which may represent what could be the single most effective area for overall risk reduction other than optimizing nutritional status. On the basis of the signaling processes that lead to the activation of various components of an inflammatory response, one general concept of cellular signaling emerges: the signal transduction pathways for activating the synthesis of proinflammatory cytokines converge on the p38–MAPK, JNK–MAPK, and NF $\kappa\beta$ pathways, leading to the activation of a large array of transcription factors. The anti-inflammatory effects of IL-10, lipoxin, resolvin, and protectin all function by inhibiting these same pathways, providing a clear indication of how to reduce proinflammatory signaling. Blocking the pathways would, of course, prevent an appropriate inflammatory response and cause severe problems in the event one is necessary; therefore, only a dampening of inflammatory responses to minimize the risk of an excessive inflammatory response is prudent.

A related signaling issue involves the various PGs, LTs, TXs, lipoxins, resolvins, protectins, and maresins. The lipid mediators made from AA tend to be

proinflammatory in nature, whereas those made from DHA and EPA tend to be either anti-inflammatory or proresolving in their effects on cell function. Again, they function through receptor-mediated effects that then alter the same signal transduction pathways. The production of anti-inflammatory lipoxins from AA complicates this issue (as discussed here); so it is not quite as simple as merely changing the availabilities of AA, EPA, and DHA by manipulating diet. On the other hand, attenuating absolute rates of synthesis of all the lipid mediators will occur primarily through dampening the activation of PLA₂, a relevant concept where initiating an inappropriate inflammatory response is a major issue. This will, not coincidently, also occur through giving appropriate attention to maintaining proper calcium/antioxidant (and nutrient) status and through attenuating stress-mediated activation of PKC and MAPK pathways.

2.5.2.1 Linoleic Acid, Linolenic Acid, Arachidonic Acid, Eicosapentaenoic Acid, and Docosahexaenoic Acid

An association between consumption of dietary DHA and EPA and reduced risk for inflammation-associated chronic disease has been known for many years, and these ω -3 fatty acids can be an important factor in helping to suppress inappropriate proinflammatory reactions, for two basic reasons. First, they are commonly known to compete with AA for incorporation into membrane phospholipids, which leads to a reduced production of PGE₂ and LTB₄ and an enhanced production of the (~1000-fold) less inflammatory 3-series PGs (PGE₃, PGA₃, and PGD₃), TXA₃/TXB₃, and LTB₅. More importantly, they are essential precursors for the D- and E-series resolvins and for protectin. From a clinical standpoint, in a variety of epidemiology-based studies consumption of these ω -3 fatty acids from marine sources appears to be strongly associated with lowered markers of inflammation (predominantly CRP) as well as a lowered risk for a variety of chronic diseases. While there is certainly sufficient biological evidence for the mechanisms of their resolving and anti-inflammatory functions; neither one, however, is an essential fatty acid.

As a nutrition issue, the essential fatty acids are linoleic acid (LA) (ω -6) with AIs of 12 and 17 g/day for adult women and men, respectively, and α -linolenic acid (ALA) (ω -3) with AIs of 1.1 and 1.6 g/day for adult women and men, respectively.³⁴³ Of these fatty acids, LA is synthesized into AA (also ω -6) and ALA is synthesized into EPA and DHA. Although the AI standards are designed (only) to prevent an essential fatty acid deficiency and, therefore, are properly considered to be adequate and not excessive, there has been a lot of recent attention on the idea of reducing the AI recommendations for LA intake and increasing the recommendations for ALA intake (to reduce inflammatory and enhance anti-inflammatory effects by reducing AA and increasing EPA and DHA levels in vivo). For such a recommendation to actually be beneficial, a simple change in substrate supply (reducing LA and increasing ALA) for AA, EPA, and DHA synthesis would need to have a greater ultimate effect on inflammatory signaling than the combined effects of all the regulatory mechanisms that govern the synthesis of AA, EPA, and DHA, and the synthesis of lipid mediators, as well as those of the regulatory mechanisms that are altered through their (activated) receptors. From an enzymology standpoint, this is simply not logical.

In a series of meta-analyses of human clinical trials where LA consumption was manipulated to determine the effect on various risks for coronary artery disease (CAD), no enhanced inflammatory functions were observed; in fact, higher LA intakes were actually associated with an anti-inflammatory effect (lower CRP), along with a lowered LDL cholesterol and a higher HDL to LDL ratio.^{344,345} The observed anti-inflammatory effects of higher LA ingestion are consistent with the anti-inflammatory effects of lipoxins, which, of course, are made from AA. The implication is that greater availability of AA for lipoxin synthesis due to greater intake of LA is important. In reality, however, AA synthesis is very tightly regulated and large changes in LA intake above AI amounts do not substantially change AA content of platelets and erythrocytes.³⁴⁶ Although this does not rule out the possibility that small changes in the AA content of cells can be beneficial, it is more likely that LA-enhanced diets alter biomarkers of inflammation through mechanisms that are independent of AA-based signaling. Because of an apparent anti-inflammatory effect of additional dietary LA, and the fact that the effect is more than likely due to effects other than those following eicosanoid signaling, it would be difficult to justify reducing the AI recommendations for LA on the grounds that lower AA will reduce inflammatory signaling, per se.

The nutrient-based story for ALA is a bit less clear but centers around a similar concept. Conversion rates of ALA to EPA are low with reported efficiencies anywhere from less than 1% to as high as 8% in men and from approximately 8% to as high as 21% in women, with the conversion rate observed in women consistently being $2.5 \times$ that of men, whereas conversion to DHA is much less than 1% for both.^{347,348} Because of the inefficient conversion to EPA and DHA, as well as the fact that ALA is a small minority fatty acid in most commonly available vegetable oils (<2% for all, except for ~50% in flax oil) and there is 9-10 g ALA/100 g of walnuts or butternuts³⁴⁹ (the two major ALA-containing non-oil foods), an awful lot of calories from oil or oily nuts would need to be consumed just to get the biological availability of a single gram of EPA (and of far less DHA); this is certainly not a prudent concept if obesity (or nutritional status) is any consideration. Assuming an average conversion of ALA to EPA of around 4% and 10% for men and women, respectively, about 25 g of ALA for men or 10 g of ALA for women would need to be consumed to produce a 1 g EPA benefit. Of course, substituting ALA-rich foods for ALA-poor foods to boost intake without increasing calories is possible. It takes about 50 g of walnuts (along with 300–350 kcal) to get 5 g of ALA, and a tablespoon of flax oil (only ~130 kcal) provides about 7 g of ALA; the taste, however, is, well, an acquired one and not really practical (or available) for everyone. These diet/ calorie issues compare to 45 or 90 kcal from a 5 or 10 g ALA supplement, probably a major reason why most clinical trials use supplements and not whole foods. Even combining supplements with dietary changes to produce a substantial increase in ALA intake still requires a moderately large dietary modification as well as reliance on single food sources, not necessarily a good dietary approach. For example, if half the dietary intake of lipid comes from walnuts and the diet is supplemented with walnut oil and flax oil, an increase in daily ALA intake of approximately 10 g will result.350 Such dietary changes to produce a potential 1 g EPA benefit are not very practical.

In addition to basic dietary considerations of practicality, the biological effects of ALA supplementation also need to be considered. EPA is preferentially incorporated into the membrane phospholipids of circulatory cells (platelets, neutrophils, and monocytes, certainly where you want them to be) and in clinical trials using ALA supplements (doses ranged from a minimum of ~5 g/day to as high as 26 g/day); EPA content was either unaltered or increased up to more than 300%, with inconsistent increases even at intakes at or slightly above 10 g/day44,344,347,348,351,352 and with little to no effect on DHA content. Thus, substantial increases in ALA intakes to a 10+ g/day range for what might be a negligible increase in EPA content of platelets, neutrophils, and monocytes and an inconsistently positive effect on risk for disease^{44,46} are simply not practical, nor are they recommended. This does not mean, however, that high-ALA foods should not be recommended as a component of a healthy diet; in fact, this would be prudent simply because only about 70% of adults in the United States meet the 1.2–1.7 g/day recommendation. In addition, simply switching from low-ALA oils to higher ALA oils for cooking is associated with a reduction in risk for CAD.353 Whether the seemingly beneficial effect of increased ALA intake in such studies is related to an increase in EPA/DHA synthesis is impossible to tell. With inconsistent changes in cellular EPA/DHA at high levels of ALA supplementation and little to no indication that any change results in faster resolution, it is doubtful. However, it certainly is possible that the switch is simply correcting an underlying ALA insufficiency. Thus, there is little justification for increasing the nutritional recommendation for ALA.

Fish oils that contain preformed EPA and DHA are the only realistic choice for a diet-based approach to modify EPA and DHA availability in order to avoid largescale dietary changes and efficiently correct a possible insufficiency in ALA intake. The basic questions become the following: how much EPA/DHA intake is necessary to have a beneficial effect? Further, what is the minimal amount of fish to eat in order to get an effective dose? Of the studies reviewed, 37-59 an intake in the range of 0.25–3 g/day of EPA reduces the risk for a variety of chronic diseases (especially CAD), whereas intakes in the higher ranges (2-4 g/day) reduce markers of inflammation and platelet adhesiveness, with the majority of the protective effect against cardiac deaths occurring in the lowest range (250-500 mg) of additional EPA intake. As might be expected, the studies that indicate a reduction in risk at the lowest intakes already have very low initial EPA/DHA (and ALA) consumption levels and along with that a high risk for CAD death. Based on many of these studies, the American Heart Association (AHA) recommends either an intake from supplements of approximately 1 g/day of a mix of both EPA and EPA or a weekly consumption of at least two servings of fish. Two 6 oz. servings of fish (salmon, ~1.7-4.5 g EPA/DHA/serving; herring, ~1.7 g EPA/DHA/serving) each week will provide a sufficient intake of these ω -3s to average well within the lowest beneficial range (with very little risk of exposure to mercury, polychlorinated biphenyls, and other toxic contaminants).³⁵⁴ This is a practical recommendation because it will provide the greatest risk reduction benefit with the least drastic change in diet. The AHA's recommendation for 1 g/day when using supplements is most likely because this is the dose in the most commonly available fish oil or EPA/DHA supplements and (similar to fish consumption) most of the risk reduction for cardiac death also occurs at this lower level of EPA/DHA

supplementation. Because the anti-inflammatory effects in the supplementation studies are observed at EPA intakes in the 3 to 4 g range, although the greatest increase in protection is achieved at intakes of 1 g or lower, it might appear that the majority of the benefits of EPA are most likely achieved through mechanisms that are not related to EPA-based lipid mediators and inflammatory signaling per se. This might be, however, because the type of data collected in many of the studies is highly limiting. Using CRP as a marker for inflammation is limiting because it is synthesized by the liver in response to IL-6 and not because of a local production of extremely short-lived resolvins and protectins from EPA elsewhere in the body. A lack of a change in a serum marker for inflammation does not necessarily indicate a lack of a local anti-inflammatory effect in areas of low-grade inflammation. In addition, EPA (and DHA) is known to be a PPARγ agonist^{355–357} and PPAR activation results in the inhibition of p38-MAPK, ERK-MAPK, and NFkβ-mediated proinflammatory signaling. Again, from a CAD standpoint (a very small and highly localized area of low-grade inflammatory response) it is doubtful that a systemic effect could be produced by a local activation of PPARs, but a local anti-inflammatory response would certainly occur. Whether the cardioprotective effects of low-dose EPA are due to PPAR-mediated effects, resolvin-mediated effects, correcting an underlying EPA/ALA insufficiency, or all three are unclear, but the protective effects do appear to be consistent.

For all the aforementioned reasons, the dietary recommendations specifically related to LA, ALA, EPA, and DHA are to include two servings of high-EPA/DHA fish each week and to eat at least one serving of a high-ALA food each day to ensure adequate intake of ALA, EPA, and DHA. There is little justification to be concerned about LA intakes simply because these are already relatively high in the American population and consuming even higher amounts are not proinflammatory. When this recommendation is considered in concert with the one recommending a proper diet that includes RDA/AI amounts of all other appropriate nutrients, the variety of cellular control mechanisms that depend on an adequate supply of nutrients should ensure that the signaling processes initiated by the lipid mediators will function optimally.

2.5.2.2 Proinflammatory Signal Transduction

The ability to ensure appropriate control of an inflammatory response also is very important. Whereas activation of the NF $\kappa\beta$ -, JNK–MAPK-, and p38–MAPK- associated signal transduction pathways is the key to inducing synthesis of the proinflammatory signals, suppressing these is central to anti-inflammatory and proresolution functions. Ideally, an appropriate array of cellular responses to the presence of pathogens, oxidizing agents, and cellular damage lead to an inflammatory response. This is then followed by a series of resolution events when the pathogens and/or damaging agents are removed. For the most part, this ordered process works very well, especially in the case of infectious agents. However, in the face of constant cellular trauma, especially that resulting from unavoidable environmental or endogenously produced toxic molecules, or constant physical trauma, an ordered resolution of inflammation is not going to happen. This is simply because all of the initiating DAMPs and other cellular stresses are continually present, resulting in a continuous production of proinflammatory signals, leading to continuous low-level inflammation. A solution to the dilemma that unavoidable chronic trauma leads to unavoidable inflammation-associated responses is to dampen the signaling processes that lead to proinflammatory signaling.

These desired effects can be accomplished by eating foods that contain compounds that can dampen the NF $\kappa\beta$, JNK–MAPK, and p38–MAPK signaling processes. This will produce a reduction in proinflammatory signaling in response to a given (chronic) initiating event. This reduction in signaling would, at the same time, be additive to any resolving mechanisms that are occurring (resulting from the chronic induction of COX-2 and mPGES1) because resolution effects are mediated through the same mechanisms. A large variety of phenolic compounds are capable of doing just that, and it may be no surprise that these compounds are found predominantly in fruits, vegetables, whole grains, teas, wine, and many other plantbased foods.

Common food sources³⁵⁸ that contain phenolic compounds (total phenolics listed high to low) are listed in Table 2.1.

From perusing the listed sources of phenolics, it may be no coincidence that the spices and foods in the high-end range for content of phenolic compounds are more common in the Mediterranean diet, whereas those in the low end of content are more common in the American diet. The reference to the Mediterranean diet, of course, is because of the well-known reduction in incidence of chronic diseases in cultures that follow the traditional diet compared to the westernized American diet.^{67,359} A seemingly similar dichotomy of choice is apparent with respect to alcoholic beverages, with wines (reds, ~200 mg/100 mL; rosés, ~80 mg/100 mL; white, ~30 mg/100 mL) being more commonly consumed with the Mediterranean diet and beers (~12-25 mg/100 mL) being more commonly consumed in the United States. Very large variations in the consumption of specific phenolic-rich foods between individuals and large variations in the content of phenolics in individual foods (depending on season, storage, and preparation) result in large variations in intake between individuals in the consumption of individual phenolics, varying by more than 100-fold. On the other hand, because phenolics are common to so many foods the average total consumption of phenolics is in the 0.5-2 g/day range with about 66% coming from flavonoids and stilbenoids and about 33% from phenolic acids. Most of the variability between individuals depends on the differences in consumption of flavonoids in vegetables, cooking oils, and fruits (including juices) and of phenolic acids in coffee, tea, and chocolate,^{360,361} with the traditional Mediterranean diet being on the high end (>1.3 g/day) and the American experience tending to be on the lower end (<0.5 g/day) of phenolic consumption.^{362,363}

Although many of the phenolic compounds found in the aforementioned foods have antioxidant properties, it is far more likely that their preventive properties are due to their ability to modify components of signal transduction pathways. As an antioxidant only, they would be susceptible to the same mechanistic limitations in prevention that other antioxidant supplements might have, as suggested at the beginning of Section 2.5.1 and by others,³⁶¹ that is, overwhelming a cell with antioxidants interferes with the regulation of cellular redox states at the microenvironment level. In addition, because of a bioavailability of 2%–20% and a half-life of anywhere

TABLE 2.1Phenolic Content of Some Common Foods

	Herbs—Spices
>5000 mg/100 g	Cloves, Ceylon cinnamon, dried pot marjoram, dried spearmint, dried wild marjoram oregano
>3000 mg/100 g	Dried summer savory, dried sweet basil, dried sweet bay, dried marjoram, capers
>1000 mg/100 g	Dried common sage, caraway, dried rosemary, dried coriander, dried turmeric, dried cumin, nutmeg, dried winter savory, dried common thyme, star anise, dried parsley, dried dill, curry, black pepper
	Dried Fruits
~1000 mg/100 g	Prunes, raisins, figs
	Fruits
>500 mg/100 g	Black elderberry, black chokeberry, skunk currant, black raspberry, black currant, Canada blueberry, gooseberry
>200 mg/100 g	Plum, lowbush blueberry, orange, American cranberry, sour cherry, strawberry, peach
>50 mg/100 g	Grapes, apple, kiwi, banana, pineapple, mango, pear, star fruit, guava
	Nuts
>1000 mg/100 g	Chestnut, pecan, walnut, pistachio
>200 mg/100 g	Hazelnut, almond, Brazil nut, cashew
>50 mg/100 g	Macadamia nut, dehulled almond, dehulled peanut
	Beans
>500 mg/100 g	Raw whole adzuki bean, raw whole lentils, raw dehulled black bean, raw whole bean, raw whole broad bean
10–150 mg/100 g	Raw whole white bean, raw whole dried pea, raw whole climbing bean, raw whole lima bean
	Vegetables
>500 mg/100 g	Swiss chard
>200 mg/100 g	Raw dandelion, red cabbage, raw green bean, chili pepper, spinach, sweet pepper, raw Brussels sprouts, broccoli, cauliflower
<100 mg/100 g	Raw green cabbage, sauerkraut, tomato, squash, zucchini, lettuce, onion, carrot, asparagus, sweet potato, potato, celery
	Oils
~500 mg/100 g	Peanut oil
~20 mg/100 g	Virgin/refined olive oil
~20 mg/100 g	Canola/rape
~2 mg/100 g	Sesame oil
	Grains
700+ mg/100 g	Whole-wheat buckwheat flour, wheat germ
~180 mg/100 g	Whole-grain hard wheat, corn flour
~70–100 mg/100 g	Whole-grain common flour, oat flour, corn flour, rice flour
from 1 to 2 hours^{360,361} for most flavonoids, antioxidant effects per se are likely to be minimal at best and relatively transient in nature. For example, curcumin (from turmeric), a more effective antioxidant than either resveratrol or quercetin, can function as an antioxidant in serum at concentrations of 1–50 μ M in vivo, a level that requires the consumption of more than 8 g/day to reach the 1+ μ M level.³⁶⁴ With a maximum observed dietary consumption of curcumin of approximately 1.5 g/day in some Asian communities³⁶⁵ (higher than the dietary consumption of resveratrol or quercetin), it is doubtful that the antioxidant properties of curcumin (or other single phenolics) significantly contribute to a diet-based protective effect. This does not rule out the possibility that high-dose supplements might achieve a sufficient concentration in vivo to have antioxidant properties but does make it unlikely that the epidemiological associations between dietary consumption of phenolics and prevention are due specifically to their antioxidant effects.

As modifiers of specific components of signal transduction pathways, however, the transient effects of dietary phenolics in vivo should be a benefit. This would certainly avoid constant alterations of regulatory pathways that might prove to be highly problematic, as indicated by the high risk for cancers when mutations render various components of the MAPK pathways constitutively active. In addition, because the molecular effects of most phenolics are targeted to a very narrow range of proteins, they are unlikely to have global detrimental effects on cell function. Large doses of individual phenolics from supplements do not have such a favorable metabolic profile and if present in large quantities for a prolonged time adverse effects might occur, such as the hepatotoxicity and altered reproductive function described in case studies in which individuals consumed supplements containing polyphenol-enriched tea and soy isoflavones.³⁶¹

In terms of diet-based risk reduction, it is also likely that phenolics do not necessarily work in isolation; rather, it is the additive and synergistic effects of different phenolics that may account for their apparent overall benefit. From the results of clinical trials, somewhat inconsistent positive effects have been observed for single phytochemicals and for phenolic extracts from single sources or single foods. Using 0.5-9 g/day of green tea extracts in a supplement produced no change in CRP or IL-6 in diabetic patients, an effect (or rather, lack thereof) also seen with 166 mg quercetin plus 133 mg vitamin C in arthritis patients.^{366–368} Similar negative results have been observed with drinking up to 900 mL/day of black tea and green tea, 800 mg of pomegranate extract, or diets rich in berries and apples.^{369–372} On the other hand, anthocyanin from blueberries (300 mg/day; equivalent to ~1 cup or ~150 g blueberries), 280 g cherries/day (equivalent to ~1.5 cups; also high in anthocyanin), grape seed extract (600 mg/day), and concentrated grape juice supplements (100 mL; 640 mg total polyphenols) have been observed to reduce CRP and/or MCP-1 in serum.^{373–376} Obviously, in some instances the differing results could be due to the differences in the total doses used. In other cases, they might be due to the different arrays of phenolics supplied with the different foods or supplements, or simply due to the fact that some studies used nondiseased normal subjects while many others have used diabetics, cancer patients, HIV-infected individuals, or other *medically* challenged subjects, each with their own possible differences. Overall, it is difficult to determine from clinical studies whether there is a consistent anti-inflammatory

effect from any individual phenolic or phenolic mixture–based supplement simply because there have not been multiple trials performed with all likely phenolics in large heterogeneous groups of subjects (and, because there are thousands of different phenolic compounds, this would actually be impossible). Based on somewhat limited evidence, however, some whole food–based phenolic mixtures or extracts from whole foods, such as grape, blueberry, and cherry sources, do appear to have a beneficial anti-inflammatory effect at concentrations that can be achieved from dietary consumption of these fruits.

Translating this information directly into a dietary recommendation, however, could be awkward. In spite of the apparent success of some supplements or single food sources of phenolics, mixes of different phenolics from several food sources are more likely to be relevant for a preventive diet. This is simply because reasonably large servings of single foods have been found necessary to produce a beneficial effect in clinical trials, and it is doubtful that such servings of the same foods are an everyday occurrence for very large numbers of people in the general population. With a 1.5–2 cup recommendation for fruit consumption (teenagers through adults; U.S. Department of Agriculture), consuming 1.5 cups of cherries each day to get a clinical dose of anthocyanins would mean the entire day's fruit recommendation would come from a single fruit, a concept that is not consistent with current recommendations for consuming a variety of fruits and vegetables in order to meet requirements for all nutrients. It is also much more likely that the observed benefits of phenolics in epidemiological studies are due to the additive and possibly synergistic effects that many different phenolics coming from many different foods can have on cell function, specifically signal transduction pathways.

Because of the technical nature of the methodologies, research on the effects of various phenolics on signal transduction pathways is pretty much limited to in vitro or ex vivo studies of human or mammalian cells. Some of the more commonly known phenolics that have been studied extensively for their ability to modify signal transduction pathways include *flavonoids* (food sources listed from high to low): quercetin (black elderberry, dark chocolate, oregano, and capers and some in vinegar, tomatoes, shallots, and red onions), apigenin (extra virgin olive oil [EVOO], Welsh onion, Italian oregano, and pistachios), epigallocatechin-3-gallate (EGCG) (green tea, oolong tea, black tea, pecan, hazelnut, pistachio, and banana), theaflavin (black tea), genistein (soy products), anthocyanins (blueberries, strawberries, cherries, and common black beans), and kaempferol (capers, cloves, black tea, broccoli, apples, cherry tomatoes, and onions); phenolic acids: ellagic acid (chestnut, Japanese walnut, walnut, blackberry, black raspberry, and pomegranate juice), capsaicin (hot peppers, Hungarian peppers, and sweet peppers), curcumin (turmeric and curry powder), and caffeic acids (coffee); and stilbenoids: resveratrol (red wine, red/black grapes, lingonberry, European cranberry, vinegar, and roasted peanuts); among many others.^{359,364,377–388}

In both cell culture and animal models, various inflammatory-response proteins have been reported to be affected by treatment with phenolic compounds. The synthesis of a variety of proinflammatory signaling molecules, including IL-6, IL-1 β , TNF- α , and IFN, has been observed to be suppressed by quercetin, apigenin, genistein, green tea polyphenols, luteolin, anthocyanins, and curcumin. In response to a reduction in synthesis of these signaling molecules, a reduction in expression of iNOS, COX-2, ICAM, and MCP-1 would be expected. Of course, these changes have also been observed with the same phenolics.^{364,377,380,384,386} Thus, phenolics that are found in many common consumed foods are capable of suppressing inflammatory signaling, a suppression of signaling that can be mediated at several different points along an activation \rightarrow signal transduction \rightarrow expression pathway.

From the standpoint of initiating synthesis of inflammatory cytokines, it is the activation of PRRs by PAMPs and DAMPs that initiates the process, and expression is then mediated through the activation of the NF $\kappa\beta$ and MAPK signaling pathways. Recent experiments have revealed that activation of TLR4 can be suppressed by curcumin and sulforaphane (cruciferous vegetables) and NOD1/2 activation can be inhibited by curcumin. In addition, resveratrol, EGCG, luteolin, and quercetin all directly inhibit TBK1 (necessary for IFN synthesis), whereas resveratrol inhibits TRIF to suppress both the MyD88 and the non-MyD88 pathways of TLR signaling, resulting in a suppression of both NF $\kappa\beta$ - and MAPK-mediated responses.³⁸⁰

Activation of NF $\kappa\beta$ signaling also can be suppressed through a variety of mechanisms including inhibition of IKK activity, which then suppresses I $\kappa\beta$ release and its subsequent degradation; this prevents free NF $\kappa\beta$ from binding in the nucleus. EGCG from green tea and theaflavins from black tea inhibit NF $\kappa\beta$ signaling through this mechanism, whereas quercetin, curcumin, kaempferol, and lycopene can inhibit NF $\kappa\beta$ signaling by suppressing nuclear translocation of p50/p65. Suppression of NF $\kappa\beta$ signaling has also been observed with apigenin, morin (strawberries), anthocyanins, and procyanidins (peaches, plums, apples, and nectarines) through a variety of related mechanisms, including suppression of phosphorylation of p65 and of DNA binding of NF $\kappa\beta$, although in many studies the specific molecular mechanism of inhibition is not always clear.^{364,377,384,386} Regardless of the exact mechanism, subsequent synthesis of IL-1, IL-6, TNF- α , and IFN is reduced.

Because MAPK pathways are activated via PRRs and are integral to activating the expression of proinflammatory cytokines, these pathways are prime targets for suppressing inflammatory signaling by phenolics as well. Apigenin, luteolin, quercetin, resveratrol, EGCG, and kaempferol can suppress ERK1/2-MAPK, JNK1/2–MAPK, and p38–MAPK pathways. Cyanidins can suppress ERK–MAPK activation, and EGCG has been observed to inhibit the activation of ERK-MAPK as well. In addition to suppressing the different MAPK pathways, curcumin has also been observed to inhibit PKC, a calcium-activated enzyme that can activate various components of the MAPK (and other) pathways.^{364,377,384,386} From these observed effects, phenolic compounds exert their suppression activities through a variety of different mechanisms that suppress the activation of MAPK and NFkβ pathways. Although not detailed here, different cell types used in different studies tend to yield differing results. Some cell types, such as respiratory epithelial cells, exhibit the suppression of all MAPK pathways as well as NFκβ pathways by single phenolics. In activated macrophages, only one or two pathways are inhibited by the same phenolics, with similar variations in response observed for a variety of different phenolics.^{364,376,377,384,386} The differences in effect indicate that different cells have different susceptibilities to inhibition by different phenolics, and because of this it is doubtful that single phenolic supplements will have the desired overall preventive effects. In addition, large doses of single or single-source phenolics can produce a prolonged

suppression of signaling or even a toxic effect; hence, a dietary recommendation designed to produce a suppression of proinflammatory signaling should include a variety of fruit and vegetable sources.

The suppressive effects on proinflammatory signaling by phenolics are not limited to NFκβ and the various MAPK pathways; a variety of phenolics also appear to affect the production of lipid mediators. EGCG, quercetin, and kaempferol have been observed to inhibit COX-1 and COX-2 and LOX with subsequent decreases in PGE₂ production in a variety of cell culture studies. Phenolics from EVOO also have the same properties. Tyrosol, hydroxytyrosol, apigenin, luteolin, and oleuropein compounds are found in EVOO, and all have been documented to inhibit the production of PGE₂, LTB₄, and TXA₂.^{377,389,390} While luteolin and apigenin reduce the production of proinflammatory PGs, TXs, and LTs by suppressing the synthesis of COX-2 (by suppressing NF $\kappa\beta$ and MAPKs, as discussed earlier), the other phenolic compounds suppress signaling by directly inhibiting COX or LOX. Regardless of the mechanism, however, in many clinical trials using virgin olive oil and EVOO in the 25-50 g/day range (certainly a dietary possibility if olive oil is the only oil used in cooking), olive oil is documented to have consistent anti-inflammatory effects. Further, the extra virgin version appears to have these beneficial effects at a lower range of intake, which is more than likely because of the much higher content of these phenolics in EVOO.^{377,391,392} Then, from a dietary perspective recommending the consumption of approximately 25 g of extra virgin oil to get an array of antiinflammatory phenolics when the caloric cost is approximately 225 kcal (or about 10% of total caloric intake) is neither harmful nor impractical. Such a recommendation also happens to be well within acceptable dietary guidelines.

In addition to the anti-inflammatory benefits obtained through the suppression of signaling discussed earlier, an increased expression of various antioxidant and redox-control enzymes would also be highly beneficial. Increasing synthesis of SOD, CAT, GPX, and GSR would be relevant for increasing protection from additional ROSs/RNSs, whereas enhanced synthesis of thioredoxin and peroxiredoxin would enhance the ability to maintain appropriate regulatory control of redox status.

The role of SOD is to catalyze the reaction $O_2 \cdot + O_2 \cdot \to H_2O_2 + O_2$ while CAT catalyzes the reaction $H_2O_2 + H_2O_2 \rightarrow O_2 + 2H_2O$ at rates far faster than the spontaneous dismutation rates that can occur in vivo. GSR and GPX work in tandem with GSH to eliminate H_2O_2 through the GPX-catalyzed reaction $2GSH + H_2O_2 \rightarrow$ $GSSG + 2H_2O$. The GSR then reduces the GSSG back to 2 GSH using NADPH as the electron donor. The benefit of these enzyme-catalyzed reactions is that they minimize the concentrations of both O_2 and H_2O_2 , which in turn minimizes the production of the highly reactive •OH. Of the eight known forms of GPX, GPX1 in the cytosol of mammalian cells reduces free H₂O₂, whereas GPX4 associates with cellular membranes and reduces lipid hydroperoxides. In concert, these antioxidant systems ultimately reduce damage to proteins, lipids, and DNA caused by ROSs.^{278,280} Of the three isozymes of SOD, the cytosolic copper-zinc form of SOD (CuZn SOD-SOD: cytosol, nucleus, and mitochondrial membrane) and the extracellular copper-zinc form (EC SOD—SOD) appear to be less inducible than the manganese form (Mn SOD—SOD: mitochondrial matrix), even though NF $\kappa\beta$ response elements are found in the promoter region of all three sod genes. This is most likely due to the fact that the sod2 promoter contains AP-1 as well as ARE binding elements.³⁹³ An increase in SOD activities mediated through NF $\kappa\beta$ is a logical response to proinflammatory signaling in order to protect cells from the inevitable increase in ROS production.

Maintaining the ability to regulate redox status at the microenvironment level is very important for ensuring proper control of signal transduction pathways. This occurs through the redox-sensing enzyme systems as well as through endogenous antioxidant enzyme activity. With this in mind, compounds that can enhance the synthesis of a variety of antioxidant enzymes through Nrf2 activation will protect cells against the damaging effects of additional inflammation-produced oxidants. Increasing antioxidant enzyme capacity should also reduce the ability of incoming oxidants to disturb redox-based regulation of signal transduction pathways. An induction of both thioredoxin and peroxiredoxin through Nrf2 activation will certainly contribute to ensuring proper regulatory control. The expression of CAT, GSR, thioredoxin, peroxiredoxin, HO-1, and glutamate cysteine ligase (GCL) (rate limiting for GSH synthesis) as well as other protective enzymes is also induced through Nrf2 activation.^{383,394–396} In addition, a variety of endogenous regulatory factors are capable of activating Nrf2, including PGJ₂; PKC; MAPKs (ERK, p38, and JNK); PI3K/PKB; and even Ca²⁺ and ROSs through PKC- and redox-mediated regulation of MAPKs, respectively.^{382,395} Thus, activating the expression of a variety of protective enzymes through the activation of Nrf2 is a more general response to cell stress than increasing only SOD through specific proinflammatory stimuli. It is through these stress-response genes that polyphenols can function to enhance protection from inflammation-produced ROSs as well as a variety of other damaging agents. The induction of enzymes such as GST, UDPGT, and NADPH:quinone oxidoreductase-1 is much more applicable to the prevention of cancer and will be discussed in Chapter 6, Sections 6.3.3.3 and 6.3.5.2.

A number of phenolics have been documented to activate Nrf2 binding to ARE and induce the expression of antioxidant and other protective proteins. Quercetin, resveratrol, diallyl sulfide, *s*-allyl cysteine, lycopene, EGCG, curcumin, and sulforaphane have all been documented to activate Nrf2 and to induce the synthesis of Nrf2/ ARE-inducible proteins, including GST, GSH, HO-1, SOD, CAT, GCL, thioredoxin, peroxiredoxin, and UDPGT.³⁹⁷⁻⁴¹⁰

The different phenolics appear to act at different points within the (albeit brief) Nrf2 signaling pathway. Nrf2 normally forms complexes with Keap1 in the cytosol, which not only prevents it from entering the nucleus but also may target it for degradation. Phosphorylation of Nrf2 by protein kinases can disassociate Nrf2 from the complex to allow it to migrate into the nucleus and bind to ARE transactivation binding sites.^{383,411} Some phenolics activate protein kinases such as PKC, PKB, ERK–MAPK, JNK–MAPK, and p38–MAPK, which in turn can phosphorylate Nrf2, whereas others appear to covalently modify susceptible thiols on Keap1. Modification of susceptible thiols on Keap1 by phytochemicals reduces ubiquitination (and subsequent degradation) of Nrf2 while stabilizing the Keap1–Nrf2 complex; this allows newly synthesized Nrf2 to remain unbound to Keap1 and enter the nucleus. These results indicate that modification of Keap1 (see later) as well as activity of various components of the MAPK pathways can activate Nrf2 and lead to the expression of the array of protective

enzymes. They also raise the conundrum of how a variety of phenolics can suppress the activation of the MAPK pathways to produce an anti-inflammatory effect while others activate p38–MAPK, ERK–MAPK, and JNK–MAPK themselves to produce a protective effect. Because this has not been specifically addressed in the research literature, conjecture will have to suffice: inhibiting the ability of the pathways to be activated by DAMPs, PAMPs, and stress should dampen the proinflammatory response to reduce the risk of an inappropriately intense response regardless of the initiator. In spite of the fact that these dampening effects would be transient, there would be a general overall reduction in the proinflammatory response. On the other hand, transiently activating the specific MAPKs may be sufficient to activate the Nrf2–Keap1 complexes to disassociate and enhance binding to ARE without them being sufficiently active to initiate a stress-induced proinflammatory response. This, of course, remains to be determined.

One potential issue with many of these cell culture–based studies is that relatively large concentrations of phenolics are used (10–50 μ M in most cases) in order to ensure that measureable differences can be seen after several hours or a day (or more) of treatment. This might make it difficult to extrapolate the data to humans because consumption of up to 10 g/day (or more) of the individual phenolics is necessary to produce such a concentration in vivo, an amount unlikely to be obtained from anything but the most extreme diet. They do reveal, however, that mixtures of phenolics can alter the activity of various signal transduction pathways to affect the expression of many different proteins.

The precise mechanism through which the phenolics work and at what step or steps of the signal transduction–expression pathway they have their effect are extremely difficult to determine from these endpoint studies. In many respects, it is simply impractical to measure the activation/oxidation state of every single kinase involved in a signal transduction pathway, as well as the binding of various relevant transcription factors to the promoter regions of the relevant genes, in addition to determining the expression of all the relevant proteins in a single experiment with one phenolic. Further, including all the complex interactions between the different pathways just elevates the impractical to the seemingly impossible (or absurd). Because phenolics happen to be metabolized fairly quickly, it is clear that their effects also will be relatively transient. Small, transient changes in activity of the different components of the signaling pathways are extremely difficult to measure. In spite of this, it is through studies that are designed to determine the time course of location-specific alterations in each of the pathways that the specific mechanisms through which the various phenolics produce their protective effects would be revealed.

Earlier in Section 2.5.2.2, it was stated that phenolics were unlikely to have an effect based on their antioxidant properties because of a requirement for micromolar concentrations of phenolics to act as antioxidant scavengers in serum. Most antioxidant function tests are based on the oxidation of target molecules in solution in vitro or in biological fluids and require moderately large concentrations of phenolics to measure the antioxidant effect. This is compounded by the fact that nutrient antioxidants are not only more effective antioxidants but also typically as much as two to three orders of magnitude greater in concentration in vivo (e.g., $50-100 \mu M$ for vitamin C) than the tested polyphenols; there simply needs to be a relatively

high concentration of phenolics in order to observe an antioxidant effect above background. This does not rule out small, transient antioxidant effects in vivo. It just means that they may be very difficult to measure; their relative contribution to total antioxidant function would still, however, be very low (less than 1% to 2%). Thus, the antioxidant abilities of phenolics may not be their most relevant feature.

Because $2O_2$ spontaneously dismutates extremely quickly into the more stable H_2O_2 and active SOD catalyzes the same reaction, H_2O_2 is the predominant nucleophilic biological oxidant. Antioxidant molecules (and antioxidant enzymes such as CAT) can, of course, reduce the peroxide to water (or rather water and oxygen). The ability of flavonoid phenolics to scavenge hydroxyl radicals, peroxyl radicals, and peroxynitrite radicals is predominantly conferred through the electron (and hydrogen)-donating capabilities of the hydroxyl groups on the B-ring.⁴¹² However, because these properties are unlikely to provide the major protective function in vivo other properties must exist. With catechol-type polyphenols, it is far more likely that autoxidation of the hydroxyl groups occurs, promoting a rearrangement of electrons within the ring to produce a quinone, which then participates in conjugation reactions with susceptible thiols such as those on Keapl.^{413,414} As already mentioned, a wide array of phenolics can covalently modify the susceptible thiols in Keap1, leading to reduced ubiquitination and degradation of Nrf2 and subsequent entry of Nrf2 into the nucleus to bind to ARE/ERE to produce its characteristic protective effects. In addition, the quinone structure can be easily reduced and H₂O₂ is produced during the redox resonance between the two states.^{395,413} H₂O₂ is a potent and common oxidant for thiols, which leads to dissociation of the Keap1-Nrf2 complex and subsequent Nrf2-ARE binding in the nucleus. Thus, it is the prooxidant and autoxidizing character of dietary phenolics that gives them their protective effects and not their antioxidant properties. With easily oxidizable thiols in Prx and Trx one could imagine that H_2O_2 generated by phenolics can exert their effects there as well, creating small, localized changes in the redox state that transiently affect components of signal transduction pathways. The transient nature of these phenolic-mediated oxidation events produces the protection effects, effects that are mediated through essentially the same signaling pathways that when continuously activated lead to detrimental effects on cell function.

Before moving on to exercise-based prevention, this is a good place for a brief discussion on HO-1 mentioned earlier in this section. The constitutive enzyme HO-2 and the inducible enzyme HO-1 catalyze the rate-limiting step in heme degradation to produce free iron (which immediately stimulates ferritin synthesis), bilirubin, and carbon monoxide (CO), with the constitutive form being expressed almost universally (with much more in spleen, liver, and bone marrow) and the inducible HO-1 being at very low to undetectable levels.^{415,416} HO is well known as being necessary for scavenging iron, so it can be recycled into new heme proteins (hemoglobin, myoglobin, respiratory chain proteins, and cytochrome P450 enzymes, among many others). As a component of a stress response, the HO-1 enzyme is highly inducible through Nfr2/ARE activation; inducing this enzyme would make sense considering the potential for oxidative damage to inactivate heme proteins (among many others) and produce a requirement for an additional capacity for scavenging iron. As a consequence of increased HO-1 activity CO production is enhanced, and this

molecule appears to be important in inflammatory responses. In cell culture models, macrophages with induced HO-1 release less TNF- α in response to LPS than noninduced cells, an effect that can be produced with externally supplied CO in noninduced macrophages. This indicates that CO is the suppressing molecule in this model. In similar models, HO-1 induction also inhibits the synthesis of IL-1, IL-6, ICAM, VCAM, and E-selectin.^{416–420} In all these studies, CO appears to affect one or another component of the different MAPK pathways to decrease proinflammatory responses to inflammatory stimuli, in a manner reminiscent of the redox effects on these pathways; because CO does not bind to any of the proteins, it appears to affect the activity of the pathways through upstream events. In any event, as a component of a generalized stress response in cells, induction of HO-1 appears to have antiinflammatory effects.

With these concepts in mind, the previous interpretation of prevention-based studies must still stand: it is the additive and synergistic effects from an array of phenolics that are more than likely responsible for their observed protective effects on inflammation and disease as illustrated in Figure 2.6. Any dietary recommendations should therefore take into account the phenolic content of many different foods and attempt to formulate a recommendation based on both the content of phenolics and their mechanisms of action.

2.6 ATTENUATION OF INFLAMMATORY PROCESSES THROUGH EXERCISE

An exercise-induced enhancement in the production of ROSs in skeletal muscle has been recognized for many years with the enhanced ROS production being thought to be the source of muscle damage.^{421,422} Many athletes took to heart the message and started consuming antioxidants, hoping for a benefit; of course, the previously documented ill effects of supplementation on performance were largely unknown at the time. Most likely this was simply a result of the entrenched idea that radicals are bad because they cause damage and that the more recent recognition that ROSs are vital regulators of signal transduction pathways was not well recognized by scientists until the past decade. Much of the earlier data on inflammatory signaling focused on damage-associated inflammatory cytokines produced following downhill running, unaccustomed strenuous exercise, or exhaustive exercise (exercise models that tend to produce some degree of muscle damage and often result in delayed-onset muscle soreness). As might be expected, increases in proinflammatory cytokines, including IL-1 β , IL-6, IL-8, TNF- α , CRP, as well as creatine kinase activity in serum (a common marker for muscle damage) and markers of ROSs, have been identified in serum under these conditions. In addition, there is a tendency for an increase in soluble TNF receptors as well as in soluble IL-1 receptor and soluble IL-1 receptor antagonist.^{70,423-426} The enhanced synthesis of various cytokines in response to the highly stressful exercises simply reinforced the idea that it was ROS that caused the damage and that a proinflammatory response resulted from the damage.

With moderately strenuous concentric exercise, there is also a production of cytokines; however, the responses are not the same. Damaging or exhaustive exercise leads to elevated TNF- α and IL-1 β that can last for 24 hours or more; with more prolonged nonexhaustive exercise, the increase in IL-1 and TNF- α does not occur, whereas the increases in IL-6, IL-10, sIL-1ra, sIL-1r, and sTNFr are far more pronounced and transient in comparison to those of exhaustive exercise.^{68,70,72,426,427} The anti-inflammatory effects of IL-6 are realized through the IL-6-mediated induction of sIL-1r, sIL-1ra, and IL-10. IL-10 suppresses NFκβ and MAPKs, increases SOCS signaling, and stimulates nonphlogistic phagocytosis. There is a large increase in soluble receptors for both IL-1 and TNF; they exert their anti-inflammatory effects by binding to circulating IL-1 α , IL-1 β , and TNF- α while sIL-1ra attenuates the ability of circulating IL-1 α and IL-1 β to bind to cellular receptors, thus suppressing proinflammatory cellular responses. A moderate-intensity cycling workload that can elevate IL-6 levels is approximately 50% of the maximum watts of power output, which is approximately a 6 MET level of intensity, a minimum level in which significant increases in IL-6 occur with 1 hour of exercise. It is also important to note that changes in IL-6 are not observed with exercise that utilizes small muscle groups, such as the upper arms, indicating that large muscle-group activities of a minimum intensity of 6 METs, sustained for approximately 1 hour, might be essential for the IL-6-mediated anti-inflammatory effects of exercise.73,428,429

As discussed previously, TNF- α activates the synthesis of IL-6 as part of a coordinated inflammatory response. A crucial difference with regard to exercise is that the production of IL-6 by muscles occurs without being activated by TNF- α . This means that the exponential (and transient) increase in IL-6 is independent of TNF- α signaling and illustrates that metabolic stress can activate signaling pathways for cytokine expression, just not for all of them. There clearly is a difference between the activation profile of cytokines due to metabolic stress and that due to activation through an elevation in DAMPs (as would be expected from damaging or exhaustive exercise) or PAMPs. This reinforces the concept that even though the various cytokines are synthesized in response to the activation of transcription factors via oxidant-mediated activation of the MAPK pathways and NFκβ pathways, the presence or absence of PAMPs and/or DAMPs makes a significant difference in the pattern of activation, resulting in different patterns of cytokine expression. In response to the highly elevated IL-6, the other anti-inflammatory molecules are subsequently expressed, with the degree of increase in expression of all anti-inflammatory molecules being dependent on the intensity and duration of exercise and the amount of muscle mass activated by the exercise. Thus, prolonged, moderately stressful exercise is anti-inflammatory while damage and pathogenic stress is proinflammatory, even though the same signaling pathways are involved.

Another issue relating to IL-6 release from skeletal muscle is the association between this cytokine and glycogen content of muscle and liver. IL-6 strongly activates glycogenolysis in liver, an effect that is greatly enhanced through increased IL-6 release from glycogen-depleted exercising muscles in comparison with nondepleted muscles.^{73,425,430,431} Activation of glycogen breakdown and glucose release by the liver via muscle-derived IL-6 is a major reason why IL-6 is considered by some to be a *glucoregulatory* cytokine. The concept raises some interesting interpretations with respect to IL-6 release from muscle and how this might relate to exercise-based anti-inflammatory effects. For instance, if glycogen levels need to decline

to produce an IL-6 response, this will not happen during short-term low-intensity physical activity and especially not if the low-intensity activity is performed in a fed or an immediate postprandial state. Thus, recommendations for physical activity to be an active anti-inflammatory agent through induction of IL-6 must be for a sufficient duration and intensity to provoke an IL-6 response. Unfortunately, clinical evidence of a consistent reduction in inflammation signaling by exercise is in short supply.

In many studies relating to inflammatory risk for chronic disease, CRP has been used as the marker of choice because of its ease of measurement as well as its documented inverse association with risk for chronic disease. Because the potential use of exercise as an anti-inflammatory agent is a relatively recent development, only a moderate number of studies that delineate the forms of exercise that can reduce CRP have been conducted. In various trials, the results that have been obtained range from no association between exercise and CRP to significant reductions in CRP due to exercise.^{69,432-440} In these studies, however, it is the longer duration exercise of at least 45 min/day at a frequency of 5 day/week or more that appears to produce a chronic reduction in CRP; in studies using 30-minute durations of exercise or resistance training, CRP was largely unchanged. These results are somewhat consistent with an earlier analysis of NHANES III data that indicated a dose effect on elevated CRP with odds ratios of 0.98, 0.85, and 0.53 for light, moderate, and vigorous exercise, respectively.⁴⁴¹ Interestingly, in some of the exercise-intervention studies in which there were consistent reductions in CRP, there was also a reduction in adiposity, not necessarily an unexpected result with the addition of 45-60minutes of exercise on most days of the week. The weight-loss effects are consistent enough that it is difficult to disassociate the effects of weight loss per se from those of exercise on reducing CRP. With IL-6 synthesis by myocytes being related to intensity, duration, and glycogen content, it is possible that low glycogen levels that would be associated with weight loss may contribute to the exercise effect. Obviously, more studies are required to sort this out. It may even be possible that simply performing exercise in a fasted state as opposed to a postprandial state may produce an elevated IL-6 response at the same exercise load without actually being in a negative calorie balance; this is sort of like being able to have your cake and eat it too.

In addition to the anti-inflammatory effects of exercise through mediation by IL-6, an increase in antioxidant enzyme activities is also relevant from a prevention standpoint. Today, it is well recognized in the research literature that the transient alterations in free radical production in skeletal muscle as a result of the metabolic stress of exercise are responsible for the anti-inflammatory response as well as the hypertrophy and mitochondrial biogenesis adaptations observed with different types of repeated exercise.^{296,326,422,442,443} H₂O₂, the major oxidant species produced that can affect susceptible thiols, transiently activates the different susceptible components of the MAPK pathways, various protein kinases, as well as Nrf2 to produce large, transient increases in antioxidant, phase II, and redox enzymes. As discussed in Section 2.5.2.1, an increase in antioxidant enzymes can be an important factor in ensuring appropriate regulatory control over the

signaling pathways. Further, in terms of prevention of chronic diseases in general, the induction of antioxidant, phase II, and redox enzymes occurs not only in the exercised muscles but also in other tissues, including the heart, liver, lung, and brain.^{296,444–449} As with the cytoprotective effect of polyphenols, these adaptations to exercise also appear to be due to oxidant-mediated activation of Keap1 with subsequent nuclear binding of Nrf2.^{450–454} Thus, sufficiently stressful exercise provokes a global anti-inflammatory and preventive effect throughout most tissues of the body.

However, these protective effects of exercise must be viewed as a relatively short-term effect. What this means is as follows: any alterations in signaling due to exercise are destined to be transient simply because of the nature of exercise; you just cannot do it for 20 hours each day. Because the levels of ROSs in the muscles would be expected to return close to baseline within a few minutes, the activation effects will last only as long as the actual exercise. With a half-life of only 90 minutes for IL-6, the cytokine profile in response to nondamaging/nonexhaustive exercise will decrease back to baseline within a few hours after the stop of exercise, as is observed. The cytoprotective effects of the antioxidant/regulatory enzymes also cannot be expected to last beyond the life of the proteins. With an approximate halflife (the time for half of newly synthesized proteins to be degraded) of approximately 10 minutes, approximately 5 hours, approximately 30 hours, 24–48 hours, and approximately 20 hours for CuZn SOD, Mn SOD, CAT, GPX, and thioredoxin reductase, respectively,^{455–458} the protective effects of enzyme induction do not last very long either. Thus, the exercise must be repeated on a daily or near daily basis if these cytoprotective and anti-inflammatory effects are to be realized on a long-term basis.

One last aspect of exercise that is just gaining a lot more interest is, well, the lack of it. Inactivity has been recognized for its strong relationship to increased risk for chronic disease for well over a decade.⁴⁵⁹⁻⁴⁶⁵ Only recently, however, has an understanding of how inactivity results in increased risk for disease been developed. As discussed earlier, the transient increase in oxidants resulting from exercise induces a global protective response. In contrast, physical inactivity produces a chronic low-level increase in radical production that leads to muscle damage, atrophy, and insulin resistance in the inactive muscles.^{466–469} From the latest 2005-2008 NHANES analyses, 33% of Americans report that they do not participate in physical activity at all and of the 76% who claim to be active approximately 50% do not meet the minimum activity guidelines.^{470–475} Thus, the majority of Americans may be under chronic metabolic stress due to inactivity (in addition to nutrient insufficiencies). Possible reasons for such a large increase in risk for chronic disease due to inactivity could include the insulin resistance that accompanies inactivity. In addition, as observed in some of the studies, an increase in abdominal fat at the expense of subcutaneous fat occurs with inactivity. As is discussed in more detail in Chapters 3 and 5, abdominal adipose tissue is a significant source of proinflammatory cytokines, which are a profound risk factor for many chronic diseases. Thus, sufficiently stressful exercise provokes a global anti-inflammatory and preventive effect throughout most tissues of the body as illustrated in Figure 2.6.



FIGURE 2.6 Attenuation of inflammation by exercise and phytochemicals: transient production of oxidants through exercise in muscle cells activates mitogen-activated protein kinase (MAPK) pathways sufficiently to lead to the synthesis of large amounts of interleukin (IL)-6, which produces a systemic reduction in the synthesis of proinflammatory cytokines and chemokines as well as in cellular responses to these. Various phytochemicals moderately inhibit the activation of MAPK pathways as well as activate nuclear factor κ - β (NF $\kappa\beta$) to dampen the activation of proinflammatory responses, whereas the transient production of oxidants from phytochemicals activates Nrf2, which, in addition to the covalent reactions of phytochemicals with Nrf2/Keap1, leads to the activation of antioxidant-response element (ARE)-/electrophile-response element (ERE)-mediated promotion events that increase the synthesis of a variety of redox-control and antioxidant enzymes to enhance redox control and further dampen proinflammatory responses.

2.7 SUMMARY AND RECOMMENDATIONS

2.7.1 INFLAMMATION

In the traditional inflammatory model, inflammation in a component of the immune system occurs as a result of pathogenic infections that produce a variety of intracellular and extracellular stimuli, which are capable of activating a proinflammatory signaling response. A sustained low-level activation by these stimuli of the various MAPK, NF $\kappa\beta$, and IFN pathways mediates the activation of a wide array of transcription factors and the subsequent synthesis of proinflammatory cytokines. These pathways are activated via a variety of TLRs, NLRs, and RLRs through the binding of these receptors with a large array of PAMPs and DAMPs. Over the last decades, it has become increasingly apparent that in addition to pathogenic stimuli increased cellular levels of calcium and ROS production resulting from either sterile damage or metabolic stress will also directly activate these same pathways through a variety of protein kinases and PI3K-mediated effects. The activation profile of these MAPK, NF $\kappa\beta$, and IFN pathways depends on the subcellular locations and relative concentrations of ROSs and calcium and on their additive and synergistic effects on different components of these pathways. The various proinflammatory PGs, LTs, and TXs also are produced as a result of the same calcium- and ROS-mediated events.

Thus, an inflammatory response is not necessarily a part of the immune system per se but rather a component of a normal continuum of graded responses to cellular stress. At moderate levels of metabolic stress, tissue perfusion will be enhanced via a temporary vasodilation response, allowing appropriate metabolic responses to the stressor to occur and returning the cell to homeostasis. As the level and duration of cellular and tissue stress increases, so does the cellular signaling response, to the point where in the continuing presence of stressors the cellular responses will include new synthesis of various antioxidant and redox regulation enzymes and may include tissue remodeling (hypertrophy) with sufficient intensity or duration of the stress. In the event of cellular damage, the stress-response signaling will intensify through the activation of PRRs by various DAMPs, increasing the possibility for recruiting macrophages, platelets, dendritic cells, and fibroblasts into the tissue. If the array of stressors includes pathogens, then additional inflammatory signaling is activated by various PAMPs to enhance the recruitment of inflammatory cells into the tissue, and the pathogens are then destroyed through a combination of enhanced ROS/RNS production and phagocytic activity by macrophages, dendritic cells, and neutrophils. The macrophages and dendritic cells can also participate in initiating an adapted immune response against the offending pathogens through interactions with various cells of the (adaptive) immune system. In the event of more extensive tissue damage (with or without pathogens), the continuing production of inflammatory signaling will activate fibroblasts, platelets, and monocytes, as well as any adult stem cells in the area to coordinate the closing of any wounds with blood clots and the replacement of destroyed tissue with new tissue cells and scar tissue. At the same time, the induction of COX-2, 12-LOX, and 15-LOX resulting from the initial proinflammatory cytokine response will start a shift in the production of lipid mediators from proinflammatory PGs, TXs, and LTs to anti-inflammatory PGJ₂, lipoxins, resolvins, and protectins. These compounds produce their anti-inflammatory effects through suppressing MAPK, NF $\kappa\beta$, PI3K, and IFN signaling; activating Nrf2; stimulating IL-10 release from macrophages; inducing apoptosis of neutrophils; promoting nonphlogistic phagocytosis by macrophages and their egress from the area; and inhibiting the synthesis of various chemokines.

Unfortunately, some components of the inflammatory response are capable of producing detrimental effects on surrounding healthy cells. Proinflammatory cytokines can alter cellular signaling to produce insulin resistance and enhance ROS production, leading to increased subcellular damage. The ROSs and RNSs produced by inflammatory cells can also diffuse into otherwise healthy cells to cause additional damage to proteins, lipids, and DNA. The excess ROSs can also interfere with normal signaling processes to disrupt the regulation of protein synthesis. In the case of properly resolved short-term inflammation, these effects are temporary and the damage is easily repaired.

In cases where the initiating stressor is relatively mild and constant, the normal cellular response to the continual presence of the stressor is to continually produce low-level amounts of proinflammatory signals, amounts that are insufficient to produce a full-blown inflammatory response. In such cases of unresolved low-level inflammation signaling that continues indefinitely, these effects can lead to progressive insulin resistance, cellular damage, enhanced fibrosis, inappropriate cell proliferation, and mutagenesis, all of which contribute to a variety of chronic diseases. Further, multiple short-term exposures to inflammatory responses due to repeated infections with the same pathogens (or chronic unresolved infections) in the same location can produce a similar array of damaging events.

The various anti-inflammatory and resolving molecules function by dampening the activation of the various MAPK pathways, NF $\kappa\beta$, and various protein kinases to suppress proinflammatory signaling as well as activate Nrf2/ARE to enhance the synthesis of a variety of antioxidant and redox system enzymes. The net effect of these changes is to greatly enhance a cell's ability to avoid the detrimental effects of ROSs as well as to maintain tighter regulatory control of signaling pathways through the endogenous regulation of redox states throughout the cell.

2.7.2 **Prevention of Inflammation**

Nutritional status is an important determinant in the ability to handle metabolic and other cellular stressors. Because a majority of Americans consume insufficient amounts of fruits and vegetables, most also have an insufficient intake of one or more of the nutrients that are essential for maintaining antioxidant and signaling control. Common nutrient insufficiencies among Americans include the minerals calcium, magnesium, iron, copper, selenium, and zinc and the vitamins C, D, and E. As a result of these dietary insufficiencies, a majority of Americans are under constant low-level metabolic stress and, therefore, in a constant low-level proinflammatory state. Thus, a primary recommendation to reduce inflammation-associated risks is to consume a nutritionally adequate diet to ensure that signaling, metabolic, and antioxidant cell functions are controlled appropriately. For convenience, the *MyPlate* approach (www.choosemyplate.gov), which includes standardized serving sizes, does provide an appropriate and nutritionally adequate model from which to start. In spite of some controversies related to the saturated fat content of the dairy recommendation, dairy products are a commonly available food that is high in calcium and currently comprise the primary source for the majority of calcium for most Americans; therefore, the current *MyPlate* dietary recommendations for three servings of dairy each day are considered acceptable. For overall nutrition, following these general recommendations will result in an adequate intake of all nutrients. In order to optimize the anti-inflammatory benefits, however, recommendations that go beyond adequate nutrient intake need to be made.

In terms of essential fatty acids, in spite of recent controversies in the academic literature and the lay press, there appears to be little evidence for worrying about moderate to high consumption of LA. In fact, there appear to be some antiinflammatory benefits from the higher levels of consumption of this essential fatty acid. The consumption of ALA, on the other hand, does appear to be below the recommended minimum for a sizeable minority of Americans. However, because it is relatively simple to ensure an adequate supply of EPA and DHA by consuming two servings of fish each week, rather than making extreme modifications to the diet in order to increase ALA consumption, this becomes the recommendation. It has the added advantage of ensuring that any additional anti-inflammatory benefits of excess EPA and DHA, such as suppression of proinflammatory PG, LT, and TX signaling and activation of PPARs, will be realized.

Regular consumption of a specific array of foods that are high in (different) phenolic content from fruits, vegetables, nuts, seeds, and whole grains is also recommended. Many phenolic compounds have beneficial effects on regulation mechanisms of cell signaling and, therefore, taking advantage of these effects is essential to developing an anti-inflammatory diet. TLR and NLR signaling pathways, as well as NF $\kappa\beta$ and the various MAPK signaling pathways, are inhibited by a wide array of phytochemicals. Through suppressing these proinflammatory signaling processes, the production of proinflammatory cytokines and prostanoids is inhibited, resulting in blunted inflammatory responses. In addition to these anti-inflammatory effects, phytochemicals can induce the synthesis of SODs, CAT, GPX, GSR, Trx, and Prx, all of which enhance the ability of cells to maintain appropriate regulatory control of redox status as well as to withstand the detrimental effects of ROSs. These latter effects are apparently mediated by affecting the regulation of Nrf2 by Keapl; ultimately enhancing the activation of the ARE/ERE regulatory element by Nrf2.

The difficulty in developing a dietary recommendation is that although individual phenolics and selected high-phenolic foods have been tested in limited short-term dietary experiments, there are no well-controlled clinical trials that actually test the long-term anti-inflammatory effects of diets that incorporate a variety of phenolic-rich foods. In spite of this, the apparent protection from inflammation-associated chronic diseases associated with phenolic intake in the 1 to 2 g range as observed for Mediterranean diets is used as a guideline.

As discussed in Section 2.5.2.2, the effective half-life of most phenolics is in the 1- to 2-hour range due to absorption and metabolic factors, resulting in a potential effect on signaling over a similar time frame. These effects are therefore highly transient; as a result, the protective effects cannot be expected to last beyond the lifetime

of the expressed proteins. Thus, for an anti-inflammatory benefit from high-phenolic foods to be realized they must be eaten on a regular basis. The basic recommendations for 2.5 cups of fruits and 3 cups of vegetables every day is an appropriate starting point to ensure nutritional requirements. As essential components of a phytochemical-rich diet, the nuts/seeds and beans/peas categories have been separated from the general protein foods or vegetable foods categories as described in the *MyPlate* approach. They are placed in their own food category in order to emphasize their importance and to ensure their consumption.

Certain common beverages such as tea, coffee, and red wine also contain beneficial phytochemicals, including EGCG, theaflavins, caffeic acids, and resveratrol. EGCG and theaflavins are commonly consumed in the form of black and green teas, whereas caffeic acids are usually obtained from drinking coffee. Because there has been little to no research to directly compare the anti-inflammatory effects of EGCG and theaflavins to those of caffeic acids, there is currently little justification to recommend tea over coffee. Because recommending alcohol consumption may be contraindicated for some, wine is not a recommended beverage. This does not mean resveratrol cannot be obtained because it is found in grapes, cranberries, and other specific foods that are recommended in Section 2.7.3.

When it comes to exercise, for an anti-inflammatory effect to be realized the best evidence so far indicates that a large increase in IL-6 to enhance serum levels of IL-10, sIL-1ra, sIL1r, and sTNFr (without a concomitant increase in IL-1 β or TNF- α) is necessary. This is usually observed after 60 minutes or more of whole-body exercise at an intensity ≥ 6 METS. IL-10 is associated with suppressed NF $\kappa\beta$ and MAPK signaling, whereas sIL-1ra, sIL1r, and sTNFr reduce the effects of circulating IL-1 α/β and TNF- α .

The benefits of reducing proinflammatory signaling and cellular responses to these signals specifically relate to the prevention of chronic diseases. By reducing inflammation, risks for cellular dysfunctions due to insulin resistance; enhanced damage to DNA, lipids, and proteins through ROS-/RNS-mediated damage; altered signaling pathways that lead to unscheduled cell proliferation; and enhanced fibrosis can be minimized. These will in turn reduce risk for cancer, neurodegenerative diseases, atherosclerosis, osteoporosis, and diabetes.

2.7.3 DIETARY RECOMMENDATIONS

In order to optimize the anti-inflammatory effects of dietary components, a variety of foods need to be consumed regularly in sufficiently large quantities. The selections for specific fruits, grains, oils, and vegetables were made on the basis of phytochemical content in order to maximize both variety and total consumption of these beneficial compounds, whereas the minimum serving selection in the meat category is based on the EPA/DHA content. The caloric content of these recommendations for fruits, vegetables, grains, oils, and meats will range from approximately 1500–1700 kcal, depending on the specific selections of individual foods. With the addition of 3 servings of dairy to maintain calcium intake, the total daily caloric intake for these recommendations will range from approximately 1760–1960 kcal. Because there is little evidence that moderate amounts of individual fats, proteins, or carbohydrates

make a significant difference in inflammatory mechanisms, the remaining selections for food servings that are necessary to satisfy caloric needs are considered discretionary, provided there is no excess consumption of calories that results in a gain in adiposity. The specific foods that make up the anti-inflammatory recommendations are as follows:

Fruits—minimum of 2.5 cups each day with at least three different selections from each category each week:

- A. Elderberries, blueberries, pomegranates, blackberries, raspberries, Saskatoon berries, black currants, raisins, figs or prunes.
- B. Plums, oranges, grapefruit, lemons, cantaloupe, cherries, cranberries, red/black grapes, or apples.
- C. Kiwi, bananas, pineapples, mango, pears, star fruit, guava, strawberries, peaches, or any other fruit.

Vegetables—minimum of 3 cups each day with at least three different selections from each category each week:

- A. Chili peppers, broccoli, water cress, garden cress, spinach, arugula, Brussels sprouts.
- B. Cauliflower, tomato, Romaine lettuce, green beans, sweet peppers, red cabbage.
- C. Onion, garlic, celery, green cabbage, asparagus, Swiss chard.
- D. Squash, zucchini, pumpkin, sweet potatoes, carrots, potato.
- Nuts and seeds—minimum of one 0.25-cup serving each day with a minimum of two different selections from each category each week:
 - A. Chestnuts, pecans, walnuts, pistachios.
 - B. Hazelnut, almonds, Brazil nuts, cashews, Macadamia nuts, or peanuts.
- Beans and peas—minimum of one 1-cup servings each day with a minimum of two different 1-cup selections from each category each week:
 - A. Black beans, adzuki beans, kidney beans, peas/pea pods, lima bean.
 - B. Lentils, pinto beans, chili beans, chick peas, navy beans.

Whole grains and cereals—minimum five 1 oz. servings of whole grains each day with a minimum of one selection from each category each week:

- A. Buckwheat, wheat, bulgur.
- B. Oats, corn, barley.
- C. Wild rice, rice.
- Oils—used for cooking, salads, and dipping—approximately 2 tbsp/day. A. EVOO.
- Meats-minimum of two servings of high-EPA/-DHA fish each week.
- Beverages—although water is the recommended beverage for fluid replacement, we acknowledge that a lot of coffee and tea is consumed all over the world for a variety of reasons. Although there is a lot of information on the preventive mechanisms associated with EGCG from green tea, there is actually little data to directly support the consumption of green tea over black tea, or over coffee, so no specific recommendation for either is made; both contain beneficial phenolics.

2.7.4 Exercise Recommendations

In order to obtain an optimal anti-inflammatory effect from physical activity, the recommendations are for whole-body, vigorous activities that are 6 METS or greater in intensity and carried out for approximately 1 hour each day for 3–5 days each week, with 3 days each week as the minimum recommended frequency. These activities include jogging, running, cycling at 10–12 mph or greater, swimming, and cross-country skiing.⁴⁷⁶

ACKNOWLEDGMENT

This work was supported in part by a salary grant from NEIU to C. Murray Ardies.

REFERENCES

- 1. Rana J, Nieuwdorp M, Jukema J, Kastelein J. Cardiovascular metabolic syndrome— An interplay of obesity, inflammation, diabetes and coronary heart disease. *Diabetes, Obesity and Metabolism.* 2007;9:218–232.
- Eizirik DL, Colli ML, Ortis F. The role of inflammation in insulitis and β-cell loss in type 1 diabetes. *Nature Reviews Endocrinology*. 2009;5:219–226.
- 3. Munro J, Cotran R. The pathogenesis of atherosclerosis: Atherogenesis and inflammation. Laboratory Investigation; A Journal of Technical Methods and Pathology. 1988;58:249.
- 4. Blum A, Miller H. The role of inflammation in atherosclerosis. *Israel Journal of Medical Sciences*. 1996;32:1059.
- Williams CS, Mann M, DuBois RN. The role of cyclooxygenases in inflammation, cancer, and development. *Oncogene*. 1999;18:7908.
- 6. Coussens LM, Werb Z. Inflammation and cancer. Nature-London. 2002;420:860-867.
- 7. Ardies CM. Inflammation as cause for scar cancers of the lung. *Integrative Cancer Therapies*. 2003;2:238–246.
- Pikarsky E, Porat RM, Stein I, Abramovitch R, Amit S, Kasem S, Gutkovich-Pyest E, Urieli-Shoval S, Galun E, Ben-Neriah Y. NFκβ functions as a tumour promoter in inflammation-associated cancer. *Nature*. 2004;431:461–465.
- 9. McGeer PL, McGeer EG. Inflammation and neurodegeneration in Parkinson's disease. *Parkinsonism & Related Disorders*. 2004;10:S3–S7.
- 10. Perry VH. The influence of systemic inflammation on inflammation in the brain: Implications for chronic neurodegenerative disease. *Brain, Behavior, and Immunity*. 2004;18:407–413.
- Akiyama H, Barger S, Barnum S, Bradt B, Bauer J, Cole GM, Cooper NR, Eikelenboom P, Emmerling M, Fiebich BL. Inflammation and Alzheimer's disease. *Neurobiology of Aging*. 2000;21:383–421.
- Sugiyama T. Involvement of interleukin-6 and prostaglandin E2 in periarticular osteoporosis of postmenopausal women with rheumatoid arthritis. *Journal of Bone and Mineral Metabolism.* 2001;19:89–96.
- 13. Mundy GR. Osteoporosis and inflammation. Nutrition Reviews. 2007;65:S147-S151.
- 14. McCormick RK. Osteoporosis: Integrating biomarkers and other diagnostic correlates into the management of bone fragility. *Alternative Medicine Review*. 2007;12:113.
- 15. Pickup JC. Inflammation and activated innate immunity in the pathogenesis of type 2 diabetes. *Diabetes Care*. 2004;27:813.
- 16. Kindt TJ, Goldsby RA, Osborne BA, Kuby J. *Kuby Immunology*. New York, NY: WH Freeman; 2007.

- 17. Abbas AK, Lichtman AH. *Basic Immunology: Functions & Disorders of the Immune System* (3rd Updated Ed.). Philadelphia, PA: Elsevier Saunders; 2010.
- 18. Abbas AK, Lichtman AH, Pillai S. *Cellular and Molecular Immunology*. Philadelphia, PA: Saunders/Elsevier; 2012.
- Murphy K, Travers P, Walport M, Janeway C. Janeway's Immunobiology. New York, NY: Garland Science; 2011.
- 20. Shalhoub J, Falck-Hansen MA, Davies AH, Monaco C. Innate immunity and monocytemacrophage activation in atherosclerosis. *Journal of Inflammation*. 2011;8:9.
- 21. Ross R. Atherosclerosis is an inflammatory disease. *American Heart Journal*. 1999;138:419–420.
- 22. Hennig B, Chow CK. Lipid peroxidation and endothelial cell injury: Implications in atherosclerosis. *Free Radical Biology and Medicine*. 1988;4:99–106.
- McCully KS, Vezeridis MP. Histopathological effects of homocysteine thiolactone on epithelial and stromal tissues. *Experimental and Molecular Pathology*. 1989;51:159–170.
- Rong JX, Rangaswamy S, Shen L, Dave R, Chang YH, Peterson H, Hodis HN, Chisolm GM, Sevanian A. Arterial injury by cholesterol oxidation products causes endothelial dysfunction and arterial wall cholesterol accumulation. *Arteriosclerosis, Thrombosis,* and Vascular Biology. 1998;18:1885.
- 25. Griendling KK, FitzGerald GA. Oxidative stress and cardiovascular injury. *Circulation*. 2003;108:1912–1916.
- 26. Taniyama Y, Griendling KK. Reactive oxygen species in the vasculature. Molecular and cellular mechanisms. *Hypertension*. 2003;42:1075–1081.
- Andersson J, Libby P, Hansson GK. Adaptive immunity and atherosclerosis. *Clinical Immunology*. 2010;134:33–46.
- Libby P, Ridker PM, Hansson GK. Inflammation in atherosclerosis: From pathophysiology to practice. *Journal of the American College of Cardiology*. 2009;54:2129–2138.
- Hansson GK. Inflammation, atherosclerosis, and coronary artery disease. New England Journal of Medicine. 2005;352:1685–1695.
- 30. Kiechl S, Egger G, Mayr M, Wiedermann CJ, Bonora E, Oberhollenzer F, Muggeo M, et al. Chronic infections and the risk of carotid atherosclerosis: Prospective results from a large population study. *Circulation*. 2001;103:1064–1070.
- Mehta M, Jawahar L, Saldeen M, Tom G, Rand M. Interactive role of infection, inflammation and traditional risk factors in atherosclerosis and coronary artery disease. *Journal* of the American College of Cardiology. 1998;31:1217–1225.
- Ericson K, Saldeen TGP, Lindquist O, Pahlson C, Mehta JL. Relationship of *Chlamydia* pneumoniae infection to severity of human coronary atherosclerosis. *Circulation*. 2000;101:2568.
- Corrado E, Rizzo M, Tantillo R, Muratori I, Bonura F, Vitale G, Novo S. Markers of inflammation and infection influence the outcome of patients with baseline asymptomatic carotid lesions. *Stroke*. 2006;37:482–486.
- Mahmoudi M, Curzen N, Gallagher P. Atherogenesis: The role of inflammation and infection. *Histopathology*. 2007;50:535–546.
- 35. Wallace JM. Nutritional and botanical modulation of the inflammatory cascade eicosanoids, cyclooxygenases, and lipoxygenases—as an adjunct in cancer therapy. *Integrative Cancer Therapies*. 2002;1:7.
- Simopoulos AP. Omega-3 fatty acids in inflammation and autoimmune diseases. *Journal* of the American College of Nutrition. 2002;21:495–505.
- Din JN, Harding SA, Valerio CJ, Sarma J, Lyall K, Riemersma RA, Newby DE, Flapan AD. Dietary intervention with oil rich fish reduces platelet-monocyte aggregation in man. *Atherosclerosis*. 2008;197:290–296.

- Giugliano D, Ceriello A, Esposito K. The effects of diet on inflammation: Emphasis on the metabolic syndrome. *Journal of the American College of Cardiology*. 2006;48:677–685.
- 39. He K, Liu K, Daviglus ML, Jenny NS, Mayer-Davis E, Jiang R, Steffen L, Siscovick D, Tsai M, Herrington D. Associations of dietary long-chain n-3 polyunsaturated fatty acids and fish with biomarkers of inflammation and endothelial activation (from the multi-ethnic study of atherosclerosis [mesa]). *The American Journal of Cardiology*. 2009;103:1238–1243.
- Pischon T, Hankinson SE, Hotamisligil GS, Rifai N, Willett WC, Rimm EB. Habitual dietary intake of n-3 and n-6 fatty acids in relation to inflammatory markers among US men and women. *Circulation*. 2003;108:155–160.
- 41. Lopez-Garcia E, Schulze MB, Fung TT, Meigs JB, Rifai N, Manson JAE, Hu FB. Major dietary patterns are related to plasma concentrations of markers of inflammation and endothelial dysfunction. *The American Journal of Clinical Nutrition*. 2004;80:1029.
- 42. Zampelas A, Panagiotakos DB, Pitsavos C, Das UN, Chrysohoou C, Skoumas Y, Stefanadis C. Fish consumption among healthy adults is associated with decreased levels of inflammatory markers related to cardiovascular disease: The ATTICA study. *Journal of the American College of Cardiology*. 2005;46:120.
- 43. Lavie CJ, Milani RV, Mehra MR, Ventura HO. Omega-3 polyunsaturated fatty acids and cardiovascular diseases. *Journal of the American College of Cardiology*. 2009;54:585–594.
- Lee JH, O'Keefe JH, Lavie CJ, Marchioli R, Harris WS. Omega-3 fatty acids for cardioprotection. *Mayo Clinic Proceedings*. 2008;83:324–332.
- 45. Kris-Etherton PM, Harris WS, Appel LJ. Fish consumption, fish oil, omega-3 fatty acids, and cardiovascular disease. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2003;23:e20–e30.
- 46. Wang C, Harris WS, Chung M, Lichtenstein AH, Balk EM, Kupelnick B, Jordan HS, Lau J. N–3 fatty acids from fish or fish-oil supplements, but not α-linolenic acid, benefit cardiovascular disease outcomes in primary- and secondary-prevention studies: A systematic review. *The American Journal of Clinical Nutrition*. 2006;84:5–17.
- 47. Marchioli R, Barzi F, Bomba E, Chieffo C, Di Gregorio D, Di Mascio R, Franzosi MG, Geraci E, Levantesi G, Maggioni AP. Early protection against sudden death by n-3 polyunsaturated fatty acids after myocardial infarction. *Circulation*. 2002;105: 1897–1903.
- 48. Yokoyama M, Origasa H, Matsuzaki M, Matsuzawa Y, Saito Y, Ishikawa Y, Oikawa S, Sasaki J, Hishida H, Itakura H. Effects of eicosapentaenoic acid on major coronary events in hypercholesterolaemic patients (jelis): A randomised open-label, blinded end-point analysis. *The Lancet*. 2007;369:1090–1098.
- 49. Geleijnse JM, Giltay EJ, Grobbee DE, Donders ART, Kok FJ. Blood pressure response to fish oil supplementation: Metaregression analysis of randomized trials. *Journal of Hypertension*. 2002;20:1493.
- Fotuhi M, Mohassel P, Yaffe K. Fish consumption, long-chain omega-3 fatty acids and risk of cognitive decline or Alzheimer disease: A complex association. *Nature Clinical Practice Neurology*. 2009;5:140–152.
- Cole GM, Lim GP, Yang F, Teter B, Begum A, Ma Q, Harris-White ME, Frautschy SA. Prevention of Alzheimer's disease: Omega-3 fatty acid and phenolic anti-oxidant interventions. *Neurobiology of Aging*. 2005;26:133–136.
- 52. Cole GM, Ma QL, Frautschy SA. Omega-3 fatty acids and dementia. *Prostaglandins, Leukotrienes and Essential Fatty Acids*. 2009;81:213–221.

- 53. Thiébaut A, Chajès V, Gerber M, Boutron-Ruault MC, Joulin V, Lenoir G, Berrino F, Riboli E, Bénichou J, Clavel-Chapelon F. Dietary intakes of ω-6 and ω-3 polyunsaturated fatty acids and the risk of breast cancer. *International Journal of Cancer*. 2009;124:924–931.
- Roynette CE, Calder PC, Dupertuis YM, Pichard C. N-3 polyunsaturated fatty acids and colon cancer prevention. *Clinical Nutrition*. 2004;23:139–151.
- 55. Rose DP, Connolly JM. Omega-3 fatty acids as cancer chemopreventive agents. *Pharmacology & Therapeutics*. 1999;83:217–244.
- Reddy BS. Omega-3 fatty acids in colorectal cancer prevention. *International Journal of Cancer*. 2004;112:1–7.
- Simopoulos AP. The importance of the omega-6/omega-3 fatty acid ratio in cardiovascular disease and other chronic diseases. *Experimental Biology and Medicine*. 2008;233:674–688.
- Rousseau JH, Kleppinger A, Kenny AM. Self-reported dietary intake of omega-3 fatty acids and association with bone and lower extremity function. *Journal of the American Geriatrics Society*. 2009;57:1781–1788.
- Chen Y, Ho S, Lam S. Higher sea fish intake is associated with greater bone mass and lower osteoporosis risk in postmenopausal Chinese women. *Osteoporosis International*. 2010;21:939–946.
- Miles EA, Zoubouli P, Calder PC. Differential anti-inflammatory effects of phenolic compounds from extra virgin olive oil identified in human whole blood cultures. *Nutrition*. 2005;21:389–394.
- Beauchamp GK, Keast RSJ, Morel D, Lin J, Pika J, Han Q, Lee CH, Smith AB, Breslin PAS. Phytochemistry: Ibuprofen-like activity in extra-virgin olive oil. *Nature*. 2005;437:45–46.
- 62. Avellone G, Di Garbo V, Campisi D, De Simone R, Raneli G, Scaglione R, Licata G. Effects of moderate Sicilian red wine consumption on inflammatory biomarkers of atherosclerosis. *European Journal of Clinical Nutrition*. 2005;60:41–47.
- Alarcón de la Lastra C, Villegas I. Resveratrol as an anti-inflammatory and anti-aging agent: Mechanisms and clinical implications. *Molecular Nutrition & Food Research*. 2005;49:405–430.
- 64. Khan N, Mukhtar H. Multitargeted therapy of cancer by green tea polyphenols. *Cancer Letters*. 2008;269:269–280.
- 65. Vinson JA. Black and green tea and heart disease: A review. Biofactors. 2000;13:127–132.
- 66. Larrosa M, Luceri C, Vivoli E, Pagliuca C, Lodovici M, Moneti G, Dolara P. Polyphenol metabolites from colonic microbiota exert anti-inflammatory activity on different inflammation models. *Molecular Nutrition & Food Research*. 2009;53:1044–1054.
- 67. Esposito K, Marfella R, Ciotola M, Di Palo C, Giugliano F, Giugliano G, D'Armiento M, D'Andrea F, Giugliano D. Effect of a Mediterranean-style diet on endothelial dysfunction and markers of vascular inflammation in the metabolic syndrome. *JAMA: The Journal of the American Medical Association*. 2004;292:1440.
- 68. Petersen AMW, Pedersen BK. The anti-inflammatory effect of exercise. *Journal of Applied Physiology*. 2005;98:1154.
- 69. Kadoglou NPE, Iliadis F, Angelopoulou N, Perrea D, Ampatzidis G, Liapis CD, Alevizos M. The anti-inflammatory effects of exercise training in patients with type 2 diabetes mellitus. *Journal of Cardiovascular Risk*. 2007;14:837.
- Ostrowski K, Rohde T, Asp S, Schjerling P, Pedersen BK. Pro- and anti-inflammatory cytokine balance in strenuous exercise in humans. *The Journal of Physiology*. 1999;515:287.
- Pedersen B. The anti-inflammatory effect of exercise: Its role in diabetes and cardiovascular disease control. *Essays in Biochemistry*. 2006;42:105–117.
- 72. Pedersen BK. Edward F. Adolph distinguished lecture: Muscle as an endocrine organ: IL-6 and other myokines. *Journal of Applied Physiology*. 2009;107:1006.

- 73. Pedersen BK, Febbraio MA. Muscle as an endocrine organ: Focus on muscle-derived interleukin-6. *Physiological Reviews*. 2008;88:1379.
- 74. Balducci S, Zanuso S, Nicolucci A, Fernando F, Cavallo S, Cardelli P, Fallucca S, Alessi E, Letizia C, Jimenez A. Anti-inflammatory effect of exercise training in subjects with type 2 diabetes and the metabolic syndrome is dependent on exercise modalities and independent of weight loss. *Nutrition, Metabolism and Cardiovascular Diseases*. 2010;20:608–617.
- Timmerman KL, Flynn MG, Coen PM, Markofski MM, Pence BD. Exercise traininginduced lowering of inflammatory (cd14+cd16+)monocytes: A role in the anti-inflammatory influence of exercise? *Journal of Leukocyte Biology*. 2008;84:1271–1278.
- Kato T, Yamashita T, Mizutani S, Honda A, Matumoto M, Umemura Y. Adolescent exercise associated with long-term superior measures of bone geometry: A cross-sectional DXA and MRI study. *British Journal of Sports Medicine*. 2009;43:932–935.
- 77. Howe T, Shea B, Dawson L, Downie F, Murray A, Ross C, Harbour R, Caldwell L, Creed G. Exercise for preventing and treating osteoporosis in postmenopausal women. *Cochrane Database of Systematic Reviews*. 2011;7:Article # CD000333.
- 78. Marcus R. Role of exercise in preventing and treating osteoporosis. *Rheumatic Disease Clinics of North America*. 2001;27:131–141.
- Vaynman SS, Ying Z, Yin D, Gomez-Pinilla F. Exercise differentially regulates synaptic proteins associated to the function of BDNF. *Brain Research*. 2006;1070:124–130.
- Lytle ME, Vander Bilt J, Pandav RS, Dodge HH, Ganguli M. Exercise level and cognitive decline: The movies project. *Alzheimer Disease & Associated Disorders*. 2004;18:57.
- 81. Bouchard C, Depres J, Tremblay A. Exercise and obesity. Obesity Research. 1993;1:133.
- 82. Jakicic JM, Otto AD. Treatment and prevention of obesity: What is the role of exercise? *Nutrition Reviews*. 2006;64:S57–S61.
- 83. Consultation WHO. Obesity: Preventing and managing the global epidemic. *World Health Organization Technical Report Series* 894, 2000.
- 84. Rundle A. Molecular epidemiology of physical activity and cancer. *Cancer Epidemiology Biomarkers & Prevention*. 2005;14:227.
- Thompson PD, Buchner D, Piña IL, Balady GJ, Williams MA, Marcus BH, Berra K, Blair SN, Costa F, Franklin B. Exercise and physical activity in the prevention and treatment of atherosclerotic cardiovascular disease. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2003;23:e42–e49.
- Manson JAE, Hu FB, Rich-Edwards JW, Colditz GA, Stampfer MJ, Willett WC, Speizer FE, Hennekens CH. A prospective study of walking as compared with vigorous exercise in the prevention of coronary heart disease in women. *New England Journal of Medicine*. 1999;341:650–658.
- 87. Ivy J. Role of exercise training in the prevention and treatment of insulin resistance and non-insulin-dependent diabetes mellitus. *Sports Medicine (Auckland, NZ)*. 1997;24:321.
- Colberg S, Sigal R, Fernhall B, Regensteiner J, Blissmer B, Rubin R, Chasan-Taber L, Albright A, Braun B. Exercise and type 2 diabetes: The American College of Sports Medicine and the American Diabetes Association: Joint position statement. *Diabetes Care*. 2010;33:e147.
- 89. Campbell KL, McTiernan A. Exercise and biomarkers for cancer prevention studies. *The Journal of Nutrition*. 2007;137:161S.
- Wolin K, Yan Y, Colditz G, Lee I. Physical activity and colon cancer prevention: A metaanalysis. *British Journal of Cancer*. 2009;100:611–616.
- 91. Gómez-Pinilla F, Dao L, So V. Physical exercise induces FGF-2 and its mRNA in the hippocampus. *Brain Research*. 1997;764:1–8.
- Gómez-Pinilla F, Ying Z, Roy RR, Molteni R, Edgerton VR. Voluntary exercise induces a BDNF-mediated mechanism that promotes neuroplasticity. *Journal of Neurophysiology*. 2002;88:2187.

- Akira S, Uematsu S, Takeuchi O. Pathogen recognition and innate immunity. *Cell*. 2006;124:783–801.
- 94. Blasius AL, Beutler B. Intracellular toll-like receptors. Immunity. 2010;32:305-315.
- 95. Pichlmair A, Reis e Sousa C. Innate recognition of viruses. Immunity. 2007;27:370-383.
- 96. Segal AW. How neutrophils kill microbes. Annual Review of Immunology. 2005;23:197.
- Dale DC, Boxer L, Liles WC. The phagocytes: Neutrophils and monocytes. *Blood*. 2008;112:935.
- Bottazzi B, Doni A, Garlanda C, Mantovani A. An integrated view of humoral innate immunity: Pentraxins as a paradigm. *Annual Review of Immunology*. 2009;28:157–183.
- 99. Rock KL, Latz E, Ontiveros F, Kono H. The sterile inflammatory response. *Annual Review of Immunology*. 2009;28:321–342.
- 100. Schroder K, Tschopp J. The inflammasomes. Cell. 2010;140:821-832.
- 101. Cinel I, Opal SM. Molecular biology of inflammation and sepsis: A primer. *Critical Care Medicine*. 2009;37:291–304.
- 102. Sims JE, Smith DE. The IL-1 family: Regulators of immunity. *Nature Reviews Immunology*. 2010;10:89–102.
- 103. Chen GY, Nuñez G. Sterile inflammation: Sensing and reacting to damage. *Nature Reviews Immunology*. 2010;10:826–837.
- 104. Chalifour A, Jeannin P, Gauchat JF, Blaecke A, Malissard M, N'Guyen T, Thieblemont N, Delneste Y. Direct bacterial protein PAMP recognition by human NK cells involves TLRs and triggers defensin production. *Blood.* 2004;104:1778.
- 105. Marshall JS. Mast-cell responses to pathogens. *Nature Reviews Immunology*. 2004;4: 787–799.
- Chen G, Shaw MH, Kim YG, Nuñez G. Nod-like receptors: Role in innate immunity and inflammatory disease. *Annual Review of Pathological Mechanical Disease*. 2009;4:365–398.
- Takeuchi O, Akira S. Innate immunity to virus infection. *Immunological Reviews*. 2009; 227:75–86.
- 108. Takeuchi O, Akira S. Pattern recognition receptors and inflammation. *Cell*. 2010; 140:805–820.
- 109. Mikkelsen SS, Jensen SB, Chiliveru S, Melchjorsen J, Julkunen I, Gaestel M, Arthur JSC, Flavell RA, Ghosh S, Paludan SR. RIG-I-mediated activation of p38–MAPK is essential for viral induction of interferon and activation of dendritic cells. *Journal of Biological Chemistry*. 2009;284:10774.
- 110. Krishnan J, Selvarajoo K, Tsuchiya M, Lee G, Choi S. Toll-like receptor signal transduction. *Experimental and Molecular Medicine*. 2007;39:421.
- 111. Mogensen TH. Pathogen recognition and inflammatory signaling in innate immune defenses. *Clinical Microbiology Reviews*. 2009;22:240.
- 112. Chen W, Li Z, Bai L, Lin Y. NFκβ in lung cancer, a carcinogenesis mediator and a prevention and therapy target. *Frontiers in Bioscience: A Journal and Virtual Library*. 2011;16:1172.
- 113. Lin Y, Bai L, Chen W, Xu S. The NFκβ activation pathways, emerging molecular targets for cancer prevention and therapy. *Expert Opinion on Therapeutic Targets*. 2010;14:45–55.
- Medzhitov R. Recognition of microorganisms and activation of the immune response. *Nature*. 2007;449:819–826.
- 115. Grutzkau A, Kruger-Krasagakes S, Baumeister H, Schwarz C, Kogel H, Welker P, Lippert U, Henz BM, Moller A. Synthesis, storage, and release of vascular endothelial growth factor/vascular permeability factor (VEGF/VPF) by human mast cells: Implications for the biological significance of VEGF206. *Molecular Biology of the Cell*. 1998;9:875.
- 116. Norrby K. Mast cells and angiogenesis. *APMIS: Acta Pathologica, Microbiologica, et Immunologica Scandinavica*. 2002;110:355–371.

- 117. Funk CD. Prostaglandins and leukotrienes: Advances in eicosanoid biology. *Science*. 2001;294:1871.
- 118. Calder PC. Polyunsaturated fatty acids and inflammation. *Prostaglandins, Leukotrienes and Essential Fatty Acids*. 2006;75:197–202.
- 119. Buczynski MW, Dumlao DS, Dennis EA. Thematic review series: Proteomics. An integrated omics analysis of eicosanoid biology. *Journal of Lipid Research*. 2009;50:1015–1038.
- 120. Tilley SL, Coffman TM, Koller BH. Mixed messages: Modulation of inflammation and immune responses by prostaglandins and thromboxanes. *Journal of Clinical Investigation*. 2001;108:15–24.
- 121. Sampey AV, Monrad S, Crofford LJ. Microsomal prostaglandin E synthase-1: The inducible synthase for prostaglandin E2. Arthritis Research & Therapy. 2005;7:1–4.
- 122. Wu KK, Liou JY. Cellular and molecular biology of prostacyclin synthase. *Biochemical and Biophysical Research Communications*. 2005;338:45–52.
- 123. Kapoor M, Kojima F, Yang L, Crofford LJ. Sequential induction of pro-and anti-inflammatory prostaglandins and peroxisome proliferators-activated receptor-gamma during normal wound healing: A time course study. *Prostaglandins, Leukotrienes and Essential Fatty Acids*. 2007;76:103–112.
- 124. Gilroy DW, Lawrence T, Perretti M, Rossi AG. Inflammatory resolution: New opportunities for drug discovery. *Nature Reviews Drug Discovery*. 2004;3:401–416.
- 125. Nelson DR, Kamataki T, Waxman DJ, Guengerich FP, Estabrook RW, Feyereisen R, Gonzalez FJ, Coon MJ, Gunsalus IC, Gotoh O. The p450 superfamily: Update on new sequences, gene mapping, accession numbers, early trivial names of enzymes, and nomenclature. *DNA and Cell Biology*. 1993;12:1–51.
- 126. Rudberg PC, Tholander F, Andberg M, Thunnissen MMGM, Haeggström JZ. Leukotriene A4 hydrolase. *Journal of Biological Chemistry*. 2004;279:27376.
- 127. Evans JH, Spencer DM, Zweifach A, Leslie CC. Intracellular calcium signals regulating cytosolic phospholipase A2 translocation to internal membranes. *Journal of Biological Chemistry*. 2001;276:30150.
- 128. Bonventre JV, Swidler M. Calcium dependency of prostaglandin E2 production in rat glomerular mesangial cells. Evidence that protein kinase C modulates the Ca²⁺-dependent activation of phospholipase A2. *Journal of Clinical Investigation*. 1988;82:168.
- 129. Handlogten ME, Huang C, Shiraishi N, Awata H, Miller RT. The Ca²⁺-sensing receptor activates cytosolic phospholipase A2 via a GQα-dependent ERK-independent pathway. *Journal of Biological Chemistry*. 2001;276:13941.
- 130. Zor U, Ferber E, Gergely P, Szücs K, Dombradi V, Goldman R. Reactive oxygen species mediate phorbol ester-regulated tyrosine phosphorylation and phospholipase A2 activation: Potentiation by vanadate. *Biochemical Journal*. 1993;295:879.
- 131. Folco G, Murphy RC. Eicosanoid transcellular biosynthesis: From cell-cell interactions to in vivo tissue responses. *Pharmacological Reviews*. 2006;58:375.
- 132. McIntyre TM, Zimmerman GA, Prescott SM. Leukotrienes C4 and D4 stimulate human endothelial cells to synthesize platelet-activating factor and bind neutrophils. *Proceedings of the National Academy of Sciences*. 1986;83:2204.
- 133. Venable ME, Zimmerman GA, McIntyre TM, Prescott SM. Platelet-activating factor: A phospholipid autacoid with diverse actions. *Journal of Lipid Research*. 1993;34:691.
- 134. Lopez-Ilasaca M, Crespo P, Pellici PG, Gutkind JS, Wetzker R. Linkage of G proteincoupled receptors to the MAPK signaling pathway through PI-3 kinase γ. Science. 1997;275:394–397.
- 135. Guijas C, Perez-Chacon G, Astudillo AM, Rubio JM, Gil-de-Gomez L, Balboa MA, Balsinde J. Simultaneous activation of p38 and JNK by arachidonic acid stimulates the cytosolic phospholipase A2-dependent synthesis of lipid droplets in human monocytes. *Journal of Lipid Research*. 2012;53:2343–2354.

- 136. Wang X, Martindale JL, Liu Y, Holbrook NJ. The cellular response to oxidative stress: Influences of mitogen-activated protein kinase signalling pathways on cell survival. *Biochemical Journal*. 1998;333:291.
- 137. Chang MC, Wu HL, Lee JJ, Lee PH, Chang HH, Hahn LJ, Lin BR, Chen YJ, Jeng JH. The induction of prostaglandin E2 production, interleukin-6 production, cell cycle arrest, and cytotoxicity in primary oral keratinocytes and KB cancer cells by areca nut ingredients is differentially regulated by MEK/ERK activation. *Journal of Biological Chemistry*. 2004;279:50676.
- 138. Chen BC, Liao CC, Hsu MJ, Liao YT, Lin CC, Sheu JR, Lin CH. Peptidoglycan-induced IL-6 production in RAW264.7 macrophages is mediated by cyclooxygenase-2, PGE2/ PGE4 receptors, protein kinase A, Ικβ kinase, and ΝFκβ. *The Journal of Immunology*. 2006;177:681.
- 139. Shek PN, Shephard RJ. Physical exercise as a human model of limited inflammatory response. *Canadian Journal of Physiology and Pharmacology*. 1998;76:589–597.
- Wilson JR, Kapoor SC. Contribution of prostaglandins to exercise-induced vasodilation in humans. American Journal of Physiology-Heart and Circulatory Physiology. 1993;265:H171.
- 141. Lin Y, Berg AH, Iyengar P, Lam TKT, Giacca A, Combs TP, Rajala MW, et al. The hyperglycemia-induced inflammatory response in adipocytes. *Journal of Biological Chemistry*. 2005;280:4617.
- 142. Kershaw EE, Flier JS. Adipose tissue as an endocrine organ. *Journal of Clinical Endocrinology & Metabolism*. 2004;89:2548.
- 143. Frühbeck G, Gómez-Ambrosi J, Muruzábal FJ, Burrell MA. The adipocyte: A model for integration of endocrine and metabolic signaling in energy metabolism regulation. *American Journal of Physiology-Endocrinology and Metabolism*. 2001;280:E827.
- 144. Wortman P, Miyazaki Y, Kalupahana NS, Kim S, Hansen-Petrik M, Saxton AM, Claycombe KJ, Voy BH, Whelan J, Moustaid-Moussa N. N3 and n6 polyunsaturated fatty acids differentially modulate prostaglandin E secretion but not markers of lipogenesis in adipocytes. *Nutrition & Metabolism*. 2009;6:5.
- 145. Serhan CN, Savill J. Resolution of inflammation: The beginning programs the end. *Nature Immunology*. 2005;6:1191–1197.
- 146. Serhan CN, Yacoubian S, Yang R. Anti-inflammatory and pro-resolving lipid mediators. *Annual Review of Pathology*. 2008;3:279.
- 147. Bannenberg G, Serhan CN. Specialized pro-resolving lipid mediators in the inflammatory response: An update. *Biochimica et Biophysica Acta (BBA)-Molecular and Cell Biology of Lipids*. 2010;1801:1260–1273.
- 148. Saban MR, Hellmich H, Nguyen NB, Winston J, Hammond TG, Saban R. Time course of LPS-induced gene expression in a mouse model of genitourinary inflammation. *Physiological Genomics*. 2001;5:147.
- Wymann MP, Schneiter R. Lipid signalling in disease. Nature Reviews Molecular Cell Biology. 2008;9:162–176.
- 150. White RE, Kryman JP, El-Mowafy AM, Han G, Carrier GO. cAMP-dependent vasodilators cross-activate the cGMP-dependent protein kinase to stimulate BKCa channel activity in coronary artery smooth muscle cells. *Circulation Research*. 2000; 86:897.
- 151. Hill S, Ganellin C, Timmerman H, Schwartz J, Shankley N, Young J, Schunack W, Levi R, Haas H. International union of pharmacology. Xiii. Classification of histamine receptors. *Pharmacological Reviews*. 1997;49:253.
- 152. Amerongen GP, Draijer R, Vermeer MA, van Hinsbergh VWM. Transient and prolonged increase in endothelial permeability induced by histamine and thrombin: Role of protein kinases, calcium, and RhoA. *Circulation Research*. 1998;83:1115.
- 153. Kumar P, Shen Q, Pivetti CD, Lee ES, Wu MH, Yuan SY. Molecular mechanisms of endothelial hyperpermeability: Implications in inflammation. *Expert Reviews in Molecular Medicine*. 2009;11:e19.

- 154. Dejana E, Orsenigo F, Lampugnani MG. The role of adherens junctions and VE-cadherin in the control of vascular permeability. *Journal of Cell Science*. 2008;121:2115.
- 155. Hudry-Clergeon H, Stengel D, Ninio E, Vilgrain I. Platelet-activating factor increases VE-cadherin tyrosine phosphorylation in mouse endothelial cells and its association with the ptdins3'-kinase. *The FASEB Journal*. 2005;19:512.
- 156. Sporri R, Reis-e-Sousa C. Inflammatory mediators are insufficient for full dendritic cell activation and promote expansion of CD4⁺ T cell populations lacking helper function. *Nature Immunology*. 2005;6:163–170.
- 157. Mosser DM. The many faces of macrophage activation. *Journal of Leukocyte Biology*. 2003;73:209.
- 158. Mosser DM, Edwards JP. Exploring the full spectrum of macrophage activation. *Nature Reviews Immunology*. 2008;8:958–969.
- 159. Elmore S. Apoptosis: A review of programmed cell death. *Toxicologic Pathology*. 2007;35:495–516.
- 160. Gouwy M, Struyf S, Noppen S, Schutyser E, Springael JY, Parmentier M, Proost P, Van Damme J. Synergy between coproduced CC and CXC chemokines in monocyte chemotaxis through receptor-mediated events. *Molecular Pharmacology*. 2008;74:485.
- Ember JA, Sanderson SD, Hugli TE, Morgan EL. Induction of interleukin-8 synthesis from monocytes by human c5a anaphylatoxin. *The American Journal of Pathology*. 1994;144:393.
- 162. Chen XL, Dodd G, Kunsch C. Sulforaphane inhibits TNF-α-induced activation of p38–MAPK and VCAM-1 and MCP-1 expression in endothelial cells. *Inflammation Research*. 2009;58:513–521.
- 163. Sobota R, Muller P, Heinrich P, Schaper F. Prostaglandin E1 inhibits IL-6-induced MCP-1 expression by interfering specifically in IL-6-dependent ERK1/2, but not STAT3, activation. *The Biochemical Journal*. 2008;412:65–72.
- Lawson C, Wolf S. ICAM-1 signaling in endothelial cells. *Pharmacological Reports*. 2009;61:22–32.
- 165. Zhou Z, Connell MC, MacEwan DJ. TNFr1-induced NF-κβ, but not ERK, p38–MAPK or JNK activation, mediates TNF-induced ICAM-1 and VCAM-1 expression on endothelial cells. *Cellular Signalling*. 2007;19:1238–1248.
- 166. Bayat H, Xu S, Pimentel D, Cohen RA, Jiang B. Activation of thromboxane receptor upregulates interleukin (IL)-1β-induced VCAM-1 expression through JNK signaling. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2008;28:127–134.
- 167. Zerfaoui M, Suzuki Y, Naura AS, Hans CP, Nichols C, Boulares AH. Nuclear translocation of p65 MFκβ is sufficient for VCAM-1, but not ICAM-1, expression in TNFstimulated smooth muscle cells: Differential requirement for PARP-1 expression and interaction. *Cellular Signalling*. 2008;20:186–194.
- 168. Kansas GS. Selectins and their ligands: Current concepts and controversies. *Blood*. 1996;88:3259.
- 169. Read MA, Whitley MZ, Gupta S, Pierce JW, Best J, Davis RJ, Collins T. Tumor necrosis factor α-induced E-selectin expression is activated by the nuclear factor-κβ and c-Jun N-terminal kinase/p38 mitogen-activated protein kinase pathways. *Journal of Biological Chemistry*. 1997;272:2753.
- 170. Ley K, Laudanna C, Cybulsky MI, Nourshargh S. Getting to the site of inflammation: The leukocyte adhesion cascade updated. *Nature Reviews Immunology*. 2007;7:678–689.
- 171. Sadik CD, Luster AD. Lipid-cytokine-chemokine cascades orchestrate leukocyte recruitment in inflammation. *Journal of Leukocyte Biology*. 2012;91:207–215.
- 172. Chou RC, Kim ND, Sadik CD, Seung E, Lan Y, Byrne MH, Haribabu B, Iwakura Y, Luster AD. Lipid-cytokine-chemokine cascade drives neutrophil recruitment in a murine model of inflammatory arthritis. *Immunity*. 2010;33:266–278.

- 173. Opitz B, Eitel J, Meixenberger K, Suttorp N. Role of toll-like receptors, NOD-like receptors and RIG-I-like receptors in endothelial cells and systemic infections. *Thrombosis and Haemostasis*. 2009;102:1103–1109.
- 174. Rose-John S. IL-6 trans-signaling via the soluble IL-6 receptor: Importance for the pro-inflammatory activities of IL-6. *International Journal of Biological Sciences*. 2012;8:1237.
- 175. Black S, Kushner I, Samols D. C-reactive protein. *Journal of Biological Chemistry*. 2004;279:48487.
- 176. Pepys MB, Hirschfield GM. C-reactive protein: A critical update. *Journal of Clinical Investigation*. 2003;111:1805–1812.
- 177. Young DP, Kushner I, Samols D. Binding of C/EBPβ to the C-reactive protein (CRP) promoter in Hep3b cells is associated with transcription of CRP mRNA. *The Journal of Immunology*. 2008;181:2420.
- 178. Heinrich PC, Behrmann I, Haan S, Hermanns HM, Müller-Newen G, Schaper F. Principles of interleukin (IL)-6-type cytokine signalling and its regulation. *Biochemical Journal*. 2003;374:1.
- 179. Leaman DW, Leung S, Li X, Stark G. Regulation of STAT-dependent pathways by growth factors and cytokines. *The FASEB Journal*. 1996;10:1578.
- Rawlings JS, Rosler KM, Harrison DA. The Jak/STAT signaling pathway. *Journal of Cell Science*. 2004;117:1281–1283.
- 181. Calvisi DF, Ladu S, Gorden A, Farina M, Conner EA, Lee JS, Factor VM, Thorgeirsson SS. Ubiquitous activation of Ras and Jak/STAT pathways in human HCC. *Gastroenterology*. 2006;130:1117–1128.
- 182. Carroll MC. Complement and humoral immunity. Vaccine. 2008;26:I28-I33.
- Gros P, Milder FJ, Janssen BJC. Complement driven by conformational changes. *Nature Reviews Immunology*. 2008;8:48–58.
- 184. Chang MK, Binder CJ, Torzewski M, Witztum JL. C-reactive protein binds to both oxidized LDL and apoptotic cells through recognition of a common ligand: Phosphorylcholine of oxidized phospholipids. *Proceedings of the National Academy of Sciences*. 2002;99:13043.
- 185. Gershov D, Kim SJ, Brot N, Elkon KB. C-reactive protein binds to apoptotic cells, protects the cells from assembly of the terminal complement components, and sustains an antiinflammatory innate immune response. *The Journal of Experimental Medicine*. 2000;192:1353.
- 186. Galve-de Rochemonteix B, Wiktorowicz K, Kushner I, Dayer JM. C-reactive protein increases production of IL-1α, IL-1β, and TNF-α, and expression of mRNA by human alveolar macrophages. *Journal of Leukocyte Biology*. 1993;53:439–445.
- 187. Pue CA, Mortensen RF, Marsh CB, Pope HA, Wewers MD. Acute phase levels of C-reactive protein enhance IL-1β and IL-1ra production by human blood monocytes but inhibit IL-1β and IL-1ra production by alveolar macrophages. *The Journal of Immunology*. 1996;156:1594.
- 188. Yeh ETH. CRP as a mediator of disease. Circulation. 2004;109:II-11-II-14.
- 189. May AE, Seizer P, Gawaz M. Platelets: Inflammatory firebugs of vascular walls. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2008;28:s5–s10.
- Borissoff JI, Spronk H, Ten Cate H. The hemostatic system as a modulator of atherosclerosis. *The New England Journal of Medicine*. 2011;364:1746–1760.
- 191. Jackson SP, Schoenwaelder SM, Yuan Y, Rabinowitz I, Salem HH, Mitchell CA. Adhesion receptor activation of phosphatidylinositol 3-kinase. Von Willebrand factor stimulates the cytoskeletal association and activation of phosphatidylinositol 3-kinase and pp60c-src in human platelets. *Journal of Biological Chemistry*. 1994;269:27093.
- 192. Kroll M, Harris T, Moake J, Handin R, Schafer A. Von Willebrand factor binding to platelet GpIb initiates signals for platelet activation. *Journal of Clinical Investigation*. 1991;88:1568.

- Lindemann S, Krämer B, Seizer P, Gawaz M. Platelets, inflammation and atherosclerosis. Journal of Thrombosis and Haemostasis. 2007;5:203–211.
- 194. Zarbock A, Polanowska-Grabowska RK, Ley K. Platelet-neutrophil interactions: Linking hemostasis and inflammation. *Blood Reviews*. 2007;21:99–111.
- 195. Ariasnegrete S, Keller K, Chadee K. Proinflammatory cytokines regulate cyclooxygenase-2 mRNA expression in human macrophages. *Biochemical and Biophysical Research Communications*. 1995;208:582–589.
- 196. Huang ZF, Massey JB, Via DP. Differential regulation of cyclooxygenase-2 (COX-2) mRNA stability by interleukin-1 β (IL-1 β) and tumor necrosis factor- α (TNF α) in human in vitro differentiated macrophages. *Biochemical Pharmacology*. 2000;59:187–194.
- 197. Vila-del Sol V, Fresno M. Involvement of TNF and NFκβ in the transcriptional control of cyclooxygenase-2 expression by IFN-γ in macrophages. *The Journal of Immunology*. 2005;174:2825.
- 198. Maloney CG, Kutchera WA, Albertine KH, McIntyre TM, Prescott SM, Zimmerman GA. Inflammatory agonists induce cyclooxygenase type 2 expression by human neutrophils. *The Journal of Immunology*. 1998;160:1402.
- 199. Pouliot M, Gilbert C, Borgeat P, Poubelle PE, Bourgoin S, Créminon C, Maclouf J, Mccoll SR, Naccache PH. Expression and activity of prostaglandin endoperoxide synthase-2 in agonist-activated human neutrophils. *The FASEB Journal*. 1998;12:1109–1123.
- Chao W, Olson MS. Platelet-activating factor: Receptors and signal transduction. Biochemical Journal. 1993;292:617.
- 201. Smith TL, Weyrich AS. Platelets as central mediators of systemic inflammatory responses. *Thrombosis Research*. 2011;127:391–394.
- Caughey GE, Cleland LG, Penglis PS, Gamble JR, James MJ. Roles of cyclooxygenase (COX)-1 and COX-2 in prostanoid production by human endothelial cells: Selective upregulation of prostacyclin synthesis by COX-2. *The Journal of Immunology*. 2001;167:2831.
- Masferrer JL, Zweifel BS, Manning PT, Hauser SD, Leahy KM, Smith WG, Isakson PC, Seibert K. Selective inhibition of inducible cyclooxygenase 2 in vivo is antiinflammatory and nonulcerogenic. *Proceedings of the National Academy of Sciences*. 1994;91:3228.
- 204. Lee SH, Soyoola E, Chanmugam P, Hart S, Sun W, Zhong H, Liou S, Simmons D, Hwang D. Selective expression of mitogen-inducible cyclooxygenase in macrophages stimulated with lipopolysaccharide. *Journal of Biological Chemistry*. 1992;267:25934–25938.
- Diegelmann RF, Evans MC. Wound healing: An overview of acute, fibrotic and delayed healing. *Frontiers in Bioscience*. 2004;9:283–289.
- 206. Werner S, Grose R. Regulation of wound healing by growth factors and cytokines. *Physiological Reviews*. 2003;83:835–870.
- 207. Werner S, Krieg T, Smola H. Keratinocyte–fibroblast interactions in wound healing. *Journal of Investigative Dermatology*. 2007;127:998–1008.
- Serhan CN, Hamberg M, Samuelsson B. Lipoxins: Novel series of biologically active compounds formed from arachidonic acid in human leukocytes. *Proceedings of the National Academy of Sciences*. 1984;81:5335.
- 209. Serhan CN, Clish CB, Brannon J, Colgan SP, Chiang N, Gronert K. Novel functional sets of lipid-derived mediators with antiinflammatory actions generated from omega-3 fatty acids via cyclooxygenase 2–nonsteroidal antiinflammatory drugs and transcellular processing. *The Journal of Experimental Medicine*. 2000;192:1197.
- 210. Levy BD. Resolvins and protectins: Natural pharmacophores for resolution biology. *Prostaglandins, Leukotrienes and Essential Fatty Acids.* 2010;82:327–332.
- Calder PC. Polyunsaturated fatty acids and inflammatory processes: New twists in an old tale. *Biochimie*. 2009;91:791–795.
- 212. Seki H, Sasaki T, Ueda T, Arita M. Resolvins as regulators of the immune system. *The Scientific World Journal*. 2010;10:818–831.

- González-Périz A, Clària J. Resolution of adipose tissue inflammation. *The Scientific World Journal*. 2010;10:832–856.
- Serhan CN, Chiang N, Van Dyke TE. Resolving inflammation: Dual anti-inflammatory and pro-resolution lipid mediators. *Nature Reviews Immunology*. 2008;8:349–361.
- Serhan CN, Krishnamoorthy S, Recchiuti A, Chiang N. Novel anti-inflammatory– pro-resolving mediators and their receptors. *Current Topics in Medicinal Chemistry*. 2011;11:629.
- 216. Dobrian AD, Lieb DC, Cole BK, Taylor-Fishwick DA, Chakrabarti SK, Nadler JL. Functional and pathological roles of the 12-and 15-lipoxygenases. *Progress in Lipid Research*. 2011;50:115–131.
- 217. Serhan C, Chiang N. Endogenous pro-resolving and anti-inflammatory lipid mediators: A new pharmacologic genus. *British Journal of Pharmacology*. 2008;153:S200–S215.
- 218. Hong S, Gronert K, Devchand PR, Moussignac RL, Serhan CN. Novel docosatrienes and 17s-resolvins generated from docosahexaenoic acid in murine brain, human blood, and glial cells. *Journal of Biological Chemistry*. 2003;278:14677.
- Kühn H, O'Donnell VB. Inflammation and immune regulation by 12/15-lipoxygenases. Progress in Lipid Research. 2006;45:334–356.
- 220. Chavis C, Chanez P, Vachier I, Bousquet J, Michel F, Godard P. 5-15-dihete and lipoxins generated by neutrophils from endogenous arachidonic acid as asthma biomarkers. *Biochemical and Biophysical Research Communications*. 1995;207:273–279.
- 221. Thomas E, Leroux J, Blotman F, Chavis C. Conversion of endogenous arachidonic acid to 5, 15-dihete and lipoxins by polymorphonuclear cells from patients with rheumatoid arthritis. *Inflammation Research*. 1995;44:121–124.
- 222. Wu SH, Liao PY, Dong L, Chen ZQ. Signal pathway involved in inhibition by lipoxin A4 of production of interleukins induced in endothelial cells by lipopolysaccharide. *Inflammation Research*. 2008;57:430–437.
- 223. Chiang N, Serhan CN, Dahlén SE, Drazen JM, Hay DWP, Rovati GE, Shimizu T, Yokomizo T, Brink C. The lipoxin receptor ALX: Potent ligand-specific and stereoselective actions in vivo. *Pharmacological Reviews*. 2006;58:463–487.
- 224. de Waal Malefyt R, Abrams J, Bennett B, Figdor CG, de Vries JE. Interleukin 10 (IL-10) inhibits cytokine synthesis by human monocytes: An autoregulatory role of IL-10 produced by monocytes. *The Journal of Experimental Medicine*. 1991;174:1209–1220.
- 225. Wang P, Wu P, Siegel MI, Egan RW, Billah MM. Interleukin (IL)-10 inhibits nuclear factor β (NFβ) activation in human monocytes. *Journal of Biological Chemistry*. 1995;270:9558.
- 226. Fukaya T, Suzuki M, Sakaguchi R, Yasukawa H, Yoshimura A. Role of SOCS proteins in inflammation and autoimmune diseases. *Inflammation and Regeneration*. 2011;31:382–392.
- 227. Mosser DM, Zhang X. Interleukin-10: New perspectives on an old cytokine. *Immunological Reviews*. 2008;226:205–218.
- 228. Godson C, Mitchell S, Harvey K, Petasis NA, Hogg N, Brady HR. Cutting edge: Lipoxins rapidly stimulate nonphlogistic phagocytosis of apoptotic neutrophils by monocyte-derived macrophages. *The Journal of Immunology*. 2000;164:1663.
- 229. Appel S, Mirakaj V, Bringmann A, Weck MM, Grünebach F, Brossart P. PPAR-γ agonists inhibit toll-like receptor-mediated activation of dendritic cells via the map kinase and NF-κβ pathways. *Blood*. 2005;106:3888.
- 230. Lawrence T, Gilroy DW. Chronic inflammation: A failure of resolution? *International Journal of Experimental Pathology*. 2007;88:85–94.
- 231. Udalova IA, Richardson A, Denys A, Smith C, Ackerman H, Foxwell B, Kwiatkowski D. Functional consequences of a polymorphism affecting NFκβ p50-p50 binding to the TNF promoter region. *Molecular and Cellular Biology*. 2000;20:9113.

- 232. Ricart KC, Bolisetty S, Johnson MS, Perez J, Agarwal A, Murphy MP, Landar A. The permissive role of mitochondria in the induction of haeme oxygenase-1 in endothelial cells. *The Biochemical Journal*. 2009;419:427.
- 233. Kobayashi M, Li L, Iwamoto N, Nakajima-Takagi Y, Kaneko H, Nakayama Y, Eguchi M, Wada Y, Kumagai Y, Yamamoto M. The antioxidant defense system Keap1-Nrf2 comprises a multiple sensing mechanism for responding to a wide range of chemical compounds. *Molecular and Cellular Biology*. 2009;29:493.
- 234. Keum YS, Chang PPJ, Kwon KH, Yuan X, Li W, Hu L, Kong ANT. 3-Morpholinopropyl isothiocyanate is a novel synthetic isothiocyanate that strongly induces the antioxidant response element-dependent Nrf2-mediated detoxifying/antioxidant enzymes in vitro and in vivo. *Carcinogenesis*. 2008;29:594.
- 235. Biswas M, Chan JY. Role of Nrf1 in antioxidant response element-mediated gene expression and beyond. *Toxicology and Applied Pharmacology*. 2010;244:16–20.
- 236. Kwak MK, Wakabayashi N, Kensler TW. Chemoprevention through the Keap1-Nrf2 signaling pathway by phase 2 enzyme inducers. *Mutation Research/Fundamental and Molecular Mechanisms of Mutagenesis*. 2004;555:133–148.
- 237. Rhee EJ. Chemerin: A novel link between inflammation and atherosclerosis? *Diabetes & Metabolism Journal*. 2011;35:216.
- Ohira T, Arita M, Omori K, Recchiuti A, Van Dyke TE, Serhan CN. Resolvin E1 receptor activation signals phosphorylation and phagocytosis. *Journal of Biological Chemistry*. 2010;285:3451.
- 239. Krishnamoorthy S, Recchiuti A, Chiang N, Yacoubian S, Lee CH, Yang R, Petasis NA, Serhan CN. Resolvin D1 binds human phagocytes with evidence for proresolving receptors. *Proceedings of the National Academy of Sciences*. 2010;107:1660.
- 240. Schwab JM, Chiang N, Arita M, Serhan CN. Resolvin E1 and protectin D1 activate inflammation-resolution programmes. *Nature*. 2007;447:869–874.
- 241. Parkin J, Cohen B. An overview of the immune system. The Lancet. 2001;357:1777–1789.
- 242. Ruddle NH, Akirav EM. Secondary lymphoid organs: Responding to genetic and environmental cues in ontogeny and the immune response. *The Journal of Immunology*. 2009;183:2205.
- 243. Zielinski CE, Corti D, Mele F, Pinto D, Lanzavecchia A, Sallusto F. Dissecting the human immunologic memory for pathogens. *Immunological Reviews*. 2011; 240:40–51.
- Goodnow CC, Vinuesa CG, Randall KL, Mackay F, Brink R. Control systems and decision making for antibody production. *Nature Immunology*. 2010;11:681–688.
- Littman DR, Rudensky AY. Th17 and regulatory T cells in mediating and restraining inflammation. *Cell*. 2010;140:845–858.
- 246. Skapenko A, Leipe J, Lipsky PE, Schulze-Koops H. The role of the T cell in autoimmune inflammation. *Arthritis Research and Therapy*. 2005;7:4.
- 247. Kim CH, Broxmeyer HE. Chemokines: Signal lamps for trafficking of T and B cells for development and effector function. *Journal of Leukocyte Biology*. 1999;65:6–15.
- 248. Schroder K, Hertzog PJ, Ravasi T, Hume DA. Interferon-γ: An overview of signals, mechanisms and functions. *Journal of Leukocyte Biology*. 2004;75:163–189.
- 249. Luster AD, Alon R, Von Andrian UH. Immune cell migration in inflammation: Present and future therapeutic targets. *Nature Immunology*. 2005;6:1182–1190.
- 250. Iwakura Y, Ishigame H. The IL-23/IL-17 axis in inflammation. *Journal of Clinical Investigation*. 2006;116:1218.
- Moseley T, Haudenschild D, Rose L, Reddi A. Interleukin-17 family and IL-17 receptors. Cytokine & Growth Factor Reviews. 2003;14:155–174.
- 252. Lee YK, Mukasa R, Hatton RD, Weaver CT. Developmental plasticity of Th17 and Treg cells. *Current Opinion in Immunology*. 2009;21:274–280.

- 253. Williams IR. CCR6 and CCL20. Annals of the New York Academy of Sciences. 2006;1072:52–61.
- 254. Kim CH. Migration and function of Th17 cells. *Inflammation & Allergy-Drug Targets*. 2009;8:221–228.
- 255. Annunziato F, Cosmi L, Liotta F, Maggi E, Romagnani S. The phenotype of human Th17 cells and their precursors, the cytokines that mediate their differentiation and the role of Th17 cells in inflammation. *International Immunology*. 2008;20:1361.
- 256. Kleinschek MA, Boniface K, Sadekova S, Grein J, Murphy EE, Turner SP, Raskin L, Desai B, Faubion WA, de Waal Malefyt R. Circulating and gut-resident human Th17 cells express cd161 and promote intestinal inflammation. *The Journal of Experimental Medicine*. 2009;206:525.
- 257. Barry M, Bleackley RC. Cytotoxic T lymphocytes: All roads lead to death. *Nature Reviews Immunology*. 2002;2:401–409.
- Gagnon SJ, Ennis FA, Rothman AL. Bystander target cell lysis and cytokine production by dengue virus-specific human CD4⁺ cytotoxic T-lymphocyte clones. *Journal of Virology*. 1999;73:3623.
- 259. Mbawuike IN, Wells J, Byrd R, Cron SG, Glezen WP, Piedra PA. HLA-restricted CD8⁺ cytotoxic T lymphocyte, interferon-γ, and interleukin-4 responses to respiratory syncytial virus infection in infants and children. *Journal of Infectious Diseases*. 2001; 83:687.
- 260. Sallusto F, Lenig D, Förster R, Lipp M, Lanzavecchia A. Two subsets of memory T lymphocytes with distinct homing potentials and effector functions. *Nature*. 1999; 402:34–38.
- Seder RA, Ahmed R. Similarities and differences in CD4⁺ and CD8⁺ effector and memory T cell generation. *Nature Immunology*. 2003;4:835–842.
- 262. Moser B, Loetscher P. Lymphocyte traffic control by chemokines. *Nature Immunology*. 2001;2:123–128.
- Mackay CR, Marston WL, Dudler L, Spertini O, Tedder TF, Hein WR. Tissue-specific migration pathways by phenotypically distinct subpopulations of memory T cells. *European Journal of Immunology*. 1992;22:887–895.
- Sheridan BS, Lefrançois L. Regional and mucosal memory T cells. *Nature Immunology*. 2011;12:485–491.
- 265. Sanz I, Wei C, Lee F, Anolik J. Phenotypic and functional heterogeneity of human memory B cells. *Seminars in Immunology*. 2008;20:67–82.
- Lund FE, Randall TD. Effector and regulatory B cells: Modulators of CD4⁺ T cell immunity. *Nature Reviews Immunology*. 2010;10:236–247.
- 267. Maghazachi AA. Role of chemokines in the biology of natural killer cells. *The Chemokine System in Experimental and Clinical Hematology*. 2010;341:37–58.
- 268. Raulet DH. Interplay of natural killer cells and their receptors with the adaptive immune response. *Nature Immunology*. 2004;5:996–1002.
- 269. Cooper MA, Colonna M, Yokoyama WM. Hidden talents of natural killers: NK cells in innate and adaptive immunity. *EMBO Reports*. 2009;10:1103–1110.
- Sun JC, Lopez-Verges S, Kim CC, DeRisi JL, Lanier LL. NK cells and immune "memory." *The Journal of Immunology*. 2011;186:1891.
- 271. Paust S, Senman B, Von Andrian UH. Adaptive immune responses mediated by natural killer cells. *Immunological Reviews*. 2010;235:286–296.
- 272. Lünemann A, Lünemann JD, Münz C. Regulatory NK-cell functions in inflammation and autoimmunity. *Molecular Medicine*. 2009;15:352.
- 273. Vivier E, Raulet DH, Moretta A, Caligiuri MA, Zitvogel L, Lanier LL, Yokoyama WM, Ugolini S. Innate or adaptive immunity? The example of natural killer cells. *Science*. 2011;331:44.
- 274. Pacher P, Beckman JS, Liaudet L. Nitric oxide and peroxynitrite in health and disease. *Physiological Reviews*. 2007;87:315–424.

- McCord JM. Free radicals and inflammation: Protection of synovial fluid by superoxide dismutase. *Science*. 1974;185:529.
- 276. Quijano C, Hernandez-Saavedra D, Castro L, McCord JM, Freeman BA, Radi R. Reaction of peroxynitrite with Mn-superoxide dismutase. *Journal of Biological Chemistry*. 2001;276:11631–11638.
- 277. Salin M, McCord J. Free radicals and inflammation. Protection of phagocytosine leukocytes by superoxide dismutase. *Journal of Clinical Investigation*. 1975;56:1319.
- 278. Halliwell B. Reactive oxygen species in living systems: Source, biochemistry, and role in human disease. *The American Journal of Medicine*. 1991;91:S14–S22.
- Halliwell B, Cross CE. Oxygen-derived species: Their relation to human disease and environmental stress. *Environmental Health Perspectives*. 1994;102:5.
- Halliwell B, Gutteridge J. Oxygen toxicity, oxygen radicals, transition metals and disease. *Biochemical Journal*. 1984;219:1.
- BeckmanJS, KoppenolWH. Nitric oxide, superoxide, and peroxynitrite: The good, thebad, and ugly. American Journal of Physiology-Cell Physiology. 1996;271:C1424–C1437.
- Byun J, Mueller DM, Fabjan JS, Heinecke JW. Nitrogen dioxide radical generated by the myeloperoxidase-hydrogen peroxide-nitrite system promotes lipid peroxidation of low density lipoprotein. *FEBS Letters*. 1999;455:243–246.
- 283. Beckman JS. Oxidative damage and tyrosine nitration from peroxynitrite. *Chemical Research in Toxicology*. 1996;9:836–844.
- Good PF, Hsu A, Werner P, Perl DP, Olanow CW. Protein nitration in Parkinson's disease. Journal of Neuropathology and Experimental Neurology. 1998;57:338.
- Turko IV, Murad F. Protein nitration in cardiovascular diseases. *Pharmacological Reviews*. 2002;54:619–634.
- Smith MA, Harris PLR, Sayre LM, Beckman JS, Perry G. Widespread peroxynitritemediated damage in Alzheimer's disease. *The Journal of Neuroscience*. 1997;17:2653–2657.
- Castegna A, Thongboonkerd V, Klein JB, Lynn B, Markesbery WR, Butterfield DA. Proteomic identification of nitrated proteins in Alzheimer's disease brain. *Journal of Neurochemistry*. 2003;85:1394–1401.
- 288. Wiseman H, Halliwell B. Damage to DNA by reactive oxygen and nitrogen species: Role in inflammatory disease and progression to cancer. *Biochemical Journal*. 1996;313:17.
- 289. Kawai Y, Morinaga H, Kondo H, Miyoshi N, Nakamura Y, Uchida K, Osawa T. Endogenous formation of novel halogenated 2'-deoxycytidine. *Journal of Biological Chemistry*. 2004;279:51241.
- 290. Pullar JM, Winterbourn CC, Vissers M. Loss of GSH and thiol enzymes in endothelial cells exposed to sublethal concentrations of hypochlorous acid. *American Journal of Physiology-Heart and Circulatory Physiology*. 1999;277:H1505.
- Pullar JM, Vissers M, Winterbourn CC. Living with a killer: The effects of hypochlorous acid on mammalian cells. *IUBMB Life*. 2000;50:259–266.
- 292. Davies MJ, Hawkins CL, Pattison DI, Rees MD. Mammalian heme peroxidases: From molecular mechanisms to health implications. *Antioxidants & Redox Signaling*. 2008;10:1199–1234.
- Hawkins CL, Davies MJ. Hypochlorite-induced damage to DNA, RNA, and polynucleotides: Formation of chloramines and nitrogen-centered radicals. *Chemical Research in Toxicology*. 2002;15:83–92.
- 294. Sies H. Oxidative stress: Oxidants and antioxidants. *Experimental Physiology*. 1997;82:291–295.
- Winrow V, Winyard P, Morris C, Blake D. Free radicals in inflammation: Second messengers and mediators of tissue destruction. *British Medical Bulletin*. 1993;49:506.
- 296. Powers SK, Duarte J, Kavazis AN, Talbert EE. Reactive oxygen species are signalling molecules for skeletal muscle adaptation. *Experimental Physiology*. 2010;95:1–9.

- 297. Nathan C. Specificity of a third kind: Reactive oxygen and nitrogen intermediates in cell signaling. *Journal of Clinical Investigation*. 2003;111:769–778.
- 298. Thannickal VJ, Fanburg BL. Reactive oxygen species in cell signaling. *American Journal* of Physiology-Lung Cellular and Molecular Physiology. 2000;279:L1005–L1028.
- 299. Morgan MJ, Liu Z. Crosstalk of reactive oxygen species and NF-κβ signaling. *Cell Research*. 2010;21:103–115.
- Nordberg J, Arnér ESJ. Reactive oxygen species, antioxidants, and the mammalian thioredoxin system1. *Free Radical Biology and Medicine*. 2001;31:1287–1312.
- 301. Sun QA, Wu Y, Zappacosta F, Jeang KT, Lee BJ, Hatfield DL, Gladyshev VN. Redox regulation of cell signaling by selenocysteine in mammalian thioredoxin reductases. *Journal of Biological Chemistry*. 1999;274:24522.
- Finkel T. Signal transduction by reactive oxygen species. *The Journal of Cell Biology*. 2011;194:7.
- Veal EA, Day AM, Morgan BA. Hydrogen peroxide sensing and signaling. *Molecular Cell*. 2007;26:1–14.
- 304. Phalen TJ, Weirather K, Deming PB, Anathy V, Howe AK, Van Der Vliet A, Jönsson TJ, Poole LB, Heintz NH. Oxidation state governs structural transitions in peroxiredoxin ii that correlate with cell cycle arrest and recovery. *The Journal of Cell Biology*. 2006;175:779–789.
- 305. Halliwell B. Reactive species and antioxidants. Redox biology is a fundamental theme of aerobic life. *Plant Physiology*. 2006;141:312–322.
- 306. Trebak M, Ginnan R, Singer HA, Jourd'heuil D. Interplay between calcium and reactive oxygen/nitrogen species: An essential paradigm for vascular smooth muscle signaling. *Antioxidants & Redox Signaling*. 2010;12:657–674.
- 307. Machaca K. Ca²⁺ signaling, genes and the cell cycle. Cell Calcium. 2011;49:323-330.
- 308. Vig M, Kinet JP. Calcium signaling in immune cells. *Nature Immunology*. 2009;10: 21–27.
- 309. Csordás G, Hajnóczky G. SR/ER–mitochondrial local communication: Calcium and ROS. *Biochimica et Biophysica Acta (BBA)-Bioenergetics*. 2009;1787:1352–1362.
- 310. Clapham DE. Calcium signaling. Cell. 2007;131:1047-1058.
- Berridge MJ, Lipp P, Bootman MD. The versatility and universality of calcium signalling. *Nature Reviews Molecular Cell Biology*. 2000;1:11–21.
- 312. Bers DM. Calcium cycling and signaling in cardiac myocytes. *Annual Review of Physiology*. 2008;70:23–49.
- Sprague AH, Khalil RA. Inflammatory cytokines in vascular dysfunction and vascular disease. *Biochemical Pharmacology*. 2009;78:539–552.
- Codreanu SG, Zhang B, Sobecki SM, Billheimer DD, Liebler DC. Global analysis of protein damage by the lipid electrophile 4-hydroxy-2-nonenal. *Molecular & Cellular Proteomics*. 2009;8:670–680.
- 315. Minko IG, Kozekov ID, Harris TM, Rizzo CJ, Lloyd RS, Stone MP. Chemistry and biology of DNA containing 1, n 2-deoxyguanosine adducts of the α, β-unsaturated aldehydes acrolein, crotonaldehyde, and 4-hydroxynonenal. *Chemical Research in Toxicology*. 2009;22:759–778.
- 316. Valko M, Rhodes C, Moncol J, Izakovic M, Mazur M. Free radicals, metals and antioxidants in oxidative stress-induced cancer. *Chemico-Biological Interactions*. 2006;160:1–40.
- 317. Bosch FX, Ribes J, Borràs J. Epidemiology of primary liver cancer. *Seminars in Liver Disease*. 2008;19:271–285.
- 318. Grivennikov SI, Greten FR, Karin M. Immunity, inflammation, and cancer. *Cell*. 2010;140:883–899.
- 319. Jones PA, Baylin SB. The epigenomics of cancer. Cell. 2007;128:683-692.

- 320. Laviano A, Meguid MM, Preziosa I, Fanelli FR. Oxidative stress and wasting in cancer. *Current Opinion in Clinical Nutrition & Metabolic Care*. 2007;10:449.
- 321. White JP, Baltgalvis KA, Puppa MJ, Sato S, Baynes JW, Carson JA. Muscle oxidative capacity during IL-6-dependent cancer cachexia. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology.* 2011;300:R201–R211.
- 322. Langstein H, Norton J. Mechanisms of cancer cachexia. *Hematology/Oncology Clinics* of North America. 1991;5:103.
- Mcnamara MJ, Alexander HR, Norton JA. Cytokines and their role in the pathophysiology of cancer cachexia. *Journal of Parenteral and Enteral Nutrition*. 1992;16:508–555.
- 324. Tisdale MJ. Molecular pathways leading to cancer cachexia. *Physiology (Bethesda, Md.)*. 2005;20:340.
- 325. Ristow M, Zarse K, Oberbach A, Klöting N, Birringer M, Kiehntopf M, Stumvoll M, Kahn CR, Blüher M. Antioxidants prevent health-promoting effects of physical exercise in humans. *Proceedings of the National Academy of Sciences*. 2009;106:8665.
- 326. Gomez-Cabrera MC, Domenech E, Viña J. Moderate exercise is an antioxidant: Upregulation of antioxidant genes by training. *Free Radical Biology and Medicine*. 2008;44:126–131.
- 327. Sharman I, Down M, Sen R. The effects of vitamin E and training on physiological function and athletic performance in adolescent swimmers. *British Journal of Nutrition*. 1971;26:265–276.
- 328. Malm C, Svensson M, Ekblom B, Sjödin B. Effects of ubiquinone-10 supplementation and high intensity training on physical performance in humans. *Acta Physiologica Scandinavica*. 1997;161:379–384.
- 329. Malm C, Svensson M, Sjöberg B, Ekblom B, Sjödin B. Supplementation with ubiquinone-10 causes cellular damage during intense exercise. *Acta Physiologica Scandinavica*. 1996;157:511–512.
- 330. Goodman M, Bostick RM, Kucuk O, Jones DP. Clinical trials of antioxidants as cancer prevention agents: Past, present, and future. *Free Radical Biology and Medicine*. 2011;51:1068.
- 331. Lin J, Cook NR, Albert C, Zaharris E, Gaziano JM, Van Denburgh M, Buring JE, Manson JAE. Vitamins C and E and β carotene supplementation and cancer risk: A randomized controlled trial. *Journal of the National Cancer Institute*. 2009; 101: 14–23.
- 332. Malila N, Taylor PR, Virtanen MJ, Korhonen P, Huttunen JK, Albanes D, Virtamo J. Effects of alpha-tocopherol and beta-carotene supplementation on gastric cancer incidence in male smokers (ATBC study, Finland). *Cancer Causes and Control*. 2002;13:617–623.
- 333. Omenn GS, Goodman GE, Thornquist MD, Balmes J, Cullen MR, Glass A, Keogh JP, Meyskens Jr FL, Valanis B, Williams Jr JH. Risk factors for lung cancer and for intervention effects in CARET, the beta-carotene and retinol efficacy trial. *Journal of the National Cancer Institute*. 1996;88:1550–1559.
- 334. Miller III ER, Pastor-Barriuso R, Dalal D, Riemersma RA, Appel LJ, Guallar E. Metaanalysis: High-dosage vitamin E supplementation may increase all-cause mortality. *Annals of Internal Medicine*. 2005;142:37–46.
- 335. Lonn E, Bosch J, Yusuf S, Sheridan P, Pogue J, Arnold J, Ross C, Arnold A, Sleight P, Probstfield J. Effects of long-term vitamin E supplementation on cardiovascular events and cancer: A randomized controlled trial. *JAMA: The Journal of the American Medical Association*. 2005;293:1338.
- 336. Jha P, Flather M, Lonn E, Farkouh M, Yusuf S. The antioxidant vitamins and cardiovascular disease: A critical review of epidemiologic and clinical trial data. *Annals of Internal Medicine*. 1995;123:860–872.

- 337. Bjelakovic G, Nikolova D, Gluud LL, Simonetti RG, Gluud C. Mortality in randomized trials of antioxidant supplements for primary and secondary prevention. JAMA: The Journal of the American Medical Association. 2007;297:842–857.
- 338. Kimmons J, Gillespie C, Seymour J, Serdula M, Blanck HM. Fruit and vegetable intake among adolescents and adults in the united states: Percentage meeting individualized recommendations. *The Medscape Journal of Medicine*. 2009;11:26.
- 339. Schleicher RL, Carroll MD, Ford ES, Lacher DA. Serum vitamin C and the prevalence of vitamin C deficiency in the United States: 2003–2004 National Health and Nutrition Examination Survey (NHANES). *The American Journal of Clinical Nutrition*. 2009;90:1252–1263.
- 340. Fulgoni III VL, Keast DR, Bailey RL, Dwyer J. Foods, fortificants, and supplements: Where do Americans get their nutrients? *The Journal of Nutrition*. 2011;141:1847–1854.
- 341. Maughan R, Greenhaff P, Hespel P. Dietary supplements for athletes: Emerging trends and recurring themes. *Journal of Sports Sciences*. 2011;29:57–66.
- 342. Clarkson PM. Minerals: Exercise performance and supplementation in athletes. *Journal* of Sports Sciences. 1991;9:91–116.
- 343. Institute of Medicine (US). Standing Committee on the Scientific Evaluation of Dietary Reference Intakes. Dietary reference intakes for energy, carbohydrate, fiber, fat, fatty acids, cholesterol, protein, and amino acids. 2005; vol. 1. National Academy Press. Washington, D.C.
- 344. Harris WS, Mozaffarian D, Rimm E, Kris-Etherton P, Rudel LL, Appel LJ, Engler MM, Engler MB, Sacks F. Omega-6 fatty acids and risk for cardiovascular disease. *Circulation*. 2009;119:902–907.
- 345. Katan MB. Omega-6 polyunsaturated fatty acids and coronary heart disease. *The American Journal of Clinical Nutrition*. 2009;89:1283–1284.
- 346. Sarkkinen ES, Agren JJ, Ahola I, Ovaskainen ML, Uusitupa MI. Fatty acid composition of serum cholesterol esters, and erythrocyte and platelet membranes as indicators of long-term adherence to fat-modified diets. *The American Journal of Clinical Nutrition*. 1994;59:364–370.
- 347. Burdge GC, Calder PC. Conversion of α -linolenic acid to longer-chain polyunsaturated fatty acids in human adults. *Reproduction Nutrition Development*. 2005;45:581–598.
- 348. Arterburn LM, Hall EB, Oken H. Distribution, interconversion, and dose response of n–3 fatty acids in humans. *The American Journal of Clinical Nutrition*. 2006;83: 1467S–1476S.
- 349. De Lorgeril M, Salen P, Laporte F, De Leiris J. Alpha-linolenic acid in the prevention and treatment of coronary heart disease. *European Heart Journal Supplements*. 2001;3:D26.
- 350. Zhao G, Etherton TD, Martin KR, West SG, Gillies PJ, Kris-Etherton PM. Dietary alpha-linolenic acid reduces inflammatory and lipid cardiovascular risk factors in hypercholesterolemic men and women. *The Journal of Nutrition*. 2004;134:2991–2997.
- 351. Simopoulos AP. Omega-6/omega-3 essential fatty acids: biological effects. In: Omega-3 Fatty Acids, the Brain and Retina. *World Review of Nutrition and Dietetics*. 2009;99:1–16.
- 352. Harris WS, Poston WC, Haddock CK. Tissue n–3 and n–6 fatty acids and risk for coronary heart disease events. *Atherosclerosis*. 2007;193:1–10.
- 353. Zatonski W, Campos H, Willett W. Rapid declines in coronary heart disease mortality in Eastern Europe are associated with increased consumption of oils rich in alpha-linolenic acid. *European Journal of Epidemiology*. 2008;23:3–10.
- 354. Mozaffarian D, Rimm EB. Fish intake, contaminants, and human health. *JAMA: The Journal of the American Medical Association*. 2006;296:1885–1899.

- 355. Li H, Ruan XZ, Powis SH, Fernando R, Mon WY, Wheeler DC, Moorhead JF, Varghese Z. EPA and DHA reduce LPS-induced inflammation responses in Hk-2 cells: Evidence for a PPAR-γ-dependent mechanism. *Kidney International*. 2005;67:867–874.
- 356. Eibl G. The role of PPAR-γ and its interaction with COX-2 in pancreatic cancer. *PPAR Research*. 2008;2008:326915.
- 357. Magee P, Pearson S, Whittingham-Dowd J, Allen J. PPARγ as a molecular target of EPA anti-inflammatory activity during TNF-α-impaired skeletal muscle cell differentiation. *The Journal of Nutritional Biochemistry*. 2012;23:1440–1448.
- 358. Neveu V, Perez-Jiménez J, Vos F, Crespy V, Du Chaffaut L, Mennen L, Knox C, Eisner R, Cruz J, Wishart D. Phenol-explorer: An online comprehensive database on polyphenol contents in foods. *Database: The Journal of Biological Databases and Curation*. 2010;doi: 10.1093/database/bap024.
- 359. Pauwels EKJ. The protective effect of the mediterranean diet: Focus on cancer and cardiovascular risk. *Medical Principles and Practice*. 2011;20:103–111.
- 360. Scalbert A, Williamson G. Dietary intake and bioavailability of polyphenols. *The Journal of Nutrition*. 2000;130:2073S–2085S.
- 361. Martin KR, Appel CL. Polyphenols as dietary supplements: A double-edged sword. *Nutrition and Dietary Supplements*. 2010;2:1–12.
- 362. Chun OK, Kim DO, Smith N, Schroeder D, Han JT, Lee CY. Daily consumption of phenolics and total antioxidant capacity from fruit and vegetables in the American diet. *Journal of the Science of Food and Agriculture*. 2005;85:1715–1724.
- 363. Dilis V, Trichopoulou A. Antioxidant intakes and food sources in Greek adults. *The Journal of Nutrition*. 2010;140:1274–1279.
- 364. Rahman I, Biswas SK, Kirkham PA. Regulation of inflammation and redox signaling by dietary polyphenols. *Biochemical Pharmacology*. 2006;72:1439–1452.
- 365. Sharma R, Gescher A, Steward W. Curcumin: The story so far. *European Journal of Cancer*. 2005;41:1955–1968.
- 366. Fukino Y, Shimbo M, Aoki N, Okubo T, Iso H. Randomized controlled trial for an effect of green tea consumption on insulin resistance and inflammation markers. *Journal of Nutritional Science and Vitaminology*. 2005;51:335.
- 367. Ryu O, Lee J, Lee K, Kim H, Seo J, Kim S, Kim N, Baik S, Choi D, Choi K. Effects of green tea consumption on inflammation, insulin resistance and pulse wave velocity in type 2 diabetes patients. *Diabetes Research and Clinical Practice*. 2006;71:356–358.
- 368. Bae SC, Jung WJ, Lee EJ, Yu R, Sung MK. Effects of antioxidant supplements intervention on the level of plasma inflammatory molecules and disease severity of rheumatoid arthritis patients. *Journal of the American College of Nutrition*. 2009;28:56–62.
- 369. Mertens-Talcott SU, Jilma-Stohlawetz P, Rios J, Hingorani L, Derendorf H. Absorption, metabolism, and antioxidant effects of pomegranate (*Punica granatum* L.) polyphenols after ingestion of a standardized extract in healthy human volunteers. *Journal of Agricultural and Food Chemistry*. 2006;54:8956–8961.
- 370. Freese R, Vaarala O, Turpeinen AM, Mutanen M. No difference in platelet activation or inflammation markers after diets rich or poor in vegetables, berries and apple in healthy subjects. *European Journal of Nutrition*. 2004;43:175–182.
- 371. Widlansky ME, Duffy SJ, Hamburg NM, Gokce N, Warden BA, Wiseman S, Keaney JF, Frei B, Vita JA. Effects of black tea consumption on plasma catechins and markers of oxidative stress and inflammation in patients with coronary artery disease. *Free Radical Biology and Medicine*. 2005;38:499–506.
- 372. De Maat M, Pijl H, Kluft C, Princen H. Consumption of black and green tea had no effect on inflammation, haemostasis and endothelial markers in smoking healthy individuals. *European Journal of Clinical Nutrition*. 2000;54:757.
- 373. Kelley DS, Rasooly R, Jacob RA, Kader AA, Mackey BE. Consumption of Bing sweet cherries lowers circulating concentrations of inflammation markers in healthy men and women. *The Journal of Nutrition*. 2006;136:981–986.
- 374. Kar P, Laight D, Rooprai H, Shaw K, Cummings M. Effects of grape seed extract in type 2 diabetic subjects at high cardiovascular risk: A double blind randomized placebo controlled trial examining metabolic markers, vascular tone, inflammation, oxidative stress and insulin sensitivity. *Diabetic Medicine*. 2009;26:526–531.
- 375. Castilla P, Dávalos A, Teruel JL, Cerrato F, Fernández-Lucas M, Merino JL, Sánchez-Martín CC, Ortuño J, Lasunción MA. Comparative effects of dietary supplementation with red grape juice and vitamin E on production of superoxide by circulating neutrophil NADPH oxidase in hemodialysis patients. *The American Journal of Clinical Nutrition*. 2008;87:1053–1061.
- 376. Karlsen A, Retterstøl L, Laake P, Paur I, Kjølsrud-Bøhn S, Sandvik L, Blomhoff R. Anthocyanins inhibit nuclear factor-κβ activation in monocytes and reduce plasma concentrations of pro-inflammatory mediators in healthy adults. *The Journal of Nutrition*. 2007;137:1951–1954.
- 377. Santangelo C, Varì R, Scazzocchio B, Di Benedetto R, Filesi C, Masella R. Polyphenols, intracellular signalling and inflammation. *Annali-Istituto Superiore Di Sanita*. 2007; 43:394.
- 378. Dolinsky VW, Dyck JRB. Calorie restriction and resveratrol in cardiovascular health and disease. *Biochimica et Biophysica Acta (BBA)-Molecular Basis of Disease*. 2011; 1812:1477–1489.
- Yap S, Qin C, Woodman OL. Effects of resveratrol and flavonols on cardiovascular function: Physiological mechanisms. *Biofactors*. 2010;36:350–359.
- Zhao L, Lee JY, Hwang DH. Inhibition of pattern recognition receptor-mediated inflammation by bioactive phytochemicals. *Nutrition Reviews*. 2011;69:310–320.
- 381. Lucas L, Russell A, Keast R. Molecular mechanisms of inflammation. Antiinflammatory benefits of virgin olive oil and the phenolic compound oleocanthal. *Current Pharmaceutical Design*. 2011;17:754–768.
- 382. Surh YJ, Kundu JK, Na HK, Lee JS. Redox-sensitive transcription factors as prime targets for chemoprevention with anti-inflammatory and antioxidative phytochemicals. *The Journal of Nutrition*. 2005;135:2993S–3001S.
- 383. Surh YJ, Na HK. NF- $\kappa\beta$ and Nrf2 as prime molecular targets for chemoprevention and cytoprotection with anti-inflammatory and antioxidant phytochemicals. *Genes & Nutrition*. 2008;2:313–317.
- González-Gallego J, García-Mediavilla M, Sánchez-Campos S, Tuñón MJ. Fruit polyphenols, immunity and inflammation. *British Journal of Nutrition*. 2010;104:S15–S27.
- Guo W, Kong EH, Meydani M. Dietary polyphenols, inflammation, and cancer. *Nutrition and Cancer*. 2009;61:807–810.
- 386. Pan MH, Lai CS, Dushenkov S, Ho CT. Modulation of inflammatory genes by natural dietary bioactive compounds. *Journal of Agricultural and Food Chemistry*. 2009; 57:4467–4477.
- 387. Kim YS, Young MR, Bobe G, Colburn NH, Milner JA. Bioactive food components, inflammatory targets, and cancer prevention. *Cancer Prevention Research*. 2009; 2:200–208.
- Park JB. 5-Caffeoylquinic acid and caffeic acid orally administered suppress P-selectin expression on mouse platelets. *The Journal of Nutritional Biochemistry*. 2009; 20:800–805.
- 389. Lee JH, Zhou HY, Cho SY, Kim YS, Lee YS, Jeong CS. Anti-inflammatory mechanisms of apigenin: Inhibition of cyclooxygenase-2 expression, adhesion of monocytes to human umbilical vein endothelial cells, and expression of cellular adhesion molecules. *Archives of Pharmacal Research*. 2007;30:1318–1327.

- 390. Harris GK, Qian Y, Leonard SS, Sbarra DC, Shi X. Luteolin and chrysin differentially inhibit cyclooxygenase-2 expression and scavenge reactive oxygen species but similarly inhibit prostaglandin-e2 formation in RAW264.7 cells. *The Journal of Nutrition*. 2006;136:1517–1521.
- 391. Covas MI. Olive oil and the cardiovascular system. *Pharmacological Research*. 2007;55:175–186.
- 392. Cicerale S, Lucas L, Keast R. Biological activities of phenolic compounds present in virgin olive oil. *International Journal of Molecular Sciences*. 2010;11:458–479.
- Miao L, St Clair DK. Regulation of superoxide dismutase genes: Implications in disease. Free Radical Biology and Medicine. 2009;47:344–356.
- 394. Na HK, Surh YJ. Modulation of Nrf2-mediated antioxidant and detoxifying enzyme induction by the green tea polyphenol EGCG. *Food and Chemical Toxicology*. 2008;46:1271–1278.
- 395. Surh YJ, Kundu JK, Na HK. Nrf2 as a master redox switch in turning on the cellular signaling involved in the induction of cytoprotective genes by some chemopreventive phytochemicals. *Planta Medica*. 2008;74:1526.
- 396. Kim YJ, Baek SH, Bogner PN, Ip C, Rustum YM, Fakih MG, Ramnath N, Park YM. Targeting the Nrf2-Prx1 pathway with selenium to enhance the efficacy and selectivity of cancer therapy. *Journal of Cancer Molecules*. 2007;3:37–43.
- 397. Miyamoto N, Izumi H, Miyamoto R, Kondo H, Tawara A, Sasaguri Y, Kohno K. Quercetin induces the expression of peroxiredoxins 3 and 5 via the Nrf2/Nrf1 transcription pathway. *Investigative Ophthalmology & Visual Science*. 2011;52:1055–1063.
- 398. Rushworth SA, Ogborne RM, Charalambos CA, O'Connell MA. Role of protein kinase Cδ in curcumin-induced antioxidant response element-mediated gene expression in human monocytes. *Biochemical and Biophysical Research Communications*. 2006;341:1007–1016.
- Hanneken A, Lin FF, Johnson J, Maher P. Flavonoids protect human retinal pigment epithelial cells from oxidative-stress-induced death. *Investigative Ophthalmology & Visual Science*. 2006;47:3164–3177.
- 400. Tanigawa S, Fujii M, Hou DX. Action of Nrf2 and Keap1 in ARE-mediated NQO1 expression by quercetin. *Free Radical Biology and Medicine*. 2007;42:1690–1703.
- 401. Kluth D, Banning A, Paur I, Blomhoff R, Brigelius-Flohé R. Modulation of pregnane X receptor- and electrophile responsive element-mediated gene expression by dietary polyphenolic compounds. *Free Radical Biology and Medicine*. 2007;42:315–325.
- 402. Chen CY, Jang JH, Li MH, Surh YJ. Resveratrol upregulates heme oxygenase-1 expression via activation of Nf-E2-related factor 2 in Pc12 cells. *Biochemical and Biophysical Research Communications*. 2005;331:993–1000.
- 403. Kode A, Rajendrasozhan S, Caito S, Yang SR, Megson IL, Rahman I. Resveratrol induces glutathione synthesis by activation of Nrf2 and protects against cigarette smoke-mediated oxidative stress in human lung epithelial cells. *American Journal of Physiology-Lung Cellular and Molecular Physiology*. 2008;294:L478–L488.
- 404. Hsieh T, Lu X, Wang Z, Wu JM. Induction of quinone reductase NQO1 by resveratrol in human K562 cells involves the antioxidant response element ARE and is accompanied by nuclear translocation of transcription factor Nrf2. *Medicinal Chemistry*. 2006;2:275–285.
- 405. Ben-Dor A, Steiner M, Gheber L, Danilenko M, Dubi N, Linnewiel K, Zick A, Sharoni Y, Levy J. Carotenoids activate the antioxidant response element transcription system. *Molecular Cancer Therapeutics*. 2005;4:177.
- 406. Gong P, Hu B, Cederbaum AI. Diallyl sulfide induces heme oxygenase-1 through MAPK pathway. *Archives of Biochemistry and Biophysics*. 2004;432:252–260.
- 407. Keum YS, Yu S, Chang PPJ, Yuan X, Kim JH, Xu C, Han J, Agarwal A, Kong ANT. Mechanism of action of sulforaphane: Inhibition of p38 mitogen-activated protein

kinase isoforms contributing to the induction of antioxidant response element-mediated heme oxygenase-1 in human hepatoma HEPG2 cells. *Cancer Research*. 2006;66:8804.

- 408. Thimmulappa RK, Mai KH, Srisuma S, Kensler TW, Yamamoto M, Biswal S. Identification of Nrf2-regulated genes induced by the chemopreventive agent sulforaphane by oligonucleotide microarray. *Cancer Research*. 2002;62:5196.
- 409. Garg R, Gupta S, Maru GB. Dietary curcumin modulates transcriptional regulators of phase I and phase II enzymes in benzo [a] pyrene-treated mice: Mechanism of its antiinitiating action. *Carcinogenesis*. 2008;29:1022–1032.
- 410. Dickinson DA, Iles KE, Zhang H, Blank V, Forman HJ. Curcumin alters EPRE and AP-1 binding complexes and elevates glutamate-cysteine ligase gene expression. *The FASEB Journal*. 2003;17:473–475.
- 411. Kaspar JW, Niture SK, Jaiswal AK. Nrf2: Inrf2 (Keap1) signaling in oxidative stress. *Free Radical Biology and Medicine*. 2009;47:1304–1309.
- 412. Heim KE, Tagliaferro AR, Bobilya DJ. Flavonoid antioxidants: Chemistry, metabolism and structure-activity relationships. *The Journal of Nutritional Biochemistry*. 2002;13:572–584.
- Nakamura Y, Miyoshi N. Electrophiles in foods: The current status of isothiocyanates and their chemical biology. *Bioscience, Biotechnology, and Biochemistry*. 2010;74:242–255.
- 414. Sang S, Yang I, Buckley B, Ho CT, Yang CS. Autoxidative quinone formation in vitro and metabolite formation in vivo from tea polyphenol (–)-epigallocatechin-3-gallate: Studied by real-time mass spectrometry combined with tandem mass ion mapping. *Free Radical Biology and Medicine*. 2007;43:362–371.
- Maines MD. Heme oxygenase: Function, multiplicity, regulatory mechanisms, and clinical applications. *The FASEB Journal*. 1988;2:2557–2568.
- 416. Ryter SW, Alam J, Choi AMK. Heme oxygenase-1/carbon monoxide: From basic science to therapeutic applications. *Physiological Reviews*. 2006;86:583–650.
- 417. Morse D, Pischke SE, Zhou Z, Davis RJ, Flavell RA, Loop T, Otterbein SL, Otterbein LE, Choi AMK. Suppression of inflammatory cytokine production by carbon monoxide involves the JNK pathway and AP-1. *Science's STKE*. 2003;278:36993.
- 418. Otterbein LE, Bach FH, Alam J, Soares M, Lu HT, Wysk M, Davis RJ, Flavell RA, Choi AMK. Carbon monoxide has anti-inflammatory effects involving the mitogenactivated protein kinase pathway. *Nature Medicine*. 2000;6:422–428.
- 419. Sawle P, Foresti R, Mann BE, Johnson TR, Green CJ, Motterlini R. Carbon monoxidereleasing molecules (CO-RMs) attenuate the inflammatory response elicited by lipopolysaccharide in RAW264. 7 murine macrophages. *British Journal of Pharmacology*. 2005;145:800–810.
- 420. Soares MP, Seldon MP, Gregoire IP, Vassilevskaia T, Berberat PO, Yu J, Tsui TY, Bach FH. Heme oxygenase-1 modulates the expression of adhesion molecules associated with endothelial cell activation. *The Journal of Immunology*. 2004;172:3553–3563.
- 421. Brady PS, Brady LJ, Ullrey DE. Selenium, vitamin E and the response to swimming stress in the rat. *The Journal of Nutrition*. 1979;109:1103.
- 422. Davies KJA, Quintanilha AT, Brooks GA, Packer L. Free radicals and tissue damage produced by exercise. *Biochemical and Biophysical Research Communications*. 1982;107:1198–1205.
- 423. Miles MP, Andring JM, Pearson SD, Gordon LK, Kasper C, Depner CM, Kidd JR. Diurnal variation, response to eccentric exercise, and association of inflammatory mediators with muscle damage variables. *Journal of Applied Physiology*. 2008;104: 451–458.
- 424. MacIntyre DL, Sorichter S, Mair J, Berg A, McKenzie DC. Markers of inflammation and myofibrillar proteins following eccentric exercise in humans. *European Journal of Applied Physiology*. 2001;84:180–186.

- 425. Shephard RJ. Cytokine responses to physical activity, with particular reference to IL-6: Sources, actions, and clinical implications. *Critical Reviews in Immunology*. 2002;22:165–182.
- 426. Peake JM, Suzuki K, Hordern M, Wilson G, Nosaka K, Coombes JS. Plasma cytokine changes in relation to exercise intensity and muscle damage. *European Journal of Applied Physiology*. 2005;95:514–521.
- 427. Pedersen BK. Muscles and their myokines. *Journal of Experimental Biology*. 2011;214: 337–346.
- 428. Pedersen M, Steensberg A, Keller C, Osada T, Zacho M, Saltin B, Febbraio MA, Pedersen BK. Does the aging skeletal muscle maintain its endocrine function. *Exercise Immunology Review*. 2004;10:42–55.
- 429. Helge JW, Stallknecht B, Pedersen BK, Galbo H, Kiens B, Richter EA. The effect of graded exercise on IL-6 release and glucose uptake in human skeletal muscle. *The Journal of Physiology*. 2003;546:299–305.
- 430. Steensberg A, Febbraio MA, Osada T, Schjerling P, Van Hall G, Saltin B, Pedersen BK. Interleukin-6 production in contracting human skeletal muscle is influenced by preexercise muscle glycogen content. *The Journal of Physiology*. 2001;537:633–639.
- 431. Keller C, Steensberg A, Pilegaard H, Osada T, Saltin B, Pedersen BK, Neufer PD. Transcriptional activation of the IL-6 gene in human contracting skeletal muscle: Influence of muscle glycogen content. *The FASEB Journal*. 2001;15:2748–2750.
- 432. Marcell TJ, McAuley KA, Traustadóttir T, Reaven PD. Exercise training is not associated with improved levels of C-reactive protein or adiponectin. *Metabolism*. 2005;54:533–541.
- 433. Mattusch F, Dufaux B, Heine O, Mertens I, Rost R. Reduction of the plasma concentration of C-reactive protein following nine months of endurance training. *International Journal of Sports Medicine*. 2000;21:21–24.
- 434. Nicklas BJ, Ambrosius W, Messier SP, Miller GD, Penninx BW, Loeser RF, Palla S, Bleecker E, Pahor M. Diet-induced weight loss, exercise, and chronic inflammation in older, obese adults: A randomized controlled clinical trial. *The American Journal of Clinical Nutrition*. 2004;79:544–551.
- 435. Friedenreich CM, Neilson HK, Woolcott CG, Wang Q, Stanczyk FZ, McTiernan A, Jones CA, Irwin ML, Yasui Y, Courneya KS. Inflammatory marker changes in a yearlong randomized exercise intervention trial among postmenopausal women. *Cancer Prevention Research*. 2012;5:98–108.
- 436. Fairey AS, Courneya KS, Field CJ, Bell GJ, Jones LW, Martin BS, Mackey JR. Effect of exercise training on C-reactive protein in postmenopausal breast cancer survivors: A randomized controlled trial. *Brain, Behavior, and Immunity.* 2005;19:381–388.
- 437. Kohut M, McCann D, Russell D, Konopka D, Cunnick J, Franke W, Castillo M, Reighard A, Vanderah E. Aerobic exercise, but not flexibility/resistance exercise, reduces serum IL-18, CRP, and IL-6 independent of β-blockers, BMI, and psychosocial factors in older adults. *Brain, Behavior, and Immunity; Brain, Behavior, and Immunity*. 2006;20:201–209.
- 438. Kelly AS, Wetzsteon RJ, Kaiser DR, Steinberger J, Bank AJ, Dengel DR. Inflammation, insulin, and endothelial function in overweight children and adolescents: The role of exercise. *The Journal of Pediatrics*. 2004;145:731–736.
- 439. Nicklas BJ, Hsu FC, Brinkley TJ, Church T, Goodpaster BH, Kritchevsky SB, Pahor M. Exercise training and plasma C-reactive protein and interleukin-6 in elderly people. *Journal of the American Geriatrics Society*. 2008;56:2045–2052.
- 440. Campbell PT, Campbell KL, Wener MH, Wood BL, Potter JD, McTiernan A, Ulrich CM. A yearlong exercise intervention decreases CRP among obese postmenopausal women. *Medicine and Science in Sports and Exercise*. 2009;41:1533.

- 441. Ford ES. Does exercise reduce inflammation? Physical activity and C-reactive protein among U.S. adults. *Epidemiology*. 2002;13:561–568.
- 442. Jackson MJ, Pye D, Palomero J. The production of reactive oxygen and nitrogen species by skeletal muscle. *Journal of Applied Physiology*. 2007;102:1664–1670.
- 443. Marini M, Veicsteinas A. The exercised skeletal muscle: A review. *European Journal Translational Myology*. 2010;20:105–120.
- 444. Duncan K, Harris S, Ardies CM. Running exercise may reduce risk for lung and liver cancer by inducing activity of antioxidant and phase ii enzymes. *Cancer Letters*. 1997;116:151–158.
- 445. Ardies CM, Smith TJ, Kim S, Yang CS. Induction of 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanone (NNK) activation in rat lung microsomes by chronic ethanol consumption and repeated running exercise. *Cancer Letters*. 1996;103:209–218.
- 446. Walsh NP, Gleeson M, Shephard RJ, Woods J, Bishop N, Fleshner M, Green C, Pedersen B, Hoffman-Goetz L, Rogers C. Position statement. Part one: Immune function and exercise. *Exercise Immunology Review*. 2011;17:6–63.
- 447. Gleeson M, Bishop NC, Stensel DJ, Lindley MR, Mastana SS, Nimmo MA. The antiinflammatory effects of exercise: Mechanisms and implications for the prevention and treatment of disease. *Nature Reviews Immunology*. 2011;11:607–615.
- 448. Jolitha A, Subramanyam M, Asha Devi S. Modification by vitamin E and exercise of oxidative stress in regions of aging rat brain: Studies on superoxide dismutase isoenzymes and protein oxidation status. *Experimental Gerontology*. 2006;41:753–763.
- 449. Gomez-Cabrera MC, Sanchis-Gomar F, Martinez-Bello VE, Ibanez-Sania S, Nascimento AL, Ji LL, Vina J. Exercise as a model to study oxidative stress. In: *Studies on Experimental Models* Eds., Basu S and Wiklund L. 2011, pp 531–542. Humana Press, Totowa, NJ.
- 450. Asghar M, George L, Lokhandwala MF. Exercise decreases oxidative stress and inflammation and restores renal dopamine d1 receptor function in old rats. *American Journal of Physiology-Renal Physiology*. 2007;293:F914–F919.
- 451. Safdar A, deBeer J, Tarnopolsky MA. Dysfunctional Nrf2-Keap1 redox signaling in skeletal muscle of the sedentary old. *Free Radical Biology and Medicine*. 2010;49: 1487–1493.
- 452. Eynon N, Sagiv M, Meckel Y, Duarte JA, Alves AJ, Yamin C, Sagiv M, Goldhammer E, Oliveira J. NRF2 intron 3 A/G polymorphism is associated with endurance athletes' status. *Journal of Applied Physiology*. 2009;107:76–79.
- 453. Muthusamy VR, Kannan S, Sadhaasivam K, Gounder SS, Davidson CJ, Boeheme C, Hoidal JR, Wang L, Soorappan RN. Acute exercise stress activates Nrf2/ARE signaling and promotes antioxidant mechanisms in the myocardium. *Free Radical Biology and Medicine*. 2012;52:366–376.
- 454. George L, Asghar M, Lokhandwala MF. Exercise stimulates transcription factors (Nrf2 & NFκβ), increases antioxidant defenses, decreases oxidative stress, and restores renal dopamine d1 receptor function in aging. *The FASEB Journal*. 2008;(Abstract) Supplement 1159.6.
- 455. Gorecki M, Beck Y, Hartman JR, Fischer M, Weiss L, Tochner Z, Slavin S, Nimrod A. Recombinant human superoxide dismutases: Production and potential therapeutical uses. *Free Radical Research*. 1991;12:401–410.
- 456. Mellman W, Schimke R, Hayflick L. Catalase turnover in human diploid cell cultures. *Experimental Cell Research*. 1972;73:399–409.
- 457. Chada S, Whitney C, Newburger PE. Post-transcriptional regulation of glutathione peroxidase gene expression by selenium in the HL-60 human myeloid cell line. *Blood*. 1989;74:2535–2541.
- 458. Gallegos A, Berggren M, Gasdaska JR, Powis G. Mechanisms of the regulation of thioredoxin reductase activity in cancer cells by the chemopreventive agent selenium. *Cancer Research*. 1997;57:4965.

- 459. Leesa SJ, Booth FW. Physical inactivity is a disease. *World Review of Nutrition and Dietetics*. 2005;95:73–79.
- 460. Booth FW, Lees SJ. Fundamental questions about genes, inactivity, and chronic diseases. *Physiological Genomics*. 2007;28:146–157.
- 461. Katzmarzyk PT, Gledhill N, Shephard RJ. The economic burden of physical inactivity in Canada. *Canadian Medical Association Journal*. 2000;163:1435.
- 462. Katzmarzyk PT, Janssen I. The economic costs associated with physical inactivity and obesity in Canada: An update. *Canadian Journal of Applied Physiology*. 2004;29:90–115.
- 463. Colditz GA. Economic costs of obesity and inactivity. *Medicine & Science in Sports & Exercise*. 1999;31:S663.
- 464. Courneya KS. Perceived severity of the consequences of physical inactivity across the stages of change in older adults. *Journal of Sport & Exercise Psychology*. 1995;17:447–457.
- 465. Booth FW, Gordon SE, Carlson CJ, Hamilton MT. Waging war on modern chronic diseases: Primary prevention through exercise biology. *Journal of Applied Physiology*. 2000;88:774–787.
- 466. Pedersen BK. The diseasome of physical inactivity–and the role of myokines in muscle– fat cross talk. *The Journal of Physiology*. 2009;587:5559–5568.
- 467. Powers SK, Kavazis AN, DeRuisseau KC. Mechanisms of disuse muscle atrophy: Role of oxidative stress. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology*. 2005;288:R337–R344.
- Powers SK, Kavazis AN, McClung JM. Oxidative stress and disuse muscle atrophy. Journal of Applied Physiology. 2007;102:2389–2397.
- 469. Hamburg NM, McMackin CJ, Huang AL, Shenouda SM, Widlansky ME, Schulz E, Gokce N, Ruderman NB, Keaney Jr JF, Vita JA. Physical inactivity rapidly induces insulin resistance and microvascular dysfunction in healthy volunteers. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2007;27:2650–2656.
- 470. Lloyd-Jones D, Adams R, Carnethon M, De Simone G, Ferguson TB, Flegal K, Ford E, Furie K, Go A, Greenlund K. Heart disease and stroke statistics—2009 update: A report from the American Heart Association statistics committee and stroke statistics subcommittee. *Circulation*. 2009;119:e21–e181.
- 471. Lloyd-Jones D, Adams RJ, Brown TM, Carnethon M, Dai S, De Simone G, Ferguson TB, Ford E, Furie K, Gillespie C. Heart disease and stroke statistics—2010 update. *Circulation*. 2010;121:e46–e215.
- 472. Roger VL, Go AS, Lloyd-Jones DM, Adams RJ, Berry JD, Brown TM, Carnethon MR, Dai S, de Simone G, Ford ES. Heart disease and stroke statistics—2011 update1. A report from the American Heart Association. *Circulation*. 2011;123:e18–e209.
- 473. Roger VL, Go AS, Lloyd-Jones DM, Benjamin EJ, Berry JD, Borden WB, Bravata DM, Dai S, Ford ES, Fox CS. Executive summary: Heart disease and stroke statistics—2012 update: A report from the American Heart Association. *Circulation*. 2012;125:188.
- 474. Rosamond W, Flegal K, Friday G, Furie K, Go A, Greenlund K, Haase N, et al. Heart disease and stroke statistics—2007 update. A report from the American Heart Association statistics committee and stroke statistics subcommittee. *Circulation*. 2007; 115:e69–e171.
- 475. Rosemond R, Flegal K, Furie K, Go A, Greenlund K, Haase N, Hailpern SM, et al. Heart disease and stroke statistics—2008 update. A report from the American Heart association Statistics Committee and Stroke Statistics Committee. *Circulation*. 2008;117:e25–e146.
- 476. Haskell WL, Lee I-M, Pate RR, Powell KE, Blair SN, Franklin BA, Macera CA, Heath GW, Thompson PD, Bauman A. Physical activity and public health. Updated recommendation for adults from the American College of Sports Medicine and the American Heart Association. *Circulation*. 2007;116:1081–1093.

This page intentionally left blank

3 Diabetes

Raymond E. Bourey, Meenakshi K. Kaw, Sumona Ghosh Lester, Simona S. Ghanem, and Sonia M. Najjar

CONTENTS

3.1	Introd	luction	102				
	3.1.1	Physiology of Glucose Metabolism	103				
	3.1.2	Effects of Exercise on Glucose Metabolism	104				
3.2	Etiology of Diabetes						
	3.2.1	Type 1 Diabetes	105				
	3.2.2	Type 2 Diabetes	106				
		3.2.2.1 Genetic Factors	106				
		3.2.2.2 Environmental Factors	107				
	3.2.3	Prediabetes and Metabolic Syndrome	107				
		3.2.3.1 Metabolic Syndrome	107				
		3.2.3.2 Visceral Obesity and Insulin Resistance	108				
		3.2.3.3 Altered Insulin Metabolism and Insulin Resistance	108				
3.3	Effects of Diet and Exercise on Insulin Resistance						
	3.3.1	Dietary Composition	109				
	3.3.2	Caloric Restriction	110				
	3.3.3	Effects of Exercise on Insulin Resistance 11					
	3.3.4	Exercise and Glucose Metabolism11					
	3.3.5	Exercise and Lipid Metabolism	111				
3.4	Treatment of Diabetes with Exercise and Diet						
	3.4.1	4.1 Requirement for a Clinical Team in Diabetes Self-Management					
		and Lifestyle	112				
	3.4.2	Goals for Glycemic Control in Adults with Diabetes Mellitus					
	3.4.3	General Principles for Use of Exercise in Therapy for					
		Diabetes Mellitus	113				
	3.4.4	Can One Exercise Too Much or the Wrong Way? 113					
	3.4.5	Avoidance of Adverse Events Related to Hypoglycemia 114					
	3.4.6	Recommendations for Exercise in the Context of					
		Complications of Diabetes	114				
	3.4.7	General Recommendations for Timing, Quality,					
		and Quantity of Diet	115				
3.5	Diet and Exercise in Prevention and Treatment of Diabetes Mellitus						
	3.5.1	Diet and Exercise in Treatment of T1DM	116				
	3.5.2	Diet and Exercise in Prevention of T2DM 117					
	3.5.3	Diet and Exercise in Treatment of T2DM	119				

	3.5.4	Diet and Exercise and Prevention of Gestational Diabetes Mellitus	120
	3.5.5	Diet and Exercise in Treatment of Gestational Diabetes Mellitus	121
3.6	Conclu	usions	121
Ackı	nowledg	gment	123
Refe	rences.	-	123

3.1 INTRODUCTION

Diabetes mellitus is a disease diagnosed by fasting hyperglycemia that results from defects in insulin secretion, insulin action, or both.¹ Impact of diabetes mellitus on the economy of the United States is overwhelming as an estimated 25.8 million people or 8.3% of the population have diabetes mellitus.² This incidence of the disease has doubled since 1980. As diabetes is the leading cause of new renal failure, blindness, and nontraumatic lower-limb amputations, as well as a major contributor to heart disease, stroke, and complications of pregnancy, it is not surprising that costs were estimated in 2007 at \$174 billion.³ The adverse effects of the twin epidemics of obesity and diabetes are expected to rise, unless we develop effective therapeutic and preventive strategies. This chapter reviews the development and pathophysiology of diabetes and discusses therapeutic and preventive value of diet and exercise against this disease.

Types of diabetes are defined by etiology. Simplistically, hyperglycemia is caused by a mismatch in glucose appearance, from gut or liver gluconeogenesis, and glucose disposal, due to insufficient insulin action or decrease in end-organ glucose metabolism. The current etiological classification of diabetes mellitus by the American Diabetes Association (ADA) specifies 51 types of diabetes mellitus in addition to "others."¹ For simplicity, diabetes can be divided into four main categories: "Type 1," caused by autoimmune mechanisms that result in pancreatic β-cell destruction; "Type 2," which has a metabolic origin that results in insulin resistance and defective insulin secretion; "Gestational," which also has a metabolic origin resulting in impaired insulin function and release; and "Other" forms with a variety of possible causes including genetic, infectious, neoplastic, pharmaceutical, hormonal, and trauma also leading to impaired insulin release and function or defective glucose metabolism. By far, the etiologies of diabetes most familiar to clinicians are autoimmune-mediated destruction of pancreatic β cells or type 1 diabetes (T1DM), which currently represents approximately 5%-10% of cases, and metabolically mediated resistance to insulin action and dysfunction of glucose regulation or type 2 diabetes mellitus (T2DM), which represents approximately 90%-95% of cases. Although gestational diabetes mellitus (GDM) is metabolically mediated and similar to T2DM, its usual transient nature and differences in definition and aggressiveness of treatment assign it a separate category.

The recent explosion in the prevalence of diabetes is associated with similar rise in obesity and associated T2DM. This rise coincides with increased availability of food and the readiness of industrialized nations to adopt a sedentary lifestyle. If one considers that we evolved to accommodate scarcity of food and relatively short life span, it is no surprise that we have a thrifty genotype⁴ and naturally carry hedonistic appetites for calorically dense food and a natural tendency to avoid physical activity that is not essential to survival and continuation of the species.

Hyperglycemia defines diabetes mellitus. Specific definitions based on fasting glucose (>7 mM; 126 mg/dL) or standardized 75-g oral glucose tolerance test (2-hour value >11.1 mM; 200 mg/dL) are well established. Diabetes can also be diagnosed in a patient with symptoms of hyperglycemia and random plasma glucose greater than 11.1 mM (200 mg/dL) or glycated hemoglobin (HbA_{1c}) >6.5%. A state of glucose intolerance or impaired fasting glucose associated with insulin resistance and visceral obesity often precedes T2DM. As this state of prediabetes is often reversible, it has now become a treatment target. Summary of definitions of prediabetes and diabetes including GDM are found in Table 3.1.

3.1.1 PHYSIOLOGY OF GLUCOSE METABOLISM

Diabetes is a complex disease that involves abnormal insulin metabolism and action. Proper regulation of insulin metabolism and action is essential for the maintenance of normal glucose levels and whole-body insulin response. In a healthy insulinsensitive individual, the rise in portal vein glucose concentrations following a meal stimulates release of insulin from pancreatic β cells that generally reside centrally in the islet. Insulin flows toward the periphery of the islet where it suppresses glucagon secretion of pancreatic α -cells.

TABLE 3.1						
Criteria for Diagnosis of Diabetes, GDM, and Prediabetes						
Туре	Criteria					
Diabetes mellitus	Fasting (>8 hours) plasma glucose >7.0 mM (126 mg/dL) Or					
	75 g oral glucose tolerance test after fast (>8 hours) with 2-hour plasma glucose >11.1 mM (200 mg/dL)					
	Or					

(200 mg/dL)

28 weeks gestation

Impaired fasting glucose

Prediabetes

Gestational diabetes mellitus

Or Impaired glucose tolerance 2-hour plasma glucose in 75 g oral glucose tolerance test of

7.8-11 mM (140-199 mg/dL)

Glycated hemoglobin (HbA_{1c}) >6.5%

Fasting plasma glucose >5.1 mM (92 mg/dL)1-hour plasma glucose >10 mM (180 mg/dL) 2-hour plasma glucose >8.5 mM (153 mg/dL)

Fasting plasma glucose 5.6-6.9 mM (100 to 125 mg/dL)

Symptomatic hyperglycemia with random plasma glucose >11.1 mM

Or

75 g oral glucose tolerance test after fast (>8 hours) between 24 and

Source: American Diabetes Association, Diabetes Care, 2012, S64-S71.

Insulin flows through the portal vein to suppress hepatic glucose production and increase glycogen synthesis. It also leads to clearance of approximately one-third of an oral glucose load following an overnight fast.⁵

In the peripheral circulation, insulin increases glucose uptake by skeletal muscle and white adipose tissue, thereby decreasing circulating glucose to sustain normal glucose levels. Additionally, insulin exerts an antilipolytic effect in white adipose tissue to increase fat storage in adipocytes and to decrease lipolysis and circulating plasma free fatty acid (FFA) levels.⁶ Decrease in circulating FFA further suppresses liver gluconeogenesis and enhances muscle glucose disposal.

Skeletal muscle comprises about 55% of body mass and, in healthy individuals, accounts for disposal of approximately two-thirds of an oral glucose meal following an overnight fast.⁵ Glucose uptake in myocytes and adipocytes is mediated through solute carrier family 2 (facilitated glucose transporter) member 4 (SLC2A4), formerly known as GLUT4 or insulin-regulated glucose transporter. In response to insulin or muscle contraction in vitro, these transporters are translocated to the plasma membrane for cellular uptake of circulating glucose.⁷ In addition to insulin, exercise is a main contributor to GLUT4 transport to the plasma membrane, and hence, the effect of insulin and exercise on glucose disposal is more than additive.^{8,9}

Insulin action begins with insulin binding to its receptor, a plasma membrane protein that undergoes phosphorylation to trigger the involvement of a cascade of signaling pathways¹⁰ that mediate the pleiotropic effects of insulin. While insulin action is mediated by these signaling events, it is regulated by the amount of circulating insulin, which is in turn determined by insulin secretion from β cells and clearance, an event that occurs mostly in liver and to a smaller extent in kidney.¹¹ Abnormal insulin metabolism and action can cause insulin resistance and diabetes.

In the postabsorptive or fasting state, about 70% of glucose disposal takes place in insulin-independent tissues with uptake by brain responsible for approximately 50% of total hepatic glucose production. As mentioned in the preceding discussion, skeletal muscle contraction will increase SLC2A4 (GLUT4) translocation and increase muscle uptake of glucose in an insulin-independent manner that is not impaired by an insulin resistance state.¹²

3.1.2 EFFECTS OF EXERCISE ON GLUCOSE METABOLISM

With exercise, a complex homeostatic interplay between the central nervous system, sympathetic nervous system, and endocrine systems maintains blood glucose in a narrow range. Although a detailed treatment is beyond the scope of this chapter, knowledge of some of the normal physiology is important for the proper use of exercise as a therapeutic modality.

With initiation of exercise, sympathetic nervous system activity suppresses insulin secretion and increases adipocyte production of FFA and liver gluconeogenesis and glycogenolysis. A small drop in blood sugar increases these effects. These mechanisms generally preclude hypoglycemia, even in prolonged exercise. With cessation of exercise, sympathetic nervous system activity and glucose counter-regulation persist for a few minutes, and transient rebound hyperglycemia can occur. Contracting muscles translocate SLC2A4 (GLUT4) to the surface to increase glucose transport and oxidative metabolism. Local factors are recruited to increase opening of capillary beds to enhance delivery of glucose and residual insulin to muscle. This can potentially explain the more than additive increase in glucose uptake when exercise is added to insulin.^{8,9}

With increased workload, glucose oxidation increases as FFA oxidation decreases. With prolonged duration, there is a shift from liver and muscle glycogenolysis to liver gluconeogenesis and increased oxidation of FFA.

After exercise, there is a prolonged period of increased glucose uptake to replenish glucose stores in liver and skeletal muscle. This residual increase in glucose uptake is especially marked in the first 1 or 2 hours and then disappears over the next 20–72 hours depending on food consumption and the level of upregulation of glycogen synthase.^{13–15} This period of relatively insulin-independent glucose uptake by liver and muscle represents an opportunity for energy substrates derived from food to go to liver and muscle, but not fat.

Sensitivity to insulin action disappears on average by 30 hours following exercise, and most of the effect of exercise is due to the last bout of exercise. Exercise training, that is regularly repeated bouts of exercise, not only induces physiological changes to allow more intense exercise (and therefore more intense effects on insulin sensitivity), but also creates structural changes of increased muscle mass, decreased fat mass, increased capillary area, and decreased capillary basal membrane thickness to modestly improve insulin sensitivity even after the effects of acute exercise have waned.^{16,17}

3.2 ETIOLOGY OF DIABETES

3.2.1 Type 1 Diabetes

T1DM is characterized by deficiency of insulin due to autoimmune damage to pancreatic β cells. Historically, T1DM was referred to as juvenile diabetes, but with recognition that adults are also susceptible to autoimmune-mediated diabetes mellitus and that T2DM now occurs frequently in children, this descriptive term has been abandoned. Because uncontrolled T2DM leads to severe insulin deficiency that requires insulin therapy, the term "insulin-dependent diabetes mellitus" should also no longer be used to refer to T1DM.

Autoimmune markers for risk of T1DM include islet cells antibodies, insulin autoantibodies, and antibodies against neuroendocrine antigens, glutamic acid decarboxylase (GAD₆₅), and tyrosine phosphatase–like protein islet antigen-2 (IA-2 and IA-2 β). One or more of these autoantibodies are present in more than 95% of newly diagnosed T1DM patients, and they present variably during the prodrome of islet autoimmunity long before the clinical onset. Two or more of these autoantibodies have a high predictive value of T1DM.¹⁸

The severity of T1DM depends on the rate and extent of the loss of β -cell mass, which is higher in children and adolescents than adults. With older age, the autoimmune response is generally less complete, and in some cases, residual function of β cells enables patients to manage without exogenous insulin for several years and in particular if good diet and exercise are followed.¹⁹ When very little or no secretion of

insulin occurs, as determined by low or undetectable levels of C-peptide in plasma, the patient develops an absolute need for exogenous insulin to survive.

T1DM patients can be at risk to develop other autoimmune diseases as part of an autoimmune polyglandular syndrome. Hypothyroidism is the most common autoimmune disorder associated with diabetes and affects 17%–30% of patients with T1DM.²⁰ Other autoimmune diseases that can occur in autoimmune polyglandular syndromes include Addison's disease, autoimmune hepatitis, celiac disease, Graves' disease, Hashimoto's thyroiditis, myasthenia gravis, pernicious anemia, and vitiligo. With risk for the development of autoimmune polyglandular syndrome, the ADA recommends antithyroid antibody testing and consideration of antitissue transglutaminase testing at diagnosis.²¹

3.2.2 Type 2 Diabetes

T2DM was previously referred to as noninsulin-dependent diabetes mellitus and adult-onset or maturity-onset diabetes mellitus. As insulin deficiency often develops due to metabolic destruction or disabling of β cells, and as the condition now occurs in preadolescent children, these obsolete terms should be avoided.

T2DM is characterized by resistance to insulin action and abnormal insulin secretion that is represented phenotypically by fasting hyperinsulinemia, impaired glucose tolerance, and reduced glucose disposal rate during glucose clamps.²² Initially, insulin secretion increases to compensate for peripheral insulin resistance. As the burden on β cells increases, especially with uncontrolled diet and physical inactivity,²³ insulin secretion progressively declines resulting in progressive insulin deficiency, hyperglycemia, and eventually T2DM.

Hence, the etiology of T2DM relates to factors that lead to compromised response to insulin. This etiology involves a complex interplay among genetic, metabolic, and environmental factors.²⁴ Although environmental factors such as persistent overnutrition might be necessary to phenotypic expression of T2DM, these are insufficient without preexisting genetic determinant.

Patients with T2DM suffer disproportionately from atherosclerosis and related macrovascular complications (stroke and heart disease),²⁵ and they show a twofold to fourfold rise in mortality related to cardiovascular disease (CVD). Five-year risk of myocardial infarction in patients with T2DM without known cardiac disease is similar to that of nondiabetic patients with prior myocardial infarction.²⁶ In this context, a joint statement from the ADA and the American Heart Association recommends structured programs that emphasize lifestyle changes, which include reducing fat and total energy intake with increased regular physical activity to produce long-term weight loss.²⁷

3.2.2.1 Genetic Factors

Although there are clear family clusters of T2DM, it does not separate in Mendelian fashion and is considered a polygenic disease. In some populations, gene polymorphisms in the insulin receptor and one of its most characterized substrates, the insulin receptor substrate 1 (IRS-1) are two to three times more common in patients with T2DM,²⁸ although studies of other populations have not confirmed this association.²⁹

More recently, genome-wide association studies and linkage analysis have been used.³⁰ Meta-analysis of genome-wide association studies of hyperglycemia or insulin resistance in nondiabetic patients identified 19 genetic loci, most of which were variably associated with insulin processing, secretion, and action.³¹ The diversity of mechanisms potentially affected by these loci only serves to emphasize the polygenic nature of glucose control.

3.2.2.2 Environmental Factors

Environmental factors commonly associated with T2DM include overeating, highfat diet, physical inactivity, and stress.^{32–37} Prolonged high-fat intake reduces tissue responsiveness to insulin and glucose-induced insulin secretion. This results in insufficient compensatory insulin secretion and subsequently T2DM.²³ Physical inactivity has also been associated with decreased activity in muscle oxidative enzymes³⁸ and insulin resistance. Importantly, as little as 7 days of exercise reverses decreased oxidative enzymes and restores insulin sensitivity.^{39,40}

Stress-related neuromodulation of metabolism can also lead to pathologically positive energy balance (hyperphagia and reduced energy expenditure).⁴¹ We know, for example, high cortisol or abnormal sex hormones levels⁴² can cause a gradual onset of hyperglycemia and hyperinsulinemia.⁴³ Sleep deprivation, another common stressor, causes insulin resistance and abnormal intravenous glucose tolerance.⁴⁴ In this context, there is no surprise to find that prevalence of sleep apnea among patients with T2DM is 86%⁴⁵ and that severity of sleep apnea independently predicts poor glucose control.⁴⁶

3.2.3 PREDIABETES AND METABOLIC SYNDROME

Prediabetes (IFG and IGT) and T2DM are components of the metabolic syndrome, previously known as insulin resistance syndrome, central obesity syndrome, syndrome X, and others.⁴⁷ Metabolic syndrome is characterized clinically by increased abdominal obesity and bodily fat content, dyslipidemia (with low HDL cholesterol) and hypertension; together representing a cluster of cardiovascular risk factors.⁴⁸

3.2.3.1 Metabolic Syndrome

Metabolic syndrome is defined by risk factors for atherosclerosis, stroke, and myocardial infarct. Several outcome data show that insulin resistance precedes atherosclerosis⁴⁹ and trumps hyperglycemia as a risk factor.⁵⁰ However, the mechanistic link has not been well elucidated, in part because insulin resistance is not a monolithic entity, but rather genetically, biochemically, and pathophysiologically heterogeneous, brought about by different mechanisms in different individuals. Regardless of its initial mechanism, insulin resistance is associated with a host of additional abnormalities, key among which are those involving lipid metabolism and visceral obesity, an inflammatory state that plays a critical role in hastening T2DM, atherosclerosis, and CVD. With diet and exercise ameliorating insulin response, treatment of insulin resistance with these modalities promises to deliver effective preventive measures against the progression of metabolic intertwined diseases. Visceral obesity is also associated with activation of inflammatory pathways in fat and associated macrophages, liver, and vascular cells, among others.^{51,52}

Insulin resistance, a key factor in the etiology of metabolic diseases, is commonly heralded by hyperinsulinemia, which reflects a need for higher than normal levels of insulin to elicit a normal response.⁵³ Insulin resistance is associated with a number of related metabolic disorders including visceral obesity, glucose intolerance, T2DM, fatty liver disease, dyslipidemia, and hypertension.^{42,43,54–56} In spite of remarkable advances in defining novel molecules involved in insulin signaling, the molecular events that result in the development of insulin resistance in humans are not fully identified.⁵⁷ However, insulin resistance is commonly attributed to defects in insulin signaling,⁵⁸ in particular at the level of IRS and IRS-associated PI3 kinase.^{59–61} Moreover, there is growing evidence that defects in insulin secretion and clearance also contribute significantly to the pathogenesis of insulin resistance.^{47,62}

3.2.3.2 Visceral Obesity and Insulin Resistance

The majority of type 2 diabetic patients are obese⁶³ and have elevated plasma FFA levels⁶⁴ that represent an independent risk factor for glucose intolerance and progression to T2DM.^{65,66} In obesity, visceral fat becomes less sensitive to the antilipolytic effect of insulin,⁶⁷ thus supplying higher levels of FFA to liver, skeletal muscle, and vasculature.⁶⁸ Excessive fat intake exaggerates the release of fatty acids from the adipose tissue and their redistribution to other tissues, leading to systemic insulin resistance.^{69–73} Increased liver fatty acid transport causes insulin resistance by promoting gluconeogenesis,^{74,75} fatty acid β-oxidation, and lipogenesis,^{76,77} in addition to inhibiting insulin-mediated suppression of glycogenolysis⁷⁸ and impairing insulin clearance.⁷⁹ Increased fatty acid transport into skeletal muscle competes with glucose oxidation and causes insulin resistance.⁸⁰

Continuous supply of fatty acids to these extra-adipocytic tissues eventually leads to incomplete oxidation, fat accumulation, and lipotoxicity.⁸¹ Increased ectopic intracellular content of lipid and of their metabolites contributes to insulin resistance⁶³ by activating signaling pathways, which impair insulin signaling directly (by impairing PI3 kinase activity⁸²) or indirectly, through engaging inflammatory signals such as activating the c-Jun N-terminal kinase (JNK) pathway and increasing secretion of TNF- α ,⁸³ which deactivates IRS proteins.^{84–86}

There is growing evidence that visceral obesity and insulin resistance are chronic subacute inflammatory states that implicate elevation in fat-derived cytokines (adipokines),^{87–90} such as TNF- $\alpha^{91,92}$ in humans^{93,94} and rodents.^{95–100} In addition to increased release of adipokines, visceral obesity is also associated with increased release of resistin¹⁰¹ and decreased adiponectin,¹⁰² both of which play a significant role in promoting insulin resistance.

3.2.3.3 Altered Insulin Metabolism and Insulin Resistance

Although hyperinsulinemia, a cardinal feature of the metabolic syndrome, reflects an increase in insulin secretion to compensate for peripheral insulin resistance,¹⁰³ considerable data support the alternative view that hyperinsulinemia can result from impaired insulin extraction,¹⁰⁴ which mostly occurs in liver and to a lower extent in kidney.¹¹ The cause-effect relationship between hyperinsulinemia and insulin resistance is complex.

Peripheral insulin resistance causes secondary compensatory hyperinsulinemia, resulting from increased insulin secretion from β cells. But chronic hyperinsulinemia, caused by impaired hepatic insulin clearance, can worsen insulin resistance by downregulating insulin receptors in peripheral tissues and causing de novo lipogenesis in liver. This, in turn, results in increased lipid distribution to the white adipose tissue for storage and consequently visceral adiposity develops.

Considerable evidence in humans supports the view that impaired hepatic insulin extraction causes chronic hyperinsulinemia in obesity.¹⁰⁴ Our studies on the role of the carcinoembryonic-related cell adhesion molecule 1 (CEACAM1) in insulin clearance provide more convincing evidence that hyperinsulinemia causes insulin resistance. CEACAM1 promotes receptor-mediated insulin endocytosis and degradation, a process that underlies insulin clearance.¹⁰⁵ Consistent with liver being the main site of endogenous insulin clearance, L-SACC1 mice with liver-specific inactivation of CEACAM1 and mice with global *Ceacam1* null mutation ($Cc1^{-/-}$) develop chronically elevated levels of insulin resulting from impaired insulin clearance. The chronic hyperinsulinemic state in these mice causes whole-body insulin resistance and increases lipogenesis in liver, followed by lipid redistribution to the white adipose tissue and subsequently visceral obesity.¹⁰⁶⁻¹⁰⁹

The similarity in the phenotype of L-SACC1 and $Cc1^{-/-}$ mice emphasizes the role of hepatic insulin clearance in regulating insulin response in extra-hepatic tissues and assigns a major role for CEACAM1 in this process. In partial support of this notion, CEACAM1 levels are markedly reduced in rats selectively bred for low aerobic running capacity (low-capacity runners [LCR]) that phenocopy the metabolic syndrome,¹¹⁰ including liver steatosis,^{111,112} and in obese humans who have fatty liver disease in the absence or presence of T2DM.¹¹³

3.3 EFFECTS OF DIET AND EXERCISE ON INSULIN RESISTANCE

In the absence of exercise, dietary composition and restriction play a significant role in ameliorating altered fat and insulin metabolism, critical parameters in the pathogenesis of insulin resistance, fatty liver disease, T2DM, and their cardiovascular comorbidities.

3.3.1 DIETARY COMPOSITION

The notion that dietary changes affect predominantly hepatic metabolism is bolstered by the fact that the liver plays a major role in regulating dietary fat metabolism. In addition to being a major site for glucose conversion into fatty acids, more than 50% of dietary fat can be transported to liver through chylomicrons. Excessive fat intake and fatty acid transport to liver cause insulin resistance by promoting fatty acid β -oxidation and lipogenesis^{76,77} and gluconeogenesis,^{74,75} in addition to inhibiting insulin-mediated suppression of glycogenolysis.⁷⁸ It can also impair insulin clearance.⁷⁹ Thus, it is not at all surprising that reduced fat intake can decrease liver fat mass and reverse the metabolic abnormalities associated with hepatic steatosis. A low-fat/low-saturated fat/low glycemic diet for 4 weeks markedly reduces liver fat mass in older subjects.¹¹⁴ Moreover, the type of dietary fat has a significant effect. A monounsaturated fat diet for 8 weeks causes significant loss of hepatic steatosis in patients with T2DM when compared to patients who fed an isoenergetic carbohydrate/fiber-enriched diet, independently of an aerobic training program.¹¹⁵

3.3.2 CALORIC RESTRICTION

Caloric restriction reverses most of metabolic abnormalities, including hepatic steatosis and insulin metabolism, more strongly than exercise alone, in rats bred for LCR.¹¹² Similarly, a calorically restricted diet with low glycemic index for 16 weeks markedly decreases serum insulin concentrations and waist circumference in insulin-resistant obese adolescents.¹¹⁶ The lowering effect of caloric restriction in man on hepatic steatosis is supported by a 6-month study, which demonstrated that caloric restriction reduces liver triglyceride and hepatic steatosis, improves hepatic histology, and reverses insulin resistance with a 10% loss of body weight.¹¹⁷ While these studies did not investigate the effect of caloric restriction on insulin metabolism (clearance), they serve to emphasize the beneficial therapeutic role of caloric restriction on lipid metabolism in liver.

3.3.3 EFFECTS OF EXERCISE ON INSULIN RESISTANCE

A number of biological mechanisms may contribute to an increase in insulin sensitivity and improvement in glucose homeostasis in response to exercise training. Exercise can cause both structural and biochemical changes in skeletal muscle. Structural changes include elevation in fiber size, fiber transformation, and an increase in capillary density with a subsequent increase in blood flow.^{23,55} Biochemical changes include, but are not limited to, increased insulin signaling and changes in the activity of enzymes associated with enhanced glucose and lipid metabolism.

3.3.4 Exercise and Glucose Metabolism

During exercise, glucose and fat serve as fuel for the working skeletal muscle. Hence, several adaptive changes occur in an exercised muscle in several steps of glucose and fat metabolism. Glucose metabolism is initiated by its cellular uptake through SLC2A4 (GLUT-4), followed by its conversion into glucose-6-phosphate, which may then undergo glycolysis to generate ATP for energy or be incorporated into glycogen by stimulating glycogen synthase enzymatic activity.¹¹⁸

Insulin resistance and T2DM are frequently associated with alterations in this process, including reduced glucose transport. Exercise training increases capacity for glucose transport into the cell,¹¹⁹ through increase in glucose transporter content,¹²⁰ and enhanced insulin signaling through the PI3 kinase/Akt pathway that leads to translocation of SLC2A4 (GLUT-4) to the membrane for glucose uptake.^{119,121} Moreover, insulin resistance is associated with reduction in the muscle oxidative and respiratory capacity due to a decrease in the number of mitochondria and key oxidative enzymes such as citrate synthase.¹²² Endurance training increases mitochondrial number¹²³ and protein expression and activity of citrate synthase in skeletal muscle,^{124,125} leading to improved glucose tolerance.

Insulin resistance is also associated with reduced glycogen synthesis,^{126–128} resulting from impaired insulin-induced stimulation of glycogen synthase activity.¹²⁸ During exercise, glycogen stores serve as an energy source for contractile activity. Hence, its synthesis is upregulated to replete the storage and meet demand imposed by the exercised muscle. This is mediated by elevating both insulin-stimulated glycolysis and glycogen synthesis during exercise.^{118,129}

3.3.5 Exercise and Lipid Metabolism

Ectopic fat accumulation in the liver and skeletal muscle is strongly associated with the development of insulin resistance in T2DM.^{130–132} Fat is synthesized primarily in the liver and the adipose tissue, a tissue that is better positioned to store fat in response to insulin. In an insulin resistance state and hyperinsulinemia, hepatic fat production and redistribution to the adipose tissue is increased, and fatty acids are mobilized from the adipose tissue. Plasma fatty acids are then cleared in skeletal muscle by β -oxidation¹³³ and in liver predominantly by undergoing esterification.^{68,134}

During exercise, β -oxidation increases in skeletal muscle, the major site of energy expenditure.¹³⁵ This is regulated by stimulating 5' adenosine monophosphate (AMP)-activated protein kinase (AMPK), through a high AMP:ATP ratio, thereby, reducing production of malonyl CoA, an inhibitor of carnitine palmitoyltransferase-1 that mediates translocation of long chain fatty acids into the mitochondria for oxidation.^{136,137} Thus, exercise increases β -oxidation, particularly in skeletal muscle.

This mechanism contributes to enhanced insulin sensitivity in patients with metabolic syndrome, as shown by several studies, in particular in response to a program that incorporates aerobic interval training (90% of highest measured heart rate), three times a week for 16 weeks.¹³⁸ This program has been shown to be superior to a continuous moderate exercise program (70% of highest measured heart rate) in terms of reversing endothelial dysfunction, insulin resistance, and fat metabolism in patients with metabolic syndrome.¹³⁸

Similarly, subjecting LCR rats to high-intensity interval endurance training for 8 weeks effectively reduces markers of cardiovascular risk, such as endothelial dysfunction and hypertension, fat mass, and glucose intolerance.¹³⁹ While this program reduces fat accumulation in skeletal muscle and adipose tissue, it does not substantially reduce hepatic triglyceride and insulin metabolism. This could be due to the need for the liver to produce and contribute to providing triglycerides (energy) to the exercised skeletal muscle to meet increasing demand. It can also suggest that endurance training does not suffice to reverse all metabolic abnormalities, especially those pertaining to the liver, which appears to be more responsive to dietary changes and caloric changes.¹³⁹

3.4 TREATMENT OF DIABETES WITH EXERCISE AND DIET

Although caloric restriction and exercise training can independently restore the altered metabolic state associated with insulin resistance, their combined therapy appears to provide a more effective treatment to fully restore metabolic regulation by insulin. In the following part of this chapter, we will translate this knowledge to patient care.

Before the past few decades, caloric restriction and physical activity were the norm for most of the world. Food was scarce and significant physical activity was necessary to seek food, shelter, and continuation of the species. In this context, diabetes mellitus was relatively rare and limited primarily to T1DM, for which there was no effective treatment. Although dietary caloric restriction therapy was widely adopted at the beginning of the past century, the natural progression of this fatal wasting disease was not changed. The advent of insulin therapy has allowed survival, and diet and exercise are now commonly recommended adjunct therapies to maintain glucose control, insulin sensitivity, and functional capacity.

The explosion in incidence of diabetes follows our ability in the industrial era to adopt a sedentary lifestyle in the face of excess energy intake. Although the relationship between caloric intake and obesity in T2DM was recognized early, only recently has it been recognized that physically inactive lifestyle initiates maladaptation to promote T2DM and other chronic disease.¹⁴⁰

3.4.1 REQUIREMENT FOR A CLINICAL TEAM IN DIABETES SELF-MANAGEMENT AND LIFESTYLE

Management of diabetes requires a capable patient and a multidisciplinary team of diabetes professionals to provide medical oversight, pharmacological therapy, hypoglycemia training, nutritional therapy, exercise, and education in self-management and psychosocial care. Diabetes self-management and education, when provided by a program that meets national standards, is reimbursed as part of the Medicare program, Indian Health Service, and most health insurance plans. Medical nutrition therapy is usually reimbursed separately.

3.4.2 GOALS FOR GLYCEMIC CONTROL IN ADULTS WITH DIABETES MELLITUS

Most physicians and patients are familiar with the use of goals based on HbA_{1c} . There has been some misconception that goals for HbA_{1c} should be rigid and universally applied, and therefore confusion about the nature of the difference between the American Association of Clinical Endocrinology recommendation of 6.5%¹⁴¹ and the ADA recommendation of 7.0% as goals for maximum HbA_{1c} . In fact both sets of guidelines acknowledge that more or less stringent glycemic goals may be appropriate for individual patients, and the ADA specifically recommends a customized range from 6.5% to 8.0% as reasonable goals depending on specific circumstances such as safety, age, proclivity to complications, and residual insulin secretory capacity.²¹

For nonpregnant adults, the ADA recommends maximal preprandial capillary plasma glucose of 3.9–7.2 mM (70–130 mg/dL) and postprandial capillary plasma glucose of 10 mM (180 mg/dL).²¹

For pregnant women with preexisting T1DM or T2DM, a recent consensus statement¹⁴² recommends premeal, bedtime, and nocturnal glucose range of 3.3-5.4 mM (60–99 mg/dL), peak postprandial glucose range of 5.4-7.1 mM (100–129 mg/dL), and HbA_{1c}<6.0%.

For pregnant women with GDM, recommendations from the 5th International Workshop-Conference on Gestational Diabetes¹⁴³ include maximal preprandial capillary

blood glucose of 5.3 mM (95 mg/dL) and either 1-hour postprandial maximal glucose of 7.8 mM (140 mg/dL) or 2-hour postmeal maximal glucose of 6.7 mM (120 mg/dL).

3.4.3 GENERAL PRINCIPLES FOR USE OF EXERCISE IN THERAPY FOR DIABETES MELLITUS

As the effects of exercise on muscle glucose disposal and insulin sensitivity wane significantly after 2 hours and become insignificant by 30 hours,^{13–15} we generally recommend daily exercise with the bulk of exercise preceding the major meal by 1 hour or 2 hours. For patients with adequate time to exercise, we recommend exercise or physical activity to precede both morning and evening meals. This differs somewhat from current guidelines for standard of care that is based on recommendations of 150 min/week of moderate intensity or 75 min/week of vigorous physical activity in three to four sessions per week.^{21,144}

Guidelines^{21,144} recommend use of both aerobic and resistance exercise. This recommendation has general support and seems reasonable. Aerobic exercise (continuous exercise below threshold for lactate production) represents an efficient means toward caloric expenditure. In addition, aerobic exercise of more than 20 minutes duration and moderate intensity predominantly uses FFA stores and can be especially effective to reduce body fat, especially if paired with caloric restriction. On the other hand, whole-body glucose disposal also correlates with muscle mass. Although muscle sensitivity to insulin of weight lifters is no better than that of sedentary subjects, whole-body insulin sensitivity and glucose disposal were nearly identical to that of marathon runners with muscle highly sensitive to insulin.¹⁴⁵

In this context, we generally recommend a combination of exercise: balance, stretching, and postural therapy for safety; aerobic activity to efficiently oxidize calories and to improve muscle sensitivity to insulin; and resistance training to maintain or increase muscle mass and to maintain joint stability. We suggest lighter exercise in the morning and more intense exercise in the evening. This not only reduces risk for hypoglycemia in those taking insulin following overnight fast but also acknowledges circadian rhythm of performance, with power output being generally better in the late afternoon or evening.¹⁴⁶ To avoid overuse injury, we usually recommend that the exercise program should vary daily and that specific exercises be repeated only every 3 days.

3.4.4 CAN ONE EXERCISE TOO MUCH OR THE WRONG WAY?

This question was addressed 20 years ago by two studies. In the first, Kirwan et al.¹⁴⁷ found eccentric exercise caused marked muscle soreness and significantly elevated creatine kinase levels 48 hours after exercise (273 ± 73 , 92 ± 27 , 87 ± 25 IU/L for the eccentric, concentric, and control conditions, respectively) 48 hours after exercise. Sensitivity to insulin action was reduced by about 40% after eccentric exercise ($3.47 \pm 0.51 \text{ mg/kg/min}$) compared to concentric exercise ($5.55 \pm 0.94 \text{ mg/kg/min}$) or control conditions ($5.48 \pm 1.0 \text{ mg/kg/min}$).

The morning after the 1991 Helsinki marathon, we found whole-body insulin sensitivity (insulin-stimulated glucose disposal) as measured by 19 concurrent

euglycemic clamps was "paradoxically" decreased by 12% in spite of glycogen depletion.¹⁴⁸ Insulin resistance following eccentric or unusually long or traumatic exercise is believed to be due to a combination of metabolic factors including elevated fatty acid oxidation and muscle damage.

3.4.5 Avoidance of Adverse Events Related to Hypoglycemia

New drugs for treatment of diabetes mellitus that minimize risk of hypoglycemia allow aggressive nutritional therapy and exercise. We believe these medications should be preferred therapy. Once insulin or sulfonylureas are used, a patient is obligated to aggressive glucose testing and constant modulation of caloric intake and expenditure to avoid hypoglycemia. In spite of this significant limitation, sulfonylureas continue to be used quite aggressively in the United States, presumably due more to reasons of therapeutic inertia or short-term economics than medical science.

Once hypoglycemic agents are used, hypoglycemia becomes a limiting and dangerous factor in treatment.^{149,150} In patients who use hypoglycemic agents, we have found the ability to analyze trends of interstitial fluid glucose in real time with continuous glucose monitor technology to be an invaluable tool for patients prescribed diet and exercise. In this context, we have tested a feed-forward neural network model for prediction of glucose in patients requiring insulin to predict future blood glucose (75-minute horizon) on the basis of continuous glucose monitor values, insulin dose, nutritional intake, and physical activity.¹⁵¹ As technology surrounding continuous glucose monitoring improves, we believe that cost analysis will demonstrate improved therapeutic effectiveness of diet and exercise through use of this technology with subsequent reduction in hypoglycemia, hyperglycemia, risk of cardiac events, and small vessel disease. In this context, we take issue with guidelines for clinical care or insurance reimbursement that emphasize early use of sulfonylureas or severely limit access of patients who require insulin to technology of continuous glucose monitoring.

3.4.6 RECOMMENDATIONS FOR EXERCISE IN THE CONTEXT OF COMPLICATIONS OF DIABETES

Complications of diabetes can in turn complicate recommendations for diet and exercise. Cardiovascular complications are especially prevalent in T2DM, which as noted in Section 3.2.2, is considered a risk factor equivalent to coronary artery disease. In spite of this, current recommendations state that routine screening for coronary artery disease is not recommended, as it does not improve outcomes as long as risk factors are treated.^{21,144,152} These guidelines recommend cardiac testing only in the face of cardiac symptoms or abnormal resting electrocardiogram.^{21,144,152} Providers are reminded, of course, to use clinical judgment in this area. From a practical perspective, if there is a question as to a patient's ability to safely exercise, then exercise testing should be undertaken.

Autonomic neuropathy due to microvascular disease can increase the risk of exercise-related adverse events through postural hypotension, impaired thermoregulation, and decreased cardiac responsiveness to exercise. Autonomic neuropathy can also cause gastroparesis. As gastric emptying is the limiting factor for glucose absorption, gastroparesis can predispose to hypoglycemia, especially if gastric emptying times are variable and unpredictable. Gastric emptying can be further slowed by commonly used medications such as antidepressants as well as hot or cold food and fat content. Pupillary dysfunction in addition to retinopathy can impair vision and increased risk of accident.

Peripheral neuropathy can increase risk for inadvertent skin and skeletal injury. Skeletal injury often leads to Charcot's joint distraction. Prevention remains critical, and the patient should use proper footwear, examination of feet, and coaching to avoid injury.

Although physical activity can acutely increase urinary protein excretion, there is no evidence that exercise increases the rate of progression of diabetic kidney disease. At this time, there is no need for specific exercise restrictions for patients with diabetic kidney disease.

3.4.7 GENERAL RECOMMENDATIONS FOR TIMING, QUALITY, AND QUANTITY OF DIET

As with exercise (see Section 3.4.3), diet should be prescribed by time, quantity, and quality to match caloric expenditure and nutritional requirements for micronutrients. Again, we note the rapidly waning residual effects of exercise on insulin sensitivity and glucose disposal to muscle and liver,^{13–15} and we therefore recommend the bulk of daily calories be eaten within 2 hours of finishing exercise. In this context, we avoid discussion with patients of the importance of breakfast, use of multiple small meals, or avoidance of evening meals, which are recommendations generally supported only by epidemiological evidence and short studies in sedentary populations. The ADA Standards of Medical Care²¹ also omits these topics from general recommendations for dietary care of diabetes mellitus, which are summarized in Table 3.2.

TABLE 3.2

General Recommendations for Nutritional Therapy from American Diabetes Association

- · Weight loss is recommended for overweight or obese individuals
- Weight loss diets may be effective for up to 2 years
- Weight loss diets can restrict either carbohydrate, fat, or specific fats (Mediterranean diet)
- · Physical activity is important to weight loss
- · Physical activity is especially important to maintenance of weight loss
- · Carbohydrates should be monitored
- Saturated fat should be less than 7% of calories
- Intake of trans fat should be minimized
- · Alcohol should be limited to one drink/day for women and two drinks/day for men
- Routine supplementation with vitamins E and C and carotene is not advised.
- Diet should meet patient preferences, metabolic goals, and current recommended daily allowance for all micronutrients

Source: American Diabetes Association, Diabetes Care, 2012, S11-S63.

Caloric content of the diet is usually aimed at weight maintenance in athletic patients with desired muscle and fat mass and weight loss or gain in others. Malnourishment is rare in the United States, and there has been no demonstration that incorporation of recommended daily allowance or dietary reference intake for micronutrients has clinical benefit. Nonetheless, misinformation and the nutraceutical industry have made education of patients on micronutrient intake an oft complicated and time-consuming affair.

Although relatively simple physiological principles might be involved, variables of patient preference, physical activity, basal metabolic rate, mood, educational background, and motivation can complicate dietary prescription. Many patients come armed with misperceptions and misinformation promoted by various media on all aspects of dietary therapy. We have found that use of a full-time professional in diabetes care with background in medical nutrition therapy and exercise seems mandatory. As noted, Medicare and the Indian Health Services reimburse medical nutritional therapy for patients with diabetes through specific legislation and rules.

3.5 DIET AND EXERCISE IN PREVENTION AND TREATMENT OF DIABETES MELLITUS

The incidence of T1DM is increasing worldwide.¹⁵³ Whether this increase in incidence is related to a higher incidence of preclinical autoimmunity or faster progression to diabetes after development of autoimmunity is not clear. Rise in T1DM has stimulated speculation that insulin resistance and obesity might modulate autoimmunity. A recent meta-analysis supports the association between childhood obesity and subsequent T1DM.¹⁵⁴ At least one paper has proposed a mechanism by which exercise might modulate preclinical autoimmunity.¹⁵⁵ In this context, exercise and diet might prove important in reduction of risk for development of T1DM and other autoimmune diseases.

3.5.1 DIET AND EXERCISE IN TREATMENT OF T1DM

Hypoglycemia is the limiting factor in management of T1DM.¹⁴⁹ To date, nutritional interventions for patients with T1DM are essentially limited to recommendations to match carbohydrate content of meals and snacks with insulin therapy and exercise. As any serious athlete with diabetes will attest, this is no mean task, especially when one considers the 30–40-minute biological half-life of insulin action and the immediately synergistic effect to lower blood glucose of insulin and exercise.^{8,9} Successful therapy and lifestyle intervention in this group require an intelligent and willing patient supported by a full team of professional diabetes educators and clinicians.

To avoid hypoglycemia with exercise, long-acting insulin should generally be avoided in patients with T1DM who exercise regularly and maintain insulin sensitivity. Pumps have been especially useful to patients with T1DM. They can be removed before exercise, and waning levels of short-acting insulin present little risk of hypoglycemia and are usually sufficient to prevent ketosis, even in prolonged exercise. The ability to analyze trends of interstitial fluid glucose in real time with continuous glucose monitor technology is an invaluable tool for patients with T1DM who exercise.¹⁵⁶ Although there is a delay of 10–15 minutes between changes in blood and interstitial fluid glucose content, software-enabled analysis of trends and appropriate alarms provide additional safety. Given longer life, obesity-related disease has not surprisingly become a problem for patients with T1DM.¹⁵⁷ Weight loss requires a careful match among caloric intake, caloric expenditure, and insulin dose. This usually requires intense glucose monitoring. More detailed recommendations pertinent to weight loss can be found in Section 3.5.2.

Patients with T1DM can find it difficult to strike a balance between insufficient insulin and hyperglycemia on one hand and too much insulin and hypoglycemia on the other hand. We have always been impressed with the minimal requirements, often in the range of less than 12 units/day, required by professional or competitive amateur athletes with T1DM. Tests of blood glucose should be undertaken before, during, and after exercise. Although hyperglycemia per se does not require postponement of exercise, hyperglycemia in the context of ketosis can worsen with exercise.¹⁵⁸ Adequate treatment and retesting should be undertaken.

It should be remembered that athletes with T1DM do not automatically enjoy improved insulin sensitivity. We previously reported that in 11 athletes with T1DM, insulin sensitivity measured across the forearm as arteriovenous difference, by euglycemic clamp, was not improved when compared with 12 more sedentary patients with T1DM.¹⁵⁹ These athletes suffered higher HbA_{1c} (8.4 \pm 0.4% vs. 7.2 \pm 0.2%), took 20% less insulin, and had higher fatty acid oxidation and lower glycogen synthase activity. We therein concluded that better education of athletes with T1DM was needed to maintain good control and insulin sensitivity.

ADA guidelines recommend consumption of additional carbohydrate if pre-exercise glucose levels are less than 5.5 mM (100 mg/dL). The patient should undertake subsequent testing during exercise with frequency dependent on how rapidly blood glucose has dropped, current insulin or sulfonylurea dose, and intensity of exercise.

3.5.2 DIET AND EXERCISE IN PREVENTION OF T2DM

Impaired fasting glucose and impaired glucose tolerance (prediabetes) are generally associated with insulin resistance and metabolic syndrome. These conditions are generally considered to be a consequence of modern lifestyle of low physical activity and consumption of excessive calories. Therefore, it follows that lifestyle changes that involve increase in physical activity and decrease in caloric intake might reverse this condition or at least prevent progression to T2DM. To date, there have been four prospective randomized studies to demonstrate that use of exercise and diet in patients with impaired fasting glucose and/or impaired glucose tolerance will prevent onset of T2DM.

The first of these trials, the Da Qing trial, was published with a 6-year follow-up in 1997.¹⁶⁰ This was the only study of this kind to directly compare effects of diet, exercise, and exercise plus diet. When compared to the control group, all three interventions were equally effective, with reduction in the incidence of diabetes at 6 years by more than 50%. Data can be found in Table 3.3. The Finnish Diabetes Prevention Study (FDPS),¹⁶¹ Indian Diabetes Prevention Program (IDPP),¹⁶² and United States Diabetes Prevention Program (USDPP)¹⁶³ all followed and demonstrated similar reductions in the development of T2DM. The USDPP was memorable in that it was closed early for ethical reasons after only 2.6 years of follow-up. Results from these studies are summarized in Table 3.3.

			Incidence of Development of Diabetes (%)			Reduction in Diabetes (%)			
	п	Duration (Years)	Control	Diet	Exercise	Diet and Exercise	Diet	Exercise	Diet and Exercise
Da Qing DPS ¹⁶⁰	577	6.0	68	44 ^a	41ª	46 ^a	55ª	59ª	51ª
Finnish DPS ¹⁶¹	400	3.2	23			11ª			58ª
U.S. DPPT ¹⁶³	3234	2.8	29ь			14 ^b			58ª
India IDPP ¹⁶²	531	2.5	55 ^b			39ь			28ª

TABLE 3.3 Prospective Intervention Trials to Prevent Diabetes in Populations with Impaired Glucose Tolerance

^a Different from control group.

^b Estimated cumulative incidence to 3 years.

In the USDPP,¹⁶³ lifestyle intervention was found twice as effective in preventing diabetes as therapy with metformin, a biguanide suppressor of gluconeogenesis. Interestingly, the IDPP¹⁶² demonstrated no additional effect of metformin on top of exercise and diet. The reduction in incidence of T2DM with lifestyle modification was virtually identical to lifestyle modification plus metformin therapy. Superiority of nondrug approaches to drug-based approaches in prevention of T2DM was confirmed in recent meta-analysis of 10 randomized, controlled trials (n = 23,152) of more than 100 patients and more than 1 year in duration.¹⁶⁴ A separate meta-analysis of 16 studies of patients with metabolic syndrome with interventions of at least 24 weeks also suggests, using mixed-treatment comparison methods, that lifestyle interventions were more probably more effective than pharmacological interventions (87%).¹⁶⁵ In that study, the odds of metabolic syndrome reversal by lifestyle were 3.8 (2.5–5.9) and by pharmacological intervention 1.6 (1.0–2.4).

Long-term follow-up of 20 years in the Da Qing trial,¹⁶⁶ 7 years in the FDPS¹⁶⁷ and 5.7 years in the USDPP Outcomes Study,¹⁶⁸ demonstrated continued reduction in cumulative incidence of diabetes of 43%, 43%, and 34% respectively. The lifestyle treatment group in the USDPP Outcomes Study continued to demonstrate twice the reduction as the group treated with metformin.

Retrospective meta-analysis of these studies was undertaken to examine whether a specific intervention was more effective.¹⁶⁹ Studies variably used low-fat, highprotein, or Mediterranean diets. All seemed equally effective. They calculated that maintenance of weight loss requires regular exercise with additional expenditure of 2000 kcal/week. They estimated it would be necessary to treat 6.4 (5.0–8.4) patients with prediabetes to prevent or delay just one case of diabetes through lifestyle intervention. In sum, these studies are encouraging as they demonstrate that exercise and diet are inexpensive and effective therapies to prevent progression of prediabetic states to T2DM and that their effectiveness can be maintained beyond a period of intensive education and intervention. On the other hand, we note that 80% of the patients in the Da Qing trial developed T2DM by the 20-year follow-up.¹⁶⁶ Although we acknowledge that those 20 years were a period of rapid industrialization in this area, and that the group treated with exercise and diet had lower incidence of diabetes than the control group, long-term intervention to prevent T2DM has room for new ideas and improvement.

3.5.3 DIET AND EXERCISE IN TREATMENT OF T2DM

The Look AHEAD (Action for Health in Diabetes) trial, designed to prevent complications in obese individuals with T2DM, has been the only one large prospective randomized controlled trial of lifestyle modification in patients with T2DM.¹⁷⁰ This trial, which was planned for up to 11 years of follow-up, recently underwent a midcourse correction when a much lower-than-expected rate of cardiovascular complications was found in the first 2 years, to expand both the definition of primary end point and follow-up of participants.¹⁷¹

Four-year follow-up of approximately 4800 participants was recently reported,¹⁷² and patients with intensive lifestyle intervention (ILI) had greater improvements in treadmill fitness, systolic and diastolic blood pressure, and HbA_{1c}, than a group assigned to more routine diabetes self-management and education (DSE) program. Weight loss was maintained in the ILI group at a somewhat disappointing 4.7% after initial weight loss of 8.6% in year 1. Meanwhile, success of both interventions is suggested by maintenance of weight loss (-1.1% in the DSE group) and occurrence of only one cardiovascular event among over 5,000 participants in the course of 4 years.

In addition, 1-year follow-up of erectile function using the International Index of Erectile Function demonstrated a trend toward improvements in the ILI group than the DSE group (p = .06).¹⁷³ Also, at 1 year, and in the face of significant weight loss, women in the ILI group had improvement in incidence, although not resolution, of urinary incontinence.¹⁷⁴ Each kilogram of weight loss was associated with a 3% reduction in odds ratio for development of urinary incontinence.

ILI and increased physical activity were not detrimental to joint disease, and in fact the pain improved in this group at 1 year.^{175,176} We await follow-up data to determine whether this is simply due to weight loss or other factors.

T2DM is a risk factor for osteoporosis. Bone density declines with weight loss, and not surprisingly the ILI group demonstrated decrease bone density at 1 year.¹⁷⁶ Again, significance is not certain as bone biopsy and quality of bone are not known. We await long-term follow-up.

The Sleep AHEAD component of the Look AHEAD trial found more than 86% of obese patients with T2DM have obstructive sleep apnea (n = 306, mean apnea–hypopnea index = 20.5 ± 16.8 respiratory events/hour).⁴⁵ This treatable condition causes insulin resistance, hypertension, weight gain, and glucose intolerance. The condition is often curable with weight loss and exercise. One-year follow-up of this subgroup demonstrated weight loss of 10.8 kg in the ILI group and 0.6 kg weight loss

in the DSE group.¹⁷⁷ The ILI had 20% decrease in AHI while the DSE group actually suffered 20% increase in AHI. A 13.6% of ILI participants had remission of obstructive sleep apnea. Greater weight loss was associated with greater improvements in sleep apnea. Although this study found a trend to improved HbA_{1c} (p = 0.08), quantitation of the effects of diet and exercise mediated through improved obstructive sleep apnea on cardiovascular risk factors and metabolic control have not been determined by a randomized, controlled trial. Sleep apnea is an independent risk factor for worsening glycemic control in patients with T2DM. It may also be an independent risk factor for complication of diabetic nephropathy¹⁷⁸ and coronary artery disease.^{179–181} Future clinical research in diabetes needs to address and control for sleep apnea.

3.5.4 DIET AND EXERCISE AND PREVENTION OF GESTATIONAL DIABETES MELLITUS

As morbidity associated with pregnancy in overweight and obese patients becomes more apparent, a more aggressive approach has been taken to prevent GDM. This is especially important as GDM is not usually diagnosed until at least the 24th week of gestation. This late date makes intervention short and perhaps less effective once a patient develops diabetes mellitus.

Unfortunately, results to date of randomized controlled trials of exercise and diet have not been as encouraging as hoped for on the basis of epidemiological studies. In part, this may reflect understandable reluctance of investigators and institutional review boards to jump into studies of effects of stressful exercise on sedentary and pregnant patients. The time to initiate exercise and diet is before pregnancy, but few young women have the luxury of this level of planning.

In this context, a very recent Norwegian study of 702 women¹⁸² declared "A 12-week exercise program performed during the second trimester does not prevent gestational diabetes in healthy pregnant women." Like many studies, this one followed recommendations for three bouts of 45 minutes of low-impact exercise each week. Another study published this year of 855 women echoed these results with demonstration that a 12-week standard exercise program provided no evidence for prevention of gestational diabetes or improvement in insulin resistance.¹⁸³ Although this study did aim for a moderate-to-high intensity activity 3 days/week, it suffered a dropout rate of 45%. A recent meta-analysis of three randomized controlled trials with moderate risk of bias found no significant difference in incidence of GDM between women receiving additional exercise intervention and routine care.¹⁸⁴ Another group undertook meta-analysis of 13 randomized and 6 nonrandomized trials of lifestyle intervention in obese pregnant women. Although lifestyle intervention reduced maternal pregnancy weight (-2.2 kg) and produced a minimal trend toward reduction in prevalence of gestational diabetes (odds ratio 0.8 with 95% confidence interval 0.6-1.1), there was no clear difference in outcomes of cesarean delivery, large for gestational age, birth weight, or macrosomia. All studies were statistically of low-to-medium quality.

In summary, data available to date suggest that exercise and diet as currently prescribed are insufficient to prevent GDM in overweight and obese women and that future studies of lifestyle intervention to prevent GDM should include daily exercise, diet designed to match the exercise and requirements of the pregnancy, and intervention at an earlier gestational age.

3.5.5 DIET AND EXERCISE IN TREATMENT OF GESTATIONAL DIABETES MELLITUS

Obesity and GDM are associated with adverse outcomes in pregnancy including macrosomia, cesarean section, hypertension, preeclampsia, and eclampsia. Slightly different recommendations have evolved for goals for glucose control in pregnant women with GDM and pregnant women with preexisting diabetes mellitus. These recommendations are in Section 3.4.2.

Medical nutritional therapy has been the cornerstone of treatment for GDM.¹⁴³ Although the Institute of Medicine report of 1990 had recommendations for weight gain with pregnancy in patients who are obese and for patients who are underweight, there are no published guidelines for weight gain in women with GDM. The situation is further complicated as diagnosis of GDM is usually delayed until weeks 24–28, at which time many patients have reached or exceeded their weight gain target.

In workshop recommendations,¹⁴³ consensus recommendations for physical activity are addressed in a mere four sentences. Physical activity of 30 min/day is recommended without recommendation for intensity or caloric expenditure aside from comment that 10 minutes of brisk walking or arm exercises while seated in a chair for 10 minutes after each meal accomplishes the stated goal. Meta-analysis of four trials based on these recommendations as carried out in 114 pregnant women found no effect of exercise on perinatal outcomes, pregnancy complications, or maternal morbidity. The authors concluded that there is insufficient evidence to recommend or advise against enrollment of pregnant women with GDM in exercise programs. It should be noted that by necessity of diagnosis, all studies started in the third trimester and duration was only 6 weeks. The authors of the study also commented that they were unable to find any studies with exercise intervention in pregnant women with T1DM or T2DM. Again, sparsity of data for use of exercise and diet in GDM or pregnant women with other types of diabetes mellitus precludes specific recommendations.

3.6 CONCLUSIONS

Consistent consumption of surplus calories and proclivity to an available sedentary lifestyle have caused an epidemic of metabolically mediated insulin resistance, T2DM, and associated morbidity. This metabolic milieu might also be associated with the rise in autoimmune-mediated T1DM. More certain is the observation that more patients with T1DM have become obese and insulin resistant as they live longer in the current environment.

As consumption of surplus calories and low levels of physical activity have created the metabolic problems of obesity, insulin resistance, and T2DM, it seems reasonable that therapy with diet and exercise can reverse these problems. A large body of work both in the laboratory and in the clinic confirms that diet and exercise have a role in the prevention of T2DM and modification of risk factors in T2DM. Unfortunately, the same cannot be said for the role of exercise and prevention or treatment of GDM or diabetes during pregnancy.

Specific guidelines from the American College of Sports Medicine¹⁴⁴ and the ADA on use of medical nutrition^{21,185} and exercise^{21,144} in the treatment of diabetes are a good starting point. Ultimately, these guidelines address nutritional therapy and exercise therapy as separate modalities. To optimize therapy, we need integrated

guidelines that allow a more physiologically rational integration of diet and exercise. If, for example, insulin-sensitizing effects of exercise persist at relatively high levels for only an hour or two after exercise, then meals should be prescribed to follow exercise in the physiologically natural rhythm of exertion, followed by eating and sleep.

We also suffer from lack of information on how best to deliver our prescription for diet and exercise to the person who will carry it out—the patient. The future might include an exercise station equipped with monitors for continuous glucose measurement, caloric consumption, substrate utilization, blood pressure, and sympathetic nervous system activity, as well as software to use these data to recommend appropriate and immediate adjustment in exercise, diet, and medication. For now, however, most patients receive prescription of diet, exercise, and medications from separate individuals with little integration of instructions that sometimes conflict and often overwhelm.

In this context, Table 3.4 provides a sample of 10 recommendations that summarize the science and clinical guidelines for an average patient with diabetes mellitus. Recommended calories are personalized on the basis of calculation or measurement of basal metabolic rate, estimate of caloric expenditure of planned daily exercise, with calories added or subtracted to accommodate needs for weight gain or loss. Recommendations can be expanded or contracted as needed.

In the end, the patient will have to integrate his/her prescription into a life already complicated by demands of survival—food, shelter, and continuation of the species. To deny ourselves food and to do physical work without reward are not natural after puberty. Health-care providers must arm themselves with enough information to convince the patient of the importance of diabetes and exercise or to convince third-party payers to reward the patient directly for healthy changes in lifestyle.

TABLE 3.4 Sample of Practical Recommendations for Diet and Exercise for Nonpregnant Adults with Diabetes Mellitus

- Aim for _____ calories daily to include 80 g of protein (320 calories)
- · Eat sufficient fiber and water to avoid constipation
- · Eat the bulk of calories within 2 hours of finishing exercise
- Try light exercise such as yoga in the morning and heavier exercise such as weight lifting in the evening
- Walking on flat ground can be hard on the knees. (Walking is not the best exercise if you are overweight.) Uphill treadmill walking and exercise cycles are easier on the knees
- · Avoid daily repetition of the same exercise to avoid injury
- · Respect your orthopedic or cardiac limitations
- Stop and call the physician for chest discomfort. Go to an emergency department for chest discomfort of greater than 5 minutes
- If needed, get help from your physical therapist or physical medicine physician to design a pain-free program and exercises to stabilize your joints
- If you are on insulin,
 - · Test glucose before exercise and every 30 minutes during exercise and as needed
 - Test 20 minutes after exercise
 - · Remember, the effects of insulin and glucose are more than additive
 - Use a snack if glucose is less than 100 mg/dL or dropping quickly

ACKNOWLEDGMENT

This work was supported by grants from the NIH R01 DK054254, R01 DK083850 and R01 HL112248 to Sonia M. Najjar.

REFERENCES

- 1. American Diabetes Association. Diagnosis and classification of diabetes mellitus. *Diabetes Care*. 2012;35 Suppl 1:S64–S71.
- 2. Centers for Disease Control and Prevention. *National Diabetes Fact Sheet: National Estimates and General Information on Diabetes and Prediabetes in the United States, 2011.* Atlanta, GA: U.S. Department of Health and Human Services, Centers for Disease Control and Prevention; 2011.
- 3. American Diabetes Association. Economic costs of diabetes in the U.S. in 2007. *Diabetes Care*. 2008;31:596–615.
- 4. Neel JV. Diabetes mellitus: A "thrifty" genotype rendered detrimental by "progress"? *Am J Hum Genet.* 1962;14:353–362.
- Kelley D, Mitrakou A, Marsh H, Schwenk F, Benn J, Sonnenberg G, Arcangeli M, et al. Skeletal muscle glycolysis, oxidation, and storage of an oral glucose load. *J Clin Invest*. 1988;81:1563–1571.
- 6. Groop LC, Bonadonna RC, Shank M, Petrides AS, DeFronzo RA. Role of free fatty acids and insulin in determining free fatty acid and lipid oxidation in man. *J Clin Invest.* 1991;87:83–89.
- Henriksen EJ, Bourey RE, Rodnick KJ, Koranyi L, Permutt MA, Holloszy JO. Glucose transporter protein content and glucose transport capacity in rat skeletal muscles. *Am J Physiol.* 1990;259:E593–E598.
- Bourey RE, Coggan AR, Kohrt WM, Kirwan JP, King DS, Holloszy JO. Effect of exercise on glucose disposal: Response to a maximal insulin stimulus. *J Appl Physiol*. 1990;69:1689–1694.
- 9. DeFronzo RA, Ferrannini E, Sato Y, Felig P, Wahren J. Synergistic interaction between exercise and insulin on peripheral glucose uptake. *J Clin Invest*. 1981;68:1468–1474.
- Saltiel AR, Kahn CR. Insulin signalling and the regulation of glucose and lipid metabolism. *Nature*. 2001;414:799–806.
- 11. Duckworth WC, Bennett RG, Hamel FG. Insulin degradation: Progress and potential. *Endocr Rev.* 1998;19:608–624.
- 12. Wang Y, Simar D, Fiatarone Singh MA. Adaptations to exercise training within skeletal muscle in adults with type 2 diabetes or impaired glucose tolerance: A systematic review. *Diabetes Metab Res Rev.* 2009;25:13–40.
- Cartee GD, Young DA, Sleeper MD, Zierath J, Wallberg-Henriksson H, Holloszy JO. Prolonged increase in insulin-stimulated glucose transport in muscle after exercise. *Am J Physiol*. 1989;256:E494–E499.
- 14. Garcia-Roves PM, Han DH, Song Z, Jones TE, Hucker KA, Holloszy JO. Prevention of glycogen supercompensation prolongs the increase in muscle GLUT4 after exercise. *Am J Physiol Endocrinol Metab.* 2003;285:E729–E736.
- King DS, Baldus PJ, Sharp RL, Kesl LD, Feltmeyer TL, Riddle MS. Time course for exercise-induced alterations in insulin action and glucose tolerance in middle-aged people. J Appl Physiol. 1995;78:17–22.
- Koivisto VA, Yki-Jarvinen H, DeFronzo RA. Physical training and insulin sensitivity. Diabetes Metab Rev. 1986;1:445–481.
- Williamson JR, Hoffmann PL, Kohrt WM, Spina RJ, Coggan AR, Holloszy O. Endurance exercise training decreases capillary basement membrane width in older nondiabetic and diabetic adults. *J Appl Physiol*. 1996;80:747–753.

- Haller MJ, Atkinson MA, Schatz D. Type 1 diabetes mellitus: Etiology, presentation, and management. *Pediatr Clin NAm*. 2005;52:1553–1578.
- Naik RG, Brooks-Worrell BM, Palmer JP. Latent autoimmune diabetes in adults. J Clin Endocrinol Metab. 2009;94:4635–4644.
- Roldan MB, Alonso M, Barrio R. Thyroid autoimmunity in children and adolescents with type 1 diabetes mellitus. *Diabetes Nutr Metab.* 1999;12:27–31.
- 21. American Diabetes Association. Standards of medical care in diabetes—2012. *Diabetes Care*. 2012;35 Suppl 1:S11–S63.
- 22. Saltiel AR. New perspectives into the molecular pathogenesis and treatment of type 2 diabetes. *Cell*. 2001;104:517–529.
- 23. LaMonte MJ, Blair SN, Church TS. Physical activity and diabetes prevention. *J Appl Physiol*. 2005;99:1205–1213.
- Kendall DM, Harmel AP. The metabolic syndrome, type 2 diabetes, and cardiovascular disease: Understanding the role of insulin resistance. *Am J Manag Care*. 2002;8:S635–S653; quiz S654–S637.
- 25. National Institute of Diabetes and Digestive and Kidney Diseases. *National Diabetes Statistics*, 2011: National Diabetes Information Clearinghouse. http://diabetes.niddk. nih.gov/dm/pubs/statistics/DM_Statistics_508.pdf. Accessed 15 December 2011.
- 26. Sethi SS, Akl EG, Farkouh ME. Diabetes mellitus and acute coronary syndrome: Lessons from randomized clinical trials. *Curr Diab Rep.* 2012;12:294–304.
- 27. Buse JB, Ginsberg HN, Bakris GL, Clark NG, Costa F, Eckel R, Fonseca V, et al. Primary prevention of cardiovascular diseases in people with diabetes mellitus: A scientific statement from the American Heart Association and the American Diabetes Association. *Diabetes Care*. 2007;30:162–172.
- Sesti G, Federici M, Hribal ML, Lauro D, Sbraccia P, Lauro R. Defects of the insulin receptor substrate (IRS) system in human metabolic disorders. *FASEB J*. 2001;15:2099–2111.
- Hager J, Zouali H, Velho G, Froguel P. Insulin receptor substrate (IRS-1) gene polymorphisms in French NIDDM families. *Lancet*. 1993;342:1430.
- Permutt MA, Hattersley AT. Searching for type 2 diabetes genes in the post-genome era. *Trends Endocrinol Metab.* 2000;11:383–393.
- Ingelsson E, Langenberg C, Hivert MF, Prokopenko I, Lyssenko V, Dupuis J, Magi R, et al. Detailed physiologic characterization reveals diverse mechanisms for novel genetic loci regulating glucose and insulin metabolism in humans. *Diabetes*. 2010;59:1266–1275.
- 32. Olefsky J, Farquhar JW, Reaven G. Relationship between fasting plasma insulin level and resistance to insulin-mediated glucose uptake in normal and diabetic subjects. *Diabetes*. 1973;22:507–513.
- Beard JC, Ward WK, Halter JB, Wallum BJ, Porte D, Jr. Relationship of islet function to insulin action in human obesity. J Clin Endocrinol Metab. 1987;65:59–64.
- Kolterman OG, Insel J, Saekow M, Olefsky JM. Mechanisms of insulin resistance in human obesity: Evidence for receptor and postreceptor defects. J Clin Invest. 1980;65:1272–1284.
- Lillioja S, Mott DM, Spraul M, Ferraro R, Foley JE, Ravussin E, Knowler WC, Bennett PH, Bogardus C. Insulin resistance and insulin secretory dysfunction as precursors of non-insulin-dependent diabetes mellitus. Prospective studies of Pima Indians. N Engl J Med. 1993;329:1988–1992.
- 36. Bjorntorp P. Visceral obesity: A "civilization syndrome". Obes Res. 1993;1:206-222.
- Peiris AN, Mueller RA, Smith GA, Struve MF, Kissebah AH. Splanchnic insulin metabolism in obesity. Influence of body fat distribution. *J Clin Invest*. 1986;78:1648–1657.
- Chi MM, Hintz CS, Coyle EF, Martin WH, 3rd, Ivy JL, Nemeth PM, Holloszy JO, Lowry OH. Effects of detraining on enzymes of energy metabolism in individual human muscle fibers. *Am J Physiol*. 1983;244:C276–C287.

- Spina RJ, Chi MM, Hopkins MG, Nemeth PM, Lowry OH, Holloszy JO. Mitochondrial enzymes increase in muscle in response to 7–10 days of cycle exercise. *J Appl Physiol*. 1996;80:2250–2254.
- 40. Short KR, Vittone JL, Bigelow ML, Proctor DN, Rizza RA, Coenen-Schimke JM, Nair KS. Impact of aerobic exercise training on age-related changes in insulin sensitivity and muscle oxidative capacity. *Diabetes*. 2003;52:1888–1896.
- 41. Schwartz MW, Porte D, Jr. Diabetes, obesity, and the brain. Science. 2005;307:375-379.
- 42. Kaku K, Rasmussen MF, Clauson P, Seino Y. Improved glycaemic control with minimal hypoglycaemia and no weight change with the once-daily human glucagonlike peptide-1 analogue liraglutide as add-on to sulphonylurea in Japanese patients with type 2 diabetes. *Diabetes Obes Metab.* 2010;12:341–347.
- 43. Tennyson GE. Understanding type 2 diabetes mellitus and associated cardiovascular disease: Linked by insulin resistance. *Am J Manag Care*. 2002;8:S450–S459.
- 44. Spiegel K, Tasali E, Penev P, Van Cauter E. Brief communication: Sleep curtailment in healthy young men is associated with decreased leptin levels, elevated ghrelin levels, and increased hunger and appetite. *Ann Intern Med.* 2004;141:846–850.
- 45. Foster GD, Sanders MH, Millman R, Zammit G, Borradaile KE, Newman AB, Wadden TA, et al. Obstructive sleep apnea among obese patients with type 2 diabetes. *Diabetes Care*. 2009;32:1017–1019.
- 46. Aronsohn RS, Whitmore H, Van Cauter E, Tasali E. Impact of untreated obstructive sleep apnea on glucose control in type 2 diabetes. *Am J Respir Crit Care Med*. 2010;181: 507–513.
- 47. Accili D. Lilly lecture 2003: The struggle for mastery in insulin action: From triumvirate to republic. *Diabetes*. 2004;53:1633–1642.
- Grundy SM, Brewer HB, Jr., Cleeman JI, Smith SC, Jr., Lenfant C. Definition of metabolic syndrome: Report of the National Heart, Lung, and Blood Institute/American Heart Association Conference on Scientific Issues Related to Definition. *Circulation*. 2004;109:433–438.
- 49. Despres JP, Lamarche B, Mauriege P, Cantin B, Dagenais GR, Moorjani S, Lupien PJ. Hyperinsulinemia as an independent risk factor for ischemic heart disease. *N Engl J Med.* 1996;334:952–957.
- 50. U.K. Prospective Diabetes Study Group. Intensive blood-glucose control with sulphonylureas or insulin compared with conventional treatment and risk of complications in patients with type 2 diabetes (UKPDS 33). *Lancet*. 1998;352:837–853.
- 51. Weisberg SP, McCann D, Desai M, Rosenbaum M, Leibel R, Ferrante AWJ. Obesity is associated with macrophage accumulation in adipose tissue. *J Clin Invest.* 2003; 112:1796–1808.
- Xu H, Barnes GT, Yang Q, Tan G, Yang D, Chou CJ, Sole J, et al. Chronic inflammation in fat plays a crucial role in the development of obesity-related insulin resistance. *J Clin Invest.* 2003;112:1821–1830.
- 53. Peters AL. The clinical implications of insulin resistance. *Am J Manag Care*. 2000;6: S668–S674; discussion S675–S681.
- 54. Lebovitz HE. Type 2 diabetes: An overview. Clin Chem. 1999;45:1339-1345.
- 55. Henriksen EJ. Invited review: Effects of acute exercise and exercise training on insulin resistance. *J Appl Physiol*. 2002;93:788–796.
- 56. Green RM. Nash—Hepatic metabolism and not simply the metabolic syndrome. *Hepatology*. 2003;38:14–17.
- Virkamaki A, Ueki K, Kahn CR. Protein-protein interaction in insulin signaling and the molecular mechanisms of insulin resistance. *J Clin Invest*. 1999;103:931–943.
- Cefalu WT. Insulin resistance: Cellular and clinical concepts. *Exp Biol Med (Maywood)*. 2001;226:13–26.

- 59. Paz K, Hemi R, LeRoith D, Karasik A, Elhanany E, Kanety H, Zick Y. A molecular basis for insulin resistance. Elevated serine/threonine phosphorylation of IRS-1 and IRS-2 inhibits their binding to the juxtamembrane region of the insulin receptor and impairs their ability to undergo insulin-induced tyrosine phosphorylation. J Biol Chem. 1997;272:29911–29918.
- Cusi K, Maezono K, Osman A, Pendergrass M, Patti ME, Pratipanawatr T, DeFronzo RA, Kahn CR, Mandarino LJ. Insulin resistance differentially affects the PI3-kinaseand MAP kinase-mediated signaling in human muscle. *J Clin Invest*. 2000;105:311–320.
- Qiao LY, Zhande R, Jetton TL, Zhou G, Sun XJ. In vivo phosphorylation of insulin receptor substrate 1 at serine 789 by a novel serine kinase in insulin-resistant rodents. *J Biol Chem.* 2002;277:26530–26539.
- 62. Kotronen A, Juurinen L, Tiikkainen M, Vehkavaara S, Yki-Jarvinen H. Increased liver fat, impaired insulin clearance, and hepatic and adipose tissue insulin resistance in type 2 diabetes. *Gastroenterology*. 2008;135:122–130.
- 63. Boden G. Fatty acid-induced inflammation and insulin resistance in skeletal muscle and liver. *Curr Diab Rep.* 2006;6:177–181.
- 64. Reaven GM. Pathophysiology of insulin resistance in human disease. *Physiol Rev.* 1995;75:473–486.
- Baldeweg SE, Golay A, Natali A, Balkau B, Del Prato S, Coppack SW. Insulin resistance, lipid and fatty acid concentrations in 867 healthy Europeans. European Group for the Study of Insulin Resistance (EGIR). *Eur J Clin Invest*. 2000;30:45–52.
- Charles MA, Eschwege E, Thibult N, Claude JR, Warnet JM, Rosselin GE, Girard J, Balkau B. The role of non-esterified fatty acids in the deterioration of glucose tolerance in Caucasian subjects: Results of the Paris Prospective Study. *Diabetologia*. 1997; 40:1101–1106.
- 67. Ostman J, Arner P, Engfeldt P, Kager L. Regional differences in the control of lipolysis in human adipose tissue. *Metabolism.* 1979;28:1198–1205.
- 68. Lewis GF, Carpentier A, Adeli K, Giacca A. Disordered fat storage and mobilization in the pathogenesis of insulin resistance and type 2 diabetes. *Endocr Rev.* 2002;23:201–229.
- 69. Rebrin K, Steil GM, Getty L, Bergman RN. Free fatty acid as a link in the regulation of hepatic glucose output by peripheral insulin. *Diabetes*. 1995;44:1038–1045.
- Bergman RN, Ader M. Free fatty acids and pathogenesis of type 2 diabetes mellitus. *Trends Endocrinol Metab.* 2000;11:351–356.
- Bays H, Mandarino L, DeFronzo RA. Role of the adipocyte, free fatty acids, and ectopic fat in pathogenesis of type 2 diabetes mellitus: Peroxisomal proliferator-activated receptor agonists provide a rational therapeutic approach. *J Clin Endocrinol Metab.* 2004;89:463–478. Review.
- 72. Kabir M, Catalano KJ, Ananthnarayan S, Kim SP, Van Citters GW, Dea MK, Bergman RN. Molecular evidence supporting the portal theory: A causative link between visceral adiposity and hepatic insulin resistance. *Am J Physiol Endocrinol Metab.* 2005;288:E454–E461.
- 73. Guilherme A, Virbasius JV, Puri V, Czech MP. Adipocyte dysfunctions linking obesity to insulin resistance and type 2 diabetes. *Nat Rev Mol Cell Biol*. 2008;9:367–377.
- Williamson JR, Kreisberg RA, Felts PW. Mechanism for the stimulation of gluconeogenesis by fatty acids in perfused rat liver. *Proc Natl Acad Sci USA*. 1966;56:247–254.
- Massillon D, Barzilai N, Hawkins M, Prus-Wertheimer D, Rossetti L. Induction of hepatic glucose-6-phosphatase gene expression by lipid infusion. *Diabetes*. 1997;46:153–157.
- Oakes ND, Cooney GJ, Camilleri S, Chisholm DJ, Kraegen EW. Mechanisms of liver and muscle insulin resistance induced by chronic high-fat feeding. *Diabetes*. 1997;46:1768–1774.
- Sidossis LS, Mittendorfer B, Walser E, Chinkes D, Wolfe RR. Hyperglycemia-induced inhibition of splanchnic fatty acid oxidation increases hepatic triacylglycerol secretion. *Am J Physiol.* 1998;275:E798–E805.

- Boden G, Cheung P, Stein TP, Kresge K, Mozzoli M. FFA cause hepatic insulin resistance by inhibiting insulin suppression of glycogenolysis. *Am J Physiol Endocrinol Metab.* 2002;283:E12–E19.
- Mittelman SD, Van Citters GW, Kim SP, Davis DA, Dea MK, Hamilton-Wessler M, Bergman RN. Longitudinal compensation for fat-induced insulin resistance includes reduced insulin clearance and enhanced beta-cell response. *Diabetes*. 2000;49:2116–2125.
- Randle PJ, Hales CN, Garland PB, Newsholm EA. The glucose-fatty acid cycle: Its role in insulin sensitivity and the metabolic disturbances of diabetes mellitus. *Lancet*. 1963;1:785–789.
- Freedland ES. Role of a critical visceral adipose tissue threshold (CVATT) in metabolic syndrome: Implications for controlling dietary carbohydrates: A review. *Nutr Metab* (*Lond*). 2004;1:12.
- 82. Kim YB, Kotani K, Ciaraldi TP, Henry RR, Kahn BB. Insulin-stimulated protein kinase C lambda/zeta activity is reduced in skeletal muscle of humans with obesity and type 2 diabetes: Reversal with weight reduction. *Diabetes*. 2003;52:1935–1942.
- Nguyen MT, Satoh H, Favelyukis S, Babendure JL, Imamura T, Sbodio JI, Zalevsky J, Dahiyat BI, Chi NW, Olefsky JM. JNK and TNFα mediate free fatty acid-induced insulin resistance in 3T3-L1 adipocytes. *J Biol Chem.* 2005;280:35361–35371.
- Hotamisligil GS, Peraldi P, Budavari A, Ellis R, White MF, Spiegelman BM. IRS-1mediated inhibition of insulin receptor tyrosine kinase activity in TNFα- and obesityinduced insulin resistance. *Science*. 1996;271:665–668.
- Sykiotis GP, Papavassiliou AG. Serine phosphorylation of insulin receptor substrate-1: A novel target for the reversal of insulin resistance. *Mol Endocrinol*. 2001;15:1864–1869.
- Le Marchand-Brustel Y, Gual P, Gremeaux T, Gonzalez T, Barres R, Tanti JF. Fatty acid-induced insulin resistance: Role of insulin receptor substrate 1 serine phosphorylation in the retroregulation of insulin signalling. *Biochem Soc Trans.* 2003; 31:1152–1156.
- 87. Kershaw EE, Flier JS. Adipose tissue as an endocrine organ. *J Clin Endocrinol Metab.* 2004;89:2548–2556. Review.
- Sjoholm A, Nystrom T. Endothelial inflammation in insulin resistance. *Lancet*. 2005;365:610–612. Review.
- 89. Lazar MA. How obesity causes diabetes: Not a tall tale. Science. 2005;307:373-375.
- Tataranni PA, Ortega E. A burning question: Does an adipokine-induced activation of the immune system mediate the effect of overnutrition on type 2 diabetes? *Diabetes*. 2005;54:917–927.
- 91. Frayn KN, Karpe F, Fielding BA, Macdonald IA, Coppack SW. Integrative physiology of human adipose tissue. *Int J Obes Relat Metab Disord*. 2003;27:875–888. Review.
- 92. Coppack SW. Pro-inflammatory cytokines and adipose tissue. *Proc Nutr Soc.* 2001;60:349–356. Review.
- 93. Chen J, Wildman RP, Hamm LL, Muntner P, Reynolds K, Whelton PK, He J. Association between inflammation and insulin resistance in U.S. nondiabetic adults: Results from the Third National Health and Nutrition Examination Survey. *Diabetes Care*. 2004;27:2960–2965.
- Calle EE, Kaaks R. Overweight, obesity and cancer: Epidemiological evidence and proposed mechanisms. *Nat Rev Cancer*. 2004;4:579–591.
- Uysal KT, Wiesbrock SM, Marino MW, Hotamisligil GS. Protection from obesityinduced insulin resistance in mice lacking TNFα function. *Nature*. 1997;389:610–614.
- Yuan M, Konstantopoulos N, Lee J, Hansen L, Li ZW, Karin M, Shoelson SE. Reversal of obesity- and diet-induced insulin resistance with salicylates or targeted disruption of Ikkbeta. *Science*. 2001;293:1673–1677.
- 97. Perreault M, Marette A. Targeted disruption of inducible nitric oxide synthase protects against obesity-linked insulin resistance in muscle. *Nat Med.* 2001;7:1138–1143.

- Hirosumi J, Tuncman G, Chang L, Gorgun CZ, Uysal KT, Maeda K, Karin M, Hotamisligil GS. A central role for JNK in obesity and insulin resistance. *Nature*. 2002;420:333–336.
- Brozovic S, Nagaishi T, Yoshida M, Betz S, Salas A, Chen D, Kaser A, et al. CD1d function is regulated by microsomal triglyceride transfer protein. *Nat Med.* 2004;10:535–539.
- 100. Cai D, Yuan M, Frantz DF, Melendez PA, Hansen L, Lee J, Shoelson SE. Local and systemic insulin resistance resulting from hepatic activation of IKK-beta and NF-kappa beta. *Nat Med.* 2005;11:183–190.
- 101. Steppan CM, Bailey ST, Bhat S, Brown EJ, Banerjee RR, Wright CM, Patel HR, Ahima RS, Lazar MA. The hormone resistin links obesity to diabetes. *Nature*. 2001;409:307–312.
- 102. Berg AH, Combs TP, Du X, Brownlee M, Scherer PE. The adipocyte-secreted protein acrp30 enhances hepatic insulin action. *Nat Med.* 2001;7:947–953.
- 103. Bonner-Weir S. Life and death of the pancreatic β cells. *Trends Endocrinol Metab.* 2000;11:375–378.
- 104. Bergman RN. Non-esterified fatty acids and the liver: Why is insulin secreted into the portal vein? *Diabetologia*. 2000;43:946–952.
- 105. Formisano P, Najjar SM, Gross CN, Philippe N, Oriente F, Kern-Buell CL, Accili D, Gorden P. Receptor-mediated internalization of insulin. Potential role of pp120/HA4, a substrate of the insulin receptor kinase. *J Biol Chem.* 1995;270:24073–24077.
- 106. Poy MN, Yang Y, Rezaei K, Fernstrom MA, Lee AD, Kido Y, Erickson SK, Najjar SM. CEACAM1 regulates insulin clearance in liver. *Nat Genet*. 2002;30:270–276.
- 107. Park SY, Cho YR, Kim HJ, Hong EG, Higashimori T, Lee SJ, Goldberg IJ, Shulman GI, Najjar SM, Kim JK. Mechanism of glucose intolerance in mice with dominant negative mutation of CEACAM1. Am J Physiol Endocrinol Metab. 2006;291: E517–E524.
- 108. DeAngelis AM, Heinrich G, Dai T, Bowman TA, Patel PR, Lee SJ, Hong EG, et al. Carcinoembryonic antigen-related cell adhesion molecule 1: A link between insulin and lipid metabolism. *Diabetes*. 2008;57:2296–2303.
- 109. Najjar SM, Yang Y, Fernstrom MA, Lee SJ, Deangelis AM, Rjaily GA, Al-Share QY, et al. Insulin acutely decreases hepatic fatty acid synthase activity. *Cell Metab*. 2005;2:43–53.
- Wisloff U, Najjar SM, Ellingsen O, Haram PM, Swoap S, Al-Share Q, Fernstrom M, et al. Cardiovascular risk factors emerge after artificial selection for low aerobic capacity. *Science*. 2005;307:418–420.
- 111. Thyfault JP, Rector RS, Uptergrove GM, Borengasser SJ, Morris EM, Wei Y, Laye MJ, et al. Rats selectively bred for low aerobic capacity have reduced hepatic mitochondrial oxidative capacity and susceptibility to hepatic steatosis and injury. J Physiol. 2009;587:1805–1816.
- 112. Bowman TA, Ramakrishnan SK, Kaw M, Lee SJ, Patel PR, Golla VK, Bourey RE, et al. Caloric restriction reverses hepatic insulin resistance and steatosis in rats with low aerobic capacity. *Endocrinology*. 2010;151:5157–5164.
- 113. Lee W. The CEACAM1 expression is decreased in the liver of severely obese patients with or without diabetes. *Diagn Pathol*. 2011;6:40.
- 114. Utzschneider KM, Bayer-Carter JL, Arbuckle MD, Tidwell JM, Richards TL, Craft S. Beneficial effect of a weight-stable, low-fat/low-saturated fat/low-glycaemic index diet to reduce liver fat in older subjects. *Br J Nutr.* 2013;109:1096–1104.
- 115. Bozzetto L, Prinster A, Annuzzi G, Costagliola L, Mangione A, Vitelli A, Mazzarella R, et al. Liver fat is reduced by an isoenergetic MUFA diet in a controlled randomized study in type 2 diabetic patients. *Diabetes Care*. 2012;35:1429–1435.
- 116. Armeno ML, Krochik AG, Mazza CS. Evaluation of two dietary treatments in obese hyperinsulinemic adolescents. *J Pediatr Endocrinol Metab.* 2011;24:715–722.

- 117. Larson-Meyer DE, Newcomer BR, Heilbronn LK, Volaufova J, Smith SR, Alfonso AJ, Lefevre M, et al. Effect of 6-month calorie restriction and exercise on serum and liver lipids and markers of liver function. *Obesity (Silver Spring)*. 2008;16:1355–1362.
- Kim JY, Hickner RC, Cortright RL, Dohm GL, Houmard JA. Lipid oxidation is reduced in obese human skeletal muscle. Am J Physiol Endocrinol Metab. 2000;279:E1039–E1044.
- 119. Luciano E, Carneiro EM, Carvalho CR, Carvalheira JB, Peres SB, Reis MA, Saad MJ, Boschero AC, Velloso LA. Endurance training improves responsiveness to insulin and modulates insulin signal transduction through the phosphatidylinositol 3-kinase/Akt-1 pathway. *Eur J Endocrinol*. 2002;147:149–157.
- 120. Gulve EA, Spina RJ. Effect of 7–10 days of cycle ergometer exercise on skeletal muscle GLUT-4 protein content. *J Appl Physiol*. 1995;79:1562–1566.
- 121. Peres SB, de Moraes SM, Costa CE, Brito LC, Takada J, Andreotti S, Machado MA, Alonso-Vale MI, Borges-Silva CN, Lima FB. Endurance exercise training increases insulin responsiveness in isolated adipocytes through IRS/PI3-kinase/Akt pathway. J Appl Physiol. 2005;98:1037–1043.
- 122. Simoneau JA, Kelley DE. Altered glycolytic and oxidative capacities of skeletal muscle contribute to insulin resistance in NIDDM. *J Appl Physiol*. 1997;83:166–171.
- 123. Hawley JA, Holloszy JO. Exercise: It's the real thing! Nutr Rev. 2009;67:172-178.
- 124. Siu PM, Donley DA, Bryner RW, Alway SE. Citrate synthase expression and enzyme activity after endurance training in cardiac and skeletal muscles. *J Appl Physiol*. 2003;94:555–560.
- Coggan AR, Spina RJ, Kohrt WM, Holloszy JO. Effect of prolonged exercise on muscle citrate concentration before and after endurance training in men. *Am J Physiol*. 1993;264:E215–E220.
- 126. Kida Y, Esposito-Del Puente A, Bogardus C, Mott DM. Insulin resistance is associated with reduced fasting and insulin-stimulated glycogen synthase phosphatase activity in human skeletal muscle. *J Clin Invest*. 1990;85:476–481.
- 127. Thorburn AW, Gumbiner B, Bulacan F, Brechtel G, Henry RR. Multiple defects in muscle glycogen synthase activity contribute to reduced glycogen synthesis in non-insulin dependent diabetes mellitus. *J Clin Invest*. 1991;87:489–495.
- Wojtaszewski JF, Jorgensen SB, Hellsten Y, Hardie DG, Richter EA. Glycogendependent effects of 5-aminoimidazole-4-carboxamide (AICA)-riboside on AMPactivated protein kinase and glycogen synthase activities in rat skeletal muscle. *Diabetes*. 2002;51:284–292.
- 129. Kraegen EW, Storlien LH, Jenkins AB, James DE. Chronic exercise compensates for insulin resistance induced by a high-fat diet in rats. *Am J Physiol*. 1989;256:E242–E249.
- 130. Roden M, Price TB, Perseghin G, Petersen KF, Rothman DL, Cline GW, Shulman GI. Mechanism of free fatty acid-induced insulin resistance in humans. J Clin Invest. 1996;97:2859–2865.
- 131. Kahn SE, Hull RL, Utzschneider KM. Mechanisms linking obesity to insulin resistance and type 2 diabetes. *Nature*. 2006;444:840–846.
- 132. Utzschneider KM, Kahn SE. Review: The role of insulin resistance in nonalcoholic fatty liver disease. *J Clin Endocrinol Metab*. 2006;91:4753–4761.
- 133. Horowitz JF, Klein S. Lipid metabolism during endurance exercise. *Am J Clin Nutr.* 2000;72:558S–563S.
- 134. Chakravarthy MV, Pan Z, Zhu Y, Tordjman K, Schneider JG, Coleman T, Turk J, Semenkovich CF. "New" hepatic fat activates PPARalpha to maintain glucose, lipid, and cholesterol homeostasis. *Cell Metab.* 2005:309–322.
- 135. Kennedy JW, Hirshman MF, Gervino EV, Ocel JV, Forse RA, Hoenig SJ, Aronson D, Goodyear LJ, Horton ES. Acute exercise induces GLUT4 translocation in skeletal muscle of normal human subjects and subjects with type 2 diabetes. *Diabetes*. 1999;48:1192–1197.
- Ruderman NB, Saha AK, Vavvas D, Witters LA. Malonyl-CoA, fuel sensing, and insulin resistance. *Am J Physiol.* 1999;276:E1–E18.
- 137. Rasmussen BB, Winder WW. Effect of exercise intensity on skeletal muscle malonyl-CoA and acetyl-CoA carboxylase. *J Appl Physiol*. 1997;83:1104–1109.
- 138. Tjonna AE, Lee SJ, Rognmo O, Stolen TO, Bye A, Haram PM, Loennechen JP, et al. Aerobic interval training versus continuous moderate exercise as a treatment for the metabolic syndrome: A pilot study. *Circulation*. 2008;118:346–354.
- 139. Haram PM, Kemi OJ, Lee SJ, Bendheim MO, Al-Share QY, Waldum HL, Gilligan LJ, et al. Aerobic interval training vs. continuous moderate exercise in the metabolic syndrome of rats artificially selected for low aerobic capacity. *Cardiovasc Res.* 2009;81:723–732.
- 140. Olsen RH, Krogh-Madsen R, Thomsen C, Booth FW, Pedersen BK. Metabolic responses to reduced daily steps in healthy nonexercising men. *JAMA*. 2008;299:1261–1263.
- 141. Handelsman Y, Mechanick JI, Blonde L, Grunberger G, Bloomgarden ZT, Bray GA, Dagogo-Jack S, et al. American Association of Clinical Endocrinologists Medical Guidelines for Clinical Practice for developing a diabetes mellitus comprehensive care plan. *Endocr Pract.* 2011;17 Suppl 2:1–53.
- 142. Kitzmiller JL, Block JM, Brown FM, Catalano PM, Conway DL, Coustan DR, Gunderson EP, et al. Managing preexisting diabetes for pregnancy: Summary of evidence and consensus recommendations for care. *Diabetes Care*. 2008;31:1060–1079.
- 143. Metzger BE, Buchanan TA, Coustan DR, de Leiva A, Dunger DB, Hadden DR, Hod M, et al. Summary and recommendations of the Fifth International Workshop-Conference on Gestational Diabetes Mellitus. *Diabetes Care*. 2007;30 Suppl 2:S251–S260.
- 144. Colberg SR, Sigal RJ, Fernhall B, Regensteiner JG, Blissmer BJ, Rubin RR, Chasan-Taber L, et al. Exercise and type 2 diabetes: The American College of Sports Medicine and the American Diabetes Association: Joint position statement. *Diabetes Care*. 2010;33:e147–e167.
- 145. Yki-Jarvinen H, Koivisto VA. Effects of body composition on insulin sensitivity. *Diabetes*. 1983;32:965–969.
- 146. Borer KT. Exercise Endocrinology. Champaign, IL: Human Kinetics; 2003.
- 147. Kirwan JP, Hickner RC, Yarasheski KE, Kohrt WM, Wiethop BV, Holloszy JO. Eccentric exercise induces transient insulin resistance in healthy individuals. *J Appl Physiol*. 1992;72:2197–2202.
- 148. Tuominen JA, Ebeling P, Bourey R, Koranyi L, Lamminen A, Rapola J, Sane T, Vuorinen-Markkola H, Koivisto VA. Postmarathon paradox: Insulin resistance in the face of glycogen depletion. *Am J Physiol*. 1996;270:E336–E343.
- 149. Cryer PE. Banting lecture. Hypoglycemia: The limiting factor in the management of iddm. *Diabetes*. 1994;43:1378–1389.
- 150. Schwartz SS, Kohl BA. Glycemic control and weight reduction without causing hypoglycemia: The case for continued safe aggressive care of patients with type 2 diabetes mellitus and avoidance of therapeutic inertia. *Mayo Clin Proc.* 2010;85:S15–S26.
- 151. Pappada SM, Cameron BD, Rosman PM, Bourey RE, Papadimos TJ, Olorunto W, Borst MJ. Neural network-based real-time prediction of glucose in patients with insulindependent diabetes. *Diabetes Technol Ther*. 2011;13:135–141.
- Bax JJ, Young LH, Frye RL, Bonow RO, Steinberg HO, Barrett EJ, ADA. Screening for coronary artery disease in patients with diabetes. *Diabetes Care*. 2007;30:2729–2736.
- 153. Ziegler AG, Pflueger M, Winkler C, Achenbach P, Akolkar B, Krischer JP, Bonifacio E. Accelerated progression from islet autoimmunity to diabetes is causing the escalating incidence of type 1 diabetes in young children. *J Autoimmun.* 2011;37:3–7.
- 154. Verbeeten KC, Elks CE, Daneman D, Ong KK. Association between childhood obesity and subsequent type 1 diabetes: A systematic review and meta-analysis. *Diabet Med*. 2011;28:10–18.

- 155. Krause Mda S, de Bittencourt PI, Jr. Type 1 diabetes: Can exercise impair the autoimmune event? The L-arginine/glutamine coupling hypothesis. *Cell Biochem Funct*. 2008;26:406–433.
- 156. Riddell M, Perkins BA. Exercise and glucose metabolism in persons with diabetes mellitus: Perspectives on the role for continuous glucose monitoring. *J Diabet Sci Technol*. 2009;3:914–923.
- 157. Conway B, Miller RG, Costacou T, Fried L, Kelsey S, Evans RW, Orchard TJ. Temporal patterns in overweight and obesity in type 1 diabetes. *Diabet Med.* 2010;27:398–404.
- Berger M, Berchtold P, Cuppers HJ, Drost H, Kley HK, Muller WA, Wiegelmann W, et al. Metabolic and hormonal effects of muscular exercise in juvenile type diabetics. *Diabetologia*. 1977;13:355–365.
- 159. Ebeling P, Tuominen JA, Bourey R, Koranyi L, Koivisto VA. Athletes with IDDM exhibit impaired metabolic control and increased lipid utilization with no increase in insulin sensitivity. *Diabetes*. 1995;44:471–477.
- 160. Pan XR, Li GW, Hu YH, Wang JX, Yang WY, An ZX, Hu ZX, et al. Effects of diet and exercise in preventing niddm in people with impaired glucose tolerance. The Da Qing IGT and Diabetes Study. *Diabetes Care*. 1997;20:537–544.
- 161. Tuomilehto J, Lindstrom J, Eriksson JG, Valle TT, Hamalainen H, Ilanne-Parikka P, Keinanen-Kiukaanniemi S, et al. Prevention of type 2 diabetes mellitus by changes in lifestyle among subjects with impaired glucose tolerance. *NEngl J Med*. 2001;344:1343–1350.
- 162. Ramachandran A, Snehalatha C, Mary S, Mukesh B, Bhaskar AD, Vijay V, Indian Diabetes Prevention Programme (IDPP). The Indian Diabetes Prevention Programme shows that lifestyle modification and metformin prevent type 2 diabetes in Asian Indian subjects with impaired glucose tolerance (IDPP-1). *Diabetologia*. 2006;49:289–297.
- 163. Knowler WC, Barrett-Connor E, Fowler SE, Hamman RF, Lachin JM, Walker EA, Nathan DM. Reduction in the incidence of type 2 diabetes with lifestyle intervention or metformin. *N Engl J Med.* 2002;346:393–403.
- 164. Hopper I, Billah B, Skiba M, Krum H. Prevention of diabetes and reduction in major cardiovascular events in studies of subjects with prediabetes: Meta-analysis of randomised controlled clinical trials. *Eur J Cardiovasc Prev Rehabil.* 2011;18:813–823.
- 165. Dunkley AJ, Charles K, Gray LJ, Camosso-Stefinovic J, Davies MJ, Khunti K. Effectiveness of interventions for reducing diabetes and cardiovascular disease risk in people with metabolic syndrome: Systematic review and mixed treatment comparison meta-analysis. *Diabetes Obes Metab.* 2012;14:616–625.
- 166. Li G, Zhang P, Wang J, Gregg EW, Yang W, Gong Q, Li H, et al. The long-term effect of lifestyle interventions to prevent diabetes in the China Da Qing Diabetes Prevention Study: A 20-year follow-up study. *Lancet*. 2008;371:1783–1789.
- 167. Lindstrom J, Ilanne-Parikka P, Peltonen M, Aunola S, Eriksson JG, Hemio K, Hamalainen H, et al. Sustained reduction in the incidence of type 2 diabetes by lifestyle intervention: Follow-up of the Finnish Diabetes Prevention Study. *Lancet*. 2006;368:1673–1679.
- 168. Diabetes Prevention Program Research Group, Knowler WC, Fowler SE, Hamman RF, Christophi CA, Hoffman HJ, Brenneman AT, et al. 10-year follow-up of diabetes incidence and weight loss in the diabetes prevention program outcomes study. *Lancet*. 2009;374:1677–1686.
- 169. Walker KZ, O'Dea K, Gomez M, Girgis S, Colagiuri R. Diet and exercise in the prevention of diabetes. *J Human Nutr Diet*. 2010;23:344–352.
- 170. Ryan DH, Espeland MA, Foster GD, Haffner SM, Hubbard VS, Johnson KC, Kahn SE, Knowler WC, Yanovski SZ, Look ARG. Look ahead (action for health in diabetes): Design and methods for a clinical trial of weight loss for the prevention of cardiovascular disease in type 2 diabetes. *Control Clin Trials*. 2003;24:610–628.

- 171. Brancati FL, Evans M, Furberg CD, Geller N, Haffner S, Kahn SE, Kaufmann PG, et al. Midcourse correction to a clinical trial when the event rate is underestimated: The Look AHEAD (Action for Health in Diabetes) Study. *Clin Trials*. 2012;9:113–124.
- 172. Look ARG, Wing RR. Long-term effects of a lifestyle intervention on weight and cardiovascular risk factors in individuals with type 2 diabetes mellitus: Four-year results of the Look AHEAD trial. *Arch Intern Med.* 2010;170:1566–1575.
- 173. Wing RR, Rosen RC, Fava JL, Bahnson J, Brancati F, Gendrano Iii IN, Kitabchi A, Schneider SH, Wadden TA. Effects of weight loss intervention on erectile function in older men with type 2 diabetes in the Look AHEAD trial. J Sex Med. 2010;7:156–165.
- 174. Phelan S, Kanaya AM, Subak LL, Hogan PE, Espeland MA, Wing RR, Burgio KL, et al. Weight loss prevents urinary incontinence in women with type 2 diabetes: Results from the Look AHEAD trial. *J Urol.* 2012;187:939–944.
- 175. Foy CG, Lewis CE, Hairston KG, Miller GD, Lang W, Jakicic JM, Rejeski WJ, et al. Intensive lifestyle intervention improves physical function among obese adults with knee pain: Findings from the Look AHEAD trial. *Obesity (Silver Spring)*. 2011;19:83–93.
- 176. Schwartz AV, Johnson KC, Kahn SE, Shepherd JA, Nevitt MC, Peters AL, Walkup MP, et al. Effect of 1 year of an intentional weight loss intervention on bone mineral density in type 2 diabetes: Results from the Look AHEAD randomized trial. *J Bone Miner Res.* 2012;27:619–627.
- 177. Foster GD, Borradaile KE, Sanders MH, Millman R, Zammit G, Newman AB, Wadden TA, et al. A randomized study on the effect of weight loss on obstructive sleep apnea among obese patients with type 2 diabetes: The sleep ahead study. *Arch Intern Med.* 2009;169:1619–1626.
- 178. Chou YT, Lee PH, Yang CT, Lin CL, Veasey S, Chuang LP, Lin SW, Lin YS, Chen NH. Obstructive sleep apnea: A stand-alone risk factor for chronic kidney disease. *Nephrol Dial Transplant*. 2011;26:2244–2250.
- 179. Sharma S, Gebregziabher M, Parker AT, Abro JA, Armstrong AM, Schoepf UJ. Independent association between obstructive sleep apnea and noncalcified coronary plaque demonstrated by noninvasive coronary computed tomography angiography. *Clin Cardiol.* 2012;35:641–645.
- Martinez D, Klein C, Rahmeier L, da Silva RP, Fiori CZ, Cassol CM, Goncalves SC, Bos AJ. Sleep apnea is a stronger predictor for coronary heart disease than traditional risk factors. *Sleep Breath*. 2012;16:695–701.
- Sorajja D, Gami AS, Somers VK, Behrenbeck TR, Garcia-Touchard A, Lopez-Jimenez F. Independent association between obstructive sleep apnea and subclinical coronary artery disease. *Chest.* 2008;133:927–933.
- 182. Ramirez-Velez R. A 12-week exercise program performed during the second trimester does not prevent gestational diabetes in healthy pregnant women. *J Physiother*. 2012;58:198.
- 183. Stafne SN, Salvesen KA, Romundstad PR, Eggebo TM, Carlsen SM, Morkved S. Regular exercise during pregnancy to prevent gestational diabetes: A randomized controlled trial. *Obstet Gynecol.* 2012;119:29–36.
- 184. Han S, Middleton P, Crowther CA. Exercise for pregnant women for preventing gestational diabetes mellitus. *Cochrane Database of Syst Rev.* 2012;7:CD009021.
- 185. American Diabetes Association, Bantle JP, Wylie-Rosett J, Albright AL, Apovian CM, Clark NG, Franz MJ, et al. Nutrition recommendations and interventions for diabetes: A position statement of the American Diabetes Association. *Diabetes Care*. 2008;31 Suppl 1:S61–S78.

4 Atherosclerosis

C. Murray Ardies and Christian K. Roberts

CONTENTS

4.1	Introduction			134
	4.1.1	Exercise	e, Diet, and Atherosclerosis	136
4.2	Molecular Etiology of Atherosclerosis			138
	4.2.1	Stress Response or Adaptation		
	4.2.2	Adaptation		141
	4.2.3	Stress-Response Events in Atherosclerosis		145
	4.2.4	Coronary Heart Disease Risks		149
		4.2.4.1	Low-Density Lipoprotein Cholesterol (and	
			High-Density Lipoprotein)	150
		4.2.4.2	Diabetes and Insulin Resistance	152
		4.2.4.3	Metabolic Syndrome and Obesity	155
		4.2.4.4	Infection	157
		4.2.4.5	Cigarette Smoking	158
		4.2.4.6	Nutritional Status/Redox Control	159
4.3	Etiology of Prevention			160
	4.3.1	Exercise		160
		4.3.1.1	Serum Lipids	161
		4.3.1.2	Antioxidant/Redox Status	162
		4.3.1.3	Inflammatory Signaling	162
		4.3.1.4	Insulin Sensitivity	164
	4.3.2	Diet		166
		4.3.2.1	Nutrient Status	168
		4.3.2.2	Serum Lipids	168
		4.3.2.3	Inflammation	169
4.4	Summary and Recommendations			177
	4.4.1	Etiology of Coronary Heart Disease17		177
	4.4.2	4.4.2 Dietary Recommendations for Prevention of Coronary Hea		
		Disease		178
	4.4.3	B Physical Activity Recommendations for Prevention of Co		
		Heart D	isease	179
Acknowledgment				180
References				180

4.1 INTRODUCTION

Coronary heart disease (CHD) and its primary cause, atherosclerosis, are the leading causes of mortality in developed countries and the number one killer of both men and women in the United States. Every year since 1919, CHD was ranked as the number one killer in the United States; in 2008, CHD accounted for approximately 32% of all deaths with direct and total costs being estimated to be \$273 and \$444 billion, respectively.¹⁻⁴ Interestingly, there has been an almost 50% decline in mortality from heart disease between the 1960s and the 1990s, a decline that has continued through the last decade with the latest reported decline in age-adjusted mortality of approximately 6.7% to 6.0% between 2006 and 2010.^{5,6} The decline in mortality is due in part to better control of most of the traditional risk factors, that is, reduced smoking rates and better pharmaceutical control of hypertension, diabetes, and the dyslipidemias.⁶⁻⁹ In addition to controlling risk factors, various technologies for treating CHD patients have undergone tremendous improvements over the same time frame and account for as much as one-half of the total decline in mortality.¹⁰⁻¹³ However, the traditional risk factor obesity has increased in prevalence, and risk behaviors such as physical activity and dietary patterns have been difficult to improve given current trends in technology, automation, and food production.

From the series of American Heart Association (AHA) updates of heart disease and stroke statistics from 2007 through 2012 (and others), there appears to have been a slight decline in overall prevalence among whites (6.6% to 6.4%) and Hispanics (6.0% to 5.2%), but a moderate increase among blacks (5.2%-6.2%) has been observed. As is expected, the prevalence increases with age from a low of 1.2% for 18-44 years to 7.1% for 45-64 years and 19.8% above 65 years, with males incurring a higher prevalence than females prior to menopause. In this same time frame, there has been a relatively large decline in the incidence of CHD in 35- to 44-year-old men from approximately 7% to approximately 3%, whereas incidence among 85- to 94-year-old men has remained essentially the same.^{3,7,14-19} Interestingly, these statistical updates also reveal that there has been a consistent reduction in the prevalence of traditional risk factors in the population over this same time, especially among younger persons. This younger age group has been on the receiving end of intensive public campaigns against smoking; against eating high-fat diets; and for getting checked and treated for hypertension, hyperglycemia/diabetes, and hyperlipidemias throughout most of their lives. The observed reduction in prevalence of these factors probably demonstrates some success for the public health education initiatives. Both physicians and patients are certainly far more aware of the importance of managing the traditional risk factors. The lack of change in incidence among the older population is more than likely a reflection of many years of poor lifestyle habits throughout most of their early lives prior to the public health initiatives and might indicate that attention to risks later in life cannot significantly alter the progression of well-developed disease.

Two other observations from these statistical updates are probably the most important from the perspective of this chapter. The first is that the percentage of Americans who participate in any physical activity has steadily declined among all ethnic groups since the 1990s; in the 2012 update, 33% of adults reported no activity at all with half of all adults not meeting either the aerobic or the strength guidelines. As is well known, self-reports of physical activity tend to be exaggerated, with men and women reporting about 44% and 138% greater activity, respectively, than they actually perform.²⁰ Because the majority of surveys on activity, diet, and health issues rely on selfreported data, such discrepancies are common and it certainly is possible that the actual increases in inactivity-based risks published may be much worse than reported. The second observation is with respect to diet. There has also been a steady increase in the reported consumption of total calories, 22% for women and 10% for men, between 1970 and 2005 and a huge increase in obesity rates during the same period. As with physical activity, self-reported dietary intake of calories is often underreported by between 15% and 25% nonobese controls compared to actual measurements using the double-labeled water technique, with obese and diet-resistant groups underreporting intake from 38% to as high as 59%.²¹ Thus, the actual increase in caloric intake between 1970 and 2005 may be higher than the reported 10% (male) and 22% (female). The prevalence of obesity in men increased from 12.2% in 1971-1974 to 20.5% and 33.3% in 1994-1999 and 2005-2008, respectively. In women, the prevalence increased from 16.8% to 26.0% and to 36.2% in the same time periods. Obviously, a decline in physical activity combined with an increase in caloric intake contributes to weight gain and increase in obesity rates. Although small declines in total fat and protein intake and an increase in carbohydrate intake have occurred over the same time frame, diet has changed very little overall in terms of fruit, vegetable, and whole-grain consumption and the average American still consumes far less than the minimum recommended servings of each. On the basis of these data, the overall reduction in mortality from atherosclerosis to date is mostly due to a combination of better surgical treatment of symptomatic atherosclerosis and more effective pharmaceutical management of hypertension, hyperglycemia, and hyperlipidemias.

While factors associated with both diet and exercise impact many chronic diseases, including atherosclerosis, none of these factors has changed in a beneficial direction over the last four decades. One possible contributor is an emphasis on the importance of medical management of the traditional CHD risks while minimizing the impact of diet and exercise. For example, who has not seen the commercial on American television for a particular cholesterol-lowering drug that starts with a phrase something along the following lines: "If diet and exercise are not enough to lower your cholesterol... you need _____? The implication of such public advertising is that cholesterol is the major cause of CHD risk and that proper exercise and diet have no real protective effect other than lowering cholesterol. Although many in the medical and health professions have decidedly different ideas on this compared to the pharmaceutical companies who are trying to sell a product, the negative effect on public opinion is still the same. In a related area of public discourse, isolated culprits such as high-fructose corn syrup (HFCS) and refined sugar have currently got significant blame for the rising obesity rates, all the while overlooking the fact that these diseases are multifactorial. As with the mistaken belief that a single food component can be the major reason for the protective effects of a diet, blaming a single dietary component for obesity and chronic diseases is open to the same level of criticism. In addition to the somewhat one-sided publicity, perhaps there is also a little bit of a general misunderstanding on just how important proper diet and exercise are in reducing risk for CHD.

4.1.1 EXERCISE, DIET, AND ATHEROSCLEROSIS

In the research literature, physical activity and exercise training have consistently been observed to be a factor in reducing risk for atherosclerosis even though the mechanisms of the protective effect have not been well described. Inverse associations have been observed between physical activity and both CHD and CHD mortality.^{22,23} Relative risks (RRs) for CHD appear to be between 1.5 and 3 times higher for inactive persons compared to active,^{24–27} and physical activity is comparable to conventional risk factors in the ability to predict risk.^{28,29} Results of meta-analyses of available clinical studies, as well as reports from the Council on Clinical Cardiology of the AHA, clearly support the effects of physical activity and exercise on CHD risk reduction in both men and women, including those with heart failure and peripheral arterial disease.^{30–33}

Among active persons, the level of fitness also appears to be associated with risk. High levels of cardiorespiratory fitness, as measured by VO₂ max, are associated with a slower progression of existing atherosclerosis and with decreased mortality in both women and men.^{34–38} In a meta-analysis of studies on fitness level and CHD risk, the RR for CHD events for the least fit subjects was 1.56 and for the mid-level fit individuals it was 1.40 in comparison to those with the highest level of fitness.³⁰ In a similar manner, reductions in RR of between approximately 20% and approximately 45% between most fit and least fit subgroups are commonly seen with the degree of reduction being directly related to the weekly levels of physical activity.³⁹ Interestingly, in the Health Professionals' Follow-Up Study, men who trained with weights for at least 30 min/week also had a 25% reduction in CHD risk.⁴⁰ This effect was independent of the general reduction in risk afforded by other forms of exercise, and they suggested this might be associated with the ability of strength training to enhance glycemic control. They also observed a reduction in risk with moderate- to low-intensity exercise that appeared to be associated with improvements in lipid profile, with additional reductions in RR occurring with more intense and longer duration exercise. These additional RR reductions appeared to be due to mechanisms other than changes in lipids or glucose levels. Although not known at the time, the additional protection observed was most likely afforded by the anti-inflammatory effects of longer term moderate-intensity exercise.

The association between diet and CHD was also investigated throughout the 1900s, although the vast majority of research focused on dietary components and their effects on serum cholesterol throughout much of the century.⁴¹ A variety of feeding studies as well as clinical and epidemiological studies support the concept that saturated fat and dietary cholesterol increase serum cholesterol and are associated with CHD mortality.^{42–45} Keys' Seven Countries Study examined risk factors for CHD in over 12,000 men, and both the average population intake of saturated fat and the changes in average serum cholesterol levels were found to be strongly related to CHD mortality rates. The Framingham Heart Study and Multiple Risk Factor Intervention Trial also emphasized the relationship between serum cholesterol, especially low-density lipoprotein cholesterol (LDL-CHOL), and CHD.^{46–48} In comparing the U.S. experience with that of China, fat intake was less than 0.5 that in the United States, fiber intake was 3 times higher than that in the United States,

animal protein intake was approximately 10% of the U.S. intake, and serum total cholesterol (total-CHOL) was 127 mg/dL in rural China versus 203 mg/dL in the United States for adults aged 20-74 years. Mortality due to CHD was 16.7-fold greater for U.S. men and 5.6-fold greater for U.S. women than for their Chinese counterparts. Although results such as these further cemented the idea that high-fat diets both raised serum cholesterol and were correlated with CHD, an interesting conclusion from this study that probably merited much more attention was that there was no evidence of a threshold beyond which further benefits did not accrue with increasing proportions of plant-based foods in the diet.⁴⁹ Migration studies have also provided compelling evidence for the relationship between saturated fat intake and CHD, whereas others such as the Nurses' Health Survey provided clear documentation that saturated fat intake is associated with CHD and that intake of trans-fatty acids also increases LDL-CHOL.50-52 These early data have been confirmed by the more recent cohort studies. The Nurses' Health Study reported that saturated and trans-fatty acids are associated with an increased risk for CHD. Interestingly, intakes of flavonols were also independent contributors in explaining population differences in CHD mortality rates, suggesting that low-density lipoprotein (LDL) oxidation may also be critical to the progression of atherosclerosis.53

All this evidence of associations between diet, serum cholesterol, and CHD risk must be qualified, however, by the fact that they are based on overall associations observed in populations and that these associations do not demonstrate cause. When these same values of cholesterol or fat consumption and serum cholesterol are correlated within individuals or compared between individuals who have CHD and those who do not, the associations do not hold up. In a variety of comparison studies (Framingham, Tecumseh, Honolulu, and Puerto Rican men), essentially no associations were observed between serum cholesterol and dietary consumption of either fat or cholesterol and differences in serum cholesterol between patients and nonpatients were minimal.⁴⁴ This does not mean that no association is possible between cholesterol and CHD; rather, this illustrates the difficulties in interpreting general associations observed from population-based data as causal.

Other components of diet also have garnered some attention, most notably the type of carbohydrate that has been associated with CHD risk. Refined carbohydrates are highly processed, resulting in the removal of some or most of the fiber, vitamins, minerals, phytonutrients, and essential fatty acids. Consumption of refined carbohydrates in comparison with whole-grain sources is associated with increased risk for CHD;^{54,55} this is possibly related to the increased glycemic load of these types of carbohydrates.⁵⁶ Increased fiber consumption also is inversely related to both CHD and all-cause mortality; it also lowers LDL-CHOL levels and improves insulin sensitivity.^{57–61} However, the direct effects of fiber per se on these parameters are very difficult to determine because it is the consumption of fruits and vegetables that provides most of the fiber and in many studies it is the consumption of fruits and vegetables (along with their phytochemicals) that is associated with a decreased CHD risk.^{62–64} Additionally, while moderate consumption of protein is associated with a reduced risk of CHD, lower protein diets tend to be higher in fruit and vegetable intake.⁶⁵

Other diet-related risks also reveal the potentially protective aspects of a healthy diet. In the Nurses' Health Study cohort, in which 84,129 women aged 30–55 years

were enrolled and followed up for 14 years,⁶⁶ a healthy lifestyle was defined as not smoking, consuming at least half a drink of alcoholic beverage per day, engaging in moderate to vigorous physical activity for >30 min/day, and a body mass index <25 kg/m². The study also defined a healthy diet as one that included components such as cereal fiber, marine *n*-3 fatty acids, folate, low trans-fatty acids, and low glycemic load foods. Adherence to these factors correlated inversely with 14-year CHD incidence. Others have noted that 82% of CHD events could be prevented by a combination of physical activity and diet, providing evidence for an additive or a synergistic effect.⁶⁶ Comparing Mediterranean-type diets to the American diet reveals that consumption of vegetables, fruit, legumes, whole grains, fish, and poultry is associated with a decreased risk of CHD, whereas the typical American diet patterns with higher intakes of red and processed meats, refined grains, sweets/ desserts, high-fat dairy products, and low fruit and vegetable intake are associated with increased risks for CHD, risks that are independent of other risk factors.^{67,68}

From the various studies conducted over the last 40 years or more, it is clear that performing regular physical exercise and consuming a proper diet are associated with a reduction in risk for CHD. Although such associations are important, they do not define how these lifestyle risk factors affect mechanisms of cause. Even though there are many clinical studies that demonstrate that various dietary and exercise treatments reduce risk factors for CHD, factors such as serum lipids and cholesterol, such studies address only a variable that is correlated with disease and not a mechanism of cause. In addition to a lack of attention to mechanisms of cause, none of the aforementioned studies conducted nutritional assessments to determine the nutritional status of their subjects. With the majority of Americans consuming insufficient amounts of one or more of the essential nutrients, there is sufficient cause to suspect that nutritional insufficiencies and deficiencies are serious confounders that are not addressed in these studies.^{69–71} The only way to understand why any of these observed associations might exist is through a thorough understanding of the molecular etiology of atherosclerosis. From that point, the mechanisms through which dietary components and exercise alter the causal mechanisms will reveal how these lifestyle variables reduce risk for atherosclerosis.

4.2 MOLECULAR ETIOLOGY OF ATHEROSCLEROSIS

4.2.1 Stress Response or Adaptation

Although macrophages were recognized to be components of atherosclerotic lesions as early as the 1930s, the predominant focus of CHD research in the early 1980s was on lipoproteins and cholesterol; lipid-transport function and metabolism; and how various pharmaceutical, dietary, and other lifestyle factors could affect serum cholesterol.⁴¹ This was in spite of the fact that the response-to-injury hypothesis for the pathogenesis of atherosclerosis was first published by Russell Ross in 1973, a hypothesis that was developed further in 1986 and expanded again in 1993 and 1999 into what has become the currently accepted inflammatory model for the etiology of atherosclerosis.^{72–75} Even though the major emphasis of most CHD research through the 1980s was still on factors associated with serum cholesterol, a plethora of new information on

the cell biology of inflammatory responses made it possible to recognize the complex relationships between the multiple factors that alter cell function and the subsequent development of inflammation-associated cellular responses and their association with atherosclerosis.⁴¹ Since that time, a massive amount of information has been developed to implicate inflammation-associated mechanisms that are initiated in various cells within the vascular wall as the cause. These mechanisms are in response to shear-and pressure-associated hemodynamic forces in addition to oxidized lipids, reactive oxygen species (ROSs) from various exogenous and endogenous sources, and glucose and fructose, implicating all of them as components of cause.^{41,75–89} In addition to these factors, both innate and adapted inflammatory processes that follow viral and bacterial infections have been proposed to contribute to the process.^{75,76,90–97}

An important concept relating to the mechanisms of cause is whether the inflammatory mechanisms that ultimately lead to CHD are a response to abnormal tissue damage and stresses or a component of an adaptive response to normal cellular stresses. The distinction is a conceptual one because if CHD (in the context of disease processes) is a result of normal adaptive functions initiated by ongoing (normal) cellular stresses then it is simply not preventable; it is only modifiable. What this means is as follows: although the mechanisms of the disease cannot be completely stopped, they might be attenuated to the point that symptomatic disease does not occur. From a tissue damage perspective, endothelial cells of the arterial circulatory system are on the receiving end of potentially damage-inducing agents. For example, molecules with accessible carbonyls, such as the aldehyde of open-chain glucose and the ketone of open-chain fructose, can form a Schiff base with available amino groups on membrane proteins and amino-phospholipids and initiate the formation of advanced glycation end products (AGEs).87,98,99 These products are involved in causing cellular damage by enhancing rates of lipid oxidation in membrane phospholipids and in lipoproteins, as well as activating proinflammatory signaling pathways by interacting with AGE receptors (RAGEs) on vascular endothelial cells, macrophages, and smooth muscle cells.84,85,100,101 As a result, glucose and fructose have been implicated as important components of atherosclerotic risk because they can lead to membrane damage, lipoprotein oxidation, and proinflammatory responses, processes that certainly help explain the greater risks for atherosclerosis in diabetics.

Atherosclerotic plaque, however, does not form equally in all locations throughout the arterial circulation, even though arterial concentrations of glucose and fructose are essentially the same everywhere. In atherosclerotic patients, the plaque forms predominantly in the wall of arteries at bifurcations, including immediately distal to or at the leading edge of the bifurcations themselves, on the lee side of branch points. The plaque also forms on the inside curve of sharply curved stretches, especially in the aorta and coronary arteries, carotid arteries, and the femoral and kidney vasculature.^{83,102–104} It is relevant to consider that the distribution throughout the arterial system of all major risk factors such as glucose, fructose, hypertension, cholesterol, and various lipoproteins and any prooxidant molecules originating from the environment is essentially the same everywhere. With an equal distribution of damaging agents, there must be nondamaging factors involved in initiating inflammation-associated responses to produce such large differences in atherosclerotic plaque formation at different vascular locations.

The major factors that differ at the different locations of the arterial system are those of blood pressure and blood flow hemodynamics. The obvious difference in pressure is the gradient in blood pressure that exists from the aorta through to the capillaries in order to ensure circulatory flow, and it happens that the atherosclerosisprone areas also tend to be in the zones of higher pressure. The tiny localized changes in vessel diameter that constantly regulate local pressures and flow characteristics in different regions of specific blood vessels are not always so obvious. As requirements for local flow change, vasoconstriction or vasodilation responses through the regulation of smooth muscle contraction occur at the local level. With changes in physical forces at these local sites, alterations in strength of the vessel wall might be necessary in order to handle the unusual stresses. Structural differences in vessels also contribute to differences in local hemodynamics; the presence of various arterial branch points, bifurcates, and sharply curved sections contributes to differences in shear forces and to the extent of laminar/turbulent flow throughout the vasculature. A much lower (relative to immediately adjacent areas) average shear stress that oscillates in response to the cardiac cycle, and in some cases includes turbulent flow, is characteristic of these locations. The arterial walls adapt to these low oscillating shear forces and turbulent flow, and to increases in tensile stress (stretching due to pressure), through a localized hypertrophy response, and the thickened vascular structure is characterized by a greater abundance of smooth muscle cells and elastic fibers.^{82,83,102,103,105,106} The additional smooth muscle cells and connective fibers are necessary to strengthen and modify the shape of that specific region of the artery to more efficiently respond to hemodynamic conditions and to ultimately ensure adequate blood flow. It is these hypertrophied regions that also happen to be the major sites of atherosclerotic plaque development, indicating that this adaptive response is fundamental to the CHD process.

Because the thickening of arterial wall does not occur in regions of high laminar shear forces, it most likely represents a response to low shear forces. This is logical because low shear would indicate low flow and the hypertrophy response is an attempt to enhance flow through the artery by slightly narrowing the diameter (and increasing the pressure) at that point. In response to increased tensile strain due to higher blood pressures, hypertrophy of the arterial wall also occurs.¹⁰³ This increase in strength of the artery reduces pressure-induced stretching and distortions in blood vessel shape that would disrupt laminar flow and induce turbulent flow immediately distal to that location, potentially compromising flow. Thus low shear, turbulent flow, and increased tensile forces are logical stimuli for initiating changes in arterial structure in order to ensure adequate perfusion of downstream tissues. One could imagine that from around the second trimester of pregnancy and throughout growth to late childhood the constant growth of the heart and vascular system would produce constant demands for continual arterial remodeling and angiogenesis in order to handle the changing growth, pressure, and flow demands of the growing individual. Results of many cadaver studies demonstrate that an initial thickening at the susceptible sites is evident by the late stage of fetal development and at a few months following birth, with the extent of the remodeling gradually increasing throughout growth to produce arterial thickening in a pattern of locations that matches the pattern of atherosclerosis-prone areas.^{103,107} In many cases

fatty lesions appear at these same sites in the late stage of fetal development and very early childhood, and then they progress throughout childhood to become a lot more extensive around the time of puberty. This demonstrates that what happens in utero and soon after birth may play a role in the development of atherosclerosis. It is relevant that some of the cadaver studies were conducted prior to the 1970s because these early studies indicate that fatty lesions were common in children long before our current diabetes and obesity "epidemics" and other potential risks ever became significant population-wide issues. Thus, the initiating CHD processes appear to be the result of an adaptive stress response to mechanical signals during periods of fast growth that are *interpreted* as insufficient blood flow and not to damage per se. The development of CHD in adults might therefore be a really unfortunate side effect of a mechanical stress response that is designed to ensure adequate perfusion of the heart. Of course, this does not mean that inflammation-associated mechanisms cannot add to the disease process because they certainly do. It is no coincidence that these hypertrophied areas have approximately a threefold increase in the number of tissue macrophages within the intima compared to the intima of other locations with higher and constant shear forces, making them far more sensitive to any subsequent damage or infection-associated proinflammatory stimuli.

Although such a concept may seem unreasonable, it is certainly logical from an evolutionary standpoint. A localized adaptive response to hemodynamics that guarantees adequate blood flow for the organism is a highly useful phenomenon when it comes to ensuring the demands of growth and development of a 10 g organism at approximately 10 weeks gestation to a 40+ kg adolescent individual. Even though the side effect of the response is a gradual buildup of plaque in susceptible locations of the vasculature that might ultimately be fatal, it happens slow enough to guarantee that the individual survives more than long enough to produce progenies and raise them to independence. Thus, a potentially fatal adaptive response that guarantees short-term survival of the individual ultimately has no detrimental effect on species survival.

4.2.2 Adaptation

Although every detail of the activation processes are not the major focus of this chapter, a summary of the important concepts is described. If low oscillating shear forces are to activate a hypertrophy response within the arteries, then mechanore-ceptor complexes in the various vascular-associated cells (endothelial cells, smooth muscle cells, platelets, fibroblasts, and tissue macrophages) would be necessary. A variety of surface proteins are capable of responding to mechanical stimulation, including the heparin sulfate proteoglycan components of the glycocalyx, caveo-lae, G-protein-coupled receptors, nicotinamide adenine dinucleotide phosphate (NADPH) oxidase, phosphatidylinositol 3 (PI3)-kinase, calcium channels, potassium channels, vascular endothelial growth factor (VEGF) receptors, platelet-derived growth factor (PDGF) receptors, platelet endothelial cell adhesion molecules, integrins, intracellular junctional proteins, and primary cilia.¹⁰⁸⁻¹²⁰ Of these, most are cell surface proteins that are involved in some form of cellular regulatory function, whereas the integrins and junctional proteins are more structural in that they ensure

cell-cell and cell-extracellular matrix connections. All have the ability to respond to changes in their three-dimensional structure from stretch, torsion, and tension forces that are caused by shear forces and by transluminal pressures. Proteins associated with the proteoglycan component of the glycocalyx, such as the syndecans, have an extracellular glycosaminoglycan component that extends into the glycocalyx as well as a cytosolic component that is physically linked to both G-protein receptors and actin filaments. These syndecans ensure that mechanical forces experienced by the cell surface are efficiently transduced to both regulatory and cytoskeletal structures within the cell. The integrins and junctional proteins ensure that the mechanical forces on the surface of cells are transduced to other cells and to structures they are in contact with, including the smooth muscle cells adjacent to the extracellular matrix. Thus, there is an extensive network of structures that not only responds to physical forces and ensures that the forces are transduced throughout the arterial wall but also can activate a variety of regulatory proteins that then affect cell function.

Repeated stretching/relaxation of the ion channels will allow the entry of calcium and potassium into the cell while stimulation of primary cilia on the surface membrane of endothelial cells activates the opening of cilia-specific as well as polycystin-1 (cilia-associated protein)-activated calcium channels. The entry of calcium stimulates a variety of cellular effects, including the activation of protein kinase C (PKC) and eNOS, with the resulting NO• activating protein kinase G (PKG) as well as inducing vasodilation.^{121,122} The elevated calcium also activates myosin light chain kinase to induce constriction of the vascular epithelial cells^{123,124} while the enhanced PKC and PKG activities activate Rho-GTPases, which interact to induce the phosphorylation of VE-cadherin. This leads to the disassembly of adherens junctions and tight junctions between endothelial cells and an enhanced permeability of the blood vessels.¹²⁴ Although these responses are classically associated with damageassociated inflammatory responses, they also occur in response to shear stress that is independent of any damage. The increased cytosolic calcium also will activate phospholipase A₂ (PLA₂), calcineurin, adenylate cyclase, phosphorylase kinase, and Ras, leading to the activation of the p38-mitogen-activated protein kinase (MAPK), JNK-MAPK, extracellular signal-related kinase (ERK)-MAPK, and nuclear factor κ -β (NF κ β) pathways as well as PI3 kinase (PI3K).^{113,125-132} As a result of activation of these pathways, expression of the angiogenesis factors PDGF, VEGF, and endothelin-1 as well as monocyte chemoattractant molecule-1 (MCP-1) is enhanced. Mechanical strain of VEGF and PDGF receptors also activates them in a manner independent of ligand binding, which adds to the growth factor-mediated increase in smooth muscle cell proliferation and fibrin deposition that ultimately enhances the strength of the vascular wall at the specific site of high shear stress.^{108,110,133} The disassembly of the cell-cell connections between endothelial cells and their proliferation that occurs in response to autocrine stimulation are events that contribute to the hypertrophy/angiogenesis responses that would be essential for the growth of arteries and heart. Thus, mechanical activation of various cellular structures activates a hypertrophy response that utilizes the same stress-response pathways that occur with inflammation-associated angiogenesis. A response that reiterates the general concept that, as described in Chapter 2, inflammation-associated responses are not necessarily a component of the (innate) immune system but rather are a series of cellular responses that make up a generalized stress response that can be activated by mechanical or metabolic stresses, or by either sterile damage or pathogen-induced damage. While the desired hypertrophy and angiogenesis responses occur following a mechanical stress-induced stress response, the signaling responses to mechanical stress unfortunately include the same signaling pathways that are involved in what is commonly known as an inflammatory response.

There is, however, a conundrum built into the aforementioned process; if shear stress induces a hypertrophy response, then logic might dictate that the hypertrophied areas should exist at locations of high shear stress and not where low (and oscillating) shear exists. To address this, comparing cellular responses to constant high shear stress with those in response to low oscillating shear forces reveals why low oscillating shear is hypertrophic/angiogenic (and atherogenic) while constant high laminar shear is antiatherogenic.^{82,105,109–111,113,134–140} From these studies. there is a consistent pattern of activation of Ras and the various MAPK and NFkß pathways by mechanical shear stress that leads to increased expression of PDGF, VEGF, MCP-1, vascular cell adhesion molecule (VCAM)-1, intracellular adhesion molecule (ICAM)-1, E-selectin, tumor necrosis factor (TNF)- α , interleukin (IL)-1, cvclooxygenase (COX)-2, and interferon- γ (INF γ) as well as other molecules that are associated with inflammatory responses (see Chapter 2 for details on these). A major difference between the two shear conditions is that with constant high laminar shear stress the responses are only temporary. Over a period of several seconds to hours, there is a downregulation of Ras and MAPK pathway activities and MCP-1 synthesis via an increased synthesis of $I\kappa\beta$ and activation of Nrf2, both of which inhibit the activation of NF $\kappa\beta$. In addition, a sustained expression of eNOS, COX-2, extracellular superoxide dismutase (eSOD; CuZn-SOD), cytosolic SOD (cSOD; CuZn-SOD), mitochondrial SOD (Mn-SOD), and the transcription factors heme oxygenase (HO)-1 and Kruppel-like factor 2 (KLF2) occurs. The enhanced synthesis of eNOS ensures that the blood vessel has an enhanced capacity for synthesizing NO• for essential vasodilation responses, whereas the increased synthesis of I $\kappa\beta$ and activation of Nrf2 ensures that NO•-mediated activation of protein kinase B (PKB) does not contribute to a continued progression into a full-blown inflammatory stress response. With the resulting suppression of NF $\kappa\beta$ and the MAPK pathways, the subsequent expressions of PDGF, VEGF, MCP-1, vascular endothelial adhesion molecule, intercellular adhesion molecule, E-selectin, TNF-a, IL-1, and INFy are reduced to the original nonactivated levels and in some cases much lower. Increased expression of the transcription factor KLF2 leads to alterations in the expression of a variety of proteins that are involved in regulating the basic phenotype of endothelial cells, including a reduction in the expression of both chemotactic ligands and their receptors, resulting in decreased activation and attraction of inflammatory cells in the immediate vicinity into the vessel wall. Other KLF2-responsive proteins are involved in the development of the overall morphology of cells such that they elongate and realign their longitudinal axis along the axis of flow. When aligned with the flow, the shear stresses on the cell are reduced and, therefore, the activation of the mechanosensitive structures is attenuated as well. In addition, primary cilia disassemble in response to prolonged shear, which eliminates a mechanosensitive activator of calcium entry.

From this discussion, many of the alterations in stress responses observed are reminiscent of those that occur during the resolution phase of an inflammatory response (described in detail in Chapter 2). The suppression of NF $\kappa\beta$ through the activation of Nrf2, along with an induction of COX-2 following chronic shear, also is observed during resolution. Although the effects of chronic shear on factors directly associated with resolution have not yet been delineated, it would certainly be a useful area of study in order to develop a more complete mechanistic understanding of the apparent antiinflammatory effects of chronic shear stress. From the observed sustained induction of COX-2, a variety of cellular effects would be expected. One of the enzymes that synthesize prostaglandin E₂ (PGE₂), the mPGE-1 form, is highly coinduced along with COX-2.141,142 This would result in an increased localized production of PGE2. Induction of PGE₂ will induce synthesis of 12/15-lipoxygenase (12/15-LOX) in endothelial cells as well as in any neutrophils or macrophages in the immediate region, resulting in an enhanced production of lipoxins (lipoxin A_4 and lipoxin B_4) from arachidonic acid (AA).^{143–147} Along with the lipoxins, resolvins and protectins are essential for the resolution phase with the E-series resolvins being produced from eicosapentaenoic acid (EPA) via a combination of COX-2 and constitutive 5-LOX activities.^{147–150} The D-series resolvins and the protectins are synthesized from docosahexaenoic acid (DHA) via a combination of 5/12/15-LOX activities.^{148–152} These resolution-activating ligands then bind to receptors on neutrophils, endothelial cells, macrophages, and monocytes^{150,153} to produce their proresolution effects by attenuating p38-MAPK, ERK-MAPK, and PI3K activities and by attenuating activation and expression of NFkB and AP-1.141,150,153-155 COX-2 also produces prostaglandin D₂, which spontaneously dehydrates to prostaglandin J₂ (PGJ₂) and activates peroxisome proliferator-activated receptor-y (PPARy).^{141,156,157} The increased binding of PGJ₂ to peroxisome proliferator-activated receptor (PPAR) would then reduce IL-1 β and TNF- α release by macrophages and dendritic cells, inhibit toll-like receptor (TLR) signaling, and inhibit NFκβ activation.^{157–160} Because Nrf2 activation has already been observed with chronic high shear stress, and PGJ₂ also can activate this factor, other resolution-associated adaptations to chronic high shear may occur. Activation of Nrf2 stimulates an upregulation of HO-1 as well as electrophile-response element (ERE) -responsive gene products such as glutathione peroxidase (GPX), SOD, catalase (CAT), glutathione-S-transferase, and UDP-glucuronosyl transferase (UDP-GT).^{159,161–165} From these, it is tantalizing to hypothesize that what appears to be an adaptive response to chronic high laminar shear is actually the ultimate expression of a chronic resolution-phase phenotype.

The endothelial cells in blood vessel walls therefore respond to chronic high laminar flow in such a manner that they are less affected by shear stress because of changes in cellular morphology and they also are far less responsive to any stress because their stress-response signaling pathways are inhibited. Unfortunately, under low oscillating shear stress the cells in that localized region retain the ability to continue the mechanically induced stress response for as long as the interrupted flow cycles remain and none of the protective responses occur.

Thus, atherosclerosis is caused by inflammation-associated mechanisms that are initiated by mechanical shear stresses and then exacerbated by the various proinflammatory mechanisms associated with risk factors such as hypertension, obesity, hyperglycemia, smoking, inactivity, poor diet, and dyslipidemia.

4.2.3 STRESS-RESPONSE EVENTS IN ATHEROSCLEROSIS

As described in Section 4.2.2, the mechanical stimulation by oscillating shear forces activates the stress-response activities of the p38–MAPK, JNK–MAPK, ERK–MAPK, and NF $\kappa\beta$ pathways as well as PI3K.^{113,125–132} Each of these stress-response actions comprises mechanisms that are classically associated with an inflammatory stress response and are described in detail in Chapter 2. Following the activation of these pathways, a variety of cellular responses occur, including (as listed in Section 4.2.2) increased expression of PDGF, VEGF, MCP-1, VCAM, ICAM, E-selectin, TNF- α , IL-1, and INF γ as well as increased PLA₂ and adenylate cyclase activities.^{82,105,109–111,113,134–138}

Activation of PLA₂ enhances the availability of AA, leading to the production of PGE₂, prostaglandin I₂ (PGI₂), thromboxane (TX) A₄, leukotriene (LT) A₄, and platelet-activating factor (PAF) through COX and LOX activities and the various prostaglandin (PG), LT, and TX synthase activities.^{84,141,166–168} The increase in PGI₂ and PGE₂ causes localized vasodilation via the activation of G-coupled IP and EP2/EP4 receptors.^{121,166,169} Although the localized vasodilation effects may not be contributing factors to the atherosclerotic process, the activation of the G-protein itself and the subsequent activation of PKG are certainly important from a signaling standpoint. An increase in vasopermeability occurs following the disassembly of adherens junctions and tight junctions, which, along with the increased expression of MCP-1, VCAM, ICAM, and E-selectin, leads to the attraction and entry of monocytes into the local area. The vasopermeability and attraction of tissue macrophages in areas of low oscillating shear forces observed in infants and young children, as already mentioned.

The processes that lead to the formation of plaque are, of course, the fundamental processes of atherosclerosis. It stands to reason that as long as the minimal processes that actively produce it continue to occur, progression to an atherosclerotic state would be the ultimate result. It does appear that mechanically induced stress response is minimally necessary to initiate the production of the first observable lesion in the development of atherosclerosis: the type I fatty lesion; the presence of this lesion in a fetus attests to that.^{103,107} The subsequent progression of type I lesions to type II through to the more clinically relevant type IV and potentially fatal type V and type VI lesions is then dependent on a continuation of the same initiating signaling processes as well as an acceleration of these processes by a variety of factors that are commonly associated with classical risks for the production of the advanced lesions,^{41,72–89} with the possible inclusion of factors associated with infection.^{75,76,90–97}

The initial development of plaque is largely a consequence of macrophages *consuming* lipids and their conversion into foam cells, thus creating the initial fatty streaks. The uptake of lipids by macrophages depends on scavenger receptor–mediated endocytosis of LDLs, with the major scavenger receptors that are highly relevant for atherosclerosis being the class A and CD36 forms.^{170–174} Both of these receptors are activated by oxidized and acetylated LDL proteins as well as oxidized cholesterol (CHOL-Ox), and they account for the majority of lipid uptake by macrophages. Both scavenger receptors also recognize a variety of bacteria,

microbial lipopolysaccharides and diacylglycerides, apoptotic cells, and AGEs, indicating that the major scavenger receptors responsible for lipid uptake and conversion of macrophages to foam cells are pattern recognition receptors (PRRs). Activation of CD36 by oxidized LDL and of oxidized CHOL leads to uptake of lipids and subsequent foam cell formation and also leads to JNK-MAPK activation of a variety of AP-1 subunits, activities essential for activating the synthesis of proinflammatory cytokines in macrophages.^{174–176} Oxidized LDL also activates TLR4 and TLR6 while AGE-LDL activates TLR4, leading to the activation of JNK-MAPK, p38-MAPK, AP-1 subunits, and NF $\kappa\beta$, resulting in increased TNF, IL-1, and IL-6 synthesis.^{177,178} Thus, uptake of oxidized and glycated LDL and CHOL by scavenger receptors on macrophages activates an inflammatory stress response through signaling pathways that are commonly activated by pathogens and damage and adds to the already existing mechanical activation of these signaling pathways. The presence of CD36 in endothelial cells and in smooth muscle cells ensures that these cells also participate in the production of proinflammatory signaling molecules through the same mechanism.170,176,179

The major mechanism missing from the aforementioned stress responses is a process that can enhance ROS-mediated oxidation of LDL and of CHOL. Further, it is the mechanical stress-mediated activation of G-proteins and their associated NADPH oxidase activity, as well as the direct activation of NADPH oxidase by the same mechanical stress, that provides the necessary mechanism.^{109,110,113,114,117} The immediate effect of stimulating G-proteins is the activation of their NADPH oxidase component on the cell membrane. Further, whether activated by mechanical stress or via G-protein stimulation, the oxidase generates superoxide ions outside the plasma membrane. The superoxide anions react with one another to produce peroxide and oxygen $O_2 \overline{\bullet} + O_2 \overline{\bullet} \rightarrow H_2 O_2 + O_2$, where the peroxide can either diffuse into the cell or react with other superoxide ions to produce hydroxyl radicals, $O_2 \overline{\bullet} + H_2 O_2 \rightarrow \bullet OH + HO^- + O_2$. The resulting $\bullet OH$ can then oxidize any LDL or CHOL (or any other molecules) in the immediate vicinity. The O_2 produced also can react with NO• to produce damaging nitroxide radicals, enhancing oxidative damage to LDL. In contrast to low oscillating shear, the increased expression of eSOD observed with chronic laminar shear stress ensures that activation of the NADPH oxidases does not necessarily result in enhanced ROS-mediated damage.

The last remaining issue is how to increase the entry of LDL into the vascular wall at the site of mechanical stress. This is accomplished through the mechanical stress response that enhances PKC, PKG, and Rho-GTPase activities that ultimately lead to the disassembly of adherens junctions and tight junctions between endothelial cells and enhances the permeability of blood vessels. While under noninflammatory conditions it is predominantly the smaller LDL particles that can penetrate the endothelial layer of the vessel wall to gain access to the tissue cells, the increased vasopermeability ensures that additional LDL particles can penetrate the layer in larger amounts to ultimately increase the concentration of LDL particles in the local region of the vessel wall. Thus, low oscillating mechanical shear stress activates all the requisite mechanisms that are necessary for the formation of foam cells: greater local concentration of tissue macrophages (via mechanical stress–induced inflammatory stress response), greater localized concentration of lipoproteins (via

increased vasopermeability), and greater production of extracellular ROSs (via enhanced NADPH oxidase activity). Further, because the uptake of oxidized LDL/ CHOL is mediated by scavenger PRRs, the mechanical stress–induced inflammatory responses are enhanced via PRR-mediated inflammatory signaling.

Because nonoxidized LDL proteins, cholesterol, and other lipids are not recognized by these scavenger receptors, it follows that cholesterol per se cannot be an initiating cause of the CHD processes. When oxidized LDL (LDL-Ox) proteins activate the scavenger receptors, uptake of the lipids that are part of the proteins is coincidental, not causal. Even when CHOL-Ox activates the scavenger receptors, the cholesterol itself can be viewed as a somewhat passive component of the process; but for the oxidizing process, it would not happen. Thus, oxygen radical mechanisms are essential to initiating events that lead to foam cell production, not cholesterol per se. LDL-CHOL may not be the direct initiating cause; it is, however, a risk factor. Increased levels of LDL-CHOL in serum obviously provide increased entry into the immediate area and therefore a greater numbers of targets for LDL and CHOL oxidation, thus explaining the general trend for an increase in risk as LDL-CHOL levels increase. At the same time, LDL-CHOL is one among many integrated risk factors, all of which affect one or more of the cellular mechanisms that are activated by mechanical stress. Thus, the large variations in associations between LDL-CHOL and CHD that are observed among different socioeconomic groups, cultures, and countries are logical.^{43,180–183} One of the more well-known differences is that originally reported by Serge Renaud¹⁸³ between France and the United States, with mean cholesterol levels being approximately 15% greater in the French and CHD deaths per 100,000 being approximately 37% of those in the United States. The greater levels of cholesterol in the serum of the French with much lower mortality due to CHD is impossible to explain if cholesterol is the cause of CHD. In light of current information, there is no paradox because cholesterol does not cause CHD; there are, however, different arrays of risk factors that directly affect the mechanisms that occur in response to those of the proximate cause, and an excess of one can be outweighed by a paucity of others.

From these discussions, intermittent low shear stress can be viewed as the proximate cause that initiates CHD, whereas inflammatory signaling, along with the normal scavenging function of macrophages, mediates the various cellular responses that comprise the processes of the disease. The risk factors of smoking, dyslipidemias, high blood pressure, obesity, infections, hyperglycemia, inadequate diet, and inadequate exercise are then viewed as accessories to the same disease processes (Figure 4.1). Further, the following does bear repeating: the role of these risk factors in the CHD process should not be underestimated because while oscillating low shear stress may be sufficient for the formation of type I and type II fatty lesions these do not necessarily progress into the more clinically relevant type IV–VI lesions.^{41,72–89,107,184,185}

Progression to more dangerous lesions demands continuing inflammationassociated processes beyond the formation of foam cells. Further, the expression of ICAM on vascular endothelial cells in response to the mechanical stress is an important consideration because platelets bind to the ICAM that is expressed on endothelial cells.^{186,187} This binding is independent of vascular damage or of exposure to von Willebrand factors or collagen proteins. On binding to ICAM, platelets degranulate and release the platelet aggregation factors fibrinogen, fibronectin, P-selectin, and



FIGURE 4.1 Proximate causes and risks for coronary heart disease: the activation of proinflammatory and prohypertrophy signaling by mechanical disturbances of vascular cells due to low oscillating shear forces and turbulent flow creates a sensitive proinflammatory environment within the cells located at bifurcates and inside the curves of arteries. The initial infiltration of inflammatory cells contributes to the hypertrophy–adaptation response to ensure adequate downstream flow, as well as to the formation of type I and II fatty lesions. Poor diet, overeating, and inadequate physical activity lead to poor redox control in cells, increased formation of advanced glycation end products (AGEs), increased oxidation of high-density lipoprotein (HDL)/low-density lipoprotein (LDL)/cholesterol (CHOL), increased blood pressure, and greatly enhanced pattern recognition receptor (PRR) activation with subsequent proinflammatory signaling, greatly increasing the risk for progression to the potentially fatal type V and VI lesions.

TXA₂, as well as PDGF; transforming growth factor β (TGF β); epidermal growth factor; basic fibroblast growth factor; and the chemokines RANTES, platelet factor 4, and IL-8 (among many others).^{186–188} These factors are very important for the formation of blood clots as well as for attracting fibroblasts, smooth muscle cells, and additional tissue macrophages into the immediate area for wound healing and

angiogenesis; other cells such as dendritic cells, mast cells, and T cells also can be attracted to the immediate region by the same chemokines. All these factors are responsible for the expansion of the asymptomatic type I, II, and III lesions into the more advanced type IV, V, and VI atherosclerotic lesions that extend into the vascular lumen, lesions that are characterized by extensive smooth muscle cell proliferation as well as the presence of fibroblasts, dendritic cells, mast cells, macrophages, T cells, neutrophils, aggregated platelets, and increased fibrosis in addition to more numerous foam cells.^{90,92,184,188,189} In this context, the proinflammatory role of the CD36 PRRs in the vascular endothelial cells (and the smooth muscle cells as well) is an important consideration in the atherosclerotic process. The endothelial cells not only provide a source of proinflammatory signals that is initiated by CHOL-Ox, a source that is in addition to the sources produced by macrophages and oscillating shear stress, but also provide a direct mechanistic link to platelet activation and the subsequent acceleration of events that are necessary for the formation of advanced atherosclerotic lesions. This is important because activated platelets appear to be an essential requirement for the development of the symptomatic lesions. Continued expansion of the lesions mediated by the expanding repertoire of local inflammatory cells will not only increase the size of the lesions but also enhance their instability through increased production of CD40 from the platelets, which activates an increased expression of matrix metalloproteinases (MMPs) by macrophages, endothelial cells, and smooth muscle cells. Any resident mast cells will secrete cathepsin G, which, along with the MMPs, can degrade collagen, VE-cadherin, and fibronectin, contributing further to the degradation of the matrix and greatly increasing the risk for rupture of the plaque.¹⁸⁸

From this viewpoint, primary prevention of CHD requires an emphasis on the central mechanisms of cause rather than on any single risk factor. It also suggests that dealing with all of the sources of risk is necessary because diminishing one risk does not change the fact that many other risk factors also contribute to the same mechanisms of cause. Prevention efforts must treat the disease process as a tightly interconnected network of biological mechanisms that all have significant contributions to the primary process and not focus on one or two isolated biological mechanisms to the exclusion of all others. Although research has been clear on this for many years, our clinical practices have unfortunately not caught up, which is an unfortunate phenomenon described by Claude Lenfent in 2003, who observed that in spite of the wide availability of published reports describing the unequivocal success of new therapies for the treatment of CHD, their universal adoption in the clinic within 10 and even 15 years is simply not seen.¹⁹⁰

4.2.4 CORONARY HEART DISEASE RISKS

In keeping with the focus on prevention that is based on the inflammatory mechanisms of the proximate cause, the molecular contributions of each of the risk factors that enhance the mechanical stress-induced CHD processes are described. In essence, anything that enhances ROS production, causes damage, or activates scavenger or other PRRs in vascular cells will either additively or synergistically enhance the production of foam cells and the progression from type I to type VI lesions. At the same time, any inflammatory signaling molecules that enter the local area from the circulation will also enhance these processes. In addition, conditions that lead to the inability to adequately regulate the stress responses can exacerbate the mechanisms of cause.

4.2.4.1 Low-Density Lipoprotein Cholesterol (and High-Density Lipoprotein)

As already discussed in detail in Section 4.2.3, LDL-CHOL levels in the blood provide a source of risk, predominantly through passive diffusion from the blood through the endothelial layer into the intima. Once the LDL-CHOL particles are localized to the immediate vicinity of the endothelial cells, smooth muscle cells, and macrophages, they are susceptible to attack by ROSs produced by metabolic and regulatory processes that are associated with normal cell functions. They are also susceptible to attack by any circulating oxidants that originate from exogenous sources such as food, contaminated air and water, tobacco smoke, or even some medications. LDL-Ox and CHOL-Ox are recognized by scavenger PRR receptors on the macrophages, smooth muscle cells, and endothelial cells, and on their uptake by the receptors proinflammatory signal transduction pathways are activated. Thus oxidation events are critical to the development of foam cells as well as to enhanced inflammatory signaling, and procedures that limit oxidative damage to LDL/CHOL will have profound effects on mechanisms of risk for CHD that are independent of LDL/CHOL levels.

There are a number of variables that can affect lipoprotein levels in the blood as well as in the tissues such as the genetic polymorphisms of LDL (and high-density lipoprotein [HDL]) receptors in the liver and in peripheral tissues, affecting cellular uptake. Genetic and metabolic factors associated with the synthesis of various apoproteins in the liver (and intestinal epithelia) and the subsequent packaging of different lipids (triglycerides, phospholipids, cholesterol) into the various classes of very-low-density lipoprotein (VLDL), intermediate-density lipoprotein, LDL, and HDL also contribute to LDL/HDL-associated risks, but all of these are beyond the scope of this chapter. This is not to mean that these are not biologically relevant issues; however, changing genes is not yet a practical part of prevention and the bottom line of the LDL/CHOL-associated risk is still how much LDL/CHOL is available for oxidation. As described later, treatments that lower LDL/CHOL levels in the blood also tend to lower the risk for atherosclerosis, although their effectiveness is not assured especially in the face of other risks.

Even though pharmacology-based prevention is not a topic for this book, the use of statins (HMG-CoA reductase inhibitors) that lower serum cholesterol has yielded some interesting results. Statins are well documented to lower rates of CHD progression as well as reduce risks for cardiac events such as heart attack, sudden cardiac death, and stroke in treated groups by as much as 25%–40%, while LDL-CHOL is typically reduced between 25% and 35%.^{191–194} Of interest though is the fact that in spite of the observed associations between lowered cholesterol and CHD events the lowered cholesterol may not be the mechanism behind the lowered risk. Relatively recent evidence has revealed that statins exert a profound anti-inflammatory effect through induction of eNOS, induction of KLF2, lipoxin synthesis by COX-2, PPAR α -mediated repression of NF $\kappa\beta$, a reduced expression of MCP-1 and PAF, and lowering of C-reactive protein (CRP) levels by as much

as 36%.^{195,196} The reason why this is relevant is that with cholestyramine therapy (a bile-acid binding resin to enhance cholesterol loss) similar reductions in serum cholesterol are observed.^{197–200} Cholestyramine therapy is also associated with a reduction in the rate of progression of early atherosclerotic lesions to more advanced lesions. However, unlike statins, similar significant reductions in fatal cardiac events are not seen. More than likely, it is the additional anti-inflammatory effects of the statins that are responsible for the reduction in fatal events and not the cholesterol-lowering effects per se. One could argue that reducing LDL/CHOL might reduce the rates of foam cell formation to attenuate the progression of early lesions to the more extensive type V or VI lesions; but without reducing inflammatory responses to attenuate the destabilization of existing lesions, fatal events associated with the formation of thrombi will still occur, even in the smaller lesions. And there are plenty of risk factors that enhance inflammatory signaling in spite of lower LDL/CHOL.

The role of HDL in increasing the risk for atherosclerosis is discussed here only briefly. HDL levels in the blood are well known to be inversely related to risk for CHD. The protective effects are thought to be due to the presence of paroxonases and apoA-I, which reduce lipid peroxides to inhibit LDL oxidation, and the ability of HDL to promote the efflux of cholesterol from foam cells and other tissue cells and transport them to the liver.^{194,201–204} HDLs are also known to attenuate local inflammation by inhibiting the formation of chemokines by vascular epithelial cells and the expression of proinflammatory cytokines by macrophages.

Such results have inspired the development of HDL-raising drugs such as Torcetrapib, Anacetrapib, and Dalcetrapib (cholesterol ester transport protein [CETP] inhibitors; CETP is associated with the transport of cholesterol from HDL to LDL, enhancing risk for CHD).²⁰⁵⁻²¹⁰ Unfortunately, as was seen with Torcetrapib, although HDLs increased and LDLs decreased, there were increases in mortality due to cardiac events and the trial was stopped due to the increase in risk. Although preliminary phase I and II trials with Anacetrapib and Dalcetrapib indicate that they do not have the same toxicity as Torcetrapib, the results are too preliminary to know if there are any actual benefits to taking these drugs. These trials, in light of the recent study by Voight and others,²¹¹ wherein genetic mechanisms that raise HDL are not associated with a reduction in risk for cardiac events raise doubts about the (exclusively) preventive effects of HDL. This last issue deserves comment. While the pharmaceutical-based studies are based on the HDL-raising effects of the drugs, the drugs themselves inhibit an enzyme that can transfer cholesterol to LDL from HDL. This is a function of HDL that contributes to risk for CHD by increasing the cholesterol content of LDL and therefore increasing the probability of LDL-CHOL oxidation. In addition, both native and oxidized HDLs bind to CD36 scavenger receptors on both macrophages and platelets, initiating a proinflammatory (and proatherogenic) response.^{212–215} Thus, HDL can participate in proatherogenic actions.

The ability of HDL to inhibit LDL oxidation is not a consistent effect; Navab and others have reported that no antioxidant effects were observed when using cocultures of human aortic endothelial cells and HDL fractions from their patients, even in patients with very high levels of HDL.^{216–220} On the other hand, the ability of HDL to protect against LDL oxidation can be improved through diet plus exercise interventions. Roberts and others^{221,222} have noted decreases in MPO, lipid

hydroperoxides, and HDL while an increase in protection against LDL oxidation per milligram of HDL was observed. They also documented a reduction in serumstimulated production of O_2 and peroxides. With such results, wherein lower HDL levels occur at the same time as enhanced protection from LDL oxidation by the HDLs, it is clear that the concept of high HDL levels being preventive based on population-based correlations is highly problematic. It is apparent that the physical properties of HDL particles are the relevant issue and not the level in serum and that the physical properties are responsive to lifestyle alterations. Because a clear protective role for raising serum HDL levels is ambiguous, various factors associated with raising serum levels of HDL as a preventive procedure will neither be discussed nor be recommended. When the physical properties of the protective forms of HDL have been elucidated and serum levels of protective HDL can be easily differentiated from those of nonprotective HDL, they (the protective forms of HDL) can become a biomarker of risk reduction.

4.2.4.2 Diabetes and Insulin Resistance

Details of the molecular etiology of diabetes are discussed in Chapters 3, so they are not detailed here. However, this condition is an important risk for CHD because overall levels of glucose in the blood will be increased. Normally, serum glucose is under very tight control through the effects of the hormones insulin (glucose uptake by insulin-responsive cells), glucagon (glycogenolysis to enhance glucose release to the blood), and cortisol (increase rates of gluconeogenesis), and under conditions of stress both epinephrine and IL-6 can enhance rates of glycogenolysis. The overall integrated effect of these hormones that are critical for glycemic control is to maintain levels somewhere within the normal fasting range of 70–100 mg/mL (4–6 mM) and somewhat less than 180 mg/mL (~10 mM) following meals. With insulin resistance, levels of glucose in the blood remain higher than normal, and as a result glycation reactions caused by glucose are increased.

As previously mentioned, the accessible carbonyls of the aldehyde of open-chain glucose and the ketone of open-chain fructose can form a Schiff base with available amino groups on DNA, phospholipids, and proteins. The Schiff bases can rearrange to produce Amadori (glucose) and Heyn's (fructose) products, which are the basis for the formation of dicarbonyls such as 3-deoxyglucosone, glyoxal, and methylglyoxal, which then form a variety of cross-linked advanced glycation end products such as pentosidine, carboxymethylethanolamine (CME), and N-(carboxymethyl)lysine (CML).87,98,99,101,223-226 From a kinetics standpoint, the formation of the freely reversible Schiff bases with open-chain glucose/fructose occurs in minutes following second-order kinetics (rate of formation decreases in proportion to the square of the glucose concentration, i.e., very quickly), whereas rearrangement to the more stable Amadori and Heyn's products takes from weeks to months to occur. In addition, the half-life of the Schiff bases is approximately 2.5 hours (hemoglobin in serum under physiological conditions); the concentration of the Schiff bases is essentially in equilibrium with serum levels of glucose. The cross-linked AGEs occur throughout all cells and tissues, and this appears to be a normal process associated with aging, indicating that these reactions occur with normal serum glucose levels but can be accelerated under high glucose loads. When these cross-links occur in vascular tissues, they are associated with decreased compliance and increased stiffness, properties that are responsible, in part, for the increased peripheral resistance and hypertension usually observed in diabetics. In addition, such cross-links could be hypothesized to enhance the efficiency at which mechanical stresses are transduced through the vascular wall and therefore enhance oscillating shear stress responses. Both of these effects (hypertension and increased mechanical transduction) would effectively enhance risk for proinflammatory signaling through mechanical stress and may be an area worth more research.

In addition to the mechanical stress issues, AGEs can bind with RAGEs on vascular endothelial cells, macrophages, and smooth muscle cells.^{84,85,100,101} Similar to scavenger receptors, RAGEs also are PRRs, in this case ones that recognize various proteins that are released by damaged cells (DAMPs). Activation of RAGE triggers the activation of PKC and NF $\kappa\beta$ as well as the expression of all the expected proinflammatory signaling molecules. Just to make things even more interesting, Amadori products can initiate lipid peroxidation of LDL apoproteins, the lipid fraction of LDLs, and the fatty acids of phospholipids in both membrane and LDL lipids.^{98,227} Thus, the formation of Amadori and Heyn's products can enhance mechanisms that directly contribute to the CHD process. In addition, the peroxidation of membrane phospholipids will enhance calcium leaking into the cell, which could then lead to additional calcium-mediated activation of cPLA₂ activity, further increasing proinflammatory eicosanoid signaling.

It is important to note here that the formation of glyoxal, CML, and CME also can occur as a result of peroxidation reactions with polyunsaturated fatty acids (in the absence of glucose) at rates comparable to, and in some cases greater, than those from glucose and that the formation of AGEs also occurs in foods as a result of cooking.^{87,99,101,223,224} This is relevant with respect to studies using AGEs as markers for diabetic risk because there are multiple and indistinguishable sources of AGE-producing reactions; they are not only derived from glucose or fructose. This does not mean that uncontrolled glucose is not a contributory factor to atherosclerosis through the production of AGEs but rather that the potential contributions of glucose levels per se must be considered in light of the fact that there are non-glucose-mediated reactions that result in the same mechanisms that contribute to the CHD process.

One last issue to discuss is that of carbohydrates and fructose. There is a lot of interest in dietary fructose as a major cause of obesity and chronic disease. This interest probably stems from the large increase in intake of HFCS since 1970 that coincides with a large increase in obesity over the same time frame. In order to help clarify some issues as they relate to CHD, a brief analysis of several of the important issues, as recently reported, is warranted.^{87,228} The predominant use of HFCS is as a sweetening agent in the same manner that sucrose is used as a sweetening agent; HFCS is used in manufactured foodstuffs (such as soft drinks) and rarely as a sweetener in the home. Use of HFCS as a commercial sweetener has steadily increased over the last five decades; simply because it is much less expensive to use than sucrose, the use of sucrose as a commercial sweetener has declined to an identical degree over the same time. This means that the contribution of fructose to total carbohydrate consumption relative to that of glucose has not changed at all; only the total amount has changed. From the recent 1999–2004 National Health and

Nutrition Examination Survey (NHANES) data in comparison with the 1977–1978 data, the intake of HFCS as a sweetener has increased by approximately 42%, which is essentially the same increase as that reported for total carbohydrate intake (41%) during the same period.

The most commonly used formulation of HFCS is 55% fructose and 45% glucose, a mixture that is essentially the same as that found in sucrose. Thus, the approximately 42% increase in fructose consumption is not due to the intake of pure fructose but rather the consumption of a formulation that is nearly indistinguishable from sucrose in metabolic terms. It makes no difference from where the fructose and the glucose come from, whether eaten in the form of sucrose or HFCS, because only the free forms are available in the blood for metabolism; therefore, the source has no effect. Because the liver metabolizes fructose very efficiently, serum levels of fructose tend to be much lower than those of glucose, with values for fructose in healthy subjects being approximately 8 µM. With over 500× more glucose in serum, it is certainly plausible that glucose might be the most important by far to deal with as a risk factor. However, this might not be the appropriate parameter for comparison. Fructose is far more likely than glucose to be in the open-chain configuration, and the subsequent rate of forming dicarbonyls is much greater. Overall, from in vivo data the formation of AGEs from fructose amounts to 10%–20% of the total AGEs. a pretty impressive amount considering that serum levels of fructose are far lower than those of glucose. Thus, the AGE-related risks for CHD from fructose are not trivial, but it should be kept in mind that they are still a small minority of the total risk when AGE formation from fructose is compared to glucose. One last consideration is that of the kinetics of Schiff base and Amadori product formation. With the normal postprandial fall in the concentration of glucose in serum (in those with normal insulin sensitivity), the decline to normal serum glucose within an hour or two of eating will automatically reduce the steady-state formation of Schiff bases to normal as well. Because the rearrangement process occurs over many weeks and months, and requires maintaining Schiff-base levels to do so, hourly and daily variations in serum glucose that ultimately return to normal will have minimal effect on enhancing the rate of AGE formation. On the other hand, a sustained elevation of serum glucose above normal over many months, regardless of the daily fluctuations due to eating behaviors, will enhance the formation of AGEs. Thus, based on the kinetics of AGE formation, an enhanced rate of AGE formation depends on serum levels of glucose that are constantly elevated above normal; transient changes due to the consumption of high-GI/-GL foods or meals would have minimal effect. In the final analysis, it is the lowest serum glucose levels that are sustained during our traditional nighttime fast that define the highest glucose (or any carbohydrate) levels that are relevant for AGE-related risk.

The ultimate concept of diabetes and AGE-related risks is, of course, insulin sensitivity and the resulting increase in serum glucose that follows increasing insulin resistance (fructose does not depend on insulin for absorption). From several recent reviews, there is a consistent association between consumption of sweetened soft drinks, obesity, and insulin resistance. However, these associations are commonly seen in population-based studies, whereas in clinical trials there is little evidence that sucrose or glucose consumption has a direct causal effect on insulin resistance

unless the carbohydrates are part of an *excess* calorie load.^{229–237} Supplementing diets with pure fructose does appear to increase risk for insulin resistance and enhance abdominal adiposity, with no evidence that dietary glycemic index is related to the development of insulin resistance.²³⁸ These results do indicate that consuming a high-fructose diet might confer additional risk. However, because consumption of HFCS-sweetened foods does not produce a higher dietary fructose consumption in comparison with sucrose-sweetened foods and because pure fructose is a rarely available commodity for public consumption, high-fructose diets such as those using fructose supplements might not be easy to generalize to the population. Thus, from currently available evidence it appears that with weight gain there is a relatively consistent association between increasing carbohydrate consumption and insulin resistance, leading to the conclusion that insulin resistance is more likely an issue directly associated with chronic weight gain than it is of carbohydrate consumption per se. This does not mean, however, that insulin sensitivity returns with weight loss.

This leads to one final issue. If carbohydrates per se are not responsible for the developing insulin resistance, then something else has to be. In many studies, the evidence indicates that inactivity is a major component of the problem.^{239–246} From a variety of detraining studies using normal daily activities to bed rest studies, inactivity may lead to insulin resistance within days. In addition, insulin resistance can be reversed through physical activity. This increase in insulin sensitivity with exercise training (both endurance and resistance training) also occurs in obese subjects, both with and without weight loss, and it can quickly return to an insensitive state on cessation of the activities.²⁴⁷⁻²⁵² Because physical activity can increase insulin sensitivity in the obese, it follows that obesity per se is not necessarily the cause of insulin resistance and that physical activity is extremely important in the ability to maintain glucose homeostasis. In light of the large increase in physical inactivity since the 1970s, it is possible that this is a major contributor to the current type 2 diabetes epidemic, whereas the increase in calorie consumption coupled with the inactivity since then has accelerated the development of the current obesity epidemic. With this model, inactivity leads to insulin resistance that then results in enhanced Amadori-/AGE-mediated inflammatory signaling, which contributes to risk for atherosclerosis; the state of being obese per se might then be a relatively passive bystander in the context of this process.

4.2.4.3 Metabolic Syndrome and Obesity

Metabolic syndrome and obesity are very tightly associated and are commonly thought to result from a combination of poor diet and lack of physical activity.^{253–255} Metabolic syndrome includes abdominal obesity, hypertension, insulin resistance, and dyslipidemia and is associated with increasing risk for many chronic diseases, including atherosclerosis. The major issue as a risk for CHD, however, is how these conditions enhance risk.

The increased risks associated with elevated lipids, hypertension, and insulin resistance have already been addressed in Sections 4.2.1, 4.2.4.1, and 4.2.4.2, which leaves abdominal obesity (and obesity in general) as the major issue to briefly review. There is a highly complex interplay of many factors that ultimately lead to the development of obesity and subsequent risk for chronic diseases, with much of the risk being associated with insulin resistance and the resulting increase in glycation events. In addition, obesity is highly associated with a *proinflammatory state* in which proinflammatory cytokines, along with other adipokines, originate from the adipose tissue itself.^{256–261} The adipokines include adiponectin, leptin, and visfatin as well as the inflammation-associated TNF- α , IL-6, IL-1 β , and MCP-1. Whereas some of these adipokines are synthesized by stromal cells and macrophages that are resident in the adipose tissue, many are expressed by adipocytes as well. As discussed in Chapter 2, the chemokine MCP-1 enhances macrophage infiltration into the adipose tissue, which then adds to the local production of proinflammatory cytokines. The entry of these proinflammatory cytokines derived from the adipose tissue into the circulation then exacerbates any localized inflammatory responses elsewhere, including those associated with the development of atherosclerotic plaque. Thus, inflammatory cytokines that originate in the adipose tissue may comprise the major source of obesity-associated risks.

The development of the proinflammatory state is initiated as a result of the demand for increased adipocyte hypertrophy and for adipogenesis from preadipocyte fibroblasts during periods of net caloric excess.^{262–268} In response to excess fatty acid uptake, adipocytes increase the rates of triglyceride synthesis, a process that also requires increased GLUT4-mediated glucose uptake for the necessary production of glycerol-3-phosphate. With prolonged periods of net caloric excess, endoplasmic reticulum dysfunction or endoplasmic reticulum stress occurs. This condition is associated with activation of JNK, increased production of TNF- α , and release of free fatty acids (FFAs) by adipocytes in addition to the hallmark accumulation of improperly folded and newly synthesized proteins that have been unfolded. As a result of autocrine activation of TNF receptors, an array of proinflammatory cytokines are released, leading to the recruitment and activation of preadipocytes (fibroblast lineage) and monocytes (which differentiate into macrophages), as well as triggering the shift of resident macrophages from an anti-inflammatory phenotype to a proinflammatory phenotype, thus greatly enhancing the production of proinflammatory cytokines with the concomitant production of PGs and other proinflammatory lipid mediators. Chronic release of TNF- α and FFAs to the circulation leads to mitochondrial dysfunction and insulin resistance in both adipocytes and skeletal muscle. Increased insulin resistance in adipocytes contributes to their compromised ability to synthesize additional triglycerides (TGs), more than likely enhancing endoplasmic reticulum dysfunction and leading to increased fatty acid levels for inflammatory signaling. PGs are a necessary component for activating the differentiation of preadipocytes into functional adipocytes, in part, through their interaction with PPARy2.^{256,264,269} Thus, it appears that prolonged TG synthesis activities resulting from caloric overload lead to hypertrophy and hyperplasia of adipocytes as well as insulin resistance, all of which are associated with the production of proinflammatory signaling molecules. In addition to these, the processes of angiogenesis are initiated, possibly in association with the localized development of hypoxia due to the adipocyte hyperplasia. As a result, additional cytokines and growth factors that are necessary for angiogenesis are produced (see Chapter 2 for details), which can then enter the circulation to affect other tissues as well. Although the order of events adipocyte hypertrophy \rightarrow endoplasmic reticulum dysfunction \rightarrow inflammatory signaling \rightarrow insulin resistance \rightarrow adipocyte hyperplasia \rightarrow adipose angiogenesis has been observed in some animal models of obesity-associated insulin resistance, it has not been confirmed in humans, but both hypertrophy and hyperplasia have been observed in subcutaneous fat. Abdominal fat, though, is more proinflammatory than subcutaneous fat with a preferential infiltration by macrophages and is known to be a more substantial risk associated with CHD as well.^{270–272}

Interestingly, it is possible to be both obese and display a normal metabolic profile, referred to as metabolically normal obese (MNO) with approximately 11%-36% of obese individuals (depending on the study) displaying this, a phenotype that includes normal insulin sensitivity.^{273–278} Paradoxically, the opposite is true as well with a similar prevalence of an obesity-associated metabolic phenotype observed in lean people.²⁷⁶ In addition to normal insulin sensitivity, these MNO individuals display normal levels of inflammatory markers in both serum and tissue biopsies as well as evidence of no additional risk for cardiovascular disease or cancer. As a relatively new area of study, very little is known about all the lifestyle variables of these MNO individuals, although in one study the lack of additional risk for CHD was directly attributable to enhanced cardiorespiratory fitness, which is suggestive of the importance of physical activity to this phenotype.²⁷³ On the basis of these studies, in addition to those documenting a return to insulin sensitivity with exercise in the obese (without weight loss), it is apparent that the lack of any physical activity is a primary driver for the increased risk for disease in the obese rather than obesity per se. Clearly, more studies in the obese that include an analysis of physical activity variables along with appropriate biomarkers of effect are needed. It is also clear that inadequate nutrition plays an important role in exacerbating inflammatory disease; therefore, nutrient status and a variety of dietary variables also need to be included. From the molecular and clinical data, it is apparent that gaining weight coupled with remaining inactive is a major cause of excess risk for CHD and simply being overweight while all other risk factors are normal is not.

4.2.4.4 Infection

As alluded to in Section 4.1, inflammatory processes that follow viral and bacterial infections may enhance the inflammatory mechanisms that are integral to the CHD process. In a variety of population-based studies, chronic infections with a variety of organisms including herpesvirus (cytomegalovirus), hepatitis A, *Chlamydia pneumonia, Porphyromonas gingivalis,* and *Helicobacter pylori* have been documented to be associated with risk for atherosclerosis independently of other risk factors.^{75,76,90–97,279} In the majority of cases, it appears that chronic infections of tissues that are distant from the coronary or carotid arteries (e.g., pulmonary infections and hepatitis) are the ones that produce the risk and the evidence for risk is often elevated serum cytokines or CRP. This indicates that it is circulating inflammatory cytokines that are most likely the predominant component of the increased risk; the additional proinflammatory signals simply add to the locally produced proinflammatory effects of the intermittent low shear forces. The difficulty with such studies is that circulating proinflammatory markers can have origins that are completely independent of an infection (see Section 4.2.4.3).

In addition to circulating cytokines that are associated with infections, a localized infection within the artery wall appears to occur with a variety of pathogens being observed in atherosclerotic plaque.²⁸⁰⁻²⁸⁵ In many cases, the identified pathogens are the same as those that cause chronic infections elsewhere, with organisms that cause pulmonary and periodontal infections appearing to be more common than others. It is of interest that in the clinical studies that analyze the specific associations between the content of pathogens in plaque and the severity of the disease either small associations or no associations are actually observed. This does not mean that a local infection with pathogens cannot initiate additional inflammatory responses to increase risk for CHD, but it does raise the possibility that the presence of pathogens in plaque is more likely coincidental with total pathogen burden. Further, in support of the passive association concept, a recent meta-analysis of 13 different clinical trials of antibiotic treatment for secondary prevention of CHD revealed no effect and in another recent study the associations with atherosclerosis and infection were dependent on smoking status and not infection alone.^{279,286} With a majority of lesions being positive for pathogen DNA, it is safe to conclude that CHD risks may be influenced by such pathogen exposures. On the other hand, if infections were a major risk for atherosclerosis then one would expect chronic antibiotic therapy to attenuate the risk. Thus, although circulating proinflammatory cytokines can certainly contribute to CHD risk, those arising from infections are likely to be a relatively minor component in comparison with the other sources of risk. Prevention strategies for atherosclerosis that relate to infection are therefore limited to the standard advice for proper personal hygiene, proper cooking and handling of foods, and appropriate precautions to avoid infectious diseases.

4.2.4.5 Cigarette Smoking

Smoking has been known for many years to be a significant risk factor for CHD, and quitting smoking significantly reduces risk for CHD.^{287–290} However, because smoking cessation is not part of a dietary or exercise approach to reducing risk for atherosclerosis, smoking cessation is not discussed here. Smoking is included here because a brief discussion on possible mechanisms of risk enhancement by smoking is quite revealing.

Formation of platelet thrombi is acutely enhanced by smoking, an effect that occurs within a few minutes of smoking and occurs with both mainstream and sidestream (environmental) smoke.^{291,292} The activation process is associated with the prooxidant effects of many of the components in cigarette smoke and can be inhibited with dietary vitamin C.^{291,293–296} These prooxidant components also cause damage to endothelial cells and can oxidize LDL and LDL-CHOL. Thus, the prooxidant and subsequent proinflammatory effects of cigarette smoke appear to be the major factor in the smoking-enhanced risk for CHD and the predominant reason why risks for CHD decline very quickly on cessation of smoking. These illustrate that any increase in exposure to prooxidant molecules, either exogenous or endogenous, will enhance risk for CHD. In addition, the ability to handle excess prooxidants from any source would be very important in reducing risk; the ability of vitamin C supplements to reduce much of the smoking-associated risk attests to this.

4.2.4.6 Nutritional Status/Redox Control

Several important issues relating to nutrient status are highly relevant to the proinflammatory stress responses that are causal. It has been estimated that only 10% of Americans eat the recommended minimum number of servings of fruits and vegetables.⁶⁹ Intakes for vitamins C, D, and E are below the estimated average requirement (EAR) for 25%, 70%, and 60% of Americans, respectively, and below the EAR for zinc (~8%), iron, copper, and selenium (~5% to 6%), as well as calcium (38%) and magnesium (45%).⁷¹ Some of these specific vitamins are antioxidants, while other vitamins and minerals are components of antioxidant enzymes or redox sensor proteins or directly involved in calcium homeostasis. This apparent large prevalence of insufficient intake of these particular nutrients in the United States directly enhances risk for CHD through a less than optimal ability to control antioxidant and redox statuses of cells.

As discussed in Section 4.2.3 and in detail in Chapter 2, the initiation of an inflammatory stress response can be mediated through the activation of both PLA₂ and the various MAPK pathways, leading to the downstream activation of NFκβ and the ultimate production of proinflammatory signaling. The regulation of these signal transduction pathways is highly dependent on the redox status of the cell. Oxidizing agents can modify the activity of signal transduction pathways by oxidizing redox sensor proteins such as thioredoxin (Trx) and peroxiredoxin (Prx).²⁹⁷⁻³⁰⁴ For example, activation of Jun/Fos (AP-1) can occur through a Trx-mediated activation of apoptosis signal-regulating kinase 1, which then activates the JNK-MAPK and p38-MAPK pathways and subsequently Jun, Fos, and a variety of other transcription factors.^{299,304–306} The antioxidant enzymes SOD, CAT, and GPX alter the local concentrations of ROSs and therefore contribute to the optimal regulation of signal transduction pathways in the cell.²⁹⁷⁻³⁰⁶ Any nutritional insufficiency that contributes to a nonoptimal antioxidant function can then compromise redox control of the signal transduction pathways and lead to enhanced inflammatory signaling. With such a large majority of Americans eating insufficient amounts of the aforementioned nutrients, it is clear that redox control and antioxidant status of cells may be compromised and may become a significant source of risk. Perhaps, this is the place to suggest that LDL-CHOL (and HDL cholesterol [HDL-CHOL] as well) is actually a component of an extracellular redox sensor network.

The role of the cellular redox sensing proteins as well as the antioxidant enzymes in regulating signal transduction pathways is an important component of the cellular stress response. As discussed in Chapter 2, increasing ROS stress within a cell may lead to damage and orchestrating a stress response that can ultimately lead to increased repair activities is an important survival mechanism. Using agents of damage (ROSs and other oxidants) as agents to activate a stress response is both logical and highly efficient; it takes time for the processes of repair to be increased, so the earlier in the damaging process that they are initiated, the more likely the cell will survive. The various redox sensor proteins within cells provide such a mechanism. In the circulation, however, it may be LDL-CHOL that provides the same redox sensor function. It is the ability of CHOL-OX (and oxidized LDL/HDL as well) to bind to scavenger receptors of macrophages, endothelial cells, and smooth muscle cells and initiate proinflammatory signaling that would be the mechanism through which this circulatory redox sensing system might work. In an ideal situation, such a (low-level) systemic response might be additive to the local cellular stress responses and thereby might enhance the local tissue responses that ultimately lead to repair. Such a system would simply prime all tissues for a local stress response when circulating oxidants are present and perhaps lead to a heightened stress and subsequent repair response in those tissues that suffer actual damage; lung and liver in the case of environmental exposures, for example. Unfortunately, such a system could have negative implications for those very few tissues that are under chronic proinflammatory stimuli, tissues such as the inner curves and sharp bends of arteries. The negative effects may even become fatal when they are seriously exacerbated by poor cellular redox control coupled with enhanced physical stress and chemical damage due to damaging environmental exposures, poor diet, and improper physical activity. The concept that LDL/HDL-CHOL may be a component of a circulatory redox sensor system remains to be tested.

4.3 ETIOLOGY OF PREVENTION

Managing risk for CHD through attempting to modify its major risk factors has been a common theme of CHD prevention over the last five decades. Declines in mortality due to CHD over the same time period has been due predominantly to pharmaceutical control of hypertension, diabetes, and the dyslipidemias⁶⁻⁹ and to improved treatment of existing CHD patients,¹⁰⁻¹³ with a lower prevalence of smoking also being an important contributor. Although the decline in mortality is important, it has done little to reduce the actual impact of the disease and CHD remains the number one cause of death in the United States.¹⁻⁴ A major reason for this is that primary prevention has not focused on the actual mechanisms of cause. To be consistent with the general theme of this book, if you do not have an effect on mechanisms of cause you will not change the development of disease. With an additional focus on attenuating the mechanisms of cause, rather than a sole focus on factors associated with risk, substantially better preventive effects should be realized. This does not mean that attention to risk factors is not important but rather that prevention should focus primarily on the actual mechanisms of cause with a secondary focus on associated factors. With inflammatory mechanisms being recognized as the primary cause of CHD, the target for a preventive focus is clear.

4.3.1 Exercise

There are several basic factors that may contribute to an exercise-based risk reduction of atherosclerosis: exercise can be used as a calorie burner to modify adiposity, reduce blood pressure, prevent insulin resistance, enhance antioxidant status in order to reduce oxidative damage to lipids, attenuate inflammatory signaling, and alter serum lipids. Of these, attenuating proinflammatory signaling, reducing ROS-mediated oxidation of LDL/HDL/CHOL, and preventing insulin sensitivity may provide the greatest benefits overall as they appear to address the greater of the population-wide risks. Although currently a common component of clinical recommendations, altering serum lipids through exercise may not be a major contributor to risk reduction for most. This is discussed first.

4.3.1.1 Serum Lipids

A number of population and cross-sectional studies relating to exercise and serum lipids reveal that physically active individuals tend to have a more favorable lipid profile.^{307–315} However, difficulties in controlling for individual differences in diet and nutritional status, body composition, and the amount/intensity of exercise make definitive conclusions of a consistent association between increased physical activity and normalization of serum lipids difficult. In spite of these limitations, the cross-sectional data reveal that there may be a consistent effect of reducing serum TG by 8–20 mg/dL when physical activity approximates an additional expenditure of 1500–2000 kcal/week (equivalent to ~15–20 mi.), with an additional decline of up to 8 mg/dL with an additional approximately 10 mi./week.^{316–320} Such activity levels can be attained by most otherwise healthy people. In addition, although such data might be informative of exercise on these parameters in individuals and therefore cannot be the sole basis of decisions regarding the preventive benefits of exercise. Clinical trials would provide much more meaningful information.

As a means to reduce risks associated with serum lipids, a variety of exercise protocols have been utilized to induce favorable changes. For example, a normalization of moderately elevated TG levels and VLDL patterns have been observed after a variety of daily walking or jogging programs.^{321,322} Similar reductions in total-CHOL and LDL-CHOL of 4%-20% have been observed with a variety of exercise protocols that range from 30 minutes of cycling at moderate intensity three times a week to continuous running of up to 45 minutes for 5 days/week.^{319,322-328} Although such changes in lipid-based risk appear to be beneficial, a serious drawback of many exercise studies is a lack of proper controls. When randomized controls are used the exercise effects tend to be much less in comparison with changes observed in controls, and in many cases the differences are nonsignificant.^{329–336} From these and a variety of other studies, moderate reductions of 4%-7% in total-CHOL and LDL-CHOL in both men and women have been observed in approximately 25% of the studies with the endurance exercise-mediated effect being observed predominantly in subjects who were initially inactive.³¹⁹ Thus, as a general concept, moderate physical activity has a small effect on serum cholesterol, whereas the effect of moderate physical activity on TG levels are often greater, possibly stemming from the metabolic demands of the activity. There is little evidence for consistent favorable reductions in total-CHOL or LDL-CHOL at these levels of physical activity.

Because serum lipids are favorably changed by either a relatively small amount (<10%) or not at all, altering this source of risk cannot be the major reason for the general and consistent decline in risk for CHD that is observed as a result of repeated exercise. Perhaps, this is one reason for some to not promote exercise as a preventive measure against CHD; exercise does not consistently reduce LDL-CHOL. Resistance and endurance exercise can, however, enhance insulin sensitivity in both normal and overweight/obese individuals with insulin resistance. This is an important protective effect for CHD that is independent of serum lipids and

one that has very often been overlooked in population studies on exercise, lipids, and CHD. Possibly, another reason why exercise may be discounted as an effective means to prevent CHD may be that the protective benefits of exercise are simply not well known.

4.3.1.2 Antioxidant/Redox Status

Another preventive effect of exercise that is independent of serum lipids relates to several antioxidant enzymes, an area of considerable importance to cellular redox control and its role in ensuring a controlled inflammatory stress response. Exercise enhances eSOD, cSOD, and GPX levels in skeletal muscle as well as in cardiac muscle, aorta, and coronary arteries, an effect that appears to be mediated through Nrf2/ARE signaling and that disappears soon after the cessation of exercise.337-343 Enhanced GPX and cSOD would enhance control of the cellular redox state and, at the same time, an induction of eSOD would ensure that extracellular superoxide anions will be reduced to attenuate the oxidation of extracellular LDL and LDL-CHOL. These exercise effects would be in addition to the *adaptation* responses to chronic high shear stress discussed in Section 4.2.2, and they occur in all tissues examined, including liver, lung, and brain.^{297,344-349} Although a systematic examination of the effects of exercise on all the redox regulatory systems in the arterial vasculature has not been conducted, the results of such studies might be highly instructive. Because CAT, glutathione reductase, Trx, Prx, HO-1, and glutamate cysteine ligase (GCL) are induced through Nrf2/ARE activation, positive results would be expected.350-353

The benefits of an increased antioxidant status would be a reduction in oxidative damage to both cellular components and HDL/LDL/CHOL. As a result, inflammatory signaling resulting from oxidative damage to cells or through the activation of scavenger receptors by oxidized HDL/LDL/CHOL will be reduced. As might be expected, the degree of LDL oxidation and the ability to activate macrophages are highly associated with the antioxidant status of the tissues, and treatment with dietary components that are known to activate Nrf2 also reduce the cell-mediated oxidation of LDL/CHOL.354-357 Exercise interventions, especially when combined with dietary modifications, also are known to enhance antioxidant and redox enzymes, reduce LDL oxidation and other markers of oxidative stress, and reduce risk for CHD.^{358–364} It is interesting to note that in studies that focus on antioxidant enzyme activities significant enhancements of activity occur following 60 minutes or more of moderate-intensity exercise and also following 20-30 minutes of continuous high-intensity exercise.^{340,365-367} Although activation status of Nrf2 is not determined in the (human) intervention studies, it is tantalizing to hypothesize that this is the mechanism that mediates, at least in part, the positive effects.

4.3.1.3 Inflammatory Signaling

In addition to optimizing control of cellular redox states, the other major protective effect of exercise is on attenuating inflammatory signaling. The sustained production of the proinflammatory cytokines IL-1 β , IL-6, IL-8, and TNF- α and CRP, as well as an increase in creatine kinase activity in serum (a common marker for muscle damage) and markers of ROSs in response to damaging or exhaustive exercise, has

been well known for many years.^{368–372} Unfortunately, such data have perpetuated the concept that exercise is a source of damaging ROSs and inflammatory signaling. With nonexhaustive exercise that is continuous for 1 hour (or more), however, the increase in IL-1 and TNF- α does not occur; at the same time, IL-6, IL-10, soluble IL-1 receptor antagonist (sIL-1ra), soluble IL-1 receptor (sIL-1r), and soluble TNF receptor (sTNFr) increase to a much greater degree. The IL-6 increase can be as much as 100-fold, and it is the large initial rise in IL-6 that then induces an increase in the other anti-inflammatory molecules, all of which return to baseline several hours after the stop of exercise.³⁷¹⁻³⁷⁶ Even at 50% of the maximum watts of power output (cycling, an approximate 6 MET level of intensity), significant increases in IL-6 are observed with 1 hour of cycling exercise. It is also important to note that changes in IL-6 are not observed with exercise that utilizes small muscle groups, such as the upper arms, indicating that large-muscle-group activities of a minimum intensity of approximately 6 METs that are sustained for approximately 1 hour are essential for these anti-inflammatory effects of exercise.^{376–378} This does not mean that higher intensity activity carried out for less than 60 minutes will not produce anti-inflammatory cytokine responses; it just remains to be seen in properly conducted experiments.

The importance of the differences between exhaustive and nonexhaustive exercise cannot be overemphasized. The large induction of IL-6 without a concomitant production of TNF- α is the major reason for the anti-inflammatory effects of prolonged exercise. The large increases in sIL-1r and sTNFr exert an anti-inflammatory effect by binding to circulating IL-1 α , IL-1 β , and TNF- α and the sIL-1ra prevents circulating IL-1 α and IL-1 β from binding to their cellular receptors, thus suppressing cellular responses to circulating proinflammatory cytokines. These effects will reduce the proatherosclerotic effects of circulating proinflammatory cytokines that originate from inflamed tissues or from abdominal adipocytes. It may also be possible for these soluble receptors and receptor antagonists to attenuate autocrine and paracrine signaling within inflamed tissues. Because TNF- α is involved in producing insulin resistance, the development of insulin resistance also might be blunted as a result of the enhanced sTNFr.

IL-10 exerts its anti-inflammatory effects through suppressing the production of the proinflammatory cytokines. IL-10 activates the Jak–STAT pathway along with prolonged STAT3 activation to result in the synthesis of suppressor of cytokine synthesis proteins, inhibiting the synthesis of IL-1 α , IL-1 β , IL-6, IL-8, and TNF- α .^{379–381} IL-10 also suppresses MAPK and NF $\kappa\beta$ signaling, which in turn attenuates the synthesis of acute-phase cytokines.^{379–382} Thus, there is a profound anti-inflammatory effect for moderate-intensity endurance-type exercise, which reduces inflammatory mechanisms at play within the arterial vessel walls. The results of these exercise-induced anti-inflammatory effects are also evident in CHD patients and high-risk individuals where reductions in CRP and enhanced eNOS activity are observed.^{383–386}

The effects of exercise on CRP are instructive because they tend to equate well with those on IL-6 induction. Associations between exercise and reductions in CRP have been observed in several clinical trials, whereas in others no associations were evident.^{387–396} Of these, the longer duration exercise periods of 45–60 min/day were more consistently associated with a chronic reduction in CRP; in studies that used

durations of exercise of 30 minutes, levels of CRP were unchanged. These results are similar to an earlier observed dose effect between exercise and risk for elevated CRP with odds ratios of 0.98, 0.85, and 0.53 for light, moderate, and vigorous exercise (NHANES III).³⁹⁷ Interestingly, in these studies the weight-loss effects of the longer duration exercise were sufficiently consistent that it was difficult to disassociate the effects of weight loss per se from those of the exercise. Because the exercise induction of IL-6 expression is related to lowered glycogen content, a lower glycogen level that might be associated with losing weight may actually be additive to the exercise effect and therefore be contributory, not confounding.

The anti-inflammatory effects of prolonged exercise are important because they help to explain the previously discussed data in which subjects with high levels of fitness (as measured by VO_2 max) have a lower risk for CHD compared to active people with lower levels of fitness. Lower intensity and duration activity would still contribute to lower risks through enhancing insulin sensitivity and increasing the efficiency of redox control. Only with longer duration activities that greatly enhance IL-6 levels do subsequent anti-inflammatory benefits accrue. Hence, this is the type of activity that is recommended for an optimal reduction in risk for CHD.

In summary, exercise can exert a variety of effects that greatly reduce risk for atherosclerosis by directly affecting the inflammatory mechanisms of cause. These effects are in addition to those mediated by increasing insulin sensitivity and enhancing cellular redox control. However, these effects are transient and the inductive mechanisms last only as long as the exercise-induced levels of regulatory molecules and cytokines remain elevated and the benefits of the induced enzymes remain only as long as the proteins exist. With half-lives from approximately 10 minutes for eSOD to less than 1 or 2 days for GPX and Trx reductase, the protective effects of enzyme induction do not last very long.^{398–401} The exercise must therefore be repeated nearly daily if these cytoprotective and anti-inflammatory effects are to be realized on a long-term basis.

4.3.1.4 Insulin Sensitivity

As described in Section 4.2.4.2, lack of physical activity is associated with the development of insulin resistance, possibly because of a proinflammatory environment in muscle cells produced by inactivity.^{239-246,402} An increase in local production of ROSs due to inactivity mediates a stress-mediated induction of TNF-a synthesis that in turn enhances phosphorylation of insulin receptor substrate (IRS)-1 on lysine to enhance its degradation as well as impair its ability to complex with p85, thus reducing insulin signaling. Because of the known effects of increased serum glucose on the development of AGE and subsequent inflammatory signaling, the increased insulin resistance resulting from inactivity contributes to risk for atherosclerosis. The obvious way to reduce risk would then be to increase levels of activity. Most interestingly, the insulin resistance appears to be reversed through almost any kind of physical activity. Both endurance training and resistance training enhance insulin sensitivity, and the effect of an acute exercise bout appears to last approximately 24 hours. In addition, enhanced insulin sensitivity occurs even in obese subjects whether they lose weight or not, although it returns very quickly to an insensitive state on cessation of the activity.247-252,403-406 Further, where

measured, decreases in glycosylated hemoglobin were also observed in association with the increased insulin sensitivity; in some cases, the amount of decrease was correlated with the intensity of the exercise training and was greater with resistance training plus endurance training in comparison with either one alone. The intensity issue, however, appeared to be associated with 3-day/week programs and not those that utilized daily activity. Possibly, this means that alternate-day activities need to be of greater intensity and volume compared to daily activities. These are important results because they indicate that risks for atherosclerosis that arise from insulin resistance can be prevented with most any kind of regular physical activity and that markers of damage from elevated glucose can be reduced. They also reveal that these effects are temporary and, therefore, the higher intensity activity must be repeated at least on an alternate-day schedule to ensure a relatively constant degree of insulin sensitivity.

There are two main issues with respect to the effects of exercise on glucose transport. The first is an enhanced glucose transport due to mechanisms relating to muscle contraction during an acute bout of exercise and an increased sensitivity to insulin that follows for a short time after the activity. As a result of muscle contraction, two events occur with respect to glucose transport, via insulin-independent mechanisms.⁴⁰⁷⁻⁴¹⁰ Translocation of the GLUT4 transporter to the membrane to activate glucose transport in skeletal muscle is dependent on Rab-GTP. The inhibitor protein AS160 is a Rab-GTPase, which hydrolyzes Rab-GTP to Rab-GDP to prevent GLUT4 translocation. Another putative Rab-GTPase is TBC1D1; although both can be phosphorylated by AMP-associated protein kinase (AMP-PK), AS160 has multiple sites for phosphorylation by Akt, rendering it the protein through which insulin activates glucose transport (insulin \rightarrow IR/IRS \rightarrow $PI3K \rightarrow PI3,4,5$ -TP:PDK1 \rightarrow Akt \rightarrow AS160 pathway). Interestingly, both inhibitor proteins also have a calmodulin-binding site, which would make both of them sensitive to calcium-mediated effects. Because the AMP to ATP ratio increases with increasing intensity of calcium-activated muscle contraction, it is an attractive model that muscle contraction can inactivate AS160 and TBC1D1 via calmodulin and AMP-PK-mediated mechanisms. Increased phosphorylation of both occurs during muscle contraction, but only AS160 remains phosphorylated for several hours after the exercise. Thus, both proteins appear to be involved in an exercisestimulated glucose transport, although only the AS160 protein would contribute to a sustained insulin sensitivity following the exercise period because it remains phosphorylated, in some cases, for 3-17 hours post exercise.

The second issue relates to an increase in capacity to respond to insulin with repeated exercise. In many studies, both resistance training and endurance training result in an increase in GLUT4 content, whereas an insulin-stimulated PI3K:IRS activity or increase in IRS-1 content has been inconsistently reported.^{411–416} Increases in the content of IRS may increase PI3K-mediated signaling by increasing the efficiency of IRS binding to IR to activate PI3K at a given insulin concentration. Increasing GLUT4 content would also increase insulin sensitivity by enhancing the amount of GLUT4 translocation at a given concentration of insulin. The mechanism of the reported increases in expression of these proteins has been associated with an increased activation and expression of PGC-1 α , and it is known that downregulation
of PGC-1a along with a decrease in PI3K activity is observed in skeletal muscles of subjects with type 2 diabetes.^{411,414,416,417} PGC-1α is known to be involved in the increased mitochondrial biogenesis that is observed following exercise training, and it is activated to enter the nucleus and bind to the promoter regions of several genes following activation by either p38-MAPK, silent information regulator (SIR) T1, SIRT3, or AMP-PK. Of these, both AMP-PK and SIRT1/2 are activated when NAD+ to NADH ratios and AMP to ATP ratios increase in tandem. AMP-PK phosphorylates PGC-1a to activate it, whereas SIRT1 deacetylates it. In addition, AMP-PK can activate p38-MAPK, which in turn can activate PGC-1a. Thus, PGC-1a activation is exquisitely sensitive to both metabolic stress and any other stressors that activate p38-MAPK. This transcription factor was originally described as a coactivator for the nuclear receptor PPAR γ coactivator-1 α and is known to act as a coactivator (along with other transcription factors) for a variety of proteins, including GLUT4 and PGC-1a. It is not known, however, to be involved in the expression of IRS-1. Interestingly, the promoter of IRS-1 contains multiple binding sites for the estrogen receptor (ER) and skeletal muscles contain both ER- α and ER- β ; in addition, PGC-1 α can act as a coactivator of ER-mediated transcription.⁴¹⁸⁻⁴²² Further, phosphorylation of ERs by a variety of exercise-activated kinases, including PKC, ERK-MAPK, p38-MAPK, and Akt, enhances the binding of ER to its coactivators to enhance ER-mediated expression.^{423–425} Thus, it may be possible that a coordinated activation of PGC-1 α and ER will also enhance the expression of IRS-1 (among many other proteins) and that inconsistencies in enhanced IRS-1 expression might be explained in part by differences in age and ER and PCG-1 α expression in subjects in different studies.

Although an increased expression of GLUT4 and other proteins involved in the insulin signal transduction pathway is a topic requiring intense investigation and there is still much to be learned, it does appear to be involved, at least in part, in the increase in insulin sensitivity observed with exercise training. However, this does not rule out that an increase in AS160 phosphorylation from an acute bout of exercise is not involved. As long as the acute periods of exercise are repeated multiple times each week, the temporary increase in insulin sensitivity due to this mechanism (lasting for as long as 17 hours in one study) would surely be additive to any longer term effects due to the increased synthesis of the proteins involved in signal transduction and glucose transport. The important message is that only with repeated exercise, resistance, endurance, or a combination of both, are these changes seen. The risk-reduction effects of exercise are summarized and illustrated in Figure 4.2.

4.3.2 DIET

As with exercise, there are several basic concepts in the role of diet in reducing risk for CHD: diet as a means to reduce hypertension, control adipose tissue, maintain insulin sensitivity and maintain proper nutrient status, alter dyslipidemias, and attenuate inflammatory signaling. These risk-reduction effects of diet are summarized and illustrated in Figure 4.3. Dietary issues relating to nutrient status and inflammatory signaling are extensively discussed in Chapter 2 (Section 2.5.1); but they are summarized here, as are the issues relating to antioxidant status/redox control and serum lipids.



FIGURE 4.2 Reductions in mechanisms of risk for atherosclerosis by exercise: exercise stimulates the release of insulin-like growth factor (IGF)-1 from muscle and activates various mechanoreceptors, leading to the transient activation of phosphatidylinositol 3-kinase (PI3K), PLC, AMP-associated protein kinase (AMP-PK), and the mitogen-activated protein kinases (MAPKs) as well as transient increases in cellular hydrogen peroxide. Subsequent activation of Nrf2, p50/p65, PGC-1 α , estrogen receptor, and many other transcription factors enhances expression of superoxide dismutase (SOD), catalase (CAT), peroxiredoxin (Prx), thioredoxin (Trx), glutathione peroxidase (GPX), GSR, GLUT4, and IRS1. Longer term exercise increases the release of interleukin (IL)-6 from muscle, activating the expression of soluble tumor necrosis factor receptor (sTNFr), soluble IL-1 receptor (sIL-1r), soluble IL-1 receptor antagonist (sIL-1ra) and suppressor of cytokine synthesis (SOCS) via IL-10–STAT3.

4.3.2.1 Nutrient Status

A relatively large minority (and in some cases a majority) of Americans consume insufficient amounts of individual nutrients, including nutrients that are important antioxidants or components of redox enzymes and coenzymes.⁶⁹⁻⁷¹ This is likely due to the fact that over 90% of Americans consume less than the minimum recommended number of servings of both fruits and vegetables, a factor in the commonly observed associations between low fruit/vegetable intake and risk for chronic disease in population-based studies. In the past, a frequent interpretation of these associations was that low fruit intake meant low antioxidant intake and that antioxidant supplements would then be useful to reduce risk. Unfortunately, in clinical trials no clear protective benefit of antioxidant supplements has been observed and in many instances negative effects such as increased deaths due to lung cancer with β-carotene or with vitamin C or E supplements and increased stomach cancer with α -tocopherol and/or β -carotene have been observed.^{426–433} A more likely interpretation of these population associations is that different degrees of nutritional deficiencies are responsible for the associations and not antioxidant deficiencies. In addition to the lack of a protective effect on human disease, antioxidant supplements appear to have little to no positive effect on athletic performance either and, in some studies, have been observed to prevent exercise-induced mitochondrial biogenesis and reduce both aerobic performance and maximum muscle force.434-440

Using antioxidant supplements may overwhelm cells and potentially interfere with the regulation of redox states at the microenvironment level. As discussed in detail in Chapter 2, the metabolic regulation of ROSs through the production of O_2 • and H_2O_2 at specific cellular locations, as well as the location-specific expression of antioxidant enzymes, plays a vital role in a wide array of normal cellular functions.⁴⁴¹ Severely reducing endogenous ROSs through supplementation with antioxidant compounds is therefore contraindicated.

For these reasons, the starting point for a dietary approach to reducing risk for CHD is a diet that provides recommended daily allowance (RDA)/adequate intake (AI) amounts of all the necessary nutrients rather than one that includes supplements. For convenience, the new *MyPlate* approach (www.choosemyplate.gov), which includes standardized serving sizes, provides an appropriate model from which to start; however, it must be modified to meet the additional anti-inflammatory effects that are necessary for optimal CHD prevention (see Section 4.4.2).

4.3.2.2 Serum Lipids

Modifying diet to alter the parameters that are associated with serum cholesterol and subsequent risk for CHD has been a common theme in CHD prevention over the years, and various beneficial changes have been observed in both patients and the nonpatient general population. Using diets that are characterized as low in fat, saturated fat, and cholesterol (typically <20% calories as fat, <200 mg cholesterol, with >50% unsaturated fats) while emphasizing increased consumption of fruits and vegetables have induced changes in serum lipids that are similar to those observed with lovastatin.^{442–445} Approximately 30% reductions in total-CHOL and LDL-CHOL along with smaller reductions in TG were seen with both statins and dietary changes in the studies from Jenkins' group, whereas smaller changes due to diet were observed in the DASH and OSLO studies. Strict dietary studies such as these and others have their drawbacks when it comes to dietary recommendations for fat and cholesterol content per se; in some cases they were too small to detect significant effects on mortality, and in others no reductions in mortality due to CHD were actually seen.^{446–448} Although it is possible that the larger studies simply were not carried out for a sufficiently long time to observe a significant effect, the modest changes in serum lipids that have been observed with these dietary studies coupled with the general lack of effect on CHD mortality indicate that, changing serum lipids without addressing mechanisms of inflammation does not significantly alter mortality.

While changing total fat and cholesterol through diet alone appears to have little effect on risk, addressing multiple risks at once does appear to provide preventive benefits. In the Stanford Coronary Risk Intervention Project, low-fat, low-cholesterol diets were prescribed along with an endurance exercise training program. Significant reductions were observed for LDL-CHOL, TG, and body weight and exercise capacity was enhanced while achieving a 47% reduction in the rate of narrowing of diseased coronary artery segments.449 Similar results have been seen with the Lifestyle Heart Trial where not only did the experimental groups exhibit a relative reduction in the diameter of stenoses of 7.9% compared with a 27.7% increase in diameter in the control group but also was the RR for a cardiac event 2.7 in the controls.^{450,451} The Pritikin residential lifestyle intervention utilizes a high-fiber, low-fat diet with high fruit and vegetable consumption and 45–60 minutes of aerobic exercise with additional flexibility and resistance exercises. Results from this program are similar, with experimental groups reducing total-CHOL, LDL-CHOL, and TG by approximately 25% along with reductions in body weight as well as in the requirement for coronary bypass surgery as treatment for CHD.^{452,453} From these results, it is difficult to determine how much of the effects were due to the changes in diet and how much were due to the extensive exercise activities. However, because of the multiple mechanisms through which exercise can reduce risk it is much more likely that the risk reduction is due to the additive effects of exercise in the diet plus exercise protocols.

As discussed earlier, oxidative damage to LDL and LDL-CHOL is an integral component of risk for CHD and larger LDL particles are more resistant to oxidation. Interestingly, using a diet plus exercise approach to reducing risk reduces the oxidizability of LDL.^{361,454–456} A reduction in LDL oxidation of up to 27% was observed in non-CHD patients, whereas a reduction of 15% was observed in CHD patients. Similar results have been observed in postmenopausal women as well.⁴⁵⁷ These results are interesting because they indicate another mechanism through which a diet plus exercise effect can reduce risk for CHD, a reduction in the oxidizability of LDL-CHOL.

4.3.2.3 Inflammation

Lifestyle modification also has been demonstrated to contribute to reduced inflammation. CRP can be used as a marker for the general inflammatory status of a person. When considered in conjunction with total-CHOL in plasma, CRP + total-CHOL serves as a better predictor of CHD risk than total-CHOL alone, is a stronger predictor of cardiac events than LDL-CHOL, and appears to add prognostic information for all degrees of metabolic syndrome.458,459 This is, of course, logical because inflammatory mechanisms are central to the cause of atherosclerosis. In diet plus exercise studies such as the Pritikin protocol, declines of 45% in CRP and similar decreases in amyloid A and sICAM have been observed, as are reductions in platelet aggregation, formation of TXs, and formation of plasminogen activator inhibitor.^{460–462} These results indicate a general suppression of inflammation by the combined treatments. Long-term weight-loss programs in the form of lifestyle modification in obese women also significantly reduce CRP, P-selectin, and ICAM-1.463,464 Furthermore, in the aforementioned diet-only studies by Jenkins and others, CRP decreased by 28% in the diet group and 33% in the lovastatin group, suggesting that the ability of diet to reduce CRP is comparable to that of statin therapy. While reductions in CRP and adipose-derived cytokines are associated with weight loss in obese individuals, the Jenkins studies did not produce weight loss but still displayed a reduction in inflammatory status. In a recent short-term diet plus exercise study by Izadpanah and others⁴⁶⁵ in overweight/obese children where there was a small change in weight, significant reductions of 20%-40% in proinflammatory and adipose-derived cytokines were observed as well. These results indicate that weight reduction is poorly correlated with the anti-inflammatory effects of diet/exercise, further supporting the concept that obesity per se is not always a high-risk phenotype.

4.3.2.3.1 Lipid Mediators of Inflammatory Signaling

As discussed in detail in Chapter 2, the mechanisms responsible for the observed reductions in inflammation by dietary components in the studies mentioned in Section 4.3.2.3 are most likely related to a combination of enhancing nutrient status by increasing fruit and vegetable consumption and of the anti-inflammatory effects of various phytochemicals and fatty acids. The effects of nutrient status on inflammatory signaling have already been discussed, whereas those of phytochemicals and other components of a healthy diet follow. The major targets for reducing inflammatory risk are the signal transduction pathways that lead to the synthesis of proinflammatory cytokines and those that lead to the activation of PLA₂, which then leads to the production of proinflammatory lipid mediators.

From the diet plus exercise studies where reductions in CRP are commonly seen, an enhanced availability of EPA and DHA from the increased intake of α -linolenic acid (ALA) (an ω -3 fatty acid from nuts, seeds, and some vegetable oils) has been cited as one of the reasons for an anti-inflammatory effect. EPA and DHA are made from ALA and are commonly known to compete with AA for incorporation into membrane phospholipids. Once they are in the membrane phospholipids, they would reduce the production of PGE₂ and LTB₄ and enhance the production of the much less inflammatory PGs, TXs, and LTs. As discussed later, however, this may not be a major mechanism of protection.

As described previously in Section 4.2.2, the mechanical activation of plasma membrane–bound calcium channels allows the entry of calcium into the endothelial cells. The increased cytosolic calcium will ultimately activate PLA_2 , Ras, and PKC, with the activated Ras and PKC enhancing the activation of PLA_2 (an effect also produced by ROSs).^{166,466–469} Once AA molecules are released from membrane phospholipids, they are metabolized into prostaglandin G_2 (PGG₂) by COX-1 and COX-2. The PGG₂ is converted into prostaglandin H_2 (PGH₂) and then into a variety of proinflammatory PGs and TXs by specific synthases.^{141,156,159,167,470} AA also is made into the proinflammatory LTs (including LTB₄) by 5-LOX in concert with other synthases.^{141,166,168} Thus, the mechanical activation of proteins in endothelial cells by intermittent shear results in the synthesis of proinflammatory lipid mediators. Dampening the production of these molecules will then have the effect of reducing proinflammatory effects. Through competing with AA for incorporation into membrane phospholipids, additional availability of DHA and EPA will lead to an enhanced production of the less-inflammatory 3-series PGs (PGE₃, PGA₃, and PGD₃), TXA₃/TXB₃, and LTB₅ by the same enzymes that modify AA. In addition to this indirect anti-inflammatory effect, as precursors for the synthesis of resolvins and protectins, DHA and EPA also might have a direct anti-inflammatory effect.

For these effects to occur, there must be some dietary or nutritional issues that are not optimal in the general population, issues that are corrected by the commonly used preventive diets. As a nutritional issue, it is the essential fatty acids linoleic acid (LA, an ω -6 fatty acid) and ALA (an ω -3 fatty acid) that are important. Because LA is made into AA while ALA is synthesized into EPA and DHA, many have proposed the idea of reducing LA intake and increasing ALA intake through dietary modification.⁴⁷¹ Such recommendations are hypothesized to reduce inflammatory and enhance anti-inflammatory effects by reducing AA and increasing EPA and DHA levels in vivo. They are also associated with an antiarrhythmic effect and, as discussed later, these effects may be the predominant protective effect of increasing ALA intake from nuts and oils, whereas different mechanisms may be responsible for the protective benefits of EPA/DHA from fish.⁴⁷²⁻⁴⁷⁴

In a series of meta-analyses of human clinical trials where LA consumption was manipulated, the higher LA intakes were associated with an anti-inflammatory effect (lower CRP), along with a lowered LDL-CHOL cholesterol and a higher HDL to LDL ratio.^{475,476} Enhanced LA intake also does not substantially change the AA content of platelets and erythrocytes.⁴⁷⁷ These diet-induced changes are certainly associated with a reduction in risk for CHD and make it difficult to justify reducing the intake of LA on the grounds that lower AA will reduce inflammatory signaling, per se.

Greatly increasing ALA intake to reduce inflammatory signaling also is suspect. Conversion rates of ALA to EPA are only 1%–8% in men and approximately 8%–21% in women; conversion to DHA is much less than 1% for both.^{478,479} ALA is also a small-minority fatty acid in available vegetable oils (<2% except for ~50% in flax oil), and there is only 9 to 10 g of ALA per 100 g of walnuts or butternuts⁴⁸⁰ (the two major ALA-containing non-oil foods). Because of the inefficient conversion to EPA and DHA, excessive calories from oil or oily nuts would need to be consumed to get the biological availability of a single gram of EPA (and of far less DHA). Even substituting high-ALA foods for low-ALA foods is impractical; 25 g of ALA for men and 10 g of ALA for women would have to be consumed just to produce an approximate 1 g benefit of EPA. Half of the dietary intake of lipid would have to come from walnuts and be supplemented with walnut oil and flax oil in order to obtain an increase in the

daily ALA intake of approximately 10 g without substantially increasing calories.⁴⁸¹ Such extensive dietary changes are not very easy for most.

Even in clinical trials using ALA supplements, the EPA content of platelets, neutrophils, and monocytes was either unaltered or inconsistently increased to greater than or equal to 300% at ALA intakes above 10 g/day, with little to no effect observed on DHA content.^{475,478,479,482-484} Thus, the far smaller increases in ALA consumption that are typical of the preventive diets discussed earlier are unlikely to be a source of additional EPA and DHA to achieve an effective competition with AA and a substantial anti-inflammatory effect through this mechanism. Recommending substantial increases in ALA intakes to a greater than or equal to 10 g/day range for what is an inconsistent increase in EPA content in membranes, along with an inconsistently positive effect on risk for disease, is not practical.^{484,485} This does not mean that high-ALA foods should not be recommended as a component of a preventive diet. Such a recommendation is appropriate because only about 70% of adults in the United States meet the AI for ALA and simply switching from low-ALA oils to higher ALA oils for cooking has been associated with a reduction in risk for CHD.⁴⁸⁶ Because clinical trials have revealed essentially no increase in EPA/DHA incorporation into cellular membranes with moderate diet-based increases in ALA consumption, it is more likely that these diets improve risk for CHD, in part, by helping to correct underlying ALA insufficiencies along with correcting other nutrient insufficiencies.

This does not mean that altering the dietary intake of EPA and DHA (as opposed to ALA intake) does not provide any benefit. Consuming fish that already contains EPA and DHA is a highly practical choice to modify EPA and DHA availability, a practice that not only avoids large-scale dietary changes but also efficiently corrects any possible insufficiency in EPA/DHA due to a low ALA intake. Based on results from a wide array of studies, there appears to be a consistent reduction in risk for CHD at EPA intakes of 0.25-3 g/day, whereas intakes of 2-4 g/day reduce markers of inflammation and platelet adhesiveness.^{484,485,487–496} Interestingly, the majority of the protective effects against cardiac deaths occur in the 250-500 mg range of additional EPA intake. In studies that indicate a reduction in risk at the lowest intakes, subjects already had very low EPA/DHA (and ALA) consumption to begin with as well as a high risk for coronary artery disease death. From a practical standpoint, two 6-oz. servings of fatty fish (e.g., salmon or herring) a week will provide a sufficient intake of EPA and DHA to average well within the lowest beneficial range as well as minimize the risk of exposure to mercury, polychlorinated biphenyls, and other toxic contaminants.⁴⁹⁷ The recommendation of the AHA for a weekly consumption of at least two servings of fish or approximately 1 g/day of a mix of both EPA and DHA in a supplement is certainly reasonable.

Although anti-inflammatory effects of EPA/DHA supplementation are observed only at higher levels of intake (>3 g/day), the recommendation for a lower intake is not necessarily inappropriate. A limitation of the studies that document a reduction in systemic inflammation with higher doses of EPA/DHA is that CRP is most often used as a marker for inflammation. Both EPA and DHA are known to be PPAR agonists,^{498–500} and PPAR activation results in an inhibition of NF $\kappa\beta$ -mediated proinflammatory signaling. This may, in fact, be the reason for the protective effect of diets that contain higher amounts of ALA but do not actually increase EPA incorporation into cellular membranes. For these reasons, the dietary recommendations that are specifically related to LA, ALA, EPA, and DHA include consuming two servings of fish high in EPA/DHA each week and at least one serving of a high-ALA food each day to ensure adequate ALA/EPA/DHA intake.

Altering the intake of ALA, EPA, and DHA is not the only way to suppress proinflammatory signaling by lipid mediators. A variety of phytochemical components that are commonly found in fruits and vegetables are known to attenuate the production of the proinflammatory lipid mediators. Epigallocatechin-3-gallate (EGCG), quercetin, and kaempferol inhibit COX-1, COX-2, and LOX activities with subsequent decreases in PGE₂ production. Phenolics from extra virgin olive oil, tyrosol, hydroxytyrosol, apigenin, and luteolin, and several oleuropein compounds also inhibit the production of PGE₂, LTB₄, and TXA₂.⁵⁰¹⁻⁵⁰³ In clinical trials using virgin and extra virgin olive oils in the 25–50 g/day range, the extra virgin version consistently provides anti-inflammatory effects at the lower range of intake.^{502,504,505} From a dietary perspective, recommending the consumption of approximately 25 g of extra virgin oil at a caloric cost of approximately 225 kcal (or about 10% of total caloric intake) is neither harmful nor impractical, especially when extra virgin olive oil is substituted for other cooking oils.

4.3.2.3.2 Cytokine Mediators of Inflammatory Signaling

As discussed in Section 4.2.1, the constant activation of proinflammatory signaling by low, intermittent shear stress is the proximate cause of CHD, whereas constantly enhancing these responses through additional activation mechanisms results in the progression of initial fatty lesions to the more deadly type V and VI lesions. Activation of the NF $\kappa\beta$ -, MAPK-, and p38/JNK-associated signal transduction pathways is essential to inducing the synthesis of proinflammatory signals, and suppressing these pathways is central to the proresolution process. Thus, in order to attenuate the unavoidable proinflammatory response to these mechanical stresses, proinflammatory signaling processes simply need to be attenuated. These desired effects can be accomplished by eating foods that contain phenolic compounds that inhibit the NF $\kappa\beta$ -, MAPK-, and p38/JNK-associated pathways.

There are a large variety of phenolic compounds that are capable of doing just this, and these compounds are predominantly found in fruits, vegetables, whole grains, teas, wine, and many other plant-based foods.⁵⁰⁶ Foods that contain clinically relevant amounts of phenolic compounds include various herbs and spices (e.g., cloves, Ceylon cinnamon, dried pot marjoram, dried spearmint, dried wild marjoram oregano, dried summer savory, dried sweet basil, dried sweet bay, dried marjoram, and capers), dried fruits (e.g., prunes, raisins, and figs), fruits (e.g., black elderberry, black chokeberry, skunk currant, black raspberry, black currant, Canada blueberry, and gooseberry), nuts (e.g., chestnut, pecan, walnut, pistachio, hazelnut, almond, Brazil nut, and cashew), beans (e.g., adzuki bean, lentils, black bean, and broad bean), vegetables (e.g., Swiss chard, dandelion, red cabbage, green bean, chili pepper, spinach, sweet pepper, raw Brussels sprouts, broccoli, and cauliflower), oils (e.g., peanut oil, extra virgin olive oil, canola/rape, and sesame), and whole grains (e.g., buckwheat and wheat flour, wheat germ, oat flour, corn flour, and rice flour).

Although many of the phenolic compounds found in the aforementioned foods have antioxidant properties, their preventive properties are likely due to their ability to modify components of signal transduction pathways. This is because with a halflife of only 1 to 2 hours and levels of consumption (at the highest levels documented) that do not come within 20% of that necessary to provide antioxidant function in vivo, it is doubtful that their antioxidant properties contribute significantly to a dietbased protective effect.^{441,507–509}

Some of the more commonly known phenolics that have been studied for their ability to modify signal transduction pathways include the flavonoids quercetin, apigenin, EGCG, theaflavin, genistein, anthocyanins, and kaempferol; the phenolic acids ellagic acid, capsaicin, and curcumin; and the stilbenoid resveratrol.^{352,502,508,510-519} In both cell culture and animal models, the syntheses of IL-6, IL-1 β , TNF- α , and interferons have been observed to be suppressed by these compounds. In response to the reduction in synthesis of these signaling molecules, a reduced expression of iNOS, COX-2, ICAM, and MCP-1 has also been observed with the same phenolics.^{502,508,513,516,518}

In a related area, the activation of PRRs is an essential component of NF $\kappa\beta$ and MAPK-mediated inflammatory signaling in response to infection and damage. As recently reviewed,⁵¹³ experiments have revealed that the activation of TLR4 can be suppressed by curcumin and sulforaphane (cruciferous vegetables) and nucleo-tide oligomerization domain (NOD) 1/2 activation can be inhibited by curcumin. Resveratrol, EGCG, luteolin, and quercetin can directly inhibit TBK1, whereas resveratrol also inhibits TRIFF to suppress both the MyD88 and the non-MyD88 pathways of TLR signaling.⁵¹³ These results are important because they will reduce the contribution of infections to the CHD process.

Because activation of NF $\kappa\beta$ is essential to proinflammatory signaling, its suppression provides an important target for attenuating inflammatory signaling. Activation of NF $\kappa\beta$ signaling can also be suppressed through a variety of mechanisms including inhibition of IKK activity, which then suppresses I $\kappa\beta$ release and its subsequent degradation. This prevents free NF $\kappa\beta$ from binding in the nucleus. EGCG and theaflavins inhibit NF $\kappa\beta$ signaling in this way, and quercetin, curcumin, kaempferol, and lycopene inhibit NF $\kappa\beta$ signaling by suppressing nuclear translocation of p50/p65. Apigenin, morin, anthocyanins, and procyanidins also suppress NF $\kappa\beta$ signaling through a variety of related mechanisms, including inhibition of phosphorylation of p65 and DNA binding of NF $\kappa\beta$.^{502,508,516,518}

The various MAPK pathways also are important targets for suppressing inflammatory signaling, and various phenolics are known to do this as well. Apigenin, luteolin, quercetin, resveratrol, EGCG, and kaempferol suppress ERK1/2–MAPK, JNK1/2–MAPK, and p38–MAPK pathways, whereas the cyanidins suppress ERK activation. EGCG has been observed to inhibit ERK–MAPK and, in addition to suppressing the MAPK pathways, curcumin inhibits PKC, a calcium-activated enzyme that can activate various components of the various MAPKs.^{502,508,516,518}

All these effects on different aspects of NF $\kappa\beta$ and MAPK signaling have been tested in a variety of different cell types. Some cell types exhibit suppression of all MAPK pathways as well as NF $\kappa\beta$ pathways by single phenolics, whereas in others only one or two pathways are inhibited by the same phenolic; similar differences in responses are observed for different phenolics as well.^{502,508,516,518,520} Thus, it is doubtful that any single-phenolic supplement will have the desired overall preventive

effects. The dietary recommendation to produce the suppression of all proinflammatory signaling pathways therefore includes a large variety of fruit and vegetable sources in order to maximize the consumption of the many different phenolics.

4.3.2.3.3 Redox Control

Preventive effects of phenolics are not limited to attenuating inflammatory signaling. An increased expression of various antioxidant and redox control enzymes also enhances redox control. SOD, CAT, GPX, and glutathione reductase are all highly relevant for increasing protection from additional ROSs/RNSs in addition to integrating with Trx and Prx to ensure appropriate control of redox status.

The role of SOD and CAT is to promote the dismutation of superoxide anions and peroxide to O_2 and H_2O at rates that are much faster than spontaneous rates of dismutation. GSR and GPX also work with glutathione to eliminate H_2O_2 . The benefit of these reactions is that they minimize both O_2 and H_2O_2 , which in turn minimizes the production of •OH. In concert, these antioxidant systems ultimately reduce damage to LDL-CHOL as well as to proteins, lipids, and DNA caused by ROSs.^{521,522}

It is highly instructive that activation of Nrf2 and its subsequent interaction with ARE increase the synthesis of a variety of protective enzymes, including Mn-SOD, CAT, glutathione reductase, Trx, Prx, HO-1, and GCL.^{350–353} A variety of endogenous factors are capable of activating Nrf2, including PGJ₂, PKC, MAPKs, and PI3K/PKB, in response to a variety of stresses that enhance cytosolic calcium and ROSs.^{351,515} It is through the Nrf2/ARE stress-response genes that polyphenols also function to enhance protection from inflammation-produced ROSs as well as to enhance control of cellular redox states.

A number of phenolics have been documented to activate Nrf2/ARE binding to DNA and induce the expression of antioxidant, redox control, and other protective enzymes. Quercetin, resveratrol, diallyl sulfide, S-allyl cysteine, lycopene, EGCG, curcumin, and sulforaphane have all been documented to activate Nrf2/ARE and induce synthesis of GST, GSH, HO-1, SOD, CAT, GCL, Trx, Prx, and UDP-GT.⁵²³⁻⁵³⁶ Nrf2 normally forms complexes with Keap1 in the cytosol, which prevents it from entering the nucleus and targets it for degradation. Oxidation of thiol residues on Keap1 or phosphorylation of NRF2 will disassociate it from the complex to allow it to enter the nucleus and bind to the ARE-binding sites.352 From induction studies, an array of phenolics alters the oxidation state of susceptible thiols on Keap1, whereas others activate protein kinases, which then phosphorylate Nrf2. Phenolic phytochemicals also can form conjugates with susceptible thiols on Keap1, which then reduces the degradation of Nrf2 to enhance its binding to the ARE/ERE response element.351,537 These results indicate that dietary polyphenols modify the oxidation or phosphorylation state of regulatory proteins such as Keap1 and various components of the MAPK pathways to activate Nrf2/ARE and the expression of a large array of protective enzymes.

It is also worth noting that HO-1 (along with the constitutive HO-2) catalyzes the rate-limiting step in heme degradation to produce free iron, bilirubin, and CO.^{538,539} The HO-1 enzyme is highly inducible through Nfr2/ARE activation; inducing this enzyme would make sense considering the potential for oxidative damage to inactivate heme proteins and produce a requirement for an additional capacity for scavenging the iron. As a consequence of increased HO-1 activity, CO production is

enhanced, and this molecule appears to be important in inflammatory responses. Macrophages with induced HO-1 release less TNF- α and, in similar cell culture models, HO-1 inhibits the synthesis of IL-1, IL-6, ICAM, VCAM, and E-selectin.^{539–543} While the chemistry of the effect is unknown, CO appears to affect the activity of MAPK pathways through upstream events.



FIGURE 4.3 Reduction in mechanisms of risk for atherosclerosis by diet: maintaining body weight through diet and increasing the consumption of fruits, vegetables, and fish reduce proinflammatory signaling through suppressing the activation of mitogen-activated protein kinases (MAPKs), nuclear factor κ - β (NF $\kappa\beta$), cyclooxygenase (COX), and lipoxygenase (LOX) as well as inducing the synthesis of antioxidant and redox control enzymes via Nrf2. EVOO refers to extra virgin olive oil.

From all of these studies, it is apparent that the additive and synergistic effects of an array of phenolics are responsible for the observed protective effects on inflammation and CHD. Therefore, dietary recommendations include specific recommendations for a wide variety of different foods with moderate to high phenolic content to produce a diet that maximizes the preventive effects.

4.4 SUMMARY AND RECOMMENDATIONS

4.4.1 ETIOLOGY OF CORONARY HEART DISEASE

CHD, caused by atherosclerosis, is an inflammatory disease. The apparent proximate cause of the initiating inflammatory events is an intermittent low shear mechanical stress, which occurs primarily at the outside walls of bifurcates, the leading edges of the bifurcates themselves, and the inside curve of curved stretches of arteries. The intermittent mechanical stress activates a nonresolving inflammatory stress response via the entry of calcium into the cell through mechanical stress-activated calcium channels and the mechanical activation of G-protein-coupled receptors, NADPH oxidase, and PI3K. The subsequent activation of PLA₂, and PKC, along with PI3K leads to the activation of various components of NF $\kappa\beta$ and the different MAPK pathways, resulting in the production of proinflammatory lipid mediators and cytokines. This then leads to the production of chemoattractant factors and a breakdown of various adhesion proteins to more easily allow the entry of monocytes, which differentiate into macrophages, as well as various lipoproteins (and eventually the entry of other inflammatory cells) into the vessel wall. At the same time, the activated G-coupled proteins and NADPH oxidases lead to an enhanced production of extracellular ROSs. The ROSs oxidize LDL and LDL-CHOL, which then bind to class A and CD36 scavenger receptors (PRRs) on macrophages, endothelial cells, and smooth muscle cells and activate their uptake by these cells. With continued uptake of oxidized lipids by the macrophages, they eventually become lipid-laden foam cells that form the primary type I and type II fatty lesions. The activation of PRRs by LDL-Ox and CHOL-Ox enhances proinflammatory signaling in the vessel wall through the activation of TLRs, adding to the proinflammatory stress responses to the intermittent mechanical forces.

When these localized inflammatory responses are enhanced through a variety of endogenous and exogenous factors, continued expansion of the lesions mediated by the expanding repertoire of local inflammatory cells and platelets will increase the size of the lesions and enhance their instability through increased production of CD40 from platelets. The resulting increased expression of MMPs by endothelial and smooth muscle cells along with cathepsin G will degrade the collagen, VE-cadherin, and fibronectin within the plaque. With continuing aggressive inflammatory responses, the lesions will progressively deteriorate and ultimately break off as a thrombus and block a smaller vessel, causing heart attack or stroke.

The role that the various risk factors play in the disease process is to accelerate the local mechanical shear-induced inflammatory stress response. Over the last several decades, due to changes in dietary patterns and activity levels, there has been a consistent increase in adiposity, resulting in expanding adipose tissue production of proinflammatory cytokines. These circulating cytokines exacerbate the local responses in the arterial vessel wall. In individuals who are very inactive, this chronic low-level proinflammatory signaling also contributes to the development of insulin resistance. With insulin resistance, the serum levels of glucose rise substantially, which leads to the formation of Amadori products. These products rearrange to form various AGEs, which then bind to RAGEs. RAGEs are PRRs that recognize damaged proteins and, as such, they also activate proinflammatory signaling pathways, contributing additional risk for CHD. AGEs can also initiate lipid peroxidation events and in this way contribute to enhancing LDL and LDL-CHOL oxidation. Hypertension increases the mechanical stretch of the vessel walls and contributes to the mechanical stress response in order to enhance a localized hypertrophy response. Although this response may be necessary to maintain appropriate regulation of blood flow in the face of increased pressures, it does add to the localized inflammatory stress responses. In addition to these factors, a large majority of Americans have an insufficient consumption of vitamins and minerals that are directly involved in antioxidant function and redox regulation and therefore do not have optimal control of either cellular redox function or antioxidant function. These nutritional insufficiencies lead to increased oxidation of LDL and cholesterol as well as to exaggerated inflammatory responses, again, enhancing the CHD processes.

In order to maximally reduce risk for atherosclerosis, attenuating the various mechanisms of cause is necessary. Because inflammatory stress responses are the primary cause of CHD, inflammatory signaling is a primary target for preventive measures. The first preventive measure would be to correct nutritional insufficiencies by following a diet that provides sufficient amounts of all the required nutrients in order to optimize redox and antioxidant control. This will minimize the likelihood of occurrence of inappropriately large inflammatory responses to low-intensity stressors. Then, the diet concept must be modified to include the consumption of only a sufficient amount of calories to maintain body weight in order to avoid inflammatory responses due to increasing adiposity. In addition to this, the diet must also include an array of foods (fruits, nuts, and vegetables) that have a high phytochemical content in order to dampen inflammatory signaling and reduce inflammatory responses to unavoidable stressors. By adding sufficient physical exercise, insulin resistance can be minimized while additional anti-inflammatory effects can be obtained.

4.4.2 DIETARY RECOMMENDATIONS FOR PREVENTION OF CORONARY HEART DISEASE

Because CHD is an inflammatory disease, the anti-inflammatory diet and exercise protocols presented in Chapter 2 become the preventive measures for CHD as well. The diet recommendations are designed to provide the greatest anti-inflammatory effects possible while ensuring compliance with current nutritional recommendations, as well as being practical. The specific foods that make up the recommendation for an anti-inflammatory diet are chosen based on their known anti-inflammatory effects. The recommendations depart from the *MyPlate* approach by expanding the food categories to include a 0.25-cup serving of nuts and seeds as well as

a 1-cup serving of beans and lentils as individual categories in order to emphasize phytochemical intake. In addition, the consumption of high-phytochemical foods is increased at the expense of grains and cereals by increasing the number of fruit servings to 3 cups/day from 2 and the number of vegetable servings to 4 cups/day from 3. The addition of a minimum of two servings of fish each week provides additional anti-inflammatory effects through increased EPA/DHA consumption. The reader will note that the caloric content of these recommendations will approximate 1500–1700 kcal, which is somewhat less than the average daily requirements for both men and women. Because the consumption of moderate amounts of individual sources of proteins, carbohydrates, or fats has little to no negative impact on CHD, the remaining calorie requirements can be obtained from whole foods at the individual's discretion, provided, of course, that no excess calories are consumed that will result in an increase in adiposity.

4.4.3 Physical Activity Recommendations for Prevention of Coronary Heart Disease

For exercise, it is apparent that different forms of exercise provide risk reduction effects through at least three basic mechanisms. Frequent short periods of low- to moderate-intensity endurance exercise and/or resistance exercise enhance insulin sensitivity and reduce risk for CHD by minimizing serum levels of glucose. Longer term periods of strenuous physical activity for 20-30 minutes appear to enhance antioxidant enzyme function, which would enhance redox control and reduce risk by dampening proinflammatory signaling. Longer term exercise of approximately 60 minutes that is of vigorous intensity (at a minimum of 6 METs) and utilizes large muscle groups provides anti-inflammatory effects due to large increases in IL-6, sIL-1ra, sIL-1r, and sTNFr while maintaining insulin sensitivity and enhancing antioxidant and redox enzyme function in most tissues, an effect that reduces ROS damage to a variety of cell components as well as to LDL/HDL/CHOL. In addition, considering the difficulties in delineating the protective effects of exercise from those of weight loss in many studies where daily exercise periods exceed 45–60 minutes in duration, a confounder that occurs far less frequently with shorter periods of activity, it appears that the longer periods of exercise at moderate intensity may be associated with reducing obesity-related risks as well.

Thus in order to obtain optimal reduction in risks for CHD through the antiinflammatory, enhanced antioxidant/redox control, weight maintenance, and insulin-sensitizing effects of exercise, the American College of Sports Medicine and the AHA recommendations for adolescents that includes vigorous activities that are 6 METs or greater in intensity and carried out for a minimum of 1 hour for 5 days each week are recommended here. These activities include jogging, running, cycling at 10–12 mph or greater, swimming, cross-country skiing, or similar vigorous activities.⁵⁴⁴

Although not providing optimum effects, reductions in risk can also be obtained from shorter duration (20–30 minutes) and higher intensity activities performed three to five times each week. Such activities include high-intensity running, cycling, swimming, and skiing and would maintain insulin resistance and enhance redox control.

Lower levels of protection might be provided through several short periods of moderate-intensity activities performed daily, such as fast walking, yard maintenance, leisurely bicycling, volleyball, baseball, skating, or canoeing, and through resistance-type weight-lifting and muscle-endurance training activities by maintaining insulin sensitivity.

ACKNOWLEDGMENT

This work was supported in part by a partial salary grant from NEIU to C. Murray Ardies.

Christian K. Roberts is supported by the American Heart Association (BGIA #0765139Y), the National Heart, Lung and Blood Institute (P50 HL105188), and the National Institute of Diabetes and Digestive and Kidney Diseases (DK090406).

REFERENCES

- Heidenreich PA, Trogdon JG, Khavjou OA, Butler J, Dracup K, Ezekowitz MD, Finkelstein EA, Hong Y, Johnston SC, Khera A. Forecasting the future of cardiovascular disease in the United States: A policy statement from the American Heart Association. *Circulation*. 2011;123:933–944.
- 2. Xu J, Kochanek KD, Murphy SL, Tejada-Vera B. Deaths: Final data for 2007. *National Vital Statistics Reports*. 2010;58:1–135.
- Roger VL, Go AS, Lloyd-Jones DM, Benjamin EJ, Berry JD, Borden WB, Bravata DM, Dai S, Ford ES, Fox CS. Executive summary: Heart disease and stroke statistics—2012 update: A report from the American Heart Association. *Circulation*. 2012;125:188.
- McGovern PG, Pankow JS, Shahar E, Doliszny KM, Folsom AR, Blackburn H, Luepker RV. Recent trends in acute coronary heart disease—mortality, morbidity, medical care, and risk factors. *New England Journal of Medicine*. 1996;334:884–890.
- McGill HC, Jr., McMahan CA, Gidding SS. Preventing heart disease in the 21st century. *Circulation*. 2008;117:1216–1227.
- CDC. Prevalence of coronary heart disease—United States, 2006–2010. Morbidity and Mortality Weekly Report. 2011;60:1377–1381.
- Berry JD, Dyer A, Cai X, Garside DB, Ning H, Thomas A, Greenland P, Van Horn L, Tracy RP, Lloyd-Jones DM. Lifetime risks of cardiovascular disease. *New England Journal of Medicine*. 2012;366:321–329.
- CDC. Million hearts: Strategies to reduce the prevalence of leading cardiovascular disease risk factors, United States, 2011. *Morbidity and Mortality Weekly Report*. 2011;60:1248–1251.
- Vaccination P. Prevalence of coronary heart disease—United States, 2006–2010. *The Americas*. 2010;2006:1385.
- Hunink MGM, Goldman L, Tosteson ANA, Mittleman MA, Goldman PA, Williams LW, Tsevat J, Weinstein MC. The recent decline in mortality from coronary heart disease, 1980–1990. JAMA: The Journal of the American Medical Association. 1997;277:535–542.
- McGovern PG, Jacobs DR, Jr., Shahar E, Arnett DK, Folsom AR, Blackburn H, Luepker RV. Trends in acute coronary heart disease mortality, morbidity, and medical care from 1985 through 1997: The Minnesota heart survey. *Circulation*. 2001;104:19–24.
- 12. Fox CS, Evans JC, Larson MG, Kannel WB, Levy D. Temporal trends in coronary heart disease mortality and sudden cardiac death from 1950 to 1999. *Circulation*. 2004;110:522–527.

- 13. Ford ES, Capewell S. Proportion of the decline in cardiovascular mortality disease due to prevention versus treatment: Public health versus clinical care. *Annual Review of Public Health*. 2011;32:5–22.
- Roger VL, Go AS, Lloyd-Jones DM, Adams RJ, Berry JD, Brown TM, Carnethon MR, Dai S, de Simone G, Ford ES. Heart disease and stroke statistics—2011 update1. A report from the American Heart Association. *Circulation*. 2011;123:e18–e209.
- Lloyd-Jones D, Adams R, Carnethon M, De Simone G, Ferguson TB, Flegal K, Ford E, Furie K, Go A, Greenlund K. Heart disease and stroke statistics—2009 update: A report from the American Heart Association Statistics Committee and Stroke Statistics Subcommittee. *Circulation*. 2009;119:e21–e181.
- Lloyd-Jones D, Adams RJ, Brown TM, Carnethon M, Dai S, De Simone G, Ferguson TB, Ford E, Furie K, Gillespie C. Heart disease and stroke statistics—2010 update. *Circulation*. 2010;121:e46–e215.
- Thom T, Haase N, Rosamond W, Howard VJ, Rumsfeld J, Manolio T, Zheng ZJ, Flegal K, O'donnell C, Kittner S. Heart disease and stroke statistics—2006 update. *Circulation*. 2006;113:e85–e151.
- Rosamond W, Flegal K, Friday G, Furie K, Go A, Greenlund K, Haase N et al. Heart dis-ease and stroke statistics—2007 update. A report from the American Heart Association Statistics Committee and Stroke Statistics Subcommittee. *Circulation*. 2007;115:e69–e171.
- Rosemond R, Flegal K, Furie K, Go A, Greenlund K, Haase N, Hailpern SM et al. Heart disease and stroke statistics—2008 update. A report from the American Heart Association Statistics Committee and Stroke Statistics Committee. *Circulation*. 2008;117:e25–e146.
- Prince SA, Adamo KB, Hamel ME, Hardt J, Gorber SC, Tremblay M. A comparison of direct versus self-report measures for assessing physical activity in adults: A systematic review. *International Journal of Behavioral Nutrition and Physical Activity*. 2008;5:56.
- Trabulsi J, Schoeller DA. Evaluation of dietary assessment instruments against doubly labeled water, a biomarker of habitual energy intake. *American Journal of Physiology-Endocrinology and Metabolism.* 2001;281:E891–E899.
- Sesso HD, Paffenbarger RS, Jr., Lee IM. Physical activity and coronary heart disease in men: The Harvard alumni health study. *Circulation*. 2000;102:975–980.
- 23. Blair SN, Kohl HW, III, Paffenbarger RS, Jr., Clark DG, Cooper KH, Gibbons LW. Physical fitness and all-cause mortality. A prospective study of healthy men and women. *JAMA: The Journal of the American Medical Association*. 1989;262:2395–2401.
- Powell KE, Thompson PD, Caspersen CJ, Kendrick JS. Physical activity and the incidence of coronary heart disease. *Annual Review Public Health*. 1987;8:253–287.
- Manson JE, Greenland P, LaCroix AZ, Stefanick ML, Mouton CP, Oberman A, Perri MG, Sheps DS, Pettinger MB, Siscovick DS. Walking compared with vigorous exercise for the prevention of cardiovascular events in women. *New England Journal of Medicine*. 2002;347:716–725.
- Manson JAE, Hu FB, Rich-Edwards JW, Colditz GA, Stampfer MJ, Willett WC, Speizer FE, Hennekens CH. A prospective study of walking as compared with vigorous exercise in the prevention of coronary heart disease in women. *New England Journal of Medicine*. 1999;341:650–658.
- Paffenbarger RS, Jr., Hyde RT, Wing AL, Hsieh CC. Physical activity, all-cause mortality, and longevity of college alumni. *New England Journal of Medicine*. 1986;314:605–613.
- Blair SN, Kampert JB, Kohl HW, 3rd, Barlow CE, Macera CA, Paffenbarger RS, Jr., Gibbons LW. Influences of cardiorespiratory fitness and other precursors on cardiovascular disease and all-cause mortality in men and women. *JAMA: The Journal of the American Medical Association*. 1996;276:205–210.

- Wei M, Kampert JB, Barlow CE, Nichaman MZ, Gibbons LW, Paffenbarger RS, Jr., Blair SN. Relationship between low cardiorespiratory fitness and mortality in normalweight, overweight, and obese men. *JAMA: The Journal of the American Medical Association*. 1999;282:1547–1553.
- 30. Kodama S, Saito K, Tanaka S, Maki M, Yachi Y, Asumi M, Sugawara A et al. Cardiorespiratory fitness as a quantitative predictor of all-cause mortality and cardiovascular events in healthy men and women: A meta-analysis. *Journal of the American Medical Association*. 2009;301:2024–2035.
- Nocon M, Hiemann T, Muller-Riemenschneider F, Thalau F, Roll S, Willich SN. Association of physical activity with all-cause and cardiovascular mortality: A systematic review and meta-analysis. *European Journal of Cardiovascular Prevention and Rehabilitation*. 2008;15:239–246.
- 32. Thompson PD, Buchner D, Pina IL, Balady GJ, Williams MA, Marcus BH, Berra K et al. Exercise and physical activity in the prevention and treatment of atherosclerotic cardiovascular disease: A statement from the Council on Clinical Cardiology (Subcommittee on Exercise, Rehabilitation, and Prevention) and the Council on Nutrition, Physical Activity, and Metabolism (Subcommittee on Physical Activity). *Circulation*. 2003;107:3109–3116.
- Mora S, Cook N, Buring JE, Ridker PM, Lee I-M. Physical activity and reduced risk of cardiovascular events: Potential mediating mechanisms. *Circulation*. 2007; 116:2110–2118.
- Laukkanen JA, Lakka TA, Rauramaa R, Kuhanen R, Venalainen JM, Salonen R, Salonen JT. Cardiovascular fitness as a predictor of mortality in men. *Archives of Internal Medicine*. 2001;161:825–831.
- Lakka TA, Laukkanen JA, Rauramaa R, Salonen R, Lakka HM, Kaplan GA, Salonen JT. Cardiorespiratory fitness and the progression of carotid atherosclerosis in middle-aged men. Archives of Internal Medicine. 2001;134:12–20.
- Laukkanen JA, Rauramaa R, Salonen JT, Kurl S. The predictive value of cardiorespiratory fitness combined with coronary risk evaluation and the risk of cardiovascular and all-cause death. *Journal of Internal Medicine*. 2007;262:263–272.
- Gregg EW, Cauley JA, Stone K, Thompson TJ, Bauer DC, Cummings SR, Ensrud KE. Relationship of changes in physical activity and mortality among older women. *Journal* of the American Medical Association. 2003;289:2379–2386.
- Lee C-D, Jacobs DR, Jr., Hankinson A, Iribarren C, Sidney S. Cardiorespiratory fitness and coronary artery calcification in young adults: The CARDIA study. *Atherosclerosis*. 2009;203:263–268.
- 39. Shiroma EJ, Lee I-M. Physical activity and cardiovascular health: Lessons learned from epidemiological studies across age, gender, and race/ethnicity. *Circulation*. 2010;122:743–752.
- 40. Tanasescu M, Leitzmann MF, Rimm EB, Willett WC, Stampfer MJ, Hu FB. Exercise type and intensity in relation to coronary heart disease in men. *JAMA: The Journal of the American Medical Association*. 2002;288:1994–2000.
- 41. Alexander R. Atherosclerosis as disease of redox-sensitive genes. *Transactions of the American Clinical and Climatological Association*. 1998;109:129.
- 42. Keys A, Parlin RW. Serum cholesterol response to changes in dietary lipids. *The American Journal of Clinical Nutrition*. 1966;19:175–181.
- 43. Keys A. Seven countries: A Mulitvariate Analysis of Death and Coronary Heart Disease. Cambridge, MA: Harvard University Press; 1980.
- McGill HC, Jr. The relationship of dietary cholesterol to serum cholesterol concentration and to atherosclerosis in man. *The American Journal of Clinical Nutrition*. 1979;32:2664–2702.

- 45. Menotti A, Blackburn H, Kromhout D, Nissinen A, Fidanza F, Giampaoli S, Buzina R et al. Changes in population cholesterol levels and coronary heart disease deaths in seven countries. *European Heart Journal*. 1997;18:566–571.
- Posner BM, Cobb JL, Belanger AJ, Cupples LA, D'Agostino RB, Stokes J, III. Dietary lipid predictors of coronary heart disease in men. The Framingham study. *Archives of Internal Medicine*. 1991;151:1181–1187.
- Castelli WP, Garrison RJ, Wilson PW, Abbott RD, Kalousdian S, Kannel WB. Incidence of coronary heart disease and lipoprotein cholesterol levels. The Framingham study. *JAMA: The Journal of the American Medical Association*. 1986;256:2835–2838.
- 48. Stamler J, Wentworth D, Neaton JD. Is relationship between serum cholesterol and risk of premature death from coronary heart disease continuous and graded? Findings in 356,222 primary screenees of the Multiple Risk Factor Intervention Trial (MRFIT). JAMA: The Journal of the American Medical Association. 1986;256:2823–2828.
- 49. Campbell TC, Parpia B, Chen J. Diet, lifestyle, and the etiology of coronary artery disease: The Cornell China study. *The American Journal of Cardiology*. 1998;82:18T–21T.
- Kato H, Tillotson J, Nichaman MZ, Rhoads GG, Hamilton HB. Epidemiologic studies of coronary heart disease and stroke in Japanese men living in Japan, Hawaii and California. *American Journal of Epidemiology*. 1973;97:372–385.
- Hu FB, Stampfer MJ, Manson JE, Ascherio A, Colditz GA, Speizer FE, Hennekens CH, Willett WC. Dietary saturated fats and their food sources in relation to the risk of coronary heart disease in women. *The American Journal of Clinical Nutrition*. 1999;70:1001–1008.
- 52. Hu FB, Stampfer MJ, Manson JE, Rimm E, Colditz GA, Rosner BA, Hennekens CH, Willett WC. Dietary fat intake and the risk of coronary heart disease in women. *The New England Journal of Medicine*. 1997;337:1491–1499.
- 53. Hertog MG, Kromhout D, Aravanis C, Blackburn H, Buzina R, Fidanza F, Giampaoli S et al. Flavonoid intake and long-term risk of coronary heart disease and cancer in the seven countries study. *Archives of Internal Medicine*. 1995;155:381–386.
- 54. Jacobs DR, Jr., Meyer KA, Kushi LH, Folsom AR. Whole-grain intake may reduce the risk of ischemic heart disease death in postmenopausal women: The Iowa women's health study. The American Journal of Clinical Nutrition. 1998;68:248–257.
- 55. Liu S, Stampfer MJ, Hu FB, Giovannucci E, Rimm E, Manson JE, Hennekens CH, Willett WC. Whole-grain consumption and risk of coronary heart disease: Results from the nurses' health study. *The American Journal of Clinical Nutrition*. 1999;70:412–419.
- 56. Liu S, Willett WC, Stampfer MJ, Hu FB, Franz M, Sampson L, Hennekens CH, Manson JE. A prospective study of dietary glycemic load, carbohydrate intake, and risk of coronary heart disease in US women. *The American Journal of Clinical Nutrition*. 2000;71:1455–1461.
- 57. Liu S, Manson JE, Buring JE, Stampfer MJ, Willett WC, Ridker PM. Relation between a diet with a high glycemic load and plasma concentrations of high-sensitivity C-reactive protein in middle-aged women. *The American Journal of Clinical Nutrition*. 2002;75:492–498.
- 58. Jacobs DR, Pereira MA, Meyer KA, Kushi LH. Fiber from whole grains, but not refined grains, is inversely associated with all-cause mortality in older women: The Iowa women's health study. *Journal of the American College of Nutrition*. 2000;19:326S–330S.
- Wolk A, Manson JE, Stampfer MJ, Colditz GA, Hu FB, Speizer FE, Hennekens CH, Willett WC. Long-term intake of dietary fiber and decreased risk of coronary heart disease among women. *JAMA: The Journal of the American Medical Association*. 1999;281:1998–2004.
- Liu S, Buring JE, Sesso HD, Rimm EB, Willett WC, Manson JE. A prospective study of dietary fiber intake and risk of cardiovascular disease among women. *Journal of the American College of Cardiology*. 2002;39:49–56.

- Brown L, Rosner B, Willett WW, Sacks FM. Cholesterol-lowering effects of dietary fiber: A meta-analysis. *The American Journal of Clinical Nutrition*. 1999;69:30–42.
- 62. Joshipura KJ, Hu FB, Manson JE, Stampfer MJ, Rimm EB, Speizer FE, Colditz G et al. The effect of fruit and vegetable intake on risk for coronary heart disease. *Annals of Internal Medicine*. 2001;134:1106–1114.
- 63. Liu S, Manson JE, Lee IM, Cole SR, Hennekens CH, Willett WC, Buring JE. Fruit and vegetable intake and risk of cardiovascular disease: The women's health study. *The American Journal of Clinical Nutrition*. 2000;72:922–928.
- 64. Bazzano LA, He J, Ogden LG, Loria CM, Vupputuri S, Myers L, Whelton PK. Fruit and vegetable intake and risk of cardiovascular disease in us adults: The first National Health and Nutrition Examination Survey epidemiologic follow-up study. *The American Journal of Clinical Nutrition*. 2002;76:93–99.
- 65. Hu FB, Stampfer MJ, Manson JE, Rimm E, Colditz GA, Speizer FE, Hennekens CH, Willett WC. Dietary protein and risk of ischemic heart disease in women. *The American Journal of Clinical Nutrition*. 1999;70:221–227.
- Stampfer MJ, Hu FB, Manson JE, Rimm EB, Willett WC. Primary prevention of coronary heart disease in women through diet and lifestyle. *The New England Journal of Medicine*. 2000;343:16–22.
- 67. Hu FB, Rimm EB, Stampfer MJ, Ascherio A, Spiegelman D, Willett WC. Prospective study of major dietary patterns and risk of coronary heart disease in men. *The American Journal of Clinical Nutrition*. 2000;72:912–921.
- Fung TT, Rimm EB, Spiegelman D, Rifai N, Tofler GH, Willett WC, Hu FB. Association between dietary patterns and plasma biomarkers of obesity and cardiovascular disease risk. *The American Journal of Clinical Nutrition*. 2001;73:61–67.
- 69. Kimmons J, Gillespie C, Seymour J, Serdula M, Blanck HM. Fruit and vegetable intake among adolescents and adults in the United States: Percentage meeting individualized recommendations. *The Medscape Journal of Medicine*. 2009;11:26.
- Schleicher RL, Carroll MD, Ford ES, Lacher DA. Serum vitamin C and the prevalence of vitamin C deficiency in the United States: 2003–2004 National Health and Nutrition Examination Survey (NHANES). *The American Journal of Clinical Nutrition*. 2009;90:1252–1263.
- Fulgoni VL, III, Keast DR, Bailey RL, Dwyer J. Foods, fortificants, and supplements: Where do Americans get their nutrients? *The Journal of Nutrition*. 2011;141:1847–1854.
- 72. Ross R, Glomset JA. Atherosclerosis and the arterial smooth muscle cell. *Science*. 1973;180:1332–1339.
- 73. Ross R. The pathogenesis of atherosclerosis—an update. *New England Journal of Medicine*. 1986;314:488–500.
- 74. Ross R. The pathogenesis of atherosclerosis: A perspective for the 1990s. *Nature*. 1993;362:801–809.
- 75. Ross R. Atherosclerosis is an inflammatory disease. *American Heart Journal*. 1999;138:419–420.
- Shalhoub J, Falck-Hansen MA, Davies AH, Monaco C. Innate immunity and monocytemacrophage activation in atherosclerosis. *Journal of Inflammation*. 2011;8:9.
- 77. Hennig B, Chow CK. Lipid peroxidation and endothelial cell injury: Implications in atherosclerosis. *Free Radical Biology and Medicine*. 1988;4:99–106.
- McCully KS, Vezeridis MP. Histopathological effects of homocysteine thiolactone on epithelial and stromal tissues. *Experimental and Molecular Pathology*. 1989;51:159–170.
- Rong JX, Rangaswamy S, Shen L, Dave R, Chang YH, Peterson H, Hodis HN, Chisolm GM, Sevanian A. Arterial injury by cholesterol oxidation products causes endothelial dysfunction and arterial wall cholesterol accumulation. *Arteriosclerosis, Thrombosis,* and Vascular Biology. 1998;18:1885.

- Griendling KK, FitzGerald GA. Oxidative stress and cardiovascular injury. *Circulation*. 2003;108:1912–1916.
- 81. Taniyama Y, Griendling KK. Reactive oxygen species in the vasculature. Molecular and cellular mechanisms. *Hypertension*. 2003;42:1075–1081.
- White CR, Haidekker M, Bao X, Frangos JA. Temporal gradients in shear, but not spatial gradients, stimulate endothelial cell proliferation. *Circulation*. 2001;103:2508–2513.
- Ku DN, Giddens DP, Zarins CK, Glagov S. Pulsatile flow and atherosclerosis in the human carotid bifurcation. Positive correlation between plaque location and low oscillating shear stress. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1985;5:293–302.
- Funk SD, Yurdagul A, Orr AW. Hyperglycemia and endothelial dysfunction in atherosclerosis: Lessons from type 1 diabetes. *International Journal of Vascular Medicine*. 2012;569654:1–19.
- van den Oever IAM, Raterman HG, Nurmohamed MT, Simsek S. Endothelial dysfunction, inflammation, and apoptosis in diabetes mellitus. *Mediators of Inflammation*. 2010;792393:1–15.
- 86. Wautier JL, Schmidt AM. Protein glycation. Circulation Research. 2004;95:233-238.
- Schalkwijk CG, Stehouwer CDA, van Hinsbergh VWM. Fructose-mediated nonenzymatic glycation: Sweet coupling or bad modification. *Diabetes/Metabolism Research and Reviews*. 2004;20:369–382.
- Dills Jr WL. Protein fructosylation: Fructose and the Maillard reaction. *The American Journal of Clinical Nutrition*. 1993;58:7798–787S.
- Tayal D, Goswami B, Koner BC, Mallika V. Role of homocysteine and lipoprotein (a) in atherosclerosis: An update. *Biomedical Research*. 2011;22:391–405.
- Andersson J, Libby P, Hansson GK. Adaptive immunity and atherosclerosis. *Clinical Immunology*. 2010;134:33–46.
- Libby P, Ridker PM, Hansson GK. Inflammation in atherosclerosis: From pathophysiology to practice. *Journal of the American College of Cardiology*. 2009;54:2129–2138.
- Hansson GK. Inflammation, atherosclerosis, and coronary artery disease. New England Journal of Medicine. 2005;352:1685–1695.
- 93. Kiechl S, Egger G, Mayr M, Wiedermann CJ, Bonora E, Oberhollenzer F, Muggeo M, Xu Q, Wick G, Poewe W, Willeit J. Chronic infections and the risk of carotid atherosclerosis: Prospective results from a large population study. *Circulation*. 2001;103:1064–1070.
- Mehta M, Jawahar L, Saldeen M, Tom G, Rand M. Interactive role of infection, inflammation and traditional risk factors in atherosclerosis and coronary artery disease. *Journal* of the American College of Cardiology. 1998;31:1217–1225.
- Ericson K, Saldeen TGP, Lindquist O, Pahlson C, Mehta JL. Relationship of Chlamydia pneumoniae infection to severity of human coronary atherosclerosis. *Circulation*. 2000;101:2568.
- Corrado E, Rizzo M, Tantillo R, Muratori I, Bonura F, Vitale G, Novo S. Markers of inflammation and infection influence the outcome of patients with baseline asymptomatic carotid lesions. *Stroke*. 2006;37:482–486.
- 97. Mahmoudi M, Curzen N, Gallagher P. Atherogenesis: The role of inflammation and infection. *Histopathology*. 2007;50:535–546.
- Bucala R, Makita Z, Koschinsky T, Cerami A, Vlassara H. Lipid advanced glycosylation: Pathway for lipid oxidation in vivo. *Proceedings of the National Academy of Sciences*. 1993;90:6434.
- 99. Goh SY, Cooper ME. The role of advanced glycation end products in progression and complications of diabetes. *Journal of Clinical Endocrinology & Metabolism*. 2008;93:1143–1152.
- Makita Z, Yanagisawa K, Kuwajima S, Bucala R, Vlassara H, Koike T. The role of advanced glycosylation end-products in the pathogenesis of atherosclerosis. *Nephrology Dialysis Transplantation*. 1996;11:31–33.

- Barlovic D, Soro-Paavonen A, Jandeleit-Dahm K. RAGE biology, atherosclerosis and diabetes. *Clinical Science*. 2011;121:43–55.
- 102. VanderLaan PA, Reardon CA, Getz GS. Site specificity of atherosclerosis. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2004;24:12–22.
- 103. Stary H, Blankenhorn D, Chandler AB, Glagov S, Insull W, Jr., Richardson M, Rosenfeld M, Schaffer S, Schwartz C, Wagner W. A definition of the intima of human arteries and of its atherosclerosis-prone regions. A report from the Committee on Vascular Lesions of the Council on Arteriosclerosis, American Heart Association. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1992;12:120.
- 104. Izzo JL, Mitchell GF. Arterial stiffness: A simplified overview in vascular medicine. In: Boyer JS, ed. Advances in Cardiology. New York, NY: Karger; 2007:1–18.
- 105. De Keulenaer GW, Chappell DC, Ishizaka N, Nerem RM, Alexander RW, Griendling KK. Oscillatory and steady laminar shear stress differentially affect human endothelial redox state: Role of a superoxide-producing NADH oxidase. *Circulation Research*. 1998;82:1094–1101.
- Gimbrone MA, Jr. Vascular endothelium, hemodynamic forces, and atherogenesis. *The* American Journal of Pathology. 1999;155:1.
- 107. Stary HC, Chandler AB, Glagov S, Guyton JR, Insull W, Jr., Rosenfeld ME, Schaffer SA, Schwartz CJ, Wagner WD, Wissler RW. A definition of initial, fatty streak, and intermediate lesions of atherosclerosis. A report from the Committee on Vascular Lesions of the Council on Arteriosclerosis, American Heart Association. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1994;14:840.
- 108. Lu D, Kassab GS. Role of shear stress and stretch in vascular mechanobiology. *Journal* of The Royal Society Interface. 2011;8:1379–1385.
- Chiu JJ, Chien S. Effects of disturbed flow on vascular endothelium: Pathophysiological basis and clinical perspectives. *Physiological Reviews*. 2011;91:327–387.
- Chatzizisis YS, Coskun AU, Jonas M, Edelman ER, Feldman CL, Stone PH. Role of endothelial shear stress in the natural history of coronary atherosclerosis and vascular remodeling: Molecular, cellular, and vascular behavior. *Journal of the American College* of Cardiology. 2007;49:2379–2393.
- 111. Nauli SM, Kawanabe Y, Kaminski JJ, Pearce WJ, Ingber DE, Zhou J. Endothelial cilia are fluid shear sensors that regulate calcium signaling and nitric oxide production through polycystin-1. *Circulation*. 2008;117:1161–1171.
- 112. Zhou J. Polycystins and primary cilia: Primers for cell cycle progression. *Annual Review* of Physiology. 2009;71:83–113.
- Chien S. Mechanotransduction and endothelial cell homeostasis: The wisdom of the cell. American Journal of Physiology-Heart and Circulatory Physiology. 2007;292: H1209–H1224.
- Hahn C, Schwartz MA. Mechanotransduction in vascular physiology and atherogenesis. Nature Reviews Molecular Cell Biology. 2009;10:53–62.
- 115. Shyy JYJ, Chien S. Role of integrins in endothelial mechanosensing of shear stress. *Circulation Research*. 2002;91:769–775.
- 116. Osawa M, Masuda M, Kusano K, Fujiwara K. Evidence for a role of platelet endothelial cell adhesion molecule-1 in endothelial cell mechanosignal transduction: Is it a mechanoresponsive molecule? *Science's STKE*. 2002;158:773.
- 117. Hwang J, Saha A, Boo YC, Sorescu GP, McNally JS, Holland SM, Dikalov S, Giddens DP, Griendling KK, Harrison DG. Oscillatory shear stress stimulates endothelial production of from p47phox-dependent NAD(P)H oxidases, leading to monocyte adhesion. *Journal of Biological Chemistry*. 2003;278:47291–47298.
- 118. Florian JA, Kosky JR, Ainslie K, Pang Z, Dull RO, Tarbell JM. Heparan sulfate proteoglycan is a mechanosensor on endothelial cells. *Circulation Research*. 2003;93:e136–e142.

- 119. Go YM, Park H, Maland MC, Darley-Usmar VM, Stoyanov B, Wetzker R, Jo H. Phosphatidylinositol 3-kinase γ mediates shear stress-dependent activation of JNK in endothelial cells. *American Journal of Physiology-Heart and Circulatory Physiology*. 1998;275:H1898–H1904.
- 120. Ngai C, Yao X. Vascular responses to shear stress: The involvement of mechanosensors in endothelial cells. *The Open Circulation and Vascular Journal*. 2010;3:85–94.
- 121. White RE, Kryman JP, El-Mowafy AM, Han G, Carrier GO. cAMP-dependent vasodilators cross-activate the cGMP-dependent protein kinase to stimulate BKCa channel activity in coronary artery smooth muscle cells. *Circulation Research*. 2000;86:897.
- 122. Hill S, Ganellin C, Timmerman H, Schwartz J, Shankley N, Young J, Schunack W, Levi R, Haas H. International union of pharmacology. XIII. Classification of histamine receptors. *Pharmacological Reviews*. 1997;49:253.
- 123. Amerongen GP, Draijer R, Vermeer MA, van Hinsbergh VWM. Transient and prolonged increase in endothelial permeability induced by histamine and thrombin: Role of protein kinases, calcium, and RhoA. *Circulation Research*. 1998;83:1115.
- 124. Kumar P, Shen Q, Pivetti CD, Lee ES, Wu MH, Yuan SY. Molecular mechanisms of endothelial hyperpermeability: Implications in inflammation. *Expert Reviews in Molecular Medicine*. 2009;11:e19.
- 125. Trebak M, Ginnan R, Singer HA, Jourd'heuil D. Interplay between calcium and reactive oxygen/nitrogen species: An essential paradigm for vascular smooth muscle signaling. *Antioxidants & Redox Signaling*. 2010;12:657–674.
- 126. Machaca K. Ca²⁺ signaling, genes and the cell cycle. *Cell calcium*. 2011;49:323–330.
- 127. Vig M, Kinet JP. Calcium signaling in immune cells. Nature Immunology. 2009;10:21-27.
- 128. Csordás G, Hajnóczky G. SR/ER–mitochondrial local communication: Calcium and ROS. *Biochimica et Biophysica Acta (BBA)-Bioenergetics*. 2009;1787:1352–1362.
- 129. Clapham DE. Calcium signaling. Cell. 2007;131:1047-1058.
- 130. Berridge MJ, Lipp P, Bootman MD. The versatility and universality of calcium signalling. *Nature Reviews Molecular Cell Biology*. 2000;1:11–21.
- 131. Bers DM. Calcium cycling and signaling in cardiac myocytes. *Annual Review of Physiology*. 2008;70:23–49.
- 132. Sprague AH, Khalil RA. Inflammatory cytokines in vascular dysfunction and vascular disease. *Biochemical Pharmacology*. 2009;78:539–552.
- 133. Hu Y, Böck G, Wick G, Xu Q. Activation of PDGF receptor α in vascular smooth muscle cells by mechanical stress. *The FASEB Journal*. 1998;12:1135–1142.
- Boo YC, Jo H. Flow-dependent regulation of endothelial nitric oxide synthase: Role of protein kinases. American Journal of Physiology-Cell Physiology. 2003;285:C499–C508.
- 135. Shyy YJ, Hsieh HJ, Usami S, Chien S. Fluid shear stress induces a biphasic response of human monocyte chemotactic protein 1 gene expression in vascular endothelium. *Proceedings of the National Academy of Sciences*. 1994;91:4678.
- 136. Bao X, Lu C, Frangos JA. Temporal gradient in shear but not steady shear stress induces PDGF-a and MCP-1 expression in endothelial cells: Role of NO, NFκβ, and EGR-1. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1999;19:996–1003.
- 137. Van Thienen JV, Fledderus JO, Dekker RJ, Rohlena J, Van IJzendoorn GA, Kootstra NA, Pannekoek H, Horrevoets AJG. Shear stress sustains atheroprotective endothelial KLF2 expression more potently than statins through mRNA stabilization. *Cardiovascular Research*. 2006;72:231–240.
- Dekker RJ, Boon RA, Rondaij MG, Kragt A, Volger OL, Elderkamp YW, Meijers JCM, Voorberg J, Pannekoek H, Horrevoets AJG. KLF2 provokes a gene expression pattern that establishes functional quiescent differentiation of the endothelium. *Blood*. 2006;107:4354–4363.
- 139. Pan S. Molecular mechanisms responsible for the atheroprotective effects of laminar shear stress. *Antioxidants & Redox Signaling*. 2009;11:1669–1682.

- 140. Yan C, Huang A, Kaley G, Sun D. Chronic high blood flow potentiates shear stressinduced release of NO in arteries of aged rats. *American Journal of Physiology-Heart* and Circulatory Physiology. 2007;293:H3105–H3110.
- 141. Buczynski MW, Dumlao DS, Dennis EA. Thematic review series: Proteomics. An integrated omics analysis of eicosanoid biology. *Journal of Lipid Research*. 2009;50:1015–1038.
- 142. Sampey AV, Monrad S, Crofford LJ. Microsomal prostaglandin E synthase-1: The inducible synthase for prostaglandin E 2. *Arthritis Research & Therapy*. 2005;7:1–4.
- 143. Serhan CN, Savill J. Resolution of inflammation: The beginning programs the end. *Nature Immunology*. 2005;6:1191–1197.
- 144. Serhan CN, Yacoubian S, Yang R. Anti-inflammatory and pro-resolving lipid mediators. Annual Review of Pathology. 2008;3:279.
- 145. Chavis C, Chanez P, Vachier I, Bousquet J, Michel F, Godard P. 5-15-Dihete and lipoxins generated by neutrophils from endogenous arachidonic acid as asthma biomarkers. *Biochemical and Biophysical Research Communications*. 1995;207:273–279.
- 146. Thomas E, Leroux J, Blotman F, Chavis C. Conversion of endogenous arachidonic acid to 5, 15-dihete and lipoxins by polymorphonuclear cells from patients with rheumatoid arthritis. *Inflammation Research*. 1995;44:121–124.
- 147. Calder PC. Polyunsaturated fatty acids and inflammatory processes: New twists in an old tale. *Biochimie*. 2009;91:791–795.
- Bannenberg G, Serhan CN. Specialized pro-resolving lipid mediators in the inflammatory response: An update. *Biochimica et Biophysica Acta (BBA)-Molecular and Cell Biology of Lipids*. 2010;1801:1260–1273.
- Serhan C, Chiang N. Endogenous pro-resolving and anti-inflammatory lipid mediators: A new pharmacologic genus. *British Journal of Pharmacology*. 2008;153:S200–S215.
- Serhan CN, Chiang N, Van Dyke TE. Resolving inflammation: Dual anti-inflammatory and pro-resolution lipid mediators. *Nature Reviews Immunology*. 2008;8:349–361.
- 151. Serhan CN, Clish CB, Brannon J, Colgan SP, Chiang N, Gronert K. Novel functional sets of lipid-derived mediators with antiinflammatory actions generated from omega-3 fatty acids via cyclooxygenase 2–nonsteroidal antiinflammatory drugs and transcellular processing. *The Journal of Experimental Medicine*. 2000;192:1197.
- 152. Hong S, Gronert K, Devchand PR, Moussignac RL, Serhan CN. Novel docosatrienes and 17S-resolvins generated from docosahexaenoic acid in murine brain, human blood, and glial cells. *Journal of Biological Chemistry*. 2003;278:14677.
- 153. Serhan CN, Krishnamoorthy S, Recchiuti A, Chiang N. Novel anti-inflammatory pro-resolving mediators and their receptors. *Current Topics in Medicinal Chemistry*. 2011;11:629.
- 154. Wu SH, Liao PY, Dong L, Chen ZQ. Signal pathway involved in inhibition by lipoxin A 4 of production of interleukins induced in endothelial cells by lipopolysaccharide. *Inflammation Research.* 2008;57:430–437.
- 155. Chiang N, Serhan CN, Dahlén SE, Drazen JM, Hay DWP, Rovati GE, Shimizu T, Yokomizo T, Brink C. The lipoxin receptor ALX: Potent ligand-specific and stereoselective actions in vivo. *Pharmacological Reviews*. 2006;58:463–487.
- 156. Kapoor M, Kojima F, Yang L, Crofford LJ. Sequential induction of pro-and antiinflammatory prostaglandins and peroxisome proliferators-activated receptor-gamma during normal wound healing: A time course study. *Prostaglandins, Leukotrienes and Essential Fatty Acids*. 2007;76:103–112.
- 157. Appel S, Mirakaj V, Bringmann A, Weck MM, Grünebach F, Brossart P. PPAR-γ agonists inhibit toll-like receptor-mediated activation of dendritic cells via the MAP kinase and NFκβ pathways. *Blood.* 2005;106:3888.
- 158. Lawrence T, Gilroy DW. Chronic inflammation: A failure of resolution? *International Journal of Experimental Pathology*. 2007;88:85–94.

- 159. Gilroy DW, Lawrence T, Perretti M, Rossi AG. Inflammatory resolution: New opportunities for drug discovery. *Nature Reviews Drug Discovery*. 2004;3:401–416.
- 160. Udalova IA, Richardson A, Denys A, Smith C, Ackerman H, Foxwell B, Kwiatkowski D. Functional consequences of a polymorphism affecting NFκβ p50-p50 binding to the TNF promoter region. *Molecular and Cellular Biology*. 2000;20:9113.
- 161. Ricart KC, Bolisetty S, Johnson MS, Perez J, Agarwal A, Murphy MP, Landar A. The permissive role of mitochondria in the induction of haeme oxygenase-1 in endothelial cells. *The Biochemical Journal*. 2009;419:427.
- 162. Kobayashi M, Li L, Iwamoto N, Nakajima-Takagi Y, Kaneko H, Nakayama Y, Eguchi M, Wada Y, Kumagai Y, Yamamoto M. The antioxidant defense system Keap1-Nrf2 comprises a multiple sensing mechanism for responding to a wide range of chemical compounds. *Molecular and Cellular Biology*. 2009;29:493.
- 163. Keum YS, Chang PPJ, Kwon KH, Yuan X, Li W, Hu L, Kong ANT. 3-Morpholinopropyl isothiocyanate is a novel synthetic isothiocyanate that strongly induces the antioxidant response element-dependent Nrf2-mediated detoxifying/antioxidant enzymes in vitro and in vivo. *Carcinogenesis*. 2008;29:594.
- 164. Biswas M, Chan JY. Role of Nrf1 in antioxidant response element-mediated gene expression and beyond. *Toxicology and Applied Pharmacology*. 2010;244:16–20.
- 165. Kwak MK, Wakabayashi N, Kensler TW. Chemoprevention through the Keap1-Nrf2 signaling pathway by phase 2 enzyme inducers. *Mutation Research/Fundamental and Molecular Mechanisms of Mutagenesis*. 2004;555:133–148.
- 166. Funk CD. Prostaglandins and leukotrienes: Advances in eicosanoid biology. *Science*. 2001;294:1871.
- 167. Wu KK, Liou JY. Cellular and molecular biology of prostacyclin synthase. *Biochemical and Biophysical Research Communications*. 2005;338:45–52.
- 168. Rudberg PC, Tholander F, Andberg M, Thunnissen MMGM, Haeggström JZ. Leukotriene a4 hydrolase. *Journal of Biological Chemistry*. 2004;279:27376.
- Wymann MP, Schneiter R. Lipid signalling in disease. Nature Reviews Molecular Cell Biology. 2008;9:162–176.
- 170. Febbraio M, Hajjar DP, Silverstein RL. Cd36: A class B scavenger receptor involved in angiogenesis, atherosclerosis, inflammation, and lipid metabolism. *Journal of Clinical Investigation*. 2001;108:785–792.
- 171. Kunjathoor VV, Febbraio M, Podrez EA, Moore KJ, Andersson L, Koehn S, Rhee JS, Silverstein R, Hoff HF, Freeman MW. Scavenger receptors class AI/II and CD36 are the principal receptors responsible for the uptake of modified low density lipoprotein leading to lipid loading in macrophages. *Journal of Biological Chemistry*. 2002;277:49982–49988.
- 172. Greaves DR, Gordon S. The macrophage scavenger receptor at 30 years of age: Current knowledge and future challenges. *Journal of Lipid Research*. 2009;50:S282–S286.
- 173. Moore KJ, Freeman MW. Scavenger receptors in atherosclerosis. Arteriosclerosis, Thrombosis, and Vascular Biology. 2006;26:1702–1711.
- 174. Baranova IN, Vishnyakova TG, Bocharov AV, Leelahavanichkul A, Kurlander R, Chen Z, Souza ACP, Yuen PST, Star RA, Csako G. Class B scavenger receptor types I and II and CD36 mediate bacterial recognition and proinflammatory signaling induced by *Escherichia coli*, lipopolysaccharide, and cytosolic chaperonin 60. *The Journal of Immunology*. 2012;188:1371–1380.
- 175. Huang W, Febbraio M, Silverstein RL. CD9 tetraspanin interacts with CD36 on the surface of macrophages: A possible regulatory influence on uptake of oxidized low density lipoprotein. *PloS One*. 2011;6:e29092.
- 176. Silverstein RL, Febbraio M. CD36, a scavenger receptor involved in immunity, metabolism, angiogenesis, and behavior. *Science's STKE*. 2009;2:re3.

- 177. Hodgkinson CP, Laxton RC, Patel K, Ye S. Advanced glycation end-product of low density lipoprotein activates the toll-like 4 receptor pathway implications for diabetic atherosclerosis. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2008;28:2275–2281.
- Hodgkinson C, Ye S. Toll-like receptors, their ligands, and atherosclerosis. *Scientific World Journal*. 2011;11:437–453.
- 179. Doran AC, Meller N, McNamara CA. Role of smooth muscle cells in the initiation and early progression of atherosclerosis. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2008;28:812–819.
- 180. Ferrières J. The French paradox: Lessons for other countries. Heart. 2004;90:107-111.
- 181. Ducimetière P, Lang T, Amouyel P, Arveiler D, Ferrières J. Why mortality from heart disease is low in France: Rates of coronary events are similar in France and southern Europe. *BMJ: British Medical Journal*. 2000;320:249.
- 182. Müller-Nordhorn J, Binting S, Roll S, Willich SN. An update on regional variation in cardiovascular mortality within Europe. *European Heart Journal*. 2008;29:1316–1326.
- 183. Renaud S, Gueguen R. The French paradox and wine drinking. *Alcohol and Cardiovascular Disease*. 2008:208.
- 184. Stary HC. Natural history and histological classification of atherosclerotic lesions: An update. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2000;20:1177–1178.
- 185. Stary HC, Chandler AB, Dinsmore RE, Fuster V, Glagov S, Insull W, Jr., Rosenfeld ME, Schwartz CJ, Wagner WD, Wissler RW. A definition of advanced types of atherosclerotic lesions and a histological classification of atherosclerosis: A report from the committee on vascular lesions of the Council on Arteriosclerosis, American Heart Association. *Circulation*. 1995;92:1355.
- 186. May AE, Seizer P, Gawaz M. Platelets: Inflammatory firebugs of vascular walls. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2008;28:s5–s10.
- Borissoff JI, Spronk H, Ten Cate H. The hemostatic system as a modulator of atherosclerosis. *The New England Journal of Medicine*. 2011;364:1746–1760.
- 188. Lindemann S, Krämer B, Seizer P, Gawaz M. Platelets, inflammation and atherosclerosis. *Journal of Thrombosis and Haemostasis*. 2007;5:203–211.
- Hansson GK, Libby P. The immune response in atherosclerosis: A double-edged sword. Nature Reviews Immunology. 2006;6:508–519.
- 190. Lenfant C. Clinical research to clinical practice—lost in translation? *New England Journal of Medicine*. 2003;349:868–874.
- 191. Law MR, Wald NJ, Rudnicka A. Quantifying effect of statins on low density lipoprotein cholesterol, ischaemic heart disease, and stroke: Systematic review and meta-analysis. *British Medical Journal*. 2003;326:1423.
- 192. Sever PS, Dahlöf B, Poulter NR, Wedel H, Beevers G, Caulfield M, Collins R, Kjeldsen SE, Kristinsson A, McInnes GT. Prevention of coronary and stroke events with atorvastatin in hypertensive patients who have average or lower-than-average cholesterol concentrations, in the Anglo-Scandinavian Cardiac Outcomes Trial—Lipid Lowering Arm (ASCOT-LLA): A multicentre randomised controlled trial. *The Lancet*. 2003;361:1149–1158.
- 193. Baigent C, Keech A, Kearney PM, Blackwell L, Buck G, Pollicino C, Kirby A, Sourjina T, Peto R, Collins R. Cholesterol treatment trialists'(CTT) collaborators. Efficacy and safety of cholesterol-lowering treatment: Prospective meta-analysis of data from 90,056 participants in 14 randomised trials of statins. *Lancet*. 2005;366:1267–1278.
- 194. Barter P, Gotto AM, LaRosa JC, Maroni J, Szarek M, Grundy SM, Kastelein JJP, Bittner V, Fruchart JC. HDL cholesterol, very low levels of LDL cholesterol, and cardiovascular events. *New England Journal of Medicine*. 2007;357:1301–1310.
- 195. Paumelle R, Blanquart C, Briand O, Barbier O, Duhem C, Woerly G, Percevault F, Fruchart JC, Dombrowicz D, Glineur C. Acute antiinflammatory properties of statins

involve peroxisome proliferator–activated receptor- α via inhibition of the protein kinase C signaling pathway. *Circulation Research*. 2006;98:361–369.

- 196. Bu D, Griffin G, Lichtman AH. Mechanisms for the anti-inflammatory effects of statins. *Current Opinion in Lipidology*. 2011;22:165.
- 197. Hashim SA, Van Itallie TB. Cholestyramine resin therapy for hypercholesteremia. *JAMA: The Journal of the American Medical Association*. 1965;192:289–293.
- 198. Mabuchi H, Sakai T, Sakai Y, Yoshimura A, Watanabe A, Wakasugi T, Koizumi J, Takeda R. Reduction of serum cholesterol in heterozygous patients with familial hypercholesterolemia. *New England Journal of Medicine*. 1983;308:609–613.
- 199. Brensike JF, Levy RI, Kelsey SF, Passamani ER, Richardson JM, Loh IK, Stone NJ, Aldrich RF, Battaglini JW, Moriarty DJ. Effects of therapy with cholestyramine on progression of coronary arteriosclerosis: Results of the NHLBI type II Coronary Intervention Study. *Circulation*. 1984;69:313–324.
- 200. Levy R, Brensike J, Epstein S, Kelsey S, Passamani E, Richardson J, Loh I, Stone N, Aldrich R, Battaglini J. The influence of changes in lipid values induced by cholestyramine and diet on progression of coronary artery disease: Results of NHLBI type II Coronary Intervention Study. *Circulation*. 1984;69:325–337.
- Berliner JA, Navab M, Fogelman AM, Frank JS, Demer LL, Edwards PA, Watson AD, Lusis AJ. Atherosclerosis: Basic mechanisms: Oxidation, inflammation, and genetics. *Circulation*. 1995;91:2488.
- 202. Wang N, Lan D, Chen W, Matsuura F, Tall AR. ATP-binding cassette transporters G1 and G4 mediate cellular cholesterol efflux to high-density lipoproteins. *Proceedings of the National Academy of Sciences of the United States of America*. 2004;101:9774.
- 203. Stein O, Stein Y. Atheroprotective mechanisms of HDL. Atherosclerosis. 1999;144:285-301.
- 204. Barter PJ, Nicholls S, Rye KA, Anantharamaiah G, Navab M, Fogelman AM. Antiinflammatory properties of HDL. *Circulation Research*. 2004;95:764–772.
- 205. Brewer HB, Jr. The evolving role of HDL in the treatment of high-risk patients with cardiovascular disease. *Journal of Clinical Endocrinology & Metabolism*. 2011;96:1246–1257.
- 206. Redondo S, Martínez-González J, Urraca C, Tejerina T. Emerging therapeutic strategies to enhance HDL function. *Lipids in Health and Disease*. 2011;10:175.
- 207. Davidson MH. HDL and CETP inhibition: Will this define the future? *Current Treatment Options in Cardiovascular Medicine*. 2012;14:384–390.
- 208. Cannon CP, Shah S, Dansky HM, Davidson M, Brinton EA, Gotto AM, Jr., Stepanavage M, Liu SX, Gibbons P, Ashraf TB. Safety of anacetrapib in patients with or at high risk for coronary heart disease. *New England Journal of Medicine*. 2010;363:2406–2415.
- Stein EA, Stroes ESG, Steiner G, Buckley BM, Capponi AM, Burgess T, Niesor EJ, Kallend D, Kastelein JJP. Safety and tolerability of dalcetrapib. *The American Journal* of Cardiology. 2009;104:82–91.
- 210. Fayad ZA, Mani V, Woodward M, Kallend D, Abt M, Burgess T, Fuster V, Ballantyne CM, Stein EA, Tardif JC. Safety and efficacy of dalcetrapib on atherosclerotic disease using novel non-invasive multimodality imaging (dal-plaque): A randomised clinical trial. *The Lancet*. 2011;378:1547–1559.
- 211. Voight BF, Peloso GM, Orho-Melander M, Frikke-Schmidt R, Barbalic M, Jensen MK, Hindy G, Hólm H, Ding EL, Johnson T. Plasma HDL cholesterol and risk of myocardial infarction: A Mendelian randomisation study. *The Lancet*. 2012;380:572–580.
- Undurti A, Huang Y, Lupica JA, Smith JD, DiDonato JA, Hazen SL. Modification of high density lipoprotein by myeloperoxidase generates a pro-inflammatory particle. *Journal of Biological Chemistry*. 2009;284:30825.
- Thorne RF, Mhaidat NM, Ralston KJ, Burns GF. CD36 is a receptor for oxidized high density lipoprotein: Implications for the development of atherosclerosis. *FEBS Letters*. 2007;581:1227–1232.

- Calvo D, Gómez-Coronado D, Suárez Y, Lasunción MA, Vega MA. Human CD36 is a high affinity receptor for the native lipoproteins HDL, LDL, and VLDL. *Journal of Lipid Research*. 1998;39:777–788.
- Assinger A, Koller F, Schmid W, Zellner M, Babeluk R, Koller E, Volf I. Specific binding of hypochlorite-oxidized HDL to platelet CD36 triggers proinflammatory and procoagulant effects. *Atherosclerosis*. 2010;212:153–160.
- 216. Ansell BJ, Navab M, Hama S, Kamranpour N, Fonarow G, Hough G, Rahmani S et al. Inflammatory/antiinflammatory properties of high-density lipoprotein distinguish patients from control subjects better than high-density lipoprotein cholesterol levels and are favorably affected by simvastatin treatment. *Circulation*. 2003;108:2751–2756.
- 217. Van Lenten BJ, Hama SY, de Beer FC, Stafforini DM, McIntyre TM, Prescott SM, La Du BN, Fogelman AM, Navab M. Anti-inflammatory HDL becomes pro-inflammatory during the acute phase response. Loss of protective effect of HDL against LDL oxidation in aortic wall cell cocultures. *The Journal of Clinical Investigation*. 1995;96:2758–2767.
- 218. Navab M, Hama SY, Ready ST, Ng CJ, Van Lenten BJ, Laks H, Fogelman AM. Oxidized lipids as mediators of coronary heart disease. *Current Opinion in Lipidology*. 2002;13:363–372.
- Navab M, Van Lenten BJ, Reddy ST, Fogelman AM. High-density lipoprotein and the dynamics of atherosclerotic lesions. *Circulation*. 2001;104:2386–2387.
- 220. Navab M, Hama SY, Anantharamaiah GM, Hassan K, Hough GP, Watson AD, Reddy ST, Sevanian A, Fonarow GC, Fogelman AM. Normal high density lipoprotein inhibits three steps in the formation of mildly oxidized low density lipoprotein: Steps 2 and 3. *Journal* of Lipid Research. 2000;41:1495–1508.
- 221. Roberts CK, Won D, Pruthi S, Kurtovic S, Sindhu RK, Vaziri ND, Barnard RJ. Effect of a short-term diet and exercise intervention on oxidative stress, inflammation, MMP-9, and monocyte chemotactic activity in men with metabolic syndrome factors. *Journal of Applied Physiology*. 2006;100:1657.
- 222. Roberts CK, Ng C, Hama S, Eliseo AJ, Barnard RJ. Effect of a short-term diet and exercise intervention on inflammatory/anti-inflammatory properties of HDL in overweight/obese men with cardiovascular risk factors. *Journal of Applied Physiology*. 2006;101:1727–1732.
- 223. Fu MX, Requena JR, Jenkins AJ, Lyons TJ, Baynes JW, Thorpe SR. The advanced glycation end product, *n*-(carboxymethyl) lysine, is a product of both lipid peroxidation and glycoxidation reactions. *Journal of Biological Chemistry*. 1996;271:9982–9986.
- 224. Requena JR, Ahmed MU, Fountain CW, Degenhardt TP, Reddy S, Perez C, Lyons TJ, Jenkins AJ, Baynes JW, Thorpe SR. Carboxymethylethanolamine, a biomarker of phospholipid modification during the Maillard reaction in vivo. *Journal of Biological Chemistry*. 1997;272:17473–17479.
- 225. Thornalley PJ, Langborg A, Minhas HS. Formation of glyoxal, methylglyoxal and 3-deoxyglucosone in the glycation of proteins by glucose. *Biochemical Journal*. 1999;344:109.
- 226. Ge SJ, Lee TC. Kinetic significance of the Schiff base reversion in the early-stage Maillard reaction of a phenylalanine-glucose aqueous model system. *Journal of Agricultural and Food Chemistry*. 1997;45:1619–1623.
- 227. Oak JH, Nakagawa K, Miyazawa T. Synthetically prepared Amadori-glycated phosphatidylethanolamine can trigger lipid peroxidation via free radical reactions. *FEBS Letters*. 2000;481:26–30.
- Tappy L, Lê KA. Metabolic effects of fructose and the worldwide increase in obesity. *Physiological Reviews*. 2010;90:23–46.
- 229. Laville M, Nazare JA. Diabetes, insulin resistance and sugars. *Obesity Reviews*. 2009;10:24–33.

- Bremer AA, Auinger P, Byrd RS. Sugar-sweetened beverage intake trends in us adolescents and their association with insulin resistance-related parameters. *Journal of Nutrition and Metabolism*. 2010;196476:1–8.
- 231. Welsh JA, Sharma A, Cunningham SA, Vos MB. Consumption of added sugars and indicators of cardiovascular disease risk among US adolescents clinical perspective. *Circulation*. 2011;123:249–257.
- Johnson RK, Appel LJ, Brands M, Howard BV, Lefevre M, Lustig RH, Sacks F, Steffen LM, Wylie-Rosett J. Dietary sugars intake and cardiovascular health. *Circulation*. 2009;120:1011–1020.
- 233. Malik VS, Popkin BM, Bray GA, Després JP, Hu FB. Sugar-sweetened beverages, obesity, type 2 diabetes mellitus, and cardiovascular disease risk. *Circulation*. 2010;121:1356–1364.
- Malik VS, Popkin BM, Bray GA, Després JP, Willett WC, Hu FB. Sugar-sweetened beverages and risk of metabolic syndrome and type 2 diabetes. *Diabetes Care*. 2010;33:2477–2483.
- 235. Hu FB, Malik VS. Sugar-sweetened beverages and risk of obesity and type 2 diabetes: Epidemiologic evidence. *Physiology & Behavior*. 2010;100:47–54.
- 236. Bravata DM, Sanders L, Huang J, Krumholz HM, Olkin I, Gardner CD, Bravata DM. Efficacy and safety of low-carbohydrate diets. *JAMA: The Journal of the American Medical Association*. 2003;289:1837–1850.
- 237. Lau C, Færch K, Glümer C, Tetens I, Pedersen O, Carstensen B, Jørgensen T, Borch-Johnsen K. Dietary glycemic index, glycemic load, fiber, simple sugars, and insulin resistance. *Diabetes Care*. 2005;28:1397–1403.
- 238. Stanhope KL, Havel PJ. Fructose consumption: Considerations for future research on its effects on adipose distribution, lipid metabolism, and insulin sensitivity in humans. *The Journal of Nutrition*. 2009;139:1236S–1241S.
- 239. Hamburg NM, McMackin CJ, Huang AL, Shenouda SM, Widlansky ME, Schulz E, Gokce N, Ruderman NB, Keaney JF, Jr., Vita JA. Physical inactivity rapidly induces insulin resistance and microvascular dysfunction in healthy volunteers. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2007;27:2650–2656.
- 240. Tremblay MSTMS, Colley RCCRC, Saunders TJSTJ, Healy GNHGN, Owen NON. Physiological and health implications of a sedentary lifestyle. *Applied Physiology, Nutrition, and Metabolism.* 2010;35:725–740.
- 241. Thijssen DHJ, Maiorana AJ, O'Driscoll G, Cable NT, Hopman MTE, Green DJ. Impact of inactivity and exercise on the vasculature in humans. *European Journal of Applied Physiology*. 2010;108:845–875.
- 242. Seider MJ, Nicholson WF, Booth FW. Insulin resistance for glucose metabolism in disused soleus muscle of mice. *American Journal of Physiology-Endocrinology and Metabolism*. 1982;242:E12–E18.
- Booth FW, Gordon SE, Carlson CJ, Hamilton MT. Waging war on modern chronic diseases: Primary prevention through exercise biology. *Journal of Applied Physiology*. 2000;88:774–787.
- 244. Booth FW, Laye MJ, Lees SJ, Rector RS, Thyfault JP. Reduced physical activity and risk of chronic disease: The biology behind the consequences. *European Journal of Applied Physiology*. 2008;102:381–390.
- 245. Booth FW, Lees SJ. Fundamental questions about genes, inactivity, and chronic diseases. *Physiological Genomics*. 2007;28:146–157.
- 246. Leesa SJ, Booth FW. Physical inactivity is a disease. *World Review of Nutrition and Dietetics*. 2005;95:73–79.
- 247. Tresierras MA, Balady GJ. Resistance training in the treatment of diabetes and obesity: Mechanisms and outcomes. *Journal of Cardiopulmonary Rehabilitation and Prevention*. 2009;29:67.

- 248. Coker RH, Williams RH, Yeo SE, Kortebein PM, Bodenner DL, Kern PA, Evans WJ. The impact of exercise training compared to caloric restriction on hepatic and peripheral insulin resistance in obesity. *Journal of Clinical Endocrinology & Metabolism*. 2009;94:4258–4266.
- Bell LM, Watts K, Siafarikas A, Thompson A, Ratnam N, Bulsara M, Finn J, O'Driscoll G, Green DJ, Jones TW. Exercise alone reduces insulin resistance in obese children independently of changes in body composition. *Journal of Clinical Endocrinology & Metabolism*. 2007;92:4230–4235.
- 250. Ferguson M, Gutin B, Le N, Karp W, Litaker M, Humphries M, Okuyama T, Riggs S, Owens S. Effects of exercise training and its cessation on components of the insulin resistance syndrome in obese children. *International Journal of Obesity and Related Metabolic Disorders: Journal of the International Association for the Study of Obesity*. 1999;23:889.
- 251. Ross R, Janssen I, Dawson J, Kungl AM, Kuk JL, Wong SL, Nguyen-Duy TB, Lee SJ, Kilpatrick K, Hudson R. Exercise-induced reduction in obesity and insulin resistance in women: A randomized controlled trial. *Obesity*. 2004;12:789–798.
- 252. Björntorp P, de Jounge K, Sjöström L, Sullivan L. The effect of physical training on insulin production in obesity. *Metabolism.* 1970;19:631–638.
- 253. Dandona P, Aljada A, Chaudhuri A, Mohanty P, Garg R. Metabolic syndrome: A comprehensive perspective based on interactions between obesity, diabetes, and inflammation. *Circulation*. 2005;111:1448.
- 254. Lund Håheim L, Wisløff T, Holme I, Nafstad P. Metabolic syndrome predicts prostate cancer in a cohort of middle-aged Norwegian men followed for 27 years. *American Journal of Epidemiology*. 2006;164:769.
- 255. Rana J, Nieuwdorp M, Jukema J, Kastelein J. Cardiovascular metabolic syndrome an interplay of obesity, inflammation, diabetes and coronary heart disease. *Diabetes, Obesity and Metabolism.* 2007;9:218–232.
- 256. Iyer A, Fairlie DP, Prins JB, Hammock BD, Brown L. Inflammatory lipid mediators in adipocyte function and obesity. *Nature Reviews Endocrinology*. 2010;6:71–82.
- 257. Monteiro R, Azevedo I, Frühbeck G. Chronic inflammation in obesity and the metabolic syndrome. *Mediators of Inflammation*. 2010;2010:116.
- González-Périz A, Clària J. Resolution of adipose tissue inflammation. Scientific World Journal. 2010;10:832–856.
- 259. Anderson EK, Gutierrez DA, Hasty AH. Adipose tissue recruitment of leukocytes. *Current Opinion in Lipidology*. 2010;21:172.
- 260. Bastard JP, Maachi M, Lagathu C, Kim MJ, Caron M, Vidal H, Capeau J, Feve B. Recent advances in the relationship between obesity, inflammation, and insulin resistance. *European Cytokine Network*. 2006;17:4–12.
- 261. Ikeoka D, Mader JK, Pieber TR. Adipose tissue, inflammation and cardiovascular disease. *Revista da Associação Médica Brasileira*. 2010;56:116–121.
- 262. de Ferranti S, Mozaffarian D. The perfect storm: Obesity, adipocyte dysfunction, and metabolic consequences. *Clinical Chemistry*. 2008;54:945–955.
- 263. El-Jack AK, Hamm JK, Pilch PF, Farmer SR. Reconstitution of insulin-sensitive glucose transport in fibroblasts requires expression of both PPARγ and C/EBPα. *Journal of Biological Chemistry*. 1999;274:7946–7951.
- Ntambi JM, Young-Cheul K. Adipocyte differentiation and gene expression. *The Journal* of Nutrition. 2000;130:3122S–3126S.
- 265. Hajer GR, Van Haeften TW, Visseren FLJ. Adipose tissue dysfunction in obesity, diabetes, and vascular diseases. *European Heart Journal*. 2008;29:2959–2971.
- 266. Brockman DA, Chen X. Proteomics in the characterization of adipose dysfunction in obesity. *Adipocyte*. 2012;1:0–1.

- Gregor MF, Yang L, Fabbrini E, Mohammed BS, Eagon JC, Hotamisligil GS, Klein S. Endoplasmic reticulum stress is reduced in tissues of obese subjects after weight loss. *Diabetes*. 2009;58:693–700.
- 268. Sun K, Kusminski CM, Scherer PE. Adipose tissue remodeling and obesity. *The Journal* of Clinical Investigation. 2011;121:2094.
- 269. Gregoire FM. Adipocyte differentiation: From fibroblast to endocrine cell. *Experimental Biology and Medicine*. 2001;226:997–1002.
- 270. Harman-Boehm I, Blüher M, Redel H, Sion-Vardy N, Ovadia S, Avinoach E, Shai I, Klöting N, Stumvoll M, Bashan N. Macrophage infiltration into omental versus subcutaneous fat across different populations: Effect of regional adiposity and the comorbidities of obesity. *Journal of Clinical Endocrinology & Metabolism*. 2007;92:2240–2247.
- Fontana L, Eagon JC, Trujillo ME, Scherer PE, Klein S. Visceral fat adipokine secretion is associated with systemic inflammation in obese humans. *Diabetes*. 2007;56:1010–1013.
- 272. Mahabadi AA, Massaro JM, Rosito GA, Levy D, Murabito JM, Wolf PA, O'Donnell CJ, Fox CS, Hoffmann U. Association of pericardial fat, intrathoracic fat, and visceral abdominal fat with cardiovascular disease burden: The Framingham Heart Study. *European Heart Journal*. 2009;30:850.
- Katzmarzyk PT, Church TS, Janssen I, Ross R, Blair SN. Metabolic syndrome, obesity, and mortality. *Diabetes Care*. 2005;28:391–397.
- 274. Calori G, Lattuada G, Piemonti L, Garancini MP, Ragogna F, Villa M, Mannino S, Crosignani P, Bosi E, Luzi L. Prevalence, metabolic features, and prognosis of metabolically healthy obese Italian individuals. *Diabetes Care*. 2011;34:210–215.
- 275. Barbarroja N, Lopez-Pedrera R, Mayas M, Garcia-Fuentes E, Garrido-Sanchez L, Macias-Gonzalez M, El Bekay R, Vidal-Puig A, Tinahones F. The obese healthy paradox: Is inflammation the answer? *The Biochemical Journal*. 2010;430:141–149.
- 276. Wildman RP, Muntner P, Reynolds K, McGinn AP, Rajpathak S, Wylie-Rosett J, Sowers MFR. The obese without cardiometabolic risk factor clustering and the normal weight with cardiometabolic risk factor clustering: Prevalence and correlates of 2 phenotypes among the US population (NHANES 1999–2004). Archives of Internal Medicine. 2008;168:1617.
- 277. Karelis AD, Faraj M, Bastard JP, St-Pierre DH, Brochu M, Prud'homme D, Rabasa-Lhoret R. The metabolically healthy but obese individual presents a favorable inflammation profile. *Journal of Clinical Endocrinology & Metabolism.* 2005;90:4145–4150.
- 278. Gauthier M, Ruderman N. Adipose tissue inflammation and insulin resistance: All obese humans are not created equal. *The Biochemical Journal*. 2010;430:e1–e4.
- 279. Wiedermann CJ, Kiechl S, Dunzendorfer S, Schratzberger P, Egger G, Oberhollenzer F, Willeit J. Association of endotoxemia with carotid atherosclerosis and cardiovascular disease 1: Prospective results from the Bruneck study. *Journal of the American College* of Cardiology. 1999;34:1975–1981.
- 280. Szklo M, Ding J, Tsai MY, Cushman M, Polak JF, Lima J, Barr RG, Sharrett AR. Individual pathogens, pathogen burden, and markers of subclinical atherosclerosis: The multi-ethnic study of atherosclerosis. *Journal of Cardiovascular Medicine (Hagerstown, Md.)*. 2009;10:747.
- Elkaim R, Dahan M, Kocgozlu L, Werner S, Kanter D, Kretz J, Tenenbaum H. Prevalence of periodontal pathogens in subgingival lesions, atherosclerotic plaques and healthy blood vessels: A preliminary study. *Journal of Periodontal Research*. 2008;43:224–231.
- Xenaki E, Hassoulas J, Apostolakis S, Sourvinos G, Spandidos DA. Detection of cytomegalovirus in atherosclerotic plaques and nonatherosclerotic arteries. *Angiology*. 2009;60:504–508.
- Epstein SE, Zhou YF, Zhu J. Infection and atherosclerosis: Emerging mechanistic paradigms. *Circulation*. 1999;100:e20–e28.

- Epstein SE, Zhu J, Burnett MS, Zhou YF, Vercellotti G, Hajjar D. Infection and atherosclerosis: Potential roles of pathogen burden and molecular mimicry. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2000;20:1417–1420.
- Marcelino SL. Presence of periodontopathic bacteria in coronary arteries from patients with chronic periodontitis. *Anaerobe*. 2010;16:629–632.
- 286. Song Z, Brassard P, Brophy JM. A meta-analysis of antibiotic use for the secondary prevention of cardiovascular diseases. *The Canadian Journal of Cardiology*. 2008;24:391.
- 287. Doyle JT, Dawber TR, Kannel WB, Kinch SH, Kahn HA. The relationship of cigarette smoking to coronary heart disease. *JAMA: The Journal of the American Medical Association*. 1964;190:886–890.
- Ockene JK, Kuller LH, Svendsen KH, Meilahn E. The relationship of smoking cessation to coronary heart disease and lung cancer in the multiple risk factor intervention trial (MRFIT). *American Journal of Public Health*. 1990;80:954–958.
- Jonas MA, Oates J, Ockene J, Hennekens C. Statement on smoking and cardiovascular disease for health care professionals. American Heart Association. *Circulation*. 1992;86:1664–1669.
- 290. Castelli W. Epidemiology of coronary heart disease: The Framingham study. *The American Journal of Medicine*. 1984;76:4–12.
- 291. Hung J, Lam JYT, Lacoste L, Letchacovski G. Cigarette smoking acutely increases platelet thrombus formation in patients with coronary artery disease taking aspirin. *Circulation*. 1995;92:2432–2436.
- Nowak J, Murray JJ, Oates JA, FitzGerald GA. Biochemical evidence of a chronic abnormality in platelet and vascular function in healthy individuals who smoke cigarettes. *Circulation*. 1987;76:6–14.
- 293. Powell JT. Vascular damage from smoking: Disease mechanisms at the arterial wall. *Vascular Medicine*. 1998;3:21–28.
- Ambrose JA, Barua RS. The pathophysiology of cigarette smoking and cardiovascular disease: An update. *Journal of the American College of Cardiology*. 2004;43:1731–1737.
- 295. Lehr HA, Weyrich AS, Saetzler RK, Jurek A, Arfors KE, Zimmerman GA, Prescott SM, McIntyre TM. Vitamin C blocks inflammatory platelet-activating factor mimetics created by cigarette smoking. *Journal of Clinical Investigation*. 1997;99:2358.
- 296. Lehr HA, Frei B, Arfors KE. Vitamin C prevents cigarette smoke-induced leukocyte aggregation and adhesion to endothelium in vivo. *Proceedings of the National Academy of Sciences*. 1994;91:7688.
- Powers SK, Duarte J, Kavazis AN, Talbert EE. Reactive oxygen species are signalling molecules for skeletal muscle adaptation. *Experimental Physiology*. 2010;95:1–9.
- 298. Nathan C. Specificity of a third kind: Reactive oxygen and nitrogen intermediates in cell signaling. *Journal of Clinical Investigation*. 2003;111:769–778.
- 299. Thannickal VJ, Fanburg BL. Reactive oxygen species in cell signaling. *American Journal* of Physiology-Lung Cellular and Molecular Physiology. 2000;279:L1005–L1028.
- Morgan MJ, Liu Z. Crosstalk of reactive oxygen species and NF-κβ signaling. *Cell Research*. 2010;21:103–115.
- 301. Nordberg J, Arnér ESJ. Reactive oxygen species, antioxidants, and the mammalian thioredoxin system1. *Free Radical Biology and Medicine*. 2001;31:1287–1312.
- 302. Sun QA, Wu Y, Zappacosta F, Jeang KT, Lee BJ, Hatfield DL, Gladyshev VN. Redox regulation of cell signaling by selenocysteine in mammalian thioredoxin reductases. *Journal of Biological Chemistry*. 1999;274:24522.
- Winrow V, Winyard P, Morris C, Blake D. Free radicals in inflammation: Second messengers and mediators of tissue destruction. *British Medical Bulletin*. 1993;49:506.
- Finkel T. Signal transduction by reactive oxygen species. *The Journal of Cell Biology*. 2011;194:7.

- 305. Veal EA, Day AM, Morgan BA. Hydrogen peroxide sensing and signaling. *Molecular Cell*. 2007;26:1–14.
- 306. Halliwell B. Reactive species and antioxidants. Redox biology is a fundamental theme of aerobic life. *Plant Physiology*. 2006;141:312–322.
- 307. Reaven P, McPhillips J, Barrett-Connor E, Criqui M. Leisure time exercise and lipid and lipoprotein levels in an older population. *Journal of the American Geriatrics Society*. 1990;38:847.
- 308. Williams PT, Krauss RM, Wood PD, Lindgren FT, Giotas C, Vranizan KM. Lipoprotein subfractions of runners and sedentary men. *Metabolism*. 1986;35:45–52.
- 309. Pate RR, Pratt M, Blair SN, Haskell WL, Macera CA, Bouchard C, Buchner D, Ettinger W, Heath GW, King AC. Physical activity and public health. *JAMA: The Journal of the American Medical Association*. 1995;273:402–407.
- Lakka T, Salonen J. Physical activity and serum lipids: A cross-sectional population study in Eastern Finnish men. *American Journal of Epidemiology*. 1992;136:806–818.
- 311. Kokkinos PF, Holland JC, Pittaras AE, Narayan P, Dotson CO, Papademetriou V. Cardiorespiratory fitness and coronary heart disease risk factor association in women. *Journal of the American College of Cardiology*. 1995;26:358–364.
- Wood PD, Haskell WL, Stern MP, Lewis S, Perry C. Plasma lippoprotein distributions in male and female runners. *Annals of the New York Academy of Sciences*. 1977;301:748–763.
- 313. Martin RP, Haskell WL, Wood PD. Blood chemistry and lipid profiles of elite distance runners. *Annals of the New York Academy of Sciences*. 1977;301:346–360.
- Hartung GH, Foreyt JP, Mitchell RE, Vlasek I, Gotto AM, Jr. Relation of diet to highdensity-lipoprotein cholesterol in middle-aged marathon runners, joggers, and inactive men. *New England Journal of Medicine*. 1980;302:357–366.
- 315. Lehtonen A, Viikari J. Serum triglycerides and cholesterol and serum high-density lipoprotein cholesterol in highly physically active men. *Acta Medica Scandinavica*. 1978;204:111–114.
- 316. Drygas W, Jegler A, Kunski H. Study on threshold dose of physical activity in coronary heart disease prevention. Part I. Relationship between leisure time physical activity and coronary risk factors. *International Journal of Sports Medicine*. 1988;9:275–278.
- 317. Drygas W, Kostka T, Jegier A, Kunski H. Long-term effects of different physical activity levels on coronary heart disease risk factors in middle-aged men. *International Journal of Sports Medicine*. 2000;21:235–241.
- 318. Williams PT. Relationships of heart disease risk factors to exercise quantity and intensity. Archives of Internal Medicine. 1998;158:237.
- Durstine JL, Grandjean PW, Davis PG, Ferguson MA, Alderson NL, DuBose KD. Blood lipid and lipoprotein adaptations to exercise: A quantitative analysis. *Sports Medicine*. 2001;31:1033–1062.
- 320. Williams PT. Relationship of distance run per week to coronary heart disease risk factors in 8283 male runners: The national runners' health study. *Archives of Internal Medicine*. 1997;157:191.
- 321. Kelley GA, Kelley KS, Tran ZV. Walking, lipids, and lipoproteins: A meta-analysis of randomized controlled trials. *Preventive Medicine*. 2004;38:651–661.
- 322. Oscai LB, Patterson JA, Bogard DL, Beck RJ, Rothermel BL. Normalization of serum triglycerides and lipoprotein electrophoretic patterns by exercise. *The American Journal* of Cardiology. 1972;30:775–780.
- 323. Huttunen JK, Lansimies E, Voutilainen E, Ehnholm C, Hietanen E, Penttila I, Siitonen O, Rauramaa R. Effect of moderate physical exercise on serum lipoproteins. A controlled clinical trial with special reference to serum high-density lipoproteins. *Circulation*. 1979;60:1220–1229.

- 324. Ponjee GAE, Janssen EME, Hermans J, Van Wersch JWJ. Effects of long-term exercise of moderate intensity on anthropometric values and serum lipids and lipoproteins. *European Journal of Clinical Chemistry and Clinical Biochemistry*. 1995;33:121–126.
- 325. Hill JO, Thiel J, Heller PA, Markon C, Fletcher G, DiGirolamo M. Differences in effects of aerobic exercise training on blood lipids in men and women. *The American Journal* of Cardiology. 1989;63:254–256.
- 326. Despres J, Pouliot M, Moorjani S, Nadeau A, Tremblay A, Lupien P, Theriault G, Bouchard C. Loss of abdominal fat and metabolic response to exercise training in obese women. *American Journal of Physiology-Endocrinology and Metabolism*. 1991;261:E159–E167.
- Vial R, Balart L, Arroyave G. Effect of exercise and physical fitness on serum lipids and lipoproteins. *Atherosclerosis*. 1974;20:1–9.
- Nye ER, Carlson K, Kirstein P, Rossner S. Changes in high density lipoprotein subfractions and other lipoproteins induced by exercise. *Clinica Chimica Acta*. 1981;113:51–57.
- 329. Wood PD, Stefanick ML, Dreon DM, Frey-Hewitt B, Garay SC, Williams PT, Superko HR, Fortmann SP, Albers JJ, Vranizan KM. Changes in plasma lipids and lipoproteins in overweight men during weight loss through dieting as compared with exercise. *New England Journal of Medicine*. 1988;319:1173–1179.
- 330. Grandjean P, Oden G, Crouse S, Brown J, Green J. Lipid and lipoprotein changes in women following 6 months of exercise training in a worksite fitness program. *Journal of Sports Medicine and Physical Fitness*. 1996;36:54–59.
- 331. Ready AE, Drinkwater DT, Ducas J, Fitzpatrick D, Brereton DG, Oades SC. Walking program reduces elevated cholesterol in women postmenopause. *The Canadian Journal* of Cardiology. 1995;11:905.
- 332. Boyden TW, Pamenter RW, Going SB, Lohman TG, Hall MC, Houtkooper LB, Bunt JC, Ritenbaugh C, Aickin M. Resistance exercise training is associated with decreases in serum low-density lipoprotein cholesterol levels in premenopausal women. Archives of Internal Medicine. 1993;153:97.
- 333. Kiens B, Jorgensen I, Lewis S, Jensen G, Lithell H, Vessby B, Hoe S, Schnohr P. Increased plasma HDL-cholesterol and apoA-1 in sedentary middle-aged men after physical conditioning. *European Journal of Clinical Investigation*. 1980;10:203–209.
- 334. Smutok M, Reece C, Kokkinos P, Farmer C, Dawson P, Shulman R, DeVane-Bell J, Patterson J, Charabogos C, Goldberg A. Aerobic versus strength training for risk factor intervention in middle-aged men at high risk for coronary heart disease. *Metabolism*. 1993;42:177–184.
- 335. Binder EF, Birge SJ, Kohrt WM. Effects of endurance exercise and hormone replacement therapy on serum lipids in older women. *Journal of the American Geriatrics Society*. 1996;44:231.
- 336. Halbert J, Silagy C, Finucane P, Withers R, Hamdorf P. Exercise training and blood lipids in hyperlipidemic and normolipidemic adults: A meta-analysis of randomized, controlled trials. *European Journal of Clinical Nutrition*. 1999;53:514.
- 337. Muthusamy VR, Kannan S, Sadhaasivam K, Gounder SS, Davidson CJ, Boeheme C, Hoidal JR, Wang L, Soorappan RN. Acute exercise stress activates Nrf2/ARE signaling and promotes antioxidant mechanisms in the myocardium. *Free Radical Biology and Medicine*. 2012;52:366–376.
- 338. Lee S, Park Y, Zuidema MY, Hannink M, Zhang C. Effects of interventions on oxidative stress and inflammation of cardiovascular diseases. *World Journal of Cardiology*. 2011;3:18.
- Faraci FM, Didion SP. Vascular protection. Arteriosclerosis, Thrombosis, and Vascular Biology. 2004;24:1367–1373.

- 340. Powers SK, Ji LL, Leeuwenburgh C. Exercise training-induced alterations in skeletal muscle antioxidant capacity: A brief review. *Medicine and Science in Sports and Exercise*. 1999;31:987–997.
- Rush JWE, Turk JR, Laughlin MH. Exercise training regulate Sod-1 and oxidative stress in porcine aortic endothelium. *American Journal of Physiology-Heart and Circulatory Physiology*. 2003;284:H1378–H1387.
- 342. Rush JWE, Laughlin MH, Woodman CR, Price EM. Sod-1 expression in pig coronary arterioles is increased by exercise training. *American Journal of Physiology-Heart and Circulatory Physiology*. 2000;279:H2068–H2076.
- Lennon SL, Quindry J, Hamilton KL, French J, Staib J, Mehta JL, Powers SK. Loss of exercise-induced cardioprotection after cessation of exercise. *Journal of Applied Physiology*. 2004;96:1299–1305.
- 344. Duncan K, Harris S, Ardies CM. Running exercise may reduce risk for lung and liver cancer by inducing activity of antioxidant and phase ii enzymes. *Cancer Letters*. 1997;116:151–158.
- 345. Ardies CM, Smith TJ, Kim S, Yang CS. Induction of 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanone (NNK) activation in rat lung microsomes by chronic ethanol consumption and repeated running exercise. *Cancer Letters*. 1996;103:209–218.
- 346. Walsh NP, Gleeson M, Shephard RJ, Woods J, Bishop N, Fleshner M, Green C, Pedersen B, Hoffman-Goetz L, Rogers C. Position statement. Part one: Immune function and exercise. *Exercise Immunology Review*. 2011;17:6–63.
- 347. Gleeson M, Bishop NC, Stensel DJ, Lindley MR, Mastana SS, Nimmo MA. The antiinflammatory effects of exercise: Mechanisms and implications for the prevention and treatment of disease. *Nature Reviews Immunology*. 2011;11:607–615.
- 348. Jolitha A, Subramanyam M, Asha Devi S. Modification by vitamin E and exercise of oxidative stress in regions of aging rat brain: Studies on superoxide dismutase iso-enzymes and protein oxidation status. *Experimental Gerontology*. 2006;41:753–763.
- 349. Gomez-Cabrera MC, Sanchis-Gomar F, Martinez-Bello VE, Ibanez-Sania S, Nascimento AL, Ji LL, Vina J. Exercise as a model to study oxidative stress. In: *Studies on Experimental Models*. Totowa, NJ: Humana Press; 2011: 531–542.
- 350. Na HK, Surh YJ. Modulation of Nrf2-mediated antioxidant and detoxifying enzyme induction by the green tea polyphenol EGCG. *Food and Chemical Toxicology*. 2008;46:1271–1278.
- 351. Surh YJ, Kundu JK, Na HK. Nrf2 as a master redox switch in turning on the cellular signaling involved in the induction of cytoprotective genes by some chemopreventive phytochemicals. *Planta medica*. 2008;74:1526.
- 352. Surh YJ, Na HK. Nf-κb and Nrf2 as prime molecular targets for chemoprevention and cytoprotection with anti-inflammatory and antioxidant phytochemicals. *Genes & Nutrition*. 2008;2:313–317.
- 353. Kim YJ, Baek SH, Bogner PN, Ip C, Rustum YM, Fakih MG, Ramnath N, Park YM. Targeting the Nrf2-Prx1 pathway with selenium to enhance the efficacy and selectivity of cancer therapy. *Journal of Cancer Molecules*. 2007;3:37–43.
- 354. Rosenblat M, Belinky P, Vaya J, Levy R, Hayek T, Coleman R, Merchav S, Aviram M. Macrophage enrichment with the isoflavan glabridin inhibits NADPH oxidase-induced cell-mediated oxidation of low density lipoprotein. *Journal of Biological Chemistry*. 1999;274:13790–13799.
- 355. Kaplan M, Aviram M. Oxidized low density lipoprotein: Atherogenic and proinflammatory characteristics during macrophage foam cell formation. An inhibitory role for nutritional antioxidants and serum paraoxonase. *Clinical Chemistry and Laboratory Medicine*. 1999;37:777–787.

- 356. Aviram M, Fuhrman B. LDL oxidation by arterial wall macrophages depends on the oxidative status in the lipoprotein and in the cells: Role of prooxidants vs. antioxidants. *Molecular and Cellular Biochemistry*. 1998;188:149–159.
- 357. Masella R, Varì R, D'Archivio M, Di Benedetto R, Matarrese P, Malorni W, Scazzocchio B, Giovannini C. Extra virgin olive oil biophenols inhibit cell-mediated oxidation of LDL by increasing the mRNA transcription of glutathione-related enzymes. *The Journal of Nutrition*. 2004;134:785–791.
- 358. Srimahachota S, Wunsuwan R, Siritantikorn A, Boonla C, Chaiwongkarjohn S, Tosukhowong P. Effects of lifestyle modification on oxidized LDL, reactive oxygen species production and endothelial cell viability in patients with coronary artery disease. *Clinical Biochemistry*. 2010;43:858–862.
- 359. Elokda AS, Nielsen DH. Effects of exercise training on the glutathione antioxidant system. *European Journal of Cardiovascular Prevention & Rehabilitation*. 2007;14:630–637.
- 360. Roberts CK, Vaziri ND, Barnard RJ. Effect of diet and exercise intervention on blood pressure, insulin, oxidative stress, and nitric oxide availability. *Circulation*. 2002;106:2530–2532.
- 361. Beard CM, Barnard RJ, Robbins DC, Ordovas JM, Schaefer EJ. Effects of diet and exercise on qualitative and quantitative measures of LDL and its susceptibility to oxidation. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1996;16:201–207.
- 362. Elosua R, Molina L, Fito M, Arquer A, Sanchez-Quesada J, Covas M, Ordonez-Llanos J, Marrugat J. Response of oxidative stress biomarkers to a 16-week aerobic physical activity program, and to acute physical activity, in healthy young men and women. *Atherosclerosis*. 2003;167:327–334.
- 363. Flores-Mateo G, Carrillo-Santisteve P, Elosua R, Guallar E, Marrugat J, Bleys J, Covas M-I. Antioxidant enzyme activity and coronary heart disease: Meta-analyses of observational studies. *American Journal of Epidemiology*. 2009;170:135–147.
- 364. García-López D, Häkkinen K, Cuevas M, Lima E, Kauhanen A, Mattila M, Sillanpää E, Ahtiainen J, Karavirta L, Almar M. Effects of strength and endurance training on antioxidant enzyme gene expression and activity in middle-aged men. *Scandinavian Journal* of Medicine & Science in Sports. 2007;17:595–604.
- 365. Takahashi M, Suzuki K, Matoba H, Sakamoto S, Obara S. Effects of different intensities of endurance exercise on oxidative stress and antioxidant capacity. *The Journal of Physical Fitness and Sports Medicine*. 2012;1:183–189.
- 366. Powers SK, Criswell D, Lawler J, Ji L, Martin D, Herb RA, Dudley G. Influence of exercise and fiber type on antioxidant enzyme activity in rat skeletal muscle. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology.* 1994;266:R375–R380.
- 367. Powers SK, Criswell D, Lawler J, Martin D, Lieu F, Ji L, Herb RA. Rigorous exercise training increases superoxide dismutase activity in ventricular myocardium. *American Journal of Physiology-Heart and Circulatory Physiology*. 1993;265:H2094–H2098.
- 368. Miles MP, Andring JM, Pearson SD, Gordon LK, Kasper C, Depner CM, Kidd JR. Diurnal variation, response to eccentric exercise, and association of inflammatory mediators with muscle damage variables. *Journal of Applied Physiology*. 2008;104:451–458.
- MacIntyre DL, Sorichter S, Mair J, Berg A, McKenzie DC. Markers of inflammation and myofibrillar proteins following eccentric exercise in humans. *European Journal of Applied Physiology*. 2001;84:180–186.
- 370. Shephard RJ. Cytokine responses to physical activity, with particular reference to IL-6: Sources, actions, and clinical implications. *Critical Reviews in Immunology*. 2002;22:165–182.
- 371. Ostrowski K, Rohde T, Asp S, Schjerling P, Pedersen BK. Pro-and anti-inflammatory cytokine balance in strenuous exercise in humans. *The Journal of Physiology*. 1999;515:287.

- 372. Peake JM, Suzuki K, Hordern M, Wilson G, Nosaka K, Coombes JS. Plasma cytokine changes in relation to exercise intensity and muscle damage. *European Journal of Applied Physiology*. 2005;95:514–521.
- 373. Pedersen BK. Edward F. Adolph distinguished lecture: Muscle as an endocrine organ: IL-6 and other myokines. *Journal of Applied Physiology*. 2009;107:1006.
- 374. Pedersen BK. Muscles and their myokines. *Journal of Experimental Biology*. 2011;214:337–346.
- 375. Petersen AMW, Pedersen BK. The anti-inflammatory effect of exercise. *Journal of Applied Physiology*. 2005;98:1154.
- 376. Helge JW, Stallknecht B, Pedersen BK, Galbo H, Kiens B, Richter EA. The effect of graded exercise on IL-6 release and glucose uptake in human skeletal muscle. *The Journal of Physiology*. 2003;546:299–305.
- 377. Pedersen BK, Febbraio MA. Muscle as an endocrine organ: Focus on muscle-derived interleukin-6. *Physiological Reviews*. 2008;88:1379.
- 378. Pedersen M, Steensberg A, Keller C, Osada T, Zacho M, Saltin B, Febbraio MA, Pedersen BK. Does the aging skeletal muscle maintain its endocrine function. *Exercise Immunology Review*. 2004;10:42–55.
- 379. de Waal Malefyt R, Abrams J, Bennett B, Figdor CG, de Vries JE. Interleukin 10 (IL-10) inhibits cytokine synthesis by human monocytes: An autoregulatory role of IL-10 produced by monocytes. *The Journal of Experimental Medicine*. 1991;174:1209–1220.
- Wang P, Wu P, Siegel MI, Egan RW, Billah MM. Interleukin (IL)-10 inhibits nuclear factor b (NFB) activation in human monocytes. *Journal of Biological Chemistry*. 1995;270:9558.
- 381. Fukaya T, Suzuki M, Sakaguchi R, Yasukawa H, Yoshimura A. Role of SOCS proteins in inflammation and autoimmune diseases. *Inflammation and Regeneration*. 2011;31:382–392.
- Mosser DM, Zhang X. Interleukin-10: New perspectives on an old cytokine. *Immunological Reviews*. 2008;226:205–218.
- 383. Smith JK, Dykes R, Douglas JE, Krishnaswamy G, Berk S. Long-term exercise and atherogenic activity of blood mononuclear cells in persons at risk of developing ischemic heart disease. JAMA: The Journal of the American Medical Association. 1999;281:1722–1727.
- 384. Hambrecht R, Adams V, Erbs S, Linke A, Krankel N, Shu Y, Baither Y et al. Regular physical activity improves endothelial function in patients with coronary artery disease by increasing phosphorylation of endothelial nitric oxide synthase. *Circulation*. 2003;107:3152–3158.
- 385. Hambrecht R, Fiehn E, Weigl C, Gielen S, Hamann C, Kaiser R, Yu J, Adams V, Niebauer J, Schuler G. Regular physical exercise corrects endothelial dysfunction and improves exercise capacity in patients with chronic heart failure *Circulation*. 1998;98:2709–2715.
- 386. Hambrecht R, Walther C, Mobius-Winkler S, Gielen S, Linke A, Conradi K, Erbs S et al. Percutaneous coronary angioplasty compared with exercise training in patients with stable coronary artery disease: A randomized trial. *Circulation*. 2004;109:1371–1378.
- Marcell TJ, McAuley KA, Traustadóttir T, Reaven PD. Exercise training is not associated with improved levels of C-reactive protein or adiponectin. *Metabolism*. 2005;54:533–541.
- 388. Mattusch F, Dufaux B, Heine O, Mertens I, Rost R. Reduction of the plasma concentration of C-reactive protein following nine months of endurance training. *International Journal of Sports Medicine*. 2000;21:21–24.
- 389. Nicklas BJ, Ambrosius W, Messier SP, Miller GD, Penninx BW, Loeser RF, Palla S, Bleecker E, Pahor M. Diet-induced weight loss, exercise, and chronic inflammation in older, obese adults: A randomized controlled clinical trial. *The American Journal of Clinical Nutrition*. 2004;79:544–551.
- 390. Friedenreich CM, Neilson HK, Woolcott CG, Wang Q, Stanczyk FZ, McTiernan A, Jones CA, Irwin ML, Yasui Y, Courneya KS. Inflammatory marker changes in a yearlong randomized exercise intervention trial among postmenopausal women. *Cancer Prevention Research*. 2012;5:98–108.
- 391. Fairey AS, Courneya KS, Field CJ, Bell GJ, Jones LW, Martin BS, Mackey JR. Effect of exercise training on Creactive protein in postmenopausal breast cancer survivors: A randomized controlled trial. *Brain, Behavior, and Immunity*. 2005;19:381–388.
- 392. Kohut M, McCann D, Russell D, Konopka D, Cunnick J, Franke W, Castillo M, Reighard A, Vanderah E. Aerobic exercise, but not flexibility/resistance exercise, reduces serum IL-18, CRP, and IL-6 independent of β-blockers, BMI, and psychosocial factors in older adults. *Brain, Behavior, and Immunity; Brain, Behavior, and Immunity*. 2006;20:201–209.
- 393. Kelly AS, Wetzsteon RJ, Kaiser DR, Steinberger J, Bank AJ, Dengel DR. Inflammation, insulin, and endothelial function in overweight children and adolescents: The role of exercise. *The Journal of Pediatrics*. 2004;145:731–736.
- 394. Nicklas BJ, Hsu FC, Brinkley TJ, Church T, Goodpaster BH, Kritchevsky SB, Pahor M. Exercise training and plasma C-reactive protein and interleukin-6 in elderly people. *Journal of the American Geriatrics Society*. 2008;56:2045–2052.
- 395. Kadoglou NPE, Iliadis F, Angelopoulou N, Perrea D, Ampatzidis G, Liapis CD, Alevizos M. The anti-inflammatory effects of exercise training in patients with type 2 diabetes mellitus. *Journal of Cardiovascular Risk*. 2007;14:837.
- 396. Campbell PT, Campbell KL, Wener MH, Wood BL, Potter JD, McTiernan A, Ulrich CM. A yearlong exercise intervention decreases CRP among obese postmenopausal women. *Medicine and Science in Sports and Exercise*. 2009;41:1533.
- 397. Ford ES. Does exercise reduce inflammation? Physical activity and C-reactive protein among U.S. Adults. *Epidemiology*. 2002;13:561–568.
- 398. Gorecki M, Beck Y, Hartman JR, Fischer M, Weiss L, Tochner Z, Slavin S, Nimrod A. Recombinant human superoxide dismutases: Production and potential therapeutical uses. *Free Radical Research*. 1991;12:401–410.
- 399. Mellman W, Schimke R, Hayflick L. Catalase turnover in human diploid cell cultures. *Experimental Cell Research*. 1972;73:399–409.
- 400. Chada S, Whitney C, Newburger PE. Post-transcriptional regulation of glutathione peroxidase gene expression by selenium in the HL-60 human myeloid cell line. *Blood*. 1989;74:2535–2541.
- 401. Gallegos A, Berggren M, Gasdaska JR, Powis G. Mechanisms of the regulation of thioredoxin reductase activity in cancer cells by the chemopreventive agent selenium. *Cancer Research*. 1997;57:4965.
- 402. Wei Y, Chen K, Whaley-Connell AT, Stump CS, Ibdah JA, Sowers JR. Skeletal muscle insulin resistance: Role of inflammatory cytokines and reactive oxygen species. *American Journal* of Physiology-Regulatory, Integrative and Comparative Physiology. 2008;294:R673–R680.
- 403. Houmard JA, Tanner CJ, Slentz CA, Duscha BD, McCartney JS, Kraus WE. Effect of the volume and intensity of exercise training on insulin sensitivity. *Journal of Applied Physiology*. 2004;96:101–106.
- 404. Koopman R, Manders RJ, Zorenc AH, Hul GB, Kuipers H, Keizer HA, van Loon LJ. A single session of resistance exercise enhances insulin sensitivity for at least 24 h in healthy men. *European Journal of Applied Physiology*. 2005;94:180–187.
- 405. Balducci S, Zanuso S, Nicolucci A, Fernando F, Cavallo S, Cardelli P, Fallucca S, Alessi E, Letizia C, Jimenez A. Anti-inflammatory effect of exercise training in subjects with type 2 diabetes and the metabolic syndrome is dependent on exercise modalities and independent of weight loss. *Nutrition, Metabolism and Cardiovascular Diseases*. 2010;20:608–617.
- 406. Zanuso S, Jimenez A, Pugliese G, Corigliano G, Balducci S. Exercise for the management of type 2 diabetes: A review of the evidence. *Acta Diabetologica*. 2010;47:15–22.

- 407. Cartee GD, Funai K. Exercise and insulin: Convergence or divergence at AS160 and TBC1D1? *Exercise and Sport Sciences Reviews*. 2009;37:188.
- 408. Funai K, Schweitzer GG, Castorena CM, Kanzaki M, Cartee GD. In vivo exercise followed by in vitro contraction additively elevates subsequent insulin-stimulated glucose transport by rat skeletal muscle. *American Journal of Physiology-Endocrinology and Metabolism*. 2010;298:E999–E1010.
- 409. Funai K, Schweitzer GG, Sharma N, Kanzaki M, Cartee GD. Increased AS160 phosphorylation, but not TBC1D1 phosphorylation, with increased postexercise insulin sensitivity in rat skeletal muscle. *American Journal of Physiology-Endocrinology and Metabolism*. 2009;297:E242–E251.
- 410. Chen S, Murphy J, Toth R, Campbell DG, Morrice NA, MacKintosh C. Complementary regulation of TBC1D1 and AS160 by growth factors, insulin and AMPK activators. *Biochemical Journal*. 2010;409:449–459.
- 411. Hawley J, Lessard S. Exercise training-induced improvements in insulin action. *Acta Physiologica*. 2007;192:127–135.
- 412. Zierath JR. Invited review: Exercise training-induced changes in insulin signaling in skeletal muscle. *Journal of Applied Physiology*. 2002;93:773–781.
- 413. Jorge MLMP, de Oliveira VN, Resende NM, Paraiso LF, Calixto A, Diniz ALD, Resende ES, Ropelle ER, Carvalheira JB, Espindola FS. The effects of aerobic, resistance, and combined exercise on metabolic control, inflammatory markers, adipocytokines, and muscle insulin signaling in patients with type 2 diabetes mellitus. *Metabolism*. 2011;60:1244–1252.
- 414. Lira VA, Benton CR, Yan Z, Bonen A. Pgc-1α regulation by exercise training and its influences on muscle function and insulin sensitivity. *American Journal of Physiology-Endocrinology and Metabolism.* 2010;299:E145–E161.
- 415. Little JP, Safdar A, Cermak N, Tarnopolsky MA, Gibala MJ. Acute endurance exercise increases the nuclear abundance of PGC-1α in trained human skeletal muscle. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology*. 2010;298:R912–R917.
- 416. Palacios OM, Carmona JJ, Michan S, Chen KY, Manabe Y, Ward Iii JL, Goodyear LJ, Tong Q. Diet and exercise signals regulate SIRT3 and activate AMPK and PGC-1α in skeletal muscle. *Aging (Albany NY)*. 2009;1:771.
- 417. Chow L, From A, Seaquist E. Skeletal muscle insulin resistance: The interplay of local lipid excess and mitochondrial dysfunction. *Metabolism: Clinical and Experimental*. 2010;59:70.
- 418. Lemoine S, Granier P, Tiffoche C, Rannou-Bekono F, Thieulant M-L, Delamarche P. Estrogen receptor alpha mRNA in human skeletal muscles. *Medicine and Science in Sports and Exercise*. 2003;35:439.
- 419. Lee AV, Jackson JG, Gooch JL, Hilsenbeck SG, Coronado-Heinsohn E, Osborne CK, Yee D. Enhancement of insulin-like growth factor signaling in human breast cancer: Estrogen regulation of insulin receptor substrate-1 expression in vitro and in vivo. *Molecular Endocrinology*. 1999;13:787–796.
- 420. Mauro L, Salerno M, Panno ML, Bellizzi D, Sisci D, Miglietta A, Surmacz E, Andò S. Estradiol increases IRS-1 gene expression and insulin signaling in breast cancer cells. *Biochemical and Biophysical Research Communications*. 2001;288:685–689.
- 421. Wiik A, Ekman M, Johansson O, Jansson E, Esbjörnsson M. Expression of both oestrogen receptor alpha and beta in human skeletal muscle tissue. *Histochemistry and Cell Biology*. 2009;131:181–189.
- 422. Tcherepanova I, Puigserver P, Norris JD, Spiegelman BM, McDonnell DP. Modulation of estrogen receptor-α transcriptional activity by the coactivator PGC-1α. *Journal of Biological Chemistry*. 2000;275:16302–16308.

- Murphy LC, Seekallu SV, Watson PH. Clinical significance of estrogen receptor phosphorylation. *Endocrine-Related Cancer*. 2011;18:R1–R14.
- 424. Lee H, Bai W. Regulation of estrogen receptor nuclear export by ligand-induced and p38-mediated receptor phosphorylation. *Molecular and Cellular Biology*. 2002;22:5835–5845.
- 425. Lannigan DA. Estrogen receptor phosphorylation. Steroids. 2003;68:1-9.
- 426. Lin J, Cook NR, Albert C, Zaharris E, Gaziano JM, Van Denburgh M, Buring JE, Manson JAE. Vitamins C and E and beta carotene supplementation and cancer risk: A randomized controlled trial. *Journal of the National Cancer Institute*. 2009;101:14–23.
- 427. Malila N, Taylor PR, Virtanen MJ, Korhonen P, Huttunen JK, Albanes D, Virtamo J. Effects of alpha-tocopherol and beta-carotene supplementation on gastric cancer incidence in male smokers (ATBC study, Finland). *Cancer Causes and Control*. 2002;13:617–623.
- 428. Omenn GS, Goodman GE, Thornquist MD, Balmes J, Cullen MR, Glass A, Keogh JP, Meyskens FL, Jr., Valanis B, Williams JH, Jr. Risk factors for lung cancer and for intervention effects in CARET, the beta-carotene and retinol efficacy trial. *Journal of the National Cancer Institute*. 1996;88:1550–1559.
- 429. Miller ER, III, Pastor-Barriuso R, Dalal D, Riemersma RA, Appel LJ, Guallar E. Meta-analysis: High-dosage vitamin E supplementation may increase all-cause mortality. *Annals of Internal Medicine*. 2005;142:37–46.
- 430. Lonn E, Bosch J, Yusuf S, Sheridan P, Pogue J, Arnold J, Ross C, Arnold A, Sleight P, Probstfield J. Effects of long-term vitamin E supplementation on cardiovascular events and cancer: A randomized controlled trial. *JAMA: The Journal of the American Medical Association*. 2005;293:1338.
- 431. Jha P, Flather M, Lonn E, Farkouh M, Yusuf S. The antioxidant vitamins and cardiovascular disease: A critical review of epidemiologic and clinical trial data. *Annals of Internal Medicine*. 1995;123:860–872.
- 432. Bjelakovic G, Nikolova D, Gluud LL, Simonetti RG, Gluud C. Mortality in randomized trials of antioxidant supplements for primary and secondary prevention. *JAMA: The Journal of the American Medical Association*. 2007;297:842–857.
- 433. Goodman M, Bostick RM, Kucuk O, Jones DP. Clinical trials of antioxidants as cancer prevention agents: Past, present, and future. *Free Radical Biology and Medicine*. 2011;51:1068.
- 434. Clarkson PM. Minerals: Exercise performance and supplementation in athletes. *Journal* of Sports Sciences. 1991;9:91–116.
- 435. Gomez-Cabrera MC, Domenech E, Viña J. Moderate exercise is an antioxidant: Upregulation of antioxidant genes by training. *Free Radical Biology and Medicine*. 2008;44:126–131.
- 436. Sharman I, Down M, Sen R. The effects of vitamin E and training on physiological function and athletic performance in adolescent swimmers. *The British Journal of Nutrition*. 1971;26:265–276.
- 437. Malm C, Svensson M, Ekblom B, Sjödin B. Effects of ubiquinone-10 supplementation and high intensity training on physical performance in humans. *Acta Physiologica Scandinavica*. 1997;161:379–384.
- 438. Malm C, Svensson M, Sjöberg B, Ekblom B, Sjödin B. Supplementation with ubiquinone-10 causes cellular damage during intense exercise. *Acta Physiologica Scandinavica*. 1996;157:511–512.
- 439. Ristow M, Zarse K, Oberbach A, Klöting N, Birringer M, Kiehntopf M, Stumvoll M, Kahn CR, Blüher M. Antioxidants prevent health-promoting effects of physical exercise in humans. *Proceedings of the National Academy of Sciences*. 2009;106:8665.
- 440. Maughan R, Greenhaff P, Hespel P. Dietary supplements for athletes: Emerging trends and recurring themes. *Journal of Sports Sciences*. 2011;29:57–66.

- 441. Martin KR, Appel CL. Polyphenols as dietary supplements: A double-edged sword. *Nutrition and Dietary Supplements*. 2010;2:1–12.
- 442. Leren P. The Oslo diet-heart study. Eleven-year report. Circulation. 1970;42:935-942.
- 443. Jenkins DJ, Kendall CW, Faulkner D, Vidgen E, Trautwein EA, Parker TL, Marchie A et al. A dietary portfolio approach to cholesterol reduction: Combined effects of plant sterols, vegetable proteins, and viscous fibers in hypercholesterolemia. *Metabolism.* 2002;51:1596–1604.
- 444. Jenkins DJ, Kendall CW, Marchie A, Faulkner DA, Wong JM, de Souza R, Emam A et al. Effects of a dietary portfolio of cholesterol-lowering foods vs lovastatin on serum lipids and C-reactive protein. JAMA: The Journal of the American Medical Association. 2003;290:502–510.
- 445. Obarzanek E, Sacks FM, Vollmer WM, Bray GA, Miller ER, III, Lin PH, Karanja NM et al. Effects on blood lipids of a blood pressure-lowering diet: The Dietary Approaches to Stop Hypertension (DASH) trial. *The American Journal of Clinical Nutrition*. 2001;74:80–89.
- 446. Frantz I, Jr., Dawson E, Ashman P, Gatewood L, Bartsch G, Kuba K, Brewer E. Test of effect of lipid lowering by diet on cardiovascular risk. The Minnesota coronary survey. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1989;9:129–135.
- 447. Corr L, Oliver M. The low fat/low cholesterol diet is ineffective. *European Heart Journal*. 1997;18:18–22.
- 448. Neaton JD, Blackburn H, Jacobs D, Kuller L, Lee DJ, Sherwin R, Shih J, Stamler J, Wentworth D. Serum cholesterol level and mortality findings for men screened in the Multiple Risk Factor Intervention Trial. *Archives of Internal Medicine*. 1992;152:1490.
- 449. Haskell WL, Alderman EL, Fair JM, Maron DJ, Mackey SF, Superko HR, Williams PT et al. Effects of intensive multiple risk factor reduction on coronary atherosclerosis and clinical cardiac events in men and women with coronary artery disease. The Stanford Coronary Risk Intervention Project (SCRIP). *Circulation*. 1994;89:975–990.
- 450. Ornish D, Brown SE, Scherwitz LW, Billings JH, Armstrong WT, Ports TA, McLanahan SM, Kirkeeide RL, Brand RJ, Gould KL. Can lifestyle changes reverse coronary heart disease? The lifestyle heart trial. *Lancet*. 1990;336:129–133.
- 451. Ornish D. Avoiding revascularization with lifestyle changes: The multicenter lifestyle demonstration project. *The American Journal of Cardiology*. 1998;82:72T–76T.
- 452. Barnard RJ, Guzy PM, Rosenberg JM, Trexler O'Brien L. Effects of an intensive exercise and nutrition program on patients with coronary artery disease: Five-year follow up. *Journal of Cardiac Rehabilitation*. 1983;3:183–190.
- 453. Barnard RJ. Effects of life-style modification on serum lipids. *The American Journal of Clinical Nutrition*. 1991;151:1389–1394.
- 454. Lusis AJ. Atherosclerosis. Nature. 2000;407:233-241.
- 455. Patrono C, FitzGerald GA. Isoprostanes: Potential markers of oxidant stress in atherothrombotic disease. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 1997;17:2309–2315.
- 456. Parks EJ, German JB, Davis PA, Frankel EN, Kappagoda CT, Rutledge JC, Hyson DA, Schneeman BO. Reduced oxidative susceptibility of LDL from patients participating in an intensive atherosclerosis treatment program. *The American Journal of Clinical Nutrition*. 1998;68:778–785.
- 457. Barnard RJ, Inkeles SB. Effects of an intensive diet and exercise program on lipids in postmenopausal women. *Women's Health Issues*. 1999;9:155–161.
- 458. Ridker PM, Buring JE, Shih J, Matias M, Hennekens CH. Prospective study of C-reactive protein and the risk of future cardiovascular events among apparently healthy women. *Circulation*. 1998;98:731–733.
- 459. Ridker PM. Clinical application of C-reactive protein for cardiovascular disease detection and prevention. *Circulation*. 2003;107:363–369.

- 460. Wegge JK, Roberts CK, Ngo TH, Barnard RJ. Effect of diet and exercise intervention on inflammatory and adhesion molecules in postmenopausal women on hormone replacement therapy and at risk for coronary artery disease. *Metabolism*. 2004;53:377–381.
- 461. Barnard RJ, Hall JA, Chaudhari A, Miller JE, Kirschenbaum MA. Effects of a low-fat, low-cholesterol diet on serum lipids, platelet aggregation and thromboxane formation. *Prostaglandins, Leukotrienes, and Medicine*. 1987;26:241–252.
- 462. Mehrabian M, Peter JB, Barnard RJ, Lusis AJ. Dietary regulation of fibrinolytic factors. *Atherosclerosis*. 1990;84:25–32.
- 463. Heilbronn LK, Noakes M, Clifton PM. Energy restriction and weight loss on verylow-fat diets reduce C-reactive protein concentrations in obese, healthy women. *Arteriosclerosis, Thrombosis, and Vascular Biology.* 2001;21:968–970.
- 464. Ziccardi P, Nappo F, Giugliano G, Esposito K, Marfella R, Cioffi M, D'Andrea F, Molinari AM, Giugliano D. Reduction of inflammatory cytokine concentrations and improvement of endothelial functions in obese women after weight loss over one year. *Circulation*. 2002;105:804–809.
- 465. Izadpanah A, Barnard RJ, Almeda AJE, Baldwin GC, Bridges SA, Shellman ER, Burant CF, Roberts CK. A short-term diet and exercise intervention ameliorates inflammation and markers of metabolic health in overweight/obese children. *American Journal* of Physiology-Endocrinology And Metabolism. 2012;303:E542–E550.
- 466. Evans JH, Spencer DM, Zweifach A, Leslie CC. Intracellular calcium signals regulating cytosolic phospholipase A₂ translocation to internal membranes. *Journal of Biological Chemistry*. 2001;276:30150.
- 467. Bonventre JV, Swidler M. Calcium dependency of prostaglandin E₂ production in rat glomerular mesangial cells. Evidence that protein kinase C modulates the Ca²⁺-dependent activation of phospholipase A₂. *Journal of Clinical Investigation*. 1988;82:168.
- 468. Handlogten ME, Huang C, Shiraishi N, Awata H, Miller RT. The Ca²⁺-sensing receptor activates cytosolic phospholipase A₂ via a GQα-dependent ERK-independent pathway. *Journal of Biological Chemistry*. 2001;276:13941.
- 469. Zor U, Ferber E, Gergely P, Szücs K, Dombradi V, Goldman R. Reactive oxygen species mediate phorbol ester-regulated tyrosine phosphorylation and phospholipase A₂ activation: Potentiation by vanadate. *Biochemical Journal*. 1993;295:879.
- 470. Nelson DR, Kamataki T, Waxman DJ, Guengerich FP, Estabrook RW, Feyereisen R, Gonzalez FJ, Coon MJ, Gunsalus IC, Gotoh O. The p450 superfamily: Update on new sequences, gene mapping, accession numbers, early trivial names of enzymes, and nomenclature. DNA and Cell Biology. 1993;12:1–51.
- 471. Simopoulos AP. The importance of the omega-6/omega-3 fatty acid ratio in cardiovascular disease and other chronic diseases. *Experimental Biology and Medicine*. 2008;233:674–688.
- 472. Albert CM, Campos H, Stampfer MJ, Ridker PM, Manson JE, Willett WC, Ma J. Blood levels of long-chain *n*-3 fatty acids and the risk of sudden death. *The New England Journal of Medicine*. 2002;346:1113–1118.
- 473. Albert CM, Gaziano JM, Willett WC, Manson JE. Nut consumption and decreased risk of sudden cardiac death in the physicians' health study. *The American Journal of Clinical Nutrition*. 2002;162:1382–1387.
- 474. Hu FB, Bronner L, Willett WC, Stampfer MJ, Rexrode KM, Albert CM, Hunter D, Manson JE. Fish and omega-3 fatty acid intake and risk of coronary heart disease in women. JAMA: The Journal of the American Medical Association. 2002;287:1815–1821.
- 475. Harris WS, Mozaffarian D, Rimm E, Kris-Etherton P, Rudel LL, Appel LJ, Engler MM, Engler MB, Sacks F. Omega-6 fatty acids and risk for cardiovascular disease. *Circulation*. 2009;119:902–907.
- 476. Katan MB. Omega-6 polyunsaturated fatty acids and coronary heart disease. *The American Journal of Clinical Nutrition*. 2009;89:1283–1284.

- 477. Sarkkinen ES, Agren JJ, Ahola I, Ovaskainen ML, Uusitupa MI. Fatty acid composition of serum cholesterol esters, and erythrocyte and platelet membranes as indicators of long-term adherence to fat-modified diets. *The American Journal of Clinical Nutrition*. 1994;59:364–370.
- 478. Burdge GC, Calder PC. Conversion of alpha-linolenic acid to longer-chain polyunsaturated fatty acids in human adults. *Reproduction Nutrition Development*. 2005;45:581–598.
- 479. Arterburn LM, Hall EB, Oken H. Distribution, interconversion, and doseresponse of n–3 fatty acids in humans. *The American Journal of Clinical Nutrition*. 2006;83:S1467– 1476S.
- 480. De Lorgeril M, Salen P, Laporte F, De Leiris J. Alpha-linolenic acid in the prevention and treatment of coronary heart disease. *European Heart Journal Supplements*. 2001;3:D26.
- 481. Zhao G, Etherton TD, Martin KR, West SG, Gillies PJ, Kris-Etherton PM. Dietary alpha-linolenic acid reduces inflammatory and lipid cardiovascular risk factors in hyper-cholesterolemic men and women. *The Journal of Nutrition*. 2004;134:2991–2997.
- 482. Simopoulos AP. Omega-6/omega-3 essential fatty acids: Biological effects. *World Review of Nutrition & Dietetics*. 2009;99:1–16.
- 483. Harris WS, Poston WC, Haddock CK. Tissue *n*-3 and *n*-6 fatty acids and risk for coronary heart disease events. *Atherosclerosis*. 2007;193:1–10.
- Lee JH, O'Keefe JH, Lavie CJ, Marchioli R, Harris WS. Omega-3 fatty acids for cardioprotection. *Mayo Clinic Proceedings*. 2008;83:324–332.
- 485. Wang C, Harris WS, Chung M, Lichtenstein AH, Balk EM, Kupelnick B, Jordan HS, Lau J. N–3 fatty acids from fish or fish-oil supplements, but not α-linolenic acid, benefit cardiovascular disease outcomes in primary-and secondary-prevention studies: A systematic review. *The American Journal of Clinical Nutrition*. 2006;84:5–17.
- 486. Zatonski W, Campos H, Willett W. Rapid declines in coronary heart disease mortality in eastern Europe are associated with increased consumption of oils rich in alpha-linolenic acid. *European Journal of Epidemiology*. 2008;23:3–10.
- 487. Din JN, Harding SA, Valerio CJ, Sarma J, Lyall K, Riemersma RA, Newby DE, Flapan AD. Dietary intervention with oil rich fish reduces platelet-monocyte aggregation in man. *Atherosclerosis*. 2008;197:290–296.
- 488. He K, Liu K, Daviglus ML, Jenny NS, Mayer-Davis E, Jiang R, Steffen L, Siscovick D, Tsai M, Herrington D. Associations of dietary long-chain *n*-3 polyunsaturated fatty acids and fish with biomarkers of inflammation and endothelial activation (from the multi-ethnic study of atherosclerosis [MESA]). *The American Journal of Cardiology*. 2009;103:1238–1243.
- 489. Pischon T, Hankinson SE, Hotamisligil GS, Rifai N, Willett WC, Rimm EB. Habitual dietary intake of *n*-3 and *n*-6 fatty acids in relation to inflammatory markers among US men and women. *Circulation*. 2003;108:155–160.
- 490. Lopez-Garcia E, Schulze MB, Fung TT, Meigs JB, Rifai N, Manson JAE, Hu FB. Major dietary patterns are related to plasma concentrations of markers of inflammation and endothelial dysfunction. *The American Journal of Clinical Nutrition*. 2004;80:1029.
- 491. Zampelas A, Panagiotakos DB, Pitsavos C, Das UN, Chrysohoou C, Skoumas Y, Stefanadis C. Fish consumption among healthy adults is associated with decreased levels of inflammatory markers related to cardiovascular disease: The ATTICA study. *Journal of the American College of Cardiology*. 2005;46:120.
- 492. Lavie CJ, Milani RV, Mehra MR, Ventura HO. Omega-3 polyunsaturated fatty acids and cardiovascular diseases. *Journal of the American College of Cardiology*. 2009;54:585–594.
- 493. Kris-Etherton PM, Harris WS, Appel LJ. Fish consumption, fish oil, omega-3 fatty acids, and cardiovascular disease. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2003;23:e20–e30.

- 494. Kris-Etherton PM, Harris WS, Appel LJ. Fish consumption, fish oil, omega-3 fatty acids, and cardiovascular disease. *Circulation*. 2002;106:2747–2757.
- 495. Marchioli R, Barzi F, Bomba E, Chieffo C, Di Gregorio D, Di Mascio R, Franzosi MG, Geraci E, Levantesi G, Maggioni AP. Early protection against sudden death by n-3 polyunsaturated fatty acids after myocardial infarction. *Circulation*. 2002;105:1897–1903.
- 496. Yokoyama M, Origasa H, Matsuzaki M, Matsuzawa Y, Saito Y, Ishikawa Y, Oikawa S, Sasaki J, Hishida H, Itakura H. Effects of eicosapentaenoic acid on major coronary events in hypercholesterolaemic patients (JELIS): A randomised open-label, blinded endpoint analysis. *The Lancet*. 2007;369:1090–1098.
- 497. Mozaffarian D, Rimm EB. Fish intake, contaminants, and human health. JAMA: The Journal of the American Medical Association. 2006;296:1885–1899.
- 498. Li H, Ruan XZ, Powis SH, Fernando R, Mon WY, Wheeler DC, Moorhead JF, Varghese Z. EPA and DHA reduce LPS-induced inflammation responses in HK-2 cells: Evidence for a PPAR-γ-dependent mechanism. *Kidney International*. 2005;67:867–874.
- 499. Eibl G. The role of PPARR-γ and its interaction with COX-2 in pancreatic cancer. PPAR Research. 2008;326915:1–6.
- 500. Magee P, Pearson S, Whittingham-Dowd J, Allen J. PPARγ as a molecular target of EPA anti-inflammatory activity during TNF-α-impaired skeletal muscle cell differentiation. *The Journal of Nutritional Biochemistry*. 2012;23:1440–1448.
- 501. Lee JH, Zhou HY, Cho SY, Kim YS, Lee YS, Jeong CS. Anti-inflammatory mechanisms of apigenin: Inhibition of cyclooxygenase-2 expression, adhesion of monocytes to human umbilical vein endothelial cells, and expression of cellular adhesion molecules. *Archives of Pharmacal research*. 2007;30:1318–1327.
- 502. Santangelo C, Varì R, Scazzocchio B, Di Benedetto R, Filesi C, Masella R. Polyphenols, intracellular signalling and inflammation. *Annali-Istituto Superiore Di Sanita*. 2007;43:394.
- 503. Harris GK, Qian Y, Leonard SS, Sbarra DC, Shi X. Luteolin and chrysin differentially inhibit cyclooxygenase-2 expression and scavenge reactive oxygen species but similarly inhibit prostaglandin-E₂ formation in Raw264.7 cells. *The Journal of Nutrition*. 2006;136:1517–1521.
- 504. Covas MI. Olive oil and the cardiovascular system. *Pharmacological Research*. 2007;55:175–186.
- 505. Cicerale S, Lucas L, Keast R. Biological activities of phenolic compounds present in virgin olive oil. *International Journal of Molecular Sciences*. 2010;11:458–479.
- 506. Neveu V, Perez-Jiménez J, Vos F, Crespy V, du Chaffaut L, Mennen L, Knox C, Eisner R, Cruz J, Wishart D, Scalbert A. Phenol-explorer: An online comprehensive database on polyphenol contents in foods. *Database*. 2010;doi: 10.1093/database/bap024.
- 507. Scalbert A, Williamson G. Dietary intake and bioavailability of polyphenols. *The Journal of Nutrition*. 2000;130:2073S–2085S.
- Rahman I, Biswas SK, Kirkham PA. Regulation of inflammation and redox signaling by dietary polyphenols. *Biochemical Pharmacology*. 2006;72:1439–1452.
- 509. Sharma R, Gescher A, Steward W. Curcumin: The story so far. *European Journal of Cancer*. 2005;41:1955–1968.
- 510. Pauwels EKJ. The protective effect of the Mediterranean diet: Focus on cancer and cardiovascular risk. *Medical Principles and Practice*. 2011;20:103–111.
- 511. Dolinsky VW, Dyck JRB. Calorie restriction and resveratrol in cardiovascular health and disease. *Biochimica et Biophysica Acta (BBA)-Molecular Basis of Disease*. 2011;1812:1477–1489.
- Yap S, Qin C, Woodman OL. Effects of resveratrol and flavonols on cardiovascular function: Physiological mechanisms. *Biofactors*. 2010;36:350–359.
- Zhao L, Lee JY, Hwang DH. Inhibition of pattern recognition receptor-mediated inflammation by bioactive phytochemicals. *Nutrition Reviews*. 2011;69:310–320.

- 514. Lucas L, Russell A, Keast R. Molecular mechanisms of inflammation. Antiinflammatory benefits of virgin olive oil and the phenolic compound oleocanthal. *Current Pharmaceutical Design*. 2011;17:754–768.
- 515. Surh YJ, Kundu JK, Na HK, Lee JS. Redox-sensitive transcription factors as prime targets for chemoprevention with anti-inflammatory and antioxidative phytochemicals. *The Journal of Nutrition*. 2005;135:2993S–3001S.
- González-Gallego J, García-Mediavilla M, Sánchez-Campos S, Tuñón MJ. Fruit polyphenols, immunity and inflammation. *British Journal of Nutrition*. 2010;104:S15–S27.
- 517. Guo W, Kong EH, Meydani M. Dietary polyphenols, inflammation, and cancer. *Nutrition and Cancer*. 2009;61:807–810.
- 518. Pan MH, Lai CS, Dushenkov S, Ho CT. Modulation of inflammatory genes by natural dietary bioactive compounds. *Journal of Agricultural and Food Chemistry*. 2009;57:4467–4477.
- 519. Kim YS, Young MR, Bobe G, Colburn NH, Milner JA. Bioactive food components, inflammatory targets, and cancer prevention. *Cancer Prevention Research*. 2009;2:200–208.
- 520. Karlsen A, Retterstøl L, Laake P, Paur I, Kjølsrud-Bøhn S, Sandvik L, Blomhoff R. Anthocyanins inhibit nuclear factor-kappa beta activation in monocytes and reduce plasma concentrations of pro-inflammatory mediators in healthy adults. *The Journal of Nutrition*. 2007;137:1951–1954.
- 521. Halliwell B. Reactive oxygen species in living systems: Source, biochemistry, and role in human disease. *The American Journal of Medicine*. 1991;91:S14–S22.
- 522. Halliwell B, Gutteridge J. Oxygen toxicity, oxygen radicals, transition metals and disease. *Biochemical Journal*. 1984;219:1.
- 523. Miyamoto N, Izumi H, Miyamoto R, Kondo H, Tawara A, Sasaguri Y, Kohno K. Quercetin induces the expression of peroxiredoxins 3 and 5 via the Nrf2/Nrf1 transcription pathway. *Investigative Ophthalmology & Visual Science*. 2011;52:1055–1063.
- 524. Rushworth SA, Ogborne RM, Charalambos CA, O'Connell MA. Role of protein kinase C [delta] in curcumin-induced antioxidant response element-mediated gene expression in human monocytes. *Biochemical and Biophysical Research Communications*. 2006;341:1007–1016.
- 525. Hanneken A, Lin FF, Johnson J, Maher P. Flavonoids protect human retinal pigment epithelial cells from oxidative-stress-induced death. *Investigative Ophthalmology & Visual Science*. 2006;47:3164–3177.
- 526. Tanigawa S, Fujii M, Hou DX. Action of Nrf2 and Keap1 in ARE-mediated NQO1 expression by quercetin. *Free Radical Biology and Medicine*. 2007;42:1690–1703.
- 527. Kluth D, Banning A, Paur I, Blomhoff R, Brigelius-Flohé R. Modulation of pregnane x receptor-and electrophile responsive element-mediated gene expression by dietary polyphenolic compounds. *Free Radical Biology and Medicine*. 2007;42:315–325.
- 528. Chen CY, Jang JH, Li MH, Surh YJ. Resveratrol upregulates heme oxygenase-1 expression via activation of NF-E2-related factor 2 in PCL2 cells. *Biochemical and Biophysical Research Communications*. 2005;331:993–1000.
- 529. Kode A, Rajendrasozhan S, Caito S, Yang SR, Megson IL, Rahman I. Resveratrol induces glutathione synthesis by activation of Nrf2 and protects against cigarette smoke-mediated oxidative stress in human lung epithelial cells. *American Journal of Physiology-Lung Cellular and Molecular Physiology*. 2008;294:L478–L488.
- 530. Hsieh T, Lu X, Wang Z, Wu JM. Induction of quinone reductase NQO1 by resveratrol in human K562 cells involves the antioxidant response element ARE and is accompanied by nuclear translocation of transcription factor Nrf2. *Medicinal Chemistry*. 2006;2:275–285.
- 531. Ben-Dor A, Steiner M, Gheber L, Danilenko M, Dubi N, Linnewiel K, Zick A, Sharoni Y, Levy J. Carotenoids activate the antioxidant response element transcription system. *Molecular Cancer Therapeutics*. 2005;4:177.

- 532. Gong P, Hu B, Cederbaum AI. Diallyl sulfide induces heme oxygenase-1 through MAPK pathway. *Archives of Biochemistry and Biophysics*. 2004;432:252–260.
- 533. Keum YS, Yu S, Chang PPJ, Yuan X, Kim JH, Xu C, Han J, Agarwal A, Kong ANT. Mechanism of action of sulforaphane: Inhibition of p38 mitogen-activated protein kinase isoforms contributing to the induction of antioxidant response element–mediated heme oxygenase-1 in human hepatoma HEPG2 cells. *Cancer Research*. 2006;66:8804.
- 534. Thimmulappa RK, Mai KH, Srisuma S, Kensler TW, Yamamoto M, Biswal S. Identification of Nrf2-regulated genes induced by the chemopreventive agent sulforaphane by oligonucleotide microarray. *Cancer Research*. 2002;62:5196.
- 535. Garg R, Gupta S, Maru GB. Dietary curcumin modulates transcriptional regulators of phase I and phase II enzymes in benzo [a] pyrene-treated mice: Mechanism of its antiinitiating action. *Carcinogenesis*. 2008;29:1022–1032.
- 536. Dickinson DA, Iles KE, Zhang H, Blank V, Forman HJ. Curcumin alters EPRE and AP-1 binding complexes and elevates glutamate-cysteine ligase gene expression. *The FASEB Journal*. 2003;17:473–475.
- 537. Nakamura Y, Miyoshi N. Electrophiles in foods: The current status of isothiocyanates and their chemical biology. *Bioscience, Biotechnology, and Biochemistry*. 2010;74:242–255.
- 538. Maines MD. Heme oxygenase: Function, multiplicity, regulatory mechanisms, and clinical applications. *The FASEB Journal*. 1988;2:2557–2568.
- 539. Ryter SW, Alam J, Choi AMK. Heme oxygenase-1/carbon monoxide: From basic science to therapeutic applications. *Physiological Reviews*. 2006;86:583–650.
- 540. Morse D, Pischke SE, Zhou Z, Davis RJ, Flavell RA, Loop T, Otterbein SL, Otterbein LE, Choi AMK. Suppression of inflammatory cytokine production by carbon monoxide involves the JNK pathway and AP-1. *Science's STKE*. 2003;278:36993.
- 541. Otterbein LE, Bach FH, Alam J, Soares M, Lu HT, Wysk M, Davis RJ, Flavell RA, Choi AMK. Carbon monoxide has anti-inflammatory effects involving the mitogenactivated protein kinase pathway. *Nature Medicine*. 2000;6:422–428.
- 542. Sawle P, Foresti R, Mann BE, Johnson TR, Green CJ, Motterlini R. Carbon monoxidereleasing molecules (CO-RMs) attenuate the inflammatory response elicited by lipopolysaccharide in Raw264. 7 murine macrophages. *British Journal of Pharmacology*. 2005;145:800–810.
- 543. Soares MP, Seldon MP, Gregoire IP, Vassilevskaia T, Berberat PO, Yu J, Tsui TY, Bach FH. Heme oxygenase-1 modulates the expression of adhesion molecules associated with endothelial cell activation. *The Journal of Immunology*. 2004;172:3553–3563.
- 544. Haskell WL, Lee I-M, Pate RR, Powell KE, Blair SN, Franklin BA, Macera CA, Heath GW, Thompson PD, Bauman A. Physical activity and public health. Updated recommendation for adults from the American College of Sports Medicine and the American Heart Association. *Circulation*. 2007;116:1081–1093.

5 Osteoporosis

David W. Dodington and Wendy E. Ward

CONTENTS

5.1	Introduction		
5.2	Molecular Etiology of Osteoporosis		
	5.2.1	Mechanisms of Bone Loss	212
	5.2.2	Prevention of Bone Loss	213
5.3	Dietar	y Mechanisms Underlying Prevention of Osteoporosis	215
	5.3.1	Estrogen Receptor-Mediated Pathways: Isoflavones	215
	5.3.2	Aryl Hydrocarbon Receptor: Resveratrol, Isoflavones	216
	5.3.3	Bone Morphogenetic Protein Signaling: Soy Foods, Hesperitin	216
	5.3.4	Wnt/β-catenin Signaling: Blueberry, Resveratrol	217
	5.3.5	Peroxisome Proliferator-Activated Receptors: Fatty Acids	218
	5.3.6	Inflammation and Reactive Oxygen Species	218
	5.3.7	Prostaglandins and Fatty Acids	219
	5.3.8	Antioxidant Activity and Collagen Synthesis: Vitamin C	220
	5.3.9	Vitamin D Receptor Pathway	221
	5.3.10	Dietary Insufficiencies	222
5.4	Physica	al Activity Mechanisms Underlying Prevention of Osteoporosis	222
5.5	Recommendations		224
	5.5.1	Whole Foods and/or Food Components	224
	5.5.2	Physical Activity	225
Acknowledgment			
References			

5.1 INTRODUCTION

Osteoporosis is characterized as a thinning of skeletal tissue that occurs due to a substantial loss of hydroxyapatite crystals and bone matrix proteins. This phenomenon results in a weakened skeleton that is more prone to debilitating fragility fractures—the ultimate outcome that lifestyle strategies including diet and exercise aim to prevent. Osteoporosis is defined as "a skeletal disorder characterized by compromised bone strength, predisposing to an increased risk of fracture."¹ The emphasis is on compromised bone strength, and recognition that bone strength is altered by both the quantity and quality of a skeletal site. Bone quantity is routinely measured clinically as bone mineral density (BMD) by dual-energy X-ray absorptiometry, whereas bone quality is more challenging to assess but includes consideration of how mineral and matrix proteins are integrated to result in structurally strong bone. Sophisticated imaging using quantitative computed tomography and magnetic resonance imaging are proving useful in more fully understanding how changes in bone structure and mineral content relate to risk of fragility fracture. In animal models, it is possible to directly measure the strength, specifically the maximum load a specific skeletal site can withstand before fracture, with a materials testing system.

The International Osteoporosis Foundation reports that "worldwide, an osteoporotic fracture is estimated to occur every 3 seconds, a vertebra fracture every 22 seconds"—with a total of 200 million women suffering from osteoporosis throughout the world.² Men are also susceptible to osteoporosis albeit at lower risk than women. Genetics is a major risk factor for developing osteoporosis. Other risk factors include lifestyle—namely nutrition and physical activity—and since they are modifiable, it is especially important to consider their potential impact. Poor nutrition and sedentary lifestyle are independently linked to poorer bone health throughout the life cycle. Although osteoporosis is often viewed as a disease of aging, bone health is in fact a lifelong process. At any stage of the life cycle, the amount of mineral in the skeleton and skeletal strength is an accumulation of what has occurred in utero through to the present age.³ As such, it can be argued that osteoporosis is a pediatric disease with geriatric consequences. Working from this premise fits well with the focus of this book, namely that a disease starts because of cellular dysfunction early in life and that actual symptoms of the disease state often do not appear until much later in the life cycle. However, most research investigating effects of whole foods and food components on bone health has focused on adult populations and using animal models that mimic aging, many of which have used ovariectomized (OVX) rodent models.

5.2 MOLECULAR ETIOLOGY OF OSTEOPOROSIS

5.2.1 MECHANISMS OF BONE LOSS

Within the context of bone health and prevention of osteoporosis, symptoms are often silent until fragility fracture occurs—the ultimate endpoint that prevention and treatment strategies aim to avoid. Osteoporosis, or a heightened risk of fracture, occurs when the rate of bone formation and the rate of bone resorption are not balanced. Essentially, bone formation may be reduced, bone resorption may be elevated, or both situations may occur. The resulting loss of mineral, matrix, and overall bone microarchitecture leads to a fragile skeleton. There are three main cell types found in bone: the osteoblast, the osteocyte, and the osteoclast. Osteoblasts are the bone-forming cells that are derived from mesenchymal stem cells (MSCs). They are responsible for producing and mineralizing the bone matrix. As osteoblasts become isolated in the bone matrix, they become osteocytes, which cease to produce osteoid and mineralized matrix. Osteoclasts are derived from the monocyte lineage, which is derived from hematopoietic stem cells. Osteoclasts are, therefore, related to other inflammatory cell types such as macrophages. Osteoclasts are responsible for the degradation of bone tissue through release of acids and proteases. Thus, balanced activity of osteoblasts and osteoclasts is essential to maintaining a healthy skeleton. Further discussion of the molecular regulation of osteoblasts and osteoclasts and their interconnected activities are extensively covered in textbooks and review articles.

Bone loss is most pronounced during aging. The loss of estrogen after menopause or after ovariectomy (in rodent models) is a well-characterized phenomenon that accounts for some of the deterioration of bone tissue during aging. Estrogen is known to stimulate osteoclast apoptosis while blocking apoptosis of osteoblasts and osteocytes. Estrogen withdrawal may also mediate its effects through the regulation of T cells. A recent review⁴ highlights that T cells are involved in the deterioration of bone tissue that occurs after estrogen withdrawal. Specifically, T cell activation results in greater production of tumor necrosis factor-alpha (TNF- α) that leads to greater production of RANKL and subsequently osteoclast formation. Thus, estrogen withdrawal is recognized as a principle event in the development of osteoporosis. Additionally, increases in oxidative stress with aging contribute to the pathogenesis of bone loss. Traditionally, osteocytes were considered to be "inactive" once they became embedded in calcified bone matrix, but a recent review makes the case that a decline in osteocyte number is associated with reductions in bone strength⁵ and that an increase in oxidative stress with aging may be a main mechanism by which a reduction in osteocyte number occurs.

5.2.2 PREVENTION OF BONE LOSS

Although the role of diet and exercise on symptoms, particularly BMD, has been extensively studied and reported on, the role of diet and exercise in preventing or attenuating cellular dysfunction in bone tissue—the focus of this chapter—has been less studied. It is now recognized that several mechanisms that mediate bone cell activity can be modulated by foods, food components, and physical activity. At the cellular level, foods and food components can impact estrogen-mediated pathways, the aryl hydrocarbon receptor (AHR), bone morphogenetic protein (BMP) signaling, Wnt/ β -catenin signaling, peroxisome proliferator–activated receptors (PPARs), and reactive oxygen species (ROS). Physical activity mediates effects on bone cell activity through mechanical signals that are transduced and transformed into biochemical signals.

This chapter focuses on specific mechanisms by which foods and/or food components or physical activity can regulate osteoblast and/or osteoclast activity and thus may prevent dysfunction of bone cells that often accompanies the aging process. From a cellular perspective, information on how nutrition, specifically food components modulate bone cell activity is not extensive and is an expanding area of research study. Likewise, the mechanisms by which physical activity converts its mechanical signals to biochemical signals within bone cells is an active and ongoing area of study. These mechanisms are shown in Figure 5.1 that was developed to summarize, to the best of our knowledge, regulatory pathways of osteoblasts and osteoclasts that are modulated by specific food components or physical activity.



FIGURE 5.1 Impact of physical exercise and dietary modification on mechanisms of osteoporosis. Physical exercise as well as phytochemicals and nutrients from whole foods alter activities of signal transduction pathways to promote apoptosis and reduce the activation of osteoclasts as well as to promote osteoblastogenesis through reducing proinflammatory signaling, thus reducing risk for osteoporosis.

5.3 DIETARY MECHANISMS UNDERLYING PREVENTION OF OSTEOPOROSIS

5.3.1 ESTROGEN RECEPTOR-MEDIATED PATHWAYS: ISOFLAVONES

There are several mechanisms by which estrogen receptor (ER) signaling affects bone metabolism. First are the genomic ER pathways. Because estrogens are hydrophobic molecules, they can readily diffuse across cell membranes and bind to the nuclear ERs.⁶ The ER contains a ligand-binding domain and a DNA-binding domain. On binding its ligand, the ERs dimerize and bind to specific DNA sites known as estrogen response elements, which regulate the transcription of estrogen responsive genes. In the classical genomic ER pathway, ER binding in osteoblasts results in the recruitment of coactivators that enhance transcription of genes responsible for osteoclast apoptosis.^{7–9} For example, increased expression of Fas-L on the surface of osteoblasts will result in increased activation of Fas on the surface of osteoclasts, leading to the apoptosis of osteoclasts. Additionally, increased expression of osteoprotegrin (OPG) will suppress osteoclast activation because it acts as a soluble RANKL receptor decoy.

In the nonclassical genomic ER pathway, the liganded ER can indirectly regulate gene expression by interacting with other DNA-bound transcription factors. When the ligated ER interacts with NF $\kappa\beta$, for example, it results in a decreased expression of macrophage colony stimulating factor (M-CSF), interleukin-6 (IL-6), and TNF- α .^{10–13} Because M-CSF, IL-6, and TNF- α activate osteoclasts by upregulating RANKL on osteoblasts, their inhibition has a protective effect on bone.

Estrogens can also exert their effects through nongenomic mechanisms. Estrogen can activate rapid signaling pathways (MAPK–ERK pathway) downstream of growth factor receptors (e.g., insulin-like growth factor-1 [IGF-1] and transforming growth factor- β [TGF- β]). Activation of the mitogen-activated protein kinase (MAPK) pathway stimulates proliferation in osteoblasts and apoptosis in osteoclasts.¹⁴ ERs located in the plasma membrane of osteoblasts and osteoclasts can induce rapid signaling pathways to also result in the stimulation of apoptosis in osteoclasts while simultaneously attenuating apoptosis in osteoblasts.^{8,15}

The term "weakly estrogenic" or "weak estrogen" is often used to describe nutrients that are thought to act in a similar way to estrogens by activating these ER pathways. These include isoflavones, coumestrol, and lignans. The best characterized of these weakly estrogenic nutrients are the isoflavones, genistein, and daidzein. These isoflavones interact with both the alpha and beta ER isoforms; however, affinity for ER beta is much greater.^{16,17} Daidzein has a lower affinity for ERs than genistein, but at saturable levels both have been shown to activate transcription from estrogen response elements (EREs) as effectively as 17 β -estradiol. Consuming a diet rich in soy produces high physiological serum levels of free genistein and activation of ER beta.¹⁸ Physiological concentrations of genistein suppress IL-6 production and boost OPG production in human osteoblast-derived cell lines.^{19,20} Similarly, genistein and daidzein, as well as coumestrol directly suppress osteoclastogenesis induced by TNF- α , an effect that was negated by an ER antagonist.²¹ Epidemiological evidence suggests that consuming whole soy foods rather than isolated isoflavones is beneficial to bone health, particularly for fracture prevention. Clinical intervention studies regarding isoflavone supplementation often yield inconsistent findings; however, McCarty¹⁸ suggests that this variability might be due to the fact that some isoflavones are administered as glycosides or in conjunction with soy protein. The soy or isoflavone effect on bone health, and whether it is better to study soy as a whole food to understand its role in preventing deterioration of bone tissue during aging continues to be actively studied.

5.3.2 ARYL HYDROCARBON RECEPTOR: RESVERATROL, ISOFLAVONES

The AHR is responsible for detecting environmental contaminants such as 2,3,7,8-tetrachlorodibenzo-p-dioxin (TCDD).²² TCDD is a toxin that can be found in cigarettes. When the AHR is activated by toxin, it recruits the aryl hydrocarbon receptor nuclear translocator (ARNT). This causes a problem because ARNT is also a coactivator of ER β , so when AHR recruits ARNT, it can interfere with ER signaling.²³ AHR signaling inhibits the proliferation of rat calvarial osteoblast-like cells and mouse calvarial clonal preosteoblastic cells. It reduces the rate of calcium deposition and depresses mRNA expression of osteocalcin, an important protein implicated in mineralization and calcium homeostasis.²⁴ Several nutrients are reported to be able to inhibit the AHR and thus may improve bone health. Phytochemicals such as quercetin, kaempferol, several catechins, curcumin, and lutein are reported to act as antagonists for AHR.25 Resveratrol, a polyphenolic compound found in the skin of red grapes and other fruits, is an AHR antagonist and prevents the damaging effects of TCDD on bone formation. TCDD alters the expression of bone specific genes; however, this effect is antagonized by resveratrol. Resveratrol has also been shown to completely antagonize the loss of mineralization induced by TCDD.²⁶ However, isoflavones such as daidzein cause AHR activation in vitro. This suggests that including small quantities of AHR inducers in foods may have some beneficial effects in the proliferation and differentiation of cells in animals.²⁷

5.3.3 BONE MORPHOGENETIC PROTEIN SIGNALING: SOY FOODS, HESPERITIN

BMPs are members of the TGF- β superfamily. More than 20 BMP-related proteins have been identified and can be subdivided into several groups based on their structures and functions.²⁸ BMP2, BMP4, BMP6, and BMP7 act as major osteogenic inducers.²⁹ BMPs bind to two distinct serine/threonine kinase receptors and induce phosphorylation of Smad proteins known as BR-Smads. BR-Smads (Smad1, Smad5, Smad8) form complexes with Co-Smad (Smad4), and move into the nucleus, where they regulate transcription of target genes.³⁰

BMP signaling is an extremely important component of bone development. DNA microarray analyses have revealed 184 BMP-early response genes that drive osteoblastic differentiation.³¹ BMP signaling is also notably important because it activates Runx2, a master transcription factor in bone development. BMP does not directly induce the expression of Runx2, but it facilitates expression of Dlx5 in osteoblasts, and Dlx5 then induces expression of Runx2 in osteoprogenitor cells.³² On activation of BMP signaling, Runx2 and BR-Smads physically interact with each other and cooperatively regulate the transcription of target genes.³³ Runx2 promotes MSCs toward the osteoblastic lineage rather than the adipocyte lineage.

There are a number of nutrients that are thought to modulate bone metabolism through this BMP/Smad/Runx2 pathway. Soy-based foods, in addition to acting through ER signaling, as previously described, may impact this pathway. When OVX mice are supplemented with estrogen, there is a decline in the level of bone formation and bone resorption markers, but when OVX mice are supplemented with soy protein isolate (SPI), bone formation markers are elevated and there is a decline in bone resorption markers.³⁴ This suggests that SPI is largely anabolic and nonestrogenic. SPI may exert its effects by activating BMP signaling. Feeding SPI increases expression of BMP2 mRNA in bone, accompanied by increased Smad 1/5/8 phosphorylation and increased Runx2 protein.³⁵ Though the active components have not yet been identified, icariin, a flavonoid component of soy foods, has been shown to activate the BMP pathway and has also shown to be more potent than genistein in promoting osteoblast differentiation and mineralization.³⁶ Another bioactive compound that may act through this pathway is hesperitin. Hesperidin, a compound found in citrus fruits, is hydrolyzed by the gut microflora into hesperitin and conjugated into hesperetin-7-O-glucuronide (Hp7G).37 In vitro studies have demonstrated Hp7G's ability to stimulate osteoblast differentiation as well as an increase in Smad1/5/8 phosphorylation and Runx2 expression.³⁷

5.3.4 WNT/ β -catenin Signaling: Blueberry, Resveratrol

The canonical Wnt signaling pathway plays a crucial role in regulating the differentiation of osteoblasts.³⁸ In the absence of the extracellular Wnt signaling molecule, β -catenin, the intracellular signaling protein is held in a complex with Axin, APC, and GSK3 β . GSK3 β will phosphorylate β -catenin, targeting it for degradation. On activation of the Wnt receptor, β -catenin is released; it enters the nucleus and complexes with the transcription factor T-cell factor (TCF)/Lef to regulate gene expression. High levels of nuclear β -catenin/TCF drive osteogenesis, whereas lower levels drive chondrogenesis. Of great importance is the fact that an increase in Wnt/ β catenin signaling leads to higher expression of Runx2 and the subsequent commitment of MSCs to the osteoblast lineage.³⁹

Recent studies have implicated this pathway in the reported beneficial effects of blueberries on the accumulation of bone mineral. Blueberry-supplemented diets have been shown to increase Wnt/ β -catenin signaling and Runx2 expression.⁴⁰ This effect may be mediated by the MAP kinase p38 as it shows higher phosphorylation after consumption of blueberries. This is further supported because blueberry-induced Wnt/ β -catenin is stunted by a p38 inhibitor. Once phosphorylated, p38 inhibits the activity of GSK3 β , resulting in increased nuclear β -catenin/TCF. Blueberry feeding has also been shown to increase serum levels of hippuric acid, phenylacetic acid, and hydrobenzoic acid. An artificial mixture of these phenolic compounds has been shown to mimic the effects of blueberries; however, it is still unclear how these compounds activate p38. There also exists the possibility that other fruits such as dried plum might be beneficial through a similar mechanism. Another bioactive that has been shown to favorably modulate Wnt/ β -catenin signaling is resveratrol. MSCs treated with

resveratrol show increased nuclear β -catenin/TCF and increased Runx2 expression.⁴¹ There is evidence that this effect is mediated through extracellular signal–related kinase 1 and 2 (ERK1/2). Resveratrol treatment increases ERK activation; ERK in turn inhibits GSK3 β via phosphorylation. The effect of resveratrol is blocked by treatment with an ERK inhibitor.

5.3.5 PEROXISOME PROLIFERATOR-ACTIVATED RECEPTORS: FATTY ACIDS

PPARs are a group of transcription factors that regulate energy metabolism and cell differentiation.⁴² In bone, PPAR γ is particularly interesting because it plays the opposite role to Runx2. PPAR γ is a master transcription factor that inhibits osteogenesis and promotes adipogenesis. An important regulator of PPAR γ is the CCAAT/ enhancer-binding protein (CEBP) that seems to be indispensable for triggering adipogenesis.⁴³ PPAR γ agonists that include certain antidiabetic drugs have been shown to cause bone loss and increase bone marrow fat accumulation.⁴² Additionally, there is a reciprocal relationship between the expression of PPAR γ and Wnt/ β -catenin signaling may suppress adipogenesis by reducing the expression of PPAR γ and CEBP mRNA,⁴⁵ but the mechanism remains to be investigated at the molecular level.

This balance may help explain the association between high-fat diet and impaired bone health. Bone marrow stromal cells, when exposed to serum from high-fat-fed rats, show increased PPAR γ levels and inhibition of Wnt/ β -catenin signaling compared to serum from low-fat-fed rats.⁴⁴ A similar pattern is observed when the cells are exposed to an artificial mixture of nonesterified fatty acids (NEFAs) mimicking the concentration and composition of NEFA after high-fat feeding. However, it is unclear whether fatty acid metabolites affect PPAR γ secondary to Wnt/ β -catenin inhibition or if they can activate PPAR γ directly. There has also been suggestion that polyunsaturated fatty acids (PUFAs) might regulate PPAR γ because they lower adipose mass and alter bone mineral content in mice.⁴⁶ Dietary fat and PUFAs may also modulate bone metabolism through inflammatory/anti-inflammatory mechanisms; this is discussed in the Section 5.3.7 of this chapter.

PPARγ may also be modulated by resveratrol independently of Wnt/β-catenin signaling. Resveratrol activates SIRT1, a nuclear protein deacetylase. The mechanism is unclear, but it is thought to affect SIRT1 indirectly by increasing the activation of adenosine monophosphate-activated protein kinase (AMPK), which increases nicotinamide adenine dinucleotide (NAD) and in turn activates SIRT1.⁴⁷ Activation of SIRT1 by resveratrol decreases adipocyte development via inhibition of PPAR_γ, which in turn promotes osteoblast differentiation.⁴⁸

5.3.6 INFLAMMATION AND REACTIVE OXYGEN SPECIES

Osteoporosis is largely an inflammatory disease. In fact, epidemiologic studies have shown that levels of systemic inflammation are able to predict bone loss and future fracture risk.⁴⁹ Furthermore, several inflammatory diseases are associated with bone resorption including rheumatoid arthritis, systemic lupus erythematosus, inflammatory bowel disease, celiac disease, cystic fibrosis, and chronic obstructive pulmonary

disease.⁵⁰ Along with inflammation, ROS plays a synergistic role in promoting bone loss as ROS leads to production of the proinflammatory cytokines TNF α , IL-1 β , and IL-6.⁵¹

One of the principle mechanisms for inflammation-induced bone loss is through the RANKL–RANK interaction. Cytokines bind to homo- or heterodimeric Janus kinase (JAK) receptors resulting in the phosphorylation of signal transducer and activator of transcription (STAT) proteins. Once the STATs are phosphorylated, they are released from the JAKs and subsequently dimerize and then enter the nucleus where they regulate gene expression. The proinflammatory cytokines exert many genomic effects on the osteoblast and one of these critical effects is the increase in expression of RANKL on the surface of osteoblasts. RANKL will then interact with RANK on the surface of preosteoclasts causing them to become mature bone-resorbing osteoclasts. ROS are also able to upregulate RANKL in osteoblasts via the activation of the MAP kinase ERK that signals through STAT3.³⁴ Inflammatory cytokines and ROS also stimulate osteoclasts directly in two ways. First, they can upregulate RANKL in osteoclasts leading to increased RANKL–RANK interaction, and second, they can actually increase the number of osteoclast precursors via stimulating the production of M-CSF.⁵¹

Inflammation and ROS can also affect bone metabolism independently of RANKL–RANK interaction. ROS production in MSCs has been shown to impair osteoblastogenesis and stimulate adipogenesis as a result of impaired Wnt/ β -catenin/Runx2 signaling and increased PPAR γ signaling.⁵² This is thought to occur because the DNA damage repair gene FoxO will divert β -catenin away from TCF-mediated transcription.⁵³ Additionally, TNF- α has been shown to inhibit osteoblast function by inducing vitamin D resistance, blocking collagen synthesis, and stimulate osteoblast apoptosis.⁵¹

Intuitively, any nutrient or activity that minimizes inflammation and ROS should be protective of bone. Given that estrogen signaling, for example, is known to antagonize the effects of ROS by upregulating antioxidant systems and inhibiting cytokine production,⁵⁴ this may be a mechanism by which food components with estrogenic activity (i.e., isoflavones) contribute to the protective effects. There are also numerous nonestrogenic methods of reducing inflammation and ROS in bone. Dietary antioxidants such as *N*-acetylcysteine and vitamin E have been shown to be effective in preventing bone loss postmenopause, in aging, and in alcohol abusers. In vitro studies using bone cells have demonstrated that exposure to dried plum polyphenol extracts inhibits TNF- α induced RANKL expression and oxidative stress with coincident upregulation of Runx2 and osterix (Osx).^{55,56} Similar inhibition of RANKL– RANK signaling has been reported by resveratrol.⁵⁷ However, it is unknown whether sufficient concentrations of plum polyphenol or resveratrol can be achieved through diet alone, supplemental levels may be required.

5.3.7 PROSTAGLANDINS AND FATTY ACIDS

Prostaglandins (PGs) are lipids that act in an autocrine/paracrine manner via G-protein coupled receptors.⁵⁸ PGs are made from arachidonic acid, an n-6 PUFA, which is released from membrane phospholipids by phospholipase A2 enzymes. Arachidonic acid is then converted by cyclooxygenase (COX) to PGH2. Terminal synthases convert PGH2 into PGs of which PGE2 is known to modulate bone

metabolism. PGE2 binds its receptors EPR2R and EP4R and activates cyclic adenosine monophsophate (cAMP) dependent pathways.

Clear roles for PGE2 in bone metabolism have been difficult to define because it seems to play a role in both bone resorption and bone formation. The effects on bone resorption are primarily mediated by the upregulation of RANKL and downregulation of OPG that results in increased osteoclast activity. Furthermore, PGE2 has been shown to enhance inflammatory cytokine production (IL-1, IL-6, TNF- α) and release of parathyroid hormone (PTH) that also enhances osteoclast activation.59 Though the principle role of PGE2 in bone was thought to be osteoclast-mediated resorption, the importance of PGE2 in bone formation is now also recognized. Local application of EP2R agonists⁶⁰ and local and systemic application of EP4R agonists⁶¹⁻⁶³ enhance bone healing in rodent models while endogenous prostaglandin production promotes osteoblast differentiation that is inhibited by COX2 deletion.64 There are a number of factors that help define the role of PGE2. When prostaglandin production is induced by BMP-2, it enhances BMP-2 stimulation of osteoblasts and osteoclasts while PTH enhances PTH stimulation of osteoclasts.58 Most importantly, research using rodent models suggests that while continual administration of PGE2 may lead to bone loss, intermittent administration is anabolic.65 Overall, more research is needed to clearly define the role of PGE2 in bone metabolism. But nonetheless, this pathway may be modulated by the fatty acid composition of the diet. By decreasing the ratio of n-6 PUFA to n-3 PUFA, there is more n-3 PUFA incorporated into the cell membrane and therefore less arachidonic acid available for PGE2 synthesis. The additional n-3 PUFA can modulate COX protein expression causing a reduction in PGE2 production.⁶⁶ Diets rich in n-3 PUFA with a low n-6/n-3 ratio are also associated with lower proinflammatory cytokine levels. Overall, the effects on bone are positive with most studies showing a decrease in RANKL and/or increase in OPG from n-3 PUFA interventions.

5.3.8 ANTIOXIDANT ACTIVITY AND COLLAGEN SYNTHESIS: VITAMIN C

Vitamin C is well known as an essential nutrient for bone metabolism due to its critical role in collagen synthesis and for its antioxidant capabilities. When endogenous ascorbic acid synthesis is inhibited, mice develop osteopenia and have spontaneous fractures under conditions of increased ascorbic acid requirements and this is reversed with ascorbic acid supplementation.⁶⁷ Vitamin C is proposed to play a dual role in bone metabolism. The first is using its antioxidant capabilities to reduce osteoclast activity, and second, it is an essential cofactor in osteoblast differentiation.⁶⁷ The mechanisms have not been thoroughly elucidated, but in vitro studies have shown that treatment with ascorbic acid causes a sixfold increase in osterix, a marker of osteoblast differentiation, in bone marrow stromal cells.⁶⁸ In this situation, treatment with ascorbic acid enhanced the binding of a transcription factor (NrF1) to an antioxidant response element that regulates the production of osterix. Similarly, ascorbic acid is a cofactor for the enzyme that hydroxylates oxygen-dependent hypoxia-inducible factor.⁶⁹ Hydroxylation of this factor increase nuclear translocation that in turn increases osteogenesis by elevating vascular endothelial growth factor levels in osteoblasts.⁷⁰ In addition to modulating osteoblast and osteoclast activity, ascorbic acid is needed for normal bone matrix

production.⁷¹ Ascorbic acid reduces the iron prosthetic group of certain hydroxylase enzymes that are essential in collagen biosynthesis and collagen cross-linking.

5.3.9 VITAMIN D RECEPTOR PATHWAY

Vitamin D is a steroid hormone produced in the skin on exposure to ultraviolet radiation or obtained in the diet. Vitamin D is hydroxylated, primarily in the liver, to its circulating form $25(OH)D_3$ by the enzyme 25-hydroxylase. It is hydroxylated again, primarily in the kidney, to its active form $1,25(OH)_2D_3$ by the enzyme 1-hydroxylase. When blood calcium levels are low, PTH is released and increases the activity of 1-hydroxylase to generate more active vitamin D. To exert its effects, $1,25(OH)_2D_3$ binds to the vitamin D receptor (VDR), a nuclear hormone receptor that is expressed throughout the entire body but is particularly abundant in kidney, intestine, and bone. When vitamin D binds to the VDR, it forms a complex with its heterodimeric partner: retinoic X receptor (RXR) so that it can penetrate the deep groove of DNA and bind to the vitamin D responsive elements of vitamin D-regulated genes.⁷²

The classical role of vitamin is to regulate calcium homeostasis; when calcium levels are low PTH is released, which enhances the activity of 1-hydroxylase and generates more $1,25(OH)_2D_3$. This active form of vitamin D will then exert genomic effects in target tissues by binding to the VDR. In the intestine, it increases the expression of proteins needed for calcium absorption. In the kidney, it increases the expression of proteins needed for calcium reabsorption. Both these events increase blood calcium levels. In osteoblasts, however, $1,25(OH)_2D_3$ increases the expression of RANKL, the membrane-bound cytokine that activates osteoclasts. The activated osteoclast will then dissolve bone, releasing calcium into the blood. Though this last effect may seem counterproductive, the net effect is beneficial to bone because it increases overall calcium levels, which prevents chronic secretion of PTH via feedback inhibition. Also, having adequate circulating calcium causes the synthesis of calcitonin by endocrine cells in the thyroid gland. Also, calcitonin promotes the mineralization of bone.

Although vitamin D is most commonly associated with calcium homeostasis, its genomic effects extend far beyond controlling this mineral. Ligation to the VDR modulates the expression of several genes important in bone metabolism. High vitamin D will decrease transcription of 1-hydroxylase and increase transcription of 24-hydroxylase instead. This results in the formation of $24,25(OH)_2D_3$, which plays a role in mineralization rather than resorption of bone.

Vitamin D is also known to be a potent inhibitor of adipogenesis, and the mechanism is starting to be elucidated. In 3T3-L1 cells, a fibroblast-like cell, vitamin D has been shown to inhibit the expression of PPAR γ .⁷³ This occurs because ligated VDR downregulates C/EBP β mRNA expression and upregulates the C/EBP β corepressor, which decreases the transcription of PPAR γ . This will prevent differentiation toward the adipocyte lineage. There is also evidence that vitamin D can drive human adipose tissue–derived stromal cells to differentiate along an osteoblastic lineage at a comparable efficacy to dexamethasone, a potent glucocorticoid.⁷⁴ In these cells, vitamin D increased the expression of Runx2, Osx, BMP-2, BMP-6, and LMP-1. Although Runx2 commits MSCs to the osteoblastic lineage, Osx is another master transcription factor that acts downstream of Runx2 and is essential for bone mineralization.

BMPs will enhance the BMP/Smad/Runx2 axis driving osteoblastogenesis, and LMP-1 is thought to help mediate the effect of BMP-6 on bone.⁷⁵ Vitamin D also has the potential to enhance the expression of ERs.⁷⁶ This may enhance the bone-building effects of estrogens and estrogenic nutrients.

5.3.10 DIETARY INSUFFICIENCIES

A major issue that impacts the mechanisms of cause is that of generally poor dietary habits. For instance, intakes of fruits and vegetables observed in American diets are traditionally low with only a very small minority of Americans actually eating the recommended minimum number of servings of fruits and vegetables.⁷⁷ This is most likely the reason for intakes of vitamin D, calcium, and magnesium to be below the estimated average requirement (EAR) for 70%, 38%, and 45% of the population.⁷⁸ Vitamin D and magnesium, of course, are vital for maintaining appropriate calcium status. As described in detail earlier, PTH is released in response to low serum calcium, stimulating synthesis of the enzyme 25-hydroxycholecalciferol-1-hydroxylase in the kidney. This results in increased synthesis of 1,25-dihydroxycholecalciferol (calcitriol), which activates synthesis of calcium and phosphate transporters in the intestine to enhance the absorption of calcium and phosphorus.^{79–83} Magnesium is an integral component of this calcium-regulatory pathway because it is required for the release of PTH from the parathyroid gland as well as for binding of PTH to its receptor. Thus, a magnesium insufficiency is a factor in risk for osteoporosis. Lower calcitriol levels also will occur because of attenuated PTH effects, an effect that also is observed with insufficient intake of magnesium. With such a high prevalence of magnesium and vitamin D insufficiency, poor regulation of calcium absorption also would occur for a majority of the population, which when combined with insufficient calcium intake for a relatively large minority of the population leads to a populationwide risk for osteoporosis due to generally poor dietary habits.

5.4 PHYSICAL ACTIVITY MECHANISMS UNDERLYING PREVENTION OF OSTEOPOROSIS

Current bone and exercise research is aiming to identify the mechanisms by which mechanical signals are transduced and transformed into biochemical signals. It is proposed that bone cells can recognize everything from deformation, to pressure, to fluid shear, to acceleration.⁸⁴ There are also several mechanisms by which cells can recognize mechanical signals. These include stretch-activated calcium channels, which activate intracellular enzymes (e.g., phospholipase C, protein kinase C) and cause membrane depolarization with subsequent voltage-gated channel opening and further calcium entry; integrins, which are activated by deformation of their extracellular binding partners (e.g., collagen, osteopontin) by fluid shear or substrate strain; activation of G-proteins in the lipid bilayer; and cytoskeleton deformation, which provides enhanced docking and activation sites for kinases.⁸⁵ It is also worth noting that nonload-bearing exercise also increases osteoblast activity and bone strength. This has been observed when animals are exercised using high-frequency oscillatory motions.^{86,87} Therefore, some of these mechanisms of signal transduction are independent of deformation.

From here, the mechanical signals converge to increase intracellular calcium, resulting in an increase in eNOS mRNA and an increase in nitric oxide (NO). NO can then potentially stimulate proliferation by binding to a regulatory site on Raf initiating a Ras–Raf–MEK–ERK cascade.⁸⁵ The role of NO is further supported by the fact that ERK activation and cell proliferation are abrogated using an NO inhibitor.⁸⁸ Ultimately, this signaling cascade will result in the modulation of gene expression, growth, and cell differentiation.

There are additional layers of complexity that play a significant role in this pathway. For example, inhibition of COX has shown that COX-1/2 induction, downstream of NO and ERK-1/2 may be a required chain in the sequence of events leading to flow-induced bone formation.⁸⁸ IGF-1 also plays a synergistic role in inducing stressrelated proliferation.⁸⁹ Exercise increases osteoblast production of IGF-1, which in turn increases ERK activation. Additionally, the effects of mechanical loading are dependent on ER α .⁹⁰ Induction of ERK by calcium/NO/PGE2 results in increased phosphorylation of ER α ,⁹¹ and experimental studies using animals and bone-derived cells in vitro suggest that functional ER α is necessary for the adaptive response of bone to mechanical loading.⁹⁰ This may help explain the reduced mechanical loading response in postmenopausal women as a consequence of a decline in ER α number and/or function, resulting from estrogen deficiency.

Current research has shown that exercise-induced signals help drive osteoblast differentiation. In mice, 6 weeks of low-magnitude mechanical signals (LMMS) have been shown to increase Runx2 and decrease PPAR γ expression, thereby promoting osteoblastogenesis and inhibiting adipogenesis.⁹² Additionally, LMMS has been shown to actually increase the number of MSCs by 46%, thereby increasing the number of potential osteoblasts. These changes in Runx2 and PPAR γ are at least in part due to increased Wnt/ β -catenin signaling. Within minutes of mechanical stimulation, we see increased activation and nuclear translocation of β -catenin.⁹³ This is consistent with the described mechanism as induction of ERK should lead to increased phosphorylation and inhibition of GSK-3, the enzyme responsible for degrading β -catenin. Indeed, mechanical strain has been shown to suppress the activity of GSK-3.⁹⁴ Another study has also found that mechanical loading reduces the expression of Sost, which reduces the secretion of its protein product, sclerostin, in osteocytes.⁹⁵ This may have favorable effects on osteoblast differentiation because sclerostin is an antagonist of Wnt/ β -catenin signaling.⁹⁶

In addition to the actual movement and mechanical strain, duration and intensity play critical roles in determining the anabolic response of the skeleton. Bone appears to have memory of its loading history, and bone structure and strength are improved if mechanical loading is separated into short bouts.⁹⁷ In vitro studies show that continual fluid shear stress (FSS) applied to bone cells desensitizes them⁹⁸ and even a gradual increase in the magnitude of FSS does not overcome this desensitization.⁹⁹ This helps explain why in conditions such as obesity, even though there is increased load on the skeleton, bone strength is actually compromised. However, when increasing FSS is applied to bone cells with rest periods, the cells do not become sensitized.¹⁰⁰ These findings may explain the in vivo BMD changes in response to the gradual increase in the strength of exercise.

One last issue in regard to exercise is that osteoporosis is essentially an inflammatory disease. As a result, the anti-inflammatory effects of repeated exercise may prove to be important in reducing risk for osteoporosis. As discussed in detail in Chapter 2, 45-60 minutes of moderately stressful aerobic exercise is necessary to illicit an antiinflammatory response through the expression of relatively large amounts of IL-6 by muscle cells. When the exercise is repeated on a chronic basis, the anti-inflammatory responses culminate in reductions in inflammatory markers. Unfortunately, although the number of clinical intervention studies on exercise and inflammation are many, as are reviews on exercise and risk for osteoporosis, the number of clinical intervention studies that address exercise, inflammatory mechanisms, and osteoporosis are very few. Interestingly, both increases in bone density and a reduction in inflammatory markers have been observed following exercise training for ~45 minutes at ~60% heart rate max and for 3-6 days per week in postmenopausal and overweight postmenopausal women.^{101,102} These results are consistent with many studies that indicate that a minimum of 45-60 minutes of continuous exercise is necessary for anti-inflammatory effects to occur. They also are consistent with the concept that inflammatory mechanisms are a component of the etiology of osteoporosis and that physical exercise is an important vehicle to reduce risk for osteoporosis.

5.5 **RECOMMENDATIONS**

As discussed, there are many potential mechanisms by which foods and/or food components as well as physical activity modulate bone cell activity. The transfer of these cellular effects to whole bone tissue has been studied for some foods and/or food components or physical activity regimens. Identifying and more comprehensively understanding potential mechanisms of action is an important area of study to ultimately develop an overall lifestyle approach—combining healthy dietary practices and regular physical activity—to promote development and maintenance of a healthy skeleton. Based on the known mechanisms, the following recommendations are made.

5.5.1 WHOLE FOODS AND/OR FOOD COMPONENTS

For the vitamins discussed, there are dietary reference intakes that are age and gender specific but it remains a challenge to make specific recommendations regarding the quantities of whole foods or food components to consume for bone health. However, many of the foods discussed in this chapter are fruits or are components of fruits and vegetables, and both the United States and Canada have general guidelines regarding intakes of fruits or vegetables for overall health. The new "MyPlate" guidelines set by the U.S. Department of Agriculture recommends that half a plate consist of fruits and vegetables. Canada's Food Guide recommends 7–10 servings of fruits and vegetables per day for adults. By following these guidelines, risks for osteoporosis that originate from insufficient intake of various nutrients will be eliminated and go a long way to reducing the incidence of osteoporosis. Although studies investigating the relationship between intake of specific fruits and vegetables and bone health are inconclusive,¹⁰³ individuals consuming higher levels of flavonoids also have been shown to have higher BMD at femur neck and lumbar spine, common sites of fragility fracture.¹⁰⁴ Based on the data on flavonoids discussed earlier, the benefits of high flavonoid consumption would be complementary to the benefits of correcting nutrient insufficiencies. Therefore, consuming a wide variety of fruits and vegetables as part of a nutritionally sufficient diet is emphasized.

There is also evidence from human studies that suggest positive effects of consuming higher intakes of n-3 relative to n-6.^{105,106} In older men and women, a higher ratio of n-6 to n-3 fatty acids is associated with lower BMD at the hip.¹⁰⁵ Lower spine BMD was also associated with a higher ratio of n-6 to n-3 fatty acids in women not receiving hormone replacement therapy. A study in young adult males showed positive associations between n-3 fatty acids (i.e., docosahexaenoic acid, a long-chain PUFA) and BMD (total and spine) between 16 and 22 years of age. Concentrations of docosahexaenoic acid were positively associated with total BMD and BMD at the spine at 22 years of age. A positive correlation was also found between docosahexaenoic acid concentrations and the changes in BMD at the spine between 16 and 22 years of age.¹⁰⁶ To most effectively change the ratio of n-6 to n-3 fatty acids, there needs to be a reduction in consumption of foods rich in n-6 fatty acids and a higher consumption of foods that are rich sources of n-3 fatty acids.

Mechanistic studies investigating effects of soy isoflavones on bone tissue show favorable effects, but intervention studies in postmenopausal women, mostly those living in North America, have largely reported small or no effects of consuming high levels of soy isoflavones (either as soy protein or isoflavone supplements) to BMD. Reasons to explain this effect include the fact that lifelong exposure may modulate response to consumption of soy isoflavones, and that BMD rather than fragility fracture has been measured in these studies. Although BMD is commonly used as a predictor of fracture, it does not provide information regarding the structure of bone that is also an important predictor of fracture. Findings from the Shanghai Women's Study demonstrated that postmenopausal women with highest intakes of soy protein (18.5 g soy protein per day) have reduced risk of fracture compared to women in the lowest quintile (3.3 g soy protein per day).¹⁰⁷ Although women consuming the higher levels of soy protein also consumed more calories, fruits and vegetables, calcium and overall protein (both soy and nonsoy protein), the relationship between soy protein intake and bone health held after adjustment for these variables.¹⁰⁷

5.5.2 PHYSICAL ACTIVITY

For recommendations regarding physical activity and bone health, the recent Canadian Physical Activity Guidelines developed by the Canadian Society for Exercise Physiology provide evidence-based guidelines aimed at preventing chronic disease, in general, through age-appropriate levels of physical activity.¹⁰⁸ In general, four types of exercises have been identified as important for bone health: strength training, weight-bearing aerobic, flexibility, and stability/balance exercises. A recent meta-analysis that considered the effects of exercise training on whole bone strength found that exercise enhanced bone strength at loaded sites in children but not in adults.¹⁰⁹ However, the authors stated the need for well-controlled studies of sufficient duration to fully determine the potential role for exercise in mediating bone strength. A meta-analysis focused on exercise for prevention and treatment of

osteoporosis specifically in postmenopausal women reported a slight improvement in BMD as well as a slight reduction in the risk of fracture.¹¹⁰ Findings from both these meta-analyses identify a need for future investigations, particularly studies of high quality to more definitely establish the type and duration of exercise and agespecific exercise programs to benefit bone health.

In addition to the general findings that four types of exercise do provide risk reduction effects, the anti-inflammatory effects of the weight-bearing aerobic exercise deserves additional mention. Because this type of exercise providing anti-inflammatory effects that are separate from the benefits of the other forms of exercise, it is prudent that this form of exercise be incorporated into any activity program that is designed to produce optimal preventive effects. And the Canadian Physical Activity Guidelines appropriately address all types of exercise.

Specific guidelines for five different age groups are provided:¹⁰⁸ 0–4 years of age, 5–11 years of age, 12–17 years of age, 18–64 years of age, and 65 years of age and older. For infants aged less than 1 year, it is recommended that they be physically active several times a day. Examples of activities include the following: tummy time, reaching for or grasping balls or toys, playing or rolling on the floor, crawling around the home. The target is that infants and young children will progress to at least 60 minutes of energetic play per day by the age of 5 years and that these activities will help with development of motor skills. Within these guidelines are specific recommendations that "activities that strengthen muscle and bone" are performed at least 2 (over 18 years of age) or 3 days per week (5–17 years of age). Moreover, adults age 65 years or older with poor mobility are recommended to perform physical activities to enhance balance and prevent falls—for the prevention of fragility fracture. By alternating days with activities that strengthen muscle and bone with days that include moderately stressful aerobic exercise, optimal risk reduction for osteoporosis would be expected.

ACKNOWLEDGMENT

Wendy E. Ward holds a Canada Research Chair in Bone and Muscle Development, and acknowledges funding from the Natural Sciences and Engineering Research Council of Canada (NSERC) that has funded her research program investigating nutrients, bioactives, and bone health.

REFERENCES

- 1. NIH Consensus Development Panel on Osteoporosis Prevention D, Therapy. Osteoporosis prevention, diagnosis, and therapy. *Journal of the American Medical Association*. 2001;285:785–795.
- International Osteoporosis Foundation. http://www.Iofbonehealth.Org/facts-and-statistics .html, 2012. Accessed December 8, 2012.
- Heaney R, Abrams S, Dawson-Hughes B, Looker A, Marcus R, Matkovic V, Weaver C. Peak bone mass. *Osteoporosis International*. 2000;11:985–1009.
- 4. Pacifici R. Role of t cells in ovariectomy induced bone loss—Revisited. *Journal of Bone and Mineral Research*. 2012;27:231–239.
- 5. Manolagas SC, Parfitt AM. What old means to bone. *Trends in Endocrinology and Metabolism*. 2010;21:369–374.

- 6. Berg J, Tymoczko J, Stryer L. *Biochemistry*. Sixth Edition. W.H. Freeman and Company, New York, NY; 2007.
- 7. Hofbauer L, Kuhne C, Viereck V. The OPG/RANKL/RANK system in metabolic bone diseases. *Journal of Musculoskeletal and Neuronal Interactions*. 2004;4:268.
- Nakamura T, Imai Y, Matsumoto T, Sato S, Takeuchi K, Igarashi K, Harada Y, Azuma Y, Krust A, Yamamoto Y. Estrogen prevents bone loss via estrogen receptor α and induction of fas ligand in osteoclasts. *Cell*. 2007;130:811–823.
- 9. Komm B, Bodine P. Regulation of bone cell function by estrogens. *Osteoporosis*. 2001;1:305–337.
- Manolagas SC, Jilka RL. Bone marrow, cytokines, and bone remodeling. Emerging insights into the pathophysiology of osteoporosis. *The New England Journal of Medicine*. 1995;332:305.
- 11. Pacifici R. Estrogen, cytokines, and pathogenesis of postmenopausal osteoporosis. *Journal of Bone and Mineral Research*. 1996;11:1043–1051.
- 12. Jilka R. Cytokines, bone remodeling, and estrogen deficiency: A 1998 update. *Bone*. 1998;23:75.
- Suda T, Takahashi N, Udagawa N, Jimi E, Gillespie MT, Martin TJ. Modulation of osteoclast differentiation and function by the new members of the tumor necrosis factor receptor and ligand families. *Endocrine Reviews*. 1999;20:345.
- Chen JR, Plotkin LI, Aguirre JI, Han L, Jilka RL, Kousteni S, Bellido T, Manolagas SC. Transient versus sustained phosphorylation and nuclear accumulation of ERKS underlie anti-versus pro-apoptotic effects of estrogens. *Journal of Biological Chemistry*. 2005;280:4632–4638.
- Krum SA, Miranda-Carboni GA, Hauschka PV, Carroll JS, Lane TF, Freedman LP, Brown M. Estrogen protects bone by inducing Fas ligand in osteoblasts to regulate osteoclast survival. *The EMBO Journal*. 2008;27:535–545.
- Kuiper GG, Carlsson B, Grandien K, Enmark E, Häggblad J, Nilsson S, Gustafsson JÅ. Comparison of the ligand binding specificity and transcript tissue distribution of estrogen receptors α and β. *Endocrinology*. 1997;138:863–870.
- Kuiper GG, Lemmen JG, Carlsson B, Corton JC, Safe SH, van der Saag PT, van der Burg B, Gustafsson JÅ. Interaction of estrogenic chemicals and phytoestrogens with estrogen receptor β. *Endocrinology*. 1998;139:4252–4263.
- 18. McCarty MF. Isoflavones made simple-genistein's agonist activity for the beta-type estrogen receptor mediates their health benefits. *Medical Hypotheses*. 2006;66:1093.
- Chen X, Garner SC, Quarles LD, Anderson JJB. Effects of genistein on expression of bone markers during mc3t3-e1 osteoblastic cell differentiation. *The Journal of Nutritional Biochemistry*. 2003;14:342–349.
- Viereck V, Gründker C, Blaschke S, Siggelkow H, Emons G, Hofbauer LC. Phytoestrogen genistein stimulates the production of osteoprotegerin by human trabecular osteoblasts. *Journal of Cellular Biochemistry*. 2002;84:725–735.
- Karieb S, Fox SW. Phytoestrogens directly inhibit TNF-α-induced bone resorption in raw264. 7 cells by suppressing c-FOS-induced NFATC1 expression. *Journal of Cellular Biochemistry*. 2011;112:476–487.
- Mimura J, Yamashita K, Nakamura K, Morita M, Takagi TN, Nakao K, Ema M, Sogawa K, Yasuda M, Katsuki M. Loss of teratogenic response to 2, 3, 7, 8-tetrachlorodibenzo-p-dioxin (TCDD) in mice lacking the AH (dioxin) receptor. *Genes* to Cells. 2003;2:645–654.
- 23. Rüegg J, Swedenborg E, Wahlström D, Escande A, Balaguer P, Pettersson K, Pongratz I. The transcription factor aryl hydrocarbon receptor nuclear translocator functions as an estrogen receptor β-selective coactivator, and its recruitment to alternative pathways mediates antiestrogenic effects of dioxin. *Molecular Endocrinology*. 2008;22:304–316.

- Naruse M, Ishihara Y, Miyagawa-Tomita S, Koyama A, Hagiwara H. 3-Methylcholanthrene, which binds to the arylhydrocarbon receptor, inhibits proliferation and differentiation of osteoblasts in vitro and ossification in vivo. *Endocrinology*. 2002;143: 3575–3581.
- Zhang S, Qin C, Safe SH. Flavonoids as aryl hydrocarbon receptor agonists/antagonists: Effects of structure and cell context. *Environmental Health Perspectives*. 2003;111:1877.
- 26. Singh S, Casper R, Fritz P, Sukhu B, Ganss B, Girard B, Savouret J, Tenenbaum H. Inhibition of dioxin effects on bone formation in vitro by a newly described aryl hydrocarbon receptor antagonist, resveratrol. *Journal of Endocrinology*. 2000;167:183–195.
- 27. Harper PA, Riddick DS, Okey AB. Regulating the regulator: Factors that control levels and activity of the aryl hydrocarbon receptor. *Biochemical Pharmacology*. 2006; 72:267–279.
- Kawabata M, Imamura T, Miyazono K. Signal transduction by bone morphogenetic proteins. *Cytokine and Growth Factor Reviews*. 1998;9:49–61.
- 29. Muruganandan S, Roman A, Sinal C. Adipocyte differentiation of bone marrow-derived mesenchymal stem cells: Cross talk with the osteoblastogenic program. *Cellular and Molecular Life Sciences*. 2009;66:236–253.
- Miyazono K, Maeda S, Imamura T. Bmp receptor signaling: Transcriptional targets, regulation of signals, and signaling cross-talk. *Cytokine and Growth Factor Reviews*. 2005;16:251–263.
- de Jong DS, Vaes BLT, Dechering KJ, Feijen A, Hendriks J, Wehrens R, Mummery CL, van Zoelen EJJ, Olijve W, Steegenga WT. Identification of novel regulators associated with early-phase osteoblast differentiation. *Journal of Bone and Mineral Research*. 2004;19:947–958.
- Miyama K, Yamada G, Yamamoto TS, Takagi C, Miyado K, Sakai M, Ueno N, Shibuya H. A BMP-inducible gene, DLX5, regulates osteoblast differentiation and mesoderm induction. *Developmental Biology*. 1999;208:123–133.
- Miyazono K, Maeda S, Imamura T. Coordinate regulation of cell growth and differentiation by TGF-β superfamily and RUNX proteins. *Oncogene*. 2004;23:4232–4237.
- 34. Chen JR, Singhal R, Lazarenko OP, Liu X, Hogue WR, Badger TM, Ronis MJJ. Short term effects on bone quality associated with consumption of soy protein isolate and other dietary protein sources in rapidly growing female rats. *Experimental Biology and Medicine*. 2008;233:1348–1358.
- Chen JR, Lazarenko OP, Blackburn ML, Badeaux JV, Badger TM, Ronis MJJ. Infant formula promotes bone growth in neonatal piglets by enhancing osteoblastogenesis through bone morphogenic protein signaling. *The Journal of Nutrition*. 2009;139:1839–1847.
- Ma HP, Ming LG, Ge BF, Zhai YK, Song P, Xian CJ, Chen KM. Icariin is more potent than genistein in promoting osteoblast differentiation and mineralization in vitro. *Journal of Cellular Biochemistry*. 2011;112:916–923.
- Trzeciakiewicz A, Habauzit V, Mercier S, Barron D, Urpi-Sarda M, Manach C, Offord E, Horcajada MN. Molecular mechanism of hesperetin-7-O-glucuronide, the main circulating metabolite of hesperidin, involved in osteoblast differentiation. *Journal* of Agricultural and Food Chemistry. 2009;58:668–675.
- Lorenzo J, Choi Y, Horowitz M, Takayanagi H. Osteoimmunology: Interactions of the Immune and Skeletal Systems. Academic Press, Burlington, MA; 2010.
- 39. Komori T. Regulation of bone development and extracellular matrix protein genes by runx2. *Cell and Tissue Research*. 2010;339:189–195.
- 40. Chen JR, Lazarenko OP, Wu X, Kang J, Blackburn ML, Shankar K, Badger TM, Ronis MJJ. Dietary-induced serum phenolic acids promote bone growth via p38 MAPK/β-catenin canonical Wnt signaling. *Journal of Bone and Mineral Research*. 2010;25:2399–2411.

- Zhou H, Shang L, Li X, Zhang X, Gao G, Guo C, Chen B, Liu Q, Gong Y, Shao C. Resveratrol augments the canonical Wnt signaling pathway in promoting osteoblastic differentiation of multipotent mesenchymal cells. *Experimental Cell Research*. 2009;315:2953–2962.
- 42. Lecka-Czernik B. Ppars in bone: The role in bone cell differentiation and regulation of energy metabolism. *Current Osteoporosis Reports*. 2010;8:84–90.
- 43. Christy R, Yang V, Ntambi J, Geiman D, Landschulz W, Friedman A, Nakabeppu Y, Kelly T, Lane M. Differentiation-induced gene expression in 3t3-11 preadipocytes: CCAAT/enhancer binding protein interacts with and activates the promoters of two adipocyte-specific genes. *Genes and Development*. 1989;3:1323–1335.
- 44. Chen JR, Lazarenko OP, Wu X, Tong Y, Blackburn ML, Shankar K, Badger TM, Ronis MJJ. Obesity reduces bone density associated with activation of PPARγ and suppression of Wnt/β-catenin in rapidly growing male rats. *PLoS One*. 2010;5:e13704.
- 45. Bennett CN, Longo KA, Wright WS, Suva LJ, Lane TF, Hankenson KD, MacDougald OA. Regulation of osteoblastogenesis and bone mass by Wnt10b. *Proceedings of the National Academy of Sciences of the United States of America*. 2005;102:3324–3329.
- Steven W, Belury MA. Impact of conjugated linoleic acid on bone physiology: Proposed mechanism involving inhibition of adipogenesis. *Nutrition Reviews*. 2011;69:123–131.
- 47. Dittenhafer-Reed KE, Feldman JL, Denu JM. Catalysis and mechanistic insights into sirtuin activation. *Chembiochem.* 2011;12:281–289.
- Bäckesjö CM, Li Y, Lindgren U, Haldosén LA. Activation of SIRT1 decreases adipocyte formation during osteoblast differentiation of mesenchymal stem cells. *Cells Tissues Organs*. 2009;189:93–97.
- 49. Schett G, Saag KG, Bijlsma JWJ. From bone biology to clinical outcome: State of the art and future perspectives. *Annals of the Rheumatic Diseases*. 2010;69:1415–1419.
- 50. Lacativa PGS, Farias MLF. Osteoporosis and inflammation. Arquivos Brasileiros de Endocrinologia & Metabologia. 2010;54:123–132.
- 51. Mundy GR. Osteoporosis and inflammation. Nutrition Reviews. 2007;65:S147-S151.
- 52. Chen JR, Lazarenko OP, Shankar K, Blackburn ML, Badger TM, Ronis MJ. A role for ethanol-induced oxidative stress in controlling lineage commitment of mesenchymal stromal cells through inhibition of Wnt/β-catenin signaling. *Journal of Bone and Mineral Research*. 2010;25:1117–1127.
- Almeida M, Han L, Martin-Millan M, Plotkin LI, Stewart SA, Roberson PK, Kousteni S, O'Brien CA, Bellido T, Parfitt AM. Skeletal involution by age-associated oxidative stress and its acceleration by loss of sex steroids. *Journal of Biological Chemistry*. 2007;282:27285–27297.
- 54. Almeida M, Han L, Ambrogini E, Bartell SM, Manolagas SC. Oxidative stress stimulates apoptosis and activates NFκβ in osteoblastic cells via a PKCβ/p66shc signaling cascade: Counter regulation by estrogens or androgens. *Molecular Endocrinology*. 2010;24:2030–2037.
- Bu SY, Lerner M, Stoecker BJ, Boldrin E, Brackett DJ, Lucas EA, Smith BJ. Dried plum polyphenols inhibit osteoclastogenesis by downregulating NFATC1 and inflammatory mediators. *Calcified Tissue International*. 2008;82:475–488.
- 56. Bu SY, Hunt TS, Smith BJ. Dried plum polyphenols attenuate the detrimental effects of TNF-α on osteoblast function coincident with up-regulation of RUNX2, Osterix and IGF-1. *Journal of Nutritional Biochemistry*. 2009;20:35–44.
- 57. He X, Andersson G, Lindgren U, Li Y. Resveratrol prevents RANKI-induced osteoclast differentiation of murine osteoclast progenitor Raw 264.7 cells through inhibition of ROS production. *Biochemical and Biophysical Research Communications*. 2010;401:356–362.
- Blackwell KA, Raisz LG, Pilbeam CC. Prostaglandins in bone: Bad cop, good cop? Trends in Endocrinology and Metabolism. 2010;21:294–301.

- 59. Bilezikian JP, Raisz LG, Martin TJ. *Principles of Bone Biology*. Academic Press, Burlington, MA; 2008.
- 60. Li M, Thompson D, Paralkar V. Prostaglandin E 2 receptors in bone formation. *International Orthopaedics*. 2007;31:767–772.
- 61. Yoshida K, Oida H, Kobayashi T, Maruyama T, Tanaka M, Katayama T, Yamaguchi K, Segi E, Tsuboyama T, Matsushita M. Stimulation of bone formation and prevention of bone loss by prostaglandin E EP4 receptor activation. *Proceedings of the National Academy of Sciences*. 2002;99:4580–4585.
- 62. Ke HZ, Crawford DT, Qi H, Simmons HA, Owen TA, Paralkar VM, Li M, Lu B, Grasser WA, Cameron KO. A nonprostanoid EP4 receptor selective prostaglandin E2 agonist restores bone mass and strength in aged, ovariectomized rats. *Journal of Bone* and Mineral Research. 2006;21:565–575.
- 63. Tanaka M, Sakai A, Uchida S, Tanaka S, Nagashima M, Katayama T, Yamaguchi K, Nakamura T. Prostaglandin E2 receptor (EP4) selective agonist (ONO-4819. Cd) accelerates bone repair of femoral cortex after drill-hole injury associated with local upregulation of bone turnover in mature rats. *Bone*. 2004;34:940–948.
- 64. Zhang X, Schwarz EM, Young DA, Puzas JE, Rosier RN, O'Keefe RJ. Cyclooxygenase-2 regulates mesenchymal cell differentiation into the osteoblast lineage and is critically involved in bone repair. *Journal of Clinical Investigation*. 2002;109:1405–1416.
- 65. Tian X, Zhang Q, Zhao R, Setterberg R, Zeng Q, Iturria S, Ma Y, Jee W. Continuous PGE2 leads to net bone loss while intermittent PGE2 leads to net bone gain in lumbar vertebral bodies of adult female rats. *Bone*. 2008;42:914–920.
- Watkins BA, Li Y, Lippman HE, Feng S. Modulatory effect of omega-3 polyunsaturated fatty acids on osteoblast function and bone metabolism. *Prostaglandins, Leukotrienes* and Essential Fatty Acids. 2003;68:387–398.
- 67. Gabbay KH, Bohren KM, Morello R, Bertin T, Liu J, Vogel P. Ascorbate synthesis pathway dual role of ascorbate in bone homeostasis. *Journal of Biological Chemistry*. 2010;285:19510–19520.
- Xing W, Singgih A, Kapoor A, Alarcon CM, Baylink DJ, Mohan S. Nuclear factor-E2-related factor-1 mediates ascorbic acid induction of osterix expression via interaction with antioxidant-responsive element in bone cells. *Journal of Biological Chemistry*. 2007;282:22052–22061.
- 69. Bruick RK, McKnight SL. A conserved family of prolyl-4-hydroxylases that modify HIF. *Science Signalling*. 2001;294:1337.
- Wan C, Gilbert SR, Wang Y, Cao X, Shen X, Ramaswamy G, Jacobsen KA, Alaql ZS, Eberhardt AW, Gerstenfeld LC. Activation of the hypoxia-inducible factor-1α pathway accelerates bone regeneration. *Proceedings of the National Academy of Sciences*. 2008;105:686–691.
- Franceschi RT. The role of ascorbic acid in mesenchymal differentiation. *Nutrition Reviews*. 2009;50:65–70.
- 72. Haussler MR, Haussler CA, Bartik L, Whitfield GK, Hsieh JC, Slater S, Jurutka PW. Vitamin D receptor: Molecular signaling and actions of nutritional ligands in disease prevention. *Nutrition Reviews*. 2008;66:S98–S112.
- Blumberg JM, Tzameli I, Astapova I, Lam FS, Flier JS, Hollenberg AN. Complex role of the vitamin D receptor and its ligand in adipogenesis in 3t3-11 cells. *Journal of Biological Chemistry*. 2006;281:11205–11213.
- 74. Zhou Y, Liu Y, Tan J. Is 1, 25-dihydroxyvitamin D3 an ideal substitute for dexamethasone for inducing osteogenic differentiation of human adipose tissue-derived stromal cells in vitro? *Chinese Medical Journal-Beijing-English Edition*. 2006;119:1278.
- Boden SD, Liu Y, Hair GA, Helms JA, Hu D, Racine M, Nanes MS, Titus L. Lmp-1, a lim-domain protein, mediates BMP-6 effects on bone formation. *Endocrinology*. 1998;139:5125–5134.

- 76. Ohsawa M, Mizunuma H, Kagami I, Miyamoto S, Kanuma T, Ibuki Y. Effect of 1,25 (OH)₂D3 on expression of estrogen receptor-alpha mRNA on rat osteosarcoma cell line (ROS 17/2.8). *Life Sciences*. 2000;66:2465–2472.
- Kimmons J, Gillespie C, Seymour J, Serdula M, Blanck HM. Fruit and vegetable intake among adolescents and adults in the United States: Percentage meeting individualized recommendations. *Medscape Journal of Medicine*. 2009;11:26.
- 78. Fulgoni III VL, Keast DR, Bailey RL, Dwyer J. Foods, fortificants, and supplements: Where do Americans get their nutrients? *Journal of Nutrition*. 2011;141:1847–1854.
- 79. Musso CG. Magnesium metabolism in health and disease. *International Urology and Nephrology*. 2009;41:357–362.
- Sahota O, Mundey M, San P, Godber I, Hosking D. Vitamin D insufficiency and the blunted PTH response in established osteoporosis: The role of magnesium deficiency. *Osteoporosis International*. 2006;17:1013–1021.
- Freitag JJ, Martin KJ, Conrades MB, Bellorin-Font E, Teitelbaum S, Klahr S, Slatopolsky E. Evidence for skeletal resistance to parathyroid hormone in magnesium deficiency: Studies in isolated perfused bone. *Journal of Clinical Investigation*. 1979;64:1238.
- Kumar R. Metabolism of 1, 25-dihydroxyvitamin D3. *Physiological Reviews*. 1984;64: 478–504.
- 83. Rude RK, Singer FR, Gruber HE. Skeletal and hormonal effects of magnesium deficiency. *Journal of the American College of Nutrition*. 2009;28:131–141.
- Ozcivici E, Luu YK, Adler B, Qin YX, Rubin J, Judex S, Rubin CT. Mechanical signals as anabolic agents in bone. *Nature Reviews Rheumatology*. 2010;6:50–59.
- Scott A, Khan KM, Duronio V, Hart DA. Mechanotransduction in human bone: In vitro cellular physiology that underpins bone changes with exercise. *Sports Medicine*. 2008;38:139–160.
- Garman R, Rubin C, Judex S. Small oscillatory accelerations, independent of matrix deformations, increase osteoblast activity and enhance bone morphology. *PLoS One*. 2007;2:e653.
- Ozcivici E, Garman R, Judex S. High-frequency oscillatory motions enhance the simulated mechanical properties of non-weight bearing trabecular bone. *Journal of Biomechanics*. 2007;40:3404–3411.
- Kapur S, Baylink DJ, Lau K. Fluid flow shear stress stimulates human osteoblast proliferation and differentiation through multiple interacting and competing signal transduction pathways. *Bone*. 2003;32:241.
- Kapur S, Mohan S, Baylink DJ, Lau KHW. Fluid shear stress synergizes with insulin-like growth factor-I (IGF-I) on osteoblast proliferation through integrin-dependent activation of IGF-I mitogenic signaling pathway. *Journal of Biological Chemistry*. 2005;280: 20163–20170.
- 90. Lee KCL, Lanyon LE. Mechanical loading influences bone mass through estrogen receptor α. *Exercise and Sport Sciences Reviews*. 2004;32:64–68.
- Jessop H, Rawlinson S, Pitsillides A, Lanyon L. Mechanical strain and fluid movement both activate extracellular regulated kinase (ERK) in osteoblast-like cells but via different signaling pathways. *Bone*. 2002;31:186–194.
- Luu YK, Capilla E, Rosen CJ, Gilsanz V, Pessin JE, Judex S, Rubin CT. Mechanical stimulation of mesenchymal stem cell proliferation and differentiation promotes osteogenesis while preventing dietary-induced obesity. *Journal of Bone and Mineral Research*. 2008;24:50–61.
- 93. Armstrong VJ, Muzylak M, Sunters A, Zaman G, Saxon LK, Price JS, Lanyon LE. Wnt/β-catenin signaling is a component of osteoblastic bone cell early responses to load-bearing and requires estrogen receptor α. *Journal of Biological Chemistry*. 2007;282:20715–20727.

- Case N, Ma M, Sen B, Xie Z, Gross TS, Rubin J. β-catenin levels influence rapid mechanical responses in osteoblasts. *Journal of Biological Chemistry*. 2008;283: 29196–29205.
- Robling AG, Niziolek PJ, Baldridge LA, Condon KW, Allen MR, Alam I, Mantila SM, Gluhak-Heinrich J, Bellido TM, Harris SE. Mechanical stimulation of bone in vivo reduces osteocyte expression of SOST/Sclerostin. *Journal of Biological Chemistry*. 2008;283:5866–5875.
- Li X, Zhang Y, Kang H, Liu W, Liu P, Zhang J, Harris SE, Wu D. Sclerostin binds to LRP5/6 and antagonizes canonical Wnt signaling. *Journal of Biological Chemistry*. 2005;280:19883–19887.
- Robling AG, Hinant FM, Burr DB, Turner CH. Improved bone structure and strength after long-term mechanical loading is greatest if loading is separated into short bouts. *Journal of Bone and Mineral Research*. 2002;17:1545–1554.
- Chen NX, Ryder KD, Pavalko FM, Turner CH, Burr DB, Qiu J, Duncan RL. Ca²⁺ regulates fluid shear-induced cytoskeletal reorganization and gene expression in osteoblasts. *American Journal of Physiology-Cell Physiology*. 2000;278:C989–C997.
- Bacabac RG, Smit TH, Mullender MG, Van Loon JJWA, Klein-Nulend J. Initial stresskick is required for fluid shear stress-induced rate dependent activation of bone cells. *Annals of Biomedical Engineering*. 2005;33:104–110.
- 100. Pan J, Zhang T, Mi L, Zhang B, Wang B, Yang L, Deng L, Wang L. Stepwise increasing and decreasing fluid shear stresses differentially regulate the functions of osteoblasts. *Cellular and Molecular Bioengineering*. 2010;3:376–386.
- 101. Tartibian B, Maleki BH, Kanaley J, Sadeghi K. Long-term aerobic exercise and omega-3 supplementation modulate osteoporosis through inflammatory mechanisms in post-menopausal women: A randomized, repeated measures study. *Nutrition and Metabolism*. 2011;8:71.
- Silverman NE, Nicklas BJ, Ryan AS. Addition of aerobic exercise to a weight loss program increases BMD, with an associated reduction in inflammation in overweight postmenopausal women. *Calcified Tissue International*. 2009;84:257–265.
- 103. Hamidi M, Boucher B, Cheung A, Beyene J, Shah P. Fruit and vegetable intake and bone health in women aged 45 years and over: A systematic review. *Osteoporosis International*. 2011;22:1681–1693.
- 104. Hardcastle AC, Aucott L, Reid DM, Macdonald HM. Associations between dietary flavonoid intakes and bone health in a Scottish population. *Journal of Bone and Mineral Research*. 2011;26:941–947.
- 105. Weiss LA, Barrett-Connor E, von Mühlen D. Ratio of n- 6 to n- 3 fatty acids and bone mineral density in older adults: The Rancho Bernardo study. *The American Journal of Clinical Nutrition*. 2005;81:934–938.
- 106. Högström M, Nordström P, Nordström A. N–3 fatty acids are positively associated with peak bone mineral density and bone accrual in healthy men: The NO₂ study. *The American Journal of Clinical Nutrition*. 2007;85:803–807.
- 107. Zhang X, Shu XO, Li H, Yang G, Li Q, Gao YT, Zheng W. Prospective cohort study of soy food consumption and risk of bone fracture among postmenopausal women. *Archives of Internal Medicine*. 2005;165:1890.
- 108. Tremblay MS, Warburton DE, Janssen I, Paterson DH, Latimer AE, Rhodes RE, Kho ME, Hicks A, LeBlanc AG, Zehr L. New Canadian physical activity guidelines. *Applied Physiology, Nutrition, and Metabolism.* 2011;36:36–46.
- 109. Nikander R, Sievänen H, Heinonen A, Daly RM, Uusi-Rasi K, Kannus P. Targeted exercise against osteoporosis: A systematic review and meta-analysis for optimising bone strength throughout life. *BMC Medicine*. 2010;8:47.
- 110. Howe T, Shea B, Dawson L, Downie F, Murray A, Ross C, Harbour R, Caldwell L, Creed G. Exercise for preventing and treating osteoporosis in postmenopausal women (review). *Cochrane Database of Systematic Reviews*. 2011;7:CD000333.

6 Cancer

Jonathan Cannizzo and C. Murray Ardies

CONTENTS

6.1	Introdu	action	234			
	6.1.1	Diet, Exercise, and Cancer: General Considerations	239			
6.2	Etiolog	gy of Cancer	241			
	6.2.1	DNA and Chromosomal Mutations	242			
	6.2.2	Cell Cycle	245			
	6.2.3	DNA Damage, Checkpoint Control, and Apoptosis	248			
	6.2.4	Cancer Stem Cells and Tumor Environment	251			
		6.2.4.1 Contribution of Obesity and Insulin Resistance to the				
		Tumor Environment	253			
	6.2.5	DNA Damage	254			
		6.2.5.1 Reactive Oxygen and Nitrogen Species	254			
		6.2.5.2 Xenibiotics	257			
	6.2.6	Nutritional Insufficiencies	264			
6.3	Etiolog	gy of Prevention	268			
	6.3.1	Nutrition and Diet	268			
	6.3.2	Inflammatory Signaling	270			
	6.3.3	Antioxidant and Redox Control	275			
		6.3.3.1 Cell Cycle	279			
		6.3.3.2 DNA Damage and Repair	280			
		6.3.3.3 Phase I and Phase II Enzymes	281			
	6.3.4	Calorie Restriction as Prevention	283			
	6.3.5	Exercise	284			
		6.3.5.1 Exercise and Insulin Sensitivity	285			
		6.3.5.2 Exercise, Redox Control, and Phase II Enzymes	286			
		6.3.5.3 DNA Repair	288			
		6.3.5.4 IL-6-Mediated Anti-Inflammatory Effects of Exercise .	289			
6.4	Summ	ary and Recommendations	290			
	6.4.1	Cancer Etiology	290			
	6.4.2	Diet-Based Prevention	293			
	6.4.3	Dietary Recommendations	296			
	6.4.4	Exercise	298			
	6.4.5	Exercise Recommendations	299			
Ackı	Acknowledgment					
References						

6.1 INTRODUCTION

Cancer is a disease that directly or indirectly affects everyone and is the second leading cause of mortality worldwide with over 7.5 million people dying from cancer each year.¹⁻³ Of that 7.5 million, the American Cancer Society estimates that 600,000 deaths occur from cancer in the United States. In 2010, 1.5 million people in the United States were diagnosed with cancer, and when stratified by age, cancer is now the leading cause of death for people under the age of 85 with lung cancer being by far the biggest killer of both men and women.^{4–8} Currently, approximately one out of every four deaths in the United States is due to cancer.^{1,7} Cancer also carries significant economic costs that continue to grow each year with 2010 costs of ~\$124 billion for cancer care being projected to increase to anywhere from ~\$157 billion to ~\$173 billion by 2020.⁹ And these figures are only for care and treatment; they do not include other indirect costs such income and productivity losses nor do they account for the substantial emotional impact of cancer on both the patients and their families.

Unlike other chronic diseases such as hypertension, diabetes, and even heart disease, cancer is a particularly difficult disease because of the severe psychological implications. The first thing many people think of when they are diagnosed with cancer is "death sentence." And when informed, most cancer patients develop high rates of anxiety and depression, with 20%–30% at a greater risk for psychiatric morbidity and ~85% of patients exhibiting well above normal levels of stress.^{10,11} This severe distress then affects activities of daily living such as appetite, sleep, and work; all of which are negatively affected by the disease.^{12–14} Although psychological distress is certainly a component of many diseases, cancer is unique in the severity and universality of these effects; leading one to believe that prevention of cancer will provide important benefits well beyond the straightforward issues of biological health and health-care expenses.

Cancer of course is not a single disease but rather a collection of different, although very much related diseases (lung cancer, breast cancer [BC], ovarian cancer, prostate cancer [PC], etc.). The various forms of cancer, however, will not be discussed individually; there are excellent texts that do that such as the latest edition of *DeVita, Hellman, and Rosenberg's Cancer: Principles & Practice of Oncology* by DeVita, Lawrence, Rosenberg, DePinho, and Weinberg.¹⁵ The different forms of cancer all share a common basis, however; one in which a variety of mutations, both inherited and acquired, lead to deregulation of cell division and ultimately result in a cancer. Because of this common basis, the general molecular and cellular mechanisms of cancer as described in *Principles and Practice* and other texts and reviews can be applied to essentially all cancers.^{15–21}

In its simplest terms, developing a cancer is generally considered to occur in three stages: initiation, promotion, and progression. The initiation process of acquiring the "first" mutation most often involves reactive oxygen species (ROS) or reactive chemical species that cause some form of DNA damage in stem cells (SCs) (and sometimes in tissue progenitor cells [PCs]). (Mutations due to infidelity of DNA replication during cell division are not a topic for this chapter.) If the damage is

not repaired, then a mutation at the genetic location of the original damage will be produced in the daughter cells following cell division. This first mutation may alter a cellular function associated with controlling cell division that then accelerates the cellular mechanisms that lead to the acquisition of more mutations that then lead to the transformation of precancerous cells into cancer cells (carcinogenesis) over multiple cycles of cell division. While there is very little one can do to change an inherited initiating mutation, the various cellular mechanisms associated with acquiring new mutations, and those of subsequent promotion and progression events that follow, are in many cases amenable to prevention efforts.

From the preceding discussion, while the importance of initiating DNA damage cannot be denied, there still must be additional mutations acquired to result in a cancer. The reactive molecules that lead to the acquisition of additional DNA damage can be of either endogenous or exogenous origin. One of the most notorious exogenous sources of carcinogenic agents is tobacco smoke and it has been implicated as a causative agent for many different cancers including those of the lung, larynx, pharynx, upper digestive tract, and oral cavity (among others).^{22,23} Of interest to cancer etiology and prevention, however, is that exposure to carcinogens does not necessarily result in cancer. Again, tobacco provides an interesting example in that based on lung cancer risks for smokers, about 15%–25% of lifetime smokers will be diagnosed with lung cancer; with the heaviest (>3 packs/day in some cases) and longest duration smokers incurring an incidence of lung cancer as high as 30%-40%.²⁴⁻²⁷ While the associations are obvious that the heaviest exposures should produce the greatest risk, the "flip side" is often ignored, that is, what is it about the 75%–85% of "average" smokers who do not get lung cancer? As discussed throughout Section 6.2, there are many different mechanisms of cell function involved other than just the simple exposure to mutagens and mitogens. Some of these mechanisms include metabolic activation of oxygen and of precarcinogens into the reactive chemical species (carcinogens) that cause DNA damage, metabolic inactivation of the damaging reactive molecules, and rates/efficiency of DNA repair among others. In addition, population studies of smokers reveal that reductions in risk for lung cancer of $\sim 40\%$ through repeated exercise activity may occur.^{28,29} Differences in diet among smokers, particularly in the consumption of higher amounts of vegetables and fruit, also account for differences in risks with higher consumption being associated with lower risks.^{30,31} And as discussed in Section 6.3, it is through their effects on mechanisms of activation, inactivation, and DNA repair (among others) that diet and exercise can exert their preventive effects.

From an individual's health perspective, individual differences in the inherent mechanisms that confer risk are important, especially if one inherits a cancer-risk genotype. That these mechanisms might be modified through lifestyle-based preventive activities is even more so, and the personal and public health benefits can be profound. Interestingly, the benefits of prevention might even be applied to a serious public health problem such as smoking-induced cancer simply because a large majority of smokers seem to be unable to quit: 55%–95% fail going "unassisted cold turkey" and 75%–85% fail with pharmacological support and/or financial incentives.^{32–37} The current norm in public health appears to be smoking cessation for

smokers and smoking prevention combined with diet, physical activity, and weight control for nonsmokers.³⁸ Perhaps a more comprehensive focus on cancer prevention that includes preventive diet and exercise behaviors for all, not just non-smokers, might prevent many more smoking-related cancers.

Understanding the molecular factors that contribute to cancer etiology and the mechanisms through which different individuals and populations are more resistant to the cancer processes is fundamental to developing effective prevention procedures that target mechanisms of cause. One of the most obvious population differences is that between women and men. Women are diagnosed far more often than men for BC even though both genders have breast tissue.^{6–8,39} It is the most common malignancy in women and the second most common cause of cancer-related death in women after lung cancer.^{4–8,39} Hormones clearly play an important role with estrogen having a dominant role in the cause of this cancer.⁴⁰⁻⁴² Estrogen can be metabolized into a reactive molecule capable of causing DNA damage; therefore, it can act as a mutagen (or an initiator). Estrogen also activates cell division in those cells with estrogen receptors, rendering estrogen a mitogen (or a promoter). Estrogen can therefore be considered to be a complete carcinogen because it is both a mutagen and a mitogen. And as a result of the effects of estrogen, women are expected to incur about 229,000 new cases in 2012 while the expected new cases in men is $\sim 2,100.7$ Of course, it is not just an "estrogen thing." There are genetic risks for BC as well with mutations in the BRCA1 and BRCA2 genes being the most well known. The overall lifetime expectation for BC in women in the general population is about 12% in the United States (one in eight) with the expectation rising to as high as 75% with mutations in the BRCA1 or BRCA2 genes; genes that code for proteins involved in the repair of double-strand (DNA) breaks.^{6–8,43–45} Interestingly, the risk for BC in carriers of these mutations can be significantly reduced through maintaining a healthy body weight and through physical activity, and in addition, through increased vegetable/ fruit consumption and physical activity in noncarriers.^{43–51} The importance of this is that even though genetic variables that confer high risk exist, they can apparently be attenuated through lifestyle modification; indicating that prevention efforts can successfully reduce risks even in the face of "bad genes."

On the male side of things, PC is the most commonly diagnosed malignancy in men.⁵² It also is the second leading cause of cancer-related death in North America in men after lung cancer.^{4,5,39,52,53} As with almost all cancers, the probability of getting invasive PC increases dramatically with age; ~12.5% of men >70 years of age were diagnosed in 2006–2008 compared to ~6.8% of men aged 60–69 and ~2.5% for men aged 40–59 with a lifetime risk of ~16.5% (one in six).⁷ Associations between high testosterone and risk for PC also are observed although the associations are not unequivocal.^{54–57} Unlike estrogen, testosterone is not known to be a mutagen, which may account for the lower consistency in the literature of a carcinogenic effect of testosterone. Testosterone and its more active metabolite dihydrotestosterone are, however, involved in stimulating proliferation and differentiation of prostate tissues (as well as apoptosis of damaged cells) to maintain normal tissue architecture, and the promoting events of PC are related to acquired mutations in the androgen receptor and in the proliferation-associated genes.^{42,58–60} As genetic lesions occur within the prostate cells, they acquire a much more proliferative response to the androgens

and lose their differentiation responses, accounting for the increase in the promoting effects of higher androgen levels observed in some studies. In its most aggressive form, however, PCs are androgen independent. Interestingly, similar to BC, there is evidence that exercise can reduce risk for PC.^{61–63} The exercise effects, however, appear to be more equivocal, with inconsistent positive results and apparently less of an absolute effect. Similar to those of BC, the physical activity associated with preventive effects in the epidemiological studies is of low intensity walking, yard work, housework, and other nonstrenuous exercise activities. This leads to the possibility that the benefits might not be due to the activity directly altering mechanisms of (cancer) cause but rather through ensuring insulin sensitivity due to simply not being completely sedentary; insulin resistance is known to be a risk factor for PC.^{64–66}

Of these three cancers mentioned, breast (one in eight) and prostate (one in six) are by far the most common malignancies in women and men, respectively. And, while lung (including bronchus) cancer is the second-most diagnosed cancer in both genders (1 in 13 for men and 1 in 16 for women), it is by far the biggest killer in both.^{4-8,39} Of particular interest is that mechanisms of inflammation have been associated with the development of each of these cancers as well as essentially, all others.^{67–78} Because cell proliferation is an obligatory component of carcinogenesis, and a variety of growth factors that stimulate cell proliferation are produced during an inflammatory response (as reviewed in detail in Chapter 2), inflammatory mechanisms are relevant to any cancer. Growth factor receptors activate various components of the different mitogen-activated protein kinase (MAPK) pathways (among others) to produce their effects. These same pathways also are cross-activated by estrogen and testosterone as part of their proliferative activities. In fact, the name of the MAPK pathways underlies the fact that these signal transduction pathways were first associated with cell proliferation, growth, and differentiation and were later recognized as being central to a variety of cellular stress responses, including inflammation. With these stress-associated pathways being an integral part of the carcinogenesis (promotion) process, anti-inflammatory-based prevention procedures should therefore have a profound effect on overall cancer risk reduction.

Although gender and behavior issues create obvious differences in risk for specific cancers, in the United States, it is a particularly difficult issue. In the United States, African American women have higher mortality rates for breast, colorectal, and endometrial cancers in comparison to Caucasian women.³⁹ Even though Caucasian women have higher adjusted incidences of BC, minorities have higher mortality rates with this disease.⁷⁹⁻⁸⁴ The increase in mortality rates of these diseases in women of color is most often because a diagnosis is typically made at a much later stage of cancer development. In most cases, this is due to the African American population in the United States not having similar access to health insurance and not receiving proper screenings as is much more common among Caucasian women.^{85,86} Others have reported that there is no relationship between income/education level and cancer diagnosis; yet, there remains a very strong difference between races in early-stage diagnosis.⁸³ Even in whites who often have less intention to get screened, greater access to screenings than their African American counterparts ensures greater levels of use of early screening regardless of gender.⁸⁶ This lack of access is mostly due to lower levels of health insurance among minorities coupled with access difficulties
such as longer waits at medical facilities in underprivileged areas that are frequented more often by minorities and a lack of child care to allow time to get to screening facilities.^{87,88} In addition to screening, there is some evidence that African-Americans also receive less counseling from their physicians on risk factors that play a role in cancer, factors such as diet, exercise, tobacco, and weight-gain/obesity.⁸⁹ This last, of course, brings us to the major issue for this chapter. As discussed very briefly in the preceding discussion, there are differences in behavior between individuals that appear to create large differences in risk, and participation in those risk-reduction behaviors, regardless of gender and genetics, plays a role in reducing cancer risk. And from both counseling and behavioral risk perspectives, without any knowledge of what those risky behaviors are, it is highly doubtful that meaningful changes in behavior can even be proposed or followed.

The information about preventive behaviors that is taught, whether in public schools, public service announcements, or in the physicians' offices, tends to come from professional organizations that specialize in cancer research (and treatment). There are many highly reputable scientific organizations that promote prevention, and a review of their recommendations published on their web sites reveals that there is considerable overlap in what they recommend. For example, the American Cancer Society and the American Institute for Cancer Research suggest that we should be as lean as possible without being underweight (BMI at the lower end); be physically active for 30 minutes 5 days/week; avoid sugary drinks; eat 2.5 cups of a variety of vegetables and fruits each day, choose whole grain instead of refined grain foods; eat less red meat; limit alcohol consumption to a maximum of two drinks/day for men and one drink/day for women; limit salt intake; do not use tobacco; and do not use supplements to prevent cancer.90,91 Other organizations such as the Mayo Clinic, the Centers for Disease Control and Prevention, and the National Cancer Institute give very similar prevention tips for the layperson but also include recommendations for timely screening for precancerous lesions and to protect yourself from the sun.^{91–94}

The professional publications of these groups, the latest from the American Cancer Society is used as an example here, typically follow the 2008 Physical Activity Guidelines for Americans. They also include more information than the web sites do as well as provide many examples of recommended activities ranging from walking, yoga, and mowing the lawn to running, cross-country skiing, and heavy manual labor.95-97 The inclusion of the low-intensity activities such as walking or yoga for 3×10 minutes is a consequence of a change in exercise guidelines for public health from those based on training for improved fitness (often measured by changes in VO₂ max) to much more moderate levels of activity based on epidemiology-based research that has associated lower intensity physical activities with improved health in comparison to completely sedentary individuals.96,97 Unfortunately, from the 1990s through to the latest 2005-2008 NHANES analyses, there has been a steady increase in the number of Americans who are inactive with 33% currently reporting no activity at all; of the remaining 76% who claim to be active ~50% do not meet the minimum guidelines for aerobic or strength activities as suggested by the Physical Activity Guidelines.⁹⁸⁻¹⁰³ Obviously claiming to be physically active and being sufficiently active for prevention are two very different things. Thus, approximately two-thirds of the country does not participate in sufficient physical activity to have a preventive effect. And interestingly enough, the increase in inactivity and in noncompliance with minimum aerobic and strength recommendations has occurred in a similar fashion across all ethnic and socioeconomic populations. Thus, the public health messages for cancer prevention through diet and physical activity simply are not getting to everyone (or to anyone?), regardless of who is delivering the messages or whether there is easy access to health care or not. Perhaps this is due in part to a general lack of understanding of just how important more intense physical activities and proper diet are in preventing cancer.

Because the published guidelines for prevention of cancer include many components in addition to physical activity and diet, it is very difficult to determine the role of each specific guideline in prevention efforts that are based on the guidelines in total. And interestingly enough, there are few studies available that evaluate the impact of following such prevention guidelines. However, a relatively recent (2011) follow-up of the Cancer Prevention Study-II Nutrition Cohort revealed that after 14 years of following the American Cancer Society general guidelines for prevention there were significant reductions in relative risk for all cancers for males and females to ~0.7 and 0.76.¹⁰⁴ In spite of a general lack of specifics, such guidelines do in fact appear to be appropriate for prevention in a general population, which then begs the question why another cancer chapter devoted to prevention through diet and exercise?

The simple answer is that since the publication of the many epidemiological studies on which the guidelines were based, there have been literally thousands of studies detailing many of the biochemical and molecular events that are a part of the etiology of cancer. In addition, many of these have detailed some of the mechanisms through which specific dietary components or exercise stress modifies mechanisms involved in the cancer process. On the basis of the mechanistic-based research, it is clear that there are different components of risk that are modified by different dietary adjustments and by different types and intensities of activity. And, based on these, more specific recommendations for diet and physical activities that target those mechanisms can be proposed.

6.1.1 DIET, EXERCISE, AND CANCER: GENERAL CONSIDERATIONS

Since the landmark publication in 1981 of Sirs Richard Doll and Richard Peto¹⁰⁵ documenting that diet may account for ~30% of the avoidable risks for cancer, there have been many thousands of publications in the area of diet and cancer. To continue with one of the first examples, population studies indicate that there are differences in the incidence of PC between countries that have large differences in cultural-based dietary practices. For example, males in China have an incidence of PC of 0.5 per 100,000 men compared to that of the United States of 110 per 100,000.⁵² There also is a significant increase in PC risk following migration of Asians (and other low-risk populations) to the United States.^{106–111} In addition to the migration studies, the incidence of PC increases in Asian countries following a "westernization" of their diet and lifestyle.^{112,113} Overall, diets that are high in dairy, red meat, and dietary fat and low in both variety and quantity of vegetables (a "westernized" diet) have been found to be positively correlated with risk while diets high in soy, fiber, cruciferous vegetables

(an "Asian" diet), as well as lycopene have been found to be inversely related to risk for PC.^{114–126} Interestingly, similar changes in risk following the adoption of a westernized lifestyle or by emigrating to a westernized culture also are observed for BC where it traditionally has been a cancer of much lower incidence in Asian countries compared to the United States.^{107,127–131} From many population studies such as these, there are consistent indications that reducing the variety and amount of vegetables in diet while increasing the amount of meat and fat contribute greatly to enhancing risk for these cancers that have a very strong endocrine component. And even in cancers that do not have a strong endocrine component, such as colorectal cancers, similar alterations in risk due to altered meat and vegetable consumption are observed in different populations and following migration to westernized cultures.^{132–135}

As reported for all cancers, in many studies, an inadequate intake of various nutrients is part of the increased risks that are associated with a poor diet, while in many others, the protective effects of phenolic components have been observed.^{4,29,31,105,136–150} In addition to nutritional adequacy and phytochemical content of diet, mounting evidence shows that obesity is linked to cancer risk as well.^{151–154} And as is well known, the prevalence of obesity in men increased from 12.2% in 1971–1974 to 33.3% in 2005–2008 while in women the prevalence increased from 16.8% to 36.2% in the same time frame.^{98–103}

While physical activity is not an absolutely essential requirement for weight maintenance, its role as a calorie burner to make it a little easier cannot be ignored. This does need to be put into perspective, however. If walking and yoga are chosen as the desired physical activities (they are recommended activities in the 2007 Physical Activity and Public Health Recommendations from the ACSM/AHA¹⁵⁵) and the minimum recommended 30 min/day of activity for 5 days/week is performed, the number of additional calories expended would be less than 150 kcal each for the 5 days. Interestingly, the self-reported consumption of energy has increased by 10% for men and by 22% in women between 1970 and 2005, easily accounting for the increase in obesity in the same time frame.98-103 These increases in consumption amount to 1701 and 2408 kcal/week for men and women, respectively. This is far more than the <750 kcal/week that would be expended following the minimum activity recommendation. Obviously, even if everyone followed the minimum recommended activity, it would still not make up for the population-wide increase in calorie intake to prevent the increase in prevalence of overweight and obesity or to prevent an increase in risk for cancer by weight gain. This does not mean of course that longer duration and higher intensity activities will not burn sufficient calories to assist with weight maintenance; they certainly will. And this is an important consideration in light of the important role that weight gain and obesity can play in promoting inflammatory signaling and risk for cancer.^{151,152,156–162} The minimum activity recommendations may simply be insufficient to have any major effect on attenuating weight gain or obesity for most people.

The idea that exercise might play a direct role in prevention of cancer has been documented since the early 1920s wherein increased physical exertion of various jobs were associated with reduced cancer prevalence.^{163,164} And the working hypothesis proposed by Siversten¹⁶⁴ (p. 376) was eerily prognostic: "That human carcinoma may be the reaction to and the result of chronic irritation of adult epithelial tissue

bathed in body fluids altered by certain metabolic products as a result of deficient muscular activity." What was not known at the time is that the weight gain associated with inactivity leads to a proinflammatory environment in abdominal adipose tissue that then leads to systemic increases in proinflammatory signaling molecules as well as insulin resistance.^{165–168} Elevated insulin levels are now known to be associated with risk for a variety of cancers including those of the breast, prostate, pancreatic, and colon.^{64,169-171} Increased proinflammatory signaling also is strongly associated with risk for almost all cancers and the elevated circulating proinflammatory cytokines coupled with increased insulin due to inactivity are almost certain to be among the major "metabolic products" that are responsible for the "irritation of the adult epithelial tissue" as hypothesized by Siversten. Since then, physical activity has been observed to be inversely associated with many different cancers in a variety of epidemiological studies and several mechanisms have been proposed on the basis of experiments in animal models.^{28,29,46,47,51,63,172–185} The proposed mechanisms have ranged from reducing circulating levels of proliferative hormones through increased metabolism to reducing weight gain and increasing insulin sensitivity, enhancing elimination of carcinogens through increased phase II enzyme activity, and increased DNA repair.

While animal models and cell-based models provide the basis of understanding mechanisms of cancer etiology and how they might be modified, the current recommendations for risk reduction appear to be based predominantly on epidemiological associations between dietary behaviors or activity behaviors and (existing) disease. This is based in part on the relatively general nature of the existing recommendations and the fact that they have been in place for many years before the current availability of a plethora of highly specific information on the cellular basis of cancer etiology. This, however, is not a fault; it is an unavoidable limitation to the development of almost all public health policy. The need for public health policies in specific areas of health are most often recognized and recommendations formulated long before all the specific details of the etiology of risks are known. Once known, more specific recommendations may then be made.

To develop targeted prevention measures, the cellular mechanisms of cause must first be known. Then the dietary components and forms of exercise that are known to modify them can be incorporated into a series of recommendations. The ultimate recommendations must, however, have a sufficiently broad appeal and be relatively easy to implement or they will never be adopted by anyone except those most dedicated to personal health. Providing target goals for optimal prevention and a rational progression of activities designed to ultimately meet those optimal goals might be prudent. And, as with the dietary components, only those specific kinds of activity that modify causal mechanisms of cancer should become the activities recommended for optimal prevention of cancer.

6.2 ETIOLOGY OF CANCER

As mentioned in Section 6.1, cancer is really a whole family of diseases that are all related through a common etiology of acquiring a sufficient number of mutations in genes that are associated with regulating cell division to cause deregulated cell

division and metastases. Although cancer is generally considered to occur in three stages—initiation, promotion, and progression—it is actually the same basic processes that lead to various mutations that occur throughout each of these stages that cause the cancer.^{15–21}

6.2.1 DNA AND CHROMOSOMAL MUTATIONS

The process of acquiring mutations often involves some sort of ROS or reactive chemical species. These reactive molecules can form adducts with DNA bases, a form of DNA damage. If the damage is not repaired, then the damaged DNA bases will be misread and mutations within the gene sequence at the location of the original damage will be produced when all the DNA is duplicated during the S phase of the cell cycle and appear in the daughter cells. Point mutations that usually arise from this process produce changes in the DNA sequence by the insertion of the wrong DNA bases. If the point mutation(s) occur(s) in the exon of a gene, then the expressed protein will likely have a single changed amino acid sequence in its primary structure, possibly affecting the ultimate three-dimensional structure and/or the physical properties of the protein.

In addition to adducts, other forms of DNA damage can occur such as DNA:DNA and DNA:protein cross-links and single- and double-strand breaks (DSBs) in the DNA. DSBs lead to extensive DNA rearrangements during the mitosis phase of the cell cycle resulting in a variety of possible chromosomal mutations in daughter cells; single-strand breaks can lead to DSBs on replication during the S phase, producing the same end result. Chromosomal mutations occur when the fragments of chromosomes fail to bind to the spindles during prometaphase and are subsequently left behind; being lost forever, along with whatever exons resided on that fragment. With multiple unrepaired DSBs on different chromosomes, components of one chromosome might rearrange with components of another to produce a translocation. Other chromosomal mutations also may occur with insertions, duplicate insertions, inversions, as well as whole chromosome duplications and deletions (aneuploidy) being seen in many cancerous cells, and they are especially common in malignancies. Where the sites of chromosomal derangements cross through exons of genes, then the protein product usually will be either nonfunctional or completely lost.

Mutations in noncoding regions of DNA also can have deleterious effects on control of cell division if they occur in the promoter region of a gene, leading to aberrant expression of the regulatory protein. Because mutations in the regulatory noncoding regions of DNA and their implications for carcinogenesis have not been as well characterized as those for specific genes, they will not be discussed here. However, with the recent publication of the various ENCODE projects, this may soon change.^{186,187} The study of epigenetics is an area that is somewhat related to this. Rather than dealing with mutations, however, it is hypermethylation of CpG islands (regions of DNA that are rich in the CpG dinucleotide) that are associated with promoter regions of genes that may play a role in cancer. Hypermethylation of these CpG islands often functions to silence expression of genes, and if the gene happens to be a tumor suppressing gene, high risk for cancer is the result.^{188,189} While the regulation of the methylation status of various cancer-associated genes is under intense investigation, it is not a topic for this chapter because there is insufficient information to determine how it might be incorporated into a prevention paradigm. There is, however, some evidence that ROS-mediated damage to DNA may initiate processes that can lead to hypermethylation of CpG dinucleotides, indicating that prevention strategies to minimize ROS-mediated damage might play a role in diminishing, at least in part, risks associated with epigenetic changes.^{190,191} ROS-associated damage and prevention will be discussed in detail in Section 6.2.5.1 and Section 6.3.

Any of these possible mutagenic or procarcinogenic changes to DNA structure may lead to functional properties of the expressed protein that are different from the native protein, that is, enhanced function (a constitutively active gene product or overexpression of normal function proteins resulting from chromosomal duplication events that produce multiple gene copies), attenuated function (partially inactivating mutations), or complete loss of function (inactivating mutations, chromosomal deletions, and gene silencing). And depending on the specific genes that are mutated or lost, the loss (or gain) in function may have little to no effect on overall cell function or profound effects; it just depends on which proteins have been affected and to what degree. Mutations also can be acquired during normal progression through the cell cycle simply due to the DNA replication errors that occur naturally. And if the initiating gene mutation(s) are inherited, then the initiation phase of the process has essentially been completed at the time of fertilization.

Where the mutations are in genes that are involved in controlling cell division, then rates of cell division might increase. The promotion phase typically involves the activation of more rapid rates of cellular division. Rates of cell division can be enhanced, for example, if the expressed mutant protein is a component of a signaling pathway that is involved in stimulating cell division. For example, a mutated K-Ras gene that produces a K-Ras protein with constitutive activity is commonly observed in colorectal cancer and less commonly in cancers of the lung and pancreas and in leukemia.¹⁹²⁻¹⁹⁶ The K-Ras protein is a GTPase cell membrane protein that is activated by binding to GTP and inactivated when the GTP is hydrolyzed to GDP. A mutated K-Ras is constitutively active in these cancers and it activates the Ras-Raf-MEK1/2-extracellular signal-related kinase 1/2 (ERK1/2) (ERK-MAPK) pathway as well as the Rac1-MEK4/7-JNK1/2/3 (JNK-MAPK) pathway through activating phosphoinositide 3-kinase (PI3K), which in turn activates Rac1. These pathways are usually activated through binding of growth factors to their membrane receptors to ultimately activate a variety of transcription factors that then initiate the synthesis of many different proteins that are necessary for normal cell function, cell growth, and cell division. Some of the transcription factors activated by these pathways include $NF\kappa\beta$, c-Myc, Elk1, HSF-1, STAT, c-Jun, and c-Fos. These pathways also can be activated by a variety of inflammatory cytokines as well as protein kinase A and protein kinase C (PKC), indicating the importance of these pathways in cellular responses to a variety of stressors. Mutations in the BRAF gene also are commonly seen in a variety of cancers including colorectal, non-small cell lung cancer (NSCLC), melanomas, and adenocarcinomas of the lung.¹⁹⁷⁻²⁰⁰ Because the B-Raf protein is immediately downstream of K-Ras and does not activate PI3K, constitutive activation of B-Raf will tend to activate the ERK-MAPK pathway only. Mutation of the growth receptors themselves also can result in a constitutively active signal transduction pathway; for

example, mutations in epidermal growth factor (EGF) receptor have been observed in a subtype of NSCLC seen in nonsmokers as well as in colorectal cancer.²⁰¹⁻²⁰³ And as mentioned in Section 6.1, mutations in the androgen receptor in PCs are associated with constitutive or enhanced proliferative signaling. Mutations such as these will lead to greater proliferative signaling and result in an increase in the rate of cell division. With greater than normal numbers of cell divisions, there is then a greater likelihood of mutations occurring due to normal errors of replication. Cells with K-Ras or B-Raf mutations obviously have an increased risk for producing cancer cells because of this, but they are not yet cancer cells; only with the acquisition of additional mutations that alter other cell cycle functions can precancerous cells transform into cancer cells. And this accumulation of more mutations in subsequent generations of daughter cells defines the progression phase. Thus, promotion and progression are intimately integrated: without promotion, there will be no progression. In essence, this is why nondividing cells that have DNA damage cannot progress to become cancer cells. Cancer originates from cells that are capable of dividing, predominantly tissue SCs and possibly tissue PCs. These are the cells that are responsible for proliferating to replace tissue cells that die through normal cell aging.

Following an indeterminate number of cell divisions during the progression phase, a sufficient array of additional mutations in a variety of genes are required to ultimately give succeeding generations of proliferating cells the ability to express a cancer phenotype with certain hallmark functions. Cancer cells typically have a sustained ability to stimulate their own cell division, an inability to differentiate into normal adult tissue cells, an inability to respond to normal cellular signals that suppress cell division, an inability to respond to apoptosis signals, the ability for indefinite proliferation, an ability for invasiveness and metastasis, and an ability for stimulating angiogenesis.^{17,204,205} Clearly a lot of different genes need to be mutated for all these functions to be acquired, which brings us to another concept of mutations—where the mutations are in genes that lead to the acquisition of the functions that are necessary for developing the cancer phenotype, they are called "driver" mutations. Essentially these are mutations that are directly involved in the transformation of precancerous cells into cancer cells. Mutations that occur during the multigeneration transformation process, which do not alter function of a protein or do not confer a change in function that contributes to the transformation process, are called "passenger" mutations. In a recent study of the entire genome of cultured metastatic melanoma cells from a single patient (in comparison to cultured noncancer cells from the same patient), over 33,000 somatic base substitutions were observed, with 187 occurring in coding regions.²⁰⁶ With approximately 2% of the entire coding genome bearing nonsynonymous mutations in this malignant melanoma, it is clear that the vast majority of mutations are passenger mutations. And while even all the driver mutations cannot possibly be discussed here, some important examples will be mentioned because they are common to many different cancers and they provide a clear illustration of the progression phase.

Driver mutations of *Ras* and *Raf* genes already have been mentioned as mutations that enhance activation of signaling pathways that ultimately lead to cell division. Other driver mutations that are common to human cancers lead to the accelerated acquisition of mutations by altering different functions that regulate the process of

cell division. Two frequently mutated genes in human cancer are the *Rb* and *P53* genes, and in fact, just about every human cancer has mutations in one or the other (or both) of these genes.^{207–212} The gene products, Rb and p53, are important regulators of the cell cycle, with Rb involved in stimulating entry into the S phase from the G1 phase (or G0) and p53 in checkpoint control. While all the details of the cell cycle and its regulation will not be discussed, a brief summary to illustrate the concepts as they relate to cancer follows.

6.2.2 CELL CYCLE

The cell cycle is essentially a model for the phases through which a stem or PC will go through as it proliferates.^{213–215} Mitosis and cytokinesis make up the M phase in which the processes of cell division occur. Following cytokinesis, there is a GAP1 (G1) phase during which the daughter cells grow and develop the regulatory controls necessary for their continuing function. From this phase, the cells can continue to progress to the S phase or exit the cell cycle. On exiting the cell cycle, the cells may remain quiescent (G0) or the cell may differentiate into a functional and nondividing tissue cell.^{216,217} This latter is the normal function of one of the two daughter cells from some adult (tissue) SCs and of PCs; of the two daughter cells, one will typically differentiate while the other retains self-renewal functions to continue to be able to replace cells as they die.

As summarized in Figure 6.1, multipotent (in contrast to the pluripotent embryonic SCs) tissue SCs can typically be seen as long-term self-renewal cells that divide infrequently in an asymmetrical manner to renew the short-term self-renewing PCs, spending a majority of their time in G0 arrest. The PCs typically continue through the cell cycle without entering the G0 phase and divide relatively quickly (and asymmetrically) to replace tissue cells that have died, until the PCs succumb to senescence (in days or weeks), die, and need to be renewed.^{216,218} When senescent cells die, most often through apoptosis, caspase 3 activates calcium-independent phospholipase A_2 (PLA₂), which cleaves arachidonic acid (AA) from membrane phospholipids leading to the subsequent release of prostaglandin E_2 (PGE₂). Along with other growth factors, PGE₂ is involved in activating G0/S or G1/S transition in tissue SCs and PCs through activation of G-coupled prostaglandin E_2 (EP) receptors that then activate Ras and PI3K, leading to activation of ERK-MAPK and JNK-MAPK pathways and subsequent production of cyclin-dependent kinases (CDKs) and the activation of mammalian target of rapamycin (mTOR) through PI3k \rightarrow Akt.²¹⁹⁻²²⁵

Regulation of the processes that promote retention of self-renewal properties or differentiation of daughter cells is not a topic of this chapter. In the simplest terms, however, partial withdrawal of MAPK signaling can stimulate exit from G1 while the self-renewal properties are maintained in part through Wnt, Notch, and EGF signaling from the niche cells. Differentiation can be activated by displacement of a daughter cell away from the immediate area on division and is promoted in part through signaling by transforming growth factor β (TGF β) along with prolonged moderate MAPK activities, cell:cell contact, and other signaling events.^{216,217,226–228}

Crossing the transition from G1 (or G0) to the S phase requires the coordination of a variety of cyclins and CDKs and their interaction with Rb and other proteins.



FIGURE 6.1 Summary of the molecular events that regulate cell cycle. (a) Stem cells (SC) slowly proliferate to replenish the faster proliferating progenitor cells (PC) while the PCs proliferate more quickly and differentiate into tissue cells. Wnt, Notch, and epidermal growth factor (EGF) signaling are important mediators that help maintain the stem cell phenotype. (b) In cases of damage in some tissues, SCs are activated to divide quickly to regenerate the tissue cells. (c) A decline in mitogen-activated protein kinase (MAPK) signaling is partially responsible for the SC exiting the cell cycle while prostaglandin E_2 (PGE₂) and other growth factors activate reentry into the cell cycle. (d) On encountering DNA damage, the sensor proteins activate a variety of DNA repair enzymes as well as induce synthesis of a variety of proteins that lead to activation of cdc25 phosphatases and p53 leading to cell cycle arrest and if sufficiently prolonged: apoptosis. Proinflammatory signaling and enabling mutations that confer constitutive activity in any of the proteins involved in activating entry into the cell cycle will enhance risk for cancer by enhancing rates of cell division. Proinflammatory signaling and disabling mutations that compromise function of any of the proteins involved in checkpoint control will increase risk for cancer through compromised repair of DNA damage, reduced ability to force cell cycle arrest, and attenuated apoptosis. (Details are discussed in the text.)

In cells in the G1 or G0 phase, the hypophosphorylated form of Rb forms a complex with E2F proteins and is maintained in a hypophosphorylated state, in part, through the actions of the INK4 and Cip/Kip (p21, p27, and p57) Cdk-inhibitor proteins, ensuring the cells will not progress into the S phase.^{215,229-231} This transition is prevented because the Rb-E2F complexes prevent the six E2F proteins from entering the nucleus and dimerizing with DP1 and/or DP2 to form transcription factors that stimulate the expression of a variety of genes that are necessary for the S phase. With sufficient stimulation by growth factors, K-Ras is activated, which in turn activates the JNK-MAPK pathway and the ERK-MAPK pathway through PI3K, resulting in activation of the transcription factors Myc and Jun (among many others) and the subsequent expression of cyclins A, D, and E. The increased numbers of cyclins then outcompete the CDK inhibitors for binding, allowing free cyclin D to form complexes with Cdk4 and Cdk6 and cyclin E to bind to Cdk2 to activate the kinases, which then phosphorylate Rb. The Rb-E2F protein complexes then break apart, allowing the E2F proteins to dimerize with DP1/DP2 and activate transcription of a large number of S phase proteins including cyclins E and A, dihydrofolate reductase, thymidine kinase, and DNA polymerase, all of which are necessary for entry into the S phase and for DNA synthesis in the S phase.^{215,229-235} The ability of E2F to induce cyclin E, which in turn regulates Cdk2 to enforce Rb phosphorylation, creates a positive feedback loop that helps contribute to the irreversibility of the G1/S transition.^{236–239} Intimately involved in this process is PI3K-mediated activation of Akt, which subsequently activates mTOR to phosphorylate p7086K, an event that is necessary for the enhanced protein synthesis associated with the proliferation activity.^{224,225,240,241} Once in the S phase, the major activity of the cell is the duplication of the chromosomes followed by the G2 phase where the growing cell prepares for mitosis.

The next major transition is from G2 to the M phase and this transition also is controlled through CDK activity. Binding of Cdk1 with either cyclin A or cyclin B activates the kinase. Activation of Cdk1 with cyclin A is a relatively early event that occurs during S phase because this cyclin is induced by E2F proteins and is associated with maintaining hyperphosphorylation of Rb to ensure progression through the S phase into the G2 phase as well as entry into the M phase.^{231,242} Cyclin B/cdk1 complexes that form in late G2 are an essential event for transition from the G2 phase to mitosis and activation of the kinase activity occurs when Cdk1 is dephosphorylated by Cdc25 phosphatases (Cdk125A, Cdk125B, and Cdk125C) that have been activated by Plk1, an activating event that also is performed by cyclin B/Cdk1 in a positive feedback loop.²⁴² This "self-activation" feedback loop ensures increasing amounts of active cyclin B/Cdc1 complexes and a definitive entry into the M phase. Once a cell progresses through the M phase, one of the two daughter cells exits the cell cycle into quiescence (G0) while the other exits to differentiate into a PC. In some cases of more extensive tissue damage, as observed for skin and intestine, SCs can take on the role of PCs and greatly increase the rate of cell cycling to replenish tissue cells, and intestinal PCs can even revert back to a SC phenotype to assist in the tissue regeneration.^{227,228}

From this brief summary of a few of the relevant events, it is clear that mutations of *Rb* that lead to constitutive activity will ensure that cells consistently enter S phase from G1, independent of environmental controls. Mutations in any other genes that

lead to the production of constitutively active CDKs and E2F proteins, overproduction of cyclins, and inactivating mutations in INK4 proteins also will lead to much greater rates of cell division, and all such mutations have been observed in a wide variety of cancers.^{207–212,231,243–245} The other major gene product for discussion is p53, an important component of DNA damage checkpoint controls.

Progression through either the S phase or the M phase of the cell cycle could be mutagenic or even catastrophic if there were DNA damage, compromised integrity of the DNA, or aberrant attachment of the chromosomes to spindles during the process. To avoid such events, there are checkpoint sensor proteins that continuously monitor chromosomes for the presence of DNA damage, for completed DNA replication, and for spindle integrity, and when defects are sensed, the cell cycle is shut down (arrested) until the damage can be repaired.^{246–248} For the purposes of this summary, only the DNA damage checkpoints will be reviewed.

6.2.3 DNA DAMAGE, CHECKPOINT CONTROL, AND APOPTOSIS

When DNA damage is detected by various sensor proteins and repair processes are initiated, the checkpoint functions are activated. Sensor protein complexes such as the Mre11-Rad50-Nbs1/MRN complex, which senses DNA DSBs, activate the transducer proteins ataxia telangiectasia mutated (ATM) kinase and DNA-dependent protein kinase (DPK).²⁴⁷⁻²⁵⁰ ATM activation leads to the immediate attraction of a variety of proteins and enzymes that form a complex and bind to the damaged DNA, proteins including Nbs1 and BRCA1/2.^{248,251,252} Nbs1 and BRCA1/2 are involved in coordinating the activity of the various proteins involved in homologous repair of the DSB, while the activated DPK can form complexes with Artemis (and others) to activate nonhomologous end-joining repair of DSBs.^{249,252,253}

Rad17 and the Rad9-Rad1-Hus1/9-1-1 complex, along with a variety of other sensor proteins, detect a diverse array of DNA lesions and then recruit DNA repair complexes that perform the functions of DNA repair: base excision repair, nucleotide excision repair, mismatch excision repair, as well as activate the ATM and Rad3-related (ATR) kinase.²⁴⁷⁻²⁵⁰ In addition to being involved in coordinating DNA repair mechanisms, the ATM, ATR kinases and DPKs also are important in initiating cell cycle arrest. ATM and ATR phosphorylate the checkpoint kinases Chk1 and Chk2, which then, along with DPK, phosphorylate a variety of proteins involved in cell cycle arrest including p53 and the Cdc25 phosphatases.^{248,251,252,254} Phosphorylation of Cdc25A inactivates the enzyme and targets it for degradation, in turn reducing activity of the cdk's and suppressing further progression through the cell cycle. Activation of p53 leads to increased expression of p21, which blocks activated cyclin E and cyclin A/Cdk2 complexes and GADD45, which blocks activated cyclin B/cdk1 complexes, effectively shutting down progression through the cell cycle.^{247,248,255} As long as the DNA damage sensor proteins detect damage, the ATM, ATR, Chk1/2 kinases, and DPK will continue to be activated and cell cycle arrest will be maintained. When the damage is repaired, the cell then reenters the cell cycle, in part due to Plk1-mediated activation of Cdc25B, which then dephosphorylates cdk1 to restore its activity.^{256,257}

From the preceding discussion, a major role for p53 in this sequence of events is to ensure cell cycle arrest through inducing expression of the cyclin/cdk complexinhibitor proteins p21 and GADD45, which then delay progression until the DNA damage is repaired. Another major function of p53 is to initiate apoptosis in the face of severe genotoxic stress.²⁵⁸⁻²⁶³ Apoptosis is an ordered process in which cells die in response to DNA damage or hypoxia (intrinsic, mitochondria-mediated pathway) or through the activation of death receptors (extrinsic, TNF receptor 1, Fas, TRAILR1/2, and other receptor-mediated pathways), or by activation of granzymes A/B by perforins that originate from cytotoxic T cells.^{264,265} The extrinsic and intrinsic pathways activate caspase 9 and then caspase 3, ultimately leading to the activation of endonucleases that then degrade chromosomal DNA and proteases that degrade nuclear and cytoskeletal proteins to completely disrupt all cellular activities. The granzymesmediated pathways involve both caspase-dependent and caspase-independent pathways to produce the same result. The cells shrink and the chromosomes condense and fragment while extensive blebbing of the plasma occurs. They then break up into smaller apoptotic bodies that are phagocytized by adjacent cells and macrophages without initiating an inflammatory response.²⁶⁴⁻²⁶⁶ For the purposes of this brief review, only the p53-mediated intrinsic pathway will be summarized.

The major proteins that activate mitochondrial-mediated apoptosis are Bax and Bak, which bind to the mitochondrial membrane and form pores that allow the release of cytochrome *c*, deoxyadenosine triphosphate, second mitochondrial-derived activator of caspase (SMAC), and other apoptogenic factors that lead to the activation of caspase 9. This caspase then cleaves caspases 3, 6, and 7 to activate them to carry out the specific molecular events of apoptosis.^{267,268} Bax and Bak are normally prevented from forming Bax:Bak heterodimers (or homodimers) by being complexed with B-cell lymphoma (BCL) and myeloid-cell leukemia (MCL) proteins. Activation of p53 by checkpoint kinase activity leads to increased transcription of Bax, Bid, Puma, and Noxa. Bid, Puma, and Noxa are BH3-only proteins that can bind to BCL and MCL proteins to block their ability to form complexes with Bax and Bak, thus releasing Bax and Bak to form dimers and bind to the mitochondrial membrane and initiate the release of the apoptogenic proteins. By inducing expression of these BH3-only proteins as well as that of Bax, the ratio of noninhibited to inhibited Bax and Bak increases and apoptosis is activated.^{262,267–270}

The preceding sequences of events described for p53 indicate that it is involved in activating both cell cycle arrest and apoptosis, a situation that produces somewhat of a conundrum. Where is the dividing line between sufficient DNA repair to allow reentry into the cell cycle and sufficient activation of apoptosis mechanisms that the cell self-destructs in spite of continuing successful efforts to repair DNA? This is unfortunately an area not very well understood although it may be related in part to calcium and ROS-mediated mitochondrial events.

Entry of calcium into the mitochondria activates the mitochondria permeability transition, which reduces electrochemical gradients to inhibit ATP synthesis as well as allow the release of mitochondrial calcium to the cytosol. This leads to an increase in cytosolic Ca⁺⁺ and the activation of numerous lipases and proteases that then participate in the apoptosis program, events that have traditionally been associated with cell death through necrosis and have since been documented to occur with apoptosis that has been stimulated by toxins.^{266,271,272} In addition to these events, activated caspase 3 can cleave the plasma Ca⁺⁺–ATPase exchanger as well as the Na⁺–Ca⁺⁺ exchanger to enhance cytosolic levels of Ca⁺⁺, also leading to the activation of numerous lipases and proteases.²⁶⁶ Thus, deregulation of Ca⁺⁺ control is an important player in the progression of apoptotic events.

Another interesting issue with mitochondrial-associated apoptosis is that redox events also can initiate cytochrome *c* release and activate caspases to initiate apoptosis. Oxidation of cardiolipin by H_2O_2 leads to disassociation of cytochrome *c* from the inner membrane of mitochondria and its subsequent efflux.^{273,274} This, coupled with the ability of ROS to activate JNK-MAPK signaling to promote apoptosis through increased expression of Bak (intrinsic pathway) and TNF α (extrinsic pathway), indicates the importance of ROS in mediating pro-apoptosis events in addition to that of p53.²⁷⁵ As calcium overload in mitochondria leads to increased ROS production and subsequent cytochrome *c* release,²⁷⁶ there is a progressive increase in mitochondrial-mediated pro-apoptotic signaling over time initiated by the binding of Bax and Bak to the outer membranes of mitochondria and mediated by both Ca⁺⁺ and ROS.

Because all mitochondria are not initially affected by Bax:Bak, it is possible that it is the duration of Bax:Bak activation that impacts the progression of the apoptotic process. The longer the duration of caspase 3 activation, the greater the deregulation of cytosolic Ca⁺⁺; leading to increasing degrees of mitochondria impairment by Ca⁺⁺, increasing production of ROS, additional cytochrome c release, and therefore further progression into cell-wide apoptosis signaling. On the other hand, when Chk-mediated activation of p53 diminishes and Bax and Bak are again inhibited by BH3-only proteins, the cell may recover. Recovery may be, however, dependent on the number of functional mitochondria remaining, coupled with the integrity of the redox-control systems. Without sufficient ATP synthesis from the remaining mitochondria, adequate function of antioxidant enzymes, and appropriate redox control, repair from the initial damage and reversal of the ROS-mediated activation of pro-apoptosis signaling would be difficult and the cell will continue to progress through apoptosis in spite of successful DNA repair. While this may be a relatively simplistic interpretation of known events, it is meant to illustrate the complexity of the cellular events that are associated with regulation of the apoptosis recovery processes.

Thus, both apoptosis and cell cycle processes are normally well regulated and the role of p53 in these processes in the face of genotoxicity is extremely important. A nonfunctional p53 will not be able to initiate cell cycle arrest or apoptosis. Under such circumstances if cells have DNA damage and then divide, they will produce daughter cells with mutations. Mutations in the genes that code for the proteins involved in the regulation of apoptosis or checkpoint controls also will have a similar end result, greater numbers of mutations in daughter cells. Thus, mutations in these regulatory gene products would all be considered driver mutations. And as expected, mutations in the *p53* gene or in any one or more of the *ATM*, *ATR*, *Chk1/2* kinases, *DPK*, *p21*, *GADD45*, or other of the regulatory or DNA repair genes mentioned in the preceding discussion have been observed in the vast majority of human cancers analyzed.^{208,246,250–252,264,269,277–283}

From the preceding discussion, ensuring proper regulation of cell cycle, DNA repair, and apoptosis is essential to minimize risk for mutagenesis, and driver

mutations in any of the genes related to these functions produce a significantly enhanced risk for developing a cancer. Mutations in the Rb gene that prevent Rb from binding to E2F proteins and inactivating mutations in the p53 gene would produce sustained proliferative signaling as well as escape from apoptosis, both hallmark functions of cancer cells.^{204,205} Mutations in any of the other genes discussed in this section also would contribute to these same hallmark functions. revealing that there are many different mutations that can contribute to the same gain in function. This is essentially why so many different mutations are observed in different cancers while there really are only a half-dozen or so hallmark functions of cancer; there is no single essential mutation that produces a specific gain in hallmark function. Thus, accumulation of specific driver mutations in the promotion phase that produces a growth advantage for subsequent generations of cells is really a selection process for function and not for specific (mutated) genes. This selection for function is one reason for the heterogeneous array of mutated genes observed in cancer tumors from different individuals with a single form of cancer as well as differences in the arrays of mutations in cancers that arise in different tissues.^{206,284–287} This leads to the last etiology topic before focusing on mechanisms of carcinogenesis that are amenable to prevention.

6.2.4 CANCER STEM CELLS AND TUMOR ENVIRONMENT

Different cells within a given tumor also express differences in phenotype in spite of the fact that they probably originated from a single mutated SC (mSC), one of the concepts that led to the relatively recent revisions to the cancer SC hypothesis.^{288,289} As mutations accumulate in subsequent generations of mSCs, the original differentiating function still may occur until a sufficient number of mutations have occurred to prevent differentiation in the daughter cells, a point when the mSCs have been transformed into cancer SCs (CSCs) that then continuously divide symmetrically. Until this happens, however, the mSCs will have produced an array of PCs with a variety of mutations (mutated PCs [mPCs]) that then produce daughter cells with slightly differing arrays of mutations and phenotypes, essentially a small array of differentiated and partially differentiated tissue cells. Some of these partially differentiated cells may retain the ability to respond to proliferating signals and subsequently divide in response to chronic stimulation. As mutations accumulate in these daughter cells, they may dedifferentiate even further to take on the proliferative characteristics of PCs. Some of these "newly dedifferentiated-progenitor" cells along with the mPCs that arose from the mSCs may even acquire the self-renewal characteristics of SCs and essentially become CSCs.^{18,290} The ability of PCs in intestinal crypts to revert to a SC phenotype following damage indicates that this is certainly a possibility.227 As these cells divide and gradually accumulate within a tumor, the tumor then acquires an array of proliferation-capable cells that can contribute to tumor mass: CSCs with unlimited self-renewal capabilities and an array of mPCs. Thus, the CSCs essentially seed the development of a heterogeneous array of larger than normal numbers of mPCs that proliferate more efficiently in response to normal proliferation signals (such as PGE_2) to drive tumor growth.

An important point for tumorigenesis and progression is that the heterogeneity of cells within a tumor is not restricted to SCs, CSCs, PCs, mPCs, and an array of dedifferentiated (and appropriately differentiated) tissue cells. Virtually all (solid) tumors contain a variety of inflammatory associated stromal cells that contribute to the tumorigenesis process, cells that include macrophages, lymphocytes, natural killer (NK) cells, mast cells, endothelial cells, and fibroblasts.78,291,292 While the actual order of events that leads to infiltration of the tissue with the individual types of stromal cells is unknown, the possibility exists that it starts as mSCs, and mPCs initially begin to accumulate and the distance from the new precancerous cells to the existing capillaries increases, causing a corresponding decrease in oxygen availability in these cells. Hypoxia induces activation of the transcription factor hypoxiainducible factor 1 (HIF-1), which subsequently activates transcription of a variety of proteins including glycolytic enzymes, pyruvate dehydrogenase kinase (PDK), glucose transporters 1 and 3 (GLUT1 and GLUT3), vascular endothelial growth factor (VEGF), insulin-like growth factor (IGF), TGF α , as well as the monocyte chemoattractant protein MCP-1.293-297 The increase in glycolytic enzymes will confer a metabolic advantage under the hypoxic conditions experienced by some tumor cells. In addition, the expression of PDK suppresses oxidative metabolism of pyruvate by inactivating pyruvate dehydrogenase. This will, through a series of moderate product-inhibition effects on the enzymes within the glycolytic pathway, enhance cellular levels of glycolytic intermediates, including glucose 6-phosphate, which is the primary substrate that enters the pentose-phosphate pathway. This metabolic pathway is necessary for nucleotide synthesis; thus, increased PDK activity enhances an essential synthesis function in cells that are dividing quickly.

The increase in MCP-1 contributes to the attraction and activation of monocytes, which, following a cascade of proinflammatory signaling events, leads to the gradual accumulation of other inflammatory cells, including platelets. Activated platelets release a large variety of factors including IL-1β, TNFa, platelet-derived growth factor (PDGF), TGFβ, EGF, basic fibroblast growth factor (bFGF), and the chemokines RANTES, platelet factor 4, and IL-8, adding to the VEGF and IGF released in response to HIF-1.²⁹⁸⁻³⁰⁰ While platelets are traditionally associated with blood clot formation, they also are intimately involved in tissue remodeling. The PDGF, TGF β , as well as the various chemokines initiate chemotaxis of tissue macrophages and fibroblasts into the area that adds to the heterogeneity of the precancerous tissue. The PDGF also has the effect of stimulating mitogenesis of fibroblasts while TGF^β stimulates the macrophages to release (among other factors) fibroblast growth factor (bFGF) and PDGF, both of which are essential for angiogenesis in tissues.^{301–303} Thus, through the infiltration and activation of inflammatory-associated cells, not only is a continuous supply of growth factors available for stimulating proliferative activity in any cells that can respond to them (SCs, mSCs, CSCs, PCs, and mPCs), but also an activation of the processes that lead to angiogenesis, which helps to sustain growth of the developing tumor, another hallmark function.78,204,205,291,292,304

Continued inflammatory signaling by the infiltrating inflammatory cells can activate expression and release of CD40 from platelets, which then activates the release of a variety of matrix metalloproteinases and serine and cysteine proteinases from endothelial cells, macrophages, mast cells, and others. These proteases then degrade

collagen, VE-cadherin, fibronectin, and other extracellular matrix adhesion molecules to contribute to the ability of a growing tumor to invade adjacent tissues, again, another hallmark function of cancer.^{304–306}

Thus, the infiltrating stromal cells provide essential contributions to the tumorigenesis process without actually being involved in a classic inflammatory response such as one that occurs with infection or tissue damage, revealing why "inflammation" has consistently been associated with risk for cancer. This does not mean that inflammatory conditions from infection or damage cannot contribute to this process because they certainly can. Any damage or infection within the precancerous tissue will activate a localized inflammatory response that will then attract more inflammatory cells to the local region and exacerbate the already protumorigenic effects of the initial infiltrating stromal cells. Inflammatory signaling molecules from damage/ infection sites elsewhere in the body or from proliferating adipose tissue also can enter the circulation and these too can exacerbate the protumorigenesis environment of a developing tumor.

6.2.4.1 Contribution of Obesity and Insulin Resistance to the Tumor Environment

Obesity, especially visceral obesity, also is associated with a "proinflammatory state" in which proinflammatory cytokines that enter the circulation originate from the adipose tissue itself.^{157,160,307–310} The inflammation-associated cytokines include TNF α , IL-6, IL-1 β , and MCP-1. Once they enter the circulation, they can then enhance proinflammatory signaling in precancerous and cancerous tumors, a major factor in the known link between obesity and an increased risk for many cancers.^{311,312} While other mechanisms associated with infiltrating inflammatory cells that contribute to tumorigenesis will not be reviewed here, it is important to note that because these inflammation-associated signaling processes are essential to the cancer process, anti-inflammatory measures should therefore reduce risk for tumorigenesis.

Closely related to the increase in risk due to obesity is an increase in risk for cancer with insulin resistance. Metabolic syndrome, which includes insulin resistance as a required symptom of the syndrome, and obesity are very tightly associated, very likely through the inactivity that often leads to obesity.^{158,313,314} The proinflammatory cytokines (especially TNFa) produced by adipocytes, coupled with the prooxidant effects of inactivity, lead to insulin resistance in peripheral tissues and a compensatory increase in insulin levels. Insulin enhances activation of the ERK-MAPK pathway (through PI3K) and increases production of IGF-1 by liver, both of which activate (through PI3K) the ERK-MAPK, JNK-MAPK, and p38-MAPK pathways as well as mTOR signaling (PI3K-PKB-Akt-mTOR-P70S6k) to produce growth and proliferative responses. It is no wonder that insulin resistance is associated with risk for a variety of cancers!64,165,315-322 Add to these signaling effects the role of glucose as a major substrate for both ATP synthesis and DNA synthesis, and the increase in serum levels of glucose and insulin that occur with insulin resistance can only be an advantage to proliferating CSCs, mSCs, and mPCs in a slightly hypoxic tumor environment. And this advantage is certain to be enhanced through the commonlyobserved increase in expression of glucose transporters and of insulin receptors in many cancer cells.321,323-328

In addition to the insulin-mediated enhanced risk for cancer, the increased levels of glucose in serum comprise an independent source of risk. The aldehyde group of open-chain glucose can form a Schiff base with available amino groups to initiate the formation of advanced glycation end-products (AGE).^{329–331} These products cause cellular and DNA damage through enhancing rates of lipid peroxidation and they also activate proinflammatory signaling pathways through interacting with AGE receptors (RAGE).^{330,332–336} Thus, obesity can contribute to risk for cancer through a cascade of events initiated by adipose-mediated inflammatory signaling that leads to insulin resistance and elevated insulin, IGF-1, and glucose. The elevated insulin and IGF-1 levels enhance activity of the proliferative and growth signaling pathways while the increased glucose can enhance cellular and DNA damage (DNA damage is discussed in Section 6.2.5) and initiate additional proinflammatory signaling as well, truly risk-enhancing for the tumor environment.

From this brief review of cancer etiology, it is clear that accumulating mutations in SCs and PCs are the driving force behind tumorigenesis, while signaling processes from infiltrating somatic cells (and circulating cytokines) are essential accessories to the process. Once formed, there is little that can be done to repair mutations. For some mutations, however, there are now drug therapies that utilize a variety of inhibitor molecules to attenuate the effects of specific gain-in-function mutations, with many more "in the pipeline." From the perspective of prevention, these therapies are "after-the-fact" attempts to manage the deleterious effect of the mutations to prevent further development of an existing tumor or the development of a secondary tumor. For prevention to be most effective, however, preventing the mutations in the first place is the best strategy. And from this standpoint, reducing the DNA damage as well as increasing repair of the inevitable remaining damage is a logical focus.

6.2.5 DNA DAMAGE

6.2.5.1 Reactive Oxygen and Nitrogen Species

DNA damage is caused by a large variety of reactive molecules that are produced during metabolism of both endogenous and exogenous molecules. ROS are one class of reactive molecules produced that can either attack DNA directly or they can initiate oxidative damage to other biological molecules, the products of which then cause the DNA damage.^{105,277,337–339} While a comprehensive discussion of radical chemistry is beyond the scope of this chapter, examples of some of the more biologically relevant reactions will be discussed.

The major endogenous sources of radical species in biological systems include the generation of superoxide anion radical (O_2^{-}) by mitochondria during aerobic metabolism, by the NADPH-oxidase activity of activated G-coupled proteins, by phagolysosomes of activated neutrophils and macrophages, by cytochrome P450 monooxygenases (CYP), as well as nitric oxide (NO·) produced by nitric oxide synthase in vascular endothelial cells and activated neutrophils and macrophages.³⁴⁰⁻³⁴⁸ Neither O_2^{-} nor NO· is particularly toxic in aqueous solutions; however, the superoxide anions can spontaneously dismutate to produce hydrogen peroxide: $O_2^{-} + O_2^{-} + 2H_2O \rightarrow H_2O_2 + O_2 + 2OH^{-}$. This dismutation is greatly accelerated by

the various superoxide dismutase (SOD) enzymes, thus ensuring that the predominant soluble oxidant in the cells will be H_2O_2 . The three isozymes of SOD include the cytosolic copper-zinc form of SOD in the cytosol, nucleus, and mitochondrial membrane (Cu/ZnSOD—SOD1), the extracellular copper-zinc form (ECSOD—SOD3), and the manganese form (MnSOD—SOD2) in the mitochondrial matrix. H_2O_2 also can be produced directly by xanthine oxidase-mediated oxidation of hypoxanthine and xanthine during uric acid synthesis. While the nonradical H_2O_2 is a relatively weak oxidant, it can easily be reduced by superoxide or by multivalent metal ions such as Fe^{2+} or Cu⁺ to produce hydroxyl radicals that are possibly the most aggressive oxidants known: $O_2^{-}(\text{or Fe}^{2+}) + H_2O_2 \rightarrow OH + HO^- + O_2(\text{or Fe}^{3+})$. Because superoxide can reduce Fe^{3+} and Cu^{2+} so much faster than it can reduce hydrogen peroxide, in all likelihood the more relevant in vivo reaction involves superoxide as a reducing agent for metal ions with the reduced metal then reacting with the hydrogen peroxide. The production of hydroxyl radicals through these reactions is of major importance to DNA damage and mutagenesis. Hydroxyl radicals have both strong oxidant and electrophilic characteristics, which make them particularly damaging to DNA (and anything else for that matter).^{349–351} It is most likely that H_2O_2 produced elsewhere in the cell diffuses into the nucleus where it then reacts with a metal ion to produce OH, which then reacts with susceptible atoms of the closest DNA molecule in the immediate vicinity.

A major endogenous reactive nitrogen species (RNS) is peroxynitrite (ONOO⁻), a strong oxidant that is produced when nitric oxide (NO) reacts with O_2^{-} . The ONOOcan react with CO₂ to produce carbonate radicals (CO $\frac{1}{3}$) and nitrogen dioxide radicals (NO_2) , both of which are oxidants. The RNS have traditionally been thought of as relevant to infection or damage-related inflammatory reactions because of the production of NO· by various inflammatory cells. With the recognition that infiltrating stromal cells are intimately involved in tumorigenesis, these NO-derived reactive species are now known to contribute to the tumorigenesis process through causing DNA damage without an actual infection or damage-associated inflammatory response being present. This source of DNA damage is in addition to that produced by hydroxyl radicals that are derived from normal metabolic and signaling processes. Thus, the major endogenous oxidants produced include ·OH, ONOO-, NO_2 , and CO_3 . While all possible reactions of these molecules with each susceptible component of the different DNA molecules cannot be described here, several of the predominant reactions will be used to illustrate the general principles of how DNA damage ultimately leads to risk for cancer.

One mechanism through which hydroxyl radicals can cause damage to DNA molecules is through abstracting a hydrogen atom from C-H bonds of the deoxyribose component of DNA; predominantly from the C5 position due to its greater exposure to the external solvent when in the double-stranded configuration.^{349,352,353} Abstracting the H leaves behind a carbon-centered allyl radical that converts to a peroxyl radical through addition of molecular oxygen and then an oxyl radical that ultimately leads to strand breaks at this position. As described in Section 6.2.1, unrepaired single-strand breaks can lead to DSBs during replication and these DSBs can then lead to a variety of chromosomal mutations in daughter cells. DSBs also can be caused by ionizing radiation such as that emitted by uranium, plutonium, and radon.

Hydroxyl radicals also can abstract a hydrogen atom from the C5 methyl group of thymine leaving an allyl radical, which then leads to the production of 5-hydroxymethyluracil and other end-products following the addition of oxygen. Addition reactions of OH with DNA bases can lead to a wide array of DNA damage products. OH adds to the various double bonds of the DNA bases, predominantly at the C4, C5, and C8 of purines and at the C5 and C6 of pyrimidines to produce a variety of OH adduct radicals.348,349,352-356 These radical products can then be oxidized, or reduced, to produce an array of stable adducts including 8-oxo-7,8-dihydrodeoxyguanosine (8-oxo-dG) and 8-oxo-7,8-dihydrodeoxyadenosine (8-oxo-dA), thymine glycol, and 5-methyldeoxycytidine (5-methyl-dC) among many others. When these lesions are not repaired and the cell progresses through the S phase, replication errors are made and mutations will occur. For example, in a variety of Escherichia *coli* and cell culture model systems, 8-oxo-dA and 8-oxo-dG lesions lead to $A \rightarrow C$ transversions and $G \rightarrow A$ transitions/ $G \rightarrow T$ transversions in daughter cells, respectively. Thymine glycol will lead to $T \rightarrow C$ transitions and 5-methyl-dC leads to $C \rightarrow T$ transitions. If these mutations occur in the coding region of a gene that is responsible for cell cycle control or DNA repair, then risk for cancer can be greatly enhanced.

Hydroxyl radicals also can initiate peroxidation reactions in the unsaturated fatty acids of phospholipids that result in the production of a wide variety of reactive products; the major reactive products include malondialdehyde, acrolein, crotonaldehyde, and 4-hydroxynonenal.^{354,355,357-361} Malondialdehyde forms exocyclic adducts by reacting with the exocyclic amino groups of C2, C4, and C6 of guanosine, cytosine, and adenosine, respectively, potentially leading to $G \rightarrow T$ transversions and $C \rightarrow T$ and $A \rightarrow G$ transitions. Acrolein, 4-hydroxynonenal, and crotonaldehyde also react with the exocyclic amino groups of DNA to form a variety of propano-adducts, while reactions of these with peroxide can produce the much more reactive epoxide derivatives, which react with exocyclic amino groups as well to form a variety of etheno adducts. These adducts are associated with producing a variety of mutations including $G \rightarrow T$ and $A \rightarrow C$ transversions, $C \rightarrow A$ transversions and $C \rightarrow T$ transitions, and $A \rightarrow T$ and $A \rightarrow C$ transversions, with etheno-dG adducts being approximately 100 times more common than those of etheno-dA and etheno-dC adducts. Most interestingly, all of these adducts are increased under conditions of oxidative stress.

In addition to the preceding, DNA lesions resulting from oxidation by ONOO⁻, NO₂·, and CO₃⁻ also occur. Of these three major products of NO + O₂⁻ reactions, peroxynitrite is not only relatively stable but it also can readily diffuse across cellular membranes, rendering it the major NO·-related oxidant capable of oxidizing DNA.^{353–355,362} ONOO⁻ (and its protonated form peroxynitrous acid) preferentially oxidizes guanosine to 8-nitroguanosine and 8-oxo-dG. 8-Nitroguanosine spontaneously depurinates leaving an apurinic site leading to G \rightarrow T transversions. The 8-oxo-dG (also produced by ·OH) is in turn far more reactive with ONOO⁻ than is the unmodified dG.^{353,363} Subsequent oxidation of 8-oxo-dG by ONOO⁻ produces oxazolone, spiroiminodihydantoin, and guanidineohydantoin. Thus, under the "normal" transforming conditions of a precancerous tumor, the mutagenic 8-oxo-dG (G \rightarrow T transversions) will be produced by both ·OH and ONOO⁻, and with a second oxidation event by ONOO⁻ oxazolone (G \rightarrow T transversions),

spiroiminodihydantoin (G \rightarrow T and G \rightarrow C transversions) and guanidineohydantoin (G \rightarrow C transversions) will result. Peroxynitrite also reacts with the exocyclic amines of dG, dA, and dC to produce reactive diazo and nitroso adducts that hydrolyze to produce xanthine, hypoxanthine, and uracil. These DNA lesions can lead to both G \rightarrow C transitions and G \rightarrow T transversions. ROS and RNS oxidation products are therefore responsible for producing G \rightarrow C and G \rightarrow T transversions as well as a variety of other mutations.

6.2.5.2 Xenibiotics

Exposures to exogenous compounds (xenobiotics) that are directly reactive, or that can be activated to reactive compounds, also are major contributors to DNA damage and subsequent risk for mutagenesis and cancer. Tobacco smoke, whether through first- or second-hand exposure, is an obvious source of exogenous compounds with carcinogenic potential. In addition to tobacco smoke, humans are exposed to a wide variety of carcinogenic xenobiotics in the air, water, and food. A major source of exogenous carcinogens is combustion products arising from a variety of common activities such as smoking tobacco (and other things), smoking and grilling meats and other foods, burning oil, gas, coal, and other biomass derivatives in mobile (automobiles) and stationary (steel mills, power stations, refineries, waste incineration, etc.) devices, and from residential wood and vegetation burning. These contain hundreds of different carcinogenic compounds including a wide variety of polycyclic aromatic hydrocarbons (PAHs), nitrosamines, ketones, aldehydes, quinones, metals, and free radicals, among others.^{364–373} In addition to breathing air that contains such compounds, these same emissions are deposited in the water and soils and they then enter the food chain such that it is difficult to eat or drink anything that does not contain at least trace amounts of these compounds. Add to this the vast array of naturally occurring toxic (and potentially carcinogenic) chemicals made by the plants themselves, and the number of different reactive compounds in the environment that could contribute to risk for cancer is far too extensive for all of them to be discussed here. Instead, a few general principles of how they contribute to the processes of DNA damage will be highlighted.

Exogenous sources of carcinogens tend to produce different effects in different tissues, effects that tend to be greater in tissues of first exposure.^{374–378} Those tissues of first exposure will get the greatest concentration of potentially harmful molecules, and if they are inherently reactive (ROS, RNS, aldehydes, ketones, epoxides, etc.), these tissues will suffer the greatest damaging effects with tissues further "downstream" suffering much less damage simply because of the relatively short half-life of the reactive molecules. With airborne carcinogens the order of exposure will most likely be oral/nasal \rightarrow pharyngeal \rightarrow trachea \rightarrow lung \rightarrow circulation whereas with water-soluble carcinogens it would more likely be oral \rightarrow pharyngeal \rightarrow esophagus \rightarrow GI tract \rightarrow liver \rightarrow circulation. Many exogenous carcinogens require activation to a reactive form before they can cause DNA damage. And, the majority of the activation reactions are catalyzed by various forms of cytochrome P450 enzymes (CYP).^{377–379}

Tobacco products probably comprise the single greatest voluntary source of exogenous carcinogens for humans. Tobacco products have been estimated to be used by 32.2% of men and 18.5% of women for a total prevalence of $\sim 25\%$ of the

U.S. population (with similar figures in other high-income countries) while in low-income countries as much as ~50% of men and ~8% of women smoke for a total of over 1.1 billion smokers worldwide.^{380,381} Tobacco smoke contains a wide variety of carcinogens including an array of PAHs, nitrosamines, aldehydes, metals, nitrogen oxides, acrolein, and many others, compounds that are found in almost all combustion-associated air pollution.^{367,370,382} Because of this, tobacco smoke provides a highly relevant model for risk for cancer from most exogenous sources. And of all of the exogenous compounds found in combustion products, the PAHs and nitrosamines produce the greatest mutagenic potential, with benzo[a]pyrene (BaP) and the tobacco-specific nitrosamine—4-(methylnitrosamino)-1-(3-pyridyl)-1-butanone (NNK) being among the best characterized of these.

BaP and NNK are both known to produce mutations as well as to cause lung cancer in a variety of animal models.^{370,382} For these compounds to exert their mutagenic effects, they must first be activated by CYP. CYP enzymes comprise a gene superfamily of over 2000 heme-containing monooxygenase enzymes in more than 200 related gene families that (collectively) are present in all eukaryotes (and many prokaryotes as well) and are responsible for metabolizing a wide array of compounds.^{379,383–386} In brief, CYP enzymes comprise the majority of phase I enzyme activities that are part of a general two-step progression in the metabolic transformation of lipid soluble xenobiotics into water-soluble compounds for easy excretion in the urine. Other phase I enzymes include the flavin-containing monooxygenases, peroxidases, dehydrogenases, and esterases among many others. The basic purpose of phase I enzymes is to add a functional group such as a hydroxyl or an oxide to the molecule so that phase II enzymes can then transfer a more water-soluble component to it. Phase II enzymes include several families of enzymes such as the UDP-glucuronosyl transferases (GT), glutathione-S-transferases (GST), and the acetyltransferases (among several others). Binding of a substrate to the heme iron of CYP triggers the reduction of the iron by an electron from CYP reductase, the ferrous iron then binds molecular oxygen. Following a second reduction to form an ironperoxo complex and a near immediate protonation to the iron-hydroperoxo form, the activated oxygen then reacts with the substrate; one oxygen atom is inserted into the substrate while the other leaves as water. In some cases, the leaving product is relatively nonreactive and subsequent conjugation reactions allow for easy excretion. In others, the products that are released are highly reactive electrophiles that readily form adducts with a variety of other molecules, including DNA. In addition to the production of reactive products from substrates, CYP reactions also can become uncoupled. Although relatively stable, the originally formed iron-oxygen complex can rearrange to produce a superoxide that dismutates within the enzyme and then disassociates to generate free H₂O₂. The superoxide also can disassociate freely from the enzyme and subsequently react with the H_2O_2 to produce $\cdot OH$; thus, uncoupled CYP enzymes can be a significant source of damaging ROS in addition to the production of electrophilic substrates. Of the 18 mammalian gene families, the CYPI, CYP2, CYP3, and CYP4 families are most active in the activation of xenobiotics, and in addition, members from each of these families have been documented to produce significant quantities of ROS as well.377,378,386,387

As mentioned earlier in this section, PAHs and nitrosamines are important contributors to environmental-associated DNA damage with BaP and NNK being well-studied models for each. The olefins of BaP can be oxidized by CYP1A1, CYP1A2, CYP1B1, and by CYP3A4 to produce an array of BaP-epoxides with BaP-7,8-diol-9,10-epoxide (from CYP1A1) being the major product of BaP metabolism because of the relative high content of this isozyme in lung.^{370,379,382,385} The various diol-epoxides produced from BaP are electrophilic compounds that are highly reactive with the exocyclic amines of DNA. Because guanine is the base that is most susceptible to attack, a preponderance of $G \rightarrow T$ transversions are produced following treatment with BaP. In some cases, oxidation of the olefins can result in the formation of an aldehyde group instead of an epoxide, a reactive group that also reacts with the same exocyclic amines.

Activation of the tobacco-specific nitrosamine NNK also is performed by a variety of CYPs. α -Methyl hydroxylation of NNK by CYP1A1, CYP1A2, CYP2A1, CYP2B1, and CYP2E1 (among others) produces 4-(3-pyridyl)-4-oxo-butane-1-diazohydroxide (plus formaldehyde) while α -methylene hydroxylation by CYP1A2, CYP3A4, and CYP2A6 produces a keto-aldehyde with the concomitant release of methyl-diazohydroxide.^{370,382,385,388} Of these products it is the diazo group that confers the reactivity resulting in pyridyloxobutyl and methyl DNA adducts, which can lead to G \rightarrow A transitions and to a lesser extent to G \rightarrow T transversions. CYP2A6 also can hydroxylate nicotine at the 2' position, which results in opening of the pyrrolidine ring, making it susceptible to nitrosylation with the subsequent formation of NNK, an endogenous reaction that adds to the NNK that is already present in tobacco smoke.

Food-borne xenobiotics comprise another potential source of environmental mutagens and the aflatoxins provide a highly relevant example. Similar to BaP, aflatoxins are (five-member) multiring structures that can be activated to diol-epoxides by CYP3A4 (a major hepatic CYP), which then can bind to susceptible DNA molecules. The resulting aflatoxin-N⁷-G adducts produced lead to $G \rightarrow T$ transversions.^{377,389–393} Aflatoxins are produced by Aspergillus flavus and Aspergillus parasiticus, which commonly infect grains and nuts such as corn, sorghum, wheat, rice, peanuts, and various tree nuts. The fungi proliferate extensively under warm and moist storage conditions to produce what may be the most important food-borne exogenous carcinogen. While exposures to aflatoxin in wealthier countries are nearly universal, the amounts tend to be relatively low and are not considered a major risk, the more modern technology available for harvesting, handling, and storage of foods limits growth of the contaminating fungi. Consumption of aflatoxin-contaminated foods is a major factor for hepatic cancer throughout parts of China and Africa where many people harvest and store their own local produce in less than adequate conditions. The risks for hepatic cancer due to aflatoxin exposures are especially great in regions where hepatitis infections are prevalent. In populations with both hepatitis infections and high aflatoxin exposures, the incidence of hepatocellular cancer is as much as 60 times greater in comparison to either condition alone, indicating the extensive synergy between chronic infections and chronic exposures to these xenobiotic carcinogens. In such populations, hepatic cancer is often the primary cause of cancer-related death and a major cause of death overall. Although aflatoxin is not considered a major exogenous carcinogen in the United States or Europe, it is a major one for many in the world and illustrates the potential for higher-level exposures to food-borne carcinogens to be a major factor in cancer, especially in the face of coexisting chronic inflammation.

Processed meats and red meats also can comprise a source of potentially mutagenic compounds, although they are only moderately associated with increased risk for cancers, including colorectal, gastric, and esophageal cancers.^{132,134,394–397} The heme content of red meat, the presence of nitrate and nitrite in processed meats that lead to the formation of N-nitroso (N-N=O) compounds in vivo, and the formation of heterocyclic amines and PAHs during high-temperature cooking (barbecuing and searing) appear to be associated with risk in these studies. Heme iron is known to contribute to the generation of ROS with subsequent increases in lipid peroxidation. And the increased formation of malondialdehyde and subsequent DNA damage can contribute to carcinogenesis. The ability of PAHs to cause DNA damage following activation by CYP has been described earlier in this section, and interestingly, total daily PAH exposures from food are similar to those from smoking a pack of cigarettes; in epidemiology studies, they even relate more closely to levels of DNA adducts in blood than smoking does.³⁹⁸ This last of course would make sense because everyone eats but only ~25% smoke and also may explain why PAH-related DNA adducts in lungs of smokers are only about 1.5-2 times higher than in nonsmokers. Along the same lines, overall exposure to nitrates from vegetable consumption is actually much higher than that from preserved meats, leading one to believe that even though CYP2E1 activation of nitrosamines can lead to the methylation of guanine and $G \rightarrow A$ transitions, the contribution of this reaction from dietary meat in and of itself is not a major source of risk for cancer.^{398,399} This is not to say that each of these reactions do not contribute to risk because they do, but it does illustrate that there are few situations where a single source of exposure to exogenous mutagens can provide the majority source of risk for any cancer. The last meat-based mutagens to mention are the heterocyclic amines that are produced from surface proteins and amino acids that are pyrolyzed during grilling and searing. N-hydroxylation of heterocyclic amines by CYP1A2 followed by esterification reactions, catalyzed by a variety of phase II transferases, lead to release of an aryl nitrenium ion that then can bind to guanine, producing $G \rightarrow T$ transversions.^{398,400}

While all of the above-mentioned mutagenic reactions occur in humans, the moderate associations between meat consumption and cancer risk indicate that rather than being a major source of risk in and of itself, high levels of processed and grilled meat consumption may more likely exacerbate the presence of other significant risks such as persistent ROS stress from chronic inflammation or high-level exposures to tobacco smoke.

CYP-mediated reactions that result in the activation of carcinogens are an important component of risk for cancer from xenobiotics. Activities of the individual CYP enzymes, however, are not constant and they also tend to be very different for different individuals.^{382,401-403} Chronic exposure to different xenobiotics (and exercise) can differentially alter expression of the CYP enzymes. Drinking alcoholic beverages, for example, can enhance CYP2E1 in many tissues and enhance the activation of NNK

and other nitrosamines, while chronic exposure to PAHs and to heterocyclic amines induces activity of CYP1A1.398,404-407 Such increases in CYP activity are associated with an increased risk for many cancers and may be related to an increase in both carcinogen activation and ROS production by these CYP isozymes. Inactivation of the electrophilic products by phase II enzymes before they can cause damage also is an important component of risk. And similar to phase I enzymes, phase II enzymes vary widely among individuals, and the activities can be modified by exposure to xenobiotics as well as by exercise.^{173,402,408-411} Because phase I and phase II enzymes are regulated through different mechanisms, it is possible that an increase in the former could be accompanied by an even larger increase in the latter, making it an awkward assumption that an increase in phase I activities necessarily means an increase in cancer risk. And to make things even more complicated, enzymes such as CYP1A1 that are involved in activating PAHs to electrophilic epoxides also are involved in hydroxylation of the epoxides to inactivate them. Because of these metabolic complexities, absolute risks due to exposure to different carcinogens in the air, water, and food are difficult to assess in the general population, and although not a topic for this chapter, the concept deserves a brief discussion.

Extensive efforts have been made over the past three decades to incorporate measures of exposure along with a variety of biological markers into risk assessment for cancer, a concept pioneered by Perera and Weinstein with the development of the field of molecular epidemiology.^{412,413} This field is essentially the study of individualized cancer risk in populations through integrating measures of exposure to different carcinogens with the measured individual differences in carcinogen metabolism (activation/inactivation), along with the observed array of DNA damage and mutations.^{207,378,401,412,414–416} Results of such studies have been instrumental in tying exposures to specific sources of carcinogens to risk for specific types of cancer, as long as there are known source-specific mutagens. For example, NNK is unique to tobacco smoke and the presence of pyridyloxobutyl DNA adducts at specific codons, and $G \rightarrow A$ transitions at the same locations in lung tumors, will tie the cancer to tobacco use. In contrast, using PAH-associated DNA damage or mutations as markers for smoking risk would not work because the PAHs found in tobacco smoke also are found in our food, air, and water.

These kinds of studies are not without their own difficulties in interpretation. For example, without the presence of a unique mutagen in a particular environmental source, it is often difficult to differentiate DNA damage due to exogenous sources from that arising from endogenous sources. One major reason for this is that the ultimate forms of DNA damage caused by various reactive compounds arising from air and water pollution are often indistinguishable from those caused through endogenous mechanisms. In instances where it is possible to differentiate the two, such as in those experiments that used labeled versions of the common air pollutants formaldehyde, vinyl chloride, and ethylene oxide, the endogenous DNA damage is more than a thousand-fold greater, and even with the administration of near toxic (to humans) doses of methanol (a common industrial solvent and fuel additive) to rats, hydroxylmethyl DNA adducts from methanol are still lower than this same adduct arising from endogenous sources.^{417,418} DNA damage from exogenous sources of pollutants in general also must be evaluated against a background of as many as 50,000 DNA

adducts/cell that arise from all endogenous sources. When such data is taken into consideration, although risks for cancer are clearly enhanced by exogenous sources, total exogenous risks have been estimated to be from 1% to as high as 19% of the total cancer risk.^{364,415,419-422} Of these values, the greater values are obtained when carcinogenic compounds in food that result from various storage (e.g., contamination from aflatoxin-producing molds) and cooking conditions (smoking and grilling) are included as environmental sources. This is not to say that environmental air and water pollution is not a source of risk and that efforts to protect the environment from pollution are not essential to minimizing risks to our future health, but it is still (relative to endogenous sources) a minority source of the total average risk for the general population.

An issue that is related to air and water pollution is that of exposures to exogenous carcinogens in the workplace and (geographically) localized exposures due to industrial accidents. Of these, the first is a major source of risk for what actually may be a relatively small segment of the population (~152,000 cancer deaths annually as currently estimated by the WHO out of the ~7.5 million annual cancer deaths worldwide).^{3,7,423} The second is certain to be even lower, but it is very difficult to estimate on a global scale simply because of the limited number of accidents and the extended time scale of the resulting carcinogenic effects. In addition, short of changing jobs or moving before an accidental release, there is little an individual can do about exposures such as these.

An area of exogenous risk that has received a lot of attention lately is that of "endocrine disruptors." Endocrine disruptor compounds (EDCs) exert their carcinogenic effect predominantly through mechanisms that ultimately lead to enhanced entry into the cell cycle; they predominantly act as promoters that in some cases (e.g., FD&C Red #3) also may cause DNA damage.424-430 EDCs were originally thought of as xenoestrogens or environmental estrogens for their ability to bind to estrogen receptors as revealed through the study of dichlorodiphenyltrichloroethane (DDT) and diethylstilbestrol. They are now recognized as having endocrine and metabolic effects through interactions with estrogen and androgen receptors, various retinoic X receptors, and the aryl hydrocarbon receptor (AhR). Common EDCs include pesticides such as the organochlorine pesticides (OCPs), which include DDT (banned in 1972 in the United States), heptachlor epoxide, and oxychlordane; the dioxins that originate from paper-bleaching agents and incinerating PVC products including polychlorinated dibenzodioxins (PCDDs) and polychlorinated biphenyls (PCBs); the plasticizers bisphenyl-A (BPA) and phthalates; and the flame retardant polybrominated diphenyl ether (PBDEs). Of these compounds, the OCPs, BPA, phthalates, and PBDEs all interact with estrogen receptors (ERs) and androgen receptors (ARs) to exert their disruptive developmental effects while the PCDDs and PCBs bind to the AhR to produce their effects.

Risks for BC will be discussed to illustrate that much of the enhanced risk for cancer following exposure to many of these compounds appears to be influenced by the timing of the exposures. In utero exposures to many of these compounds are known from animal studies to enhance risk for BC by increasing proliferation of the ductal tissues during organogenesis and early growth (and adolescence) and delaying differentiation of the terminal buds to lobules to enhance the duration of their sensitivity to proliferative signaling, therefore increasing risk for cancer. While these risk-enhancing effects can be mediated by the estrogenic activity of a variety of EDCs, the various dioxins also may play a role in these developmental effects. Dioxins bind to the AhR that then dimerizes with the ARNT protein to bind to the xenobiotic response element to induce expression of a variety of proteins (including CYP1A1). Because of its proximity to the estrogen response element, there is considerable cross talk between AhR- and ER-mediated effects with 2,3,7,8-tetrachlorodibenzodioxin (TCDD) being capable of eliciting estrogenic effects.^{431,432} In addition, ERs act as co-activators of AhR-mediated responses, indicating that it is possible for dioxins and xenoestrogens to act in an additive or possibly even synergistic fashion, leading to enhanced risks during susceptible periods of growth.

Induction of CYP1A1 (msp1 polymorphism) by dioxins is associated with hydroxylation of estrogen to a metabolite that retains its biological activity and also causes DNA damage.^{433–435} These effects would be consistent with the elevated risk for BC observed with occupational exposures to both classes of EDCs.⁴²⁶ Environmental exposures during adolescence also might increase risk for BC through the potential additive effects of these EDCs with the ovarian estrogens, and possibly compounded by the effect of the many other growth factors that are present during adolescence and that lead to hyperphosphorylation of ERs to greatly enhance their promoting activities. Risks for PC also appear to be affected by EDCs when exposures occur during fetal development and adolescence through similar mechanisms, a logical result because prostate SCs and PCs express the full repertoire of ERs as well as ARs and AhRs.^{436,437}

As with other environmental risks, there may be little that can be done about exposures to EDCs from air and water pollution, however, because many of the exposures to these compounds come from what might be termed voluntary exposures, some risks can be minimized. Simply avoiding using any pesticides if at all possible, minimizing the use of plastic storage containers for foods and beverages and avoiding purchasing canned foods or those already in plastic will go a long way to reduce exposures to these compounds during critical times. While an awful lot of commercial interests might not like it, repetitious public health announcements expressing such sentiments might actually help to reduce behaviors that enhance risk for BC and PC in the same way a variety of antismoking campaigns have modestly reduced smoking behaviors in the United States and other countries.^{438–440}

From the preceding discussion, it is clear that there are many endogenous and exogenous sources of compounds that lead to DNA damage and to enhanced rates of cell division that in combination lead to elevated risks for cancer. There also are a wide variety of possible DNA alterations in addition to those already mentioned. As one can imagine, many of the activated lipid-peroxidation species mentioned in the preceding discussion (such as malondialdehyde and acrolein) also may react with each other, in addition to DNA, to form more complex adducts including a variety of DNA cross-links, each with their own mutagenic potential.^{348,354,360,441} The activated compounds that react with exocyclic amines of DNA also can react with exposed amino groups of any nearby protein, leading to inactivation of these proteins, as well creating a variety of inactivating protein:protein and protein:DNA cross-links. While the details of these forms of DNA and protein damage are beyond the scope

of this chapter, the potential carcinogenic effect is not. The DNA:protein cross-links will obviously have their own direct mutagenic potential because they are a form of DNA damage. In instances where the protein damage produces a loss of function of any of the enzymes that are associated with the regulation of signal transduction pathways, or cell cycle control, or those involved in phase II activities, or in protein or DNA synthesis and repair, then risks for mutations also may be enhanced.

The repair or replacement of damaged proteins and damaged DNA in proliferating cells ensures that risks for mutagenesis are minimized. When the processes of repair and synthesis are compromised and the damage persists, then the probability of producing mutations increases. Not coincidently, it is the same ROS, RNS, and other electrophilic oxidant species that cause damage to proteins and to DNA that also can alter activities of the various signal-transduction pathways to initiate (directly and indirectly) proinflammatory and proliferative signaling. Some of these same oxidants also are involved in enhancing synthesis and repair functions (through altering activity of the signal transduction pathways). Therefore, anything that can interfere with the tight integration of oxidant/antioxidant status with the regulation and integrity of signal transduction pathways also can contribute to risk for cancer. This leads us to our last major topic of cancer etiology.

6.2.6 NUTRITIONAL INSUFFICIENCIES

Nutritional insufficiencies comprise a risk factor for cancer that is common in the United States. For instance, intakes for vitamins C, D, and E are well below the EAR for 25%, 70%, and 60% of all Americans, respectively. Intakes also are below the EAR for folate and zinc (>10%); iron, copper, and selenium (\sim 5%–6%); and for calcium (38%) and magnesium (>45%), and each of these insufficient intakes are associated with increased DNA damage, mutagenesis, and risk for cancer.^{105,136,139,145,442-446} The implications of these insufficiencies are very important for cancer risk in the United States, especially when one considers the endemic nature of them; the majority of Americans consume insufficient amounts of more than one nutrient. As described in his triage theory of insufficiency-based risk, Dr. Ames hypothesizes that shortterm survival of an organism takes precedence over long-term survival in the face of micronutrient insufficiencies as an evolutionary trait, ensuring that the immediate nutrient requirements for current metabolic and cell function requirements (presumably, e.g., activities such as ATP generation, protein and membrane synthesis, and production of membrane gradients) are satisfied at the expense of longer-term functions such as DNA synthesis and repair.445,446 This theory is both attractive and practical in part because it puts into a biological and evolutionary perspective one of the realities of basic chemistry, that is, diffusion. As specific nutrients are being modified or conjugated to other macromolecules while being used, then the local concentration of the original form must decline. If the rate of use is high, then the local concentration will decline faster than in other locations of slower use, causing the nutrient to diffuse from the areas of higher concentration to those of the fastest use. This will of course compromise other functions at some other locations, unless optimal amounts of the nutrient are available. With such a high penetration of insufficient nutrient intake in the U.S. population, optimal nutrient availability for most is highly doubtful. And as a result, many cell functions will be compromised to some extent. From the preceding discussion, it is seen that a majority of Americans consume insufficient amounts of vitamin C, D, and E, with obvious implications for both ROS-mediated damage and inflammatory and proliferative signaling.

The antioxidant function of vitamins C and E is well known with vitamin C acting as a soluble antioxidant that can quench a variety of soluble ROS including superoxide anions and hydroxyl radicals while vitamin E acts as a lipid-soluble antioxidant to arrest lipid peroxidation reactions.^{447–451} Vitamin C also reduces vitamin E radicals back to the functional form while glutathione (GSH) reduces (directly or through glutaredoxin activity) the ascorbate radical or the dehydroascorbate forms back to ascorbate. GSH reductase (GSR) then uses NADPH as an electron source to reduce the glutathione disulfide (GSSG) back to GSH + GSH, while thioredoxin reductase (TrxR) can reduce oxidized ascorbate to the ascorbate form using NADPH as the electron source. The importance of these reactions is twofold, with the first relating to ROS-mediated damage. A reduction in antioxidant function due to an insufficient intake of vitamin C can lead to a reduction and DNA damage from ROS, malondial-dehyde, and other peroxidation products. The second is the effect this may have on activity of various signal transduction pathways.

ROS (and other oxidizing agents) can modify activity of signal transduction pathways by oxidizing redox sensor proteins such as thioredoxin (Trx) and peroxiredoxin (Prx). Trx is a component of a complex redox-regulation system that includes Trx, TrxR, GSH, GSR, GSH peroxidase (GPX), and Prx.450,452 These sensor proteins function as antioxidant enzymes through their peroxidase activities and also are important regulators of activity of the signal transduction pathways.^{168,450,453-458} For example, Trx forms inhibitory heterodimers with apoptosis signal-regulating kinase 1 (ASK1) and disassociation can occur when Trx is oxidized, leading to activation of the p38 and JNK-MAPK pathways. The transcription factors Jun/Fos (AP-1), heat shock factor 1, and p53 also can be indirectly activated by Trx through Trx-mediated oxidation of redox-factor 1 (Ref-1).454,458-462 Because GSH is an essential component of this redox system, a nutritional insufficiency that contributes to a nonoptimal antioxidant status can enhance use of GSH as an antioxidant while compromising its use as a cofactor in the redox system. This could then compromise redox-based control of the signal transduction pathways, inappropriately enhancing their activation. Selenium insufficiency (5%-6%) of the population) also contributes to this dysregulation because the selenoprotein TrxR is an important component of the Trx/ Prx redox-regulation system, explaining in part the common associations between insufficiencies of selenium and increased risk for cancer.

In addition to the antioxidant compounds mentioned, a variety of antioxidant enzymes also contribute to redox-based control of signal transduction pathways. The antioxidant enzymes SOD, catalase (CAT), and GPX (also a selenoenzyme) contribute to redox regulation in the cell through the inactivation of ROS, a function that also is commonly known to reduce ROS-induced damage to DNA, proteins, and phospholipids.^{168,344,345,450,453–460} Reducing ROS will attenuate ROS-mediated oxidation of Trx and therefore help to ensure "appropriate" signal transduction activities. Because of the heterogeneous distribution of these antioxidant enzymes,

the redox-sensor enzymes, and the various sources of ROS with differing rates of ROS production within cells, there also will be differing spatial distributions of ROS oxidants within the cell as well, producing an activation pattern of the various signal transduction pathways that is specific to the particular spatial distribution of ROS.^{458,459,463} As the redox environment within a cell changes due to changes in metabolic requirements, activities of the different signal transduction pathways also change and the resulting patterns of activation of various transcription factors are altered as well. It is through these temporal and spatial alterations in redox potential that alterations in protein synthesis are made to modify cell functions to better handle the various stressors. Other factors that can alter the internal redox state of a cell include responses to various growth factors and other signaling molecules, production of ROS from either endogenous or exogenous sources, or even exposures to exogenous toxins and xenobiotics. "Flooding" cells with damaging ROS or toxins could interfere with the delicate regulatory balance between activity of the antioxidant (and redox) enzymes and the activity of the signal transduction pathways, resulting in overactivating or underactivating the pathways. On the other hand, excessive amounts of soluble antioxidants could disrupt the redox environment to inappropriately alter regulation as well. Both situations could obviously alter overall cell function, and the direct activation of ERK-MAPK and p38-MAPK pathways by cigarette smoke extracts resulting in proinflammatory signaling (independent of damage or mutagenic effects) is indicative of the former.^{464,465} The latter concept is illustrated by the prevention of the mitochondrial biogenesis and enhanced insulin sensitivity that is normally stimulated by exercise through supplementation with vitamin C(1 g) and vitamin E (400 IU).⁴⁶⁶ Maintaining tight control of the signaling processes within the cell therefore depends on the localized concentration of the various antioxidant and redox enzymes and their cofactors. Because iron, copper, zinc, and selenium are components of one or more of all of the antioxidant and redox enzymes, the prevailing insufficient intake (5%-10%) of the population) of these minerals is certain to be a factor in preventing optimal control of the various signal transduction pathways and thereby enhance risk for cancer.

In addition to the direct effects of ROS on the various signaling pathways, they also can affect them indirectly. PGE_2 is an important activator of proliferative signaling in tissue SCs and PCs and cPLA₂ can be activated by ROS, leading to enhanced production of PGE_2 .^{467–471} This is one more avenue for enhanced proliferative signaling to occur as a result of enhanced ROS stress. It also is one that would be enhanced in the event of nonoptimal antioxidant status. Thus, from all of the preceding discussion, nutritional insufficiencies in antioxidant-associated nutrients and cofactors exacerbate the risks associated with exposures to exogenous and endogenous ROS, mutagens, and mitogens.

The last issue to discuss in regard to nutrient insufficiencies is that of vitamin D, calcium, and magnesium. These represent the dietary (intake) insufficiencies with the greatest prevalence for both vitamins and minerals, 70%, 38%, and >45%, respectively. The importance of vitamin D and magnesium for maintaining calcium status is well known. When levels of calcium in the serum decline, parathyroid hormone (PTH) is released, stimulating synthesis of the enzyme 25-hydroxycholecalciferol-1-hydroxylase in the kidney and ultimately enhancing

intestinal calcium and phosphorus absorption.^{472–476} Because magnesium is required for both the release of PTH from the parathyroid gland and binding of PTH to its receptor, a magnesium insufficiency can have detrimental effects on both calcium regulation and absorption and is known to be a major factor in risk for osteoporosis. Lower calcitriol levels also will occur because of attenuated PTH effects, an effect that also is observed with insufficient intake of magnesium.

Inadequate vitamin D is commonly known to be associated somewhat with increased risk for cancer; the associations have not consistently been observed in all studies.477-480 The risk-reduction effects of vitamin D have been proposed to be mediated in part through an increased expression of p21 and p27 that leads to cell cycle arrest and by enhancing differentiation through modulating JNK-MAPK activities, effects that have been observed in cancer cells in culture. Calcitriol also inhibits the production of proinflammatory prostaglandins and cytokines by suppressing COX2 expression and NF $\kappa\beta$ signaling in both cancer and noncancer cells. These latter effects would be important in prevention and illustrate how nutritional deficits (vitamin D) might enhance risk factors (inflammation) that appear to be far removed from the commonly known function of the nutrient (promoting calcium absorption). In the one double-blind, placebo-controlled clinical trial for prevention in postmenopausal women that used calcium and vitamin D supplements (individually or in combination), relative risks of less than 0.4 were observed for all cancers in each of the supplemented groups in comparison to nonsupplemented.⁴⁷⁹ Although these results support the associations between low vitamin D and enhanced risk for cancer, the study was of relatively short duration (4 years) and may not necessarily apply to the "life-time-to-date" associations often tested for in many epidemiology studies. These results also support the concept that not knowing the nutritional status of subjects in epidemiological studies can limit the interpretation of the results; calcium, vitamin D, and magnesium insufficiencies (and any others for that matter) may be a serious confounder for interpreting risks in such studies because they exist in a large majority of subjects.

This brings us to the various avenues that are amenable to prevention. Minimizing exposure as much as possible (or practical) to potentially carcinogenic compounds is clearly a very important behavior. While moving or changing jobs are not often convenient strategies to avoid exogenous mutagens, reducing the voluntary exposures to them certainly is. The use of tobacco products and self-exposure to second-hand smoke are voluntary behaviors that should be avoided or at the very least minimized. Preparing and eating smoked meats also can be limited and food storage conditions can be altered to minimize contamination. Risks that arise from endogenous mechanisms also can be attenuated, although they are certainly less amendable to complete prevention. The major endogenous factors that are best associated with prevention include the metabolic production and inactivation of ROS and RNS, activation and inactivation of electrophiles from xenobiotics, and the repair of the damage to DNA and proteins that is caused by the various ROS, RNS, oxidants, and electrophiles. In addition to these, the various signaling pathways that are associated with cell cycle regulation also are important targets, with the primary prevention effort being directed at correcting nutrient deficiencies and insufficiencies. If this primary objective can be met through optimal dietary habits, then many of the other prevention targets that relate to signaling processes, such as those resulting in enhanced phase II activities, will be addressed at the same time.

6.3 ETIOLOGY OF PREVENTION

As described in Section 6.1.1, there is considerable evidence from both epidemiologic and mechanistic-based researches that a variety of factors associated with eating a proper diet and exercising regularly may significantly reduce risk for many cancers. Perhaps, this should be restated from a slightly different perspective: it is the variety of nutritional insufficiencies, coupled with an inadequate consumption of plant-based foods, and a lack of sufficient physical exercise that profoundly increase the risks for cancer. The difference in phrasing is philosophical; cancer should not be considered "normal" but rather it should be considered an unfortunate consequence of our poor lifestyle choices. In the context of our cellular and molecularbased approach to disease etiology and prevention, exposures to various carcinogens can lead to a variety of damaging events that result in cellular-stress responses and dysfunctions. And, continuing exposures to these damaging agents can lead to the development of a cancer. Unfortunately, many in the United States already have subclinical cellular dysfunctions due to an inadequate consumption of different essential nutrients, a situation that leads to greater signaling through cellular stress-response pathways than would otherwise occur, with subsequent enhanced proinflammatory and proliferative effects. When these nutritional insufficiencies are combined with prolonged exposures to multiple carcinogens from a variety of sources, the result is a perfect recipe for enhancing risk for cancer. They also are the starting point for prevention.

6.3.1 NUTRITION AND DIET

Logic dictates that eliminating nutritional insufficiencies should reduce risks for cancer in part by ensuring optimal antioxidant status and redox regulation of signal transduction pathways. Proper nutrition also should help through maintaining adequate availability of nutrients for the purposes of synthesis and repair, and it should be a relatively simple thing to do. After all, all one has to do is eat a proper diet, such as that described in the MyPlate (www.choosemyplate.gov) recommendations. The difficulty with this concept, as explained in Chapter 8, is that people eat for many reasons and the habits of eating that are learned and ingrained over a lifetime are extremely difficult to break. The simplest way to deal with part of this problem is to use dietary supplements, and many attempts to reduce risk for cancer through the use of dietary supplements have been made, albeit with varying degrees of success.

In many of the prevention trials, supplements that contain antioxidant-related compounds such as β -carotene, vitamins E and C, and selenium have been tried.^{481–489} The use of antioxidant supplements has often been based on the negative associations observed between cancer incidence and fruit and vegetable intake in epidemiological studies and on the assumption that the protective effect was apparently due mainly to the antioxidant and β -carotene content of these foods. Although decreased

incidences of gastric cancer have been observed in China with vitamin E, selenium, plus β -carotene supplements; supplementation with vitamin C and molybdenum, retinol and zinc, vitamin E alone, vitamin C alone, or with multivitamins, in both China and the United States, have had little to no positive effect. Similar inconsistent results are seen with PC as well as with many other cancers, and increased risks for cancer, while certainly unintended, have been observed with some supplements. In the cases of PC (SELECT), supplementation with vitamin E, selenium, or both significantly enhanced risk for cancer. In both the ATBC and CARET studies, supplements with antioxidant vitamins had little to no effect. In several other recent longer-term (>10 years) clinical trials, the use of multivitamin/multimineral supplements (or multi-antioxidant supplements) produced only very small nonsignificant preventive effects, or no preventive effects at all, while the evidence for their use in epidemiological surveys is also weak to nonexistent.^{487,490–496}

With a focus on mechanisms of prevention for this section, a comprehensive review of these types of studies is not the point. These few studies do, however, illustrate several important issues. In the smoking-lung cancer studies, the enhanced risk was associated with heavy alcohol consumption. Although the enhanced carcinogenic effect of β -carotene in conjunction with heavy drinking in smokers was one that could not be predicted on the basis of the epidemiological evidence at that time, there was earlier evidence from animal and human studies that chronic ethanol consumption may significantly alter β -carotene and retinoic acid metabolism to enhance vitamin A toxicity.^{497,498} Perhaps, if such information were more widely known, it might have been incorporated into concepts of risk and the potential for risks when using supplements in special populations. One more issue of unintended consequences is that of antioxidants interfering with normal cell function. While antioxidant supplements have often been thought to be beneficial as a prophylactic to reduce ROS-mediated damage, increases in mitochondrial function and in insulin sensitivity that are normally caused by exercise and that are partially responsible for the preventive effects of exercise can be prevented by supplementation with vitamin C (1 g) and vitamin E (400 IU) in amounts that are often used in supplements.⁴⁶⁶ Whether this phenomenon contributes to the lack of beneficial effect of using supplements in many studies is unknown; however, with insulin resistance being a source of risk for carcinogenesis, it should not be overlooked.

In relation to the special populations' issue, the preventive effect of the antioxidant compounds on gastric cancer appeared to occur predominantly in populations with a relatively high incidence of chronic undernutrition and malnutrition. In hindsight, one would expect that providing nutrient supplements to a predominantly malnourished population to correct nutrient deficiencies might have a much greater impact on prevention than on people with marginally insufficient diets and no overt deficiencies. And this may be especially true when the population with the much poorer diet also is exposed to greater amounts of carcinogens, aflatoxins in the case of the referenced studies on gastric cancer conducted in China and the United States. Although nutrient insufficiencies can enhance risk for cancer by enhancing mechanisms of cause, nutrient-based supplements may only be a major contributing factor for prevention in the face of chronic undernutrition or outright malnutrition. Another issue may be that nutrient supplements simply cannot overcome the detrimental risks associated with moderate exposures to carcinogens when they are coupled with behaviors that enhance risks, behaviors such as consuming too much alcohol, smoking (or using) tobacco products, eating too many calories, eating too few fruits and vegetables, and not participating in any physical activity. The first few of these behaviors have been addressed earlier and reducing risks for a variety of cancers includes altering behaviors to maintain weight, to avoid using tobacco products or exposures to tobacco smoke, and to minimize the use of alcohol. These preventive activities have one thing in common; they all reduce exposures to endogenous and exogenous sources of carcinogens and proliferative signaling but do very little to *enhance* protection. The latter two, however, deserve a detailed discussion because increasing fruit and vegetable consumption (and participating in appropriate physical activity) can enhance a variety of mechanisms of "protection" to reduce risk for cancer.

Because of the known inverse association between consumption of plant-based foods and cancer, it stands to reason that increasing consumption of a variety of fruits and vegetables confers risk-reduction benefits. And from the preceding, it apparently is not because of any one specific nutrient or compound with antioxidant properties. Rather, it is far more likely that it is an optimal nutrient status gained from consuming such a diet combined with the consumption of an array of bioactive non-nutrient compounds called phytochemicals that protects against cancer.^{148,410,499–506}

The major forms of phytochemicals include the various phenolics (flavonoids, phenolic acids, stilbenes, coumarins, and tannins), carotenoids (α -carotene, β -carotene, lutein, lycopene, zeaxanthin, and astaxanthin), and organosulfur compounds (isothiocyanates, indoles, and allelic sulfur compounds), many of which are weak to moderate antioxidants.⁴¹⁰ The highest concentrations of phytochemicals are found in herbs, spices, and dried fruits (-1%-5% by weight) with somewhat lower amounts in fresh fruits, nuts, grains, vegetables, and oils (0.05%–1% by weight).⁵⁰⁷ Common foods and their phenolic contents are listed in Chapter 2, Section 2.5.2.2 and won't be listed here. It is important to note that many of the foods that are high in phytochemical content also are consumed much more frequently in a traditional Mediterranean diet than in the American diet and it is consumption of these foods that is associated with the reduced incidence of cancer (and other chronic diseases) in the Mediterranean cultures in comparison to the United States. 49,140,150,508-510 The mechanisms of prevention that are enhanced by phytochemicals include reducing proinflammatory signaling, attenuating the production of mutagenic electrophiles, enhancing the elimination of mutagenic compounds, enhancing antioxidant control through increasing expression of antioxidant and redox enzymes, and increasing DNA and protein repair.

6.3.2 INFLAMMATORY SIGNALING

As described in Section 6.2.4, inflammatory signaling by infiltrating stromal cells is an important factor in the development of tumors, stromal cells that include macrophages, lymphocytes, NK cells, mast cells, endothelial cells, and fibroblasts. Each of these cells can produce an array of proinflammatory prostaglandins and cytokines (for details please see Chapter 2) in response to the activation of pattern recognition receptors (PRRs) by pathogen-associated molecular patterns and damage-associated molecular patterns, and through paracrine and autocrine activation by the same prostaglandins and cytokines.^{511–517} The receptor-mediated activation process for production of the proinflammatory cytokines involves activation of the NF $\kappa\beta$, p38-MAPK, and JNK-MAPK pathways, which in turn activate the various transcription factors that then induce expression of IL-6, pro-IL-1 β , TNF α , and IFN- β .^{512,514,518,519} And because both elevated calcium and ROS can activate these same signal transduction pathways, any metabolic stress and cellular damage also will lead to the activation of proinflammatory signaling as well.^{520–523}

One means to attenuate inflammatory signaling is to attenuate activation of the NF $\kappa\beta$, p38-MAPK, and JNK-MAPK pathways; this will in turn reduce the expression of the proinflammatory cytokines. And in a variety of animal studies as well as cell culture studies, a variety of flavonoids (quercetin, apigenin, epigallocatechin gallate (EGCG), theaflavin, genistein, anthocyanins, and kaempferol), phenolic acids (ellagic acid, capsaicin, and curcumin), and the stilbenoid resveratrol significantly inhibit activity of one or more of these pathways.^{147,508,524–534} And where tested, in many of these same studies, a significantly reduced expression of IL-6, IL-1 β , TNF α , IFN, ICAM, and MCP-1 have been observed as well, indicating that consumption of these phytochemicals can reduce proinflammatory signaling.

The mechanisms through which they reduce proinflammatory signaling are diverse. Sulforaphane can inhibit PRR signaling by directly attenuating TLR4 activation while curcumin inhibits both TLR4 and NOD1/2. Resveratrol, EGCG, quercetin, and luteolin inhibit TBK1 (MyD88-independent pathway) and resveratrol also inhibits TRIFF (MyD88 activation pathway), all of which attenuate the NF $\kappa\beta$, p38-MAPK, and JNK-MAPK signal transduction pathways.⁵²⁸ As described in detail in Chapter 2, PRR-mediated activation of these pathways is essential for both damage- and infection-mediated inflammatory responses. JNK-MAPK and p38-MAPK activate a variety of transcription factors (including cJun, ATF2, IRF3, IRF5, IRF7, JunB, JunD, and I $\kappa\beta\zeta$) that dimerize in various combinations and, along with NF $\kappa\beta$ (p50/p65 dimer), induce expression of the proinflammatory cytokines. The ability of curcumin, resveratrol, sulforaphane, EGCG, and luteolin to attenuate PRR activation and subsequent proinflammatory signaling therefore is an important factor in reducing inflammation-associated risks for cancer.

In addition to blocking PRR activation, NF $\kappa\beta$ signaling also can be attenuated through inhibiting IKK activity. IKK phosphorylates I $\kappa\beta$, which is normally complexed to NF $\kappa\beta$ in the cytosol. On phosphorylation, NF $\kappa\beta$ is released and it then can enter the nucleus and bind to its promoter. A variety of theaflavins (including EGCG) directly inhibit IKK while quercetin, curcumin, kaempferol, apigenin, morin, anthocyanins, procyanidins, and lycopene suppress either nuclear translocation of NF $\kappa\beta$ or binding of NF $\kappa\beta$ to DNA, attenuating expression of the proinflammatory cytokines.^{524,525,532,534}

Activity of the MAPK pathways also is attenuated by a variety of phenolics including apigenin, luteolin, quercetin, resveratrol, EGCG, and kaempferol. These suppress ERK1/2-MAPK, JNK1/2-MAPK, and p38-MAPK pathways while cyanidins can attenuate ERK activation and curcumin can inhibit PKC, a calcium-activated enzyme that activates various components of the MAPK pathways.^{524,525,532,534}

These effects also lead to a suppression of proinflammatory signaling. Different cell types, however, respond differently to phenolics; for example, luteolin attenuates all MAPK pathways in respiratory epithelial cells but not the JNK-MAPK in macrophages.^{524,525,532,534–536} EGCG also is known to have different effects on different lineages of the same cell type with inhibition of IKK occurring in some keratinocytes and blocking of p38 activation in others (HaCa cells).

Because both ROS and increased cytosolic calcium also can activate these same MAPK pathways to produce a metabolic stress-mediated proinflammatory response, inhibiting these effects is an important preventive measure. The inhibition of PKC by curcumin has already been noted, an effect that also occurs with quercetin and luteolin and contributes to the dampening of the various MAPK pathways.⁵³⁷ As discussed in Chapter 2 (section 2.5.2.2), the antioxidant effects of individual phytochemicals tend to be relatively weak, and even at the highest recorded levels, consumption of any individual compound from specific foods rarely approaches within 10%–20% of that necessary for achieving effective antioxidant effects. However, when mixtures of foods (with very different phytochemical contents) are consumed, then antioxidant effects are observed at a total phytochemical dose that is up to five times less than the dose necessary when a single food source of phytochemicals is eaten.^{410,463,538} With half-lives in the 1–2-hour range for most phytochemicals, the antioxidant effects are transient at best, but may still have a dampening effect on ROS-mediated activation. From a wide array of studies, it is clear that different components of the MAPK and NFkß signaling pathways are attenuated by different phytochemicals, as are the different activating mechanisms (PRRs, ROS, and PKC), and in some respects in a cell-specific manner. Because of this, to have an overall preventive effect through suppressing cytokine-mediated inflammatory signaling, an array of phytochemicals would need to be consumed.

In addition to the cytokines, the eicosanoids PGE₂, LTB₄, and TXA₂ also have potent proinflammatory effects and they are usually the first ones produced in response to any type of damage or stress. These molecules are produced in almost all cells as a direct response to the calcium-, ROS-, or MAPK-mediated activation of PLA₂ along with the coordinated activities of cyclooxygenases, lipoxygenases, and the various eicosanoid synthases.^{467–471,520,539,540} These eicosanoids are responsible for initiating proinflammatory responses; responses that include the chemoattraction of macrophages, neutrophils, and other inflammatory cells into the area as well as their activation. These effects are in addition to activating the proliferation of any SCs or PCs by PGE₂ in the immediate area. Once activated by eicosanoids, the inflammatory cells produce the inflammatory cytokines that then enhance the attraction and activation of additional inflammatory responses within a tumor environment can be an important contributor to the carcinogenesis process.

Inhibiting cyclooxygenase (COX) and lipoxygenase (LOX) enzymes would reduce synthesis of eicosanoids, and the phytochemicals EGCG, quercetin, and kaempferol have been observed to directly inhibit either COX or LOX activity to attenuate production of PGE₂, LTB₄, and TXA₂, as do tyrosol, hydroxytyrosol, apigenin, luteolin, and oleuropein from extra virgin olive oil.^{525,541,542} And, similar to their effects on MAPK and NF $\kappa\beta$ signaling, single phenolics rarely attenuate activity of both enzymes in all cells, necessitating the consumption of a variety of phytochemicals to obtain an optimal preventive effect. Interestingly, the anti-inflammatory effects of extra virgin olive oil are observed in clinical trials with a daily consumption of as little as 25 g/day (less than 12% of the average daily caloric requirements for adult males), an easily attainable amount if extra-virgin olive oil (EVOO) is used for most culinary needs. These anti-inflammatory effects also are observed in conjunction with the Mediterranean diets where the average consumption of a wide array of phenolics (including those from EVOO) is greater than 1 g/day (~1.3 g/day for females and ~ 1.5 g/day for men), but not observed with the typical U.S. diet that has an intake of phenolics that averages less than 450 mg/day.^{49,140,150,508-510,543,544} Perhaps, the three times greater intake of phenolics common to the traditional Mediterranean diet (in comparison to the U.S. diet) represents some form of a threshold effect, maybe it takes more than 1 g/day of a mixture of phenolics that are obtained from whole foods to produce a reduction in synthesis of proinflammatory eicosanoids. On the other hand, it is possible that consuming an insufficient amount of the essential fatty acid α -linoleic acid (ALA) in combination with a low consumption of eicosapentaenoic acid (EPA) and docosahexaenoic acid (DHA) (from marine sources) contributes to the relatively proinflammatory effect of the "typical" U.S. diet.

Both EPA and DHA, commonly found in cold-water fish oils, are known to have anti-inflammatory properties. EPA is synthesized from the essential fatty acid ALA, with conversion rates of approximately 1%-8% in men and 8%-21% in women; DHA is subsequently synthesized from EPA with less than 1% of dietary ALA being converted to DHA for both.545,546 Less well known are the anti-inflammatory effects of AA, which is synthesized from the essential fatty acid linoleic acid. As reviewed in detail in Chapter 2, lipoxins A_4 and B_4 are synthesized from AA, resolvins D1–4 and protectin D1 are synthesized from DHA, and resolvin E1 is synthesized from EPA. These anti-inflammatory eicosanoids are produced through the coordinated activities of COX2 and 5-, 12-, and 15-LOX and are responsible for initiating and maintaining the resolution phase of inflammation. Lipoxins produce their pro-resolution effects through activating G-protein-coupled receptors on neutrophils, macrophages, endothelial cells, and monocytes that then result in a suppression of p38-MAPK, p42/44-MAPK (ERK 1/2), PI3K activities and inhibiting the activation of NF $\kappa\beta$, producing potent anti-inflammatory effects. Resolvins also have been observed to inhibit activation of NF $\kappa\beta$ and both resolvins and protectins greatly enhance the phagocytosis of neutrophils by macrophages without the concomitant production of ROS (thus reducing inflammation-associated ROS-mediated damage) and to inhibit the migration of neutrophils into tissues. In addition to these direct anti-inflammatory effects, lipoxins enhance the release of IL-10 from macrophages, which also suppresses synthesis of the proinflammatory cytokines through activating expression of the suppressor of cytokine synthesis (SOCS) proteins. While the resolution effects of these eicosanoids are essential for shutting down inflammation responses following pathogen and sterile damage-associated stress, their role in diminishing the pro-carcinogenic effects of inflammatory signaling in a tumor environment is largely unknown. The possibility exists, however, that with ~30% of the U.S. population not meeting the AI for ALA, an insufficient availability of endogenously formed EPA and DHA might contribute to an enhanced risk for cancer. This last concept is a
possibility already proposed for CHD based on a significant decline in risk for CHD in Eastern European populations that change from using predominantly low-ALA oils to high-ALA oils for cooking.⁵⁴⁷

EPA and DHA also may exert some of their resolution effects through direct activation of receptors in addition to being precursors for lipoxins and resolvins. For example, both EPA and DHA are peroxisome proliferator-activated receptor gamma (PPARy) agonists, and activation of this receptor is known to inhibit p38-MAPK, ERK-MAPK, and the NFκβ pathways, resulting in decreased inflammatory signaling in a variety of inflammatory cells.⁵⁴⁸⁻⁵⁵² This direct anti-inflammatory effect is important because it indicates that increasing consumption of exogenous EPA and DHA not only could correct an insufficiency produced through inadequate consumption of ALA, but it also may provide a pharmacological effect. What we mean by this is that with an adequate consumption of ALA, existing synthesis activities will produce sufficient amounts of EPA and DHA to ensure normal proresolving functions and therefore reduce risks for disease that were produced from an inadequate consumption of ALA. Synthesis of EPA, DHA, and the various lipoxins, resolvins, and protectins is, however, highly regulated and it is doubtful that simply increasing EPA and DHA as substrates would significantly enhance rates of synthesis of resolvins and protectins beyond that which would exist in cases of an adequate intake of ALA. However, due to their role as direct acting PPARy agonists, these PPARy-mediated anti-inflammatory effects would be in addition to their normal resolution effects mediated through the regulated synthesis of resolvins and protections. If this is the case, then increasing EPA/DHA supply beyond that obtained from adequate ALA consumption could produce enhanced anti-inflammatory effects. From an array of studies, there appear to be consistent protective effects of consuming EPA and DHA in the 250-500 mg/day range without consistently observing (systemic) antiinflammatory effects while at higher intakes in the 2–3 g/day range a reduction in systemic markers of inflammation such as C-reactive protein or IL-6 are consistently observed.553-559 In addition, PPARy agonists are known to inhibit platelet aggregation, and increased blood-clotting times are observed only at the higher doses of EPA and DHA.^{560–563} These general observations would be consistent with such a concept; consuming lower amounts of EPA/DHA essentially corrects an inadequate consumption of ALA while consuming higher amounts produces an additional pharmacological effect through direct activation of PPARy receptors.

From the preceding discussion, it is clear that consuming mixtures of different phytochemicals that are obtained from a wide variety of fruits, vegetables, and EVOO as well as EPA and DHA from fish oils can reduce proinflammatory signaling by attenuating the synthesis of both proinflammatory cytokines and eicosanoids. These effects are mediated through dampening the activation of the p38-MAPK, ERK-MAPK, and NF $\kappa\beta$ pathways, inhibiting both COX and LOX activities and activating PPAR γ receptors. The fact that only partial inhibition of these pathways is achieved from these dietary components is very important. Complete blocking of these pathways would prevent any inflammatory responses from happening and that obviously could ultimately result in severe damage and death. By dampening the inflammatory responses, the essential inflammatory responses are preserved while at the same time the risks for excessive or chronic responses are reduced. These

anti-inflammatory effects also appear to occur with a total consumption of phenolics that is well above 1 g/day, although an exact threshold for this effect is unknown. Even without the knowledge of an exact threshold level, it is clear that consuming effective amounts of phytochemicals within the context of a predominantly vegetarian diet is not only practical but also common to some traditional cultures. In addition to their effects on inflammatory signaling, mixtures of phytochemicals also can have transient antioxidant effects that might contribute to reducing ROS-associated risks for mutations. And as with the dampening effects on inflammatory signaling, the transient nature of the antioxidant effects also may be a desirable characteristic. As described in Section 6.3.1, antioxidant supplements (even in typical OTC doses) that appear to produce a prolonged excessive antioxidant effect can have detrimental effects, presumably through interfering with the normal regulation of redox control and resulting in detrimental effects on activities of the various signal transduction pathways. For optimal cellular antioxidant function, it may be more desirable to enhance activities of the various antioxidant and redox enzymes to maintain appropriate regulatory control while at the same time increasing the ability to deal with higher levels of ROS and other oxidants.

6.3.3 ANTIOXIDANT AND REDOX CONTROL

Enhancing antioxidant enzyme activity should provide two benefits in regard to reducing risk for cancer. The first is through reducing the amount of ROS-mediated damage within cells. As described in Section 6.2.5, a variety of RNS and ROS (whether from intracellular or extracellular sources) can cause damage to DNA, cellular phospholipids, and proteins increasing risk for the production of an array of mutations in subsequent generations of proliferating cells. In addition to the mutagenic potential of ROS and the products of ROS reactions, they also can modify activity of various signal transduction pathways. Chronic activation of these pathways, especially the various MAPKs, by excessive levels of ROS can lead to enhanced proliferative responses. This occurs directly in proliferating cells and indirectly through activating proinflammatory signaling, which then enhances proliferative responses. Activated inflammatory cells also produce ROS and RNS themselves, which in turn add to the potential for damage and detrimental signaling. The integration of the effects of ROS/RNS-mediated damage and stress responses in proliferating cells, with the same effects resulting from proinflammatory signaling, would be especially important in the tumor environment. And because of the known detrimental effects of supplementing otherwise healthy (and presumably nutritionally replete) athletes with antioxidants, 466,564-567 enhancing expression of antioxidant enzymes and redoxcontrol enzymes may be the more prudent approach to reducing ROS-mediated risks for cancer. Increasing expression of antioxidant enzymes would enhance the ability to reduce damage caused by excessive entry or production of ROS/RNS that may occur as a result of inflammation or transient exposures to metabolic or chemical stress. By enhancing expression of redox-control enzymes, overactivation of signal transduction pathways by oxidants also may be minimized.

As discussed earlier in Section 6.2.6, the redox sensor proteins Trx and Prx are integral components of the redox sensing system, which are important regulators of

activity of various enzymes throughout the cell, including those of signal transduction pathways. As described, the inhibitory complex formed between Trx and ASK1 disassociates on oxidation of Trx, leading to the activation of various MAPK kinases (MAPKKs) by ASK1 and ultimately to that of p38 and JNK.^{450,452,458,459,568} The consequences of JNK/p38 activation on cell proliferation and inflammatory signaling activities have already been discussed. By increasing the activity of the various antioxidant enzymes and antioxidant compounds in a cell, excessive or inappropriate activation of proliferative or inflammatory signaling might be minimized.

Trx also mediates the activation of Ref-1 through its reduction in the nucleus, which also may be important in terms of risk for tumorigenesis. Ref-1 was first recognized as being the mammalian AP-endonuclease (base excision repair of apurinic and apyrimidinic sites from ROS damage) and subsequently as a direct activator of the Jus/Fos heterodimer AP-1, HIF-1, and p53 and as an enhancer for p21 expression.^{461,462,569} These latter two activities of Ref-1 are indicative of one possible mechanism for reducing risk for tumorigenesis from ROS and ROS-mediated DNA damage. As discussed in Section 6.2.3, activation of ATM, DPK, and ATR by the DNA-damage sensor proteins initiates a series of events that lead to cell cycle arrest and the activation of p53 and DNA repair. Through the reduction of Ref-1 by Trx, these same events will occur, increased DNA repair by Ref-1 and increased cell cycle arrest through enhanced p53 activation and p21 expression. This leads to the concept that increasing content of reduced Trx also may increase the amount of reduced Ref-1 and thereby reduce risk for tumorigenesis by increasing DNA repair and increasing the time available in the cell cycle for repair. Sensitivity to apoptosis also may be enhanced through the increased activation of p53, another mechanism through which risk for tumorigenesis may be reduced. While these functions may have possible preventive effects, greatly enhanced Trx content has been observed in several cancer cell lines and many cancers, and high levels of Trx are associated with high risk for metastatic progression of cancers.⁴⁶² This association is likely due in part to the activation of HIF-1 by Ref-1. HIF-1 is a transcription factor that initiates expression of a variety of proteins necessary for maintaining cellular metabolism in a hypoxic environment (PDK, GLUT1/3), for angiogenesis (VEGF, IGF, and TGF), and for enhanced protein synthesis and growth (AP-1). The possibility exists, however, that these effects that comprise some of the hallmark functions of a tumor result from an array of driver mutations that lead to in part a greatly increased expression of Trx. This would not necessarily preclude the preventive role that an enhanced expression of Trx might play. When Trx expression is induced, along with that of an array of other protective proteins that function as an integrated system of redox sensors and antioxidants, the total effect should be protective, as opposed to the risk-enhancing effects of a greatly increased Trx content alone. If indeed this is the case, it indicates the potential difficulty in making global conclusions of risk on the basis of the functions of a single protein in transformed cells.

Trx functions within a complex system of cellular redox enzymes that includes Trx, TrxR, GSH, GSH, GPX, and Prx with the antioxidant enzymes SOD and CAT also being important in regulating the overall cellular redox state.^{450,452,462,570–573} GPX utilizes two GSH as a source of electrons to reduce H_2O_2 to water while GSR reduces

the resulting GSSG back to two GSH using NADPH as a source of electrons. The antioxidant activities of both Trx and Prx work in a similar manner with the SH groups of these peroxidases providing the electrons to reduce the H₂O₂ to water. To regain their reduced state, TrxR reduces Trx using electrons from NADPH while (reduced) Trx can directly reduce Prx. Similar to Trx, Prx also forms dimers with a variety of proteins to form inhibitory complexes, including c-Myc, macrophage migration inhibitory factor (inhibits migration of macrophages away from areas of inflammation, a necessary component of the resolution phase of inflammation), and p66 in mitochondria. The involvement of c-Myc in enhanced Cdk synthesis for entering cell cycle was mentioned in Section 6.2.2, and p66 is involved in promoting mitochondria-mediated apoptosis following its phosphorylation by JNK. Prx also can bind with ASK1 to attenuate activation of p38/JNK-MAPK in a manner similar to Trx. While Prx is involved in many other signaling pathways, it is evident that at least in the case of p66, Prx can attenuate signaling that leads to apoptosis. An overabundance of Trx that might maintain Prx in a reduced state may ensure a sustained antiapoptotic effect that might lead to increased risk for cancer. An overabundance of Prx might do the same, and as discussed in the most recent reviews, 427,535,538 an overexpression of either one alone significantly enhances a variety of functions that increase risk for cancer, including inhibited mitochondria-mediated apoptosis. The various functions that lead to enhanced risk will not be discussed here because the point of this is to relate that the uncoordinated increase in expression of either Trx or Prx can lead to dysregulation of cellular functions that then increase risk for cancer, and an increased expression of both (individually) has been observed in many cancers. On the other hand, a coordinated increase in both might ensure that a balanced regulatory function of these redox sensors can be maintained, while at the same time an increase in antioxidant protection would be realized. And, such a coordinated induction is possible because Trx and Prx as well as mitochondrial SOD (Sod2), CAT, GSR, glutamate cysteine ligase (rate limiting for GSH synthesis), GST, heme oxygenase-1 (HO-1), UDP-GT, and NADPH:quinone oxidoreductase-1 (NQO1) are all induced through binding of the Nrf2:Mah heterodimer transcription factor to the antioxidant response element (ARE) (aka: electrophilic response element [ERE]) of their promoter regions.411,531,574,575

The Nrf2:Keap1 complex is a cytosolic sensor protein in the cytosol that responds to a wide array of oxidants and electrophilic compounds. Phosphorylation of Nrf2 by protein kinases, oxidation of SH groups by H_2O_2 or other oxidants, and covalent modification of SH groups through a variety of mechanisms can disassociate the Nrf2:Keap1 complex and lead to binding of Nrf2 to the ARE transactivation binding sites.^{531,576–578} The ability for H_2O_2 to disassociate the complex is a major factor in how a transient increase in ROS can lead to enhanced antioxidant control through the induction of the various antioxidant enzymes and redox sensor proteins. And a wide array of phytochemicals can contribute to all three mechanisms of Nrf2 activation, revealing an important component of dietary-based prevention.

Expression of GSH, GSR, HO-1, Sod2, CAT, GCL, Trx, Prx, and UDPGT are all induced by a variety of phytochemicals including quercetin, resveratrol, diallyl sulfide, s-allyl cysteine, lycopene, EGCG, curcumin, and sulforaphane.^{408,433,579–590} Although the step-by-step mechanisms through which each of the phytochemicals

work to activate Nrf2:ARE binding have not been completely worked out, from several recent reviews there is good evidence for several possibilities.^{411,530,576,578,591} The hydroxyl groups on the catechol polyphenols (including flavonoids and phenolic acids such as EGCG, quercetin, catechin, cyanidin, and caffeic acid) are susceptible to autoxidation, which leads to the formation of a quinone. The quinone can then react with and bind to susceptible SH groups of the Keap1 to disassociate the complex. Other phytochemicals that have a Michael acceptor center such the α , β -unsaturated ketone of curcumin and the caffeic acid phenethyl ester, or an isothiocyanate group such as phenethyl isothiocyanate, benzyl isothiocyanate, and sulforaphane also are known to form adducts with Keap1 because of the electrophilic nature of these reactive centers. The formation of quinones through autoxidation of the catechol polyphenols also can lead to the formation of H₂O₂ because they participate in a redox resonance reaction where the quinone cycles between being reduced and then oxidized again, producing H_2O_2 at each redox cycle. The hydrogen peroxide generated by this process could directly oxidize the SH groups to force the disassociation. Another possibility would involve the variety of protein kinases that are known to phosphorylate Nrf2, including p38-MAPK, JNK-MAPK, ERK-MAPK, and PKC. Oxidation of the redox-sensor Trx by H₂O₂ leads to the disassociation of the Trx:ASK1 complex and ASK1 can then phosphorylate MEKKs, which subsequently activate p38- and JNK-MAPKs. Generation of H₂O₂ also is associated with activating a variety of forms of PKC through phosphorylation of tyrosine, again, indicating the possibility that redox cycling of catechol polyphenols can contribute to phosphorylation of Nrf2. PKC also can contribute to this process by phosphorylating Raf, which subsequently activates MEK1/2 and then ERK-MAPK.

From a variety of studies reviewed, 392,543 an array of phytochemicals activate Nrf2 through the different mechanisms: resveratrol (grapes and other fruits) through Nrf2 phosphorylation; capsaicin (hot peppers) through Nrf2 phosphorylation following PI3K/AKT activation; diallyl sulfide (garlic) through Nrf2 phosphorylation by p38-MAPK and ERK-MAPK; EGCG (green tea) through Nrf2 phosphorylation by AKT, ERK-MAPK, and p38-MAPK, as well as direct oxidation by H₂O₂, and through covalent modification of Keap1; curcumin (turmeric) through phosphorylation by PKC, p38-MAPK, and subsequent to activation of PI3K; isothiocyanates (cruciferous vegetables) through phosphorylation by ERK-MAPK, p38-MAPK, AKT, and by covalent modification Keap1; and caffeic acid (coffee, tea, and wine) through covalent modification of Keap1. From these examples, Nrf2 is commonly activated through phosphorylation by a variety of redox-sensitive kinases and covalent modification of Keap1, presumably through a quinone intermediate, also occurs relatively frequently. These mechanisms reveal that rather than antioxidant properties, it is much more likely that it is the prooxidant and electrophilic properties of phytochemicals that confer their protective effect. Because Nrf2 activation leads to an increased expression of all of the major classes of antioxidant and redox enzymes, diets that include an array of these compounds should enhance both redox control of signal transduction pathways as well as increase protection from (temporary) exposures to large amounts of ROS and RNS as might occur during an active infection or exposures to exogenous-derived reactive compounds. These protective functions may, in part, explain the strong associations between diets that contain relative high amounts of various phytochemicals and a reduced risk for breast, prostate, and colorectal cancers described in Section 6.1.1 as well as for lung, renal, and GI cancers.^{501,504,592–594}

In Section 6.3.2, the ability of various phytochemicals to suppress proinflammatory signaling through attenuating the various MAPK pathways was described. These effects are, however, seemingly directly contradictory to the enhanced activities of various MAPKs by the same phytochemicals for Nrf2 activation. Rather than being a conundrum, these opposing results are expected due to the timing of the events and the specific effects that are measured. The activation processes reflect the rates of the chemical reactions necessary for the events to occur. These would include the autoxidation of the hydroxyls and subsequent generation of H₂O₂ (followed by oxidation of susceptible –SH groups) or the various Michael reactions that also can occur, all of which would be relatively short-term phenomena considering the half-life of the phytochemicals. From the data on flavonoids, the bioavailability of these compounds is in the 2%–20% range, which results in biological concentrations in the 250 nM to 30 µM range.^{463,538,595,596} With a half-life of from 1 to 2 hours, the biological effects of these phytochemicals following ingestion are sure to be transient; possibly a moderate effect that declines quickly over a couple hours (or less). As a guess, considering typical eating patterns and the amounts eaten, the activation effects of phytochemicals might last for an hour or so, possibly once or twice a day (well, on days they are actually consumed). Thus, the activation time of the various MAPK pathways and Nrf2 would be relatively transient. The half-lives of the proteins expressed are much longer; MnSOD has a half-life of approximately 5 hours while that of CAT, GPX, and Trx are in the 24-48-hour range. 597-600 Ultimately, it is the transient activation of intracellular signaling that produces a longer-term attenuation of the signaling processes through increased expression of the antioxidant and redox enzymes. The net effect is to dampen the average or chronic signaling processes that can lead to an enhanced risk for tumorigenesis while still allowing for enhanced signaling due to short-term cellular stresses. In these situations, regular consumption of these phytochemicals would dampen the overall total daily signaling activity and result in a protective effect.

6.3.3.1 Cell Cycle

Because cell progression into and through the cell cycle is dependent on activity of the cellular signaling pathways as well as the redox state of the cell, a phytochemical-induced increase in redox control coupled with dampened activity of the signaling pathways would then alter the progression of stem and PCs through the cell cycle. From a number of reviews on the subject, a variety of cell signaling functions in cancer cells have been observed to be altered by a wide variety of phytochemicals, changes that can affect different aspects of the progression phase of carcinogenesis.^{500,578,601–606} For example, benyl isothionate (BITC), phenethyl isothionate (PEITC), curcumin, quercetin, resveratrol, theaflavins, genistein, sulforaphane, apigenin, and EGCG (among others) have all been documented to enhance rates of apoptosis in cancer cells. Mechanisms of this effect have been associated, in part, with increased caspase activation (isothiocyanates, sulforaphane, and apigenin), decreases in cyclin D1 and cdk (curcumin and resveratrol), and inhibition of NFk β (BITC, resveratrol, genistein, theaflavins, EGCG, and curcumin). Inhibiting NF $\kappa\beta$ is an important mechanism not only because it attenuates apoptosis (in part through inducing Bcl proteins) but also because it is a component of the activation pathways for proinflammatory signaling.

Cell cycle arrest and suppression of growth also have been observed with many phytochemicals, including BITC, luteolin, resveratrol, apigenin, genistein, EGCG, curcumin and quercetin.^{601–605,607} These effects have been observed to be associated with suppression of ERK-MAPK and JNK-MAPK activities, reduction in activation of Ras, Myc, Jun, and/or Fos (curcumin, EGCG, quercetin, genistein, and resveratrol), and reductions in expression of p38, Cdc2, and Fos while increased expression of p21 also has been observed. And each of these growth suppression and enhanced apoptosis effects from cell culture studies has been associated with phytochemical-mediated reductions in chemical-induced tumorigenesis in a variety of animal studies.

From the standpoint of redox control, nutrient-based dietary supplements have had little effect other than correcting various nutritional deficiencies while supplements that contain individual phytochemicals have their own limitations due to issues of toxicity. In addition, the majority of the mechanistic studies from the preceding discussion are based on treating transformed cells (or animals with chemical-induced or transplanted tumors) with individual phytochemicals in amounts that are unlikely to be obtained from a diet. This raises some issues as to how well these studies apply to diet-based prevention in humans, before they have transformed cells (i.e., cancer) and because human diets actually contain a wide array of different phytochemicals (in differing and lower amounts), each with their own array of (often overlapping) effects. The value of these studies is not that they reveal how any single phytochemical can prevent cancer through dietary modification or supplementation but rather how a variety of phytochemicals through an array of relatively subtle effects on many different mechanisms of cellular function may integrate into an overall protective effect. And considering the integrated risks of proinflammatory signaling and prooxidant stresses within the tumor environment, the suppression of inflammatory signaling as well as the enhanced antioxidant capacity to optimize control of the cellular redox state could have profound overall risk-reduction effects.

6.3.3.2 DNA Damage and Repair

As described, the enhanced antioxidant capacity through induction of SOD, CAT, GPX, Trx, and Prx that is mediated through Nrf2 activation will reduce the levels of ROS in the cell and enhance redox control of signal transduction pathways. The ability of ROS and RNS to cause mutations directly through DNA damage and indirectly through lipid peroxidation–induced DNA damage also will be reduced. Reductions in DNA damage in peripheral lymphocytes in humans and various tissues in animals have been observed where diets have been modified by increasing consumption of fruit juice, fruit and/or vegetables, and whole berry extracts.^{608–612} In many of these studies, whether the reductions were due to antioxidant effects of the dietary compounds or induction of the various antioxidant enzymes is difficult to tell. Based on the ability of various phenolics to activate Nrf2 responses through a variety of mechanisms, it is, however, reasonable to infer that a Nrf2-mediated induction of antioxidant and redox-control enzymes is at least partially responsible for the

reduction in oxidative DNA damage. Enhanced DNA repair also could explain some of the observed reduction in oxidative DNA damage.

As described in Section 6.3.3, one of the effects of phytochemicals is the transient production of H_2O_2 through quinone/semiquinone cycling. This may lead to a transient oxidation of Trx and the subsequent activation of Ref-1. Because this endonuclease is a functional component of base excision repair (BER), its transient (and moderate) activation by Trx oxidation may be an important factor in enhancing DNA repair. This increase, coupled with a Ref-1-mediated activation of p53 and expression of p21, would help explain the enhanced apoptosis and cell cycle arrest observed for a variety of phytochemicals as well. Selenium has been observed to enhance BER in human fibroblasts in association with enhanced REF-1 activation while β -cryptoxanthin (a carotenoid) as well as ascorbate plus α -tocopherol appear to enhance DNA repair and reduce markers of DNA damage in HeLa cells and Ref-1 heterozygous mice, respectively.^{613–615} And interestingly, silibinin (from milk thistle) enhances p53 stabilization as well as upregulates GADD45 following UVB radiation of mouse epidermal cells. Because GADD45 is a downstream target of p53 (as well as being implicated in DNA repair and cell cycle arrest), each of these results is consistent with a role for regulation of Ref-1 activity by Trx in enhancing DNA repair.

In studies where regulatory mechanisms of DNA repair have not been determined, there are relatively consistent results that various phytochemicals including curcumin, indole-3-carbinol, resveratrol, ellagic acid, red raspberry extract, and quercetin can enhance rates of DNA repair.^{609,616-618} In human dietary studies, flavonoid-rich diets (multiple fruits and vegetables) or addition of kiwi to the diet significantly enhanced DNA repair with enhanced expression of DNA repair enzymes being observed in some, although not in all.^{619–621} In some instances, the enhanced DNA repair was related to nutritional deficits being corrected rather than to a direct effect of the phytochemicals per se, again indicating the difficulty of interpreting human studies where nutritional insufficiencies are common. In one particular study using human cells exposed to 2-amino-1-methyl-6-phenylimidazo[4,5-b] pyridine (a heterocyclic amine found in grilled meats), a reduction in DNA adducts was observed following cotreatment with sulforaphane and quercetin; however, this effect was associated with a sulforaphane-mediated increase in UDP-GT and GST enzymes and a quercetin-mediated reduction in CYP1A2 activity.⁶²² The results of this last study indicate other mechanisms through which dietary phytochemicals can modify risk for carcinogenesis, inhibition of phase I activity that leads to the production of DNA-damaging electrophiles and induction of phase II activities that lead to the inactivation of them.

6.3.3.3 Phase I and Phase II Enzymes

As discussed in Section 6.2.5.2, the activation of various PAHs, nitrosamines, heterocyclic amines, and other xenobiotics to electrophilic intermediates by phase I enzymes such as CYP1A1, CYP1A2, CYP2A1, CYP2B1, and CYP2E1 are responsible for the actual production of the carcinogens from these parent compounds. And it is the phase II enzymes GST, UDP-GT, and NQO1 that are responsible in part for conjugating hydrophilic compounds to the electrophiles in order that they can be inactivated and easily excreted. Phase II enzymes also can be induced by binding of Nrf2 to the ARE/ERE response element of their genes, an event activated by a wide array of phytochemicals.

In a variety of studies, a consistent induction of GST, UDP-GT, and/or NQO1 has been observed following consumption of curcumin, benzyl isothiocyanate, coumarin, indole-3-carbinol, sulforaphane, or infusion of cruciferous extracts (isothiocyanate mixtures).⁶²²⁻⁶²⁷ Similar results are observed in cell cultures following treatment with a wide variety of phytochemicals including EGCG, sulforaphane, quercetin, catechin, indole-3-carbinol, phenethyl isothiocyanate, organosulfur compounds from garlic, and coumarins.^{628–632} In addition, chemically induced lung, liver, mammary, esophageal, and colon cancers also are significantly reduced with EGCG, N-acetylcysteine, quercetin, genistein, multiple types of berries, curcumin, resveratrol, tea polyphenols, and indole-3-carbinol in a variety of animal models.^{143,626,633-640} The reduction in risk in many these studies is associated with increased inactivation of the reactive carcinogens. While certainly not an exhaustive list, phytochemicals are capable of inducing phase II enzyme activity and reducing incidence and multiplicity of tumors that are caused by specific carcinogens. The application of each individual model to the general population, however, is a little less direct because relatively high doses of individual carcinogens that are unlikely to be seen in humans are used, as is the amount of specific phytochemical. The basic message, though, is that different phytochemicals can reduce risk for a variety of chemical-induced cancers through inducing expression of these phase II enzymes as observed in a variety of animal models and because humans are exposed to a wide array of possible carcinogens (at much lower doses) consuming a variety of phytochemicals should produce a general protective effect against all (most?).

In addition to reducing risk by inactivating active carcinogens, inhibiting cytochrome P450 activities also may contribute to risk reduction in some cases. In a wide variety of studies, activation of many different procarcinogens to carcinogens by CYP1A1, CYP1A2, CYP2A1, CYP2B1, CYP2E1, and CYP3A4 activities (as well those of other activating CYPs) is inhibited to differing degrees by quercetin, EGCG, curcumin, phenethyl isothiocyanate, benzyl isothiocyanate, and sulforaphane.^{601,641–646} On the other hand, some phytochemicals also have been observed to enhance expression of several CYP enzymes. Indole-3-carbinol, quercetin, flavone, tangeretin, and phenethyl isothiocyanate, for example, induce expression of CYP1A1 and CYP1A2 in a variety of tissues including the intestines, lung, and liver while St. John's Wort is a very effective inducer of CYP3A4 and CYP2B6.601,647-650 Some phytochemicals also can induce activity of certain CYPs while inhibiting others. Diallyl disulfide from garlic, for example, can increase CYP2B1 activities while inhibiting CYP2E1, and the anthraquinone emodin can induce CYP1A1 while at the same time inhibiting its activity. Because the absolute activities of individual CYP enzymes are difficult to associate with specific risks for individual cancers and because of the different effects that various phytochemicals can have on CYP activities, it is even more difficult to determine how risks for cancer may be modified through such a mechanism. Because of the consistent induction of phase II enzymes by a variety of phytochemicals and the consistent association with a reduction in risk for cancer by such changes, it is far more likely that the risk reduction benefits of a high-phytochemical diet are conferred through this mechanism rather than through

inhibiting phase I enzymes. This brings us to the last dietary component that consistently reduces risk for cancer through enhancing antioxidant and redox-control enzymes and the phase II enzymes—caloric restriction.

6.3.4 CALORIE RESTRICTION AS PREVENTION

Of all the dietary modifications, calorie restriction (CR) while maintaining appropriate nutrient intake appears to be the most consistent means to reduce risk for cancer.^{651–654} In animal models, calories are typically restricted to approximately 60%-80% of what an ad libitum-fed animal might consume with mechanisms of risk reduction including reduced insulin levels, increased insulin sensitivity, reduced ROS/RNS production, enhanced antioxidant and redox control enzyme activities, and reductions in circulating proinflammatory cytokines.^{652,655,656} In addition to these, circulating IGF-1 is reduced while that of adiponectin is increased. These last two would be expected to occur because of the reduced calorie intake and the resulting reduction in adiposity. In response to increases in serum glucose following eating (and overeating), the insulin that is released not only activates glucose transport into cells through translocation of GLUT4 transporters to the cell membrane but it also stimulates synthesis of IGF-1 by the liver; resulting in increased circulating IGF-1. Because of the reduction in calories with CR, glycemic loads are reduced, along with both insulin and IGF-1. Adiponectin is inversely associated with adiposity and as logic would dictate with continuing CR, adiposity is reduced, and circulating adiponectin increases concomitantly.

The risk-reduction benefits of reduced IGF-1 and increased adiponectin are somewhat interrelated because of their common effects on mTOR. mTOR phosphorylates p70S6K and it is activated by Akt to promote both cell growth and proliferation. The PI3K/Akt pathway is activated by insulin and IGF-1 as well as by other growth factors and by increased nutrient supply. The subsequent proliferative and growth effects of activated mTOR are factors in obesity-related risks for cancer as well as being commonly seen in tumors.^{224,657} Under conditions of CR, the PI3K-mediated activation of Akt is suppressed, resulting in decreased mTOR activity. In addition, the lower-energy environment in cells produced by CR increases AMP levels in the cell, enhancing AMP-dependent protein kinase (AMPK) activity, which directly inhibits mTOR.658 The increased levels of adiponectin that also occur with CR not only enhance AMPK activity to mediate a decrease in mTOR activity but they also enhance expression of PGC-1a.659 Activation of PGC-1a (described in detail in Chapter 4) is associated with increased insulin sensitivity by acting as a coactivator of GLUT4 synthesis as well as possibly being involved in enhancing expression of insulin receptor substrate-1 (IRS1), providing an additional suppressing effect on circulating insulin by enhancing insulin sensitivity, which in turn lowers IGF-1 even further. Thus, the growth and proliferative suppressing effects of CR appear to be mediated in part through a coordinated increase in AMPK activity that inhibits mTOR as well as a decline in insulin and IGF-1-mediated activation of the PI3K/Akt pathway, further reducing mTOR activity.

As mentioned earlier in this section, a reduced production of ROS and RNS with enhanced redox control also occurs with CR. These responses are consistent with those mediated by Nrf2 binding to the ARE/ERE response element and indeed CR is associated with activation of this important regulator.^{656,660,661} As one might expect with activation of Nrf2, activities of phase II enzymes also are induced by CR and are considered to be a major reason for decreased DNA damage and tumorigenesis in carcinogen-exposed, CR animals.^{651,660,662–665}

CR thus appears to be an effective treatment for reducing risk for cancer on the basis of cellular mechanisms that are altered in a variety of animal CR models. Data on risk reduction by CR in humans is in extremely short supply although there are a few studies documenting reduced risk in populations with traditionally low (or forced) caloric intake, and in CR trials, similar alterations in insulin, IGF-1, adiponectin, and in metabolic indices occur. Unfortunately, such severe reductions in calorie intake are very difficult for the majority of humans. And while reductions in IGF-1, enhanced insulin sensitivity, and enhanced phase II activities occur in humans under conditions of CR, the altered IGF-1 levels appear to be associated in part with reduced protein intake. Elevated IGF-1 levels are observed with protein intakes that exceed the 0.8 g/kg recommendation while diets that restrict protein intake to this level from customary higher intakes, even without CR, are associated with lowered IGF-1 levels.^{666–668} With a median intake of ~91 g/day for adults between 19 and 50 years, 1.4 and 1.2 g/kg/day for American males and females (19-50 years) respectively,⁶⁶⁹ protein intake in sedentary people might be considered an independent risk for cancer through elevating IGF-1 and a factor to be considered for non-CR recommendations for prevention.

6.3.5 Exercise

Physical activity has been recognized as an important lifestyle issue in risk for cancer for many years and it is a key component of risk-reduction recommendations of many organizations.^{95,97,155,670} Physical activity is widely documented in epidemiology studies to be inversely associated with risk for all cancers. In cases where individual cancers are studied, however, the inverse association is relatively modest or inconsistent for some (e.g., prostate, lung, and ovarian cancers) while there is a consistent and strongly protective association for colon, breast, and endometrial cancers.^{46,671} From the standpoint of how exercise affects cellular functions, however, one might expect a more consistent as well as a stronger inverse association. Perhaps, some of the limitations inherent in the epidemiology research are in part responsible for this. Diet and physical activity are notoriously difficult to accurately assess in such studies and an accurate assessment of possible confounding nutritional, body composition, and environmental variables are essentially unfeasible.^{182,672} Also, with an approximate 33% prevalence of complete inactivity coupled with an additional $\sim 32\%$ prevalence of physical activity that does not meet the minimal requirements for preventive effects, we would conjecture that there simply may be too few subjects who actually exercise at enough different levels of intensity to observe a statistically significant inverse association in all but the largest epidemiological samples. In addition, a pervasive lack of a nutritionally adequate diet for a large majority of Americans may be a serious confounding factor that prevents some of the positive effects of exercise to be observed. This last issue, coupled with the long latency period for the development of cancer, also would make it very difficult to assess the ability of physical activity to reduce risk for cancer in all but the longest duration clinical trials as well. As discussed in Chapter 1, it is precisely these limitations that form the basis for developing a focus on using various biomarkers of risk in the context of properly designed clinical trials in this area of research.¹⁸²

When it comes to specific mechanisms of cancer etiology that are affected by exercise, the array of cellular functions that are modified are very similar to those described for dietary components. While not typically determined in cancer-associated studies, these mechanisms have been extensively studied in a wide variety of exercise-specific experiments and are highly relevant to cancer.

6.3.5.1 Exercise and Insulin Sensitivity

As discussed in Section 6.2.4.1, elevated insulin enhances risk for cancer through enhancing growth and proliferative signaling within the tumor environment through direct activation of PI3K activities and indirectly through insulin-mediated increases in hepatic expression of IGF-1. Increased glucose levels lead to enhanced cellular damage through the formation of Amadori products and to DNA damage from enhanced lipid peroxidation as well as increased proinflammatory signaling through activation of RAGE. Enhancing insulin sensitivity would therefore reduce risk for cancer through two principal mechanisms: reducing circulating levels of insulin and reducing circulating levels glucose. The ability of physical activities to enhance insulin sensitivity and reduce circulating levels of both insulin and glucose has been discussed in detail in Chapters 3 and 4 and will be summarized here.

A major component of the exercise-insulin issue is that inactivity can cause insulin resistance in skeletal muscle, a component of risk for approximately 33% of Americans who report that they are completely inactive. An enhanced local production of ROS due to inactivity mediates an induction of TNF α synthesis that in turn enhances lysine phosphorylation of IRS1. This enhances its degradation as well as impairs its ability to complex with p85, thus reducing insulin sensitivity.^{165–167,673–678} Fortunately, insulin sensitivity can return to normal relatively quickly with exercise.

The ability of both endurance training and resistance training to enhance insulin sensitivity has been well documented, as has a transient increase (~24 hours) in insulin sensitivity following acute exercise.^{679–688} Interestingly, insulin sensitivity is enhanced to a greater degree at higher intensity activities when the exercise is performed 3 days/week, with even greater insulin sensitivity observed when weight training is combined with endurance exercise, although little to no association with increasing intensity is observed with daily exercise. Repeated exercise training is known to enhance GLUT4 and IRS synthesis.^{689–694} These increases are likely initiated in part through phosphorylation or deacetylation of PGC-1 α and phosphorylation of estrogen receptors following an exercise-mediated activation of MAPKs, AMPK, and SIRT1. Increased expression of both GLUT4 and IRS has the effect of increasing the activation of glucose transport at a given level of insulin. Acute exercise also is known to increase insulin sensitivity for as long as 17 hours following the activity, with this effect also mediated in part by AMPK.⁶⁹⁵⁻⁶⁹⁸ The Rab-GTPase:AS160 is normally inhibited following phosphorylation by Akt (insulin \rightarrow $IR/IRS \rightarrow PI3K \rightarrow PI3,4,5-TP:PDK1 \rightarrow Akt)$ to allow Rab-GTP-mediated translocation of GLUT4 proteins to the cell membrane. This Rab-GTPase as well as TBC1D1 also can be phosphorylated by AMPK, leading to insulin-independent glucose transport. In addition to AMPK-mediated effects, muscle contraction also appears to activate Akt. Thus, through the transient inhibition of Rab-GTPases and the increased expression of GLUT4 and IRS proteins, repeated exercise can enhance insulin sensitivity. With daily exercise, one would expect the prolonged acute (and transient) increase in insulin sensitivity through AS160/TBC1D1 phosphorylation (the "last bout effect") to be additive to the increase due to an increased expression of GLUT4 and IRS (the "repeated bout effect") to maximize sensitivity while with alternate-day activity the more transient last bout effect would have dissipated by the second day to reduce the effective daily average level of insulin sensitivity. This average, however, appears to be enhanced by more strenuous alternate-day activities as observed in some studies. Whether a slightly greater average insulin sensitivity that may occur as a result of more strenuous alternate-day activities results in significantly lower risk for cancer compared to less strenuous alternate-day activities is unknown. Based on the kinetics of the formation of Amadori products and their contribution to damage and proinflammatory signaling, however (as discussed in Chapter 4), it is doubtful there may be a significant difference.

Because skeletal muscle is the largest organ that is responsive to insulin and is an important contributor to glucose homeostasis through its removal, alterations in skeletal muscle sensitivity to insulin produce significant changes in risk for cancer. Importantly, most kinds of physical activity enhance insulin sensitivity in skeletal muscle and also are associated with a reduction in risk for cancer, with cellular mechanisms that are directly associated with tumorigenesis being reduced through enhanced insulin sensitivity. Thus, to contribute to a reduction in risk for cancer through maintaining "optimal" insulin sensitivity, moderately stressful exercise activities of either strength training or endurance training should be performed at a minimum of every other day.

6.3.5.2 Exercise, Redox Control, and Phase II Enzymes

Exercise training is commonly known to improve athletic performance in part by enhancing metabolic function of skeletal muscle through increased biogenesis of mitochondria. And it is the transient increase in production of the cellular oxidant H_2O_2 due to the high metabolic demands of the exercise that is responsible in part for many of the metabolic adaptations to exercise.^{168,564,699–703} These adaptations not only include mitochondrial biogenesis, but in several of the more recent studies, enhanced activities or expression of one or more of the redox control enzymes (GSR, GPX, Trx, and Prx) and/or of the phase II enzymes (UDP-GT, GST, and NQO1) have been observed in a variety of tissues.^{173,704–710} The ability of acute endurance-type exercise to activate Nrf2:ARE binding also has been observed and this would be expected because Nrf2 is a transcription factor that is known to be associated directly with mitochondrial biogenesis, as is PGC-1 α , MEF2, and Tfam (among others).^{710–719} The importance of Nrf2 binding to ARE/ERE promoter sequences is that Nrf2 is a major transcription factor responsible for initiating transcription of the redox control enzymes as well as phase II enzymes as discussed in Section 6.3.3. Because it also is necessary for mitochondrial biogenesis, it is apparent that physical activities that lead to mitochondrial biogenesis in skeletal muscle also can lead to enhanced redox control and phase II enzyme activities.

A brief summary of the activation of mitochondrial biogenesis follows.^{694,710-720} To sustain muscle contractions, the metabolic production of ATP is greatly enhanced, with both the muscle contractions themselves and the enhanced activity of the various metabolic enzymes activated by increased cytosolic calcium (as well as a variety of allosteric regulators). The enhanced metabolic production of H₂O₂, which occurs concomitant with the enhanced ATP synthesis, then activates the various MAPKs, Nrf2, and NF $\kappa\beta$. Increased cytosolic calcium also activates PKC to further activate the MAPK pathways as well as calcineurin A (CnA) and Cacalmodulin-dependent protein kinase (CaMK). The CnA and CaMK in turn activate the transcription factors cyclic AMP response element-binding protein (CREB) and myocyte enhancer factor-2 (MEF2) while p38-MAPK activity mediates activation of the transcription factor ATF2 as well as MEF2 and CREB. MEF2, CREB, and ATF2 then enhance expression of PGC-1 α . PGC-1 α then coactivates Nrf2, Nrf1, NFAT, and other factors to initiate transcription of the mitochondrial genes that are necessary in part for mitochondrial biogenesis. Posttranslational modification of PGC-1α such as phosphorylation by AMPK or deacetylation by SIRT1 also activates PGC-1α to enhance its coactivation activity and increased AMPK activity also activates SIRT1. Thus, there are essentially two "pathways" to activate mitochondrial biogenesis through exercise, enhanced MAPK, CaMK, and CnA activities through ROS and Ca⁺⁺-mediated mechanisms (among others) and additionally through direct activation of PGC-1a through AMPK and SIRT1. And one would expect maximal activation of both of these pathways with maximal exercise, a concept that has been tested to its extreme relatively recently where in a variety of studies extremely low-volume intervals of maximal or near-maximal intensity exercise (e.g., 5×30 seconds of maximum effort on a cycle-ergometer with 4 minutes rest between each interval performed 3 days/week) produces significant elevations in aerobic capacity, induces mitochondrial biogenesis, and enhances insulin sensitivity.693,721-727

The reason for this digression into mitochondrial biogenesis is to illustrate the signaling processes that mediate mitochondrial biogenesis also can enhance insulin sensitivity and activity of redox control enzymes and phase II enzymes. These responses are initiated through activating the MAPK signaling pathways, CaMK, and CnA as well as AMPK and SIRT1 activities and coordinated by PGC-1a and Nrf2, each of which can be activated directly or indirectly through physical activitymediated release of Ca++ and/or enhanced production of H2O2 and/or elevated AMP. While the activation of PGC-1a appears to be required for both enhanced mitochondrial biogenesis and insulin sensitivity, Nrf2 activation does not appear to be essential for enhancing insulin sensitivity nor is it considered a requirement for muscle hypertrophy. In addition, muscle hypertrophy responses and mitochondrial biogenesis responses tend to be mutually exclusive.^{660,728,729} One reason for this last item is that enhanced AMPK and SIRT1 activities that typically occur with longer-term endurance exercise (both) inhibit mTOR, an essential mediator of muscle hypertrophy. While this does not necessarily mean that resistance exercises cannot activate Nrf2 sufficiently to enhance expression of the redox control and phase II enzymes, no research was found to document any changes in redox-control or phase II enzymes following resistance training. Thus, based on currently available data, it is the longer-duration moderate to high stress and shorter-duration maximal stress endurance-type exercise that appears to enhance mitochondrial function, and in some cases VO_{2max} , which likely mediates an enhanced expression of redox control and phase II enzymes and not resistance exercise alone.

Much of the molecular-based information on mitochondrial biogenesis and insulin sensitivity has been performed on muscle cells of one type or another, as has much of the research on redox control enzymes and exercise. Because sarcomas of skeletal muscle are very rare in comparison to cancers of the lung, breast, colon, liver, and prostate (and others), cancer risk reduction through enhanced Nrf2 signaling in nonmuscle tissue is an important consideration. In the relatively few studies that have investigated the induction of either the various redox-control or phase II enzyme activities, some inconsistencies in results have been observed.^{173,704-710,730} In studies that used 6-8 weeks of endurance training in rats, GPX and GSR activities have been increased in heart and skeletal muscle but not in liver. In a longer-term study with beagle dogs and a similar duration study using swimming rats, enhanced hepatic GST activities have been observed with a substantial increase in the Ya_1 subunit. This is relevant because the Ya1 subunit is associated with activation of the ARE response element to which Nrf2 binds. Eight weeks of running exercise training in rats also has been observed to enhance Trx activity in the brain and a similar duration of running exercise enhances UDP-GT activity in the lung. In these studies, the duration of the exercise periods ranged from 40 minutes/day for the beagles and from 60 minutes to 120 minutes for the rat studies at what would be termed moderately stressful intensity. Why some studies saw no change is very difficult to say and may reflect differences in strain and possibly age of the animals used. Ultimately, there simply are not enough studies performed to make any specific definitive conclusions other than to recognize that in the majority of available studies where enhanced activities of Nrf2-dependent enzymes are observed in nonmuscle tissues, it is following longer-duration moderate-intensity exercise training, a form of exercise that also has relevance for DNA repair.

6.3.5.3 DNA Repair

As discussed in Section 6.3.3.2, oxidation of Ref-1 by Trx in the nucleus enhances DNA repair by Ref-1 and increases cell cycle arrest through a Ref-1-mediated enhanced p53 activation and p21 expression. An enhanced content of Trx mediated by activation of Nrf2 would be expected to enhance DNA repair and reduce risk for tumorigenesis through this mechanism and in addition through enhanced p53/ p21. As might be expected, in those few studies that investigated exercise and DNA repair, reductions in DNA damage as well as enhanced activity of DNA repair has been observed.^{731–736} These effects appear to be related in part to the enhanced ROS production during acute exercise periods and occur in muscle, brain, liver, and circulating leukocytes. The DNA repair enzyme oxoguanine DNA glycosylase (OGG1) is mainly responsible for the repair of oxidative damage to guanine and is the specific enzyme identified as being induced by exercise. Interestingly, the gene for this enzyme has Nrf2-binding sites in its promoter region.^{737–739} With Nrf2 activation associated with mitochondrial biogenesis as well as being necessary for induction of

OGG1, it is no surprise that the type of repeated exercise used in the referenced studies was moderately stressful endurance exercise (running or swimming) for 60–90 minutes and is suggestive that longer-term exercise might be necessary to enhance DNA repair. The implication of a duration effect (60–90 minutes) is very interesting. The IL-6 promoter contains an ARE consensus sequence that is activated by Nrf2, and IL-6 also has been implicated as a required signaling molecule for repair of hepatic mitochondrial DNA through induction of OGG1 and the prevention of ROS-induced DNA fragmentation in lung, preventive effects that do not happen in IL-6 knockout mice.^{740–742} Thus, the possibility exists that IL-6 is a coactivator along with Nrf2 for augmenting the enhanced DNA repair following longer-term exercise. Thus, the form and duration of exercise that increases IL-6 production may be optimal for enhancing DNA repair while mitochondrial biogenesis, antioxidant and redox control enzyme, and phase II enzyme induction can occur with shorter duration exercise.

6.3.5.4 IL-6-Mediated Anti-Inflammatory Effects of Exercise

The ability of physical activity to reduce inflammatory signaling has been discussed in detail in Chapters 2 and 4, so it will be presented in summary here. With approximately 1 hour or more of continuous moderate-intensity exercise, serum levels of IL-6 can be elevated as much as 100-fold before diminishing to resting levels over a period of from 3 to 5 hours.^{743–750} This increase in IL-6 originates from the active skeletal muscle cells during nonexhaustive whole-body exercise and has been observed with activity as low as ~6 METS; the IL-6 response does not occur with small-muscle group exercise. The nonexhaustive nature of the exercise is associated with no increase in TNF α , a response that does occur with exhaustive and/or damaging exercise. As a result of the large increase in IL-6, expression of IL-10, sIL-1ra (soluble IL-1 receptor antagonist), sIL-1r (soluble IL-1 receptor), and sTNFr (soluble TNF receptor) is induced.

The sIL-1r and sTNFr bind to circulating IL-1 α , IL-1 β , and TNF α while the sIL-1ra inhibits IL-1 α and IL-1 β from binding to their cellular receptors. The IL-10 activates the Jak-STAT pathway to activate synthesis of the SOCS proteins, which then inhibits the synthesis of IL-1 α , IL-1 β , IL-6, IL-8, and TNF α .^{751–753} IL-10 also suppresses both MAPK and NF $\kappa\beta$ signaling.^{751–754} Thus, as a result of the large initial production of IL-6, there is a profound anti-inflammatory effect of the longer-term moderate intensity endurance-type exercise. As a result of repeated exercise of this type, one would expect a reduction in serum markers of inflammation such as C-reactive protein (CRP).

From a variety of clinical trials testing the association between repeated exercise and CRP, those trials that utilized moderate to strenuous exercise of a duration from 45 minutes to 1 hour tended to demonstrate a reduction in CRP while in those with exercise durations of 30 minutes there were no changes in CRP observed.^{755–764} These results are consistent with an anti-inflammatory effect mediated by IL-6 induction from the longer duration exercise periods. They also reveal a possible reason for the longer duration activities to be associated with enhanced DNA repair.

In summary, exercise can produce a variety of effects that can reduce mechanisms that contribute to cause for a wide variety of cancers. Daily or alternate daily exercise

training of almost any form and resistance exercise training as well as most daily low to moderate physical activities (gardening, dancing, light-effort bicycle riding, golf, household chores, etc.) are associated with an increase in insulin sensitivity, which lowers circulating insulin and glucose and therefore reduces their associated promotional effects. Moderate intensity (30 or more minutes continuous; five times a week), vigorous (20 minutes or more; three times week), and high or maximum intensity ($\sim 0.5-10$ minutes duration repeated approximately three to six times, three times a week) exercises that enhance mitochondrial biogenesis are associated with transiently activating MAPKs, NF $\kappa\beta$, and Nrf2 to enhance expression of antioxidant enzymes, redox control enzymes, and phase II enzymes and to increase DNA repair activity. While enhancing insulin sensitivity occurs with less intensive and shorter duration activities to reduce risk for cancer, the enhanced redox control, phase II, and DNA repair benefits appear to accrue only with the more intense and longer duration activities that also enhance mitochondrial biogenesis. It is only with the longer duration (~1 hour or longer) of moderate intensity activities that the significant anti-inflammatory effects and possibly the enhanced DNA repair activities mediated in part by IL-6 will occur.

6.4 SUMMARY AND RECOMMENDATIONS

Cancer is a major public health issue that causes approximately one in fourth deaths in the United States each year and costs over \$125 billion each year just for care and treatment. While there are many different cancers that arise from different tissues, the general processes of carcinogenesis are remarkably similar for each.

6.4.1 CANCER ETIOLOGY

Essentially, cancer arises from tissue SCs and possibly PCs that divide and accumulate mutations in genes that code for any of a variety of proteins that are directly involved in regulating different aspects of the cell cycle or that are involved in signaling processes that in turn lead to entry into the cell cycle. The initiation stage is the process that leads to the acquisition of the first driver mutation that enhances risk for cancer. This mutation could occur when a SC divides before the complete repair of existing DNA damage or it could be inherited. Most chemical modifications to DNA bases (DNA damage) lead to the production of point mutations during the S phase of the cell cycle. If single-strand breaks are present, they can lead to the production of DSBs during the DNA-replication process and DSBs that persist into the M phase ultimately lead to a variety of chromosomal mutations. These point or chromosomal mutations are then present in the daughter cells, and if they are again stimulated to divide in the face of additional unrepaired DNA damage, then additional mutations also will be acquired in the subsequent generation of cells. While the vast majority of mutations acquired during the promotion phase are passenger mutations, the relatively few driver mutations that enhance rates of cell division, that interfere with DNA repair, or that impede checkpoint control ultimately promote the development of a tumor and its subsequent progression to malignancy over many cell division cycles.

Because regulating entry into the S phase of the cell cycle, preserving the integrity of checkpoint control, and maintaining efficient DNA repair are so important to the fidelity of DNA duplication and cell division, mutations in proteins that are involved in one or more of these processes are very common to all human cancers. Mutations in the genes that code for proteins of the signal-transduction pathways that subsequently lead to growth and cell division such as BRAF and KRAS, or genes that are directly involved in $G0 \rightarrow S$ or $G1 \rightarrow S$ transition such as *RB*, are very common in human cancers. Enabling mutations in these will enhance rates of cell division. The TP53 gene also is very commonly mutated in almost all human cancers and appears to be an important mutation that leads to malignancy. This is most likely because P53 is vital to maintaining checkpoint control. When sensor proteins detect DNA damage, this protein is activated to enhance cell cycle arrest and prevent progression through the cell cycle until the DNA damage is repaired. P53 also induces expression of a variety of proteins that block BCL and MXL proteins from inhibiting Bax and Bak, two regulatory proteins that stimulate the release of cytochrome c and other apoptotic proteins from mitochondria. On release of the apoptotic proteins, caspases are activated to initiate the process of apoptosis, ultimately ending in cell death (unless the DNA is repaired in a timely manner to stop the process and continue safely through the cell cycle).

If the dividing cells acquire disabling mutations in any of the genes that code for proteins that are involved in checkpoint control or DNA repair, then the cells will likely progress through the cell cycle to produce additional mutations in subsequent generations. As the dividing mutated cells accumulate additional mutations in a variety of genes over an indeterminate number of generations, they ultimately acquire the functional characteristics of cancer cells, a phenotype with specific hallmark functions: a sustained ability to stimulate their own cell division, an inability to differentiate into normal adult tissue cells, an inability to respond to normal cellular signals that suppress cell division, an inability to respond to apoptosis signals, the ability for indefinite proliferation, an ability for invasiveness and metastasis, and an ability for stimulating angiogenesis.

Once the carcinogenesis process is initiated and SCs and PCs proliferate, a small array of mSC and PCs will accumulate within the tumor environment and develop slightly hypoxic conditions as they accumulate some distance from the existing capillaries. The slight hypoxic environment activates HIF-1 and subsequently a variety of proteins that confer a growth advantage for cells in this environment are induced, along with chemoattractant proteins that enhance the infiltration of the local region with activated macrophages, lymphocytes, NK cells, mast cells, endothelial cells, platelets, and fibroblasts. Thus, the developing tumor environment is really a heterogeneous mixture of mSCs, mPCs, and tumor-associated cells that include a large number of inflammatory cells and platelets. The activated inflammatory cells are intimately involved in the promotion and progression phase. Activated inflammatory cells release a variety of growth and proliferative signaling molecules including PGE₂, IL-1β, TNFα, PDGF, TGFβ, EGF, bFGF, RANTES, platelet factor 4, IL-8, and IL-6 (among others). The wide array of inflammatory signaling molecules released from the activated proinflammatory cells initiate cellular responses in the mSCs and mPCs that are tightly integrated with the already impaired cell cycle control, inadequate DNA repair, and deficient apoptosis signaling processes. This integration of the extracellular initiated and intracellular initiated/impaired signaling processes leads to the development of tumors with immortal, self-stimulating, and invasive CSCs that can independently stimulate their own angiogenesis to sustain continued tumorigenesis essentially anywhere in the body.

Of the two fundamental processes that are essential for carcinogenesis, the first is enhanced proliferative signaling that accelerates cell division of SCs and PCs. One source of proliferative signaling is derived from driver mutations within the SCs and PCs. Another is from the array of growth and proliferative signaling molecules released from activated inflammatory cells. A third is PGE_2 that is released from apoptotic cells that have died through normal senescence; very often senescent PCs. A fourth would be PGE_2 that is released by any cell following almost any kind of cell stress; thus, there are plenty of additional sources of proliferative signaling available. While short-term production of the prostaglandin due to transient cell stresses very rarely leads to a proliferative response, longer-term signaling from chronic damage–associated stresses will. Chronic inflammation is especially important to this process because it is not only a source of a steady supply of many different growth and proliferation signaling prostaglandins, cytokines, and growth factors but it also greatly enhances production of a variety of ROS and RNS. This is, of course, the second fundamental process that is necessary for carcinogenesis: a source of reactive molecules to cause DNA damage.

The major endogenous sources of radical species include mitochondrial respiration, NADPH-oxidase activity of activated G-coupled proteins and by phagolysosomes of activated neutrophils and macrophages, cytochrome P450 monooxygenases, and nitric oxide synthase in vascular endothelial cells and activated neutrophils and macrophages. The major radical molecules produced from these processes are $O_{\overline{2}}$ and NO, which then participate in a variety of reactions to produce H_2O_2 , ONOO⁻, CO_{3}^{-} , NO₂, and $\cdot OH$. Of these, the H₂O₂ does not actually cause DNA damage, but as the major cellular oxidant, it can enhance activity of the various signal transduction pathways to contribute to growth and proliferative signaling. Of the DNA-damaging species, the most aggressive oxidant is the hydroxyl radical. In addition to causing DNA damage, the OH also causes lipid peroxidation to produce a variety of DNAdamaging products that include malondialdehyde, acrolein, crotonaldehyde, and 4-hydroxynonenal. Because H₂O₂, NO, and ONOO⁻ readily diffuse across cellular membranes, their production by activated inflammatory cells in the tumor environment contribute to the carcinogenesis process by providing a steady supply of additional ROS and RNS that then leads to DNA damage and the ultimate production of a variety of transitions and transversions.

Exogenous compounds arising from the environment also can play an important role in carcinogenesis by adding to the DNA damage from endogenous sources, compounds such as PAHs, nitrosamines, aldehydes, metals, nitrogen oxides, acrolein, and many others that are common to combustion products. These carcinogens are present in tobacco as well as being common air and water pollutants. Tobacco, of course, is probably the most "effective" source of carcinogens due to environmental exposures because of the very high concentrations of carcinogens achieved during normal and repeated use of tobacco products. Although relatively very rare, environmental exposures due to industrial accidents also may produce high-concentration exposures, albeit far less frequently in comparison to smoking. Other environmental carcinogens include the aflatoxins, the N-nitroso compounds formed from nitrates and nitrites in processed meats, and the heterocyclic amines and PAHs formed from barbecuing and searing meats.

Many of the "environmental" compounds are not in and of themselves reactive. They require metabolic activation to become active DNA-damaging species, usually by the CYP1, CYP2, CYP3, and CYP4 families of the phase I enzymes. Once activated to an electrophilic species, they then react with susceptible components of the different DNA bases to produce DNA damage that can lead to the production of an array of transitions and transversions. These reactions are a normal component of a series of reactions that are designed to modify the original compounds to make them more easily eliminated in the urine. The production of reactive intermediates by CYP is sometimes necessary for a second CYP reaction that often involves the hydroxylation or demethylation of the electrophile to make it more water soluble. Thus, some CYP reactions generate electrophiles while others can eliminate them. Additional reactions by phase II enzymes also are involved in this process. UDP-GT, GST, and the acetyltransferases all are involved in conjugating relatively large water-soluble components to the electrophiles, essentially inactivating them. Phase II enzymes are therefore important in reducing the potential for DNA damage by the electrophiles that arise from xenobiotics following phase I activation.

Overall, DNA damage from both exogenous and endogenous sources are very important in producing risk for cancer. Although DNA damage from exogenous sources may comprise up to 20% of the total damage, individual behaviors such as smoking and frequent consumption of BBQ meats can significantly enhance risk for cancer by greatly increasing exposures to exogenous carcinogens. Reducing significant exposures to environmental carcinogens by not smoking is a relatively straightforward affair; do not smoke or use tobacco products (and so much easier said than done), while avoiding breathing contaminated air, and drinking contaminated food and water can be more problematic because of the endemic nature of air and water pollution. The behavioral risks associated with diet and physical activity may be more amenable to change.

The etiology of cancer is complex when individual cancers are considered, however; there are mechanisms common to all with a majority of the risk arising from endogenous mechanisms that enhance proliferative signaling and DNA damage. These endogenous mechanisms can be exacerbated by a variety of exogenous factors that relate to environmental exposures to carcinogenic xenobiotics that can enhance DNA damage and/or proliferative signaling. All of these risks can be enhanced to a remarkable degree through poor diet and through inadequate (or nonexistent) physical activity.

6.4.2 DIET-BASED PREVENTION

Diet-based reduction in risk for cancer is based on two general concepts. The first is that poor dietary habits can significantly enhance existing risks. Perhaps, the most common detrimental dietary behavior today is the overconsumption of calories, a situation that applies to many of the over 70% of the population that is either overweight or obese. The cellular dysfunction that ensues in fat cells that are subject to chronic lipid overload leads to the production of a variety of adipokines and proinflammatory molecules including adiponectin, leptin, and visfatin as well as TNF α , IL-6, IL-1 β , and MCP-1. These signaling molecules are necessary to activate the infiltration of the growing adipose tissue with the proinflammatory cells and fibroblasts that are necessary for adopogenesis. These signaling molecules also enter the circulation and can enhance growth and proliferative signaling in the tumor environment as well as to produce additional risks.

The additional risks from weight gain are associated with the enhanced circulating levels of TNF α that are implicated in developing insulin resistance in skeletal muscles. The insulin resistance leads to higher levels of circulating insulin and glucose, both of which provide growth advantages to the proliferating cells within the tumor environment. The chronic increase in insulin due to insulin resistance and the intermittent elevations in insulin due to overeating at mealtime enhance production of IGF-1 by the liver to provide an additional source of growth and proliferative signaling. The increased glucose levels provide essential substrates for the developing tumor cells as well as contributing to an enhanced production of AGEs, which then activate proinflammatory signaling through binding to the RAGEs of macrophages that are within the tumor environment. Thus, overeating behaviors are a significant source of risk for the development of cancers.

Eating an inadequate amount of nutrients also is a very common source of risk for cancers. Unfortunately, a majority of Americans consume a diet that is insufficient in one or more of the essential nutrients. While outright deficiencies are very rare, the very low consumption of fruits and vegetables that is endemic to American culture is associated with an insufficient intake of vitamins C, D, and E as well as folate, zinc, selenium, calcium, and magnesium. Each of these insufficiencies is associated with an increased risk for various cancers, most likely through compromised antioxidant status that leads to greater DNA damage and to inadequate control of various signaling pathways to enhance proliferative and proinflammatory signaling.

The second concept relates to the possibility that specific dietary components may reduce risk to a greater degree than correcting inadequate nutrient consumption alone will. A variety of foods contain beneficial phytochemicals that modify cellular functions such that proinflammatory and proliferative responses to common cellular stresses might be attenuated. In addition, the inactivation of reactive carcinogens and DNA repair also may be enhanced by these phytochemicals.

Production of the proinflammatory eicosanoids and cytokines in response to cellular stress and to infection and infection-related damage is mediated by activation of the MAPK signal transduction pathways, NF $\kappa\beta$, PLA2, COX, and LOX. Regulation of these activities depends in part on the activation of PRRs by components of infectious agents or cellular debris. The phytochemicals curcumin, resveratrol, sulforaphane, EGCG, and luteolin are known to attenuate PRR activation and subsequent proinflammatory signaling. Others, including quercetin, curcumin, EGCG, kaempferol, apigenin, morin, anthocyanins, procyanidins, and lycopene, directly suppress NF $\kappa\beta$ signaling to reduce expression of proinflammatory cytokines. In addition, MAPK pathways also can be directly inhibited by these compounds; this ultimately moderates inflammatory responses to infectious agents and reduces the likelihood of excessive "peripheral damage" due to exaggerated inflammatory responses. A variety of compounds from EVOO such as tyrosol, hydroxytyrosol, apigenin, luteolin, and oleuropein as well as EGCG, quercetin, and kaempferol also directly inhibit COX and/or LOX to reduce the production of proinflammatory eicosanoids. In addition to moderating inflammatory responses, all of these inhibitory effects also would reduce the activation of proinflammatory signaling by noninfectious cellular stress.

Other compounds with anti-inflammatory properties are EPA and DHA. American diets tend to be low in α -linolenic acid, which is the precursor essential fatty acid for EPA and DHA synthesis. These fatty acids are essential for production of the resolvins and protectins that contribute to the resolution phase of inflammation. They also can directly activate PPAR γ to inhibit MAPK pathways and platelet aggregation, both of which attenuate proinflammatory signaling.

The activities of the various MAPK pathways also are relevant to regulation of the cell cycle. Cell cycle arrest and suppression of growth by BITC, luteolin, resveratrol, apigenin, genistein, EGCG, curcumin, and quercetin in association with suppression of ERK-MAPK and Jnk-MAPK activities and a reduction in activation of Ras, Myc, Jun, and/or Fos have been observed. In addition, rates of apoptosis are enhanced by these phytochemicals and all these alterations are associated with reductions in tumorigenesis.

MAPK activities also are dependent in part on the redox state of the cell that in turn is dependent on the degree of metabolic stress, ROS/RNS production, and the relative activities of the antioxidant and redox-control enzymes. Trx, TrxR, GSH, GSR, GPX, and Prx along with the antioxidant enzymes SOD and CAT are essential to regulating the overall cellular redox state. Expressions of these enzymes are all induced through binding of the Nrf2:Mah heterodimer transcription factor to the ARE. A variety of phytochemicals including quercetin, resveratrol, diallyl sulfide, s-allyl cysteine, lycopene, EGCG, curcumin, catechin, cyanidin, caffeic acid, and sulforaphane have been observed to enhance Nrf2 binding as well as expression of the various antioxidant redox-control enzymes. Nrf2 activation is associated with covalent binding of some phytochemicals to susceptible SH groups of Keap1, while others generate H2O2, which then disassociates Keap1 from Nrf2. The activation of Nrf2 also may occur through the transient H2O2-mediated activation of p38-MAPK, JNK-MAPK, ERK-MAPK, and PKC, which are known to phosphorylate Nrf2. Increased expression of these antioxidant and redox-control enzymes optimize control of the redox state of cells to reduce the likelihood of exaggerated proinflammatory or proliferative responses to a variety of cellular stresses.

Other enzymes that are regulated by Nrf2 include glutamate cysteine ligase, GST, HO-1, UDP-GT, and NQO1. Induction of these phase II enzymes by administration of EGCG, N-acetylcysteine, quercetin, genistein, multiple types of berries, curcumin, resveratrol, tea polyphenols, and indole-3-carbinol leads to an enhanced inactivation of carcinogens, a reduction in DNA damage, and attenuation of lung, liver, mammary, esophageal, and colon cancers in a variety of studies. The reduction in DNA damage observed in some of the studies reveals an additional possibility for risk reduction—DNA repair. While inconsistent in the few human studies, induction in DNA repair activities following treatment with curcumin, indole-3-carbinol,

resveratrol, ellagic acid, red raspberry extract, and quercetin have been observed consistently in animal studies. In some studies, this increase has been related to the activation of Ref-1, a component of the BER as well as an activator of p53.

Though not classically considered a risk for cancer, consumption of protein in amounts greater than the recommended 0.8 g/kg/day are associated with an increase in IGF-1 levels. Increased synthesis of IGF-1 also occurs due to the chronic "insulin overload" that results from calorie excess and from insulin resistance and is a source of risk due to the proliferative effects of these hormones. Interestingly, a reduction in IGF-1 with calorie restricted diets is one reason for the effectiveness of this dietary strategy. The association with protein intake is that reductions in protein intake to AI levels lead to reductions in IGF-1 that are similar to those seen with CR. Other effects of CR include enhanced insulin sensitivity mediated by activation of PGC-1 α and AMPK by adiponectin as well as an induced expression of antioxidant and redox-control enzymes and of phase II enzymes, effects mediated by activation of Nrf2. As expected from all of these changes, CR is associated with reduced DNA damage and attenuated tumorigenesis. The reason to single out the protein intake earlier is to suggest that while CR is an extremely difficult lifestyle to follow for most people, reducing protein intake to achieve lowered IGF-1 levels might be more a more palatable dietary strategy to reduce risk.

6.4.3 DIETARY RECOMMENDATIONS

Reducing diet-based risks for the development of cancers entails two basic guidelines. The first is to consume a diet that matches caloric intake with caloric expenditures and meets the RDA/AI guidelines for lipids, protein, carbohydrates, and each of the nutrients. The second is to include in a variety of foods that contain beneficial phytochemicals as integral components of the diet. The various dietary insufficiencies that are common to the "average" American diet are relatively easy to correct and the USDA has published dietary guidelines many times over the past decades that describe how to do just that. The latest *My Plate* guidelines are an excellent starting point. To maximize the risk-reduction benefits of a nutritious diet through the consumption of the various phytochemicals, the additional servings of nuts/seeds and beans/peas that are outlined in Chapter 2 also are recommended.

As is obvious from the fact that the majority of the population consumes too many calories with too few nutrients, the constant barrage of information about proper diet, food groups, "healthy foods," and serving sizes in health classes in schools, "health clubs," and in the media simply is not having much of a positive effect. Perhaps, correcting poor dietary habits is not such as easy thing to do after all; the current approach of giving good advice without providing detailed guidance on how to change behavior certainly has not worked very well. Because altering risk for cancer entails altering a lifetime of poor dietary habits, a behavioral approach to learning a new diet rather than a knowledge-only approach may be more successful. Most details of the dietary recommendations have already been described in other chapters, so they will not be reproduced here. For a general proposal on how to integrate the dietary recommendations within a behavioral approach to lifestyle change, the reader is referred to Chapter 9 where all of the dietary recommendations have been incorporated into a recommendation for behavioral modification (Figure 6.2).



FIGURE 6.2 Summary of effects of diet and exercise on risks for cancer. Nutrient insufficiencies, calorie overload, and inactivity lead to compromised antioxidant and redox status, enhanced proinflammatory signaling, and elevated reactive oxygen species (ROS)/oxidant-mediated damage. These lead to increase cell proliferation and the ultimate production of a variety of enabling and disabling mutations that drive the cancer processes and may account for as much as 80% or more of the total risk. These effects are enhanced through voluntary and involuntary exposures to a variety of procarcinogenic compounds from the environment. Consumption of a calorie-balanced, phytochemical-rich diet along with 60 continuous minutes of daily exercise will optimize redox and antioxidant control, DNA repair, and phase II activities and attenuate proinflammatory signaling to minimize both cell proliferation and mutagenesis. Altering behavior to reduce the voluntary use of higher-risk products such as tobacco, BBQ meats, pesticides, and plastic products also will reduce risk for a large minority of the risks. (Details are discussed in the text.)

6.4.4 Exercise

While an active lifestyle has been known to be inversely associated with risk for cancer for many decades, the recognition that inactivity is a specific risk is a relatively new phenomenon. The role of inactivity in increasing cancer risks is, of course, intimately related to that of obesity. Physical activity certainly can be an effective calorie burner to avoid weight gain if performed frequently and for a sufficiently long time. Inactivity also is closely associated with the development of insulin resistance and therefore all of its inherent proinflammatory and proliferative risks. The inactivityinduced insulin resistance is independent of that arising from weight gain and therefore applies to the normal-weight component of the population that is inactive. An important issue is that almost any kind of physical activity that is performed for 10 minutes or more several times each day is associated with maintaining insulin sensitivity. Whether resistance training, "aerobic" type training, or just being involved in activities such as gardening and other yard work or social dancing, regular and daily physical activities avoid inactivity-induced insulin resistance. The more intense physical activities appear to increase insulin sensitivity in part through increased expression of GLUT4 and IRS, most likely mediated through repeated activation of PGC-1 α by an exercise-mediated activation of MAPKs, AMPK, and SIRT1. Acute exercise also can enhance AMPK activity to phosphorylate AS160 and TBC1D1 to mediate a prolonged activation of existing GLUT4 transporters.

In addition to being highly beneficial in maintaining insulin sensitivity and in promoting weight maintenance, other benefits of exercise also can be very important. The enhanced mitochondrial biogenesis and the improved "aerobic capacity" that accompanies some forms of exercise are mediated in part through PGC-1 α , MEF2, Tfam, and Nrf2. The type of physical activity that improves "aerobic capacity" is typically of moderate intensity and longer duration (30–60 minutes) although repeated bouts (three to five) of very short-term (60 seconds) maximum intensity exercise also are known to induce mitochondrial biogenesis through the same molecular pathways. Most notable is that Nrf2 is involved in mediating mitochondrial biogenesis and that the forms of exercise that increase aerobic capacity also enhance activity of a variety of important protective enzymes in nonmuscle tissue. These enzymes include the antioxidant and redox control enzymes SOD, GPX, Trx, and GSR as well as the phase II enzymes GST and UDP-GT to enhance redox control and to reduce ROS- and non-ROS-mediated DNA damage.

Rates of DNA repair also can occur with the more stressful, longer duration exercise. Enhanced activity of the DNA repair enzyme OGG1 has been observed following moderately stressful endurance exercise such as swimming or running for 60–90 minutes. This duration is the same time range for induction of IL-6 synthesis by muscle cells and both OGG and IL-6 have Nrf2-binding sites in their promoter. With IL-6 being implicated as a possible coinducer of DNA repair along with its known anti-inflammatory effects, it is evident that longer duration aerobic exercise may produce the greatest cancer-preventive effects.

The anti-inflammatory effects of exercise are mediated by IL-6 through increased expression of IL-10, soluble IL-1 receptor antagonist, soluble IL-1 receptor, and soluble TNF receptor. Binding of any circulating IL-1 α , IL-1 β , and TNF α to sIL-r and

sTNFr will reduce the effect of these circulating proinflammatory cytokines while sIL-1ra is an antagonist for IL-1 binding to its receptor, thus reducing the proinflammatory effects of these cytokines. The IL-10 activates synthesis of the SOCS, which then inhibits the synthesis of IL-1 α , IL-1 β , IL-6, IL-8, and TNF α to further reduce proinflammatory signaling. In clinical trials, the longer duration aerobic exercise is associated with reductions in CRP and is consistent with the concept that longer durations of Nrf2 activation are necessary for an anti-inflammatory effect to be realized.

6.4.5 Exercise Recommendations

Based on the molecular mechanisms of risk for carcinogenesis and for altering those mechanisms, there are several levels of risk reduction that can be obtained through exercise. Daily or alternate daily "aerobic" and resistance exercise training or daily low to moderate intensity physical activities such as gardening, dancing, light-effort bicycle riding, and golf increase insulin sensitivity.

To obtain the greater protective benefits of enhanced expression of antioxidant enzymes, redox control enzymes, phase II enzymes moderate intensity aerobic exercise that enhances mitochondrial biogenesis (30 or more minutes continuous; five times a week), vigorous (20 minutes or more; three times a week) and high or maximum intensity (~0.5–10 minutes duration repeated three to five times; three times a week) is necessary.

To maximize the cancer-preventive effects through additional anti-inflammatory and enhanced DNA repair, the longer duration (1 hour or longer) moderate intensity activities that enhance expression of IL-6 by myocytes need to be performed on a daily basis, which thus becomes the recommendation for exercise. These longer duration activities also are important in helping to avoid weight gain. If the long duration activities cannot be performed everyday due to time limitations, then it may be possible that similar reductions in risk may be attained through daily exercise using an alternate-day schedule: alternating the longer duration activities with shorter duration high intensity activities.

ACKNOWLEDGMENT

This work was supported in part by a partial salary grant from NEIU to C. Murray Ardies.

REFERENCES

- Fontham ETH, Thun MJ, Ward E, Balch AJ, Delancey JOL, Samet JM. American cancer society perspectives on environmental factors and cancer. *CA: A Cancer Journal for Clinicians*. 2009;59:343–351.
- 2. World Health Organization. *The World Health Report 2002: Reducing Risks, Promoting Healthy Life.* Geneva, CH: World Health Organization; 2002.
- 3. Jemal A, Bray F, Center MM, Ferlay J, Ward E, Forman D. Global cancer statistics. *CA: A Cancer Journal for Clinicians*. 2011;61:69–90.

- 4. Jemal A, Siegel R, Ward E, Murray T, Xu J, Smigal C, Thun MJ. Cancer statistics, 2006. *CA: A Cancer Journal for Clinicians*. 2006;56:106–130.
- Jemal A, Siegel R, Ward E, Hao Y, Xu J, Thun MJ. Cancer statistics, 2009. CA: A Cancer Journal for Clinicians. 2009;59:225–249.
- Jemal A, Siegel R, Xu J, Ward E. Cancer statistics, 2010. CA: A Cancer Journal for Clinicians. 2010;60:277–300.
- 7. Siegel R, Naishadham D, Jemal A. Cancer statistics, 2012. CA: A Cancer Journal for Clinicians. 2012;62:10–29.
- 8. Siegel R, Ward E, Brawley O, Jemal A. Cancer statistics, 2011. *CA: A Cancer Journal for Clinicians*. 2011;61:212–236.
- Mariotto AB, Yabroff KR, Shao Y, Feuer EJ, Brown ML. Projections of the cost of cancer care in the United States: 2010–2020. *Journal of the National Cancer Institute*. 2011;103:117–128.
- Hagedoorn M, Buunk BP, Kuijer RG, Wobbes T, Sanderman R. Couples dealing with cancer: Role and gender differences regarding psychological distress and quality of life. *Psycho-Oncology*. 2000;9:232–242.
- 11. Payne S, Smith P, Dean S. Identifying the concerns of informal carers in palliative care. *Palliative Medicine*. 1999;13:37.
- 12. Carter PA. Family caregivers' sleep loss and depression over time. *Cancer Nursing*. 2003;26:253.
- 13. Thomas C, Morris SM. Informal carers in cancer contexts. *European Journal of Cancer Care*. 2002;11:178–182.
- 14. Matthews AB. Role and gender differences in cancer-related distress: A comparison of survivor and caregiver self-reports. *Oncology Nursing Society*. 2003;30:493–499.
- 15. DeVita VT, Lawrence TS, Rosenberg SA, DePinho RA, Weinberg RA. *DeVita, Hellman, and Rosenberg's Cancer: Principals and Practice of Oncology.* Philadelphia, PA: Lippincott Williams & Wilkins; 2011.
- Weinberg RA. *The Biology of Cancer*. New York: Garland Science, Taylor & Francis Group; 2006.
- 17. Bertram JS. The molecular biology of cancer. *Molecular Aspects of Medicine*. 2000;21: 167–223.
- Lobo NA, Shimono Y, Qian D, Clarke MF. The biology of cancer stem cells. *Annual Review of Cell and Developmental Biology*. 2007;23:675–699.
- Finkel T, Serrano M, Blasco MA. The common biology of cancer and ageing. *Nature*. 2007;448:767–774.
- 20. Ruddon RR. Cancer Biology. London: Oxford University Press, Inc.; 2007.
- Vincent T. DeVita J, Lawrence TS, Rosenberg SA. Cancer Principals & Practice of Oncology—Primer of the Molecular Biology of Cancer. Philadelphia, PA: Lippincott Williams & Wilkins; 2011.
- 22. Kuper H, Adami HO, Boffetta P. Tobacco use, cancer causation and public health impact. *Journal of Internal Medicine*. 2002;251:455–466.
- Gandini S, Botteri E, Iodice S, Boniol M, Lowenfels AB, Maisonneuve P, Boyle P. Tobacco smoking and cancer: A meta-analysis. *International Journal of Cancer*. 2008;122:155–164.
- Bach PB, Kattan MW, Thornquist MD, Kris MG, Tate RC, Barnett MJ, Hsieh LJ, Begg CB. Variations in lung cancer risk among smokers. *Journal of the National Cancer Institute*. 2003;95:470–478.
- Brennan P, Crispo A, Zaridze D, Szeszenia-Dabrowska N, Rudnai P, Lissowska J, Fabiánová E, Mates D, Bencko V, Foretova L. High cumulative risk of lung cancer death among smokers and nonsmokers in central and eastern Europe. *American Journal of Epidemiology*. 2006;164:1233–1241.
- Godtfredsen NS, Prescott E, Osler M. Effect of smoking reduction on lung cancer risk. Journal of the American Medical Association. 2005;294:1505–1510.

- Villeneuve P, Mao Y. Lifetime probability of developing lung cancer, by smoking status, Canada. Canadian Journal of Public Health [Revue Canadienne de Santé Publique]. 1994;85:385.
- Kubík A, Zatloukal P, Tomasek L, Pauk N, Havel L, Dolezal J, Plesko I. Interactions between smoking and other exposures associated with lung cancer risk in women: Diet and physical activity. *Neoplasma*. 2007;54:83.
- 29. Thune I, Lund E. The influence of physical activity on lung-cancer risk: A prospective study of 81,516 men and women. *International Journal of Cancer*. 1997;70:57–62.
- Tang L, Zirpoli G, Jayaprakash V, Reid M, McCann S, Nwogu C, Zhang Y, Ambrosone C, Moysich K. Cruciferous vegetable intake is inversely associated with lung cancer risk among smokers: A case-control study. *BMC Cancer*. 2010;10:162.
- Büchner FL, Bueno-de-Mesquita HB, Ros MM, Overvad K, Dahm CC, Hansen L, Tjønneland A, Clavel-Chapelon F, Boutron-Ruault MC, Touillaud M. Variety in fruit and vegetable consumption and the risk of lung cancer in the European prospective investigation into cancer and nutrition. *Cancer Epidemiology, Biomarkers & Prevention*. 2010;19:2278–2286.
- 32. Fiore MC, Novotny TE, Pierce JP, Giovino GA, Hatziandreu EJ, Newcomb PA, Surawicz TS, Davis RM. Methods used to quit smoking in the United States. *Journal of the American Medical Association*. 1990;263:2760–2765.
- 33. Moore D, Aveyard P, Connock M, Wang D, Fry-Smith A, Barton P. Effectiveness and safety of nicotine replacement therapy assisted reduction to stop smoking: Systematic review and meta-analysis. *British Medical Journal*. 2009;338:b1024.
- Ward KD, Klesges RC, Zbikowski SM, Bliss RE, Garvey AJ. Gender differences in the outcome of an unaided smoking cessation attempt. *Addictive Behaviors*. 1997;22:521–533.
- 35. Gonzales D, Rennard SI, Nides M, Oncken C, Azoulay S, Billing CB, Watsky EJ, Gong J, Williams KE, Reeves KR. Varenicline, an α4β2 nicotinic acetylcholine receptor partial agonist, vs. sustained-release bupropion and placebo for smoking cessation. *Journal of the American Medical Association*. 2006;296:47–55.
- Zhu SH, Melcer T, Sun J, Rosbrook B, Pierce JP. Smoking cessation with and without assistance: A population-based analysis. *American Journal of Preventive Medicine*. 2000;18:305–311.
- Volpp KG, Troxel AB, Pauly MV, Glick HA, Puig A, Asch DA, Galvin R, Zhu J, Wan F, DeGuzman J. A randomized, controlled trial of financial incentives for smoking cessation. *New England Journal of Medicine*. 2009;360:699–709.
- 38. Drake BF, Colditz GA. Assessing cancer prevention studies—A matter of time. *Journal* of the American Medical Association. 2009;302:2152–2153.
- Garcia M, Jemal A, Ward E, Center M, Hao Y, Siegel R, Thun M. Global Cancer Facts and Figures 2007. Atlanta, GA: American Cancer Society; 2007;1.
- 40. Russo J, Russo IH. The role of estrogen in the initiation of breast cancer. *Journal of Steroid Biochemistry and Molecular Biology*. 2006;102:89–96.
- 41. Clemons M, Goss P. Estrogen and the risk of breast cancer. *New England Journal of Medicine* 2001;344:276–285.
- 42. Henderson BE, Feigelson HS. Hormonal carcinogenesis. *Carcinogenesis*. 2000;21: 427–433.
- 43. King MC, Marks JH, Mandell JB. Breast and ovarian cancer risks due to inherited mutations in BRCA1 and BRCA2. *Science*. 2003;302:643–646.
- 44. Antoniou A, Pharoah P, Narod S, Risch HA, Eyfjord JE, Hopper J, Loman N, Olsson H, Johannsson O, Borg Å. Average risks of breast and ovarian cancer associated with BRCA1 or BRCA2 mutations detected in case series unselected for family history: A combined analysis of 22 studies. *American Journal of Human Genetics*. 2003;72:1117–1130.
- 45. Levy-Lahad E, Friedman E. Cancer risks among BRCA1 and BRCA2 mutation carriers. *British Journal of Cancer*. 2007;96:11–15.

- 46. Friedenreich CM, Orenstein MR. Physical activity and cancer prevention: Etiologic evidence and biological mechanisms. *Journal of Nutrition*. 2002;132:3456S–3464S.
- 47. Friedenreich CM, Woolcott CG, McTiernan A, Ballard-Barbash R, Brant RF, Stanczyk FZ, Terry T, Boyd NF, Yaffe MJ, Irwin ML. Alberta physical activity and breast cancer prevention trial: Sex hormone changes in a year-long exercise intervention among post-menopausal women. *Journal of Clinical Oncology*. 2010;28:1458–1466.
- 48. Hauner H, Hauner D. The impact of nutrition on the development and prognosis of breast cancer. *Breast Care*. 2010;5:377.
- Trichopoulou A, Bamia C, Lagiou P, Trichopoulos D. Conformity to traditional Mediterranean diet and breast cancer risk in the Greek EPIC (European Prospective Investigation into Cancer and Nutrition) cohort. *American Journal of Clinical Nutrition*. 2010;92:620–625.
- McEligot AJ, Mouttapa M, Ziogas A, Anton-Culver H. Diet and predictors of dietary intakes in women with family history of breast and/or ovarian cancer. *Cancer Epidemiology*. 2009;33:419–423.
- 51. Thune I, Brenn T, Lund E, Gaard M. Physical activity and the risk of breast cancer. *New England Journal of Medicine*. 1997;336:1269–1275.
- 52. Parkin DM, Pisani P, Ferlay J. Estimates of the worldwide incidence of 25 major cancers in 1990. *International Journal of Cancer*. 1999;80:827–841.
- Lu-Yao G, Stukel TA, Yao SL. Prostate-specific antigen screening in elderly men. *Journal* of the National Cancer Institute. 2003;95:1792–1797.
- 54. Gann PH, Hennekens CH, Ma J, Longcope C, Stampfer MJ. Prospective study of sex hormone levels and risk of prostate cancer. *Journal of the National Cancer Institute*. 1996;88:1118–1126.
- Shaneyfelt T, Husein R, Bubley G, Mantzoros CS. Hormonal predictors of prostate cancer: A meta-analysis. *Journal of Clinical Oncology*. 2000;18:847–853.
- Isbarn H, Pinthus JH, Marks LS, Montorsi F, Morales A, Morgentaler A, Schulman C. Testosterone and prostate cancer: Revisiting old paradigms. *European Urology*. 2009; 56:48–56.
- Morgentaler A. Testosterone and prostate cancer: An historical perspective on a modern myth. *European Urology*. 2006;50:935–939.
- Feldman BJ, Feldman D. The development of androgen-independent prostate cancer. *Nature Reviews Cancer*. 2001;1:34–45.
- 59. Hodgson MC, Bowden WA, Agoulnik IU. Androgen receptor footprint on the way to prostate cancer progression. *World Journal of Urology*. 2012;30:279–285.
- Henderson BE, Ross RK, Pike MC, Casagrande JT. Endogenous hormones as a major factor in human cancer. *Cancer Research*. 1982;42:3232–3239.
- 61. Oliveria S, Lee I. Is exercise beneficial in the prevention of prostate cancer? *Sports Medicine*. 1997;23:271.
- 62. Torti DC, Matheson GO. Exercise and prostate cancer. Sports Medicine. 2004;34:363-369.
- Young-McCaughan S. Potential for prostate cancer prevention through physical activity. World Journal of Urology. 2012:30:167–179.
- 64. Hsing AW, Gao YT, Chua Jr S, Deng J, Stanczyk FZ. Insulin resistance and prostate cancer risk. *Journal of the National Cancer Institute*. 2003;95:67–71.
- Barnard R, Aronson W, Tymchuk C, Ngo T. Prostate cancer: Another aspect of the insulin-resistance syndrome? *Obesity Reviews*. 2002;3:303–308.
- Weinstein D, Laron Z, Werner H. Evidence of direct mitogenic activity of insulin and the insulin receptor in prostate cancer cell lines. *Endocrine Abstracts*. 2012;29:P984.
- 67. Chen W, Li Z, Bai L, Lin Y. NF-kappaβ in lung cancer, a carcinogenesis mediator and a prevention and therapy target. *Frontiers in Bioscience*. 2011;16:1172–1185.
- Ben-Neriah Y, Karin M. Inflammation meets cancer, with NFκβ as the matchmaker. *Nature Immunology*. 2011;12:715–723.

- 69. Spitz MR, Gorlov IP, Amos CI, Dong Q, Chen W, Etzel CJ, Gorlova OY, Chang DW, Pu X, Zhang D. Variants in inflammation genes are implicated in risk of lung cancer in never smokers exposed to second-hand smoke. *Cancer Discovery*. 2011;1:420–429.
- Engels EA. Inflammation in the development of lung cancer: Epidemiological evidence. Expert Review of Anticancer Therapy. 2008;8:605–615.
- 71. Klein EA, Silverman R. Inflammation, infection, and prostate cancer. *Current Opinion in Urology*. 2008;18:315.
- 72. Nelson WG, De Marzo AM, DeWeese TL, Isaacs WB. The role of inflammation in the pathogenesis of prostate cancer. *Journal of Urology*. 2004;172:S6–S12.
- 73. De Nunzio C, Kramer G, Marberger M, Montironi R, Nelson W, Schröder F, Sciarra A, Tubaro A. The controversial relationship between benign prostatic hyperplasia and prostate cancer: The role of inflammation. *European Urology*. 2011;60:106–117.
- Radisky ES, Radisky DC. Stromal induction of breast cancer: Inflammation and invasion. *Reviews in Endocrine and Metabolic Disorders*. 2007;8:279–287.
- 75. Grivennikov SI, Greten FR, Karin M. Immunity, inflammation, and cancer. *Cell*. 2010;140:883–899.
- 76. Mantovani A, Allavena P, Sica A, Balkwill F. Cancer-related inflammation. *Nature*. 2008;454:436–444.
- 77. Pikarsky E, Porat RM, Stein I, Abramovitch R, Amit S, Kasem S, Gutkovich-Pyest E, Urieli-Shoval S, Galun E, Ben-Neriah Y. NF-kappaβ functions as a tumour promoter in inflammation-associated cancer. *Nature*. 2004;431:461–465.
- Tlsty TD, Coussens LM. Tumor stroma and regulation of cancer development. Annual Review of Pathology: Mechanisms of Disease. 2006;1:119–150.
- 79. Bradley CJ, Given CW, Roberts C. Race, socioeconomic status, and breast cancer treatment and survival. *Journal of the National Cancer Institute*. 2002;94:490.
- Reis L, Eisner M, Kosary C, Hankey B, Miller B, Clegg L, Mariotto A, Feuer E, Edwards B. Seer Cancer Statistics Review, 1975–2002. Bethesda, MD: National Cancer Institute; 2005;6.
- Clegg LX, Li FP, Hankey BF, Chu K, Edwards BK. Cancer survival among us whites and minorities: A seer (surveillance, epidemiology, and end results) program population-based study. *Archives of Internal Medicine*. 2002;162:1985.
- 82. Moormeier J. Breast cancer in black women. Annals of Internal Medicine. 1996;124:897.
- Lantz PM, Mujahid M, Schwartz K, Janz NK, Fagerlin A, Salem B, Liu L, Deapen D, Katz SJ. The influence of race, ethnicity, and individual socioeconomic factors on breast cancer stage at diagnosis. *American Journal of Public Health*. 2006;96:2173–2178.
- 84. Weir HK, Thun MJ, Hankey BF, Ries LAG, Howe HL, Wingo PA, Jemal A, Ward E, Anderson RN, Edwards BK. Annual report to the nation on the status of cancer, 1975– 2000, featuring the uses of surveillance data for cancer prevention and control. *Journal* of the National Cancer Institute. 2003;95:1276.
- 85. Reiter PL, Linnan LA. Cancer screening behaviors of African American women enrolled in a community-based cancer prevention trial. *Journal of Women's Health*. 2011;20:429–438.
- Smith-McLallen A, Fishbein M. Predicting intentions to engage in cancer prevention and detection behaviors: Examining differences between black and white adults. *Psychology, Health and Medicine*. 2009;14:180–189.
- Nelson A. Unequal treatment: Confronting racial and ethnic disparities in health care. Journal of the National Medical Association. 2002;94:666.
- Copeland VC. African Americans: Disparities in health care access and utilization. *Health and Social Work*. 2005;30:265–270.
- Friedman C, Brownson RC, Peterson DE, Wilkerson JC. Physician advice to reduce chronic disease risk factors. *American Journal of Preventive Medicine*. 1994;10:367–371.

- AICR. Reduce your cancer risk. http://www.aicr.org/reduce-your-cancer-risk/recommendations-for-cancer-prevention/. 2012. Accessed 12 September, 2012.
- 91. Stay healthy. http://www.cancer.org/healthy/index. 2012. Accessed 12 September, 2012.
- 92. Cancer prevention: 7 tips to reduce your risk. http://www.mayoclinic.com/health /cancer-prevention/ca00024. 2012. Accessed 12 September, 2012.
- 93. Cancer prevention. http://www.cdc.gov/cancer/dcpc/prevention/. 2012. Accessed 12 September, 2012.
- Cancer prevention. http://www.cancer.gov/cancertopics/pdq/prevention/overview/patient. 2012. Accessed 12 September, 2012.
- Kushi LH, Doyle C, McCullough M, Rock CL, Demark-Wahnefried W, Bandera EV, Gapstur S, Patel AV, Andrews K, Gansler T. American Cancer Society Guidelines on nutrition and physical activity for cancer prevention. *CA: A Cancer Journal for Clinicians*. 2012;62:30–67.
- 96. Pate RR, Pratt M, Blair SN, Haskell WL, Macera CA, Bouchard C, Buchner D, Ettinger W, Heath GW, King AC. Physical activity and public health. *Journal of the American Medical Association*. 1995;273:402–407.
- Pate RR, Yancey AK, Kraus WE. The 2008 physical activity guidelines for Americans: Implications for clinical and public health practice. *American Journal of Lifestyle Medicine*. 2010;4:209.
- Lloyd-Jones D, Adams R, Carnethon M, De Simone G, Ferguson TB, Flegal K, Ford E, Furie K, Go A, Greenlund K. Heart disease and stroke statistics—2009 update: A report from the American Heart Association Statistics Committee and Stroke Statistics Subcommittee. *Circulation*. 2009;119:e21–e181.
- Lloyd-Jones D, Adams RJ, Brown TM, Carnethon M, Dai S, De Simone G, Ferguson TB, Ford E, Furie K, Gillespie C. Heart disease and stroke statistics—2010 update. *Circulation*. 2010;121:e46–e215.
- 100. Roger VL, Go AS, Lloyd-Jones DM, Adams RJ, Berry JD, Brown TM, Carnethon MR, Dai S, de Simone G, Ford ES. Heart disease and stroke statistics—2011 update 1. A report from the American Heart Association. *Circulation*. 2011;123:e18–e209.
- 101. Roger VL, Go AS, Lloyd-Jones DM, Benjamin EJ, Berry JD, Borden WB, Bravata DM, Dai S, Ford ES, Fox CS. Executive summary: Heart disease and stroke statistics—2012 update: A report from the American Heart Association. *Circulation*. 2012;125:188.
- 102. Rosamond W, Flegal K, Friday G, Furie K, Go A, Greenlund K, Haase N, et al. Heart disease and stroke statistics—2007 update. A report from the American Heart Association Statistics Committee and Stroke Statistics Subcommittee. *Circulation*. 2007;115:e69–e171.
- 103. Rosemond R, Flegal K, Furie K, Go A, Greenlund K, Haase N, Hailpern SM, et al. Heart disease and stroke statistics—2008 update. A report from the American Heart Association Statistics Committee and Stroke Statistics Committee. *Circulation*. 2008;117:e25–e146.
- McCullough ML, Patel AV, Kushi LH, Patel R, Willett WC, Doyle C, Thun MJ, Gapstur SM. Following cancer prevention guidelines reduces risk of cancer, cardiovascular disease, and all-cause mortality. *Cancer Epidemiology, Biomarkers & Prevention*. 2011;20:1089.
- 105. Doll R, Peto R. The causes of cancer: Quantitative estimates of avoidable risks of cancer in the United States today. *Journal of the National Cancer Institute*. 1981;66:1191.
- 106. Tominaga S. Cancer incidence in Japanese in Japan, Hawaii, and western United States. *National Cancer Institute Monograph*. 1985;69:83.
- 107. Shimizu H, Ross R, Bernstein L, Yatani R, Henderson B, Mack T. Cancers of the prostate and breast among Japanese and white immigrants in Los Angeles County. *British Journal of Cancer*. 1991;63:963.
- 108. Staszewski J, Haenszel W. Cancer mortality among the Polish-born in the United States. *Journal of the National Cancer Institute*. 1965;35:291.

- 109. Cook LS, Goldoft M, Schwartz SM, Weiss NS. Incidence of adenocarcinoma of the prostate in Asian immigrants to the United States and their descendants. *The Journal of Urology*. 1999;161:152–155.
- 110. Edwards BK, Howe HL, Ries LAG, Thun MJ, Rosenberg HM, Yancik R, Wingo PA, Jemal A, Feigal EG. Annual report to the nation on the status of cancer, 1973–1999, featuring implications of age and aging on US cancer burden. *Cancer*. 2002;94: 2766–2792.
- 111. Stanford JL, Stephenson RA, Coyle LM, Cerhan J, Correa R, Eley J, Gilliland F, Hankey B, Kolonel L, Kosary C, Ross R, Severson R, West D. Prostate Cancer Trends 1973–1995, SEER Program, National Cancer Institute. NIH Pub. No.99-4543; 1999, Bethesda MD.
- 112. Pu Y, Chiang H, Lin C, Huang C, Huang K, Chen J. Changing trends of prostate cancer in Asia. *The Aging Male*. 2004;7:120–132.
- 113. Clinton SK, Giovannucci E. Diet, nutrition, and prostate cancer. Annual Review of Nutrition. 1998;18:413–440.
- 114. Armstrong B, Doll R. Environmental factors and cancer incidence and mortality in different countries, with special reference to dietary practices. *International Journal of Cancer*. 1975;15:617–631.
- 115. Giovannucci E, Rimm EB, Colditz GA, Stampfer MJ, Ascherio A, Chute CC, Willett WC. A prospective study of dietary fat and risk of prostate cancer. *Journal of the National Cancer Institute*. 1993;85:1571–1579.
- 116. Hebert JR, Hurley TG, Olendzki BC, Teas J, MaY, Hampl JS. Nutritional and socioeconomic factors in relation to prostate cancer mortality: A cross-national study. *Journal of the National Cancer Institute*. 1998;90:1637–1647.
- 117. Michaud DS, Augustsson K, Rimm EB, Stampfer MJ, Willett WC, Giovannucci E. A prospective study on intake of animal products and risk of prostate cancer. *Cancer Causes & Control*. 2001;12:557–567.
- Ganmaa D, Li XM, Wang J, Qin LQ, Wang PY, Sato A. Incidence and mortality of testicular and prostatic cancers in relation to world dietary practices. *International Journal* of Cancer. 2002;98:262–267.
- 119. Chan JM, Holick CN, Leitzmann MF, Rimm EB, Willett WC, Stampfer MJ, Giovannucci EL. Diet after diagnosis and the risk of prostate cancer progression, recurrence, and death (United States). *Cancer Causes & Control*. 2006;17:199–208.
- Jacobsen BK, Knutsen SF, Fraser GE. Does high soy milk intake reduce prostate cancer incidence? The Adventist health study (United States). *Cancer Causes & Control*. 1998;9:553–557.
- 121. Kolonel LN, Hankin JH, Whittemore AS, Wu AH, Gallagher RP, Wilkens LR, John EM, Howe GR, Dreon DM, West DW. Vegetables, fruits, legumes and prostate cancer: A multiethnic case-control study. *Cancer Epidemiology, Biomarkers & Prevention*. 2000;9:795–804.
- 122. Cohen JH, Kristal AR, Stanford JL. Fruit and vegetable intakes and prostate cancer risk. *Journal of the National Cancer Institute*. 2000;92:61–68.
- 123. Key TJ, Allen N, Appleby P, Overvad K, Tjønneland A, Miller A, Boeing H, Karalis D, Psaltopoulou T, Berrino F. Fruits and vegetables and prostate cancer: No association among 1,104 cases in a prospective study of 130,544 men in the European prospective investigation into cancer and nutrition (epic). *International Journal of Cancer*. 2004;109:119–124.
- 124. Giovannucci E, Ascherio A, Rimm EB, Stampfer MJ, Colditz GA, Willett WC. Intake of carotenoids and retino in relation to risk of prostate cancer. *Journal of the National Cancer Institute*. 1995;87:1767–1776.
- 125. Giovannucci E, Clinton SK. Tomatoes, lycopene, and prostate cancer. *Proceedings of the Society for Experimental Biology and Medicine, Society for Experimental Biology and Medicine*. 1998;218:129–139.

- Berkow SE, Barnard ND, Saxe GA, Ankerberg-Nobis T. Diet and survival after prostate cancer diagnosis. *Nutrition Reviews*. 2007;65:391–403.
- 127. Nomura A, Henderson BE, Lee J. Breast cancer and diet among the Japanese in Hawaii. *The American Journal of Clinical Nutrition*. 1978;31:2020–2025.
- Deapen D, Liu L, Perkins C, Bernstein L, Ross RK. Rapidly rising breast cancer incidence rates among Asian-American women. *International Journal of Cancer*. 2002;99:747–750.
- 129. Porter P. "Westernizing" women's risks? Breast cancer in lower-income countries. *New England Journal of Medicine*. 2008;358:213–216.
- 130. Buell P. Changing incidence of breast cancer in Japanese-American women. *Journal of the National Cancer Institute*. 1973;51:1479.
- 131. Ziegler RG, Hoover RN, Pike MC, Hildesheim A, Nomura AMY, West DW, Wu-Williams AH, Kolonel LN, Horn-Ross PL, Rosenthal JF. Migration patterns and breast cancer risk in Asian-American women. *Journal of the National Cancer Institute*. 1993;85:1819–1827.
- Santarelli RL, Pierre F, Corpet DE. Processed meat and colorectal cancer: A review of epidemiologic and experimental evidence. *Nutrition and Cancer*. 2008;60:131–144.
- Potter JD, Slattery ML, Bostick RM, Gapstur SM. Colon cancer: A review of the epidemiology. *Epidemiologic Reviews*. 1993;15:499–545.
- 134. Norat T, Riboli E. Meat consumption and colorectal cancer: A review of epidemiologic evidence. *Nutrition Reviews*. 2001;59:37–47.
- 135. Wilmink ABM. Overview of the epidemiology of colorectal cancer. *Diseases of the Colon & Rectum*. 1997;40:483–493.
- 136. Ames BN. DNA damage from micronutrient deficiencies is likely to be a major cause of cancer. *Mutation Research*. 2001;475:7–20.
- 137. Govindan R, Page N, Morgensztern D, Read W, Tierney R, Vlahiotis A, Spitznagel EL, Piccirillo J. Changing epidemiology of small-cell lung cancer in the United States over the last 30 years: Analysis of the surveillance, epidemiologic, and end results database. *Journal of Clinical Oncology*. 2006;24:4539.
- 138. Lam TK, Gallicchio L, Lindsley K, Shiels M, Hammond E, Tao XG, Chen L, Robinson KA, Caulfield LE, Herman JG. Cruciferous vegetable consumption and lung cancer risk: A systematic review. *Cancer Epidemiology, Biomarkers & Prevention*. 2009;18:184–195.
- 139. Gonzalez CA, Riboli E. Diet and cancer prevention: Contributions from the European Prospective Investigation into Cancer and Nutrition (EPIC) study. *European Journal of Cancer*. 2010;46:2555–2562.
- 140. Kontou N, Psaltopoulou T, Panagiotakos D, Dimopoulos MA, Linos A. The Mediterranean diet in cancer prevention: A review. *Journal of Medicinal Food*. 2011;14:1065–1078.
- 141. Hurst R, Hooper L, Norat T, Lau R, Aune D, Greenwood DC, Vieira R, Collings R, Harvey LJ, Sterne JAC. Selenium and prostate cancer: Systematic review and metaanalysis. *The American Journal of Clinical Nutrition*. 2012;96:111–122.
- 142. Amin ARMR, Kucuk O, Khuri FR, Shin DM. Perspectives for cancer prevention with natural compounds. *Journal of Clinical Oncology*. 2009;27:2712–2725.
- 143. Banerjee S, Manna S, Mukherjee S, Pal D, Panda CK, Das S. Black tea polyphenols restrict benzopyrene-induced mouse lung cancer progression through inhibition of Cox-2 and induction of caspase-3 expression. *Asian Pacific Journal of Cancer Prevention*. 2006;7:661–666.
- 144. Basile V, Ferrari E, Lazzari S, Belluti S, Pignedoli F, Imbriano C. Curcumin derivatives: Molecular basis of their anti-cancer activity. *Biochemical Pharmacology*. 2009;78:1305–1315.
- 145. Block G, Patterson B, Subar A. Fruit, vegetables, and cancer prevention: A review of the epidemiological evidence. *Nutrition and Cancer*. 1992;18:1–29.

- 146. Hong WK, Sporn MB. Recent advances in chemoprevention of cancer. *Science*. 1997;278:1073.
- 147. Kim YS, Young MR, Bobe G, Colburn NH, Milner JA. Bioactive food components, inflammatory targets, and cancer prevention. *Cancer Prevention Research*. 2009;2:200–208.
- 148. Surh YJ. Cancer chemoprevention with dietary phytochemicals. *Nature Reviews. Cancer*. 2003;3:768–780.
- 149. Tachibana H. Molecular basis for cancer chemoprevention by green tea polyphenol EGCG. In: *Food Factors for Health Promotion*. Ed: T. Yoshikawa. Basel, Switzerland: Karger; 2009;156–169.
- 150. Trichopoulou A, Lagiou P, Kuper H, Trichopoulos D. Cancer and Mediterranean dietary traditions. *Cancer Epidemiology, Biomarkers & Prevention*. 2000;9:869–873.
- 151. Calle EE, Rodriguez C, Walker-Thurmond K, Thun MJ. Overweight, obesity, and mortality from cancer in a prospectively studied cohort of US adults. *New England Journal of Medicine*. 2003;348:1625–1638.
- Calle EE, Kaaks R. Overweight, obesity and cancer: Epidemiological evidence and proposed mechanisms. *Nature Reviews. Cancer*. 2004;4:579–591.
- 153. Adams KF, Schatzkin A, Harris TB, Kipnis V, Mouw T, Ballard-Barbash R, Hollenbeck A, Leitzmann MF. Overweight, obesity, and mortality in a large prospective cohort of persons 50 to 71 years old. *New England Journal of Medicine*. 2006;355:763–778.
- 154. Adami HO, Trichopoulos D. Obesity and mortality from cancer. *New England Journal* of *Medicine*. 2003;348:1623–1624.
- 155. Haskell WL, Lee I-M, Pate RR, Powell KE, Blair SN, Franklin BA, Macera CA, Heath GW, Thompson PD, Bauman A. Physical activity and public health. Updated recommendation for adults from the American College of Sports Medicine and the American Heart Association. *Circulation*. 2007;116:1081–1093.
- 156. Ziccardi P, Nappo F, Giugliano G, Esposito K, Marfella R, Cioffi M, D'Andrea F, Molinari AM, Giugliano D. Reduction of inflammatory cytokine concentrations and improvement of endothelial functions in obese women after weight loss over one year. *Circulation*. 2002;105:804–809.
- 157. Bastard JP, Maachi M, Lagathu C, Kim MJ, Caron M, Vidal H, Capeau J, Feve B. Recent advances in the relationship between obesity, inflammation, and insulin resistance. *European Cytokine Network*. 2006;17:4–12.
- 158. Dandona P, Aljada A, Chaudhuri A, Mohanty P, Garg R. Metabolic syndrome: A comprehensive perspective based on interactions between obesity, diabetes, and inflammation. *Circulation*. 2005;111:1448.
- 159. Heilbronn LK, Noakes M, Clifton PM. Energy restriction and weight loss on verylow-fat diets reduce C-reactive protein concentrations in obese, healthy women. *Arteriosclerosis, Thrombosis, and Vascular Biology.* 2001;21:968–970.
- 160. Monteiro R, Azevedo I, Frühbeck G. Chronic inflammation in obesity and the metabolic syndrome. *Mediators of Inflammation*. 2010;2010:116.
- 161. van Kruijsdijk RCM, Van Der Wall E, Visseren FLJ. Obesity and cancer: The role of dysfunctional adipose tissue. *Cancer Epidemiology, Biomarkers & Prevention*. 2009;18:2569–2578.
- 162. Morris PG, Hudis CA, Giri D, Morrow M, Falcone DJ, Zhou XK, Du B, Brogi E, Crawford CB, Kopelovich L. Inflammation and increased aromatase expression occur in the breast tissue of obese women with breast cancer. *Cancer Prevention Research*. 2011;4:1021.
- 163. Cherry T. A theory of cancer. Medical Journal of Australia. 1922;1:425-438.
- 164. Sivertsen I, Dahlstrom A. The relation of muscular activity to carcinoma. A preliminary report. *Journal of Cancer Research*. 1922;6:365–378.

- 165. Hamburg NM, McMackin CJ, Huang AL, Shenouda SM, Widlansky ME, Schulz E, Gokce N, Ruderman NB, Keaney Jr JF, Vita JA. Physical inactivity rapidly induces insulin resistance and microvascular dysfunction in healthy volunteers. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2007;27:2650–2656.
- 166. Leesa SJ, Booth FW. Physical inactivity is a disease. *World Review of Nutrition and Dietetics*. 2005;95:73–79.
- 167. Booth FW, Laye MJ, Lees SJ, Rector RS, Thyfault JP. Reduced physical activity and risk of chronic disease: The biology behind the consequences. *European Journal of Applied Physiology*. 2008;102:381–390.
- 168. Powers SK, Duarte J, Kavazis AN, Talbert EE. Reactive oxygen species are signalling molecules for skeletal muscle adaptation. *Experimental Physiology*. 2010;95:1–9.
- 169. Stoll B. Western nutrition and the insulin resistance syndrome: A link to breast cancer. *European Journal of Clinical Nutrition*. 1999;53:83.
- Colangelo LA, Gapstur SM, Gann PH, Dyer AR, Liu K. Colorectal cancer mortality and factors related to the insulin resistance syndrome. *Cancer Epidemiology, Biomarkers & Prevention.* 2002;11:385–391.
- 171. Michaud DS, Liu S, Giovannucci E, Willett WC, Colditz GA, Fuchs CS. Dietary sugar, glycemic load, and pancreatic cancer risk in a prospective study. *Journal of the National Cancer Institute*. 2002;94:1293–1300.
- 172. Ainsworth BE, Sternfeld B, Slattery ML, Daguisé V, Zahm SH. Physical activity and breast cancer. *Cancer*. 1998;83:611–620.
- 173. Duncan K, Harris S, Ardies CM. Running exercise may reduce risk for lung and liver cancer by inducing activity of antioxidant and phase II enzymes. *Cancer Letters*. 1997;116:151–158.
- 174. Friedenreich CM. Physical activity and cancer prevention. *Cancer Epidemiology, Biomarkers & Prevention.* 2001;10:287–301.
- 175. Gregg EW, Cauley JA, Stone K, Thompson TJ, Bauer DC, Cummings SR, Ensrud KE. Relationship of changes in physical activity and mortality among older women. *Journal of the American Medical Association*. 2003;289:2379–2386.
- 176. Kiningham RB. Physical activity and the primary prevention of cancer. *Primary Care: Clinics in Office Practice*. 1998;25:515–536
- 177. Kohl HW, LaPorte RE, Blair SN. Physical activity and cancer. An epidemiological perspective. *Sports Medicine (Auckland, NZ)*. 1988;6:222.
- 178. Lee I. Physical activity and cancer prevention-data from epidemiologic studies. *Medicine* & Science in Sports & Exercise. 2003;35:1823.
- 179. Monninkhof EM, Elias SG, Vlems FA, van der Tweel I, Schuit AJ, Voskuil DW, van Leeuwen FE. Physical activity and breast cancer: A systematic review. *Epidemiology*. 2007;18:137.
- 180. Oliveria S, Christos P. The epidemiology of physical activity and cancer. *Annals of the New York Academy of Sciences*. 1997;833:79–90.
- Rogers CJ, Colbert LH, Greiner JW, Perkins SN, Hursting SD. Physical activity and cancer prevention: Pathways and targets for intervention. *Sports Medicine*. 2008;38:271–296.
- 182. Rundle A. Molecular epidemiology of physical activity and cancer. *Cancer Epidemiology, Biomarkers & Prevention.* 2005;14:227.
- 183. Shephard RJ, Futcher R. Physical activity and cancer: How may protection be maximized? *Critical Reviews in Oncogenesis.* 1997;8:219.
- 184. Sternfeld B. Cancer and the protective effect of physical activity: The epidemiological evidence. *Medicine & Science in Sports & Exercise*. 1992;24:1195.
- Wolin K, Yan Y, Colditz G, Lee I. Physical activity and colon cancer prevention: A metaanalysis. *British Journal of Cancer*. 2009;100:611–616.

- Consortium TEP. An integrated encyclopedia of DNA elements in the human genome. *Nature*. 2012;489:57–74.
- 187. Feingold E, Good P, Guyer M, Kamholz S, Liefer L, Wetterstrand K, Collins F, Gingeras T, Kampa D, Sekinger E. The encode (encyclopedia of DNA elements) project. *Science*. 2004;306:636–640.
- 188. Das PM, Singal R. DNA methylation and cancer. *Journal of Clinical Oncology*. 2004;22:4632–4642.
- McCabe MT, Brandes JC, Vertino PM. Cancer DNA methylation: Molecular mechanisms and clinical implications. *Clinical Cancer Research*. 2009;15:3927–3937.
- 190. Valinluck V, Tsai HH, Rogstad DK, Burdzy A, Bird A, Sowers LC. Oxidative damage to methyl-CpG sequences inhibits the binding of the methyl-CpG binding domain (MBD) of methyl-CpG binding protein 2 (MECP2). *Nucleic Acids Research*. 2004;32:4100–4108.
- Franco R, Schoneveld O, Georgakilas AG, Panayiotidis MI. Oxidative stress, DNA methylation and carcinogenesis. *Cancer Letters*. 2008;266:6–11.
- 192. Riely GJ, Marks J, Pao W. Kras mutations in non–small cell lung cancer. *Proceedings of the American Thoracic Society*. 2009;6:201–205.
- 193. Artale S, Sartore-Bianchi A, Veronese SM, Gambi V, Sarnataro CS, Gambacorta M, Lauricella C, Siena S. Mutations of KRAS and BRAF in primary and matched metastatic sites of colorectal cancer. *Journal of Clinical Oncology*. 2008;26:4217–4219.
- 194. Agarwal A, Eide C, Harlow A, Corbin A, Mauro M, Druker B, Corless C, Heinrich M, Deininger M. An activating KRAS mutation in imatinib-resistant chronic myeloid leukemia. *Leukemia*. 2008;22:2269–2272.
- 195. Croce CM. Oncogenes and cancer. New England Journal of Medicine. 2008;358:502-511.
- 196. Blackford A, Parmigiani G, Kensler TW, Wolfgang C, Jones S, Zhang X, Parsons DW, Lin JCH, Leary RJ, Eshleman JR. Genetic mutations associated with cigarette smoking in pancreatic cancer. *Cancer Research*. 2009;69:3681.
- 197. Brose MS, Volpe P, Feldman M, Kumar M, Rishi I, Gerrero R, Einhorn E, Herlyn M, Minna J, Nicholson A. BRAF and Ras mutations in human lung cancer and melanoma. *Cancer Research*. 2002;62:6997.
- 198. Davies H, Bignell GR, Cox C, Stephens P, Edkins S, Clegg S, Teague J, Woffendin H, Garnett MJ, Bottomley W. Mutations of the BRAF gene in human cancer. *Nature*. 2002;417:949–954.
- 199. Gorden A, Osman I, Gai W, He D, Huang W, Davidson A, Houghton AN, Busam K, Polsky D. Analysis of BRAF and n-Ras mutations in metastatic melanoma tissues. *Cancer Research*. 2003;63:3955.
- 200. Naoki K, Chen TH, Richards WG, Sugarbaker DJ, Meyerson M. Missense mutations of the BRAF gene in human lung adenocarcinoma. *Cancer Research*. 2002;62:7001.
- 201. Soung YH, Lee JW, Kim SY, Seo SH, Park WS, Nam SW, Song SY, Han JH, Park CK, Lee JY. Mutational analysis of EGFR and k-RAS genes in lung adenocarcinomas. *Virchows Archives*. 2005;446:483–488.
- Chan S, Gullick W, Hill M. Mutations of the epidermal growth factor receptor in nonsmall cell lung cancer—search and destroy. *European Journal of Cancer*. 2006;42:17–23.
- 203. Linardou H, Dahabreh IJ, Kanaloupiti D, Siannis F, Bafaloukos D, Kosmidis P, Papadimitriou CA, Murray S. Assessment of somatic k-Ras mutations as a mechanism associated with resistance to EGFR-targeted agents: A systematic review and meta-analysis of studies in advanced non-small-cell lung cancer and metastatic colorectal cancer. *Lancet Oncology*. 2008;9:962–972.
- 204. Hanahan D, Weinberg RA. Hallmarks of cancer: The next generation. *Cell*. 2011;144:646–674.
- 205. Hanahan D, Weinberg RA. The hallmarks of cancer. Cell. 2000;100:57-70.
- 206. Pleasance ED, Cheetham RK, Stephens PJ, McBride DJ, Humphray SJ, Greenman CD, Varela I, Lin ML, Ordóñez GR, Bignell GR. A comprehensive catalogue of somatic mutations from a human cancer genome. *Nature*. 2009;463:191–196.
- 207. Harris CC. P53 tumor suppressor gene: At the crossroads of molecular carcinogenesis, molecular epidemiology, and cancer risk assessment. *Environmental Health Perspectives*. 1996;104:435.
- Greenblatt M, Bennett W, Hollstein M, Harris C. Mutations in the p53 tumor suppressor gene: Clues to cancer etiology and molecular pathogenesis. *Cancer Research*. 1994;54:4855.
- 209. Dash BC, El-Deiry WS. Cell cycle checkpoint control mechanisms that can be disrupted in cancer. *Methods in Molecular Biology*. 2004;280:99–161.
- 210. Murphree AL, Benedict WF. Retinoblastoma: Clues to human oncogenesis. *Science*. 1984;223:1028.
- Shackelford RE, Kaufmann WK, Paules RS. Cell cycle control, checkpoint mechanisms, and genotoxic stress. *Environmental Health Perspectives*. 1999;107 Suppl 1:5–24.
- 212. Fang SC, de los Reyes C, Umen JG. Cell size checkpoint control by the retinoblastoma tumor suppressor pathway. *PLoS Genetics*. 2006;2:e167.
- 213. Sherr CJ. Gl phase progression: Cycling on cue. Cell. 1994;79:551–555.
- 214. Sherr CJ. Cancer cell cycles. Science. 1996;274:1672-1677.
- 215. Sherr CJ, Roberts JM. CDK inhibitors: Positive and negative regulators of G1-phase progression. *Genes and Development*. 1999;13:1501.
- 216. Shenghui H, Nakada D, Morrison SJ. Mechanisms of stem cell self-renewal. *Annual Review of Cell and Developmental Biology*. 2009;25:377–406.
- Orford KW, Scadden DT. Deconstructing stem cell self-renewal: Genetic insights into cell-cycle regulation. *Nature Reviews Genetics*. 2008;9:115–128.
- 218. Rodier F, Campisi J. Four faces of cellular senescence. *Journal of Cell Biology*. 2011;192:547–556.
- Greenhough A, Smartt HJM, Moore AE, Roberts HR, Williams AC, Paraskeva C, Kaidi A. The COX-2/PGE2 pathway: Key roles in the hallmarks of cancer and adaptation to the tumour microenvironment. *Carcinogenesis*. 2009;30:377–386.
- 220. Kriegler A, Bradley T, Hodgson G. The effect of prostaglandins E1 and E2 on macrophage progenitor cells with high proliferative potential in mouse bone marrow in vitro. *Blood.* 1984;63:1348–1352.
- 221. North TE, Goessling W, Walkley CR, Lengerke C, Kopani KR, Lord AM, Weber GJ, Bowman TV, Jang IH, Grosser T. Prostaglandin E2 regulates vertebrate haematopoietic stem cell homeostasis. *Nature*. 2007;447:1007–1011.
- Hoggatt J, Singh P, Sampath J, Pelus LM. Prostaglandin E2 enhances hematopoietic stem cell homing, survival, and proliferation. *Blood.* 2009;113:5444–5455.
- 223. Wu CH, Shih YW, Chang CH, Ou TT, Huang CC, Hsu JD, Wang CJ. EP4 upregulation of Ras signaling and feedback regulation of Ras in human colon tissues and cancer cells. *Archives of Toxicology*. 2010;84:731–740.
- 224. Memmott RM, Dennis PA. Akt-dependent and -independent mechanisms of mTOR regulation in cancer. *Cellular Signalling*. 2009;21:656–664.
- 225. Laplante M, Sabatini DM. mTOR signaling. Cold Spring Harbor Perspectives in Biology. 2012;41–3.
- 226. Nusse R. Wnt signaling and stem cell control. Cell Research. 2008;18:523-527.
- 227. van Es JH, Sato T, van de Wetering M, Lyubimova A, Nee ANY, Gregorieff A, Sasaki N, Zeinstra L, van den Born M, Korving J. DII1+ secretory progenitor cells revert to stem cells upon crypt damage. *Nature Cell Biology*. 2012;14:1099–1104.
- Mascré G, Dekoninck S, Drogat B, Youssef KK, Brohée S, Sotiropoulou PA, Simons BD, Blanpain C. Distinct contribution of stem and progenitor cells to epidermal maintenance. *Nature*. 2012;489:257–262.

- 229. Neganova I, Lako M. G1 to S phase cell cycle transition in somatic and embryonic stem cells. *Journal of Anatomy*. 2008;213:30–44.
- Bartek J, Lukas J. Pathways governing G1/S transition and their response to DNA damage. FEBS Letters. 2001;490:117–122.
- 231. Johnson D, Walker C. Cyclins and cell cycle checkpoints. Annual Review of Pharmacology and Toxicology. 1999;39:295–312.
- 232. Meyerson M, Harlow E. Identification of G1 kinase activity for cdk6, a novel cyclin D partner. *Molecular and Cellular Biology*. 1994;14:2077–2086.
- Matsushime H, Quelle D, Shurtleff S, Shibuya M, Sherr C, Kato J. D-type cyclindependent kinase activity in mammalian cells. *Molecular and Cellular Biology*. 1994;14:2066–2076.
- 234. Taira N, Mimoto R, Kurata M, Yamaguchi T, Kitagawa M, Miki Y, Yoshida K. DYRK2 priming phosphorylation of c-Jun and c-Myc modulates cell cycle progression in human cancer cells. *Journal of Clinical Investigation*. 2012;122:859.
- 235. Marshall C. How do small GTPase signal transduction pathways regulate cell cycle entry? *Current Opinion in Cell Biology*. 1999;11:732–736.
- 236. Girard F, Strausfeld U, Fernandez A, Lamb NJC. Cyclin A is required for the onset of DNA replication in mammalian fibroblasts. *Cell*. 1991;67:1169–1179.
- 237. Pagano M, Pepperkok R, Verde F, Ansorge W, Draetta G. Cyclin A is required at two points in the human cell cycle. *EMBO Journal*. 1992;11:961.
- 238. Zindy F, Lamas E, Chenivesse X, Sobczak J, Wang J, Fesquet D, Henglein B, Brechot C. Cyclin A is required in S phase in normal epithelial cells. *Biochemical and Biophysical Research Communications*. 1992;182:1144–1154.
- 239. Ohtsubo M, Theodoras AM, Schumacher J, Roberts JM, Pagano M. Human cyclin E, a nuclear protein essential for the G1 to S phase transition. *Molecular and Cellular Biology*. 1995;15:2612.
- 240. Laplante M, Sabatini DM. mTOR signaling at a glance. *Journal of Cell Science*. 2009;122:3589–3594.
- 241. Sengupta S, Peterson TR, Sabatini DM. Regulation of the mTOR complex 1 pathway by nutrients, growth factors, and stress. *Molecular Cell*. 2010;40:310–322.
- Schmit TL, Ahmad N. Regulation of mitosis via mitotic kinases: New opportunities for cancer management. *Molecular Cancer Therapeutics*. 2007;6:1920.
- 243. Easton J, Wei T, Lahti JM, Kidd VJ. Disruption of the cyclin D/cyclin-dependent kinase/ INK4/retinoblastoma protein regulatory pathway in human neuroblastoma. *Cancer Research*. 1998;58:2624.
- 244. Wolfel T, Hauer M, Schneider J, Serrano M, Wolfel C, Klehmann-Hieb E, De Plaen E, Hankeln T, Zum Buschenfelde KHM, Beach D. A p16INK4a-insensitive CDK4 mutant targeted by cytolytic Tlymphocytes in a human melanoma. *Science*. 1995;269:1281–1284.
- 245. Ortega S, Malumbres M, Barbacid M. Cyclin D-dependent kinases, INK4 inhibitors and cancer. *Biochimica et Biophysica Acta*. 2002;1602:73–87.
- 246. Elledge SJ. Cell cycle checkpoints: Preventing an identity crisis. *Science*. 1996;274:1664–1672.
- 247. Dai Y, Grant S. New insights into checkpoint kinase 1 in the DNA damage response signaling network. *Clinical Cancer Research*. 2010;16:376–383.
- 248. Medema RH, Macurek L. Checkpoint control and cancer. Oncogene. 2012;31:2601-2613.
- Warmerdam DO, Kanaar R. Dealing with DNA damage: Relationships between checkpoint and repair pathways. *Mutation Research*. 2010;704:2–11.
- 250. Lindahl T, Wood RD. Quality control by DNA repair. Science. 1999;286:1897.
- 251. Gudmundsdottir K, Ashworth A. The roles of BRCA1 and BRCA2 and associated proteins in the maintenance of genomic stability. *Oncogene*. 2006;25:5864–5874.
- 252. Wood RD, Mitchell M, Sgouros J, Lindahl T. Human DNA repair genes. *Science*. 2001;291:1284.

- Hefferin ML, Tomkinson AE. Mechanism of DNA double-strand break repair by nonhomologous end joining. DNA Repair. 2005;4:639–648.
- 254. Siliciano JD, Canman CE, Taya Y, Sakaguchi K, Appella E, Kastan MB. DNA damage induces phosphorylation of the amino terminus of p53. *Genes and Development*. 1997;11:3471.
- 255. Levine AJ. P53, the cellular gatekeeper review for growth and division. *Cell*. 1997;88:323–331.
- 256. van Vugt MATM, Bràs A, Medema RH. Restarting the cell cycle when the checkpoint comes to a halt. *Cancer Research*. 2005;65:7037.
- 257. Bansal P, Lazo JS. Induction of Cdc25B regulates cell cycle resumption after genotoxic stress. *Cancer Research*. 2007;67:3356.
- 258. Sheikh MS, Fornace Jr AJ. Role of p53 family members in apoptosis. *Journal of Cellular Physiology*. 2000;182:171–181.
- 259. Liebermann DA, Hoffman B, Steinman RA. Molecular controls of growth arrest and apoptosis: p53-dependent and independent pathways. *Oncogene*. 1995;11:199.
- Brown JM, Wouters BG. Apoptosis, p53, and tumor cell sensitivity to anticancer agents. *Cancer Research*. 1999;59:1391.
- Sionov RV, Haupt Y. The cellular response to p53: The decision between life and death. Oncogene. 1999;18:6145.
- 262. Fridman JS, Lowe SW. Control of apoptosis by p53. Oncogene. 2003;22:9030–9040.
- Morgenbesser SD, Williams BO, Jacks T, DePinho RA. p53-dependent apoptosis produced by Rb-deficiency in the developing mouse lens. 1994;371:72–74.
- 264. Li J, Yuan J. Caspases in apoptosis and beyond. Oncogene. 2008;27:6194-6206.
- 265. Elmore S. Apoptosis: A review of programmed cell death. *Toxicologic Pathology*. 2007;35:495–516.
- Orrenius S, Nicotera P, Zhivotovsky B. Cell death mechanisms and their implications in toxicology. *Toxicological Sciences*. 2011;119:3–19.
- Wyllie AH. "Where, O death, is thy sting?": A brief review of apoptosis biology. Molecular Neurobiology. 2010;42:4–9.
- 268. Wang C, Youle RJ. The role of mitochondria in apoptosis. *Annual Review of Genetics*. 2009;43:95–118.
- 269. Miyashita T, Krajewski S, Krajewska M, Wang HG, Lin H, Liebermann DA, Hoffman B, Reed JC. Tumor suppressor p53 is regulator of BCL-2 and BAX gene expression in vitro and in vivo. *Oncogene*. 1994;9:1799–1805.
- Miyashita T, Reed JC. Tumor suppressor p53 is a direct transcriptional activator of the human bax gene. *Cell*. 1995;80:293.
- 271. Wyllie A. Apoptosis: Cell death under homeostatic control. *Archives of Toxicology*. 1987;11:3.
- 272. He L, Perkins GA, Poblenz AT, Harris JB, Hung M, Ellisman MH, Fox DA. Bcl-xL overexpression blocks Bax-mediated mitochondrial contact site formation and apoptosis in rod photoreceptors of lead-exposed mice. *Proceedings of the National Academy of Sciences of the United States of America*. 2003;100:1022.
- 273. Ott M, Robertson JD, Gogvadze V, Zhivotovsky B, Orrenius S. Cytochrome c release from mitochondria proceeds by a two-step process. *Proceedings of the National Academy of Sciences of the United States of America*. 2002;99:1259.
- 274. Ott M, Zhivotovsky B, Orrenius S. Role of cardiolipin in cytochrome c release from mitochondria. *Cell Death and Differentiation*. 2007;14:1243–1247.
- Circu ML, Aw TY. Reactive oxygen species, cellular redox systems, and apoptosis. Free Radical Biology & Medicine. 2010;48:749–762.
- Brookes PS, Yoon Y, Robotham JL, Anders M, Sheu SS. Calcium, ATP, and ROS: A mitochondrial love-hate triangle. *American Journal of Physiology*. 2004;287:C817–C833.

- 277. Kastan MB, Onyekwere O, Sidransky D, Vogelstein B, Craig RW. Participation of p53 protein in the cellular response to DNA damage. *Cancer Research*. 1991;51:6304.
- 278. Levine AJ, Momand J, Finlay CA. The p53 tumour suppressor gene. *Nature*. 1991;351:453.
- Hollstein M, Sidransky D, Vogelstein B, Harris CC. p53 mutations in human cancers. Science. 1991;253:49.
- 280. Vogelstein B, Lane D, Levine AJ. Surfing the p53 network. Nature. 2000;408:307.
- Ziegler A, Jonason AS, Leffellt DJ, Simon JA, Sharma HW, Kimmelman J, Remington L, Jacks T, Brash DE. Sunburn and p53 in the onset of skin cancer. *Nature*. 1994;372:773–776.
- 282. Lane DP. Cancer. p53, guardian of the genome. Nature. 1992;358:15.
- Whibley C, Pharoah PDP, Hollstein M. p53 polymorphisms: Cancer implications. *Nature Reviews Cancer*. 2009;9:95–107.
- 284. Wang Z, Shen D, Parsons DW, Bardelli A, Sager J, Szabo S, Ptak J, Silliman N, Peters BA, Van Der Heijden MS. Mutational analysis of the tyrosine phosphatome in colorectal cancers. *Science*. 2004;304:1164.
- Pleasance ED, Stephens PJ, Oâ S. A small-cell lung cancer genome with complex signatures of tobacco exposure. *Nature*. 2009;463:184–190.
- 286. McLendon R, Friedman A, Bigner D, Van Meir EG, Brat DJ, Mastrogianakis GM, Olson JJ, Mikkelsen T, Lehman N, Aldape K. Comprehensive genomic characterization defines human glioblastoma genes and core pathways. *Nature*. 2008;455: 1061–1068.
- 287. Bignell GR, Greenman CD, Davies H, Butler AP, Edkins S, Andrews JM, Buck G, Chen L, Beare D, Latimer C. Signatures of mutation and selection in the cancer genome. *Nature*. 2010;463:893–898.
- 288. Reya T, Morrison SJ, Clarke MF, Weissman IL. Stem cells, cancer, and cancer stem cells. *Nature*. 2001;414:105–111.
- Wicha MS, Liu S, Dontu G. Cancer stem cells: An old idea—a paradigm shift. *Cancer Research*. 2006;66:1883.
- 290. Gupta PB, Chaffer CL, Weinberg RA. Cancer stem cells: Mirage or reality? *Nature Medicine*. 2009;15:1010–1012.
- 291. Balkwill F. Cancer and the chemokine network. Nature Reviews Cancer. 2004;4:540-550.
- Melillo R, Guarino V, Avilla E, Galdiero M, Liotti F, Prevete N, Rossi F, Basolo F, Ugolini C, de Paulis A. Mast cells have a protumorigenic role in human thyroid cancer. *Oncogene*. 2010;29:6203–6215.
- 293. Mojsilovic-Petrovic J, Callaghan D, Cui H, Dean C, Stanimirovic DB, Zhang W. Hypoxia-inducible factor-1 (HIF-1) is involved in the regulation of hypoxia-stimulated expression of monocyte chemoattractant protein-1 (MCP-1/CCL2) and MCP-5 (CCL12) in astrocytes. *Journal of Neuroinflammation*. 2007;4:12.
- 294. Ke Q, Costa M. Hypoxia-inducible factor-1 (HIF-1). *Molecular Pharmacology*. 2006;70:1469–1480.
- 295. Semenza GL. HIF-1: Mediator of physiological and pathophysiological responses to hypoxia. *Journal of Applied Physiology*. 2000;88:1474–1480.
- Semenza GL. HIF-1: Upstream and downstream of cancer metabolism. *Current Opinion* in Genetics and Development. 2010;20:51–56.
- 297. Lee PJ, Jiang BH, Chin BY, Iyer NV, Alam J, Semenza GL, Choi AMK. Hypoxiainducible factor-1 mediates transcriptional activation of the heme oxygenase-1 gene in response to hypoxia. *Journal of Biological Chemistry*. 1997;272:5375–5381.
- Borissoff JI, Spronk H, Ten Cate H. The hemostatic system as a modulator of atherosclerosis. *New England Journal of Medicine*. 2011;364:1746–1760.
- 299. May AE, Seizer P, Gawaz M. Platelets: Inflammatory firebugs of vascular walls. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2008;28:s5–s10.

- Lindemann S, Krämer B, Seizer P, Gawaz M. Platelets, inflammation and atherosclerosis. Journal of Thrombosis and Haemostasis. 2007;5:203–211.
- Diegelmann RF, Evans MC. Wound healing: An overview of acute, fibrotic and delayed healing. *Front Bioscience*. 2004;9:283–289.
- 302. Werner S, Grose R. Regulation of wound healing by growth factors and cytokines. *Physiological Reviews*. 2003;83:835–870.
- Werner S, Krieg T, Smola H. Keratinocyte–fibroblast interactions in wound healing. Journal of Investigative Dermatology. 2007;127:998–1008.
- Hanahan D, Coussens LM. Accessories to the crime: Functions of cells recruited to the tumor microenvironment. *Cancer Cell*. 2012;21:309–322.
- 305. Kumar P, Shen Q, Pivetti CD, Lee ES, Wu MH, Yuan SY. Molecular mechanisms of endothelial hyperpermeability: Implications in inflammation. *Expert Reviews in Molecular Medicine*. 2009;11:e19.
- Lu P, Weaver VM, Werb Z. The extracellular matrix: A dynamic niche in cancer progression. *Journal of Cell Biology*. 2012;196:395–406.
- Iyer A, Fairlie DP, Prins JB, Hammock BD, Brown L. Inflammatory lipid mediators in adipocyte function and obesity. *Nature Reviews Endocrinology*. 2010;6:71–82.
- González-Périz A, Clària J. Resolution of adipose tissue inflammation. Scientific World Journal. 2010;10:832–856.
- 309. Anderson EK, Gutierrez DA, Hasty AH. Adipose tissue recruitment of leukocytes. *Current Opinion in Lipidology*. 2010;21:172.
- 310. Ikeoka D, Mader JK, Pieber TR. Adipose tissue, inflammation and cardiovascular disease. *Revista da Associacao Medica Brasileira [Journal of the Brazilian Medical Association]*. 2010;56:116–121.
- 311. Basen-Engquist K, Chang M. Obesity and cancer risk: Recent review and evidence. *Current Oncology Reports*. 2011;13:71–76.
- 312. Collaboration PS. Body-mass index and cause-specific mortality in 900 000 adults: Collaborative analyses of 57 prospective studies. *Lancet*. 2009;373:1083–1096.
- 313. Lund Håheim L, Wisløff T, Holme I, Nafstad P. Metabolic syndrome predicts prostate cancer in a cohort of middle-aged Norwegian men followed for 27 years. *American Journal of Epidemiology*. 2006;164:769.
- 314. Rana J, Nieuwdorp M, Jukema J, Kastelein J. Cardiovascular metabolic syndrome— An interplay of obesity, inflammation, diabetes and coronary heart disease. *Diabetes, Obesity and Metabolism.* 2007;9:218–232.
- Facchini FS, Hua N, Abbasi F, Reaven GM. Insulin resistance as a predictor of age-related diseases. *Journal of Clinical Endocrinology and Metabolism*. 2001;86:3574–3578.
- Pedersen BK. The diseasome of physical inactivity—and the role of myokines in muscle—fat cross talk. *Journal of Physiology*. 2009;587:5559–5568.
- Powers SK, Kavazis AN, DeRuisseau KC. Mechanisms of disuse muscle atrophy: Role of oxidative stress. *American Journal of Physiology*. 2005;288:R337–R344.
- Powers SK, Kavazis AN, McClung JM. Oxidative stress and disuse muscle atrophy. Journal of Applied Physiology. 2007;102:2389–2397.
- Donohoe CL, Doyle SL, Reynolds JV. Visceral adiposity, insulin resistance and cancer risk. *Diabetology and Metabolic Syndrome*. 2011;3:12.
- 320. Milazzo G, Giorgino F, Damante G, Sung C, Stampfer MR, Vigneri R, Goldfine ID, Belfiore A. Insulin receptor expression and function in human breast cancer cell lines. *Cancer Research*. 1992;52:3924–3930.
- 321. Gallagher EJ, LeRoith D. Minireview: IGF, insulin, and cancer. *Endocrinology*. 2011;152:2546–2551.
- 322. Laviola L, Natalicchio A, Giorgino F. The IGF-1 signaling pathway. *Current Pharmaceutical Design*. 2007;13:663–669.

- 323. Smith T. Facilitative glucose transporter expression in human cancer tissue. *British Journal of Biomedical Science*. 1999;56:285.
- Macheda ML, Rogers S, Best JD. Molecular and cellular regulation of glucose transporter (GLUT) proteins in cancer. *Journal of Cellular Physiology*. 2005;202:654–662.
- 325. Barron C, Tsiani E, Tsakiridis T. Expression of the glucose transporters GLUT1, GLUT3, GLUT4 and GLUT12 in human cancer cells. *BMC Proceedings*. 2012;6:P4.
- 326. Cairns RA, Harris IS, Mak TW. Regulation of cancer cell metabolism. *Nature Reviews. Cancer*. 2011;11:85–95.
- 327. LeRoith D, Roberts Jr CT. The insulin-like growth factor system and cancer. *Cancer Letters*. 2003;195:127.
- 328. Belfiore A, Malaguarnera R. Insulin receptor and cancer. *Endocrine Related Cancer*. 2011;18:R125–R147.
- Schalkwijk CG, Stehouwer CDA, van Hinsbergh VWM. Fructose-mediated non-enzymatic glycation: Sweet coupling or bad modification. *Diabetes/Metabolism Research and Reviews*. 2004;20:369–382.
- 330. Bucala R, Makita Z, Koschinsky T, Cerami A, Vlassara H. Lipid advanced glycosylation: Pathway for lipid oxidation in vivo. *Proceedings of the National Academy of Sciences of the United States of America*. 1993;90:6434.
- 331. Goh SY, Cooper ME. The role of advanced glycation end products in progression and complications of diabetes. *Journal of Clinical Endocrinology and Metabolism*. 2008;93:1143–1152.
- 332. Makita Z, Yanagisawa K, Kuwajima S, Bucala R, Vlassara H, Koike T. The role of advanced glycosylation end-products in the pathogenesis of atherosclerosis. *Nephrology, Dialysis, Transplantation.* 1996;11:31–33.
- Barlovic D, Soro-Paavonen A, Jandeleit-Dahm K. RAGE biology, atherosclerosis and diabetes. *Clinical Science*. 2011;121:43–55.
- 334. Funk SD, Yurdagul A, Orr AW. Hyperglycemia and endothelial dysfunction in atherosclerosis: Lessons from type 1 diabetes. *International Journal of Vascular Medicine*. 2012;2012:569654.
- 335. van den Oever IAM, Raterman HG, Nurmohamed MT, Simsek S. Endothelial dysfunction, inflammation, and apoptosis in diabetes mellitus. *Mediators of Inflammation*. 2010;2010:792393.
- Oak JH, Nakagawa K, Miyazawa T. Synthetically prepared Amadori-glycated phosphatidylethanolamine can trigger lipid peroxidation via free radical reactions. *FEBS Letters*. 2000;481:26–30.
- 337. Ames BN. Endogenous oxidative DNA damage, aging, and cancer. *Free Radical Research*. 1989;7:121–128.
- 338. Bartkova J, Hořejší Z, Koed K, Krämer A, Tort F, Zieger K, Guldberg P, Sehested M, Nesland JM, Lukas C. DNA damage response as a candidate anti-cancer barrier in early human tumorigenesis. *Nature*. 2005;434:864–870.
- Loft S, Poulsen H. Cancer risk and oxidative DNA damage in man. *Journal of Molecular Medicine*. 1996;74:297–312.
- 340. Pacher P, Beckman JS, Liaudet L. Nitric oxide and peroxynitrite in health and disease. *Physiological Reviews*. 2007;87:315–424.
- McCord JM. Free radicals and inflammation: Protection of synovial fluid by superoxide dismutase. *Science*. 1974;185:529.
- 342. Quijano C, Hernandez-Saavedra D, Castro L, McCord JM, Freeman BA, Radi R. Reaction of peroxynitrite with Mn-superoxide dismutase. *Journal of Biological Chemistry*. 2001;276:11631–11638.
- Salin M, McCord J. Free radicals and inflammation. Protection of phagocytosine leukocytes by superoxide dismutase. *Journal of Clinical Investigation*. 1975;56:1319.

- 344. Halliwell B. Reactive oxygen species in living systems: Source, biochemistry, and role in human disease. *American Journal of Medicine*. 1991;91:S14–S22.
- 345. Halliwell B, Cross CE. Oxygen-derived species: Their relation to human disease and environmental stress. *Environmental Health Perspectives*. 1994;102:5.
- Halliwell B, Gutteridge J. Oxygen toxicity, oxygen radicals, transition metals and disease. *Biochemical Journal*. 1984;219:1.
- 347. Beckman JS, Koppenol WH. Nitric oxide, superoxide, and peroxynitrite: The good, the bad, and ugly. *American Journal of Physiology*. 1996;271:C1424–C1437.
- 348. Valko M, Rhodes C, Moncol J, Izakovic M, Mazur M. Free radicals, metals and antioxidants in oxidative stress-induced cancer. *Chemico-Biological Interactions*. 2006;160:1–40.
- Cooke MS, Evans MD, Dizdaroglu M, Lunec J. Oxidative DNA damage: Mechanisms, mutation, and disease. *FASEB Journal*. 2003;17:1195–1214.
- Dizdaroglu M, Jaruga P. Mechanisms of free radical-induced damage to DNA. Free Radical Research. 2012:1–73.
- 351. Von Sonntag C. The chemistry of free-radical-mediated DNA damage. *Basic Life Sciences*. 1991;58:287.
- 352. Balasubramanian B, Pogozelski WK, Tullius TD. DNA strand breaking by the hydroxyl radical is governed by the accessible surface areas of the hydrogen atoms of the DNA backbone. *Proceedings of the National Academy of Sciences of the United States of America.* 1998;95:9738–9743.
- 353. Aust AE, Eveleigh JF. Mechanisms of DNA oxidation. 1999;222:246-252.
- 354. Marnett LJ. Oxyradicals and DNA damage. Carcinogenesis. 2000;21:361-370.
- 355. Marnett LJ, Riggins JN, West JD. Endogenous generation of reactive oxidants and electrophiles and their reactions with DNA and protein. *Journal of Clinical Investigation*. 2003;111:583–594.
- Valko M, Izakovic M, Mazur M, Rhodes CJ, Telser J. Role of oxygen radicals in DNA damage and cancer incidence. *Molecular and Cellular Biochemistry*. 2004;266:37–56.
- 357. Łuczaj W, Skrzydlewska E. DNA damage caused by lipid peroxidation products. *Cellular and Molecular Biology Letters*. 2003;8:391.
- 358. Chung FL, Chen HJC, Nath RG. Lipid peroxidation as a potential endogenous source for the formation of exocyclic DNA adducts. *Carcinogenesis*. 1996;17:2105–2111.
- Chaudhary AK, Nokubo M, Reddy GR, Yeola SN, Morrow JD, Blair IA, Marnett LJ. Detection of endogenous malondialdehyde-deoxyguanosine adducts in human liver. *Science*. 1994;265:1580.
- De Bont R, Van Larebeke N. Endogenous DNA damage in humans: A review of quantitative data. *Mutagenesis*. 2004;19:169–185.
- 361. Stone MP, Cho YJ, Huang H, Kim HY, Kozekov ID, Kozekova A, Wang H, Minko IG, Lloyd RS, Harris TM. Interstrand DNA cross-links induced by α, β-unsaturated aldehydes derived from lipid peroxidation and environmental sources. Accounts of Chemical Research. 2008;41:793–804.
- 362. Hiraku Y. Formation of 8-nitroguanine, a nitrative DNA lesion, in inflammation-related carcinogenesis and its significance. *Environmental Health and Preventive Medicine*. 2010;15:63–72.
- 363. Yu H, Venkatarangan L, Wishnok JS, Tannenbaum SR. Quantitation of four guanine oxidation products from reaction of DNA with varying doses of peroxynitrite. *Chemical Research in Toxicology*. 2005;18:1849–1857.
- 364. Ames BN, Profet M, Gold LS. Dietary pesticides (99.99% all natural). Proceedings of the National Academy of Sciences of the United States of America. 1990;87:777–7781.
- 365. Gullett BK, Touati A, Hays MD. PCDD/F, PCB, HxCBz, PAH, and PM emission factors for fireplace and woodstove combustion in the San Francisco Bay region. *Environmental Science and Technology*. 2003;37:1758–1765.

- 366. Grimmer G, Naujack KW, Dettbarn G. Gas chromatographic determination of polycyclic aromatic hydrocarbons, aza-arenes, aromatic amines in the particle and vapor phase of mainstream and sidestream smoke of cigarettes. *Toxicology Letters*. 1987;35:117–124.
- 367. Stedman RL. Chemical composition of tobacco and tobacco smoke. *Chemical Reviews*. 1968;68:153–207.
- 368. Nordholm L, Espensen IM, Jensen HS, Holst E. Polycyclic aromatic hydrocarbons in smokehouses. *Scandinavian Journal of Work, Environment and Health*. 1986;12:614–618.
- Gomaa EA, Gray JI, Rabie S, Lopez-Bote C, Booren AM. Polycyclic aromatic hydrocarbons in smoked food products and commercial liquid smoke flavourings. *Food Additives* and Contaminants. 1993;10:503–521.
- 370. Pfeifer GP, Denissenko MF, Olivier M, Tretyakova N, Hecht SS, Hainaut P. Tobacco smoke carcinogens, DNA damage and p53 mutations in smoking-associated cancers. *Oncogene*. 2002;21:7435–7451.
- 371. Pfeifer GP. Environmental exposures and mutational patterns of cancer genomes. *Genome Medicine*. 2010;2:54.
- 372. Meij R, te Winkel H. The emissions of heavy metals and persistent organic pollutants from modern coal-fired power stations. *Atmospheric Environment*. 2007;41:9262–9272.
- 373. Dyke PH, Foan C, Fiedler H. PCB and PAH releases from power stations and waste incineration processes in the UK. *Chemosphere*. 2003;50:469–480.
- 374. Perera FP, Mooney L, Dickey CP, Santella RM, Bell D, Blaner W, Tang D, Whyatt RM. Molecular epidemiology in environmental carcinogenesis. *Environmental Health Perspectives*. 1996;104:441.
- 375. Perera FP. Molecular cancer epidemiology: A new tool in cancer prevention. *Journal of the National Cancer Institute*. 1987;78:887.
- 376. Goldstein B, Gibson J, Henderson R, Hobbie J, Landrigan P, Mattison D, Perera E, Sübergeld E, Wogan G. Biological markers in environmental health research. *Environmental Health Perspectives*. 1987;74:3–9.
- 377. Wogan GN, Hecht SS, Felton JS, Conney AH, Loeb LA. Environmental and chemical carcinogenesis. *Seminars in Cancer Biology*. 2004;14:473–486.
- 378. Shields PG, Harris CC. Molecular epidemiology and the genetics of environmental cancer. *Journal of the American Medical Association*. 1991;266:681–687.
- Nebert DW, Dalton TP. The role of cytochrome P450 enzymes in endogenous signalling pathways and environmental carcinogenesis. *Nature Reviews Cancer*. 2006;6:947–960.
- 380. Jha P. Avoidable global cancer deaths and total deaths from smoking. *Nature Reviews Cancer*. 2009;9:655–664.
- 381. King BA, Dube SR, Tynan MA. Current tobacco use among adults in the United States: Findings from the National Adult Tobacco Survey. *American Journal of Public Health*. 2012;102:e93–e100.
- Hecht SS. Tobacco smoke carcinogens and lung cancer. *Journal of the National Cancer Institute*. 1999;91:1194–1210.
- 383. Nebert DW, Adesnik M, Coon MJ, Estabrook RW, Gonzalez FJ, Guengerich FP, Gunsalus IC, Johnson EF, Kemper B, Levin W. The P450 gene superfamily: Recommended nomenclature. DNA. 1987;6:1–11.
- 384. Yang X, Zhang B, Molony C, Chudin E, Hao K, Zhu J, Gaedigk A, Suver C, Zhong H, Leeder JS. Systematic genetic and genomic analysis of cytochrome P450 enzyme activities in human liver. *Genome Research*. 2010;20:1020–1036.
- Meunier B, De Visser SP, Shaik S. Mechanism of oxidation reactions catalyzed by cytochrome P450 enzymes. *Chemical Reviews-Columbus*. 2004;104:3947–3980.
- 386. Puntarulo S, Cederbaum AI. Production of reactive oxygen species by microsomes enriched in specific human cytochrome P450 enzymes. *Free Radical Biology and Medicine*. 1998;24:1324–1330.

- Anzenbacher P, Anzenbacherova E. Cytochromes P450 and metabolism of xenobiotics. Cellular and Molecular Life Sciences. 2001;58:737–747.
- Hecht SS. Biochemistry, biology, and carcinogenicity of tobacco-specific n-nitrosamines. *Chemical Research in Toxicology*. 1998;11:559–603.
- 389. Essigmann J, Croy R, Nadzan A, Busby W, Reinhold V, Büchi G, Wogan G. Structural identification of the major DNA adduct formed by aflatoxin B1 in vitro. *Proceedings of the National Academy of Sciences of the United States of America*. 1977;74:1870.
- 390. Guengerich FP, Johnson WW, Shimada T, Ueng YF, Yamazaki H, Langouët S. Activation and detoxification of aflatoxin B1. *Mutation Research*. 1998;402:121–128.
- Kensler TW, Roebuck BD, Wogan GN, Groopman JD. Aflatoxin: A 50-year odyssey of mechanistic and translational toxicology. *Toxicological Sciences*. 2011;120:S28–S48.
- Nigam S, Ghosh S, Malaviya R. Aflatoxin, its metabolism and carcinogenesis—A historical review. *Toxin Reviews*. 1994;13:179–203.
- 393. Williams JH, Phillips TD, Jolly PE, Stiles JK, Jolly CM, Aggarwal D. Human aflatoxicosis in developing countries: A review of toxicology, exposure, potential health consequences, and interventions. *American Journal of Clinical Nutrition*. 2004;80:1106–1122.
- 394. Bastide NM, Pierre FHF, Corpet DE. Heme iron from meat and risk of colorectal cancer: A meta-analysis and a review of the mechanisms involved. *Cancer Prevention Research*. 2011;4:177–184.
- 395. Cross AJ, Freedman ND, Ren J, Ward MH, Hollenbeck AR, Schatzkin A, Sinha R, Abnet CC. Meat consumption and risk of esophageal and gastric cancer in a large prospective study. *American Journal of Gastroenterology*. 2011;106:432–442.
- 396. Sandhu MS, White IR, McPherson K. Systematic review of the prospective cohort studies on meat consumption and colorectal cancer risk: A meta-analytical approach. *Cancer Epidemiology, Biomarkers and Prevention*. 2001;10:439–446.
- 397. Norat T, Bingham S, Ferrari P, Slimani N, Jenab M, Mazuir M, Overvad K, Olsen A, Tjønneland A, Clavel F. Meat, fish, and colorectal cancer risk: The European Prospective Investigation into cancer and nutrition. *Journal of the National Cancer Institute*. 2005;97:906–916.
- 398. Goldman R, Shields PG. Food mutagens. Journal of Nutrition. 2003;133:965S-973S.
- 399. Hord NG, Tang Y, Bryan NS. Food sources of nitrates and nitrites: The physiologic context for potential health benefits. *American Journal of Clinical Nutrition*. 2009;90:1–10.
- 400. Schut HAJ, Snyderwine EG. DNA adducts of heterocyclic amine food mutagens: Implications for mutagenesis and carcinogenesis. *Carcinogenesis*. 1999;20:353–368.
- 401. Shields PG. Molecular epidemiology of smoking and lung cancer. *Oncogene*. 2002;21:6870–6876.
- 402. Alexandrov K, Cascorbi I, Rojas M, Bouvier G, Kriek E, Bartsch H. CYP1A1 and GSTM1 genotypes affect benzo [a] pyrene DNA adducts in smokers' lung: Comparison with aromatic/hydrophobic adduct formation. *Carcinogenesis*. 2002;23:1969–1977.
- Hecht SS. Human urinary carcinogen metabolites: Biomarkers for investigating tobacco and cancer. *Carcinogenesis*. 2002;23:907.
- 404. Ma XL, Baraona E, Lasker JM, Lieber CS. Effects of ethanol consumption on bioactivation and hepatotoxicity of *n*-nitrosodimethylamine in rats. *Biochemical Pharmacology*. 1991;42:585–591.
- 405. Ardies CM, Smith TJ, Kim S, Yang CS. Induction of 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanone (NNK) activation in rat lung microsomes by chronic ethanol consumption and repeated running exercise. *Cancer Letters*. 1996;103:209–218.
- 406. Bagnardi V, Randi G, Lubin J, Consonni D, Lam TK, Subar AF, Goldstein AM, Wacholder S, Bergen AW, Tucker MA. Alcohol consumption and lung cancer risk in the Environment and Genetics in Lung Cancer Etiology (EAGLE) study. *American Journal* of Epidemiology. 2010;171:36–44.

- 407. Jin M, Cai S, Guo J, Zhu Y, Li M, Yu Y, Zhang S, Chen K. Alcohol drinking and all cancer mortality: A meta-analysis. *Annals of Oncology*. 2013;24:807–816.
- 408. Garg R, Gupta S, Maru GB. Dietary curcumin modulates transcriptional regulators of phase I and phase II enzymes in benzo [a] pyrene-treated mice: Mechanism of its antiinitiating action. *Carcinogenesis*. 2008;29:1022–1032.
- 409. Kundu JK, Surh YJ. Molecular basis of chemoprevention with dietary phytochemicals: Redox-regulated transcription factors as relevant targets. *Phytochemistry Reviews*. 2009;8:333–347.
- 410. Liu RH. Potential synergy of phytochemicals in cancer prevention: Mechanism of action. *Journal of Nutrition*. 2004;134:3479S.
- 411. Surh YJ, Kundu JK, Na HK. Nrf2 as a master redox switch in turning on the cellular signaling involved in the induction of cytoprotective genes by some chemopreventive phytochemicals. *Planta Medica*. 2008;74:1526.
- 412. Perera FP, Weinstein IB. Molecular epidemiology and carcinogen-DNA adduct detection: New approaches to studies of human cancer causation. *Journal of Chronic Diseases*. 1982;35:581.
- Perera FP, Weinstein IB. Molecular epidemiology: Recent advances and future directions. *Carcinogenesis*. 2000;21:517–524.
- 414. Perera FP. Molecular epidemiology: Insights into cancer susceptibility, risk assessment, and prevention. *Journal of the National Cancer Institute*. 1996;88:496.
- 415. Vineis P, Perera F. Molecular epidemiology and biomarkers in etiologic cancer research: The new in light of the old. *Cancer Epidemiology, Biomarkers and Prevention*. 2007;16:1954–1965.
- Hussain SP, Harris CC. Molecular epidemiology of human cancer: Contribution of mutation spectra studies of tumor suppressor genes. *Cancer Research*. 1998;58:4023.
- 417. Lu K, Gul H, Upton PB, Moeller BC, Swenberg JA. Formation of hydroxymethyl DNA adducts in rats orally exposed to stable isotope labeled methanol. *Toxicological Sciences*. 2012;126:28–38.
- 418. Swenberg JA, Lu K, Moeller BC, Gao L, Upton PB, Nakamura J, Starr TB. Endogenous versus exogenous DNA adducts: Their role in carcinogenesis, epidemiology, and risk assessment. *Toxicological Sciences*. 2011;120:S130–S145.
- 419. Ames BN, Gold LS, Willett WC. The causes and prevention of cancer. *Proceedings of the National Academy of Sciences of the United States of America*. 1995;92:5258–5265.
- 420. Prüss-Üstün A, Corvalán C. How much disease burden can be prevented by environmental interventions? *Epidemiology*. 2007;18:167–178.
- 421. Wild CP. Environmental exposure measurement in cancer epidemiology. *Mutagenesis*. 2009;24:117–125.
- 422. Vineis P, Husgafvel-Pursiainen K. Air pollution and cancer: Biomarker studies in human populations. *Carcinogenesis*. 2005;26:1846–1855.
- 423. Huff J, Melnick RL. Occupational chemical carcinogenesis. *Patty's Toxicology*. 2012:449–494.
- 424. Dees C, Askari M, Garrett S, Gehrs K, Henley D, Ardies CM. Estrogenic and DNAdamaging activity of Red No. 3 in human breast cancer cells. *Environmental Health Perspectives*. 1997;105 Suppl 3:625–632.
- 425. Ardies CM, Dees C. Xenoestrogens significantly enhance risk for breast cancer during growth and adolescence. *Medical Hypotheses*. 1998;50:457–464.
- 426. Brophy JT, Keith MM, Watterson A, Park R, Gilbertson M, Maticka-Tyndale E, Beck M, Abu-Zahra H, Schneider K, Reinhartz A. Breast cancer risk in relation to occupations with exposure to carcinogens and endocrine disruptors: A Canadian case–control study. *Environmental Health*. 2012;11:1–17.
- 427. Birnbaum LS, Fenton SE. Cancer and developmental exposure to endocrine disruptors. *Environmental Health Perspectives*. 2003;111:389.

- 428. Casals-Casas C, Desvergne B. Endocrine disruptors: From endocrine to metabolic disruption. *Annual Review of Physiology*. 2011;73:135–162.
- 429. Soto AM, Sonnenschein C. Environmental causes of cancer: Endocrine disruptors as carcinogens. *Nature Reviews Endocrinology*. 2010;6:363–370.
- 430. Doherty LF, Bromer JG, Zhou Y, Aldad TS, Taylor HS. In utero exposure to diethylstilbestrol (DES) or bisphenol-A (BPA) increases EZH2 expression in the mammary gland: An epigenetic mechanism linking endocrine disruptors to breast cancer. *Hormones and Cancer*. 2010;1:146–155.
- 431. Matthews J, Wihlén B, Thomsen J, Gustafsson J-Å. Aryl hydrocarbon receptormediated transcription: Ligand-dependent recruitment of estrogen receptor α to 2, 3, 7, 8-tetrachlorodibenzo-p-dioxin-responsive promoters. *Molecular and Cellular Biology*. 2005;25:5317–5328.
- 432. Boverhof DR, Kwekel JC, Humes DG, Burgoon LD, Zacharewski TR. Dioxin induces an estrogen-like, estrogen receptor-dependent gene expression response in the murine uterus. *Molecular Pharmacology*. 2006;69:1599–1606.
- 433. Hsieh T, Lu X, Wang Z, Wu JM. Induction of quinone reductase NQO1 by resveratrol in human K562 cells involves the antioxidant response element ARE and is accompanied by nuclear translocation of transcription factor Nrf2. *Medicinal Chemistry*. 2006;2:275–285.
- 434. Hsieh T-H, Tsai C-F, Hsu C-Y, Kuo P-L, Lee J-N, Chai C-Y, Wang S-C, Tsai E-M. Phthalates induce proliferation and invasiveness of estrogen receptornegative breast cancer through the AHR/HDAC6/c-Myc signaling pathway. *FASEB Journal*. 2012;26:778–787.
- 435. Huang C-S, Chern H-D, Chang K-J, Cheng C-W, Hsu S-M, Shen C-Y. Breast cancer risk associated with genotype polymorphism of the estrogen-metabolizing genes CYP17, CYP1A1, and COMT: A multigenic study on cancer susceptibility. *Cancer Research*. 1999;59:4870–4875.
- 436. Prins GS. Endocrine disruptors and prostate cancer risk. *Endocrine Related Cancer*. 2008;15:649–656.
- 437. Hu W-Y, Shi G-B, Hu D-P, Nelles JL, Prins GS. Actions of estrogens and endocrine disrupting chemicals on human prostate stem/progenitor cells and prostate cancer risk. *Molecular and Cellular Endocrinology*. 2012;354:63–73.
- 438. Pierce JP. International comparisons of trends in cigarette smoking prevalence. *American Journal of Public Health.* 1989;79:152–157.
- 439. Fortmann SP, Taylor CB, Flora JA, Jatulis DE. Changes in adult cigarette smoking prevalence after 5 years of community health education: The Stanford Five-City Project. *American Journal of Epidemiology*. 1993;137:82–96.
- 440. Friend K, Levy DT. Reductions in smoking prevalence and cigarette consumption associated with mass-media campaigns. *Health Education Research*. 2002;17:85–98.
- 441. Marnett LJ. Lipid peroxidation—DNA damage by malondialdehyde. *Mutation Research*. 1999;424:83–95.
- 442. Fulgoni III VL, Keast DR, Bailey RL, Dwyer J. Foods, fortificants, and supplements: Where do Americans get their nutrients? *Journal of Nutrition*. 2011;141:1847–1854.
- 443. Kimmons J, Gillespie C, Seymour J, Serdula M, Blanck HM. Fruit and vegetable intake among adolescents and adults in the united states: Percentage meeting individualized recommendations. *Medscape Journal of Medicine*. 2009;11:26.
- 444. Schleicher RL, Carroll MD, Ford ES, Lacher DA. Serum vitamin C and the prevalence of vitamin C deficiency in the United States: 2003–2004 National Health and Nutrition Examination Survey (NHANES). *American Journal of Clinical Nutrition*. 2009;90:1252–1263.
- 445. Ames BN. Prevention of mutation, cancer, and other age-associated diseases by optimizing micronutrient intake. *Journal of Nucleic Acids*. 2010;2010:725071.

Cancer

- 446. Ames BN. Low micronutrient intake may accelerate the degenerative diseases of aging through allocation of scarce micronutrients by triage. *Proceedings of the National Academy of Sciences of the United States of America*. 2006;103:17589–17594.
- 447. Schlueter AK, Johnston CS. Vitamin C: Overview and update. *Journal of Evidence-Based Complementary and Alternative Medicine*. 2011;16:49–57.
- 448. Burton GW, Traber MG. Vitamin E: Antioxidant activity, biokinetics, and bioavailability. *Annual Review of Nutrition*. 1990;10:357–382.
- 449. Packer JE, Slater T, Willson R. Direct observation of a free radical interaction between vitamin E and vitamin C. 1979;278:737–738.
- 450. Nordberg J, Arnér ESJ. Reactive oxygen species, antioxidants, and the mammalian thioredoxin system. *Free Radical Biology and Medicine*. 2001;31:1287–1312.
- 451. Lillig CH, Berndt C, Holmgren A. Glutaredoxin systems. *Biochimica et Biophysica Acta*. 2008;1780:1304–1317.
- 452. Winyard PG, Moody CJ, Jacob C. Oxidative activation of antioxidant defence. *Trends in Biochemical Sciences*. 2005;30:453–461.
- 453. Nathan C. Specificity of a third kind: Reactive oxygen and nitrogen intermediates in cell signaling. *Journal of Clinical Investigation*. 2003;111:769–778.
- 454. Thannickal VJ, Fanburg BL. Reactive oxygen species in cell signaling. *American Journal of Physiology*. 2000;279:L1005–L1028.
- 455. Morgan MJ, Liu Z. Crosstalk of reactive oxygen species and NF-κB signaling. *Cell Research*. 2010;21:103–115.
- 456. Sun QA, Wu Y, Zappacosta F, Jeang KT, Lee BJ, Hatfield DL, Gladyshev VN. Redox regulation of cell signaling by selenocysteine in mammalian thioredoxin reductases. *Journal of Biological Chemistry*. 1999;274:24522.
- 457. Winrow V, Winyard P, Morris C, Blake D. Free radicals in inflammation: Second messengers and mediators of tissue destruction. *British Medical Bulletin*. 1993;49:506.
- 458. Finkel T. Signal transduction by reactive oxygen species. *Journal of Cell Biology*. 2011;194:7.
- 459. Veal EA, Day AM, Morgan BA. Hydrogen peroxide sensing and signaling. *Molecular Cell*. 2007;26:1–14.
- 460. Halliwell B. Reactive species and antioxidants. Redox biology is a fundamental theme of aerobic life. *Plant Physiology*. 2006;141:312–322.
- 461. Ueno M, Masutani H, Arai RJ, Yamauchi A, Hirota K, Sakai T, Inamoto T, Yamaoka Y, Yodoi J, Nikaido T. Thioredoxin-dependent redox regulation of p53-mediated p21 activation. *Journal of Biological Chemistry*. 1999;274:35809–35815.
- Karlenius TC, Tonissen KF. Thioredoxin and cancer: A role for thioredoxin in all states of tumor oxygenation. *Cancers*. 2010;2:209–232.
- 463. Martin KR, Appel CL. Polyphenols as dietary supplements: A double-edged sword. *Nutrition and Dietary Supplements*. 2010;2:1–12.
- 464. Vikman P, Xu CB, Edvinsson L. Lipid-soluble cigarette smoking particles induce expression of inflammatory and extracellular-matrix-related genes in rat cerebral arteries. *Vascular Health and Risk Management*. 2009;5:333.
- 465. Szulakowski P, Crowther AJL, Jiménez LA, Donaldson K, Mayer R, Leonard TB, MacNee W, Drost EM. The effect of smoking on the transcriptional regulation of lung inflammation in patients with chronic obstructive pulmonary disease. *American Journal* of Respiratory and Critical Care Medicine. 2006;174:41–50.
- 466. Ristow M, Zarse K, Oberbach A, Klöting N, Birringer M, Kiehntopf M, Stumvoll M, Kahn CR, Blüher M. Antioxidants prevent health-promoting effects of physical exercise in humans. *Proceedings of the National Academy of Sciences of the United States of America*. 2009;106:8665.
- 467. Funk CD. Prostaglandins and leukotrienes: Advances in eicosanoid biology. *Science*. 2001;294:1871.

- Evans JH, Spencer DM, Zweifach A, Leslie CC. Intracellular calcium signals regulating cytosolic phospholipase A2 translocation to internal membranes. *Journal of Biological Chemistry*. 2001;276:30150.
- 469. Bonventre JV, Swidler M. Calcium dependency of prostaglandin E2 production in rat glomerular mesangial cells. Evidence that protein kinase C modulates the Ca²⁺-dependent activation of phospholipase A2. *Journal of Clinical Investigation*. 1988;82:168.
- 470. Handlogten ME, Huang C, Shiraishi N, Awata H, Miller RT. The Ca²⁺-sensing receptor activates cytosolic phospholipase A2 via a GQα-dependent ERK-independent pathway. *Journal of Biological Chemistry*. 2001;276:13941.
- 471. Zor U, Ferber E, Gergely P, Szücs K, Dombradi V, Goldman R. Reactive oxygen species mediate phorbol ester-regulated tyrosine phosphorylation and phospholipase A2 activation: Potentiation by vanadate. *Biochemical Journal*. 1993;295:879.
- 472. Musso CG. Magnesium metabolism in health and disease. *International Urology and Nephrology*. 2009;41:357–362.
- 473. Sahota O, Mundey M, San P, Godber I, Hosking D. Vitamin D insufficiency and the blunted PTH response in established osteoporosis: The role of magnesium deficiency. *Osteoporosis International*. 2006;17:1013–1021.
- 474. Freitag JJ, Martin KJ, Conrades MB, Bellorin-Font E, Teitelbaum S, Klahr S, Slatopolsky E. Evidence for skeletal resistance to parathyroid hormone in magnesium deficiency: Studies in isolated perfused bone. *Journal of Clinical Investigation*. 1979;64:1238.
- 475. Kumar R. Metabolism of 1, 25-dihydroxyvitamin d3. Physiological Reviews. 1984;64:478-504.
- 476. Rude RK, Singer FR, Gruber HE. Skeletal and hormonal effects of magnesium deficiency. *Journal of the American College of Nutrition*. 2009;28:131–141.
- 477. Garland CF, Gorham ED, Mohr SB, Garland FC. Vitamin D for cancer prevention: Global perspective. *Annals of Epidemiology*. 2009;19:468–483.
- 478. Krishnan AV, Trump DL, Johnson CS, Feldman D. The role of vitamin D in cancer prevention and treatment. *Endocrinology and Metabolism Clinics of North America*. 2010;39:401.
- 479. Lappe JM, Travers-Gustafson D, Davies KM, Recker RR, Heaney RP. Vitamin D and calcium supplementation reduces cancer risk: Results of a randomized trial. *American Journal of Clinical Nutrition*. 2007;85:1586–1591.
- 480. Peterlik M, Cross H. Vitamin D and calcium deficits predispose for multiple chronic diseases. *European Journal of Clinical Investigation*. 2005;35:290–304.
- 481. Lin J, Cook NR, Albert C, Zaharris E, Gaziano JM, Van Denburgh M, Buring JE, Manson JAE. Vitamins C and E and beta carotene supplementation and cancer risk: A randomized controlled trial. *Journal of the National Cancer Institute*. 2009;101:14–23.
- 482. Malila N, Taylor PR, Virtanen MJ, Korhonen P, Huttunen JK, Albanes D, Virtamo J. Effects of alpha-tocopherol and beta-carotene supplementation on gastric cancer incidence in male smokers (ATBC study, Finland). *Cancer Causes and Control*. 2002;13:617–623.
- 483. Omenn GS, Goodman GE, Thornquist MD, Balmes J, Cullen MR, Glass A, Keogh JP, Meyskens Jr FL, Valanis B, Williams Jr JH. Risk factors for lung cancer and for intervention effects in CARET, the Beta-Carotene and Retinol Efficacy Trial. *Journal of the National Cancer Institute*. 1996;88:1550–1559.
- 484. Miller III ER, Pastor-Barriuso R, Dalal D, Riemersma RA, Appel LJ, Guallar E. Metaanalysis: High-dosage vitamin E supplementation may increase all-cause mortality. *Annals of Internal Medicine*. 2005;142:37–46.
- 485. Lonn E, Bosch J, Yusuf S, Sheridan P, Pogue J, Arnold J, Ross C, Arnold A, Sleight P, Probstfield J. Effects of long-term vitamin E supplementation on cardiovascular events and cancer: A randomized controlled trial. *Journal of the American Medical Association*. 2005;293:1338.

- 486. Goodman M, Bostick RM, Kucuk O, Jones DP. Clinical trials of antioxidants as cancer prevention agents: Past, present, and future. *Free Radical Biology and Medicine*. 2011;51:1068.
- 487. Albanes D, Heinonen OP, Taylor PR, Virtamo J, Edwards BK, Rautalahti M, Hartman AM, Palmgren J, Freedman LS, Haapakoski J. A-tocopherol and β-carotene supplements and lung cancer incidence in the alpha-tocopherol, beta-carotene cancer prevention study: Effects of base-line characteristics and study compliance. *Journal of the National Cancer Institute*. 1996;88:1560–1570.
- 488. Jacobs EJ, Connell CJ, McCullough ML, Chao A, Jonas CR, Rodriguez C, Calle EE, Thun MJ. Vitamin C, vitamin E, and multivitamin supplement use and stomach cancer mortality in the Cancer Prevention Study II cohort. *Cancer Epidemiology, Biomarkers* and Prevention. 2002;11:35–41.
- 489. Blot WJ, Li JY, Taylor PR, Guo W, Dawsey S, Wang GQ, Yang CS, Zheng SF, Gail M, Li GY. Nutrition intervention trials in Linxian, China: Supplementation with specific vitamin/mineral combinations, cancer incidence, and disease-specific mortality in the general population. *Journal of the National Cancer Institute*. 1993;85:1483–1491.
- 490. Albanes D, Heinonen OP, Huttunen JK, Taylor PR, Virtamo J, Edwards B, Haapakoski J, Rautalahti M, Hartman A, Palmgren J. Effects of alpha-tocopherol and beta-carotene supplements on cancer incidence in the Alpha-Tocopherol Beta-Carotene Cancer Prevention Study. *American Journal of Clinical Nutrition*. 1995;62:1427S.
- 491. Klein EA, Thompson Jr IM, Tangen CM, Crowley JJ, Lucia MS, Goodman PJ, Minasian LM, Ford LG, Parnes HL, Gaziano JM. Vitamin E and the risk of prostate cancer. *Journal of the American Medical Association*. 2011;306:1549–1556.
- 492. Park SY, Murphy SP, Wilkens LR, Henderson BE, Kolonel LN. Multivitamin use and the risk of mortality and cancer incidence: The multiethnic cohort study. *American Journal of Epidemiology*. 2011;173:906–914.
- 493. Gaziano JM, Sesso HD, Christen WG, Bubes V, Smith JP, MacFadyen J, Schvartz M, Manson JAE, Glynn RJ, Buring JE. Multivitamins in the prevention of cancer in men: The Physicians' Health Study II randomized controlled trialmultivitamins in the prevention of cancer in men. *Journal of the American Medical Association*. 2012;308:1871–1880.
- 494. Myung SK, Kim Y, Ju W, Choi H, Bae W. Effects of antioxidant supplements on cancer prevention: Meta-analysis of randomized controlled trials. *Annals of Oncology*. 2010;21:166–179.
- 495. Kristal AR, Arnold KB, Neuhouser ML, Goodman P, Platz EA, Albanes D, Thompson IM. Diet, supplement use, and prostate cancer risk: Results from the prostate cancer prevention trial. *American Journal of Epidemiology*. 2010;172:566–577.
- 496. Zhang Y, Coogan P, Palmer JR, Strom BL, Rosenberg L. Vitamin and mineral use and risk of prostate cancer: The case–control surveillance study. *Cancer Causes and Control*. 2009;20:691–698.
- 497. Leo MA, Lieber CS. Interaction of ethanol with vitamin A. *Alcoholism: Clinical and Experimental Research*. 1983;7:15–21.
- 498. Leo M, Arai M, Sato M, Lieber CS. Hepatotoxicity of vitamin A and ethanol in the rat. *Gastroenterology*. 1982;82:194.
- 499. Talalay P, Fahey JW. Phytochemicals from cruciferous plants protect against cancer by modulating carcinogen metabolism. *Journal of Nutrition*. 2001;131:3027S–3033S.
- 500. Shu L, Cheung KL, Khor TO, Chen C, Kong AN. Phytochemicals: Cancer chemoprevention and suppression of tumor onset and metastasis. *Cancer and Metastasis Reviews*. 2010:29:483–502.
- 501. Ziegler RG, Colavito EA, Hartge P, McAdams MJ, Schoenberg JB, Mason TJ, Fraumeni Jr JF. Importance of alpha-carotene, beta-carotene, and other phytochemicals in the etiology of lung cancer. *Journal of the National Cancer Institute*. 1996;88:612.

- 502. McCann SE, Freudenheim JL, Marshall JR, Graham S. Risk of human ovarian cancer is related to dietary intake of selected nutrients, phytochemicals and food groups. *Journal* of Nutrition. 2003;133:1937.
- 503. Liu RH. Health benefits of fruit and vegetables are from additive and synergistic combinations of phytochemicals. *American Journal of Clinical Nutrition*. 2003;78:517S.
- Campbell JK, Canene-Adams K, Lindshield BL, Boileau TWM, Clinton SK, Erdman JW. Tomato phytochemicals and prostate cancer risk. *Journal of Nutrition*. 2004;134:34865.
- Wang H, Cao G, Ronald L. Total antioxidant capacity of fruits. *Journal of Agricultural* and Food Chemistry. 1996;44:701–705.
- 506. Lako J, Trenerry VC, Wahlqvist M, Wattanapenpaiboon N, Sotheeswaran S, Premier R. Phytochemical flavonols, carotenoids and the antioxidant properties of a wide selection of Fijian fruit, vegetables and other readily available foods. *Food Chemistry*. 2007;101:1727–1741.
- 507. Neveu V, Perez-Jiménez J, Vos F, Crespy V, Du Chaffaut L, Mennen L, Knox C, Eisner R, Cruz J, Wishart D. Phenol-explorer: An online comprehensive database on polyphenol contents in foods. *Database: The Journal of Biological Databases and Curation*. 2010;2010:bap024.
- Pauwels EKJ. The protective effect of the Mediterranean diet: Focus on cancer and cardiovascular risk. *Medical Principles and Practice*. 2011;20:103–111.
- 509. Esposito K, Marfella R, Ciotola M, Di Palo C, Giugliano F, Giugliano G, D'Armiento M, D'Andrea F, Giugliano D. Effect of a Mediterranean-style diet on endothelial dysfunction and markers of vascular inflammation in the metabolic syndrome. *Journal of the American Medical Association*. 2004;292:1440.
- 510. Willett WC, Sacks F, Trichopoulou A, Drescher G, Ferro-Luzzi A, Helsing E, Trichopoulos D. Mediterranean diet pyramid: A cultural model for healthy eating. *American Journal of Clinical Nutrition*. 1995;61:1402S.
- 511. Akira S, Uematsu S, Takeuchi O. Pathogen recognition and innate immunity. *Cell*. 2006;124:783–801.
- 512. Takeuchi O, Akira S. Pattern recognition receptors and inflammation. *Cell*. 2010;140:805–820.
- Bottazzi B, Doni A, Garlanda C, Mantovani A. An integrated view of humoral innate immunity: Pentraxins as a paradigm. *Annual Review of Immunology*. 2009;28:157–183.
- 514. Cinel I, Opal SM. Molecular biology of inflammation and sepsis: A primer. *Critical Care Medicine*. 2009;37:291–304.
- 515. Chen GY, Nuñez G. Sterile inflammation: Sensing and reacting to damage. *Nature Reviews Immunology*. 2010;10:826–837.
- Chen G, Shaw MH, Kim YG, Nuñez G. NOD-like receptors: Role in innate immunity and inflammatory disease. *Annual Review of Pathology*. 2009;4:365–398.
- 517. Takeuchi O, Akira S. Innate immunity to virus infection. *Immunological Reviews*. 2009;227:75–86.
- 518. Krishnan J, Selvarajoo K, Tsuchiya M, Lee G, Choi S. Toll-like receptor signal transduction. *Experimental and Molecular Medicine*. 2007;39:421.
- 519. Mogensen TH. Pathogen recognition and inflammatory signaling in innate immune defenses. *Clinical Microbiology Reviews*. 2009;22:240.
- 520. Lopez-Ilasaca M, Crespo P, Pellici PG, Gutkind JS, Wetzker R. Linkage of G proteincoupled receptors to the MAPK signaling pathway through PI3-kinase γ. Science. 1997;275:394.
- 521. Wang X, Martindale JL, Liu Y, Holbrook NJ. The cellular response to oxidative stress: Influences of mitogen-activated protein kinase signalling pathways on cell survival. *Biochemical Journal*. 1998;333:291.

- 522. Chang MC, Wu HL, Lee JJ, Lee PH, Chang HH, Hahn LJ, Lin BR, Chen YJ, Jeng JH. The induction of prostaglandin E2 production, interleukin-6 production, cell cycle arrest, and cytotoxicity in primary oral keratinocytes and KB cancer cells by areca nut ingredients is differentially regulated by MEK/ERK activation. *Journal of Biological Chemistry*. 2004;279:50676.
- 523. Chen BC, Liao CC, Hsu MJ, Liao YT, Lin CC, Sheu JR, Lin CH. Peptidoglycan-induced IL-6 production in RAW 264.7 macrophages is mediated by cyclooxygenase-2, PGE2/ PGE4 receptors, protein kinase A, Ικβ kinase, and ΝFκβ. *Journal of Immunology*. 2006;177:681.
- 524. Rahman I, Biswas SK, Kirkham PA. Regulation of inflammation and redox signaling by dietary polyphenols. *Biochemical Pharmacology*. 2006;72:1439–1452.
- 525. Santangelo C, Varì R, Scazzocchio B, Di Benedetto R, Filesi C, Masella R. Polyphenols, intracellular signalling and inflammation. *Annali-Istituto Superiore Di Sanita*. 2007;43:394.
- 526. Dolinsky VW, Dyck JRB. Calorie restriction and resveratrol in cardiovascular health and disease. *Biochimica et Biophysica Acta*. 2011;1812:1477–1489.
- 527. Yap S, Qin C, Woodman OL. Effects of resveratrol and flavonols on cardiovascular function: Physiological mechanisms. *Biofactors*. 2010;36:350–359.
- Zhao L, Lee JY, Hwang DH. Inhibition of pattern recognition receptor-mediated inflammation by bioactive phytochemicals. *Nutrition Reviews*. 2011;69:310–320.
- 529. Lucas L, Russell A, Keast R. Molecular mechanisms of inflammation. Antiinflammatory benefits of virgin olive oil and the phenolic compound oleocanthal. *Current Pharmaceutical Design*. 2011;17:754–768.
- 530. Surh YJ, Kundu JK, Na HK, Lee JS. Redox-sensitive transcription factors as prime targets for chemoprevention with anti-inflammatory and antioxidative phytochemicals. *Journal of Nutrition*. 2005;135:2993S–3001S.
- 531. Surh YJ, Na HK. NF- κ B and Nrf2 as prime molecular targets for chemoprevention and cytoprotection with anti-inflammatory and antioxidant phytochemicals. *Genes and Nutrition*. 2008;2:313–317.
- González-Gallego J, García-Mediavilla M, Sánchez-Campos S, Tuñón MJ. Fruit polyphenols, immunity and inflammation. *British Journal of Nutrition*. 2010;104:S15–S27.
- 533. Guo W, Kong EH, Meydani M. Dietary polyphenols, inflammation, and cancer. *Nutrition and Cancer*. 2009;61:807–810.
- 534. Pan MH, Lai CS, Dushenkov S, Ho CT. Modulation of inflammatory genes by natural dietary bioactive compounds. *Journal of Agricultural and Food Chemistry*. 2009;57:4467–4477.
- 535. Karlsen A, Retterstøl L, Laake P, Paur I, Kjølsrud-Bøhn S, Sandvik L, Blomhoff R. Anthocyanins inhibit NFκβ activation in monocytes and reduce plasma concentrations of pro-inflammatory mediators in healthy adults. *Journal of Nutrition*. 2007;137:1951–1954.
- 536. Park OJ, Surh YJ. Chemopreventive potential of epigallocatechin gallate and genistein: Evidence from epidemiological and laboratory studies. *Toxicology Letters*. 2004;150:43–56.
- 537. Middleton Jr E, Kandaswami C, Theoharides TC. The effects of plant flavonoids on mammalian cells: Implications for inflammation, heart disease, and cancer. *Pharmacological Reviews*. 2000;52:673–751.
- 538. Scalbert A, Williamson G. Dietary intake and bioavailability of polyphenols. *Journal of Nutrition*. 2000;130:2073S–2085S.
- 539. Calder PC. Polyunsaturated fatty acids and inflammation. *Prostaglandins, Leukotrienes and Essential Fatty Acids*. 2006;75:197–202.

- 540. Tilley SL, Coffman TM, Koller BH. Mixed messages: Modulation of inflammation and immune responses by prostaglandins and thromboxanes. *Journal of Clinical Investigation*. 2001;108:15–24.
- 541. Lee JH, Zhou HY, Cho SY, Kim YS, Lee YS, Jeong CS. Anti-inflammatory mechanisms of apigenin: Inhibition of cyclooxygenase-2 expression, adhesion of monocytes to human umbilical vein endothelial cells, and expression of cellular adhesion molecules. *Archives of Pharmacal Research*. 2007;30:1318–1327.
- 542. Harris GK, Qian Y, Leonard SS, Sbarra DC, Shi X. Luteolin and chrysin differentially inhibit cyclooxygenase-2 expression and scavenge reactive oxygen species but similarly inhibit prostaglandin-E2 formation in RAW 264.7 cells. *Journal of Nutrition*. 2006;136:1517–1521.
- 543. Dilis V, Trichopoulou A. Antioxidant intakes and food sources in Greek adults. *Journal* of Nutrition. 2010;140:1274–1279.
- 544. Chun OK, Kim DO, Smith N, Schroeder D, Han JT, Lee CY. Daily consumption of phenolics and total antioxidant capacity from fruit and vegetables in the American diet. *Journal of the Science of Food and Agriculture*. 2005;85:1715–1724.
- 545. Burdge GC, Calder PC. Conversion of alpha-linolenic acid to longer-chain polyunsaturated fatty acids in human adults. *Reproduction, Nutrition, Development.* 2005;45:581–598.
- 546. Arterburn LM, Hall EB, Oken H. Distribution, interconversion, and dose response of n-3 fatty acids in humans. *American Journal of Clinical Nutrition*. 2006;83:S1467–1476S.
- 547. Zatonski W, Campos H, Willett W. Rapid declines in coronary heart disease mortality in Eastern Europe are associated with increased consumption of oils rich in alpha-linolenic acid. *European Journal of Epidemiology*. 2008;23:3–10.
- 548. Appel S, Mirakaj V, Bringmann A, Weck MM, Grünebach F, Brossart P. PPAR-γ agonists inhibit toll-like receptor-mediated activation of dendritic cells via the MAP kinase and NF-κb pathways. *Blood.* 2005;106:3888.
- 549. Li H, Ruan XZ, Powis SH, Fernando R, Mon WY, Wheeler DC, Moorhead JF, Varghese Z. EPA and DHA reduce LPS-induced inflammation responses in HK-2 cells: Evidence for a PPAR-γ-dependent mechanism. *Kidney International*. 2005;67:867–874.
- 550. Eibl G. The role of PPAR-γ and its interaction with COX-2 in pancreatic cancer. *PPAR Research*. 2008;2008:326915.
- 551. Magee P, Pearson S, Whittingham-Dowd J, Allen J. PPARγ as a molecular target of EPA anti-inflammatory activity during TNF-α-impaired skeletal muscle cell differentiation. *Journal of Nutritional Biochemistry*. 2012;23:1440–1448.
- 552. Spinelli S, O'Brien J, Bancos S, Lehmann G, Springer D, Blumberg N, Francis C, Taubman M, Phipps R. The PPAR-platelet connection: Modulators of inflammation and potential cardiovascular effects. *PPAR Research*. 2008;2008:328172.
- 553. Yokoyama M, Origasa H, Matsuzaki M, Matsuzawa Y, Saito Y, Ishikawa Y, Oikawa S, Sasaki J, Hishida H, Itakura H. Effects of eicosapentaenoic acid on major coronary events in hypercholesterolaemic patients (JELIS): A randomised open-label, blinded endpoint analysis. *Lancet.* 2007;369:1090–1098.
- 554. Giugliano D, Ceriello A, Esposito K. The effects of diet on inflammation: Emphasis on the metabolic syndrome. *Journal of the American College of Cardiology*. 2006;48:677–685.
- 555. Din JN, Harding SA, Valerio CJ, Sarma J, Lyall K, Riemersma RA, Newby DE, Flapan AD. Dietary intervention with oil rich fish reduces platelet-monocyte aggregation in man. *Atherosclerosis*. 2008;197:290–296.
- 556. He K, Liu K, Daviglus ML, Jenny NS, Mayer-Davis E, Jiang R, Steffen L, Siscovick D, Tsai M, Herrington D. Associations of dietary long-chain n-3 polyunsaturated fatty acids and fish with biomarkers of inflammation and endothelial activation (from the Multi-Ethnic Study of Atherosclerosis [MESA]). *American Journal of Cardiology*. 2009;103:1238–1243.

- 557. Zampelas A, Panagiotakos DB, Pitsavos C, Das UN, Chrysohoou C, Skoumas Y, Stefanadis C. Fish consumption among healthy adults is associated with decreased levels of inflammatory markers related to cardiovascular disease: The ATTICA study. *Journal of the American College of Cardiology*. 2005;46:120.
- 558. Kris-Etherton PM, Harris WS, Appel LJ. Fish consumption, fish oil, omega-3 fatty acids, and cardiovascular disease. *Arteriosclerosis, Thrombosis, and Vascular Biology*. 2003;23:e20–e30.
- 559. Wang C, Harris WS, Chung M, Lichtenstein AH, Balk EM, Kupelnick B, Jordan HS, Lau J. n-3 Fatty acids from fish or fish-oil supplements, but not α-linolenic acid, benefit cardiovascular disease outcomes in primary-and secondary-prevention studies: A systematic review. *American Journal of Clinical Nutrition*. 2006;84:5–17.
- 560. Akbiyik F, Ray DM, Gettings KF, Blumberg N, Francis CW, Phipps RP. Human bone marrow megakaryocytes and platelets express PPARγ, and PPARγ agonists blunt platelet release of CD40 ligand and thromboxanes. *Blood*. 2004;104:1361–1368.
- 561. Ahmed AA, Holub BJ. Alteration and recovery of bleeding times, platelet aggregation and fatty acid composition of individual phospholipids in platelets of human subjects receiving a supplement of cod-liver oil. *Lipids*. 1984;19:617–624.
- 562. Herold PM, Kinsella JE. Fish oil consumption and decreased risk of cardiovascular disease: A comparison of findings from animal and human feeding trials. *American Journal* of Clinical Nutrition. 1986;43:566–598.
- 563. Rogers S, James KS, Butland BK, Etherington M, O'Brien J, Jones J. Effects of a fish oil supplement on serum lipids, blood pressure, bleeding time, haemostatic and rheological variables: A double blind randomised controlled trial in healthy volunteers. *Atherosclerosis*. 1987;63:137–143.
- 564. Gomez-Cabrera MC, Domenech E, Viña J. Moderate exercise is an antioxidant: Upregulation of antioxidant genes by training. *Free Radical Biology and Medicine*. 2008;44:126–131.
- 565. Sharman I, Down M, Sen R. The effects of vitamin E and training on physiological function and athletic performance in adolescent swimmers. *British Journal of Nutrition*. 1971;26:265–276.
- 566. Malm C, Svensson M, Ekblom B, Sjödin B. Effects of ubiquinone-10 supplementation and high intensity training on physical performance in humans. *Acta Physiologica Scandinavica*. 1997;161:379–384.
- 567. Malm C, Svensson M, Sjöberg B, Ekblom B, Sjödin B. Supplementation with ubiquinone-10 causes cellular damage during intense exercise. Acta Physiologica Scandinavica. 1996;157:511–512.
- 568. Matsuzawa A, Nishitoh H, Tobiume K, Takeda K, Ichijo H. Physiological roles of ASK1-mediated signal transduction in oxidative stress- and endoplasmic reticulum stress-induced apoptosis: Advanced findings from ASK1 knockout mice. *Antioxidants* and Redox Signaling. 2002;4:415–425.
- 569. Bhakat KK, Mantha AK, Mitra S. Transcriptional regulatory functions of mammalian AP-endonuclease (APE1/Ref-1), an essential multifunctional protein. *Antioxidants and Redox Signaling*. 2009;11:621–637.
- 570. Rhee SG, Woo HA. Multiple functions of peroxiredoxins: Peroxidases, sensors and regulators of the intracellular messenger H2O2, and protein chaperones. *Antioxidants and Redox Signaling*. 2011;15:781–794.
- 571. Tanaka T, Hosoi F, Yamaguchi-Iwai Y, Nakamura H, Masutani H, Ueda S, Nishiyama A, Takeda S, Wada H, Spyrou G. Thioredoxin-2 (Trx-2) is an essential gene regulating mitochondria-dependent apoptosis. *EMBO Journal*. 2002;21:1695–1703.
- 572. Neumann CA, Cao J, Manevich Y. Peroxiredoxin 1 and its role in cell signaling. *Cell Cycle*. 2009;8:4072–4078.

- 573. Ishii T, Warabi E, Yanagawa T. Novel roles of peroxiredoxins in inflammation, cancer and innate immunity. *Journal of Clinical Biochemistry and Nutrition*. 2012;50:91–105.
- 574. Na HK, Surh YJ. Modulation of Nrf2-mediated antioxidant and detoxifying enzyme induction by the green tea polyphenol EGCG. *Food and Chemical Toxicology*. 2008;46:1271–1278.
- 575. Kim YJ, Baek SH, Bogner PN, Ip C, Rustum YM, Fakih MG, Ramnath N, Park YM. Targeting the Nrf2-Prx1 pathway with selenium to enhance the efficacy and selectivity of cancer therapy. *Journal of Cancer Molecules*. 2007;3:37–43.
- 576. Kaspar JW, Niture SK, Jaiswal AK. Nrf2: INrf2 (Keap1) signaling in oxidative stress. *Free Radical Biology and Medicine*. 2009;47:1304–1309.
- 577. Nerland DE. The antioxidant/electrophile response element motif. *Drug Metabolism Reviews*. 2007;39:235–248.
- 578. Nakamura Y, Miyoshi N. Electrophiles in foods: The current status of isothiocyanates and their chemical biology. *Bioscience, Biotechnology, and Biochemistry*. 2010;74:242–255.
- 579. Miyamoto N, Izumi H, Miyamoto R, Kondo H, Tawara A, Sasaguri Y, Kohno K. Quercetin induces the expression of peroxiredoxins 3 and 5 via the Nrf2/Nrf1 transcription pathway. *Investigative Ophthalmology and Visual Science*. 2011;52:1055–1063.
- 580. Rushworth SA, Ogborne RM, Charalambos CA, O'Connell MA. Role of protein kinase C [delta] in curcumin-induced antioxidant response element-mediated gene expression in human monocytes. *Biochemical and Biophysical Research Communications*. 2006;341:1007–1016.
- 581. Hanneken A, Lin FF, Johnson J, Maher P. Flavonoids protect human retinal pigment epithelial cells from oxidative-stress-induced death. *Investigative Ophthalmology and Visual Science*. 2006;47:3164–3177.
- 582. Tanigawa S, Fujii M, Hou DX. Action of Nrf2 and Keap1 in ARE-mediated NQO1 expression by quercetin. *Free Radical Biology and Medicine*. 2007;42:1690–1703.
- 583. Kluth D, Banning A, Paur I, Blomhoff R, Brigelius-Flohé R. Modulation of pregnane X receptor- and electrophile responsive element-mediated gene expression by dietary polyphenolic compounds. *Free Radical Biology and Medicine*. 2007;42:315–325.
- 584. Chen CY, Jang JH, Li MH, Surh YJ. Resveratrol upregulates heme oxygenase-1 expression via activation of NF-E2-related factor 2 in PC12 cells. *Biochemical and Biophysical Research Communications*. 2005;331:993–1000.
- 585. Kode A, Rajendrasozhan S, Caito S, Yang SR, Megson IL, Rahman I. Resveratrol induces glutathione synthesis by activation of Nrf2 and protects against cigarette smoke-mediated oxidative stress in human lung epithelial cells. *American Journal of Physiology*. 2008;294:L478–L488.
- 586. Ben-Dor A, Steiner M, Gheber L, Danilenko M, Dubi N, Linnewiel K, Zick A, Sharoni Y, Levy J. Carotenoids activate the antioxidant response element transcription system. *Molecular Cancer Therapeutics*. 2005;4:177.
- 587. Gong P, Hu B, Cederbaum AI. Diallyl sulfide induces heme oxygenase-1 through MAPK pathway. *Archives of Biochemistry and Biophysics*. 2004;432:252–260.
- 588. Keum YS, Yu S, Chang PPJ, Yuan X, Kim JH, Xu C, Han J, Agarwal A, Kong ANT. Mechanism of action of sulforaphane: Inhibition of p38 mitogen-activated protein kinase isoforms contributing to the induction of antioxidant response element–mediated heme oxygenase-1 in human hepatoma HepG2 cells. *Cancer Research*. 2006;66:8804.
- 589. Thimmulappa RK, Mai KH, Srisuma S, Kensler TW, Yamamoto M, Biswal S. Identification of Nrf2-regulated genes induced by the chemopreventive agent sulforaphane by oligonucleotide microarray. *Cancer Research*. 2002;62:5196.
- 590. Dickinson DA, Iles KE, Zhang H, Blank V, Forman HJ. Curcumin alters EPRE and AP-1 binding complexes and elevates glutamate-cysteine ligase gene expression. *FASEB Journal*. 2003;17:473–475.

- 591. Sang S, Yang I, Buckley B, Ho CT, Yang CS. Autoxidative quinone formation in vitro and metabolite formation in vivo from tea polyphenol (-)-epigallocatechin-3-gallate: Studied by real-time mass spectrometry combined with tandem mass ion mapping. *Free Radical Biology and Medicine*. 2007;43:362–371.
- 592. Birt DF, Hendrich S, Wang W. Dietary agents in cancer prevention: Flavonoids and isoflavonoids. *Pharmacology and Therapeutics*. 2001;90:157–177.
- 593. Miller PE, Snyder DC. Phytochemicals and cancer risk: A review of the epidemiological evidence. *Nutrition in Clinical Practice*. 2012;27:599–612.
- 594. Gorlova OY, Weng SF, Hernandez L, Spitz MR, Forman MR. Dietary patterns affect lung cancer risk in never smokers. *Nutrition and Cancer*. 2011;63:842–849.
- 595. Moiseeva EP, Almeida GM, Jones GDD, Manson MM. Extended treatment with physiologic concentrations of dietary phytochemicals results in altered gene expression, reduced growth, and apoptosis of cancer cells. *Molecular Cancer Therapeutics*. 2007;6:3071–3079.
- 596. Howells LM, Moiseeva EP, Neal CP, Foreman BE, Andreadi CK, Sun Y, Hudson E, Manson MM. Predicting the physiological relevance of in vitro cancer preventive activities of phytochemicals. *Acta Pharmacologica Sinica*. 2007;28:1274–1304.
- 597. Gorecki M, Beck Y, Hartman JR, Fischer M, Weiss L, Tochner Z, Slavin S, Nimrod A. Recombinant human superoxide dismutases: Production and potential therapeutical uses. *Free Radical Research*. 1991;12:401–410.
- 598. Mellman W, Schimke R, Hayflick L. Catalase turnover in human diploid cell cultures. *Experimental Cell Research*. 1972;73:399–409.
- 599. Chada S, Whitney C, Newburger PE. Post-transcriptional regulation of glutathione peroxidase gene expression by selenium in the HL-60 human myeloid cell line. *Blood*. 1989;74:2535–2541.
- 600. Gallegos A, Berggren M, Gasdaska JR, Powis G. Mechanisms of the regulation of thioredoxin reductase activity in cancer cells by the chemopreventive agent selenium. *Cancer Research*. 1997;57:4965.
- Cheung KL, Kong AN. Molecular targets of dietary phenethyl isothiocyanate and sulforaphane for cancer chemoprevention. *AAPS Journal*. 2010;12:87–97.
- 602. Tan AC, Konczak I, Sze DMY, Ramzan I. Molecular pathways for cancer chemoprevention by dietary phytochemicals. *Nutrition and Cancer*. 2011;63:495–505.
- 603. Milner JA, McDonald SS, Anderson DE, Greenwald P. Molecular targets for nutrients involved with cancer prevention. *Nutrition and Cancer*. 2001;41:1–16.
- 604. Bansal P, Gupta V V, Bansal R, Sapra R. Dietary phytochemicals in cell cycle arrest and apoptosis—An insight. *Journal of Drug Delivery and Therapeutics*. 2012;2:8–17.
- 605. Russo M, Spagnuolo C, Tedesco I, Russo GL. Phytochemicals in cancer prevention and therapy: Truth or dare? *Toxins*. 2010;2:517–551.
- 606. Hartojo W, Silvers AL, Thomas DG, Seder CW, Lin L, Rao H, Wang Z, Greenson JK, Giordano TJ, Orringer MB. Curcumin promotes apoptosis, increases chemosensitivity, and inhibits nuclear factor κβ in esophageal adenocarcinoma. *Translational Oncology*. 2010;3:99.
- 607. Frey RS, Li J, Singletary KW. Effects of genistein on cell proliferation and cell cycle arrest in nonneoplastic human mammary epithelial cells: Involvement of Cdc2, p21(waf/cip1), p27(kip1), and Cdc25C expression. *Biochemical Pharmacology*. 2001;61:979–989.
- 608. Cui X, Jin Y, Singh UP, Chumanevich AA, Harmon B, Cavicchia P, Hofseth AB, Kotakadi V, Poudyal D, Stroud B. Suppression of DNA damage in human peripheral blood lymphocytes by a juice concentrate: A randomized, double-blind, placebo-controlled trial. *Molecular Nutrition and Food Research*. 2012;56:666–670.
- 609. Aiyer HS, Kichambare S, Gupta RC. Prevention of oxidative DNA damage by bioactive berry components. *Nutrition and Cancer*. 2008;60:36–42.

- 610. Smith MJ, Inserra PF, Watson RR, Wise JA, O'Neill KL. Supplementation with fruit and vegetable extracts may decrease DNA damage in the peripheral lymphocytes of an elderly population. *Nutrition Research*. 1999;19:1507–1518.
- 611. Park YK, Park E, Kim JS, Kang MH. Daily grape juice consumption reduces oxidative DNA damage and plasma free radical levels in healthy Koreans. *Mutation Research*. 2003;529:77–86.
- 612. Mišík M, Hoelzl C, Wagner KH, Cavin C, Moser B, Kundi M, Simic T, Elbling L, Kager N, Ferk F. Impact of paper filtered coffee on oxidative DNA-damage: Results of a clinical trial. *Mutation Research*. 2010;692:42–48.
- 613. Lorenzo Y, Azqueta A, Luna L, Bonilla F, Domínguez G, Collins AR. The carotenoid β-cryptoxanthin stimulates the repair of DNA oxidation damage in addition to acting as an antioxidant in human cells. *Carcinogenesis*. 2009;30:308–314.
- 614. Raffoul JJ, Heydari AR, Hillman GG. DNA repair and cancer therapy: Targeting Ape1/ Ref-1 using dietary agents. *Journal of Oncology*. 2012;2012:370481.
- 615. Roy S, Deep G, Agarwal C, Agarwal R. Silibinin prevents ultraviolet B radiationinduced epidermal damages in JB6 cells and mouse skin in a p53-GADD45α-dependent manner. *Carcinogenesis*. 2012;33:629–636.
- 616. Chakraborty S, Roy M, Bhattacharya RK. Prevention and repair of DNA damage by selected phytochemicals as measured by single cell gel electrophoresis. *Journal of Environmental Pathology, Toxicology and Oncology: Official Organ of the International Society for Environmental Toxicology and Cancer.* 2004;23:215.
- 617. Aiyer HS, Vadhanam MV, Stoyanova R, Caprio GD, Clapper ML, Gupta RC. Dietary berries and ellagic acid prevent oxidative DNA damage and modulate expression of DNA repair genes. *International Journal of Molecular Sciences*. 2008;9:327–341.
- 618. Min K, Ebeler SE. Quercetin inhibits hydrogen peroxide-induced DNA damage and enhances DNA repair in CACO-2 cells. *Food and Chemical Toxicology*. 2009;47:2716–2722.
- 619. Guarnieri S, Loft S, Riso P, Porrini M, Risom L, Poulsen HE, Dragsted LO, Moller P. DNA repair phenotype and dietary antioxidant supplementation. *British Journal of Nutrition*. 2008;99:1018–1024.
- 620. Guarrera S, Sacerdote C, Fiorini L, Marsala R, Polidoro S, Gamberini S, Saletta F, Malaveille C, Talaska G, Vineis P. Expression of DNA repair and metabolic genes in response to a flavonoid-rich diet. *British Journal of Nutrition*. 2007;98:525–533.
- 621. Collins AR, Harrington V, Drew J, Melvin R. Nutritional modulation of DNA repair in a human intervention study. *Carcinogenesis*. 2003;24:511–515.
- 622. Bacon JR, Williamson G, Garner RC, Lappin G, Langouët S, Bao Y. Sulforaphane and quercetin modulate PHIP–DNA adduct formation in human HEPG2 cells and hepatocytes. *Carcinogenesis*. 2003;24:1903–1911.
- 623. Iqbal M, Sharma SD, Okazaki Y, Fujisawa M, Okada S. Dietary supplementation of curcumin enhances antioxidant and phase II metabolizing enzymes in DDY male mice: Possible role in protection against chemical carcinogenesis and toxicity. *Pharmacology* and Toxicology. 2003;92:33–38.
- 624. Melega S, Canistro D, Pagnotta E, Iori R, Sapone A, Paolini M. Effect of sprout extract from Tuscan black cabbage on xenobiotic-metabolizing and antioxidant enzymes in rat liver. *Mutation Research*. 2012;751:450–451.
- 625. Nho CW, Jeffery E. The synergistic upregulation of phase II detoxification enzymes by glucosinolate breakdown products in cruciferous vegetables. *Toxicology and Applied Pharmacology*. 2001;174:146–152.
- 626. Kelly VP, Ellis EM, Manson MM, Chanas SA, Moffat GJ, McLeod R, Judah DJ, Neal GE, Hayes JD. Chemoprevention of aflatoxin B1 hepatocarcinogenesis by coumarin, a natural benzopyrone that is a potent inducer of aflatoxin B1-aldehyde reductase, the glutathione S-transferase A5 and P1 subunits, and NAD(P)H: Quinone oxidoreductase in rat liver. *Cancer Research*. 2000;60:957–969.

- 627. Wang H, Khor TO, Yang Q, Huang Y, Wu T, Saw CLL, Lin W, Androulakis IP, Kong AN. Pharmacokinetics and pharmacodynamics of phase II drug metabolizing/antioxidant enzymes gene response by anti-cancer agent sulforaphane in rat lymphocytes. *Molecular Pharmaceutics*. 2012;9:2819–2827.
- 628. Han SG, Han SS, Toborek M, Hennig B. EGCG protects endothelial cells against PCB 126-induced inflammation through inhibition of AhR and induction of Nrf2-regulated genes. *Toxicology and Applied Pharmacology*. 2012;261:181–188.
- 629. Warwick E, Cassidy A, Hanley B, Jouni ZE, Bao Y. Effect of phytochemicals on phase II enzyme expression in infant human primary skin fibroblast cells. *British Journal of Nutrition*. 2012;108:2158–2165.
- 630. Saw CLL, Cintrón M, Wu TY, Guo Y, Huang Y, Jeong WS, Kong ANT. Pharmacodynamics of dietary phytochemical indoles I3C and DIM: Induction of Nrf2-mediated phase II drug metabolizing and antioxidant genes and synergism with isothiocyanates. *Biopharmaceutics and Drug Disposition*. 2011;32:289–300.
- 631. Chen C, Pung D, Leong V, Hebbar V, Shen G, Nair S, Li W, Kong A. Induction of detoxifying enzymes by garlic organosulfur compounds through transcription factor Nrf2: Effect of chemical structure and stress signals. *Free Radical Biology and Medicine*. 2004;37:1578.
- 632. Kleiner HE, Vulimiri SV, Miller L, Johnson WH, Whitman CP, DiGiovanni J. Oral administration of naturally occurring coumarins leads to altered phase I and II enzyme activities and reduced DNA adduct formation by polycyclic aromatic hydrocarbons in various tissues of SENCAR mice. *Carcinogenesis*. 2001;22:73–82.
- 633. Yang YM, Conaway CC, Chiao J, Wang CX, Amin S, Whysner J, Dai W, Reinhardt J, Chung FL. Inhibition of benzo (a) pyrene-induced lung tumorigenesis in A/J mice by dietary N-acetylcysteine conjugates of benzyl and phenethyl isothiocyanates during the postinitiation phase is associated with activation of mitogen-activated protein kinases and p53 activity and induction of apoptosis. *Cancer Research*. 2002;62:2.
- 634. Xu Y, Ho C-T, Amin SG, Han C, Chung F-L. Inhibition of tobacco-specific nitrosamineinduced lung tumorigenesis in A/J mice by green tea and its major polyphenol as antioxidants. *Cancer Research*. 1992;52:3875–3879.
- 635. Verma AK, Johnson JA, Gould MN, Tanner MA. Inhibition of 7, 12-dimethylbenz(a) anthracene- and N-nitrosomethylurea-induced rat mammary cancer by dietary flavonol quercetin. *Cancer Research*. 1988;48:5754–5758.
- 636. Lamartiniere CA, Murrill WB, Manzolillo PA, Zhang J-X, Barnes S, Zhang X, Wei H, Brown NM. Genistein alters the ontogeny of mammary gland development and protects against chemically-induced mammary cancer in rats. *Proceedings of the Society for Experimental Biology and Medicine*. 1998;217:358–364.
- 637. Stoner GD, Wang L-S, Seguin C, Rocha C, Stoner K, Chiu S, Kinghorn AD. Multiple berry types prevent n-nitrosomethylbenzylamine-induced esophageal cancer in rats. *Pharmaceutical Research*. 2010;27:1138–1145.
- 638. Pereira MA, Grubbs CJ, Barnes LH, Li H, Olson GR, Eto I, Juliana M, Whitaker LM, Kelloff GJ, Steele VE. Effects of the phytochemicals, curcumin and quercetin, upon azoxymethane-induced colon cancer and 7, 12-dimethylbenz[a]anthracene-induced mammary cancer in rats. *Carcinogenesis*. 1996;17:1305–1311.
- 639. Banerjee S, Bueso-Ramos C, Aggarwal BB. Suppression of 7, 12-dimethylbenz(a) anthracene-induced mammary carcinogenesis in rats by resveratrol role of nuclear factor-κβ, cyclooxygenase 2, and matrix metalloprotease 9. *Cancer Research*. 2002;62:4945–4954.
- 640. Lubet RA, Heckman BM, De Flora SL, Steele V, Crowell JA, Juliana M, Grubbs CJ. Effects of 5, 6-benzoflavone, indole-3-carbinol (I3C) and diindolylmethane (DIM) on chemically-induced mammary carcinogenesis: Is DIM a substitute for I3C? *Oncology Reports*. 2011;26:731.

- 641. Bear W, Teel R. Effects of citrus phytochemicals on liver and lung cytochrome P450 activity and on the in vitro metabolism of the tobacco-specific nitrosamine NNK. *Anticancer Research*. 2000;20:3323.
- 642. Muto S, Fujita K, Yamazaki Y, Kamataki T. Inhibition by green tea catechins of metabolic activation of procarcinogens by human cytochrome P450. *Mutation Research*. 2001;479:197–206.
- 643. Misaka S, Kawabe K, Onoue S, Werba JP, Giroli M, Tamaki S, Kan T, Kimura J, Watanabe H, Yamada S. Effects of green tea catechins on cytochrome P450 2B6, 2C8, 2C19, 2D6 and 3A activities in human liver and intestinal microsomes. *Drug Metabolism* and Pharmacokinetics. 2012;28:244–249.
- 644. Thapliyal R, Maru G. Inhibition of cytochrome P450 isozymes by curcumins in vitro and in vivo. *Food and Chemical Toxicology*. 2001;39:541–547.
- 645. Jiao D, Smith T, Yang C, Pittman B, Desai D, Amin S, Chung F. Chemopreventive activity of thiol conjugates of isothiocyanates for lung tumorigenesis. *Carcinogenesis*. 1997;18:2143–2147.
- 646. Barcelo S, Mace K, Pfeifer A, Chipman J. Production of DNA strand breaks by N-nitrosodimethylamine and 2-amino-3-methylimidazo[4, 5-f]quinoline in THLE cells expressing human CYP isoenzymes and inhibition by sulforaphane. *Mutation Research*. 1998;402:111–120.
- 647. Renwick A, Mistry H, Barton P, Mallet F, Price R, Beamand J, Lake B. Effect of some indole derivatives on xenobiotic metabolism and xenobiotic-induced toxicity in cultured rat liver slices. *Food and Chemical Toxicology*. 1999;37:609–618.
- 648. Broadbent TA, Broadbent HS. The chemistry and pharmacology of indole-3-carbinol (indole-3-methanol) and 3-(methoxymethyl) indole [Part I]. *Current Medicinal Chemistry*. 1998;5:52.
- 649. Moon YJ, Wang X, Morris ME. Dietary flavonoids: Effects on xenobiotic and carcinogen metabolism. *Toxicology In Vitro*. 2006;20:187–210.
- 650. Zhou S, Gao Y, Jiang W, Huang M, Xu A, Paxton JW. Interactions of herbs with cytochrome P450. *Drug Metabolism Reviews*. 2003;35:35–98.
- 651. Kritchevsky D. The effect of over- and undernutrition on cancer. *European Journal of Cancer*. 1995;4:445–452.
- 652. Hursting SD, Dunlap SM, Ford NA, Hursting MJ, Lashinger LM. Calorie restriction and cancer prevention: A mechanistic perspective. *Cancer and Metabolism*. 2013;1:10.
- 653. Hursting SD, Lavigne JA, Berrigan D, Perkins SN, Barrett JC. Calorie restriction, aging, and cancer prevention: Mechanisms of action and applicability to humans. *Annual Review of Medicine*. 2003;54:131–152.
- 654. Hine CM, Mitchell JR. Nrf2 and the phase II response in acute stress resistance induced by dietary restriction. *Journal of Clinical and Experimental Pathology*. 2012;4:pii 7329.
- 655. Bordone L, Guarente L. Calorie restriction, SIRT1 and metabolism: Understanding longevity. *Nature Reviews Molecular Cell Biology*. 2005;6:298–305.
- 656. Longo VD, Fontana L. Calorie restriction and cancer prevention: Metabolic and molecular mechanisms. *Trends in Pharmacological Sciences*. 2010;31:89–98.
- 657. Moore T, Beltran L, Carbajal S, Strom S, Traag J, Hursting SD, DiGiovanni J. Dietary energy balance modulates signaling through the Akt/mammalian target of rapamycin pathways in multiple epithelial tissues. *Cancer Prevention Research*. 2008;1:65–76.
- 658. Cantó C, Auwerx J. Calorie restriction: Is AMPK a key sensor and effector? *Physiology*. 2011;26:214–224.
- 659. Iwabu M, Yamauchi T, Okada-Iwabu M, Sato K, Nakagawa T, Funata M, Yamaguchi M, Namiki S, Nakayama R, Tabata M. Adiponectin and AdipoR1 regulate PGC-1α [AGR] and mitochondria by Ca²⁺ and AMPK/SIRT1. *Nature*. 2010;464:1313–1319.

- 660. Pearson KJ, Lewis KN, Price NL, Chang JW, Perez E, Cascajo MV, Tamashiro KL, Poosala S, Csiszar A, Ungvari Z. Nrf2 mediates cancer protection but not prolongevity induced by caloric restriction. *Proceedings of the National Academy of Sciences of the United States of America*. 2008;105:2325–2330.
- Martín-Montalvo A, Villalba JM, Navas P, de Cabo R. Nrf2, cancer and calorie restriction. *Oncogene*. 2010;30:505–520.
- 662. Wen H, Yang HJ, An YJ, Kim JM, Lee DH, Jin X, Park SW, Min KJ, Park S. Enhanced phase II detoxification contributes to beneficial effects of dietary restriction as revealed by multi-platform metabolomics studies. *Molecular and Cellular Proteomics*. 2013;12:575–586.
- 663. Cao SX, Dhahbi JM, Mote PL, Spindler SR. Genomic profiling of short- and longterm caloric restriction effects in the liver of aging mice. *Proceedings of the National Academy of Sciences of the United States of America*. 2001;98:10630–10635.
- 664. Gao P, Chou MW. Effect of caloric restriction on hepatic nuclear DNA damage in male Fischer 344 rats treated with aflatoxin B1. *Toxicology Letters*. 1992;61:233–242.
- 665. Kritchevsky D. Caloric restriction and experimental carcinogenesis. *Toxicological Sciences*. 1999;52:13–16.
- 666. Fontana L, Weiss EP, Villareal DT, Klein S, Holloszy JO. Long-term effects of calorie or protein restriction on serum IGF-1 and IGFBP-3 concentration in humans. *Aging Cell*. 2008;7:681–687.
- 667. Holmes MD, Pollak MN, Willett WC, Hankinson SE. Dietary correlates of plasma insulin-like growth factor I and insulin-like growth factor binding protein 3 concentrations. *Cancer Epidemiology, Biomarkers and Prevention*. 2002;11:852–861.
- 668. Giovannucci E, Pollak M, Liu Y, Platz EA, Majeed N, Rimm EB, Willett WC. Nutritional predictors of insulin-like growth factor I and their relationships to cancer in men. *Cancer Epidemiology, Biomarkers and Prevention*. 2003;12:84–89.
- 669. Fulgoni VL. Current protein intake in America: Analysis of the National Health and Nutrition Examination Survey, 2003–2004. *The American Journal of Clinical Nutrition*. 2008;87:1554S–1557S.
- 670. Tremblay MS, Warburton DE, Janssen I, Paterson DH, Latimer AE, Rhodes RE, Kho ME, Hicks A, LeBlanc AG, Zehr L. New Canadian physical activity guidelines. *Applied Physiology, Nutrition, and Metabolism.* 2011;36:36–46.
- 671. Friedenreich CM, Neilson HK, Lynch BM. State of the epidemiological evidence on physical activity and cancer prevention. *European Journal of Cancer*. 2010;46:2593.
- 672. Campbell KL, McTiernan A. Exercise and biomarkers for cancer prevention studies. *Journal of Nutrition*. 2007;137:161S.
- 673. Tremblay MS, Colley RC, Saunders TJ, Healy GN, Owen N. Physiological and health implications of a sedentary lifestyle. *Applied Physiology, Nutrition, and Metabolism*. 2010;35:725–740.
- 674. Thijssen DHJ, Maiorana AJ, O'Driscoll G, Cable NT, Hopman MTE, Green DJ. Impact of inactivity and exercise on the vasculature in humans. *European Journal of Applied Physiology*. 2010;108:845–875.
- 675. Seider MJ, Nicholson WF, Booth FW. Insulin resistance for glucose metabolism in disused soleus muscle of mice. *American Journal of Physiology*. 1982;242:E12–E18.
- 676. Booth FW, Gordon SE, Carlson CJ, Hamilton MT. Waging war on modern chronic diseases: Primary prevention through exercise biology. *Journal of Applied Physiology*. 2000;88:774–787.
- 677. Booth FW, Lees SJ. Fundamental questions about genes, inactivity, and chronic diseases. *Physiological Genomics*. 2007;28:146–157.
- 678. Wei Y, Chen K, Whaley-Connell AT, Stump CS, Ibdah JA, Sowers JR. Skeletal muscle insulin resistance: Role of inflammatory cytokines and reactive oxygen species. *American Journal of Physiology*. 2008;294:R673–R680.

- 679. Tresierras MA, Balady GJ. Resistance training in the treatment of diabetes and obesity: Mechanisms and outcomes. *Journal of Cardiopulmonary Rehabilitation and Prevention*. 2009;29:67.
- 680. Coker RH, Williams RH, Yeo SE, Kortebein PM, Bodenner DL, Kern PA, Evans WJ. The impact of exercise training compared to caloric restriction on hepatic and peripheral insulin resistance in obesity. *Journal of Clinical Endocrinology and Metabolism*. 2009;94:4258–4266.
- 681. Bell LM, Watts K, Siafarikas A, Thompson A, Ratnam N, Bulsara M, Finn J, O'Driscoll G, Green DJ, Jones TW. Exercise alone reduces insulin resistance in obese children independently of changes in body composition. *Journal of Clinical Endocrinology and Metabolism*. 2007;92:4230–4235.
- 682. Ferguson M, Gutin B, Le N, Karp W, Litaker M, Humphries M, Okuyama T, Riggs S, Owens S. Effects of exercise training and its cessation on components of the insulin resistance syndrome in obese children. *International Journal of Obesity and Related Metabolic Disorders*. 1999;23:889.
- 683. Ross R, Janssen I, Dawson J, Kungl AM, Kuk JL, Wong SL, Nguyen-Duy TB, Lee SJ, Kilpatrick K, Hudson R. Exercise-induced reduction in obesity and insulin resistance in women: A randomized controlled trial. *Obesity*. 2004;12:789–798.
- 684. Björntorp P, de Jounge K, Sjöström L, Sullivan L. The effect of physical training on insulin production in obesity. *Metabolism.* 1970;19:631–638.
- 685. Houmard JA, Tanner CJ, Slentz CA, Duscha BD, McCartney JS, Kraus WE. Effect of the volume and intensity of exercise training on insulin sensitivity. *Journal of Applied Physiology*. 2004;96:101–106.
- 686. Koopman R, Manders RJ, Zorenc AH, Hul GB, Kuipers H, Keizer HA, van Loon LJ. A single session of resistance exercise enhances insulin sensitivity for at least 24 h in healthy men. *European Journal of Applied Physiology*. 2005;94:180–187.
- 687. Balducci S, Zanuso S, Nicolucci A, Fernando F, Cavallo S, Cardelli P, Fallucca S, Alessi E, Letizia C, Jimenez A. Anti-inflammatory effect of exercise training in subjects with type 2 diabetes and the metabolic syndrome is dependent on exercise modalities and independent of weight loss. *Nutrition, Metabolism, and Cardiovascular Diseases*. 2010;20:608–617.
- 688. Zanuso S, Jimenez A, Pugliese G, Corigliano G, Balducci S. Exercise for the management of type 2 diabetes: A review of the evidence. *Acta Diabetologica*. 2010;47:15–22.
- 689. Hawley J, Lessard S. Exercise training-induced improvements in insulin action. *Acta Physiologica*. 2007;192:127–135.
- 690. Zierath JR. Invited review: Exercise training-induced changes in insulin signaling in skeletal muscle. *Journal of Applied Physiology*. 2002;93:773–781.
- 691. Jorge MLMP, de Oliveira VN, Resende NM, Paraiso LF, Calixto A, Diniz ALD, Resende ES, Ropelle ER, Carvalheira JB, Espindola FS. The effects of aerobic, resistance, and combined exercise on metabolic control, inflammatory markers, adipocytokines, and muscle insulin signaling in patients with type 2 diabetes mellitus. *Metabolism*. 2011;60:1244–1252.
- 692. Lira VA, Benton CR, Yan Z, Bonen A. PGC-1α regulation by exercise training and its influences on muscle function and insulin sensitivity. *American Journal of Physiology*. 2010;299:E145–E161.
- 693. Little JP, Safdar A, Cermak N, Tarnopolsky MA, Gibala MJ. Acute endurance exercise increases the nuclear abundance of PGC-1α in trained human skeletal muscle. *American Journal of Physiology*. 2010;298:R912–R917.
- 694. Palacios OM, Carmona JJ, Michan S, Chen KY, Manabe Y, Ward III JL, Goodyear LJ, Tong Q. Diet and exercise signals regulate SIRT3 and activate AMPK and PGC-1α in skeletal muscle. *Aging*. 2009;1:771.

- 695. Cartee GD, Funai K. Exercise and insulin: Convergence or divergence at AS160 and TBC1D1? *Exercise and Sport Sciences Reviews*. 2009;37:188.
- 696. Funai K, Schweitzer GG, Castorena CM, Kanzaki M, Cartee GD. In vivo exercise followed by in vitro contraction additively elevates subsequent insulin-stimulated glucose transport by rat skeletal muscle. *American Journal of Physiology*. 2010;298:E999–E1010.
- 697. Funai K, Schweitzer GG, Sharma N, Kanzaki M, Cartee GD. Increased AS160 phosphorylation, but not TBC1D1 phosphorylation, with increased postexercise insulin sensitivity in rat skeletal muscle. *American Journal of Physiology*. 2009;297:E242–E251.
- 698. Chen S, Murphy J, Toth R, Campbell DG, Morrice NA, MacKintosh C. Complementary regulation of TBC1D1 and AS160 by growth factors, insulin and AMPK activators. *Biochemical Journal*. 2010;409:449–459.
- 699. Jackson MJ, Pye D, Palomero J. The production of reactive oxygen and nitrogen species by skeletal muscle. *Journal of Applied Physiology*. 2007;102:1664–1670.
- 700. Davies KJA, Quintanilha AT, Brooks GA, Packer L. Free radicals and tissue damage produced by exercise. *Biochemical and Biophysical Research Communications*. 1982;107:1198–1205.
- 701. Gleeson M, Bishop NC, Stensel DJ, Lindley MR, Mastana SS, Nimmo MA. The antiinflammatory effects of exercise: Mechanisms and implications for the prevention and treatment of disease. *Nature Reviews Immunology*. 2011;11:607–615.
- 702. Jolitha A, Subramanyam M, Asha Devi S. Modification by vitamin E and exercise of oxidative stress in regions of aging rat brain: Studies on superoxide dismutase isoenzymes and protein oxidation status. *Experimental Gerontology*. 2006;41:753–763.
- 703. Gomez-Cabrera MC, Sanchis-Gomar F, Martinez-Bello VE, Ibanez-Sania S, Nascimento AL, Ji LL, Vina J. Exercise as a model to study oxidative stress. *Studies on Experimental Models*. 2011:531–542.
- 704. Sen CK, Marin E, Kretzschmar M, Hanninen O. Skeletal muscle and liver glutathione homeostasis in response to training, exercise, and immobilization. *Journal of Applied Physiology*. 1992;73:1265–1272.
- 705. Rush JWE, Turk JR, Laughlin MH. Exercise training regulates SOD-1 and oxidative stress in porcine aortic endothelium. *American Journal of Physiology-Heart and Circulatory Physiology*. 2003;284:H1378–H1387.
- 706. Kavazis AN, Alvarez S, Talbert E, Lee Y, Powers SK. Exercise training induces a cardioprotective phenotype and alterations in cardiac subsarcolemmal and intermyofibrillar mitochondrial proteins. *American Journal of Physiology-Heart and Circulatory Physiology*. 2009;297:H144–H152.
- 707. Lappalainen Z, Lappalainen J, Oksala NK, Laaksonen DE, Khanna S, Sen CK, Atalay M. Diabetes impairs exercise training-associated thioredoxin response and glutathione status in rat brain. *Journal of Applied Physiology*. 2009;106:461–467.
- 708. Kumar R, Negi P, Singh B, Ilavazhagan G, Bhargava K, Sethy NK. *Cordyceps sinensis* promotes exercise endurance capacity of rats by activating skeletal muscle metabolic regulators. *Journal of Ethnopharmacology*. 2011;136:260–266.
- Sen CK. Glutathione homeostasis in response to exercise training and nutritional supplements. *Molecular and Cellular Biochemistry*. 1999;196:31–42.
- 710. Muthusamy VR, Kannan S, Sadhaasivam K, Gounder SS, Davidson CJ, Boeheme C, Hoidal JR, Wang L, Soorappan RN. Acute exercise stress activates Nrf2/ARE signaling and promotes antioxidant mechanisms in the myocardium. *Free Radical Biology and Medicine*. 2012;52:366–376.
- 711. Fernandez-Marcos PJ, Auwerx J. Regulation of PGC-1α, a nodal regulator of mitochondrial biogenesis. *American Journal of Clinical Nutrition*. 2011;93:884S–890S.
- 712. Garesse R, Vallejo CG. Animal mitochondrial biogenesis and function: A regulatory cross-talk between two genomes. *Gene*. 2001;263:1–16.

- 713. Niklas P, Li W, Jens W, Michail T, Kent S. Mitochondrial gene expression in elite cyclists: Effects of high-intensity interval exercise. *European Journal of Applied Physiology*. 2010;110:597–606.
- 714. Perry CG, Lally J, Holloway GP, Heigenhauser GJ, Bonen A, Spriet LL. Repeated transient mRNA bursts precede increases in transcriptional and mitochondrial proteins during training in human skeletal muscle. *Journal of Physiology*. 2010;588:4795–4810.
- 715. Asghar M, George L, Lokhandwala MF. Exercise decreases oxidative stress and inflammation and restores renal dopamine D1 receptor function in old rats. *American Journal* of Physiology-Renal Physiology. 2007;293:F914–F919.
- 716. George L, Asghar M, Lokhandwala MF. Exercise stimulates transcription factors (Nrf2 & NFκβ), increases antioxidant defenses, decreases oxidative stress, and restores renal dopamine D1 receptor function in aging. *FASEB Journal*. 2008;22:159.6 (Meeting Abstract Supplement).
- 717. Stepto NK, Benziane B, Wadley GD, Chibalin AV, Canny BJ, Eynon N, McConell GK. Short-term intensified cycle training alters acute and chronic responses of PGC1α and cytochrome c oxidase IV to exercise in human skeletal muscle. *PLoS One*. 2012;7:e53080.
- 718. Diaz F, Moraes CT. Mitochondrial biogenesis and turnover. Cell Calcium. 2008;44:24-35.
- Ventura-Clapier R, Garnier A, Veksler V. Transcriptional control of mitochondrial biogenesis: The central role of PGC-1α. *Cardiovascular Research*. 2008;79:208–217.
- 720. Cantó C, Jiang LQ, Deshmukh AS, Mataki C, Coste A, Lagouge M, Zierath JR, Auwerx J. Interdependence of AMPK and SIRT1 for metabolic adaptation to fasting and exercise in skeletal muscle. *Cell Metabolism.* 2010;11:213–219.
- 721. Babraj JA, Vollaard NB, Keast C, Guppy FM, Cottrell G, Timmons JA. Extremely short duration high intensity interval training substantially improves insulin action in young healthy males. *BMC Endocrine Disorders*. 2009;9:3.
- 722. Gormley SE, Swain DP, High R, Spina RJ, Dowling EA, Kotipalli US, Gandrakota R. Effect of intensity of aerobic training on VO₂max. *Medicine and Science in Sports and Exercise*. 2008;40:1336.
- 723. Hood MS, Little JP, Tarnopolsky MA, Myslik F, Gibala MJ. Low-volume interval training improves muscle oxidative capacity in sedentary adults. *Medicine and Science in Sports and Exercise*. 2011;43:1849.
- 724. Gibala MJ, McGee SL. Metabolic adaptations to short-term high-intensity interval training: A little pain for a lot of gain? *Exercise and Sport Sciences Reviews*. 2008;36:58–63.
- 725. Gibala MJ, Little JP, Van Essen M, Wilkin GP, Burgomaster KA, Safdar A, Raha S, Tarnopolsky MA. Short-term sprint interval versus traditional endurance training: Similar initial adaptations in human skeletal muscle and exercise performance. *Journal* of Physiology. 2006;575:901–911.
- 726. Hawley J, Gibala M. Exercise intensity and insulin sensitivity: How low can you go? *Diabetologia*. 2009;52:1709–1713.
- 727. Bartlett JD, Joo CH, Jeong T-S, Louhelainen J, Cochran AJ, Gibala MJ, Gregson W, Close GL, Drust B, Morton JP. Matched work high-intensity interval and continuous running induce similar increases in PGC-1α mRNA, AMPK, p38, and p53 phosphorylation in human skeletal muscle. *Journal of Applied Physiology*. 2012;112:1135–1143.
- 728. Gundersen K. Excitation-transcription coupling in skeletal muscle: The molecular pathways of exercise. *Biological Reviews*. 2011;86:564–600.
- 729. Marini M, Veicsteinas A. The exercised skeletal muscle: A review. *European Journal Translational Myology—Myology Reviews*. 2010;20:105–120.
- 730. Lew H, Quintanilha A. Effects of endurance training and exercise on tissue antioxidative capacity and acetaminophen detoxification. *European Journal of Drug Metabolism and Pharmacokinetics*. 1991;16:59–68.

- 731. Radak Z, Kumagai S, Nakamoto H, Goto S. 8-Oxoguanosine and uracil repair of nuclear and mitochondrial DNA in red and white skeletal muscle of exercise-trained old rats. *Journal of Applied Physiology*. 2007;102:1696–1701.
- 732. Radak Z, Bori Z, Koltai E, Fatouros IG, Jamurtas AZ, Douroudos II, Terzis G, Nikolaidis MG, Chatzinikolaou A, Sovatzidis A. Age-dependent changes in 8-oxoguanine-DNA glycosylase activity are modulated by adaptive responses to physical exercise in human skeletal muscle. *Free Radical Biology & Medicine*. 2011;51:417–423.
- 733. Radák Z, Naito H, Kaneko T, Tahara S, Nakamoto H, Takahashi R, Cardozo-Pelaez F, Goto S. Exercise training decreases DNA damage and increases DNA repair and resistance against oxidative stress of proteins in aged rat skeletal muscle. *Pflügers Archives: European Journal of Physiology*. 2002;445:273–278.
- 734. Nakamoto H, Kaneko T, Tahara S, Hayashi E, Naito H, Radak Z, Goto S. Regular exercise reduces 8-oxodG in the nuclear and mitochondrial DNA and modulates the DNA repair activity in the liver of old rats. *Experimental Gerontology*. 2007;42:287–295.
- 735. Radak Z, Chung HY, Goto S. Systemic adaptation to oxidative challenge induced by regular exercise. *Free Radical Biology and Medicine*. 2008;44:153–159.
- 736. Asami S, Hirano T, Yamaguchi R, Itoh H. Reduction of 8-hydroxyguanine in human leukocyte DNA by physical exercise. *Free Radical Research*. 1998;29:581–584.
- 737. Sampath H, McCullough AK, Lloyd RS. Regulation of DNA glycosylases and their role in limiting disease. *Free Radical Research*. 2012;46:460–478.
- 738. Piao MJ, Kim KC, Choi J-Y, Choi J, Hyun JW. Silver nanoparticles down-regulate Nrf2-mediated 8-oxoguanine DNA glycosylase 1 through inactivation of extracellular regulated kinase and protein kinase B in human Chang liver cells. *Toxicology Letters*. 2011;207:143–148.
- 739. Dhénaut A, Boiteux S, Radicella JP. Characterization of the HOGG1 promoter and its expression during the cell cycle. *Mutation Research*. 2000;461:109–118.
- 740. Zhang X, Tachibana S, Wang H, Hisada M, Williams GM, Gao B, Sun Z. Interleukin-6 is an important mediator for mitochondrial DNA repair after alcoholic liver injury in mice. *Hepatology*. 2010;52:2137–2147.
- 741. Kida H, Yoshida M, Hoshino S, Inoue K, Yano Y, Yanagita M, Kumagai T, Osaki T, Tachibana I, Saeki Y. Protective effect of IL-6 on alveolar epithelial cell death induced by hydrogen peroxide. *American Journal of Physiology-Lung Cellular and Molecular Physiology*. 2005;288:L342–L349.
- 742. Wruck CJ, Streetz K, Pavic G, Götz ME, Tohidnezhad M, Brandenburg L-O, Varoga D, Eickelberg O, Herdegen T, Trautwein C. Nrf2 induces interleukin-6 (IL-6) expression via an antioxidant response element within the IL-6 promoter. *Journal of Biological Chemistry*. 2011;286:4493–4499.
- 743. Helge JW, Stallknecht B, Pedersen BK, Galbo H, Kiens B, Richter EA. The effect of graded exercise on IL-6 release and glucose uptake in human skeletal muscle. *Journal* of Physiology. 2003;546:299–305.
- 744. Pedersen M, Steensberg A, Keller C, Osada T, Zacho M, Saltin B, Febbraio MA, Pedersen BK. Does the aging skeletal muscle maintain its endocrine function. *Exercise Immunology Review*. 2004;10:42–55.
- 745. Pedersen BK, Febbraio MA. Muscle as an endocrine organ: Focus on muscle-derived interleukin-6. *Physiological Reviews*. 2008;88:1379.
- 746. Peake JM, Suzuki K, Hordern M, Wilson G, Nosaka K, Coombes JS. Plasma cytokine changes in relation to exercise intensity and muscle damage. *European Journal of Applied Physiology*. 2005;95:514–521.
- 747. Pedersen BK, Edward F. Adolph distinguished lecture: Muscle as an endocrine organ: IL-6 and other myokines. *Journal of Applied Physiology*. 2009;107:1006.
- 748. Pedersen BK. Muscles and their myokines. *Journal of Experimental Biology*. 2011;214:337–346.

- Ostrowski K, Rohde T, Asp S, Schjerling P, Pedersen BK. Pro- and anti-inflammatory cytokine balance in strenuous exercise in humans. *Journal of Physiology*. 1999;515:287.
- 750. Petersen AMW, Pedersen BK. The anti-inflammatory effect of exercise. *Journal of Applied Physiology*. 2005;98:1154.
- 751. de Waal Malefyt R, Abrams J, Bennett B, Figdor CG, de Vries JE. Interleukin 10 (IL-10) inhibits cytokine synthesis by human monocytes: An autoregulatory role of IL-10 produced by monocytes. *Journal of Experimental Medicine*. 1991;174:1209–1220.
- 752. Wang P, Wu P, Siegel MI, Egan RW, Billah MM. Interleukin (IL)-10 inhibits nuclear factor B (NFB) activation in human monocytes. *Journal of Biological Chemistry*. 1995;270:9558.
- 753. Fukaya T, Suzuki M, Sakaguchi R, Yasukawa H, Yoshimura A. Role of SOCS proteins in inflammation and autoimmune diseases. *Inflammation and Regeneration*. 2011;31:382–392.
- 754. Mosser DM, Zhang X. Interleukin-10: New perspectives on an old cytokine. *Immunological Reviews*. 2008;226:205–218.
- 755. Marcell TJ, McAuley KA, Traustadóttir T, Reaven PD. Exercise training is not associated with improved levels of C-reactive protein or adiponectin. *Metabolism*. 2005;54:533–541.
- 756. Mattusch F, Dufaux B, Heine O, Mertens I, Rost R. Reduction of the plasma concentration of C-reactive protein following nine months of endurance training. *International Journal of Sports Medicine*. 2000;21:21–24.
- 757. Nicklas BJ, Ambrosius W, Messier SP, Miller GD, Penninx BW, Loeser RF, Palla S, Bleecker E, Pahor M. Diet-induced weight loss, exercise, and chronic inflammation in older, obese adults: A randomized controlled clinical trial. *American Journal of Clinical Nutrition*. 2004;79:544–551.
- 758. Friedenreich CM, Neilson HK, Woolcott CG, Wang Q, Stanczyk FZ, McTiernan A, Jones CA, Irwin ML, Yasui Y, Courneya KS. Inflammatory marker changes in a yearlong randomized exercise intervention trial among postmenopausal women. *Cancer Prevention Research*. 2012;5:98–108.
- 759. Fairey AS, Courneya KS, Field CJ, Bell GJ, Jones LW, Martin BS, Mackey JR. Effect of exercise training on C-reactive protein in postmenopausal breast cancer survivors: A randomized controlled trial. *Brain, Behavior, and Immunity*. 2005;19:381–388.
- 760. Kohut M, McCann D, Russell D, Konopka D, Cunnick J, Franke W, Castillo M, Reighard A, Vanderah E. Aerobic exercise, but not flexibility/resistance exercise, reduces serum IL-18, CRP, and IL-6 independent of β-blockers, BMI, and psychosocial factors in older adults. *Brain, Behavior, and Immunity*. 2006;20:201–209.
- 761. Kelly AS, Wetzsteon RJ, Kaiser DR, Steinberger J, Bank AJ, Dengel DR. Inflammation, insulin, and endothelial function in overweight children and adolescents: The role of exercise. *Journal of Pediatrics*. 2004;145:731–736.
- 762. Nicklas BJ, Hsu FC, Brinkley TJ, Church T, Goodpaster BH, Kritchevsky SB, Pahor M. Exercise training and plasma C-reactive protein and interleukin-6 in elderly people. *Journal of the American Geriatrics Society*. 2008;56:2045–2052.
- 763. Kadoglou NPE, Iliadis F, Angelopoulou N, Perrea D, Ampatzidis G, Liapis CD, Alevizos M. The anti-inflammatory effects of exercise training in patients with type 2 diabetes mellitus. *Journal of Cardiovascular Risk*. 2007;14:837.
- 764. Campbell PT, Campbell KL, Wener MH, Wood BL, Potter JD, McTiernan A, Ulrich CM. A yearlong exercise intervention decreases CRP among obese postmenopausal women. *Medicine and Science in Sports and Exercise*. 2009;41:1533.

7 Neurodegenerative Disease

Aparna Raghavan and Zahoor A. Shah

CONTENTS

7.1	Introd	uction	340
	7.1.1	Aging and Neurodegenerative Diseases: The Current	
		Perspective	340
		7.1.1.1 Alzheimer's Disease	341
		7.1.1.2 Parkinson's Disease	342
		7.1.1.3 Huntington's Disease	343
		7.1.1.4 Amyotrophic Lateral Sclerosis	344
	7.1.2	Neurodegenerative Diseases, Diet, and Exercise: What Is the	
		Connection?	345
7.2	Molec	ular Etiology	347
	7.2.1	Genetics	347
	7.2.2	Environment	349
	7.2.3	Aging	351
	7.2.4	Oxidative Stress	351
	7.2.5	Mitochondrial Dysfunction	354
	7.2.6	Protein Aggregation	356
	7.2.7	Excitotoxicity	359
	7.2.8	ND Etiology: At the Crossroads of Prevention and Treatment	360
7.3	Preven	ntive Mechanisms Mediated by Diet	362
	7.3.1	Dietary Restriction: Less Is Good for the Brain	362
		7.3.1.1 DR Mechanisms: Neurotrophic Factors	363
		7.3.1.2 DR Mechanisms: Heat-Shock Response and Protein	
		Chaperones	363
		7.3.1.3 DR Mechanisms: Epigenetics	364
		7.3.1.4 DR Mimetics: A Shortcut?	364
	7.3.2	Mediterranean Diet: What You Eat Is as Important as	
		How Much	365
7.4	Preven	ntive Mechanisms Mediated by Exercise	367
	7.4.1	Exercise Mechanisms: BDNF Is the Key Player	367
	7.4.2	Exercise Mechanisms: Energy Metabolism–Synaptic	
		Plasticity Interface	367
	7.4.3	Exercise Mechanisms: Epigenetics	368

7.5	Summary and Recommendations		
	7.5.1	Summary: Etiology and the Preventive Role of Diet and	
		Exercise	
	7.5.2	Dietary Recommendations	370
	7.5.3	Exercise Recommendations	372
Acknowledgment			
Refe	rences.	-	

7.1 INTRODUCTION

7.1.1 Aging and Neurodegenerative Diseases: The Current Perspective

The past decade saw the oldest baby boomers step into their 60s, many of whom are entering their retirement today, and will continue to do so at a precipitous rate until 2030. It is estimated that the number of people aged 65 years and above will soar to about 72.1 million, which is more than twice their number in 2000 (Aging Statistics-Administration on Aging, Department of Health and Human Services). This meteoric demographic shift conflated with the increased life expectancy estimated for the coming years has overarching implications for health care and the economy as a whole as it encroaches into already weakened pension and social security systems.^{1,2} This pivots the global health-care priorities more toward management of age-related chronic illnesses such as cardiovascular diseases, osteoporosis, and neurodegenerative diseases (NDs).^{3,4} Many of these illnesses pose a challenge to assessing their true nature and severity as they are often underrepresented in mortality statistics and there are not enough reliable epidemiological studies that focus on the suffering and disability such diseases cause.⁵ Neuropsychiatric disorders cover the top 5 of the 10 leading causes of disability worldwide and account for about 37% of years lived with disability-a metric of equivalent years of healthy life lost through time spent in states of less than full health.^{1,6} Moreover, the societal, economic, as well as emotional costs of managing such conditions are not restricted to the individual and often involve loss of the caregiver's productivity and psychological well-being. In the face of such staggering facts juxtaposed with dwindling resources to cope with them, it is imperative that we regroup our efforts toward better prevention or management of neurological and psychiatric disorders.

NDs form a significant component of neurological disorders and are inextricable from the current global concern of age-related chronic illnesses because their incidence and the attendant cognitive and motor deficits are known to increase exponentially with age.⁷ NDs are conventionally characterized by a progressive loss of neurons in distinct anatomical subunits of the nervous system, thus leading to different clinical manifestations.⁸ The longstanding hallmark of these disorders has been the presence of signature molecular lesions or plaques upon pathological examination of the diseased brains.⁹ However, this classification has been debunked by recent progress in the neuropathological diagnosis of these conditions, which have revealed that these lesions are not singularly associated with a particular disease; rather, there is considerable overlap within the molecular as well as clinical features of this group of diseases.^{9,10} This naturally confounds the diagnosis—clinicians often have to rely on highly subjective family history profiling to supplement genetic testing—which is complicated by the lack of specific biomarkers for the diseases. The status quo of ND management lacks in two prominent areas: (1) a unifying dogma pinning the molecular events common to the progression of these diseases, which would tremendously reduce efforts aimed at targeting each disorder separately and (2) impetus for preventive strategies by identifying modifiable lifestyle factors that could delay or alter the progression of the disease. These two areas form the crux of this chapter. We shall begin with an introduction of some of the most common diseases in this category, followed by a detailed assessment of their molecular etiology and mechanisms amenable to preventive manipulation. We shall then discuss the potential of diet and exercise as plausible means to prevent/attenuate neuronal degeneration and delineate the putative mechanisms for their proposed effects. Lastly, we will look into some recommended regimens for diet and physical activity aimed to slow down disease progression and thus eventually delay the onset of morbidity organic to the disease.

There are close to 20 major NDs,^{8,9} but our discussion will be limited to the 4 most commonly encountered, and which we believe would cover the salient aspects of this group of illnesses: Alzheimer's disease (AD), Parkinson's disease (PD), Huntington's disease (HD), and amyotrophic lateral sclerosis (ALS).

7.1.1.1 Alzheimer's Disease

More than a hundred years have passed since Alois Alzheimer published his seminal report on senile plaques (SPs) causing cognitive dysfunction in a middle-aged woman;¹¹ AD is now the most common form of dementia in the United States, accounting for almost 60%–80% of dementia cases.¹² It is estimated that approximately 5.4 million Americans have AD, 96% of whom are aged 65 years or more. This number is expected to nearly triple and to account for more than \$1.1 trillion in health-care costs by 2050.¹² Although AD has not been a major contributor to the burden of mortality when compared to other illnesses such as heart disease, cancer, and stroke, it rose from the eighth leading cause of death in 1999 to the sixth in 2010, surpassing diabetes mellitus.¹³ This is aggravated by the profound socioeconomic burden on immediate family members and/or caregivers during the disease course and after the patient's death.^{14–16}

The neuropathogenesis of AD is linked to two types of lesions or protein aggregates—SPs and neurofibrillary tangles (NFTs)—although their presence has not been unequivocally tied to the disease itself.¹⁷ NFTs are intraneuronal protein clusters of hyperphosphorylated form of the microtubule-associated protein tau (MAPT), whereas SPs are extracellular aggregates of beta-amyloid (A β)—a cleavage product of amyloid precursor protein (APP). It is posited that the neurodegenerative changes at the molecular level kick-start 5–10 years before a clinical diagnosis is made based on the cognitive deficits and other neuropsychiatric changes impairing normal life functions.^{17,18} Typically, AD manifests itself in the form of amnesic type of memory impairment, language, and visuospatial deficits, whereas the motor and sensory abnormalities and seizures do not surface until the late stages of the disease.¹⁸ AD is initially thought to progress from a pre-dementia stage featuring mild cognitive impairment beneath the level of clinical diagnosis to the clinical onset of mild dementia during which declarative memory changes and depression become prominent. The next stage of moderate dementia affects other domains of cognition

and also the noncognitive behavioral aspects, which put tremendous strain on the caregiver. This then develops into a full-blown impairment of cognitive function or the final stage of dementia, wherein patients are unable to articulate their needs or carry out even the simplest of motor tasks and are entirely dependent on their caregivers. The life expectancy of an AD patient is reduced by one-third upon clinical diagnosis, and the average duration of survival is between 5 and 8 years,¹⁹ but it is hoped that future interventions targeting the crucial time-window of subclinical dementia would delay onset or even prevent the development of the disease.

7.1.1.2 Parkinson's Disease

PD—or "Shaking Palsy" as James Parkinson first described it in 1817²⁰—is the second most common ND after AD, affecting more than 1% of 55-year-old individuals and more than 3% of those over age 75 years.²¹ It is primarily a motor disorder resulting in the progressive loss of functionality in the upper extremities, although cognitive dysfunction and other psychiatric problems are fairly common.²² With the incidence of PD expected to double within the next 20 years, it is predicted to overburden the caregiving system considering the level of disability it causes.²³ The annual economic burden of PD in the United States is estimated to be \$10.8 billion a combination of direct and indirect costs—of which 58% is related to direct medical costs.²³ The indirect costs, including loss of productivity for patients and caregivers due to early retirement or termination, have societal and psychological costs associated with it and often take a greater toll than represented.

The classic clinical features of PD include tremors, hypokinesia (decreased movement), bradykinesia (slow movement), and rigidity. Psychiatric abnormalities include depression and visual hallucinations. Dementia is seen in at least 20% of the cases.²¹ Pathologically, the brains of PD patients are found to be deficient in dopaminergic neurons in the substantia nigra region, with intracytoplasmic inclusions called Lewy bodies in the remaining nigral neurons.²⁴ With the discovery of new subtypes of the disease that do not display Lewy body pathology, a combination of neuropathological and genetic studies is needed to make an informed decision. Since there are no definitive diagnostic tests for PD, a high premium is placed on the clinical manifestations of the disease, especially the motor symptoms. Bradykinesia is an easily recognizable and the principal diagnostic criterion; it involves difficulty in programming and executing actions involving movement and also features slower reaction times. Patients have difficulty with daily activities like eating or buttoning a shirt. It is posited that a combination of muscle weakness, rigidity, tremors, inaccurate movement, and the slowing of thought contribute to bradykinesia.²⁵ It is also known to correlate best with the extent of dopamine deficiency in the substantia nigra region. "Freezing" or motor block, in which patients are unable to initiate voluntary movements and appear "stuck" to the ground, is another classical symptom of PD. Apart from this, patients also present with postural deformities ("dropped head" or "bent spine"), decreased stride length while walking, speech disorders (failure to find words), and drooling. There is also significant cognitive retardation, dementia, occurrence of sleep disorders, and a general lack of initiative to carry out daily activities.26,27

Despite the promise offered by levodopa and other such symptomatic medications, the prognosis of PD patients is bleak, involving considerable motor disability within 5–10 years following the onset of the disease.²⁷ Lack of potent drug therapy and specific biomarkers for early detection of the disease underscore the urgent need to develop better diagnostic techniques as well as investigate the role of any preventive measures that could reduce disability in susceptible individuals.

7.1.1.3 Huntington's Disease

George Huntington's landmark description of inherited chorea—an involuntary movement disorder—spawned a cascade of scientific research on the disease that later came to bear his name.²⁸ We now know that it is an incurable autosomal dominant ND caused due to an aberration in the *HTT* gene coding for the protein huntingtin.²⁹ The mutated gene contains an expanded CAG trinucleotide repeat sequence. While the mean onset of disease is 40 years, it could decrease in people with high CAG repeat sizes.³⁰ HD is found in about 4–10 people per 100,000 in western countries, although the number could be as low as 0.38 or as high as 15 in some population groups, possibly due to their geographical isolation.^{31,32} As with any disease causing severe disability, HD has farreaching ramifications for the caregiver and the physician, who respond differently from the patient considering that they are influenced by different aspects of the disease,³³ thus stressing the need for developing a holistic approach toward management of HD.

HD neuropathology depicts selective atrophy in the neostriatal region comprising the caudate and putamen, although with disease progression, there is widespread brain atrophy.³¹ The medium spiny neurons in the striatum are most vulnerable to degeneration resulting in an increased undesirable motor stimulation of the body.³⁴ which translates to the clinical manifestations of the disease. The diagnosis of HD is not as confounding as that for other NDs owing to its high genetic predictability. Upon positive genetic diagnosis, the classical features essential for a clinical diagnosis could be grouped under extrapyramidal movement disorder, cognitive dysfunction, and psychiatric or behavioral disturbance.³¹ The breakdown of motor function is due to a combination of voluntary and involuntary movement disorders. The involuntary movements initially proceed through Chorea-a condition whose etymology is linked to the Greek word for "dance"-characterized by rapid, irregular, jerky movements of the trunk, face, and limbs. As the disease progresses, the chorea usually becomes less pronounced and is replaced by bradykinesia and rigidity, and eventually, akinesia, which are functionally more debilitating.³⁵ The deficits in voluntary movements start off as gait abnormalities and slowed performance in executing functions such as handwriting and equipment usage and lead to a progressive decline of fine motor skills, speech, and mobility to a point where the patient loses volitional control over limb movements and becomes nonambulatory and completely dependent on the caregiver for daily activities.^{22,31} Cognitive deficits in HD are often apparent before a clinical diagnosis is made; they include bradyphrenia (slowness of thought), defective recall, and personality changes. Patients suffer from "subcortical dementia," which affects the ability to learn and recall new skills as well as the visuospatial memory. In advanced stages, dementia leads to a global decline of all cognitive faculties.³¹ As far as the psychiatric manifestations of the disease is concerned, depression offers the most cause for concern, considering that it is prevalent in about 40% of the patients, and that they are at a higher risk to commit suicide.^{31,36}

What makes HD the most tractable of ND in addition to its genetic predictability is the bounty of reliable biomarker and clinical flags available for the subclinical or premanifest stage of the disease—the stage most amenable to preventive interventions in progressive disorders.³² This could probably serve as a perfect model system for designing early strategies for diseases with more equivocal diagnostic criteria such as AD and PD.

7.1.1.4 Amyotrophic Lateral Sclerosis

The term "ALS" was first coined by Jean-Martin Charcot in 1874, who is not only credited for expertly outlining the link between the clinical features of ALS to the autopsy findings but also for pioneering the clinicopathological approach in the medical community.³⁷ ALS-also popularly known as Lou Gehrig's disease-is an adult onset, rapidly progressive, and the most common form of motor neuron disease resulting in the degeneration of both the upper and lower motor neurons. It is characterized by muscular atrophy, weakness, and fasciculation resulting in paralysis, and eventually, death within 3-5 years of onset of symptoms.³⁸ Not only does ALS entail the morbidity and the heavy socioeconomic costs common to most neurodegenerative conditions, but also one of the worst mortality rates.³⁹⁻⁴¹ It is estimated to be present in 4-6 individuals per 100,000, with the prevalence peaking in the 60-75 age group, although some pockets in the Western Pacific have much higher prevalence.⁴² The most common ALS cases are sporadic, that is, lack a genetic component (95%), whereas the rest have a family history (familial).⁴³ A significant proportion of the familial cases result from mutations in the gene coding for the copper-zinc superoxide dismutase enzyme (SOD1).⁴⁴ There is no specific etiology for the disease, although some environmental factors such as exposure to heavy metals, epidemiological features, and trauma are implicated in addition to genetic predisposing factors.^{38,45,46}

The clinical presentation of the disease is a combination of upper and lower motor neuron symptoms. It may originate most commonly as an asymmetric focal weakness of the extremities (difficulty gripping and stumbling) or have bulbar origins (dysphagia or dysarthria),⁴⁷ or less commonly, have respiratory muscle onset.⁴⁸ Regardless of onset, the disease eventually spreads to, and wastes away other muscles of the body. Another cardinal feature of ALS is the hyperreflexia of the atrophied muscles, with the lack of any sensory disturbance.⁴⁷ The classical upper motor neuron symptoms include poor dexterity and gait, difficulty in swallowing, stiffness, and imbalance. Lower motor neuron symptoms are a result of muscle weakness, atrophy, fasciculations, and cramps³⁸ and include difficulty in performing fine motor tasks such as buttoning and writing, leg weakness leading to problems in climbing, arising, as well as balance. Weakness of the tongue and pharyngeal muscles could lead to coughing, choking, and aspiration of food, which could be potentially fatal. Although essentially a motor neuron disease, a small number of ALS patients develop frontotemporal dementia, and about 30%-50% show frontotemporal executive function deficits.⁴⁷ These may become apparent before or after the onset of motor dysfunction and are usually subtler than the symptoms of full-blown dementia. Early decline in interpersonal conduct, mental rigidity, distractibility, and impairment in verbal fluency and attention are some of the characteristics observed in such patients.⁴⁹ The clinical course of the disease is nearly linear with the degeneration inexorably progressing throughout the body and death usually occurring due to respiratory failure.⁴⁷

The knowledge gleaned from the prevalent, clinical manifestations of the most common NDs along with the monumental societal and economic costs associated with their symptomatic management clearly highlight one thing: the need for costeffective strategies to counter the etiological mechanisms preceding the clinical or morbid states of the disease, thus greatly reducing the disability burden inherent to them. In this regard, the potential of diet and exercise as disease-modifying lifestyle factors has generated a lot of optimism in the neurological community, due to their reported benefits as well as their relatively seamless incorporation into our daily lives.

7.1.2 NEURODEGENERATIVE DISEASES, DIET, AND EXERCISE: What Is the Connection?

There is no doubt of the tremendous influence our environment and lifestyle have on our overall state of well-being; they are known to influence the progression of many chronic diseases, including those affecting the nervous system.⁵⁰ However, in line with our ever-increasing obsession with short-term gains, we have found means to circumvent our nutritional and exercise needs, to our own detriment. Very often, the negative consequences of this do not manifest until mid-to-late life, but when they do, there is little scope for reparative measures. Thus, to have a fighting chance against such chronic illnesses, we need to start early.

In this regard, it is important to understand the dynamic relationship between the brain and the environment. Our lifestyle has a dramatic influence on brain plasticity and neuronal function, which in turn determine the clinical course of NDs, when corrected for genetic factors.⁵¹ The etiology of most NDs being multifactorial, diet and exercise as potential preventive modalities offer several advantages: (1) they have a broad scope of action, (2) they are noninvasive and safe, (3) they have strong translational potential,⁵² and (4) they are relatively cheap and can be easily incorporated in our daily life. Moreover, there is evidence to suggest a synergistic role between diet and exercise in brain function.^{53,54} These implications have spawned a number of epidemiological studies investigating their role in neurodegenerative conditions, and the results have been encouraging, if not conclusive.

Studies correlating a high-calorie diet with increased risk for AD indirectly point to the salutary role of dietary restriction (DR) in preventing the disease.^{55,56} In particular, a diet rich in saturated fat and cholesterol is associated with an increased risk for dementia, and AD in particular,^{55,57,58} whereas fish consumption and high intake of monounsaturated fatty acids (MUFA) and polyunsaturated fatty acids (PUFA) is thought to reduce the risk in most cases.^{59–63} Also, there is overwhelming evidence linking adherence to a Mediterranean-type diet (vegetables, legumes, cereals, fish, and MUFA) to reduced risk of AD, cognitive decline, and overall, reduced AD-related mortality.^{64–67} However, it remains to be established whether or not these dietary components work synergistically, thus warranting their use as a whole diet, as opposed to the individual components. ⁶⁸ In addition to diet, exercise is another modifiable factor known to retard disease progression, modulate pathogenic mechanisms, and reduce cognitive decline associated with AD and pre-dementia syndromes.^{69–72} Interestingly, exercise is independently associated
with reduced AD risk, even when considered together with diet, thus suggesting that a combination of diet and exercise could provide additive benefits to patients.⁷³

In PD, although there is no conclusive basis for the role of overall fat intake in pathogenesis,⁷⁴ consumption of unsaturated fatty acids may proffer a reduced risk.⁷⁵ Similar to AD, a Mediterranean-type diet plus reduced consumption of saturated fat and moderate intake of alcohol may protect against PD.^{76,77} The preconceived dismal role of exercise in rehabilitation of PD patients has been recently debunked, thanks to the abundant bench and translational research in the area.^{78,79} Exercise is now associated with improvement in the motor symptoms such as postural stability and tremor,^{80,81} nonmotor symptoms including depression,⁸² and overall functional outcome.^{83,84} Also, the studies utilizing resistive and high-intensity exercise paradigms in early and late stages of the disease underscore the importance of forced rather than self-selected physical activity, to attenuate disease progression.^{85–87}

There is not a lot of epidemiological data supporting the role of dietary modification in delaying progression of HD. However, a diet low in tryptophan—an amino acid whose abnormal metabolism is implicated in the pathogenesis—is argued to attenuate the progression of disease or delay onset of symptoms in high-risk candidates.^{88,89} Also, animal studies suggest that DR could have a neuroprotective effect in HD by boosting the body's defense against protein aggregation and apoptosis.⁹⁰ The link between exercise and HD progression is yet to be clinically established, but there are some positive results from animal studies.^{91–93}

Dietary and exercise interventions in ALS have shown moderate effects in the clinical level and warrant further investigation.⁹⁴ Dietary vitamin E supplementation has by far been the most promising intervention at the clinical level.^{95,96} Contrary to conventional knowledge, a low-calorie diet or DR does not offer any benefits in the pathogenesis or functional outcome of the disease.^{97–99} This discrepancy is attributed to the malnourished and hypermetabolic state of ALS patients, which would only worsen with caloric restriction.¹⁰⁰ Thus, it is postulated that ALS patients could show increased survival and improved quality of life by consuming a high-energy or a ketogenic diet.^{101,102} As far as exercise and ALS are concerned, there has been a conspicuous lack of consensus due to studies associating vigorous physical activity to increased ALS risk.^{103–105} However, evidence supporting the use of moderate physical activity and resistance training in improving muscle strength and function and enhancing the quality of life in ALS patients justifies further consideration.^{106–108}

A comprehensive analysis of the clinical as well as the preclinical data from the various nutritional and physical activity–based studies strongly point to their plausibility as effective preventive and/or adjuvant therapy. They promise to relieve the overwhelming burden on drug therapies by offering means to delay the need for their intervention. However, there are many caveats associated with such epidemiological studies. For example, many of these studies were case-control and not randomized control trials, which introduces a huge correlation-causation bias. Also, there was a disproportionate focus on functional outcomes of diet and exercise modalities rather than the biochemical/molecular changes they initiate; this could have rejected some of the nascent, promising interventions that did not have overt functional effects yet. Some of the other confounding issues are that regarding statistical power, poor patient compliance, and dosage standardization.⁹⁴ This calls for not only better designed randomized

control trials but also a comprehensive understanding of the molecular mechanisms in play. Some questions that need to be addressed at this stage, and which would appraise the potential of diet and exercise in the etiological process of NDs, are as follows: What mechanisms do diet and exercise modulate, relevant to ND prevention? Where do these mechanisms feature in the global molecular etiology of the disease?

To be able to better answer these questions, a detailed assessment of the molecular etiology of NDs is needed, which forms the basis of Section 7.2.

7.2 MOLECULAR ETIOLOGY

7.2.1 GENETICS

The multifactorial nature of NDs is evident from the presence of both familial and sporadic forms of diseases. Although the familial forms of diseases represent a minority of the cases (5%–10%), their genetic causes have provided many insights into the pathogenic mechanisms associated with the sporadic forms as well. Although they lack a Mendelian component, sporadic forms have many susceptibility genes associated with them, which—in conjunction with various environmental factors and the aging of the body—are known to increase the likelihood of the disease.¹⁰⁹

AD: Familial forms of AD are caused by fully penetrant mutations in three genes: *Amyloid precursor protein (APP), Presenilin 1 (PSEN1)*, and *Presenilin 2 (PSEN2)*. The APP, as the name suggests, is the precursor of A β , a 4-kDa peptide and principal component of SPs found in AD patients. APP is proteolytically processed by a group of enzymes called α , β , and γ secretases to obtain two major A β fragments, A β_{40} and A β_{42} , consisting of 40 and 42 residues, respectively. A β_{42} is the pernicious one of the two and is the main culprit implicated in AD pathogenesis.¹¹⁰ The presenilins form a component of the γ secretase complex and along with the other secretases determine the balance between amyloidogenic and nonamyloidogenic fragment content of A β . The mutations in the three key genes involved in synthesis and processing of APP is thought to tip the balance toward production A β_{42} and cause the familial form of the disease.¹¹¹

One of the well-established genetic susceptibility factors for AD is the *apolipoprotein* E (*APOE*) gene.^{112,113} The *APOE-e4* allele is linked to a three-fold increased risk of AD, although its impact on the disease risk decreases with age. It is associated with increased plasma concentrations of cholesterol and thought to enhance A β deposition and formation of SPs; however, it is neither sufficient nor necessary to cause the disease.¹¹³ The other susceptibility genes whose polymorphs carry an increased risk for the sporadic forms of AD include the β -secretases, *BACE1* and *BACE2*, and other components of γ -secretase activity, *Nicastrin* and *PEN2*,^{114–116} thus establishing errant A β biosynthesis as the common theme linking both the familial and the sporadic forms of the disease.

PD: As with AD, the etiology of the most commonly encountered sporadic form of PD seems to be a complex interplay of susceptibility genes, environmental exposure, and aging. The key causative genes for the rare, familial form of the disease are α -synuclein, parkin, ubiquitin carboxyl-terminal hydrolase L1 (UCH-L1), PTEN-induced putative kinase 1 (PINK1), and leucine-rich repeat kinase 2 (LRRK2). The protein α -synuclein is thought to function as a molecular chaperone and be involved in vesicular trafficking.^{117,118} Mutations in the gene (PARK1 and PARK4) are known

to increase the tendency of α -synuclein to form aggregates and eventually form Lewy bodies.^{119,120} Mutations in the *parkin* gene (*PARK2*) are known to cause early-onset PD devoid of the classical Lewy body formation.¹²¹ Mutated parkin, a ubiquitin E3 ligase, is thought to interfere with the normal functioning of the ubiquitin-proteasome system essential for the degradation of misfolded proteins.¹²² The implication of a mutation in the *UCH-L1* gene in PD cemented the role of the ubiquitin-proteasomal dysfunction in the pathogenesis.¹²³ The mutation in *UCH-L1* impairs its hydrolase activity, thus affecting ubiquitin hydrolysis and eventual death of dopaminergic neurons.¹²⁴ The *PINK-1* gene encodes a mitochondrial protein involved in stress response pathways, mutations in which could hinder the ability of neurons to cope with stress.¹²⁵ The other major causative gene is the *LRR2* gene, whose physiological role is not precisely established, but is known to cause an autosomal dominant form of the disease.¹²⁶

Many of the susceptibility factors of PD include variants of the genes implicated in the familial forms. The Rep1 polymorphism in the promoter region of the α -synuclein gene,¹²⁷ a variant in the *LRRK2* gene,¹²⁸ and mutations in the gene encoding the *MAPT*¹²⁹ have been associated with increased PD risk. The other susceptibility factors include genes coding for proteins involved in dopamine transport and metabolism and the detoxification of xenobiotics. A plethora of studies have been carried out in this regard, yet conclusive results of an increased risk of PD have been associated only with polymorphisms in the genes encoding for proteins glutathione S-transferase P1¹³⁰ and cytochrome P450 2D6.¹³¹

HD: HD is inherited in an autosomal dominant manner and is caused by expanded CAG repeats in the exon 1 of the *HTT/huntingtin* gene. The number of CAG repeats—which code for a track of glutamine residues in the protein—is the major contributor of variability in terms of age of onset and severity of symptoms as it determines the ability of the mutated huntingtin to form aggregates.^{132,133} Wild-type huntingtin is thought to be involved in embryonic development, vesicle transport, postsynaptic signaling, transcriptional regulation, and cell survival pathways.¹³⁴ Some of the other genetic variants altering the age of onset of the disease are those in the NR2A and the NR2B subunits of the N-methyl-D-aspartate receptor (NMDAR).¹³⁵ Also, the polyglutamine expression in HD has been shown to impair the expression of a number of genes encoding synaptic and intraneuronal signaling proteins, suggesting that synaptic dysregulation could be a key pathogenic mechanism.¹³⁶

ALS: ALS is mainly a sporadic disorder with about 5%–10% prevalence of familial cases. The key causative genes for familial ALS are *SOD1*, *alsin* (*ALS2*), *senataxin*, and *VAMP-associated protein B* (*VAPB*). Superoxide dismutases are a class of free-radical scavenging enzymes, with SOD1 specifically involved in converting the highly reactive superoxide anion, a by-product of oxidative phosphorylation (OXPHOS) in the mitochondria, to the relatively benign hydrogen peroxide. SOD1 mutations account for about 20% of the familial cases, and although most mutations reduce enzyme activity, others do not affect catalytic function.¹³⁷ However, neither age of onset nor the course of the disease seem to correlate with SOD1 activity,^{138,139} suggesting that the effects of mutant SOD1 are a result of a "gain-of-toxicity" rather than "loss-of-function." Mutations in the *ALS2* gene coding for alsin are predominantly loss-of-function and are known to render neurons vulnerable to oxidative stress and disrupt vesicle trafficking.^{140,141} The function of alsin in unknown,

but it was found to suppress mutant SOD1-mediated neurotoxicity in motor neuron cell lines.¹⁴² Mutations in the *senataxin* gene encoding for a DNA and RNA helicase involved in the defense against oxidative DNA damage¹⁴³ were associated with juve-nile autosomal dominant form of ALS.¹⁴⁴ Defects in the *VAPB* gene are linked to adult-onset autosomal dominant and atypical ALS and cause decreased anchoring of lipid-bound proteins and disruption of vesicular transport.^{137,145}

Studies aimed at identifying candidate susceptibility genes for ALS have so far met with conflicting results. However, information gathered from the genes causing the familial forms of the disease have highlighted the importance of polymorphisms in DNA repair genes, angiogenesis factors, and genes coding for neurofilaments in ALS etiology.¹⁰⁹

7.2.2 Environment

The impact of the environment on the etiology of NDs cannot be underestimated. In fact, many of the susceptibility factors discussed in Section 7.2.1 have emerged from gene–environment interactions accumulating over a period of time, thereby necessitating studies to identify such factors relevant to ND etiology. The results from such studies have not always been conclusive due to biases and confounding factors inherent in such case-control groups. Nevertheless, few have clearly emerged as potential risk factors for NDs (Table 7.1).^{109,136,146–148}

Metal exposure is probably one the most thoroughly studied environmental factors in ND pathogenesis and has led to a better understanding of the intracellular processes that could trigger the disease. Higher levels of metals have been detected in regions of the brain specific to the insult caused by the disease. For example, higher copper and zinc levels were found in the SPs and NFTs in AD brains, and PD midbrains have higher iron levels. High levels of these metals are linked to oxidative stress conditions in the brain, as they are capable of generating free radicals and setting off the oxidative stress cascade.¹⁴⁹

Exposure to metals in the perinatal period is deleterious to neuronal development. Low levels of lead exposure during the developmental stage is thought to have a protracted negative effect on APP processing and A β biosynthesis in later stages of life.¹⁴⁶ It is also known to induce oxidative DNA damage and reduce DNA methyltransferase activity. The ability of lead to have late onset effects on the activities of enzymes and on DNA oxidation revealed an interesting facet to the gene-environment interaction—epigenetics. Epigenetics is an umbrella term for all the mechanisms involved in gene expression changes and their subsequent inheritance, by means not involving DNA sequence changes.¹⁵⁰ It principally involves processes such as DNA methylation, histone deacetylation, and RNA editing, which occur after DNA synthesis. DNA methylation patterns are established in utero and are susceptible to environmental modification until postnatal development, thus establishing a gene imprint that lasts for a lifetime.¹⁵¹ Early life exposure to lead is thought to enhance or repress the functions of various genes by influencing DNA methylation. In AD, hypomethylation of the APP promoter and the subsequent increase in APP and A β production could contribute to the oxidative damage. DNA methylation is also linked to DNA repair and oxidative DNA damage, thus potentially modifying

TABLE 7.1

Some of the Important Environmental Factors Associated with ND Etiology

Disease	Environmental Factor	Comment
AD	Metals (iron, copper, zinc, lead)	Increased risk; inconclusive results
	Pesticides	Increased risk
	Solvents	Increased risk; inconclusive results
	Brain injuries	Increased risk
	Infection and inflammation	Increased risk
	Caloric restriction	Protection
	Antioxidants	Protection
	Mediterranean diet, fruits, vegetables	Protection
	Fish, MUFA, PUFA	Protection
PD	Metals (iron, copper, manganese, lead)	Increased risk; conflicting results
	Pesticides, herbicides	Increased risk
	Head injuries	Increased risk
	Tobacco	Protection
	Caffeine from coffee and tea	Protection
	Mediterranean diet, fruits, vegetables	Protection
	Fish	Protection
HD	Metals (iron, copper, manganese)	Increased risk; conflicting results
	Environmental enrichment (cognitive and physical activities)	Protection
ALS	Metals (lead)	Increased risk
	Pesticides, insecticides	Increased risk
	High-intensity sports	Increased risk
	Head injuries	Increased risk

Source: Modified from Migliore, L. and Coppedè, F., Mutat. Res., 674, 73-84, 2009. With permission.

stress-response pathways. Moreover, such changes in gene expression patterns are not limited to metal exposure, but also other environmental components such as diet and chemicals. Epigenetic changes can thus translate early environmental exposure to far-reaching changes in cellular mechanisms, which could eventually be useful in understanding our susceptibility to age-related diseases. Although its role in cancer is well established, its contribution to NDs is still in its infancy.

The role of "designer drugs" and other agricultural chemicals is best established in PD, wherein the effect of 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine (MPTP) led to the establishment of MPTP-based animal models of the disease. MPTP and related pesticides such as rotenone, paraquat, and maneb possess oxidative stressgenerating properties,¹⁵² which cause mitochondrial dysfunction, a classical feature of many NDs. Some of the other interesting environmental components implicated in NDs are air pollution and diet. Air pollution is associated with neuroinflammation and deposition of A β_{42} and α -synuclein, beginning at childhood.¹⁵³ There is evidence that nano-sized particulate matter could gain entry into the brain and cause NDs.¹⁵⁴ The influence of diet in NDs has been previously described and will continue to feature in the following discussion.

7.2.3 AGING

Aging is an inevitable physiological process that slows down our sensory, motor, and cognitive machinery with time. While the probability of developing NDs increases remarkably with age, it is not irreversibly linked to neurodegeneration and death. What separates people who age and people who are struck with a ND while they age is the rate of aging and the ability of the body to compensate for age-associated deficits.¹⁵⁵

Aging is broadly characterized by increased oxidative stress,¹⁵⁶ impaired energy metabolism,¹⁵⁷ accumulation of dysfunctional proteins,¹⁵⁸ and DNA damage.¹⁵⁹ Disease-specific genes, susceptibility factors, and the environment trigger molecular events similar to aging and exacerbate its effects to a toxic level in specific neuronal populations.¹⁵⁵ Aging is associated with increased levels of oxidized proteins, DNA, and lipids in the brain. Age-related protein modifications such as carbonylation, nitration, and lipid peroxidation markers such as 4-hyrodynonenal (4-HNE) and malondialdehyde are markedly elevated in NDs.^{160–162} Protein accumulation is a prominent feature of aging and is thought to be a combination of increased protein damage and the breakdown of the proteasomal and autophagic systems handling such proteins.¹⁶³ However, accumulation of aggregate components such as A β , α -synuclein, and SOD1 in the diseased brains is greater than that seen in normally aging brains.^{164,165} Aging also alters neurotransmitter pathways and the production of neurotrophic factors like brain-derived neurotrophic factor (BDNF)—changes that are enhanced in diseased brains.^{166,167}

Taken together, the etiological model of NDs is built on the premise that aging exposes the chinks in the neuronal armor, and a combination of genetic and environmental factors decide whether the weakened neurons succumb to the disease or not. Aging is a perpetual component in almost all cellular processes responsible for neurodegeneration and shall be reintroduced wherever relevant in the following sections (Sections 7.2.4–7.2.7) delineating the molecular mechanisms of causation.

7.2.4 OXIDATIVE STRESS

Oxygen is essential for most cellular processes in our body, yet the by-products of its metabolism result in the formation of reactive intermediates, which could lead to oxidative stress, if left unchecked. These reactive oxygen species (ROS) serve important physiological functions such as signal transduction, gene transcription, and response to noxious stimuli, at low concentrations.¹⁶⁸ The deleterious effects of ROS become apparent at higher concentrations, which under normal circumstances are nullified by the robust antioxidant defense consisting of enzymes and free-radical scavengers, thus maintaining homeostasis. When the oxidant-antioxidant balance goes awry, oxidative stress and its attendant toxic events ensue.

Oxidative stress is the hub of all processes leading to neurodegenerative damage. Whether it triggers the cascade of events that eventually lead to the death of neurons is debatable, but it is intimately involved in the major events leading to the formation of the protein aggregates—the pathological hallmark of NDs—as well as the consequent damage exacted by them resulting in neuronal demise. The brain is particularly vulnerable to the noxious effects of oxidative stress due to many potentiating factors. It is one of the highest consumers of oxygen and glucose on a per weight basis, making it more susceptible to ROS overload.¹⁶⁹ Neuronal membranes are rich in unsaturated fatty acids, making them easy candidates for free-radical-mediated lipid peroxidation, causing membrane damage and the disruption of the activity of membrane-bound proteins.¹⁷⁰ It also shows lower basal antioxidant capacity, which is aggravated by the reduced antioxidant surveillance and regenerative capacity of the aged brain.¹⁷¹

The mitochondrion serves as the main source of intracellular ROS through the OXPHOS reaction occurring in the electron transport chain, wherein molecular oxygen is reduced to generate the highly reactive superoxide anion ($O_2^{\bullet-}$). This reaction is essential as it activates molecular oxygen, whose reaction with physiological molecules is otherwise energetically unfavorable.¹⁷¹ Once generated, the superoxide ion is capable of triggering the production of more ROS such as hydrogen peroxide (H₂O₂), hypochlorous acid (HOCl), hydroxyl radical (HO[•]), and hydroperoxyl radical (HOO[•]). Of particular interest is the Fenton chemistry that involves the iron catalyzed reaction of hydrogen peroxide with superoxide ion forming the highly reactive HO[•].^{172,173}

$$Fe^{3+} + O_2^- \rightarrow Fe^{2+} + O_2(Step I)$$

$$Fe^{2+} + H_2O_2 \rightarrow Fe^{3+} + OH^- + OH(Step II)$$

Overall,

$$O_2^- + H_2O_2 \rightarrow OH + OH^- + O_2$$

These transition-metal catalyzed reactions are key to the disease pathogenesis because the brains in most NDs show elevated levels of Cu²⁺, Zn²⁺, and Fe³⁺, thus potentiating the reaction and the formation of ROS.¹⁴⁷ A β aggregation and α -synuclein-mediated damage in AD¹⁷⁴ and PD,¹⁷⁵ respectively, have been shown to involve interaction with transition metals and the concurrent formation of hydroxyl radicals. ROS is also generated by the peroxisomes that mainly produce H₂O₂, and NADPH oxidase (NOX), which releases H₂O₂ and 'O₂⁻. Both aging and AD show an increase in NOX activity.¹⁷⁶ Another mitochondrial ROS source is the enzyme monoamine oxidase that catalyzes the oxidation of biogenic amines such as dopamine, serotonin, and norepinephrine and is implicated in PD pathology.¹⁷⁷

Along with ROS, reactive nitrogen species (RNS) generated as by-products of nitric oxide (NO[•]), also contribute to oxidative stress. Nitric oxide, a free radical in itself, is an important signaling molecule functioning almost like a neurotransmitter in the

CNS.¹⁷⁸ It plays a role in immune response and vasodilation of smooth muscles, thus regulating blood flow.¹⁷⁶ During inflammatory conditions prevalent in the ND brain, NO is released by activated microglia and endothelial cells by the action of inducible nitric oxide synthase (iNOS) and endothelial nitric oxide synthase, respectively. NO then reacts with superoxide to form peroxynitrite (ONOO⁻)—a potent RNS, which could decompose to form HO[•].¹⁶⁹ iNOS-mediated NO release and ONOO⁻ generation is thought to be one of mechanisms of mutant-SOD1 toxicity in ALS¹⁷⁹ as well as the reduction in levels of aconitase—an enzyme of the tricarboxylic acid cycle sensitive to inhibition by ROS—in the striatum of HD patients.¹³⁴

Our body has devised various adaptive responses to combat the oxidative onslaught of ROS and RNS. The antioxidant system comprises enzymes such as the superoxide dismutases (Mn-SOD and Cu/Zn-SOD) that convert O_2^- to H_2O_2 and O₂ and catalase and glutathione peroxidases (GSHPx) that convert H₂O₂ to water. Also, there are nonenzymatic components that act as direct antioxidants by scavenging/chelating ROS or by inducing antioxidant enzymes that include thioredoxin, glutathione (GSH), vitamins E and C, uric acid, and ubiquinone.^{176,177} Evidence of reduction in antioxidant enzyme activity and scavengers is noted in both AD¹⁸⁰ and PD.¹⁸¹ GSH depletion is one of the markers of nigral degeneration, and its levels parallel the severity of PD.¹⁷⁹ In addition to antioxidants, there are protein chaperones responsible for tagging oxidatively modified proteins for their eventual disposal, such as heat shock protein 70 (HSP70), glucose-regulated protein 78 (GRP78), and HSP27, whose activities are also compromised in NDs.¹⁸² One of the crucial targets of ROS is proteins, particularly at their cysteine residues.¹⁸³ In addition to oxidation by transition metals, proteins are also vulnerable to covalent modifications by lipid peroxidation products such as 4-HNE, acrolein, and malondialdehyde. The formation of these adducts sets into motion a host of other processes that result from either the loss of protein function¹⁸⁴ or the toxicity of the adduct itself.¹⁸⁵ Acrolein and 4-HNE are known to disrupt sugar and glutamate uptake, by acting on membrane Na⁺/K⁺- and Ca²⁺-ATPases and glutamate and glucose transporters, which in turn cause severe energy depletion and Ca²⁺ influx, leading to synaptic degeneration and neuronal death.^{171,186} Ca²⁺ ions are signaling gateways responsible for cell-cell communication, synaptic plasticity, and neuronal integrity.¹⁸⁷ Its cytosolic levels are tightly regulated, and any process—or in this case ROS—causing Ca^{2+} overload leads to a suicidal cascade fueled by further influx of mitochondria-derived ROS and RNS.177

There is considerable ROS-mediated DNA damage in NDs that hampers DNA repair mechanisms leading to mutations in the sequences coding for essential proteins.¹⁸⁸ Mutant SOD1 toxicity in ALS is thought to involve oxidative DNA damage due to reduced capacity of DNA repair enzymes.¹⁸⁹ The CAG trinucleotide expansion seen in HD is also found to be due to erroneous DNA repair induced by damaged DNA.¹⁹⁰ While the primary hypothesis of oxidative stress–triggered neurodegeneration rests on the breakdown of antioxidant defense in the face of ROS overload, there are many other processes that result from and further contribute to the ROS insult (Figure 7.1). There is continuous crosstalk between oxidative stress and other cellular mechanisms (discussed in Sections 7.2.5–7.2.7), forming a self-amplifying loop of events that cause synaptic damage and neuronal death.



FIGURE 7.1 Oxidative stress fuels many other toxic insults in ND etiology. Oxidative stress causes and results from mitochondrial damage, protein aggregation, and excitotoxicity thereby forming a self-amplifying loop.

7.2.5 MITOCHONDRIAL DYSFUNCTION

Mitochondria are responsible for ATP production that fuels most cellular processes. Being an energy-intensive organ, the brain depends on the mitochondria for most of its energy supply by a process called oxidative phosphorylation. Mitochondria are not only the major sites of ROS generation but are also involved in the adaptive response against them by activating antioxidant enzymes.¹⁹¹ They control apoptotic pathways, buffer cytosolic Ca²⁺, and maintain respiratory efficiency by their ability to undergo fission/fusion.¹⁹² Mitochondrial axonal transport is extremely important for maintaining neuronal polarity, and thus their signaling ability.¹⁹³ It is evident that the unimpeded functioning of mitochondria is vital to neuronal integrity and survival. Mitochondrial dysfunction in NDs occurs due to a combination of several factors, the most important triggers of which are disruption in its ability to undergo fission/fusion and changes in its DNA. As discussed in Section 7.2.4, oxidative stress is found to play a role both upstream and downstream of mitochondrial deregulation.

Human mitochondrial DNA (mtDNA) is circular, double-stranded and exists in multiple copies within the cell.¹⁹⁴ As this gives rise to the possibility of coexistence of wild-type and mutant forms, the deleterious effects of mutations do not have any functional significance until they exceed a threshold, which is lower in neurons due to their sole reliance on OXPHOS for energy. This along with higher ROS exposure,

lack of histone protection, and limited DNA repair machinery make mtDNA highly vulnerable to oxidative damage leading to mutations.¹⁹⁵ These mutations could gradually impair the electron transport chain, effectively reducing ATP production and increasing ROS generation, which in turn act as a positive feedback spawning further mutations and oxidative damage.¹⁹⁵ Decline of mitochondrial function due to ROS and mtDNA mutations is prevalent in aged brains and is exacerbated in NDs due to several disease-specific susceptibility factors.¹⁹⁶

mtDNA mutations are believed to cause the increased oxidative damage seen in AD patients,¹⁹⁷ although it is not a direct causative agent of the disease.¹⁹⁸ mtDNA single-nucleotide polymorphisms and haplogroups seem to influence AD risk;¹⁹⁹ mtDNA haplogroups have been linked to APOE-4 in some cases.²⁰⁰ ROS-mediated impairment of mtDNA base excision repair could also contribute to mitochondrial damage.²⁰¹ Certain haplogroups have also been associated with the modification of mitochondrial complex I activity and increased oxidative stress-vulnerable nigral neurons of PD patients, which indicates that they could be partially responsible for the widespread mitochondrial damage seen in PD—an implication strengthened by both clinical and preclinical data.^{203,204} Although the causality of mtDNA mutations in PD has been challenged,^{205,206} its interplay with other causative factors of PD like familial gene mutations²⁰⁷ suggests that its role cannot be disregarded.

In addition to fidelity of mtDNA, mitochondrial dynamics are also crucial to maintaining cellular homeostasis. Mitochondria undergo fission (splitting) and fusion (combining) that facilitate exchange of essential components and the dilution of toxic ones, thus establishing a healthy equilibrium. Fusion is carried out by the GTPases mitofusin 1 & 2 (Mfn 1 & 2) and optic atrophy protein 1 (OAP1), and fission, by dynamin-like protein 1 (DLP1), and the outer mitochondrial membrane protein Fis1.195 The balance of fission and fusion is contingent upon changes in metabolic status, oxidative stress, and ion homeostasis and could have serious impact on mitochondrial functions involving calcium buffering, apoptosis, and ROS generation.²⁰⁸ Dysfunctional mitochondrial dynamics in AD is evident from the changes in the number and morphology of mitochondria in the hippocampal neurons in addition to lower levels of fusion proteins (Mfn 1 & 2 and OAP1) and higher levels of Fis1, suggesting an equilibrium shift toward increased mitochondrial fragmentation.^{209,210} APP and subsequent A^β overproduction have been shown to impair mitochondrial dynamics in very early stages of the disease, even before the appearance of plaques.²¹¹ Nitrosylation of DLP1 by A β -mediated NO release is thought to be one of the mechanisms for the observed effects, thus providing a cellular mechanism tying oxidative stress and mitochondrial dysfunction in Aβ-toxicity.²¹² Recently, it was found that the A β -mediated mitochondrial damage was aggravated by a truncated form of the protein tau—the aberrantly phosphorylated form of which is a major component of NFTs in AD.²¹³ Caspase-mediated truncation of tau enhances aggregate formation,²¹⁴ thus suggesting that mitochondrial deregulation probably mediates tau toxicity. In turn, mitochondrial damage may also promote tangle formation by enhancing tau phosphorylation.²¹⁵

Mitochondrial dysfunction is probably one of the most pronounced pathogenic features of PD. This is no surprise considering that the causative genes for the familial form of the disease as well as the susceptibility factors including α -synuclein, parkin, PINK-1, and LRR2 are associated with the mitochondria.¹⁹⁵ Both environmental and genetic factors implicated in the disease etiology seem to perturb mitochondrial homeostasis by inhibiting the activity of complex I. MPTP or rotenone-mediated inhibition of its activity causes pronounced oxidative stress, which further drives mitochondrial activity and ATP production down.²¹⁶ Lower ATP levels reduce the energy-intensive vesicular loading of dopamine, causing a spike in the cytoplasmic dopamine levels and an increase in oxidative stress due to its enhanced oxidation.¹⁷⁹ The by-products of dopamine oxidation—semiquinones and quinones—are capable of inhibiting complex I activity by interacting with GSH, thus establishing a bidirectional relationship between dopamine oxidation and complex I inhibition.¹⁷⁹

The proteins implicated in the familial form of the disease are generally associated with the mitochondrial membrane and influence the activity of complex I. α -synuclein associated with the inner mitochondrial membrane reduces complex I activity, triggering apoptosis,²¹⁷ whereas parkin and PINK1 have prosurvival roles probably by working in conjunction to inhibit the fatal mitochondrial cytochrome c release.¹⁹¹ Cytochrome c release is also one of the events triggered by mutant SOD1 aggregates in ALS, in addition to blocking protein importation and ROS production.¹⁹¹ Reduced levels of the antioxidant Mn-SOD and reduced function of the electron transport chain are the other features in ALS contributing to mitochondrial damage. The attendant energy deficit affects mitochondrial membrane potential, causing Ca²⁺ influx and membrane permeability eventually causing cell death.²¹⁸ This mechanism is also common to huntingtin-mediated toxicity in HD.²¹⁹ Impaired complex-II activity and disruption of fusion-fission complexes are the other mechanisms of mitochondrial dysfunction in HD.¹⁹¹ In summary, the mitochondrial dysfunction-oxidative stress cycle mediates both the downstream effects of pathological components of NDs and the prodromal effects of the genetic and environmental factors causing them.

7.2.6 PROTEIN AGGREGATION

Protein aggregates are considered to be of seminal importance in ND etiology as they provide the first visual pathological indicator for the disease. Protein inclusions/ amyloid plaques have been the stand-alone theme tying diverse NDs together (Table 7.2), until further deliberative studies unraveled other mechanisms associated with the disease. An intriguing feature of these aggregates in addition to their characteristic β -pleated structure is that they are composed of otherwise ubiquitous proteins found throughout the body, but form plaques in only in disease-specific regions in the CNS.²²⁰ It is thought that the genetic mutations in the proteins implicated in the familial forms of the disease accelerate this process of aggregation, whereas environmental and susceptibility factors might mirror the same at a lower rate, in the sporadic forms. For example, mutations in the gene encoding α -synuclein (*SNCA*) increases its propensity to form aggregates (seen in Lewy bodies) in the familial form, whereas environmental factors such as rotenone or MPTP are known

INDLL /	• 4				
Protein Aggregates in Some of the Common NDs					
Disease	Protein Component of Aggregates	Cellular Location			
AD	Αβ	Extracellular			
	Hyperphosphorylated tau	Cytoplasmic			
PD	α-synuclein	Cytoplasmic			
HD	Huntingtin (with polyglutamine tracks)	Nuclear			
ALS	SOD1 (only in the familial form)	Intracellular			

TABLE 7.2		
Protein Aggregates	in Some of the C	Common NDs
	_	

to promote α -synuclein-containing inclusions in the sporadic form.²²¹ Similarly, the preponderance of the amyloidogenic $A\beta_{42}$ fragment caused by mutations in genes encoding APP and presenilins is also promoted by the APOE-4 haplotype (APOE-4), a known susceptibility factor for sporadic AD. Although the underlying mechanism is not clear, a combination of reduced Aß clearance and cholesterol modulating effect of APOE-4 is thought to contribute to Aβ deposition.²²⁰ In some cases, the aggregates show mixed toxicity-toxicity of the aggregate itself (toxic gain of function) as well as toxicity due to loss of function of the aggregated protein. An excellent example of this is the MAPT, which is involved in axonal transport.²²⁰ Overall, the protein aggregates are thought to exert their toxicity by a complex mix of overlapping mechanisms including mitochondrial dysfunction, altered gene expression, oxidative stress, sequestration of other proteins, disruption of axonal transport, excitotoxicity (discussed in Section 7.2.7), and synaptic dysfunction.²²⁰

Although the "amyloid hypothesis" proposing that protein aggregation triggers the cascade of events culminating in neurodegeneration has been the gold standard of the ND paradigm,²²² an increasing body of evidence challenges this notion.^{17,223,224} One of the reasons for this conflict is the lack of correlation of number and the size of plaques with the severity of the disease in AD postmortem tissue;²²⁵ the other is the fact that SPs are seen in the cortical tissue of many cognitively normal 70-year-olds.²²⁶ The intensive research spurred on by these revelations now suggest that the fibrillar aggregates are an epiphenomenon of the disease and that a common trigger sets off both the neurodegeneration and protein aggregation.¹⁷ There is evidence for the presence of soluble oligomeric species of AB,²²⁷ globular or ringlike protofibrils of α -synuclein,²²⁸ and protofibrillar intermediates of huntingtin,²²⁹ all of which are toxic in their respective disease conditions.²²⁴ It is speculated that the preaggregate protofibrils might be the actual malignant entity, possibly even more toxic than the fibrillar aggregates. Hence, it is pragmatic to inhibit the formation of these neurotoxic protofibrillar intermediates to prevent neuronal death. This requires a detailed assessment of the processes in play that influence the formation of these moieties.²²⁴

Since the prefibrillar theory has not been unequivocally proven, there is no knowledge of the concrete steps responsible for their formation, or which of the several intermediate species is most toxic. However, we do have knowledge of the possible factors that could seed the process, or in other words, present avenues for its prevention. Oxidative damage, again, is one of the culprits initiating the misfolding of proteins and thought to be a primary event in the generation of protofibrils.²²⁴ Other covalent modifications such as nitration,²³⁰ SUMOylation (small ubiquitin-like modifier),²³¹ and phosphorylation²³² are also involved in the aggregation process. Under normal conditions, these misfolded proteins are handled by various housekeeping proteolytic machinery (proteasomal and autophagic) that serve to either refold them or tag them for degradation, thereby circumventing their toxic effects. Cellular stress also stimulates the assembly of a bevy of stress response proteins called molecular chaperones, which aid in combating the stress. This bastion of stress-handling pathways is broken down in NDs and contributes to the formation and/or reduced clearance of the toxic protofibrils and amyloid aggregates.

Proteins misfolded due to genetic mutations, environmental factors, or oxidative stress induce the synthesis of HSPs that chaperone them and modulate their conformation, movement across membranes, or tag them for degradation.²³³ Some of the most common HSPs include HSP70, HSP27, ubiquitin, and HSP32/heme oxygenase. Overexpression of HSPs seems to reverse the neuronal loss caused by the proteins implicated in various NDs, thus suggesting that misfolding is one of the causes of neurodegeneration.²²⁰ Induction of HSP70 reversed the neurotoxicity seen in α -synuclein and polyglutamine-toxicity models in *Drosophila*.²³⁴ The induction of heme oxygenase-1 is being pursued as a promising therapeutic strategy in combating AD neurodegeneration.²³³ Ubiquitin is another chaperone and the smallest of the HSPs involved in targeting proteins to the proteasome-the degradation hub of short-lived proteins.²³⁵ The ubiquitin-proteasome system (UPS) plays a crucial role in regulating protein turnover and controls functions as diverse as neuronal functioning, DNA repair, synaptic plasticity, and cell signaling.¹⁷⁶ When proteins get misfolded, they are tagged by a chain of ubiquitin molecules to the 26S proteasome-a multiprotein complex with proteolytic activity. The ubiquitination reaction is catalyzed by a series of activating (E1), conjugating (E2), and ligating enzymes (E3) and reversed by deubiquitinating enzymes (DUBs).¹²⁴ The dysfunction of any of the components of the UPS could conceivably result in the accumulation of damaged proteins and eventually, cell death.

Both oxidative stress and aging are known to impair the UPS by increased oxidative modifications of the proteasomal proteins as well as the ubiquitinating enzymes involved in the system.^{236,237} Hence, it is no surprise that UPS dysfunction is also a prominent phenomenon in ND etiology, wherein both aging and oxidative stress have been exhaustively implicated. Aggregates positive for ubiquitin have been found in almost all ND pathologies,²³⁸⁻²⁴¹ suggesting either that the UPS defense system was probably engaged, yet overpowered by the excess production of the toxic proteins or its efficiency was reduced or some combination of both. Decreased proteasomal function has also been found throughout the ND spectrum,²⁴²⁻²⁴⁴ thus lending further credibility to this hypothesis. The fact that some of the familial forms of PD are caused by mutations in the genes encoding parkin—an E3 ligase, and UHCL1—a DUB, underscores the importance of this system in PD etiology. Proteasomal dysfunction promotes accumulation of aggregate-prone α -synuclein, which in turn is known to depress UPS function further, thus setting up a vicious circle.^{124,242} Thus, disease-specific genetic factors or global oxidative stress could first activate and then impair the UPS, which could eventually cause toxic protein buildup in addition to other ramifications such as additional oxidative stress, mitochondrial damage, and neuronal death.^{67,176}

In addition to the UPS-mediated protein degradation, other proteolytic systems such as lysosome-mediated autophagy,²⁴⁵ matrix metalloproteinases,²⁴⁶ and calpains²⁴⁷ are also implicated in NDs. In all cases, maintaining the tenuous balance between synthesis of essential proteins and the degradation of damaged ones seems to be of utmost importance in preventing the neuronal damage caused by these protein aggregates.

7.2.7 **E**xcitotoxicity

Glutamate is the principal excitatory neurotransmitter in the nervous system that plays a role in synaptic plasticity, learning, memory acquisition, and other cognitive functions.²⁴⁸ It is involved in fast transmission of neuronal signals by binding to and activating excitatory amino acid receptors/ligand-gated ion channels, which cause receptor depolarization and neuronal excitation. While deficits in this excitatory transmission are known to cause several neuropsychiatric conditions,²⁴⁹ excessive or prolonged stimulation by glutamate could cause a condition called excitotoxicity, which eventually causes neuronal death. Excitotoxicity has been implicated in the etiology of several acute as well as chronic NDs.²⁵⁰

Glutamate receptors can be divided into two categories: ionotrophic and metabotrophic, wherein the ionotrophic receptors are subdivided into NMDA, α -amino-3hydroxy-5-methyl-4-isoxazolepropionic acid (AMPA), and kainate. AMPA receptors can be directly activated by glutamate, mediating rapid neurotransmission, whereas NMDARs require prior depolarization by AMPA/kainate receptors for activation.²⁵⁰ NMDAR activation causes release of Mg²⁺ and an influx of Ca²⁺, which activate intracellular signaling pathways modulating surface presence of AMPA receptors on the postsynaptic membrane, thus forming a regulatory loop for synaptic plasticity.²⁵¹ Usually, glutamate neurotransmission is temporary and it is rapidly reabsorbed by high-affinity pumps located on nerve terminals and astrocytes.²⁴⁹ Excitotoxicity presents a condition where the reuptake ability of these pumps is compromised due to excess extracellular glutamate, causing prolonged depolarization spiraling into a sequence of events involving disruption of calcium balance, nitric oxide synthesis, generation of ROS, and activation of cell death pathways.²⁵² Elevated intracellular levels of calcium is the secondary trigger of excitotoxic damage and probably the most profound as it marks the beginning of an irreversible sequence of events culminating in cell death.²⁵⁰ High intracellular levels of calcium could activate nucleases causing DNA fragmentation, calcium-dependent proteases like calpain causing cytoskeletal breakdown, and lipases causing membrane disruption, all of which are detrimental to cell survival.²⁵³ Hence, the ability to buffer calcium may be the determinant of which cells survive the excitotoxic insult.254

Mitochondria are both regulators of calcium and the chief downstream targets of its overload. Mitochondria can help reduce cytosolic calcium by an energy-dependent process, but could also dump toxic levels of Ca²⁺ through the Na⁺/Ca²⁺ exchanger, when metabolically impaired or physically damaged.¹⁸⁷ Impaired energetics in the mitochondria and the concurrent reduction in ATP levels hamper energy-driven processes within the cell, which is no longer able to maintain its membrane potential.

The already depolarized cell potentials reduce the threshold of glutamate required to cause excitotoxicity, further adding to the damage. The resultant surge in intracellular calcium increases ROS formation, which aggravates the mitochondrial dysfunction that initiated this cycle in the first place.²⁵⁰ Thus, agents that cause mitochondrial dysfunction by interfering with the electron transport chain or other metabolic processes could theoretically render neurons more vulnerable to even mild excitotoxic insults. This type of slow excitotoxicity is seen in both HD and PD, both shown to have profound impairment in mitochondrial energy metabolism.²⁵⁵ It is postulated that the loss of nigral cells in PD could have an excitotoxic component,²⁵⁶ as the damage induced by MPTP-a known mitochondrial toxin-involves glutamate release, and is reversed upon administration of NMDAR antagonists.²⁵⁷ Also, parkin is known to negatively regulate the number of glutamatergic synapses; mutated parkin implicated in PD causes an increase in the number of synapses, thereby increasing vulnerability to excitotoxic damage.²⁵⁸ In HD, there is marked decrease in mitochondrial complex II activity, which causes long-term potentiation of the NMDA-mediated synaptic excitation in the striatum.²⁵⁹ Also, mutant huntingtin aggregates directly disrupt mitochondria, reducing membrane potential and the threshold of calcium required to open the mitochondrial permeability transition pore, whose prolonged opening causes the release of apoptotic factors.¹⁸⁷

Oxidative stress is also closely tied to excitotoxicity and is often a downstream effect of increased intracellular Ca²⁺. Free radicals are produced by the action of calcium-dependent enzymes such as phospholipase A_2 (PLA₂), nitric oxide synthase, and xanthine oxidase as well as by dysfunctional mitochondria that result from such high Ca2+ levels.250 In addition to the deleterious effects of free radicals on membrane lipids, essential proteins, and DNA (discussed in Section 7.2.4), they could also cause inhibition of glutamate transporters that reuptake glutamate from the synapses, causing prolonged stimulation of postsynaptic receptors and increased calcium influx, triggering yet another feed-forward cycle resulting in cell death.¹⁸⁹ Inhibition of glutamate transporter (EAAT2) by oxidative modification is one of the mechanisms by which mutant SOD1 causes motor neuron death.²⁶⁰ In fact, this uptake deficit was found specifically in ALS motor neurons, and not in other NDs, suggesting that excitotoxicity could play a significant role in ALS pathogenesis.²⁶¹ Experiments investigating the influence of excitotoxicity on AD etiology revealed that β-amyloid sensitizes neurons to excitotoxic cell death by deregulating calcium homeostasis in cultured neurons.²⁶² Also, the oxidative stress induced by Aβ oligomers involves an NMDAR mechanism,²⁶³ thus highlighting the potential of excitotoxicity to amplify the neurodegenerative process seen in AD. Thus, although excitotoxicity is generally thought to play a secondary role in the ND etiology, its mechanisms dovetail with the cardinal features seen in most NDs: oxidative stress and mitochondrial dysfunction, causing neuronal death.

7.2.8 ND ETIOLOGY: AT THE CROSSROADS OF PREVENTION AND TREATMENT

The etiological spectrum of NDs forms an intermeshing network of mechanisms such as oxidative stress, mitochondrial dysfunction, accumulation of protein aggregates, and excitotoxicity (Figure 7.2). In addition to these, inflammatory



FIGURE 7.2 The molecular etiology of neurodegenerative diseases in a capsule. Genetic and environmental factors combined with accelerated aging are thought to cause NDs. Crosstalk between oxidative stress pathways and mitochondrial dysfunction seems to drive the process. Impaired heat-shock proteins (HSPs), antioxidants (anti-ox), and proteasomal (UPS) degradation of misfolded proteins cause them to aggregate and form plaques, disrupting normal cellular function. Also, reduced ATP production by damaged mitochondria coupled with reduced levels of protective and trophic factors (anti-ox, brain derived neurotrophic factor, [BDNF]) causes excessive membrane depolarization and excess activation by glutamate. This causes a heavy influx of calcium leading to excitotoxicity that eventually leads to synaptic dysfunction, release of apoptotic factors, and neuronal death.

responses have also been noted in the later stages of pathogenesis, especially in AD.²⁶⁴ Genetic and environmental factors aggravate the effects of the aging nervous system by accelerating the molecular events that cause neuronal dysfunction. Save for purely Mendelian NDs, it is seen that, often what start off as seemingly insidious triggers accrue over a period of many years, gradually weakening the

defense and stress response mechanisms of the body causing neuronal dysfunction and death, thus accounting for the characteristic progressive nature of deficits seen in most NDs. Neuronal death is thought to mainly occur by apoptosis, with oxidative stress, glutamate overactivation, genetic mutations, and decreased levels of neurotrophic factors serving as death signals. The apoptosis is mediated by opening of mitochondrial pores and release of proapoptotic cytochrome c, which triggers the caspase cascade culminating in cell phagocytosis and nuclear fragmentation.²⁶⁵

One of the prime goals of preventive therapy in NDs is increasing the healthy interim period before debilitation sets in. This involves identifying mechanisms that deteriorate with age—those that could be fortified to afford the neurons increased ability to fight degeneration. Focus on the early events of ND etiology has delineated key mechanisms that could be preventively averted: oxidative stress, mitochondrial impairment, and impaired stress response. Attenuation of these deficits could involve boosting the adaptive response of neurons by increased synthesis of neurotrophic factors, HSPs, and other prosurvival factors as well as by improving energy metabolism.¹⁵⁵ Diet and exercise are thought to bolster this adaptive response by various mechanisms, thus considerably reducing risk of ND. An understanding of the mechanistic underpinnings of this salutary role is essential for designing better drugs and alternative therapies aimed at preventing the disease.

7.3 PREVENTIVE MECHANISMS MEDIATED BY DIET

Two representative diet paradigms are considered here, as they cover most of the neuroprotective effects of diet vis-à-vis ND prevention.

7.3.1 DIETARY RESTRICTION: LESS IS GOOD FOR THE BRAIN

We are constantly barraged with facts about how overeating is the root cause of all maladies, creating an almost mass paranoia about everything we eat. The situation may actually not be as dire as it seems, but the principle behind the health benefits of low calorie intake is unshakeable. DR—a 10%–40% reduction of calories with maintenance of macronutrient intake—is widely reported to increase longevity and "healthspan" in laboratory animals.²⁶⁶ Recently, it showed similar beneficial effects on the metabolic health and overall function in primates, although its effects on longevity are ambiguous.²⁶⁷ Nevertheless, the extrapolation of the beneficial effects of DR to living species beyond the lab spells good news for its potential in humans.

In addition to its effects on survival and overall well-being, DR also slows down age-related changes in the nervous system. It reduces age-inflicted oxidative damage as well as cognitive and motor deficits in mice.^{268,269} Since accelerated aging is one of the chief risk factors for neurodegeneration, DR mediated resistance to age-related deficits could potentially delay onset or progression of NDs. Encouragingly so, DR has been found to reverse damage to hippocampal neurons and increase resistance to lipid peroxidation and excitotoxicity in animals models of AD.²⁷⁰ It also reduced dopaminergic and striatal neuronal loss in animal models of PD²⁷¹ and HD,²⁷² respectively and improved motor function in both. This along with strong epidemiological evidence supporting the benefits of reduced calorie intake in reducing ND risk^{55,75} warrant further investigation.

7.3.1.1 DR Mechanisms: Neurotrophic Factors

Neurotrophic factors, along with neurotransmitters and hormones, belong to the global class of signaling proteins responsible for neuronal survival and synaptic plasticity. The most notable of them-BDNF-is induced by DR. BDNF plays a crucial role in synaptogenesis, nerve outgrowth, and ultimately neuronal survival.²⁷³ Its reduction has also been implicated in the age-related changes leading to memory decline and synaptic dysfunction.²⁷⁴ BDNF has been known to promote resistance to oxidative damage, metabolic deficits, and excitotoxicity inherent in ND pathogenesis.²⁷³ Its neuroprotective effects stem from its ability to induce the expression of genes encoding for proteins that combat stress: antioxidant enzymes²⁷⁵ (Cu/Zn-SOD, Mn-SOD, GSHPx, and catalase) and antiapoptotic proteins²⁷⁶ (Bcl-2 family members). It also stabilizes calcium homeostasis by interacting with glutamate receptors, thereby promoting synaptic stability, as well as by inducing the expression of genes coding for calcium-dependent proteins and the glutamate receptor subunits.²⁷⁷ These prosurvival actions of BDNF are mediated by two sets of transducers within the cell: the kinases (mitogen-activated protein kinase, protein kinase C, tyrosine kinase [TrkB], Ca^{2+/} calmodulin-dependent kinase), which are the primary transducers and transcription factors (nuclear factor kappa-B and cAMP response element-binding protein [CREB]), which eventually modulate the expression of genes coding for antioxidant enzymes, HSPs, antiapoptotic proteins, and more neurotrophic factors.¹⁸²

The ability of neural progenitor cells to differentiate into neurons and glial cells is crucial to replace cells damaged due to aging or disease. DR is also known to promote this process, known as neurogenesis. BDNF is thought to play a role in enhancing the survival of such newly divided cells, thus linking DR and BDNF to maintenance or perhaps enhancement of cognitive function during aging and diseased conditions.²⁷³

7.3.1.2 DR Mechanisms: Heat-Shock Response and Protein Chaperones

Heat-shock proteins, as their name suggests, form the first line of defense against high temperatures usually associated with inflammation or infection. Their roles as protein chaperones and their involvement in ND etiology has been previously discussed (Section 7.2.6). DR increases the levels of HSP70 and GRP78 in cortical, hippocampal, and striatal neurons.²⁷³ These molecular chaperones can protect neurons against oxidative and excitotoxic injury²⁷⁸ and participate in protein remodeling and disposal pathways.²⁷⁹ It is presumed that DR activates the synthesis of these chaperones through the BDNF-mediated induction of transcription factors as mentioned earlier (Section 7.3.1.1).

DR was also found to offer increased resistance to metabolic and oxidative distress in the synapses, as indicated by maintenance of intact glutamate and glucose transport as well as mitochondrial function.²⁸⁰ It is also thought that the protective proteins induced by DR concentrate in the synaptic compartments.²⁷⁷ These findings are important because synapses are pivotal to signal transduction, yet extremely vulnerable to damage due to repetitive calcium influx and oxyradical production.²⁷³ In a way, they are the "Achilles' heel" of neuronal circuitry with their dysfunction being regarded as a primary marker of aging and neurodegeneration, even before neuronal death. DR thus augments the ability of synapses to cope with the stress prevalent in NDs.

7.3.1.3 DR Mechanisms: Epigenetics

Aging is generally associated with globally decreased, yet locally increased DNA methylation, leading to the silencing of many genes essential for survival. DR is known to reverse these DNA methylation changes and increase genomic stability.⁵⁰ DR also plays a role in histone modification, which could cause either gene activation or repression, depending on the type of modification. Histone deacetylation is associated with transcriptional repression as it leads to a more condensed chromatin structure. The enzymes that catalyze the deacetylation of histones are called histone deacetylases (HDACs); their actions are critical to understand how DR affects neuronal survival at the transcriptional level.

HDAC activity is increased during DR, suggesting that global deacetylation may be protective against age-related stress.²⁸¹ Of particular importance among the HDACs is the Class III NAD⁺-dependent enzyme sirtuin-1 (SIRT1), which is a prime mediator of the health-promoting effects of DR.^{282,283} SIRT1 is highly sensitive to the metabolic status of the cell and improves stress resistance of the cell by deacetylating and thus negatively regulating the activities of transcription factors and regulatory proteins.⁵⁰ For example, it deacetylates and represses the O subclass of forkhead family of transcription factors (FOXO), thereby inhibiting FOXO-mediated apoptosis.²⁸⁴ Similarly, its effect on the DNA repair protein Ku70 also prevents stress-induced cell death by sequestering proapoptotic proteins away from the mitochondria.²⁸² DR-mediated SIRT1 induction could promote cell survival and retard degeneration in many NDs.²⁸⁵

7.3.1.4 DR Mimetics: A Shortcut?

Reducing calorie intake is a draconian regimen and likely to have poor adherence by most of the population. Not surprisingly, this has sparked intense research toward finding more convenient alternatives that provide the benefits of DR by circumventing the arduous path of calorie-cutback. DR mimetics such as resveratrol (RES)⁵⁰ and 2-deoxy-D-glucose²⁷⁷ have demonstrated DR-like effects in various pathological conditions. RES, a polyphenol found in grapes, red wine, and berries, has potent neuroprotective effects in many models of NDs.²⁸⁵ RES-mediated activation of SIRT1, and the consequent suppression of FOXO and p53-mediated apoptosis, is thought to be one of its chief neuroprotective mechanisms.²⁸⁶ RES also has potent antioxidant effects, although its low bioavailability makes a direct radical scavenging effect less likely.⁵⁰ Majority of its antioxidant effects probably stem from its ability to induce antioxidant enzymes and maintain mitochondrial function.²⁸⁷ In addition, RES also mirrors DR-mediated HSP induction and improved protein handling through a mechanism not related to its antioxidant effects, thus highlighting the versatility of its actions.²⁸⁸

DR has met with positive results in all ND models except ALS so far.⁹⁹ This could perhaps mean that the negative effects of reduced energy intake nullify/outweigh the prosurvival effects of DR in already undernourished ALS patients or that DR

does not equally benefit all populations of neurons.²⁶⁶ In fact, ALS patients seem to benefit from a high-energy ketogenic diet.¹⁰² A ketogenic diet might also delay the weight loss seen in the later stages of HD.²⁸⁹ Thus, DR has overarching effects on all levels of neuronal circuitry including synapses (glutamate and calcium homeostasis), postsynaptic membranes (BDNF and glutamate), intracellular machinery (kinases, transcription factors, and HSPs), the genome (epigenetic modifications), and finally at the level of neuronal differentiation itself (neurogenesis). These mechanisms pose a stronghold against the cellular stress rampant in most NDs.

7.3.2 MEDITERRANEAN DIET: WHAT YOU EAT IS AS IMPORTANT AS HOW MUCH

The Mediterranean diet (MeDi) is heralded as the model for good eating, due to not only its health-promoting effects but also its high palatability.²⁹⁰ It is characterized by a higher proportion of plant-derived foods (fruits, vegetables, nuts, legumes, and cereals), fish, olive oil, low to moderate intake of wine, and a low intake of red meat and poultry.²⁹¹ High adherence to MeDi has been associated with reduced risks of mild cognitive impairment, AD, and PD.^{77,292} Although the MeDi combines nutrients that have individually been associated with reduced ND risks, its study as a diet is more pragmatic as people consume food as a whole rather than a mix of disparate nutrients.²⁹² To date, no synergistic relationship among the MeDi components has been established; however, they seem to impinge on similar molecular pathways en route to their beneficial effects.

Vegetables, especially the green leafy and cruciferous ones (broccoli, cabbage, brussels sprouts, and kale) have been associated with a slower decline of cognitive function with age.^{293,294} The MeDi owes most of its neuroprotective properties to a chemical class of compounds called flavonoids found in many fruits, vegetables, cereals, wine, and tea. They are a subclass of naturally occurring polyphenolic compounds historically known for their potent antioxidant effects. However, just like RES, their actions are not restricted to their free-radical scavenging activity; rather, they modulate diverse signaling pathways influencing neuronal survival, memory, and cognition.²⁹⁵ For a detailed description of the types of flavonoids and the foods that contain them, see Chapter 2, Section 2.5.2.2.

The polyphenols found in fruits and vegetables are mainly of the anthocyanin type and abundant in bilberries, black raspberries, and chokeberries (concentration ranging from 600 to 1500 mg/100 g of fresh weight).²⁹⁶ Polyphenols mediate some of their antioxidant effects by inhibiting the activity of the calcium-dependent enzyme PLA₂, which is involved in lipid peroxidation and mitochondrial deregulation.²⁹⁷ They are also excellent metal chelators (e.g., green tea catechins), which are particularly beneficial in AD pathogenesis, wherein transition metals promote A β deposition, and also in other neurodegenerative conditions, where metals could lead to free-radical production.²⁹⁸ Also, some grape-derived polyphenols may potentially inhibit A β oligomerization, which is one of the primary goals of AD prevention.²⁹⁹

Another fascinating aspect of the repertoire of beneficial effects of flavonoids is their ability to trigger cascades that oversee all aspects of neuronal function including synaptic plasticity, memory, and neurogenesis.²⁹² It is postulated that they

do so by either increasing the expression of BDNF, thereby initiating the cascade of events similar to those activated by DR, or by directly activating the CREB pathway.³⁰⁰ Flavonoids presumably bind to the TrkB receptor, which activates CREB-mediated transcription of genes essential for synaptic remodeling.³⁰¹ This also activates the PI3 kinase/Akt pathway, which inhibits proapoptotic proteins and promotes neurogenesis.²⁹² These actions of flavonoids are mildly reminiscent of the prosurvival mechanisms activated by RES, which incidentally is a nonflavonoid polyphenol and another major component of the MeDi found in red wine, grapes, and berries. Since there is considerable overlap between the rescue mechanisms mediated by polyphenols and DR, such polyphenols could probably make excellent candidates as DR mimetics.

Unsaturated fatty acids (UFA) form the next significant component of the MeDi and are found mainly in olive oil, nuts, seeds (MUFA), and fish (PUFA). Olive oil is also rich in antioxidant principles and tocopherol and polyphenols, which probably contribute to the beneficial effects of MUFA. Fish (especially fatty fish) are rich in n-3 PUFA (e.g., docosahexaenoic acid [DHA] and eicosapentaenoic acid [EPA]), which are thought to have neuroprotective effects in NDs.³⁰² DHA, the representative n-3 PUFA, is associated with general memory and learning³⁰³ and with reduced Aß & tau accumulation³⁰⁴ and reduced apoptotic activity in AD models.³⁰⁵ Of particular importance in the effectiveness of PUFA is the balance between n-3 and n-6 (linoleic acid is the most common n-6 PUFA), as n-6 PUFA is suspected to have atherogenic effects. Hence a high n-3/n-6 ratio (found in fish oil) provides a better risk/benefit profile in terms of neuronal heath.⁶⁸ The neuroprotective effects of UFA mainly originate in the membrane and the processes associated with it. They help maintain the fluidity and the integrity of the membrane, thereby controlling synaptic function and neuronal transmission. They may also modify the activities of several membrane-bound proteins (PLA₂, protein kinase C, and acetyltransferase), neurotransmitter receptors, and ion channels.²⁹² They could thus conceivably work in conjunction with other factors like BDNF and flavonoids that activate processes beginning at the membrane level.

MeDi is also an excellent source of vitamins that have diverse neurophysiological functions, and whose augmentation could have several beneficial effects. Vitamin E found in nuts, seeds, and green leafy vegetables and is a potent antioxidant with possible effects on gene expression. The verdict from clinical studies of vitamin E as a preventive agent in NDs (AD, PD, and ALS) although not conclusive is promising.³⁰⁶ Vitamin D₃ found in fish (salmon, mackerel, and blue fish) is associated with enhancement of cognitive functions, mediated by its antioxidant, anti-inflammatory, and neutrophin-inducing effects.²⁹² Folic acid or folate found in green vegetables, citrus fruits, and whole grains is an essential cofactor in enzymatic processes involved in cell differentiation, proliferation, and survival.²⁷³ Folic acid deficiency is associated with reduced S-adenosylmethionine and increased homocysteine levels; the former is responsible for DNA methylation and could modulate epigenetic pathways, whereas the latter is known to cause oxidative DNA damage and impaired DNA repair.¹⁴⁷

The MeDi diet thus offers a potpourri of nutrients in a palatable format that individually or through composite mechanisms promotes neuronal defenses and survival. In addition to the two paradigms discussed in the preceding discussion, a lot of dietary supplements consisting of neuroprotective herbs and antioxidants are proposed to prevent the progression of NDs. They include herbs such as *Panax ginseng*, *Ginkgo biloba*, *Withania somnifera*, *Curcuma longa* and antioxidants such as N-acetylcysteine, Coenzyme Q₁₀, creatine, GSH, and lycopene.^{273,307}

7.4 PREVENTIVE MECHANISMS MEDIATED BY EXERCISE

Physical inactivity is positively correlated with the risk of many age-related chronic illnesses including NDs. Studies in humans and rodents have unequivocally established the beneficial effects of regular exercise on cognitive functions.^{308–310} It can improve nuanced tasks such as planning, decision making, as well as reverse cognitive dysfunction in both aged and diseased brains.^{310,311} Regular aerobic exercise combined with muscle strengthening is expected to act as a disease-delaying strategy by promoting neuronal activity and brain plasticity.

7.4.1 EXERCISE MECHANISMS: BDNF IS THE KEY PLAYER

One of the notable molecular consequences of exercise is the increase in the levels of BDNF—the neurotrophic factor also responsible for most of the prosurvival effects of DR. Exercise promotes the expression of BDNF in the hippocampus, thus providing a molecular base for its actions on learning and memory.³¹² Interestingly, inhibiting BDNF also abrogates the exercise-mediated beneficial effects on synaptic plasticity, suggesting that BDNF is essential for exercise-mediated neuroprotection.³¹³ BDNF activates downstream pathways similar to those discussed under DR mechanisms involving kinases such as mitogen-activated protein kinase (MAPK) and calcium/ calmodulin-dependent protein kinase (CAMK), which converge on CREB-mediated gene transcription of essential proteins and maintenance of Ca^{2+} homeostasis and glutamate activity, which eventually lead to efficacious signal transduction at the synapses and neuronal survival. Exercise also promotes hippocampal neurogenesis with BDNF suggested to support survival of the newly formed cells.³¹⁴ There is a growing body of evidence to show that BDNF acts as a "metabotrophin," a portmanteau coined to suggest that its effects derive from its crosstalk with processes involved in both energy metabolism and synaptic function.³¹⁵

7.4.2 EXERCISE MECHANISMS: ENERGY METABOLISM-Synaptic Plasticity Interface

Energy balance has a marked influence on neuronal and cognitive plasticity.³¹⁶ As we now know, defective energy metabolism has multifaceted effects on cellular mechanisms that affect neuronal transmission such as Ca²⁺ homeostasis, glutamate activity, and stress response, which are crucial components of ND etiology. Any entity that perturbs the energy status of the cell (like physical activity) could thus theoretically affect neuronal function. It is postulated that exercise exerts its protective effects on neuronal function by a combined action on processes involved in both energy metabolism and synaptic plasticity.⁵¹ One of the reasons for this assumption

is that its actions primarily stem from BDNF upregulation, which as previously discussed modulates both these systems. Further evidence that exercise upregulates proteins, which are both components of metabolic and synaptic pathways and are also implicated in cognitive function, support the hypothesis that it functions at energy metabolism-cognition interface.³¹⁷ Voluntary exercise decreases oxidative stress and improves energy homeostasis in the mitochondria, possibly by increasing the levels of mitochondrial uncoupling protein (UCP2)²⁹³—a member of the family of mitochondrial inner membrane proteins that allow protons to leak across membranes, thereby reducing OXPHOS and the production of superoxides.¹⁸² They can thus protect neurons against oxidative damage and maintain mitochondrial function under conditions of duress. The oxidative stress-alleviating effects of exercise also contribute to the actions of BDNF and CREB-which in turn activate BDNF expression. Moreover, negative manipulation of mitochondrial metabolism also inhibits the exercise-mediated elevation of BDNF expression and its downstream events, thus confirming the role of exercise in modulating these interdependent systems.

In addition to neural cues, visceral signals of the energy status of the cell are also picked up by the CNS to modulate neuronal functions.⁵¹ Insulin-like growth factor-1 (IGF-1) plays a role in regulating lipid metabolism and insulin action.¹⁰² The IGF-1 receptor is expressed in the hippocampus and known to be involved in synaptic and cognitive function.³¹⁸ Exercise stimulates the uptake of IGF-1 into the brain, wherein it modulates pathways downstream to BDNF activation and improves synaptic plasticity.^{51,319}

7.4.3 EXERCISE MECHANISMS: EPIGENETICS

Exercise can control the expression of BDNF by modifying the epigenome, thus providing another example of how the environment can produce long-lasting changes that eventually affect our neuronal circuitry.52 Exercise promotes chromatin remodeling of the *Bdnf* gene by modulating histone deacetylation and methylation, which in turn cause stable elevations in gene expression, thus providing for a means to regulate synaptic function at the genomic level. Along with enhancing cognitive function, both exercise and BDNF have been linked to reduction of depression-an interesting prospect considering that depression is a prevalent comorbidity in most NDs.^{320,321} These epigenetic modifications provide a channel to transfer the salutary effects of exercise across generations. With an ever-growing trove of encouraging results emerging from studies on diet and exercise, it is only natural to investigate whether they could synergize to produce better results. A combination of exercise and DHA supplementation enhances the BDNF-mediated synaptic regulation as well as learning ability.54 Flavonoids and exercise also synergize to promote cell survival and neuronal plasticity.³²² Additionally, exercise has also been found to reverse the noxious effects of unhealthy diets.³²³ Thus, in addition to interaction among themselves, components of a diet also interact with other lifestyle interventions like exercise to amplify the health-promoting influence of one another. This presents an interesting modality that when subjected to standardization and well-designed trials could open up new avenues for the preventive care of NDs.

7.5 SUMMARY AND RECOMMENDATIONS

7.5.1 SUMMARY: ETIOLOGY AND THE PREVENTIVE ROLE OF DIET AND EXERCISE

An intricate combination of age, environmental factors, and genetic factors decides whether our nervous system degenerates or not. Environmental factors such as heavy metal exposure or poor diet could potentiate the effects of aging by contributing to the oxidative stress, metabolic dysfunction, and weakened stress response mechanisms. Epigenetic modifications triggered by environmental influences beginning early in life could also have long-term repercussions in terms of ND risk. Finally, disease-specific genes and susceptibility factors are what determine the degeneration of specific anatomical regions within the nervous system, the latter playing a dominant role in the sporadic forms of the disease. Oxidative stress seems to be the constant factor in the complex etiological process. It generates ROS that detrimentally modify proteins, lipids, and DNA. Oxidative modification of proteins by ROS as well as lipid peroxidation products causes them to lose their activity or enhances their tendency to aggregate. This is further enhanced by an age-related decline of protein chaperones and the UPS—two systems responsible for the processing of damaged proteins. Aggregated proteins or amyloid plaques add to the oxidative insult and impair signal transduction, setting the cells up for a programmed death cascade. Oxidative stress also damages mitochondria resulting in metabolic deregulation, Ca²⁺ dysfunction, and further oxidative stress. This self-propagating cycle causes the mitochondrial release of proapoptotic factors and subsequent cell death. Energetically impaired mitochondria also render neurons vulnerable to excessive activation by glutamate culminating in excitotoxicity—a condition characterized by synaptic dysfunction, excess calcium influx, further mitochondrial deterioration, and apoptotic cell death. It is therefore clear that the molecular cascade of events surrounding neuronal degeneration is far from linear, and more often than not, involves cross-communication among various processes. However, there are some key features that stand out, which when targeted offer considerable promise for preventive intervention in NDs. They are oxidative stress, energy dysfunction, and stress response.

The environment has a tremendous influence on our brain functions. Channeling this impact to our benefit is a challenge; modifying lifestyle parameters like what we eat and whether we run may play a significant role in achieving that. Proper diet and regular exercise have favorable effects on the nervous system. They positively influence learning, memory, and cognition as well as motor functions. Eating a low-calorie diet rich in fruits, vegetables, nuts, fish, and olive oil with moderate consumption of wine seems to protect from neurodegeneration. Regular aerobic exercises and muscle strengthening activities also stimulate processes in the body that could ward-off neuronal damage. Both diet and exercise are able to exert their effects by bolstering the expression of neurotrophic factors and stress response proteins, which maintain energy homeostasis and safeguard neuronal function (Figure 7.3). They seem to augment antioxidant and protein chaperone response and initiate favorable epigenetic changes, thus acting at an incipient stage of the disease course.¹⁵⁵ They also aid in the genesis of neuronal cells and help maintain synaptic plasticity. It is speculated that restricting calories and subjecting the body to physical activity provide a kind of preconditioning effect by delivering small bouts of stress, so that the



Neuroprotection, synaptic plasticity, neurogenesis

FIGURE 7.3 Diet and exercise stimulate adaptive stress response to provide resistance against neurodegenerative diseases. Diet and exercise stimulate the release of brain-derived neurotrophic factor (BDNF), heat-shock proteins (HSPs), and antioxidants that have multifaceted actions boosting the body's defense mechanisms against stress, thereby promoting synaptic and neuronal plasticity as well as neuronal differentiation.

body could boost its adaptive response for a future, more severe insult. This effect is called hormesis¹⁸² and probably explains the actions of many dietary constituents whose effective body concentrations are too low to have a direct consequence.

Diet and exercise could potentially intervene early, which is ideal for the almost traceless progression of these diseases. ND diagnosis is usually based on the manifestation of clinical symptoms, which is often a point of no return as far as the disease course is concerned. Diet and exercise have multimodal neuroprotective effects, which are desirable in preventing or delaying the progression of diseases with such complex etiology. Thus, maintaining a low-calorie diet with sufficient "brain-healthy" foods and thorough exercise regimen (see recommendations in Section 7.5.2) may well be the elixir to keeping these malicious, yet fascinating diseases at bay.

7.5.2 DIETARY RECOMMENDATIONS

Oxidative stress being a global factor in the molecular etiology of NDs as well as inflammatory diseases, the recommendations suggested in Chapter 2 on Inflammation (Section 2.7.3) mostly apply to ND prevention. The crux of the dietary

protocol rests on reduction of calories as well as consumption of brain-healthy foods. It relies heavily on plant-derived foods and fish to meet the body's need for energy and macronutrients. It is inspired from the elements of the MeDi, which consists of foods rich in polyphenols, especially flavonoids. Consumption of one serving of low-fat dairy per day is recommended (approximately 120 kcals), to maintain calcium intake. The caloric content of the total recommendations including one serving of dairy is approximately 1500–1700 kcals, with fat constituting a relatively lower share of the total caloric intake. An exception to this caloric restriction is for ALS patients; they seem to benefit from a high-calorie, high-fat diet, which is low in carbohydrates^{101,102} (fat:carbohydrate ratio—approximately 4:1). Nevertheless, to maintain a healthy eating pattern, it is recommended they follow a similar diet plan but with modifications (ALS modifications are listed alongside in parentheses).

Finally, since disease prevention is a long-term goal, it is impractical to expect an austere adherence to the recommended diet. Indulging in a treat once in a while is okay, as long as there is not too much digression from the optimum calorie intake and adequate intake of healthy foods.

Fruits—Minimum of 2.5 cups each day with at least three different selections from each category each week:

- A: elderberries, blueberries, pomegranate, blackberries, raspberries, Saskatoon berries, blackcurrant, raisins, figs, or prunes
- B: plums, oranges, grapefruit, lemons, cantaloupe, cherries, cranberries, red/ black grapes, or apples
- C: kiwi, bananas, pineapples, mangos, pears, star fruit, guava, strawberries, peaches, or any other fruit

Vegetables—Minimum of three one-cup servings each day with at least three different one-cup selections from each category each week:

- A: chili peppers, broccoli, water cress, garden cress, spinach, arugula, and Brussels sprouts
- B: cauliflower, tomato, lettuce, green beans, sweet peppers, and red cabbage
- C: onion, garlic, celery, green cabbage, asparagus, and Swiss chard
- D: squash, zucchini, pumpkin, sweet potatoes, carrots, and potatoes

Nuts and Seeds—Minimum of one one-quarter cup serving each day with a minimum of three different selections from each category each week:

A: chestnuts, pecans, walnuts, and pistachios

B: hazelnut, almonds, Brazil nuts, cashews, macadamia nuts, or peanuts

Beans and Peas—Minimum of one one-cup serving each day with a minimum of two different one-cup selections from each category each week:

A: black beans, Adzuki beans, kidney beans, peas/pea pods, and lima beans B: lentils, pinto beans, chili beans, chick peas, and navy beans Whole Grains and Cereals—Minimum five 1-oz servings of whole grains each day with a minimum of one selection from each category each week: (—one to two servings per day in the case of ALS patients)

A: buckwheat, wheat, and bulgur B: oats, corn, and barley C: wild rice and rice

Oils—Used for cooking, salads, and dipping; approximately 1 tbsp/day (2 tbsp in case of ALS patients)

A: Extra virgin olive oil

Meats—Minimum of two servings of high EPA/DHA fish each week (four servings in case of ALS patients)

Beverages—In addition to water, tea, and coffee, consumption of red wine (one 4–5 oz glass for women and two for men per day) is recommended for its high phenolic content.

7.5.3 Exercise Recommendations

In addition to the requirements for aerobic exercises recommended in Chapter 2, additional muscle strengthening, flexibility, and balance training exercises as suggested by the American College of Sports Medicine/American Heart Association nexus are also recommended. The muscle strengthening exercises consist of 8–10 exercises (10–15 repetitions for each exercise) on two or more nonconsecutive days per week, involving the major muscle groups. Weight-bearing calisthenics and other resistance training exercises (0–10 exercises), once again focusing on major muscle groups (3–4 repetitions and holding each repetition for 10–30 seconds), for a minimum of 2–3/week, as well as balance training exercises for 3 days/week are also recommended.

ACKNOWLEDGMENT

This work was supported by the grant NIH-National Complementary and Alternative Medicine R00-AT004197 to Zahoor A. Shah

REFERENCES

- 1. Lopez AD, Murray CC. The global burden of disease, 1990–2020. *Nature Medicine*. 1998;4:1241–1243.
- Payne G, Laporte A, Deber R, Coyte PC. Counting backward to health care's future: Using time-to-death modeling to identify changes in end-of-life morbidity and the impact of aging on health care expenditures. *The Milbank Quarterly*. 2007;85:213–257.
- 3. Yazdanyar A, Newman AB. The burden of cardiovascular disease in the elderly: Morbidity, mortality, and costs. *Clinics in Geriatric Medicine*. 2009;25:563–577, vii.

- Mesterton J, Wimo A, By A, Langworth S, Winblad B, Jonsson L. Cross sectional observational study on the societal costs of Alzheimer's disease. *Current Alzheimer Research*. 2010;7:358–367.
- 5. Chandra V, Pandav R, Laxminarayan R, Tanner C, Manyam B, Rajkumar S, Silberberg D, et al. Neurological disorders. In: Jamison DT, Breman JG, Measham AR, Alleyne G, Claeson M, Evans DB, Jha P, Mills A, Musgrove P, eds. *Disease Control Priorities in Developing Countries*. Washington, DC: Oxford University Press/The World Bank; 2006.
- Mathers CD, Lopez AD, Murray CJL. The burden of disease and mortality by condition: Data, methods, and results for 2001. In: Lopez AD, Mathers CD, Ezzati M, Jamison DT, Murray CJL, eds. *Global Burden of Disease and Risk Factors*. Washington, DC: Oxford University Press/The World Bank; 2006.
- 7. Finkel T. Radical medicine: Treating ageing to cure disease. *Nature Reviews. Molecular Cell Biology*. 2005;6:971–976.
- 8. Kovacs GG, Budka H. Current concepts of neuropathological diagnostics in practice: Neurodegenerative diseases. *Clinical Neuropathology*. 2010;29:271–288.
- Armstrong RA. On the 'classification' of neurodegenerative disorders: Discrete entities, overlap or continuum? *Folia Neuropathologica/Association of Polish Neuropathologists* and Medical Research Centre, Polish Academy of Sciences. 2012;50:201–208.
- Armstrong RA, Lantos PL, Cairns NJ. Overlap between neurodegenerative disorders. Neuropathology: Official Journal of the Japanese Society of Neuropathology. 2005;25:111–124.
- Alzheimer A, Stelzmann RA, Schnitzlein HN, Murtagh FR. An English translation of Alzheimer's 1907 paper, "Uber eine eigenartige Erkankung der Hirnrinde". *Clinical Anatomy*. 1995;8:429–431.
- 12. Alzheimer's A, Thies W, Bleiler L. Alzheimer's disease facts and figures. *Alzheimer's & Dementia: The Journal of the Alzheimer's Association*. 2011;7:208–244.
- 13. Alzheimer's A. Alzheimer's disease facts and figures. *Alzheimer's & Dementia: The Journal of the Alzheimer's Association*. 2012;8:131–168.
- 14. Monin JK, Schulz R, Kershaw TS. Caregiving spouses' attachment orientations and the physical and psychological health of individuals with Alzheimer's disease. *Aging & Mental Health*. 2013;17:508–516.
- 15. Scott JP, Roberto KA, Hutton JT. Families of Alzheimer's victims. Family support to the caregivers. *Journal of the American Geriatrics Society*. 1986;34:348–354.
- 16. Vitaliano PP, Zhang J, Scanlan JM. Is caregiving hazardous to one's physical health? A meta-analysis. *Psychological Bulletin*. 2003;129:946–972.
- 17. Lansbury PT, Lashuel HA. A century-old debate on protein aggregation and neurodegeneration enters the clinic. *Nature*. 2006;443:774–779.
- 18. Cummings JL. Alzheimer's disease. *The New England Journal of Medicine*. 2004;351:56–67.
- Forstl H, Kurz A. Clinical features of Alzheimer's disease. European Archives of Psychiatry and Clinical Neuroscience. 1999;249:288–290.
- 20. Parkinson J. An essay on the shaking palsy. 1817. *The Journal of Neuropsychiatry and Clinical Neurosciences*. 2002;14:223–236; discussion 222.
- Pankratz ND, Wojcieszek J, Foroud T. Parkinson disease overview. In: Pagon RA, Bird TD, Dolan CR, Stephens K, Adam MP, eds. *Genereviews*. Seattle, WA: University of Washington; 1993.
- 22. Quinn L, Busse M, Dal Bello-Haas V. Management of upper extremity dysfunction in people with Parkinson disease and Huntington disease: Facilitating outcomes across the disease lifespan. *Journal of Hand Therapy: Official Journal of the American Society of Hand Therapists*. 2013;26:148–154.

- Chen JJ. Implications for managed care for improving outcomes in Parkinson's disease: Balancing aggressive treatment with appropriate care. *The American Journal of Managed Care*. 2011;17 Suppl 12:S322–S327.
- Braak H, Braak E. Pathoanatomy of Parkinson's disease. *Journal of Neurology*. 2000;247 Suppl 2:II3–II10.
- Berardelli A, Rothwell JC, Thompson PD, Hallett M. Pathophysiology of bradykinesia in Parkinson's disease. *Brain: A Journal of Neurology*. 2001;124:2131–2146.
- 26. Jankovic J. Parkinson's disease: Clinical features and diagnosis. *Journal of Neurology, Neurosurgery, and Psychiatry.* 2008;79:368–376.
- 27. Dauer W, Przedborski S. Parkinson's disease: Mechanisms and models. *Neuron*. 2003;39:889–909.
- 28. Lanska DJ. George Huntington (1850–1916) and hereditary chorea. *Journal of the History of the Neurosciences*. 2000;9:76–89.
- [No authors listed]. A novel gene containing a trinucleotide repeat that is expanded and unstable on Huntington's disease chromosomes. The Huntington's disease collaborative research group. *Cell*. 1993;72:971–983.
- 30. Langbehn DR, Hayden MR, Paulsen JS, Group P-HIotHS. CAG-repeat length and the age of onset in Huntington disease (HD): A review and validation study of statistical approaches. *American Journal of Medical Genetics. Part B, Neuropsychiatric Genetics: The Official Publication of the International Society of Psychiatric Genetics.* 2010;153B:397–408.
- 31. Sturrock A, Leavitt BR. The clinical and genetic features of Huntington disease. *Journal* of Geriatric Psychiatry and Neurology. 2010;23:243–259.
- 32. Tabrizi SJ, Scahill RI, Durr A, Roos RA, Leavitt BR, Jones R, Landwehrmeyer GB, et al. Biological and clinical changes in premanifest and early stage Huntington's disease in the TRACK-HD study: The 12-month longitudinal analysis. *Lancet Neurology*. 2011;10:31–42.
- Banaszkiewicz K, Sitek EJ, Rudzinska M, Soltan W, Slawek J, Szczudlik A. Huntington's disease from the patient, caregiver and physician's perspectives: Three sides of the same coin? *Journal of Neural Transmission*. 2012;119:1361–1365.
- 34. Lobo MK. Molecular profiling of striatonigral and striatopallidal medium spiny neurons past, present, and future. *International Review of Neurobiology*. 2009;89:1–35.
- Rosenblatt A, Abbott MH, Gourley LM, Troncoso JC, Margolis RL, Brandt J, Ross CA. Predictors of neuropathological severity in 100 patients with Huntington's disease. *Annals of Neurology*. 2003;54:488–493.
- Schoenfeld M, Myers RH, Cupples LA, Berkman B, Sax DS, Clark E. Increased rate of suicide among patients with Huntington's disease. *Journal of Neurology, Neurosurgery,* and Psychiatry. 1984;47:1283–1287.
- Rowland LP. How amyotrophic lateral sclerosis got its name: The clinical-pathologic genius of Jean-Martin Charcot. Archives of Neurology. 2001;58:512–515.
- Rowland LP, Shneider NA. Amyotrophic lateral sclerosis. *The New England Journal of Medicine*. 2001;344:1688–1700.
- 39. van der Steen I, van den Berg JP, Buskens E, Lindeman E, van den Berg LH. The costs of amyotrophic lateral sclerosis, according to type of care. *Amyotrophic Lateral Sclerosis: Official Publication of the World Federation of Neurology Research Group on Motor Neuron Diseases*. 2009;10:27–34.
- Jennum P, Ibsen R, Pedersen SW, Kjellberg J. Mortality, health, social and economic consequences of amyotrophic lateral sclerosis: A controlled national study. *Journal of Neurology*. 2013;260:785–793.
- Steenland K, MacNeil J, Seals R, Levey A. Factors affecting survival of patients with neurodegenerative disease. *Neuroepidemiology*. 2010;35:28–35.
- 42. Majoor-Krakauer D, Willems PJ, Hofman A. Genetic epidemiology of amyotrophic lateral sclerosis. *Clinical Genetics*. 2003;63:83–101.

- 43. Bastos AF, Orsini M, Machado D, Mello MP, Nader S, Silva JG, da Silva Catharino AM, et al. Amyotrophic lateral sclerosis: One or multiple causes? *Neurology International*. 2011;3:e4.
- 44. Rosen DR, Siddique T, Patterson D, Figlewicz DA, Sapp P, Hentati A, Donaldson D, et al. Mutations in Cu/Zn superoxide dismutase gene are associated with familial amyotrophic lateral sclerosis. *Nature*. 1993;362:59–62.
- 45. Sutedja NA, Veldink JH, Fischer K, Kromhout H, Heederik D, Huisman MH, Wokke JH, van den Berg LH. Exposure to chemicals and metals and risk of amyotrophic lateral sclerosis: A systematic review. Amyotrophic Lateral Sclerosis: Official Publication of the World Federation of Neurology Research Group on Motor Neuron Diseases. 2009;10:302–309.
- 46. Trojsi F, Sagnelli A, Vanacore N, Piccirillo G, Daniele L, Izzo F, Laiola A, et al. Clinical features and lifestyle of patients with amyotrophic lateral sclerosis in Campania: Brief overview of an Italian database. *Annali dell'Istituto Superiore di Sanita*. 2012;48:287–291.
- Kinsley L, Siddique T. Amyotrophic lateral sclerosis overview. In: Pagon RA, Bird TD, Dolan CR, Stephens K, Adam MP, eds. *Genereviews*. Seattle, WA: University of Washington; 1993.
- 48. Shoesmith CL, Findlater K, Rowe A, Strong MJ. Prognosis of amyotrophic lateral sclerosis with respiratory onset. *Journal of Neurology, Neurosurgery, and Psychiatry*. 2007;78:629–631.
- 49. Giordana MT, Ferrero P, Grifoni S, Pellerino A, Naldi A, Montuschi A. Dementia and cognitive impairment in amyotrophic lateral sclerosis: A review. *Neurological Sciences: Official Journal of the Italian Neurological Society and of the Italian Society of Clinical Neurophysiology*. 2011;32:9–16.
- 50. Ribaric S. Diet and aging. Oxidative Medicine and Cellular Longevity. 2012; 2012: 741468.
- 51. Gomez-Pinilla F. The influences of diet and exercise on mental health through hormesis. *Ageing Research Reviews*. 2008;7:49–62.
- 52. Gomez-Pinilla F. The combined effects of exercise and foods in preventing neurological and cognitive disorders. *Preventive Medicine*. 2011;52 Suppl 1:S75–S80.
- Chytrova G, Ying Z, Gomez-Pinilla F. Exercise contributes to the effects of DHA dietary supplementation by acting on membrane-related synaptic systems. *Brain Research*. 2010;1341:32–40.
- Wu A, Ying Z, Gomez-Pinilla F. Docosahexaenoic acid dietary supplementation enhances the effects of exercise on synaptic plasticity and cognition. *Neuroscience*. 2008;155:751–759.
- Luchsinger JA, Tang MX, Shea S, Mayeux R. Caloric intake and the risk of Alzheimer disease. Archives of Neurology. 2002;59:1258–1263.
- 56. Gustafson D, Rothenberg E, Blennow K, Steen B, Skoog I. An 18-year follow-up of overweight and risk of Alzheimer disease. *Archives of Internal Medicine*. 2003;163:1524–1528.
- 57. Grant WB. Dietary links to Alzheimer's disease: 1999 update. *Journal of Alzheimer's Disease*. 1999;1:197–201.
- Kalmijn S, Launer LJ, Ott A, Witteman JC, Hofman A, Breteler MM. Dietary fat intake and the risk of incident dementia in the Rotterdam study. *Annals of Neurology*. 1997;42:776–782.
- Barberger-Gateau P, Letenneur L, Deschamps V, Peres K, Dartigues JF, Renaud S. Fish, meat, and risk of dementia: Cohort study. *British Medical Journal*. 2002;325:932–933.
- Morris MC, Evans DA, Bienias JL, Tangney CC, Bennett DA, Aggarwal N, Schneider J, Wilson RS. Dietary fats and the risk of incident Alzheimer disease. *Archives of Neurology*. 2003;60:194–200.
- Morris MC, Evans DA, Bienias JL, Tangney CC, Bennett DA, Wilson RS, Aggarwal N, Schneider J. Consumption of fish and n-3 fatty acids and risk of incident Alzheimer disease. *Archives of Neurology*. 2003;60:940–946.

- 62. Huang TL, Zandi PP, Tucker KL, Fitzpatrick AL, Kuller LH, Fried LP, Burke GL, Carlson MC. Benefits of fatty fish on dementia risk are stronger for those without ApoE epsilon4. *Neurology*. 2005;65:1409–1414.
- Barberger-Gateau P, Raffaitin C, Letenneur L, Berr C, Tzourio C, Dartigues JF, Alperovitch A. Dietary patterns and risk of dementia: The three-city cohort study. *Neurology*. 2007;69:1921–1930.
- 64. Scarmeas N, Stern Y, Tang MX, Mayeux R, Luchsinger JA. Mediterranean diet and risk for Alzheimer's disease. *Annals of Neurology*. 2006;59:912–921.
- Scarmeas N, Luchsinger JA, Mayeux R, Stern Y. Mediterranean diet and Alzheimer disease mortality. *Neurology*. 2007;69:1084–1093.
- 66. Gu Y, Luchsinger JA, Stern Y, Scarmeas N. Mediterranean diet, inflammatory and metabolic biomarkers, and risk of Alzheimer's disease. *Journal of Alzheimer's disease*. 2010;22:483–492.
- Tangney CC, Kwasny MJ, Li H, Wilson RS, Evans DA, Morris MC. Adherence to a mediterranean-type dietary pattern and cognitive decline in a community population. *The American Journal of Clinical Nutrition*. 2011;93:601–607.
- Solfrizzi V, Frisardi V, Capurso C, D'Introno A, Colacicco AM, Vendemiale G, Capurso A, Panza F. Dietary fatty acids in dementia and predementia syndromes: Epidemiological evidence and possible underlying mechanisms. *Ageing Research Reviews*. 2010;9:184–199.
- Rolland Y, Pillard F, Klapouszczak A, Reynish E, Thomas D, Andrieu S, Riviere D, Vellas B. Exercise program for nursing home residents with Alzheimer's disease: A 1-year randomized, controlled trial. *Journal of the American Geriatrics Society*. 2007;55:158–165.
- Rolland Y, Abellan van Kan G, Vellas B. Physical activity and Alzheimer's disease: From prevention to therapeutic perspectives. *Journal of the American Medical Directors Association*. 2008;9:390–405.
- Lautenschlager NT, Cox KL, Flicker L, Foster JK, van Bockxmeer FM, Xiao J, Greenop KR, Almeida OP. Effect of physical activity on cognitive function in older adults at risk for Alzheimer disease: A randomized trial. *The Journal of the American Medical Association*. 2008;300:1027–1037.
- Geda YE, Roberts RO, Knopman DS, Christianson TJ, Pankratz VS, Ivnik RJ, Boeve BF, Tangalos EG, Petersen RC, Rocca WA. Physical exercise, aging, and mild cognitive impairment: A population-based study. *Archives of Neurology*. 2010;67:80–86.
- Scarmeas N, Luchsinger JA, Schupf N, Brickman AM, Cosentino S, Tang MX, Stern Y. Physical activity, diet, and risk of Alzheimer disease. *The Journal of the American Medical Association*. 2009;302:627–637.
- 74. Chen H, Zhang SM, Hernan MA, Willett WC, Ascherio A. Dietary intakes of fat and risk of Parkinson's disease. *American Journal of Epidemiology*. 2003;157:1007–1014.
- de Lau LM, Bornebroek M, Witteman JC, Hofman A, Koudstaal PJ, Breteler MM. Dietary fatty acids and the risk of Parkinson disease: The Rotterdam study. *Neurology*. 2005;64:2040–2045.
- Gao X, Chen H, Fung TT, Logroscino G, Schwarzschild MA, Hu FB, Ascherio A. Prospective study of dietary pattern and risk of Parkinson disease. *The American Journal* of Clinical Nutrition. 2007;86:1486–1494.
- 77. Alcalay RN, Gu Y, Mejia-Santana H, Cote L, Marder KS, Scarmeas N. The association between mediterranean diet adherence and Parkinson's disease. *Movement Disorders: Official Journal of the Movement Disorder Society*. 2012;27:771–774.
- Ang ET, Tai YK, Lo SQ, Seet R, Soong TW. Neurodegenerative diseases: Exercising toward neurogenesis and neuroregeneration. *Frontiers in Aging Neuroscience*. 2010;2:pii:25.

- Hirsch MA, Farley BG. Exercise and neuroplasticity in persons living with Parkinson's disease. *European Journal of Physical and Rehabilitation Medicine*. 2009;45:215–229.
- Dibble LE, Addison O, Papa E. The effects of exercise on balance in persons with parkinson's disease: A systematic review across the disability spectrum. *Journal of Neurologic Physical Therapy*. 2009;33:14–26.
- Schalow G, Paasuke M, Jaigma P. Integrative re-organization mechanism for reducing tremor in Parkinson's disease patients. *Electromyography and Clinical Neurophysiology*. 2005;45:407–415.
- Peluso MA, Guerra de Andrade LH. Physical activity and mental health: The association between exercise and mood. *Clinics*. 2005;60:61–70.
- Crizzle AM, Newhouse IJ. Is physical exercise beneficial for persons with Parkinson's disease? *Clinical Journal of Sport Medicine: Official Journal of the Canadian Academy* of Sport Medicine. 2006;16:422–425.
- Goodwin VA, Richards SH, Taylor RS, Taylor AH, Campbell JL. The effectiveness of exercise interventions for people with Parkinson's disease: A systematic review and meta-analysis. *Movement Disorders: Official Journal of the Movement Disorder Society*. 2008;23:631–640.
- Ridgel AL, Vitek JL, Alberts JL. Forced, not voluntary, exercise improves motor function in Parkinson's disease patients. *Neurorehabilitation and Neural Repair*. 2009;23:600–608.
- 86. Hirsch MA, Toole T, Maitland CG, Rider RA. The effects of balance training and high-intensity resistance training on persons with idiopathic Parkinson's disease. *Archives of Physical Medicine and Rehabilitation*. 2003;84:1109–1117.
- Falvo MJ, Schilling BK, Earhart GM. Parkinson's disease and resistive exercise: Rationale, review, and recommendations. *Movement Disorders: Official Journal of the Movement Disorder Society*. 2008;23:1–11.
- 88. Pascoe M. Huntington's disease and low tryptophan diet. *Medical Hypotheses*. 1993;41:325–326.
- Buruma OJ, Van der Kamp W, Barendswaard EC, Roos RA, Kromhout D, Van der Velde EA. Which factors influence age at onset and rate of progression in Huntington's disease? *Journal of the Neurological Sciences*. 1987;80:299–306.
- 90. Maalouf M, Rho JM, Mattson MP. The neuroprotective properties of calorie restriction, the ketogenic diet, and ketone bodies. *Brain Research Reviews*. 2009;59:293–315.
- Renoir T, Pang TY, Zajac MS, Chan G, Du X, Leang L, Chevarin C, Lanfumey L, Hannan AJ. Treatment of depressive-like behaviour in Huntington's disease mice by chronic sertraline and exercise. *British Journal of Pharmacology*. 2012;165:1375–1389.
- Cepeda C, Cummings DM, Hickey MA, Kleiman-Weiner M, Chen JY, Watson JB, Levine MS. Rescuing the corticostriatal synaptic disconnection in the r6/2 mouse model of Huntington's disease: Exercise, adenosine receptors and ampakines. *PLoS Currents*. 2010;2:pii:RRN1182.
- Pang TY, Stam NC, Nithianantharajah J, Howard ML, Hannan AJ. Differential effects of voluntary physical exercise on behavioral and brain-derived neurotrophic factor expression deficits in Huntington's disease transgenic mice. *Neuroscience*. 2006;141:569–584.
- 94. Patel BP, Hamadeh MJ. Nutritional and exercise-based interventions in the treatment of amyotrophic lateral sclerosis. *Clinical Nutrition*. 2009;28:604–617.
- Ascherio A, Weisskopf MG, O'Reilly EJ, Jacobs EJ, McCullough ML, Calle EE, Cudkowicz M, Thun MJ. Vitamin E intake and risk of amyotrophic lateral sclerosis. *Annals of Neurology*. 2005;57:104–110.
- 96. Veldink JH, Kalmijn S, Groeneveld GJ, Wunderink W, Koster A, de Vries JHM, van der Luyt J, Wokke JHJ, Van den Berg LH. Intake of polyunsaturated fatty acids and vitamin E reduces the risk of developing amyotrophic lateral sclerosis. *Journal of Neurology, Neurosurgery & Psychiatry*. 2007;78:367–371.

- Hamadeh MJ, Rodriguez MC, Kaczor JJ, Tarnopolsky MA. Caloric restriction transiently improves motor performance but hastens clinical onset of disease in the Cu/Zn-superoxide dismutase mutant g93a mouse. *Muscle & Nerve*. 2005;31:214–220.
- Hamadeh MJ, Tarnopolsky MA. Transient caloric restriction in early adulthood hastens disease endpoint in male, but not female, Cu/Zn-sod mutant g93a mice. *Muscle & Nerve*. 2006;34:709–719.
- Pedersen WA, Mattson MP. No benefit of dietary restriction on disease onset or progression in amyotrophic lateral sclerosis Cu/Zn-superoxide dismutase mutant mice. *Brain Research*. 1999;833:117–120.
- Kasarskis EJ, Berryman S, Vanderleest JG, Schneider AR, McClain CJ. Nutritional status of patients with amyotrophic lateral sclerosis: Relation to the proximity of death. *American Journal of Clinical Nutrition*. 1996;63:130–137.
- 101. Rio A, Cawadias E. Nutritional advice and treatment by dietitians to patients with amyotrophic lateral sclerosis/motor neurone disease: A survey of current practice in England, Wales, Northern Ireland and Canada. *Journal of Human Nutrition and Dietetics*. 2007;20:3–13.
- 102. Zhao Z, Lange DJ, Voustianiouk A, MacGrogan D, Ho L, Suh J, Humala N, Thiyagarajan M, Wang J, Pasinetti GM. A ketogenic diet as a potential novel therapeutic intervention in amyotrophic lateral sclerosis. *BMC Neuroscience*. 2006;7:29.
- 103. Scarmeas N, Shih T, Stern Y, Ottman R, Rowland LP. Premorbid weight, body mass, and varsity athletics in ALS. *Neurology*. 2002;59:773–775.
- Strickland D, Smith SA, Dolliff G, Goldman L, Roelofs RI. Physical activity, trauma, and ALS: A case-control study. *Acta Neurologica Scandinavica*. 1996;94:45–50.
- 105. Siciliano G, Pastorini E, Pasquali L, Manca ML, Iudice A, Murri L. Impaired oxidative metabolism in exercising muscle from ALS patients. *Journal of the Neurological Sciences*. 2001;191:61–65.
- Bello-Haas VD, Florence JM, Kloos AD, Scheirbecker J, Lopate G, Hayes SM, Pioro EP, Mitsumoto H. A randomized controlled trial of resistance exercise in individuals with ALS. *Neurology*. 2007;68:2003–2007.
- 107. Pinto AC, Alves M, Nogueira A, Evangelista T, Carvalho J, Coelho A, de Carvalho M, Sales-Luis ML. Can amyotrophic lateral sclerosis patients with respiratory insufficiency exercise? *Journal of the Neurological Sciences*. 1999;169:69–75.
- Drory VE, Goltsman E, Reznik JG, Mosek A, Korczyn AD. The value of muscle exercise in patients with amyotrophic lateral sclerosis. *Journal of the Neurological Sciences*. 2001;191:133–137.
- 109. Migliore L, Coppede F. Genetics, environmental factors and the emerging role of epigenetics in neurodegenerative diseases. *Mutation Research*. 2009;667:82–97.
- 110. Findeis MA. The role of amyloid beta peptide 42 in Alzheimer's disease. *Pharmacology & Therapeutics*. 2007;116:266–286.
- 111. Sambamurti K, Greig NH, Lahiri DK. Advances in the cellular and molecular biology of the beta-amyloid protein in Alzheimer's disease. *Neuromolecular Medicine*. 2002;1:1–31.
- 112. Strittmatter WJ, Saunders AM, Schmechel D, Pericak-Vance M, Enghild J, Salvesen GS, Roses AD. Apolipoprotein e: High-avidity binding to beta-amyloid and increased frequency of type 4 allele in late-onset familial Alzheimer disease. *Proceedings of the National Academy of Sciences of the United States of America*. 1993;90:1977–1981.
- 113. Farrer LA, Cupples LA, Haines JL, Hyman B, Kukull WA, Mayeux R, Myers RH, Pericak-Vance MA, Risch N, van Duijn CM. Effects of age, sex, and ethnicity on the association between apolipoprotein E genotype and Alzheimer disease. A meta-analysis. ApoE and Alzheimer Disease Meta Analysis Consortium. *The Journal of the American Medical Association*. 1997;278:1349–1356.

- 114. Kan R, Wang B, Zhang C, Jin F, Yang Z, Ji S, Lu Z, Zheng C, Wang L. Genetic association of BACE1 gene polymorphism C786G with late-onset Alzheimer's disease in Chinese. *Journal of Molecular Neuroscience*. 2005;25:127–131.
- 115. Confaloni A, Terreni L, Piscopo P, Crestini A, Campeggi LM, Frigerio CS, Blotta I, et al. Nicastrin gene in familial and sporadic Alzheimer's disease. *Neuroscience Letters*. 2003;353:61–65.
- 116. Jia L, Ye J, L V H, Wang W, Zhou C, Zhang X, Xu J, Wang L, Jia J. Genetic association between polymorphisms of PEN2 gene and late onset Alzheimer's disease in the North Chinese population. *Brain Research*. 2007;1141:10–14.
- 117. Bonini NM, Giasson BI. Snaring the function of alpha-synuclein. Cell. 2005;123:359–361.
- 118. Cooper AA, Gitler AD, Cashikar A, Haynes CM, Hill KJ, Bhullar B, Liu K, et al. Alpha-synuclein blocks ER-golgi traffic and Rab1 rescues neuron loss in Parkinson's models. *Science*. 2006;313:324–328.
- 119. Zarranz JJ, Alegre J, Gomez-Esteban JC, Lezcano E, Ros R, Ampuero I, Vidal L, et al. The new mutation, E46K, of alpha-synuclein causes Parkinson and Lewy body dementia. *Annals of Neurology*. 2004;55:164–173.
- Singleton AB, Farrer M, Johnson J, Singleton A, Hague S, Kachergus J, Hulihan M, et al. Alpha-synuclein locus triplication causes Parkinson's disease. *Science*. 2003;302:841.
- 121. Kitada T, Asakawa S, Hattori N, Matsumine H, Yamamura Y, Minoshima S, Yokochi M, Mizuno Y, Shimizu N. Mutations in the parkin gene cause autosomal recessive juvenile Parkinsonism. *Nature*. 1998;392:605–608.
- 122. Gasser T. Genetics of Parkinson's disease. Journal of Neurology. 2001;248:833-840.
- 123. Leroy E, Boyer R, Auburger G, Leube B, Ulm G, Mezey E, Harta G, et al. The ubiquitin pathway in Parkinson's disease. *Nature*. 1998;395:451–452.
- 124. Lim KL, Tan JM. Role of the ubiquitin proteasome system in Parkinson's disease. *BMC Biochemistry*. 2007;8 Suppl 1:S13.
- 125. Bonifati V, Rohe CF, Breedveld GJ, Fabrizio E, De Mari M, Tassorelli C, Tavella A, et al. Early-onset parkinsonism associated with PINK1 mutations: Frequency, geno-types, and phenotypes. *Neurology*. 2005;65:87–95.
- 126. West AB, Moore DJ, Biskup S, Bugayenko A, Smith WW, Ross CA, Dawson VL, Dawson TM. Parkinson's disease-associated mutations in leucine-rich repeat kinase 2 augment kinase activity. *Proceedings of the National Academy of Sciences of the United States of America*. 2005;102:16842–16847.
- 127. Lorincz MT. Clinical implications of Parkinson's disease genetics. Seminars in Neurology. 2006;26:492–498.
- 128. Tan EK. The role of common genetic risk variants in Parkinson disease. *Clinical Genetics*. 2007;72:387–393.
- 129. McCulloch CC, Kay DM, Factor SA, Samii A, Nutt JG, Higgins DS, Griffith A, et al. Exploring gene-environment interactions in Parkinson's disease. *Human Genetics*. 2008;123:257–265.
- 130. Menegon A, Board PG, Blackburn AC, Mellick GD, Le Couteur DG. Parkinson's disease, pesticides, and glutathione transferase polymorphisms. *Lancet*. 1998;352:1344–1346.
- 131. Elbaz A, Levecque C, Clavel J, Vidal JS, Richard F, Amouyel P, Alperovitch A, Chartier-Harlin MC, Tzourio C. Cyp2D6 polymorphism, pesticide exposure, and Parkinson's disease. *Annals of Neurology*. 2004;55:430–434.
- 132. DiFiglia M, Sapp E, Chase KO, Davies SW, Bates GP, Vonsattel JP, Aronin N. Aggregation of huntingtin in neuronal intranuclear inclusions and dystrophic neurites in brain. *Science*. 1997;277:1990–1993.
- 133. Perutz MF. Glutamine repeats and inherited neurodegenerative diseases: Molecular aspects. *Current Opinion in Structural Biology*. 1996;6:848–858.
- 134. Gil JM, Rego AC. Mechanisms of neurodegeneration in Huntington's disease. *The European Journal of Neuroscience*. 2008;27:2803–2820.

- 135. Arning L, Kraus PH, Valentin S, Saft C, Andrich J, Epplen JT. NR2A and NR2B receptor gene variations modify age at onset in Huntington disease. *Neurogenetics*. 2005;6:25–28.
- 136. van Dellen A, Grote HE, Hannan AJ. Gene-environment interactions, neuronal dysfunction and pathological plasticity in Huntington's disease. *Clinical and Experimental Pharmacology & Physiology*. 2005;32:1007–1019.
- 137. Pasinelli P, Brown RH. Molecular biology of amyotrophic lateral sclerosis: Insights from genetics. *Nature Reviews. Neuroscience*. 2006;7:710–723.
- Yamanaka K, Cleveland DW. Determinants of rapid disease progression in ALS. *Neurology*. 2005;65:1859–1860.
- Cleveland DW, Laing N, Hurse PV, Brown RH, Jr. Toxic mutants in Charcot's sclerosis. *Nature*. 1995;378:342–343.
- 140. Cai H, Lin X, Xie C, Laird FM, Lai C, Wen H, Chiang HC, et al. Loss of ALS2 function is insufficient to trigger motor neuron degeneration in knock-out mice but predisposes neurons to oxidative stress. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*. 2005;25:7567–7574.
- 141. Hadano S, Benn SC, Kakuta S, Otomo A, Sudo K, Kunita R, Suzuki-Utsunomiya K, et al. Mice deficient in the Rab5 guanine nucleotide exchange factor ALS2/Alsin exhibit age-dependent neurological deficits and altered endosome trafficking. *Human Molecular Genetics*. 2006;15:233–250.
- 142. Kanekura K, Hashimoto Y, Niikura T, Aiso S, Matsuoka M, Nishimoto I. Alsin, the product of ALS2 gene, suppresses SOD1 mutant neurotoxicity through RhoGEF domain by interacting with SOD1 mutants. *The Journal of Biological Chemistry*. 2004;279:19247–19256.
- 143. Suraweera A, Becherel OJ, Chen P, Rundle N, Woods R, Nakamura J, Gatei M, et al. Senataxin, defective in ataxia oculomotor apraxia type 2, is involved in the defense against oxidative DNA damage. *The Journal of Cell Biology*. 2007;177:969–979.
- 144. Chen YZ, Bennett CL, Huynh HM, Blair IP, Puls I, Irobi J, Dierick I, et al. DNA/RNA helicase gene mutations in a form of juvenile amyotrophic lateral sclerosis (ALS4). *American Journal of Human Genetics*. 2004;74:1128–1135.
- 145. Teuling E, Ahmed S, Haasdijk E, Demmers J, Steinmetz MO, Akhmanova A, Jaarsma D, Hoogenraad CC. Motor neuron disease-associated mutant vesicle-associated membrane protein-associated protein (VAP) B recruits wild-type VAPs into endoplasmic reticulum-derived tubular aggregates. *The Journal of Neuroscience: The Official Journal* of the Society for Neuroscience. 2007;27:9801–9815.
- 146. Migliore L, Coppede F. Environmental-induced oxidative stress in neurodegenerative disorders and aging. *Mutation Research*. 2009;674:73–84.
- 147. Rivera-Mancia S, Perez-Neri I, Rios C, Tristan-Lopez L, Rivera-Espinosa L, Montes S. The transition metals copper and iron in neurodegenerative diseases. *Chemico-Biological Interactions*. 2010;186:184–199.
- 148. Williams BB, Li D, Wegrzynowicz M, Vadodaria BK, Anderson JG, Kwakye GF, Aschner M, Erikson KM, Bowman AB. Disease-toxicant screen reveals a neuroprotective interaction between Huntington's disease and manganese exposure. *Journal of Neurochemistry*. 2010;112:227–237.
- Jomova K, Vondrakova D, Lawson M, Valko M. Metals, oxidative stress and neurodegenerative disorders. *Molecular and Cellular Biochemistry*. 2010;345:91–104.
- Holliday R. Epigenetics: A historical overview. *Epigenetics: Official Journal of the DNA* Methylation Society. 2006;1:76–80.
- 151. Zawia NH, Lahiri DK, Cardozo-Pelaez F. Epigenetics, oxidative stress, and Alzheimer disease. *Free Radical Biology & Medicine*. 2009;46:1241–1249.
- 152. Drechsel DA, Patel M. Role of reactive oxygen species in the neurotoxicity of environmental agents implicated in Parkinson's disease. *Free Radical Biology & Medicine*. 2008;44:1873–1886.

- 153. Calderon-Garciduenas L, Solt AC, Henriquez-Roldan C, Torres-Jardon R, Nuse B, Herritt L, Villarreal-Calderon R, et al. Long-term air pollution exposure is associated with neuroinflammation, an altered innate immune response, disruption of the blood-brain barrier, ultrafine particulate deposition, and accumulation of amyloid beta-42 and alpha-synuclein in children and young adults. *Toxicologic Pathology*. 2008;36:289–310.
- 154. Block ML, Wu X, Pei Z, Li G, Wang T, Qin L, Wilson B, Yang J, Hong JS, Veronesi B. Nanometer size diesel exhaust particles are selectively toxic to dopaminergic neurons: The role of microglia, phagocytosis, and NADPH oxidase. *Federation of American Societies for Experimental Biology Journal*. 2004;18:1618–1620.
- 155. Mattson MP, Magnus T. Ageing and neuronal vulnerability. *Nature Reviews*. *Neuroscience*. 2006;7:278–294.
- 156. Sohal RS, Weindruch R. Oxidative stress, caloric restriction, and aging. *Science*. 1996;273:59–63.
- 157. Bratic I, Trifunovic A. Mitochondrial energy metabolism and ageing. *Biochimica et Biophysica Acta*. 2010;1797:961–967.
- 158. Hipkiss AR. Accumulation of altered proteins and ageing: Causes and effects. *Experimental Gerontology*. 2006;41:464–473.
- 159. Lu T, Pan Y, Kao SY, Li C, Kohane I, Chan J, Yankner BA. Gene regulation and DNA damage in the ageing human brain. *Nature*. 2004;429:883–891.
- 160. Reed TT. Lipid peroxidation and neurodegenerative disease. *Free Radical Biology & Medicine*. 2011;51:1302–1319.
- 161. Good PF, Hsu A, Werner P, Perl DP, Olanow CW. Protein nitration in Parkinson's disease. Journal of Neuropathology and Experimental Neurology. 1998;57:338–342.
- 162. Aksenov MY, Aksenova MV, Butterfield DA, Geddes JW, Markesbery WR. Protein oxidation in the brain in Alzheimer's disease. *Neuroscience*. 2001;103:373–383.
- 163. Keller JN, Gee J, Ding Q. The proteasome in brain aging. *Ageing Research Reviews*. 2002;1:279–293.
- 164. Trojanowski JQ, Mattson MP. Overview of protein aggregation in single, double, and triple neurodegenerative brain amyloidoses. *Neuromolecular Medicine*. 2003;4:1–6.
- 165. Rakhit R, Crow JP, Lepock JR, Kondejewski LH, Cashman NR, Chakrabartty A. Monomeric Cu,Zn-superoxide dismutase is a common misfolding intermediate in the oxidation models of sporadic and familial amyotrophic lateral sclerosis. *The Journal of Biological Chemistry*. 2004;279:15499–15504.
- 166. Mattson MP. Pathways towards and away from Alzheimer's disease. *Nature*. 2004;430:631–639.
- 167. Zuccato C, Tartari M, Crotti A, Goffredo D, Valenza M, Conti L, Cataudella T, et al. Huntingtin interacts with REST/NRSF to modulate the transcription of NRSE-controlled neuronal genes. *Nature Genetics*. 2003;35:76–83.
- 168. Valko M, Leibfritz D, Moncol J, Cronin MT, Mazur M, Telser J. Free radicals and antioxidants in normal physiological functions and human disease. *The International Journal of Biochemistry & Cell Biology*. 2007;39:44–84.
- Emerit J, Edeas M, Bricaire F. Neurodegenerative diseases and oxidative stress. Biomedicine & Pharmacotherapy. 2004;58:39–46.
- 170. Montine TJ, Neely MD, Quinn JF, Beal MF, Markesbery WR, Roberts LJ, Morrow JD. Lipid peroxidation in aging brain and Alzheimer's disease. *Free Radical Biology & Medicine*. 2002;33:620–626.
- 171. Uttara B, Singh AV, Zamboni P, Mahajan RT. Oxidative stress and neurodegenerative diseases: A review of upstream and downstream antioxidant therapeutic options. *Current Neuropharmacology*. 2009;7:65–74.
- 172. Valko M, Morris H, Cronin MT. Metals, toxicity and oxidative stress. *Current Medicinal Chemistry*. 2005;12:1161–1208.
- 173. Lloyd RV, Hanna PM, Mason RP. The origin of the hydroxyl radical oxygen in the Fenton reaction. *Free Radical Biology & Medicine*. 1997;22:885–888.
- 174. Bolognin S, Messori L, Drago D, Gabbiani C, Cendron L, Zatta P. Aluminum, copper, iron and zinc differentially alter amyloid-Aβ(1-42) aggregation and toxicity. *The International Journal of Biochemistry & Cell Biology*. 2011;43:877–885.
- 175. Lotharius J, Brundin P. Impaired dopamine storage resulting from alpha-synuclein mutations may contribute to the pathogenesis of Parkinson's disease. *Human Molecular Genetics*. 2002;11:2395–2407.
- Dasuri K, Zhang L, Keller JN. Oxidative stress, neurodegeneration, and the balance of protein degradation and protein synthesis. *Free Radical Biology & Medicine*. 2013;62:170–185.
- 177. Melo A, Monteiro L, Lima RM, Oliveira DM, Cerqueira MD, El-Bacha RS. Oxidative stress in neurodegenerative diseases: Mechanisms and therapeutic perspectives. *Oxidative Medicine and Cellular Longevity*. 2011;2011:467180.
- 178. Knott AB, Bossy-Wetzel E. Nitric oxide in health and disease of the nervous system. *Antioxidants & Redox Signaling*. 2009;11:541–554.
- 179. Andersen JK. Oxidative stress in neurodegeneration: Cause or consequence? *Nature Medicine*. 2004;10 Suppl:S18–S25.
- Pappolla MA, Omar RA, Kim KS, Robakis NK. Immunohistochemical evidence of oxidative [corrected] stress in Alzheimer's disease. *The American Journal of Pathology*. 1992;140:621–628.
- Perry TL, Godin DV, Hansen S. Parkinson's disease: A disorder due to nigral glutathione deficiency? *Neuroscience Letters*. 1982;33:305–310.
- 182. Texel SJ, Mattson MP. Impaired adaptive cellular responses to oxidative stress and the pathogenesis of Alzheimer's disease. *Antioxidants & Redox Signaling*. 2011;14:1519–1534.
- 183. Berlett BS, Stadtman ER. Protein oxidation in aging, disease, and oxidative stress. *The Journal of Biological Chemistry*. 1997;272:20313–20316.
- Cecarini V, Ding Q, Keller JN. Oxidative inactivation of the proteasome in Alzheimer's disease. *Free Radical Research*. 2007;41:673–680.
- 185. Castellani RJ, Perry G, Siedlak SL, Nunomura A, Shimohama S, Zhang J, Montine T, Sayre LM, Smith MA. Hydroxynonenal adducts indicate a role for lipid peroxidation in neocortical and brainstem Lewy bodies in humans. *Neuroscience Letters*. 2002;319:25–28.
- 186. Keller JN, Pang Z, Geddes JW, Begley JG, Germeyer A, Waeg G, Mattson MP. Impairment of glucose and glutamate transport and induction of mitochondrial oxidative stress and dysfunction in synaptosomes by amyloid beta-peptide: Role of the lipid peroxidation product 4-hydroxynonenal. *Journal of Neurochemistry*. 1997;69:273–284.
- 187. Marambaud P, Dreses-Werringloer U, Vingtdeux V. Calcium signaling in neurodegeneration. *Molecular Neurodegeneration*. 2009;4:20.
- 188. Gonfloni S, Maiani E, Di Bartolomeo C, Diederich M, Cesareni G. Oxidative stress, DNA damage, and c-Abl signaling: At the crossroad in neurodegenerative diseases? *International Journal of Cell Biology*. 2012;2012:683097.
- 189. Barber SC, Mead RJ, Shaw PJ. Oxidative stress in ALS: A mechanism of neurodegeneration and a therapeutic target. *Biochimica et Biop hysica Acta*. 2006;1762:1051–1067.
- Kovtun IV, Liu Y, Bjoras M, Klungland A, Wilson SH, McMurray CT. OGG1 initiates age-dependent CAG trinucleotide expansion in somatic cells. *Nature*. 2007;447:447–452.
- 191. Federico A, Cardaioli E, Da Pozzo P, Formichi P, Gallus GN, Radi E. Mitochondria, oxidative stress and neurodegeneration. *Journal of the Neurological Sciences*. 2012;322:254–262.
- 192. Karbowski M. Mitochondria on guard: Role of mitochondrial fusion and fission in the regulation of apoptosis. *Advances in Experimental Medicine and Biology*. 2010;687:131–142.

- 193. Hollenbeck PJ, Saxton WM. The axonal transport of mitochondria. *Journal of Cell Science*. 2005;118:5411–5419.
- 194. Taylor RW, Turnbull DM. Mitochondrial DNA mutations in human disease. *Nature Reviews. Genetics*. 2005;6:389–402.
- 195. Yan MH, Wang X, Zhu X. Mitochondrial defects and oxidative stress in Alzheimer disease and Parkinson disease. *Free Radical & Medicine*. 2013;62:90–101.
- 196. Hsieh RH, Hou JH, Hsu HS, Wei YH. Age-dependent respiratory function decline and DNA deletions in human muscle mitochondria. *Biochemistry and Molecular Biology International*. 1994;32:1009–1022.
- 197. Mecocci P, MacGarvey U, Beal MF. Oxidative damage to mitochondrial DNA is increased in Alzheimer's disease. *Annals of Neurology*. 1994;36:747–751.
- 198. Elson JL, Herrnstadt C, Preston G, Thal L, Morris CM, Edwardson JA, Beal MF, Turnbull DM, Howell N. Does the mitochondrial genome play a role in the etiology of Alzheimer's disease? *Human Genetics*. 2006;119:241–254.
- 199. Edland SD, Silverman JM, Peskind ER, Tsuang D, Wijsman E, Morris JC. Increased risk of dementia in mothers of Alzheimer's disease cases: Evidence for maternal inheritance. *Neurology*. 1996;47:254–256.
- 200. Carrieri G, Bonafe M, De Luca M, Rose G, Varcasia O, Bruni A, Maletta R, et al. Mitochondrial DNA haplogroups and apoE4 allele are non-independent variables in sporadic Alzheimer's disease. *Human Genetics*. 2001;108:194–198.
- 201. Weissman L, Jo DG, Sorensen MM, de Souza-Pinto NC, Markesbery WR, Mattson MP, Bohr VA. Defective DNA base excision repair in brain from individuals with Alzheimer's disease and amnestic mild cognitive impairment. *Nucleic Acids Research*. 2007;35:5545–5555.
- 202. Autere J, Moilanen JS, Finnila S, Soininen H, Mannermaa A, Hartikainen P, Hallikainen M, Majamaa K. Mitochondrial DNA polymorphisms as risk factors for Parkinson's disease and Parkinson's disease dementia. *Human Genetics*. 2004;115:29–35.
- 203. Pickrell AM, Pinto M, Hida A, Moraes CT. Striatal dysfunctions associated with mitochondrial DNA damage in dopaminergic neurons in a mouse model of Parkinson's disease. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*. 2011;31:17649–17658.
- 204. Luoma P, Melberg A, Rinne JO, Kaukonen JA, Nupponen NN, Chalmers RM, Oldfors A, et al. Parkinsonism, premature menopause, and mitochondrial DNA polymerase gamma mutations: Clinical and molecular genetic study. *Lancet*. 2004;364:875–882.
- 205. van der Walt JM, Nicodemus KK, Martin ER, Scott WK, Nance MA, Watts RL, Hubble JP, et al. Mitochondrial polymorphisms significantly reduce the risk of Parkinson disease. *American Journal of Human Genetics*. 2003;72:804–811.
- 206. Simon DK, Lin MT, Zheng L, Liu GJ, Ahn CH, Kim LM, Mauck WM, Twu F, Beal MF, Johns DR. Somatic mitochondrial DNA mutations in cortex and substantia nigra in aging and Parkinson's disease. *Neurobiology of Aging*. 2004;25:71–81.
- 207. Gegg ME, Cooper JM, Schapira AH, Taanman JW. Silencing of pink1 expression affects mitochondrial DNA and oxidative phosphorylation in dopaminergic cells. *PloS One*. 2009;4:e4756.
- Su B, Wang X, Zheng L, Perry G, Smith MA, Zhu X. Abnormal mitochondrial dynamics and neurodegenerative diseases. *Biochimica et Biophysica Acta*. 2010;1802:135–142.
- 209. Wang X, Su B, Lee HG, Li X, Perry G, Smith MA, Zhu X. Impaired balance of mitochondrial fission and fusion in Alzheimer's disease. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*. 2009;29:9090–9103.
- Wang X, Su B, Fujioka H, Zhu X. Dynamin-like protein 1 reduction underlies mitochondrial morphology and distribution abnormalities in fibroblasts from sporadic Alzheimer's disease patients. *The American Journal of Pathology*. 2008;173:470–482.

- 211. Trushina E, Nemutlu E, Zhang S, Christensen T, Camp J, Mesa J, Siddiqui A, et al. Defects in mitochondrial dynamics and metabolomic signatures of evolving energetic stress in mouse models of familial Alzheimer's disease. *PloS One*. 2012;7:e32737.
- Cho DH, Nakamura T, Fang J, Cieplak P, Godzik A, Gu Z, Lipton SA. S-nitrosylation of Drp1 mediates beta-amyloid-related mitochondrial fission and neuronal injury. *Science*. 2009;324:102–105.
- Quintanilla RA, Dolan PJ, Jin YN, Johnson GV. Truncated tau and Aβ cooperatively impair mitochondria in primary neurons. *Neurobiology of Aging*. 2012;33:619 e625– 619 e635.
- 214. Rissman RA, Poon WW, Blurton-Jones M, Oddo S, Torp R, Vitek MP, LaFerla FM, Rohn TT, Cotman CW. Caspase-cleavage of tau is an early event in Alzheimer disease tangle pathology. *The Journal of Clinical Investigation*. 2004;114:121–130.
- Melov S, Adlard PA, Morten K, Johnson F, Golden TR, Hinerfeld D, Schilling B, et al. Mitochondrial oxidative stress causes hyperphosphorylation of tau. *PloS One*. 2007;2:e536.
- 216. Schapira AH, Gu M, Taanman JW, Tabrizi SJ, Seaton T, Cleeter M, Cooper JM. Mitochondria in the etiology and pathogenesis of Parkinson's disease. *Annals of Neurology*. 1998;44:S89–S98.
- 217. Battisti C, Formichi P, Radi E, Federico A. Oxidative-stress-induced apoptosis in PBLS of two patients with Parkinson disease secondary to alpha-synuclein mutation. *Journal of the Neurological Sciences*. 2008;267:120–124.
- 218. Cluskey S, Ramsden DB. Mechanisms of neurodegeneration in amyotrophic lateral sclerosis. *Molecular Pathology*. 2001;54:386–392.
- Lin MT, Beal MF. Mitochondrial dysfunction and oxidative stress in neurodegenerative diseases. *Nature*. 2006;443:787–795.
- 220. Forman MS, Trojanowski JQ, Lee VM. Neurodegenerative diseases: A decade of discoveries paves the way for therapeutic breakthroughs. *Nature Medicine*. 2004;10:1055–1063.
- 221. Betarbet R, Sherer TB, MacKenzie G, Garcia-Osuna M, Panov AV, Greenamyre JT. Chronic systemic pesticide exposure reproduces features of Parkinson's disease. *Nature Neuroscience*. 2000;3:1301–1306.
- 222. Hardy J, Selkoe DJ. The amyloid hypothesis of Alzheimer's disease: Progress and problems on the road to therapeutics. *Science*. 2002;297:353–356.
- Wolozin B, Behl C. Mechanisms of neurodegenerative disorders: Part 1: Protein aggregates. Archives of Neurology. 2000;57:793–796.
- 224. Ross CA, Poirier MA. Protein aggregation and neurodegenerative disease. *Nature Medicine*. 2004;10 Suppl:S10–S17.
- 225. Terry RD, Masliah E, Salmon DP, Butters N, DeTeresa R, Hill R, Hansen LA, Katzman R. Physical basis of cognitive alterations in Alzheimer's disease: Synapse loss is the major correlate of cognitive impairment. *Annals of Neurology*. 1991;30:572–580.
- 226. Dickson DW, Crystal HA, Mattiace LA, Masur DM, Blau AD, Davies P, Yen SH, Aronson MK. Identification of normal and pathological aging in prospectively studied nondemented elderly humans. *Neurobiology of Aging*. 1992;13:179–189.
- 227. Lue LF, Kuo YM, Roher AE, Brachova L, Shen Y, Sue L, Beach T, Kurth JH, Rydel RE, Rogers J. Soluble amyloid beta peptide concentration as a predictor of synaptic change in Alzheimer's disease. *The American Journal of Pathology*. 1999;155:853–862.
- 228. Volles MJ, Lee SJ, Rochet JC, Shtilerman MD, Ding TT, Kessler JC, Lansbury PT, Jr. Vesicle permeabilization by protofibrillar alpha-synuclein: Implications for the pathogenesis and treatment of Parkinson's disease. *Biochemistry*. 2001;40:7812–7819.
- 229. Poirier MA, Li H, Macosko J, Cai S, Amzel M, Ross CA. Huntingtin spheroids and protofibrils as precursors in polyglutamine fibrilization. *The Journal of Biological Chemistry*. 2002;277:41032–41037.

- 230. Giasson BI, Duda JE, Murray IV, Chen Q, Souza JM, Hurtig HI, Ischiropoulos H, Trojanowski JQ, Lee VM. Oxidative damage linked to neurodegeneration by selective alpha-synuclein nitration in synucleinopathy lesions. *Science*. 2000;290:985–989.
- 231. Krumova P, Weishaupt JH. Sumoylation in neurodegenerative diseases. *Cellular and Molecular Life Sciences*. 2013;70:2123–2138.
- 232. Cookson MR. Alpha-synuclein and neuronal cell death. *Molecular Neurodegeneration*. 2009;4:9.
- 233. Calabrese V, Scapagnini G, Colombrita C, Ravagna A, Pennisi G, Giuffrida Stella AM, Galli F, Butterfield DA. Redox regulation of heat shock protein expression in aging and neurodegenerative disorders associated with oxidative stress: A nutritional approach. *Amino Acids*. 2003;25:437–444.
- 234. Bonini NM. Chaperoning brain degeneration. *Proceedings of the National Academy of Sciences of the United States of America*. 2002;99 Suppl 4:16407–16411.
- 235. Hershko A, Ciechanover A. The ubiquitin system. *Annual Review of Biochemistry*. 1998;67:425–479.
- 236. Farout L, Friguet B. Proteasome function in aging and oxidative stress: Implications in protein maintenance failure. *Antioxidants & Redox Signaling*. 2006;8:205–216.
- 237. Yao D, Gu Z, Nakamura T, Shi ZQ, Ma Y, Gaston B, Palmer LA, et al. Nitrosative stress linked to sporadic Parkinson's disease: S-nitrosylation of parkin regulates its E3 ubiquitin ligase activity. *Proceedings of the National Academy of Sciences of the United States of America*. 2004;101:10810–10814.
- 238. Bruijn LI, Miller TM, Cleveland DW. Unraveling the mechanisms involved in motor neuron degeneration in ALS. *Annual Review of Neuroscience*. 2004;27:723–749.
- 239. Perry G, Friedman R, Shaw G, Chau V. Ubiquitin is detected in neurofibrillary tangles and senile plaque neurites of Alzheimer disease brains. *Proceedings of the National Academy of Sciences of the United States of America*. 1987;84:3033–3036.
- Skinner PJ, Koshy BT, Cummings CJ, Klement IA, Helin K, Servadio A, Zoghbi HY, Orr HT. Ataxin-1 with an expanded glutamine tract alters nuclear matrix-associated structures. *Nature*. 1997;389:971–974.
- 241. Cheroni C, Peviani M, Cascio P, Debiasi S, Monti C, Bendotti C. Accumulation of human SOD1 and ubiquitinated deposits in the spinal cord of SOD1G93A mice during motor neuron disease progression correlates with a decrease of proteasome. *Neurobiology of Disease*. 2005;18:509–522.
- 242. Giasson BI, Lee VM. Are ubiquitination pathways central to Parkinson's disease? *Cell*. 2003;114:1–8.
- Bence NF, Sampat RM, Kopito RR. Impairment of the ubiquitin-proteasome system by protein aggregation. *Science*. 2001;292:1552–1555.
- 244. Oddo S. The ubiquitin-proteasome system in Alzheimer's disease. *Journal of Cellular and Molecular Medicine*. 2008;12:363–373.
- 245. Wong E, Cuervo AM. Autophagy gone awry in neurodegenerative diseases. *Nature Neuroscience*. 2010;13:805–811.
- 246. Miller JP, Holcomb J, Al-Ramahi I, de Haro M, Gafni J, Zhang N, Kim E, et al. Matrix metalloproteinases are modifiers of huntingtin proteolysis and toxicity in Huntington's disease. *Neuron*. 2010;67:199–212.
- 247. Camins A, Verdaguer E, Folch J, Pallas M. Involvement of calpain activation in neurodegenerative processes. *CNS Drug Reviews*. 2006;12:135–148.
- 248. Nakanishi S. Molecular diversity of glutamate receptors and implications for brain function. *Science*. 1992;258:597–603.
- 249. Javitt DC, Zukin SR. The role of excitatory amino acids in neuropsychiatric illness. *The Journal of Neuropsychiatry and Clinical Neurosciences*. 1990;2:44–52.
- 250. Doble A. The role of excitotoxicity in neurodegenerative disease: Implications for therapy. *Pharmacology & Therapeutics*. 1999;81:163–221.

- 251. Mehta A, Prabhakar M, Kumar P, Deshmukh R, Sharma PL. Excitotoxicity: Bridge to various triggers in neurodegenerative disorders. *European Journal of Pharmacology*. 2013;698:6–18.
- 252. Dong XX, Wang Y, Qin ZH. Molecular mechanisms of excitotoxicity and their relevance to pathogenesis of neurodegenerative diseases. *Acta Pharmacologica Sinica*. 2009;30:379–387.
- Arundine M, Tymianski M. Molecular mechanisms of calcium-dependent neurodegeneration in excitotoxicity. *Cell Calcium*. 2003;34:325–337.
- 254. Abdel-Hamid KM, Tymianski M. Mechanisms and effects of intracellular calcium buffering on neuronal survival in organotypic hippocampal cultures exposed to anoxia/ aglycemia or to excitotoxins. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*. 1997;17:3538–3553.
- 255. Beal MF, Hyman BT, Koroshetz W. Do defects in mitochondrial energy metabolism underlie the pathology of neurodegenerative diseases? *Trends in Neurosciences*. 1993;16:125–131.
- 256. Beal MF. Mechanisms of excitotoxicity in neurologic diseases. *Federation of American Societies for Experimental Biology Journal*. 1992;6:3338–3344.
- 257. Carboni S, Melis F, Pani L, Hadjiconstantinou M, Rossetti ZL. The non-competitive NMDA-receptor antagonist MK-801 prevents the massive release of glutamate and aspartate from rat striatum induced by 1-methyl-4-phenylpyridinium (MPP+). *Neuroscience Letters*. 1990;117:129–133.
- 258. Helton TD, Otsuka T, Lee MC, Mu Y, Ehlers MD. Pruning and loss of excitatory synapses by the parkin ubiquitin ligase. *Proceedings of the National Academy of Sciences of the United States of America*. 2008;105:19492–19497.
- 259. Calabresi P, Gubellini P, Picconi B, Centonze D, Pisani A, Bonsi P, Greengard P, Hipskind RA, Borrelli E, Bernardi G. Inhibition of mitochondrial complex II induces a long-term potentiation of NMDA-mediated synaptic excitation in the striatum requiring endogenous dopamine. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience.* 2001;21:5110–5120.
- Trotti D, Rolfs A, Danbolt NC, Brown RH, Jr., Hediger MA. SOD1 mutants linked to amyotrophic lateral sclerosis selectively inactivate a glial glutamate transporter. *Nature Neuroscience*. 1999;2:427–433.
- 261. Rothstein JD, Martin LJ, Kuncl RW. Decreased glutamate transport by the brain and spinal cord in amyotrophic lateral sclerosis. *The New England Journal of Medicine*. 1992;326:1464–1468.
- 262. Mattson MP, Cheng B, Davis D, Bryant K, Lieberburg I, Rydel RE. Beta-amyloid peptides destabilize calcium homeostasis and render human cortical neurons vulnerable to excitotoxicity. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*. 1992;12:376–389.
- 263. De Felice FG, Velasco PT, Lambert MP, Viola K, Fernandez SJ, Ferreira ST, Klein WL. Abeta oligomers induce neuronal oxidative stress through an N-methyl-D-aspartate receptor-dependent mechanism that is blocked by the Alzheimer drug memantine. *The Journal of Biological Chemistry*. 2007;282:11590–11601.
- 264. Akiyama H, Barger S, Barnum S, Bradt B, Bauer J, Cole GM, Cooper NR, et al. Inflammation and Alzheimer's disease. *Neurobiology of Aging*. 2000;21:383–421.
- Mattson MP. Apoptosis in neurodegenerative disorders. Nature Reviews. Molecular Cell Biology. 2000;1:120–129.
- 266. Mattson MP. Neuroprotective signaling and the aging brain: Take away my food and let me run. *Brain Research*. 2000;886:47–53.
- 267. Mattison JA, Roth GS, Beasley TM, Tilmont EM, Handy AM, Herbert RL, Longo DL, et al. Impact of caloric restriction on health and survival in rhesus monkeys from the NIA study. *Nature*. 2012;489:318–321.

- 268. Dubey A, Forster MJ, Lal H, Sohal RS. Effect of age and caloric intake on protein oxidation in different brain regions and on behavioral functions of the mouse. *Archives* of Biochemistry and Biophysics. 1996;333:189–197.
- Ingram DK, Weindruch R, Spangler EL, Freeman JR, Walford RL. Dietary restriction benefits learning and motor performance of aged mice. *Journal of Gerontology*. 1987;42:78–81.
- 270. Zhu H, Guo Q, Mattson MP. Dietary restriction protects hippocampal neurons against the death-promoting action of a presenilin-1 mutation. *Brain Research*. 1999;842:224–229.
- 271. Duan W, Mattson MP. Dietary restriction and 2-deoxyglucose administration improve behavioral outcome and reduce degeneration of dopaminergic neurons in models of Parkinson's disease. *Journal of Neuroscience Research*. 1999;57:195–206.
- Bruce-Keller AJ, Umberger G, McFall R, Mattson MP. Food restriction reduces brain damage and improves behavioral outcome following excitotoxic and metabolic insults. *Annals of Neurology*. 1999;45:8–15.
- 273. Mattson MP, Chan SL, Duan W. Modification of brain aging and neurodegenerative disorders by genes, diet, and behavior. *Physiological Reviews*. 2002;82:637–672.
- Tapia-Arancibia L, Aliaga E, Silhol M, Arancibia S. New insights into brain BDNF function in normal aging and Alzheimer disease. *Brain Research Reviews*. 2008;59:201–220.
- 275. Mattson MP, Lovell MA, Furukawa K, Markesbery WR. Neurotrophic factors attenuate glutamate-induced accumulation of peroxides, elevation of intracellular Ca²⁺ concentration, and neurotoxicity and increase antioxidant enzyme activities in hippocampal neurons. *Journal of Neurochemistry*. 1995;65:1740–1751.
- 276. Schabitz WR, Sommer C, Zoder W, Kiessling M, Schwaninger M, Schwab S. Intravenous brain-derived neurotrophic factor reduces infarct size and counterregulates Bax and Bcl-2 expression after temporary focal cerebral ischemia. *Stroke: A Journal of Cerebral Circulation*. 2000;31:2212–2217.
- 277. Mattson MP, Duan W, Lee J, Guo Z. Suppression of brain aging and neurodegenerative disorders by dietary restriction and environmental enrichment: Molecular mechanisms. *Mechanisms of Ageing and Development*. 2001;122:757–778.
- Lowenstein DH, Chan PH, Miles MF. The stress protein response in cultured neurons: Characterization and evidence for a protective role in excitotoxicity. *Neuron*. 1991;7:1053–1060.
- 279. Kakimura J, Kitamura Y, Takata K, Umeki M, Suzuki S, Shibagaki K, Taniguchi T, et al. Microglial activation and amyloid-beta clearance induced by exogenous heat-shock proteins. *Federation of American Societies for Experimental Biology Journal*. 2002;16:601–603.
- 280. Guo Z, Ersoz A, Butterfield DA, Mattson MP. Beneficial effects of dietary restriction on cerebral cortical synaptic terminals: Preservation of glucose and glutamate transport and mitochondrial function after exposure to amyloid beta-peptide, iron, and 3-nitropropionic acid. *Journal of Neurochemistry*. 2000;75:314–320.
- 281. Li Y, Liu L, Tollefsbol TO. Glucose restriction can extend normal cell lifespan and impair precancerous cell growth through epigenetic control of hTERT and p16 expression. *Federation of American Societies for Experimental Biology Journal*. 2010;24:1442–1453.
- 282. Cohen HY, Miller C, Bitterman KJ, Wall NR, Hekking B, Kessler B, Howitz KT, Gorospe M, de Cabo R, Sinclair DA. Calorie restriction promotes mammalian cell survival by inducing the SIRT1 deacetylase. *Science*. 2004;305:390–392.
- 283. Guarente L, Picard F. Calorie restriction-the SIR2 connection. Cell. 2005;120:473-482.
- Motta MC, Divecha N, Lemieux M, Kamel C, Chen D, Gu W, Bultsma Y, McBurney M, Guarente L. Mammalian SIRT1 represses forkhead transcription factors. *Cell*. 2004;116:551–563.

- 285. Raghavan A, Shah ZA. Sirtuins in neurodegenerative diseases: A biological-chemical perspective. *Neuro-Degenerative Diseases*. 2012;9:1–10.
- 286. Singh M, Arseneault M, Sanderson T, Murthy V, Ramassamy C. Challenges for research on polyphenols from foods in Alzheimer's disease: Bioavailability, metabolism, and cellular and molecular mechanisms. *Journal of Agricultural and Food Chemistry*. 2008;56:4855–4873.
- 287. Robb EL, Winkelmolen L, Visanji N, Brotchie J, Stuart JA. Dietary resveratrol administration increases MnSOD expression and activity in mouse brain. *Biochemical and Biophysical Research Communications*. 2008;372:254–259.
- 288. Putics A, Vegh EM, Csermely P, Soti C. Resveratrol induces the heat-shock response and protects human cells from severe heat stress. *Antioxidants & Redox Signaling*. 2008;10:65–75.
- 289. Ruskin DN, Ross JL, Kawamura M, Jr., Ruiz TL, Geiger JD, Masino SA. A ketogenic diet delays weight loss and does not impair working memory or motor function in the R6/2 1J mouse model of Huntington's disease. *Physiology & Behavior*. 2011;103:501–507.
- 290. Willett WC, Sacks F, Trichopoulou A, Drescher G, Ferro-Luzzi A, Helsing E, Trichopoulos D. Mediterranean diet pyramid: A cultural model for healthy eating. *The American Journal of Clinical Nutrition*. 1995;61:1402S–1406S.
- 291. Solfrizzi V, Panza F, Frisardi V, Seripa D, Logroscino G, Imbimbo BP, Pilotto A. Diet and Alzheimer's disease risk factors or prevention: The current evidence. *Expert Review of Neurotherapeutics*. 2011;11:677–708.
- 292. Frisardi V, Panza F, Seripa D, Imbimbo BP, Vendemiale G, Pilotto A, Solfrizzi V. Nutraceutical properties of mediterranean diet and cognitive decline: Possible underlying mechanisms. *Journal of Alzheimer's Disease*. 2010;22:715–740.
- 293. Garry PJ, Wayne SJ, Vellas B. The New Mexico aging process study (1979–2003). A longitudinal study of nutrition, health and aging. *The Journal of Nutrition, Health & Aging*. 2007;11:125–130.
- Morris MC, Evans DA, Tangney CC, Bienias JL, Wilson RS. Associations of vegetable and fruit consumption with age-related cognitive change. *Neurology*. 2006;67:1370–1376.
- 295. Spencer JP. Flavonoids and brain health: Multiple effects underpinned by common mechanisms. *Genes & Nutrition*. 2009;4:243–250.
- 296. Wu X, Beecher GR, Holden JM, Haytowitz DB, Gebhardt SE, Prior RL. Concentrations of anthocyanins in common foods in the United States and estimation of normal consumption. *Journal of Agricultural and Food Chemistry*. 2006;54:4069–4075.
- 297. Dreiseitel A, Korte G, Schreier P, Oehme A, Locher S, Hajak G, Sand PG. sPhospholipase A(2) is inhibited by anthocyanidins. *Journal of Neural Transmission*. 2009;116:1071–1077.
- Weinreb O, Amit T, Mandel S, Youdim MB. Neuroprotective molecular mechanisms of (-)-epigallocatechin-3-gallate: A reflective outcome of its antioxidant, iron chelating and neuritogenic properties. *Genes & Nutrition*. 2009;4:283–296.
- Porat Y, Abramowitz A, Gazit E. Inhibition of amyloid fibril formation by polyphenols: Structural similarity and aromatic interactions as a common inhibition mechanism. *Chemical Biology & Drug Design*. 2006;67:27–37.
- 300. Williams CM, El Mohsen MA, Vauzour D, Rendeiro C, Butler LT, Ellis JA, Whiteman M, Spencer JP. Blueberry-induced changes in spatial working memory correlate with changes in hippocampal CREB phosphorylation and brain-derived neurotrophic factor (BDNF) levels. *Free Radical Biology & Medicine*. 2008;45:295–305.
- 301. Spencer JP, Vauzour D, Rendeiro C. Flavonoids and cognition: The molecular mechanisms underlying their behavioural effects. *Archives of Biochemistry and Biophysics*. 2009;492:1–9.

- 302. Zhang W, Li P, Hu X, Zhang F, Chen J, Gao Y. Omega-3 polyunsaturated fatty acids in the brain: Metabolism and neuroprotection. *Frontiers in Bioscience: A Journal and Virtual Library*. 2011;16:2653–2670.
- Moriguchi T, Greiner RS, Salem N, Jr. Behavioral deficits associated with dietary induction of decreased brain docosahexaenoic acid concentration. *Journal of Neurochemistry*. 2000;75:2563–2573.
- 304. Green KN, Martinez-Coria H, Khashwji H, Hall EB, Yurko-Mauro KA, Ellis L, LaFerla FM. Dietary docosahexaenoic acid and docosapentaenoic acid ameliorate amyloid-beta and tau pathology via a mechanism involving presenilin 1 levels. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*. 2007;27:4385–4395.
- 305. Florent S, Malaplate-Armand C, Youssef I, Kriem B, Koziel V, Escanye MC, Fifre A, et al. Docosahexaenoic acid prevents neuronal apoptosis induced by soluble amyloid-beta oligomers. *Journal of Neurochemistry*. 2006;96:385–395.
- Ricciarelli R, Argellati F, Pronzato MA, Domenicotti C. Vitamin E and neurodegenerative diseases. *Molecular Aspects of Medicine*. 2007;28:591–606.
- 307. Kumar GP, Khanum F. Neuroprotective potential of phytochemicals. *Pharmacognosy Reviews*. 2012;6:81–90.
- Kramer AF, Hahn S, Cohen NJ, Banich MT, McAuley E, Harrison CR, Chason J, et al. Ageing, fitness and neurocognitive function. *Nature*. 1999;400:418–419.
- 309. van Praag H, Christie BR, Sejnowski TJ, Gage FH. Running enhances neurogenesis, learning, and long-term potentiation in mice. *Proceedings of the National Academy of Sciences of the United States of America*. 1999;96:13427–13431.
- Suominen-Troyer S, Davis KJ, Ismail AH, Salvendy G. Impact of physical fitness on strategy development in decision-making tasks. *Perceptual and Motor Skills*. 1986;62:71–77.
- 311. Bohannon RW. Physical rehabilitation in neurologic diseases. *Current Opinion in Neurology*. 1993;6:765–772.
- Neeper SA, Gomez-Pinilla F, Choi J, Cotman CW. Physical activity increases mRNA for brain-derived neurotrophic factor and nerve growth factor in rat brain. *Brain Research*. 1996;726:49–56.
- 313. Vaynman S, Ying Z, Gomez-Pinilla F. Hippocampal BDNF mediates the efficacy of exercise on synaptic plasticity and cognition. *The European Journal of Neuroscience*. 2004;20:2580–2590.
- Cotman CW, Berchtold NC. Exercise: A behavioral intervention to enhance brain health and plasticity. *Trends in Neurosciences*. 2002;25:295–301.
- Gomez-Pinilla F, Vaynman S, Ying Z. Brain-derived neurotrophic factor functions as a metabotrophin to mediate the effects of exercise on cognition. *The European Journal of Neuroscience*. 2008;28:2278–2287.
- Vaynman S, Ying Z, Wu A, Gomez-Pinilla F. Coupling energy metabolism with a mechanism to support brain-derived neurotrophic factor-mediated synaptic plasticity. *Neuroscience*. 2006;139:1221–1234.
- 317. Ding Q, Vaynman S, Souda P, Whitelegge JP, Gomez-Pinilla F. Exercise affects energy metabolism and neural plasticity-related proteins in the hippocampus as revealed by proteomic analysis. *The European Journal of Neuroscience*. 2006;24:1265–1276.
- Ding Q, Vaynman S, Akhavan M, Ying Z, Gomez-Pinilla F. Insulin-like growth factor I interfaces with brain-derived neurotrophic factor-mediated synaptic plasticity to modulate aspects of exercise-induced cognitive function. *Neuroscience*. 2006;140:823–833.
- 319. Trejo JL, Carro E, Torres-Aleman I. Circulating insulin-like growth factor I mediates exercise-induced increases in the number of new neurons in the adult hippocampus. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*. 2001;21:1628–1634.

- 320. Tsankova NM, Berton O, Renthal W, Kumar A, Neve RL, Nestler EJ. Sustained hippocampal chromatin regulation in a mouse model of depression and antidepressant action. *Nature Neuroscience*. 2006;9:519–525.
- 321. ConnVS. Depressive symptom outcomes of physical activity interventions: Meta-analysis findings. Annals of Behavioral Medicine: A Publication of the Society of Behavioral Medicine. 2010;39:128–138.
- 322. van Praag H. Exercise and the brain: Something to chew on. *Trends in Neurosciences*. 2009;32:283–290.
- 323. Molteni R, Wu A, Vaynman S, Ying Z, Barnard RJ, Gomez-Pinilla F. Exercise reverses the harmful effects of consumption of a high-fat diet on synaptic and behavioral plasticity associated to the action of brain-derived neurotrophic factor. *Neuroscience*. 2004;123:429–440.
- 324. Nelson ME, Rejeski WJ, Blair SN, Duncan PW, Judge JO, King AC, Macera CA, Castaneda-Sceppa C. Physical activity and public health in older adults: Recommendation from the American College of Sports Medicine and the American Heart Association. *Medicine and Science in Sports and Exercise*. 2007;39:1435–1445.

8 Hunger and Satiety Signaling

Denovan P. Begg and Stephen C. Woods

CONTENTS

8.1	Introduction		391		
	8.1.1	Why We Eat	391		
8.2	Neurobiological Controls over Food Intake				
8.3	Homeostatic versus Nonhomeostatic Influences over Food Intake				
	8.3.1	Meals	394		
	8.3.2	Signals That Contribute to Meal Termination (Satiation Signals)	394		
	8.3.3	Signals Related to Adiposity	397		
	8.3.4	Interactions of Satiation and Obesity Signals	398		
8.4	Summary and Recommendations		399		
	8.4.1	Summary	399		
	8.4.2	Recommendation	400		
Ackr	nowledg	gment	400		
Refe	rences.		401		

8.1 INTRODUCTION

Food intake is a complex behavior that is nonetheless quite familiar since most of us engage in it several times a day. The complexity derives from the flexibility inherent in most of its parameters and the integrated activity of numerous and diverse influences. This short chapter reviews what is known of the factors that influence food intake in humans and other general omnivores, that is, animals (mammals in this instance) that consume, and are able to derive usable energy from, almost any type of food or nutrient.

8.1.1 WHY WE EAT

Eating is the behavior that takes nutrients into the body from the environment. Nutrients include both micronutrients (minerals and vitamins) and energy-containing macronutrients (carbohydrates, lipids, and proteins). It is axiomatic that under normal circumstances eating is the primary source of energy for the body and that if an individual is going to be weight stable energy intake must equal the energy expended during exercise and metabolism over an extended period of time. Taking in more energy than is expended necessarily leads to weight gain and vice versa. The other important factor in the calculus of eating and energy balance is that unlike the case for other commodities such as water, consumed but not yet used energy can be stored in the body, mainly in adipose depots (fat tissue). Eating less energy than is expended, at least in the short term, results in more of this previously stored energy being used, and when this occurs chronically body weight is lost primarily from body fat. Taking in more energy than is expended builds up the adipose tissue reservoir, and this is presumably what is happening in the contemporary *epidemic of obesity*.

8.2 NEUROBIOLOGICAL CONTROLS OVER FOOD INTAKE

Eating is a behavior and as such is controlled and coordinated by the brain; an excellent review of this control is found in the work by Langhans and Geary.¹ Historically, it was thought that there are two critical brain *centers* that control the intake of food. The lateral hypothalamic area (LHA) was called the eating center because when it was electrically or pharmacologically stimulated in experiments, animals started eating. Chronic stimulation of the LHA resulted in animals becoming obese, that is, having excess stored fat. Experimental destruction of the LHA resulted in animals that ate very little or not at all and that often died without food being infused directly into their stomachs. In contrast, the ventromedial hypothalamic nucleus (VMN) was called the satiety center because when it was stimulated animals stopped eating and when it was destroyed animals overate and became obese. Both the LHA and the VMN receive sensory input related to energy needs and reserves, and the two were considered to work in a reciprocally coordinated manner to determine eating.

It is now recognized that this model is overly simplistic and that diverse circuits interconnecting numerous brain areas are involved.^{2–5} Relying on receptors located throughout the body as well as within the brain itself, numerous brain circuits are made continuously aware of relevant information regarding energy status throughout the body. This includes knowing about immediately available energy (e.g., blood glucose), energy being consumed or that was recently consumed and awaiting entry into the blood from the intestines, energy stored in adipose tissue, energy stored in liver and other tissues, and so on. This information is integrated with knowledge of current energetic needs or anticipated needs, environmental factors such as where and when food might be available, the social situation (predators, offspring, conspecifics), memory of past experiences, hedonic factors, and many other factors as well (see the reviews by Seeley and Woods⁶ and Levin et al.⁷).

Based on these myriad inputs, the brain is able to make informed decisions about when to eat, what to eat, how much to eat, and so on and to integrate food seeking and consumption with other behavioral and metabolic priorities. Another fundamental mandate of the brain is to ensure adequate circulating energy (i.e., blood glucose and fatty acids) for immediate tissue needs.^{6–8} Eating is consequently well coordinated with the control of plasma glucose, fatty acids, and other nutrients. Under most circumstances, the levels of energy-rich fuels in the blood are relatively constant, with extraction from the blood by active tissues matched well to secretion into the blood by liver and fat cells, and this is true at rest as well as during exercise.

Meals are the major exception to this rule as they are times of elevated plasma glucose and other nutrients. An important mandate of the brain is to restrain meal size so as to circumvent excessively large increments of plasma nutrients while none-theless allowing adequate nutrients to be consumed.^{9–11} This is accomplished in large part by knowing when meals are likely to occur and making appropriate anticipatory responses that keep plasma fuels as low as possible during and soon after eating. This requires coordinating ongoing information about when food is likely to be available, calories being consumed (via satiation signals), the levels of fuels already in the plasma (via direct sensing by specialized cells in the brain and elsewhere), and the amount of energy present in various storage depots (via adiposity signals). In sum, a key function of the brain is to coordinate diverse processes that allow optimal circulating and stored levels of energy-rich nutrients.

8.3 HOMEOSTATIC VERSUS NONHOMEOSTATIC INFLUENCES OVER FOOD INTAKE

Eating is often considered a homeostatic behavior, that is, a reflexive response to insufficient energy, either in the blood or in its ongoing usage by the brain. Glucose supply to the brain is a key variable since unlike most other tissues, nerve cells obligatorily derive energy mainly from glucose and, because very little energy can be stored within the brain, it requires a steady stream of adequate glucose input via the circulation. This need for continuous glucose was the basis of the popular glucostatic theory of eating,^{12–14} and it posited that when immediately available glucose is low the brain responds by initiating eating. The intended consequence of this low glucose-induced eating was an increase of glucose entering the blood from the intestines, and this in turn was thought to provide a signal to the brain to stop the ongoing meal. Although simple and appealing, the theory fell into disfavor because levels of glucose within the normal physiological range have little or no impact on eating behavior.^{15,16} It is true that if the blood glucose level or glucose utilization in the brain is lowered to extremely low levels eating is initiated, but this is considered an emergency response necessary to prevent coma or death.¹⁷ Thus, although eating in this situation of extreme immediate glucose need can certainly be considered homeostatic, it does not reflect normal eating.

Eating can perhaps better be considered a homeostatic behavior with respect to adiposity.¹⁸ The amount of fat in the body is generally thought to be regulated. When individuals lose weight due to dieting or insufficient available food, they tend to regain the weight to precisely their former level given sufficient time with adequate food availability. Likewise, when otherwise weight-stable individuals voluntarily overeat and gain weight in experiments, they tend to lose the excess weight and return to basal levels over time once the period of overeating is over.¹⁹ All of these data are consistent with the concept of a strict regulation of body weight (body fat) by the brain, and with the act of eating being recruited to assist in the regulation. An important point is that most of the energy available from any individual meal is not available for a considerable interval after the meal is ended, such that the time constant for considering the level of adiposity as a homeostatically regulated behavior is much longer than for other instances of homeostasis. In this light, eating can

be considered a behavior that is used in the homeostatic regulation of body weight, but only minor adjustments are possible to be made at any one time or meal.^{11,18} In fact, considerable evidence suggests that in humans matching food intake to energy expenditure often takes days or weeks to be accurate.^{20,21} In contrast to food intake being a homeostatic slave to the regulation of body weight, there is compelling evidence that most instances of food intake are influenced to a large extent by nonhomeostatic factors.¹⁰

8.3.1 MEALS

Meals are the units of food intake, and an important point is that the factors that determine when meals will start are different from those that determine when meals will end. As indicated in Section 8.3, single factors such as glucose availability or stomach contractions were historically thought to determine both meal onset and meal offset, but these single-factor theories have not withstood the test of time. With respect to when meals are taken, the best evidence is that meal initiation is based on habit, convenience, social situation, opportunities, time of day, and so on. Such a flexible system that is not tied to metabolic need allows optimal coordination of meals with other activities.¹⁰

In contrast to meal initiation, the amount of food that is eaten once a meal has begun is more directly linked with metabolism and overall energy homeostasis. During a meal, ingested food is continuously analyzed for its nutrient composition, and this begins even before it enters the mouth. The sight and smell of many foods initiate the secretion of hormones and gastrointestinal (GI) enzymes that will facilitate the processing of what is about to be eaten.²² For individuals who habitually eat at the same time each day, these same secretions are initiated as that time approaches. These secretions are collectively called cephalic responses since they are initiated by the brain around mealtime but prior to any ingested food actually being absorbed into the blood. Cephalic responses readily become associated via a classical conditioning process with stimuli that reliably predict the onset of meals, such as time of day or certain environments.^{9,22} This is a useful adaptation in that it enables the GI system to better prepare to digest and absorb nutrients efficiently while preventing prandial excursions of blood glucose from getting excessively high. If an individual is prevented from making premeal anticipatory cephalic responses, and nonetheless eats the same-sized meal as he or she otherwise would, blood glucose rises into the diabetic range during and soon after the meal.9

8.3.2 SIGNALS THAT CONTRIBUTE TO MEAL TERMINATION (SATIATION SIGNALS)

Satiation is the phenomenon of *fullness*, and in experiments its occurrence heralds the end of a meal and hence determines meal size. Satiety, on the other hand, refers to factors that prolong the start of a second meal after a first one has ended in freely feeding individuals. Satiety is generally considered to result from a signal that is generated in the processing of a first meal and that has a long half-life; as its levels wane below some threshold, the urge to eat is initiated. However, although the concept of satiety may apply to freely feeding individuals in an environment devoid of temporal and other possible food cues, other, nonhomeostatic, factors, as discussed earlier in Section 8.3, are more likely to determine when meals begin. Although little is known about satiety, considerable information is now available about satiation.

The GI system begins at the mouth and ends at the anus. Each segment along the way is characterized by numerous specialized cell types. These include muscles, secretory cells, receptor cells, and many more. Taste receptors are clustered in groups (taste buds) on the tongue as well as being scattered over much of the oropharyngeal region, and they have recently been identified on intestinal cells as well, implying that specific tastants in meals are being detected throughout the GI tract and the progress of individual nutrients can thus be tracked and appropriate hormones and enzymes secreted at the right time, customizing the digestive process to what is being eaten.^{23,24}

In the mid-twentieth century, a paradigm shift occurred that moved theorizing about the controls of food intake from the simplistic single-factor theories such as glucose utilization to a neuropeptide-centric model of energy balance control. In 1973, it was discovered that the duodenal peptide cholecystokinin (CCK) decreases food intake when administered to animals.^{25,26} CCK is produced predominantly in I-cells in the initial part of the small intestine after the stomach, the duodenum, and it is released prandially in response to specific nutrient signals in consumed foods such as fatty acids and amino acids.²⁷ Prior to this breakthrough work on satiation, CCK had been characterized for its role in other digestive processes, including gastric emptying, stimulation of the gall bladder and bile flow and pancreatic secretions.^{28–30} The hypophagia produced by CCK administration was considered to be part of a negative feedback loop, such that when partially digested nutrients are sensed in the duodenum CCK is released to aid digestion and reduce further food intake. When the flow of newly consumed nutrients into the duodenum stops, CCK secretion is no longer stimulated.

When CCK is released from I-cells in the wall of the duodenum, some of it enters the blood to produce hormonal effects in other organs. In addition, some of the secreted CCK molecules act locally on CCK receptors (CCK-1R) located on sensory endings of fibers of vagal afferent nerves that pass to the hindbrain where they interact with other signals to produce satiation. If CCK activity during a meal is prevented by administering antagonists to CCK-1R, animals and humans eat larger than normal meals.^{31,32} This implies that our own endogenous CCK normally limits meal size and is an important component of satiation response.

Although there has been considerable interest in utilizing CCK-1R agonists to reduce food intake in overweight or obese humans, experiments to date have not been efficacious.³³ This is due in part to the fact that whereas the administration of CCK at the onset of each meal always reduces the amount eaten in that meal, individuals compensate by eating more often, that is, they increase meal number, leading to no net reduction in daily energy intake.³⁴ More recently, several longer acting formulations of CCK agonists have been developed and may prove therapeutically beneficial for weight loss.³³ Despite the disappointment of CCK as a weight loss agent, the demonstration that a GI peptide secreted during meals can impact food intake changed the way the neurobiological controls of food intake have been examined over the past 40 years.

Numerous other peptides secreted by the GI tract during meals have since been reported to act as satiation signals.²⁷ The list is long and growing, and it includes peptide YY (PYY), pancreatic polypeptide, glucagon-like peptide-1 (GLP-1), glucagon, amylin, somatostatin, uroguanylin, and others. It is important to note that each of these gut peptides is secreted in response to specific properties of the food being digested, and each has unique functions in causing appropriate enzymes or other molecules to be secreted into the intestinal lumen or else to speed up or slow down the passage of the partially digested food (or chyme) along the GI tract. What these peptides share in common is an ability to limit meal size when they are administered just prior to the onset of a meal. One gut peptide, ghrelin, is synthesized primarily in the stomach and has a functionally opposite action on food intake. When administered to humans or animals, ghrelin initiates meals and causes overeating.^{35,36} If the time that a meal will occur is well known, individuals secrete most of these peptides cephalically, beginning an hour or so before the meal begins^{9,22,37–40} (see Figure 8.1), presumably to jump-start the digestive system and consequently allow more efficient digestion without the risk of hyperglycemia.

To summarize, food intake (i.e., meals) tends to be initiated at times that are convenient or habitual. At every step of the way along the GI tract, the food is analyzed by sensory cells. Depending on the nutrient content of the food, a customized cocktail of peptides is secreted to facilitate digestion and nutrient absorption. These same peptides generate hormonal and neural signals that are relayed to the brain, enabling coordination among the various organs involved in digestion. The signals also combine to create the sensation/perception of fullness and satiation.



FIGURE 8.1 Different parameters and signals in anticipation of, during, and following a meal: cephalically induced hormone and neuropeptide release occurs to prepare the body for an influx of nutrients and to prevent potentially damaging hyperglycemia. During a meal, many factors may act as satiation signals. When the meal is completed, many of the same peptides help complete the digestion process.

8.3.3 SIGNALS RELATED TO ADIPOSITY

Food intake is also influenced by the amount of fat stored in the body, or adiposity. Several hormones are secreted into the blood in proportion to adiposity, and they circulate to the brain and interact with receptors in many brain areas, especially in the hypothalamus. These hormones are called adiposity signals because they provide continuous feedback to the brain about how much fat is in storage, and this information can then be integrated with other information related to metabolism and energy balance. Insulin, a hormone secreted by the endocrine pancreas that is best known for its blood glucose-lowering action, was the first identified adiposity signal.^{4,41} Insulin is secreted in direct proportion to adiposity, and it acts in the hypothalamus to cause a net catabolic action, that is, it reduces food intake and lowers body weight. Leptin, a hormone secreted by fat cells, is also secreted in direct proportion to body fat and, like insulin, is transported from the blood into the hypothalamus and other brain areas where it acts on leptin receptors on neurons influencing energy homeostasis. Also like insulin, when leptin is administered into the brain animals eat less food and lose weight. There are many reviews of these phenomena.^{4,42–44} The point is that an experimental or therapeutic increase of either the insulin signal or the leptin signal in the brain conveys the message that the body is fatter than it actually is, and the brain responds by reducing food intake. Conversely, when either the insulin signal or the leptin signal is reduced within the brain, overeating and body weight gain occur. A number of other compounds have been proposed to be functional adiposity signals, including amylin,⁴⁵ which is cosecreted with insulin from the pancreas, and several adipose tissue-derived cytokines including interleukin-6 and tumor necrosis factor- α . However, most research on adiposity signals has been focused on leptin and insulin.

Adiposity signals are transported across the blood-brain barrier from the blood into the brain by a receptor-mediated, saturable pathway in brain capillary endothelial cells.^{46,47} Within the brain, adiposity signals act on their respective receptors located in several areas, especially within the arcuate nucleus (ARC) of the hypothalamus, a region that receives and integrates diverse information pertinent to energy balance; that is, in addition to having receptors for insulin and leptin, ARC neurons are sensitive to individual nutrients (such as glucose and some fatty acids and amino acids), to satiation signals, and to inputs from numerous brain areas involved with hedonics, learning, biological rhythms, and others.^{48,49} The ARC contains two major types of neurons that integrate all of these inputs and project to other hypothalamic nuclei as well as to other brain areas (Figure 8.2). The neurons located predominantly in the ventromedial portion of the ARC synthesize and secrete two neuropeptides, agouti-related protein (AgRP) and neuropeptide Y (NPY).443,50 These neurons primarily project to the hypothalamic paraventricular nuclei (PVNs) and the lateral hypothalamus (LH). Electrical or pharmacological stimulation of NPY/ AgRP neurons produces an anabolic response that includes a rapid initiation of food intake and reduced energy expenditure. Animals consume excessive food (i.e., are hyperphagic) and become obese when AgRP or NPY is administered chronically. The adiposity signals insulin and leptin have an inhibitory action on these neurons, reducing food intake and causing weight loss when administered chronically.



FIGURE 8.2 Neuropeptide Y (NPY)/agouti-related protein (AgRP) neurons in the arcuate nucleus of the hypothalamus (ARC) project to the paraventricular nucleus of the hypothalamus (PVN): NPY acts on Y1/Y5 receptors in PVN to stimulate food intake; simultaneously, AgRP acts to inhibit MC3/MC4 receptor signaling to further increase food intake. Proopiomelanocortin (POMC) neurons project to MC3/MC4 receptors in the lateral hypothalamic area (LHA) and the PVN, releasing α -melanocyte-stimulating hormone to inhibit food intake. The depicted unilateral view occurs on both sides of the 3rd ventricle (V3). DMH, dorsomedial hypothalamus; VMH, ventromedial hypothalamus.

The lateral region of the ARC (see Figure 8.2) contains neurons that synthesize two different neuropeptides, proopiomelanocortin (POMC) and cocaine- and amphetamineregulated transcript (CART). POMC is a large prohormone that is the precursor molecule for several active neuropeptides, the exact neuropeptides generated in any cell depending on the cleavage enzymes present. In the ARC, POMC is cleaved into the active neuropeptide α -melanocyte-stimulating hormone (α MSH), which is secreted from the axon terminals of the ARC POMC neurons and acts on melanocortin receptors (MC3 and MC4). The axons of POMC/CART neurons project to the PVN and LH, where they elicit their catabolic effects by inhibiting food intake and increasing energy expenditure.^{5,43} Insulin and leptin act on POMC/CART neurons to release α MSH, and intact melanocortin receptor signaling is required for their hypophagic effects. Several excellent reviews on this ARC integrative system that influences energy homeostasis are available.^{5,51,52}

8.3.4 INTERACTIONS OF SATIATION AND OBESITY SIGNALS

Given all of these different kinds of controllers over energy balance and influences on food intake, it is important to consider how they interact. As discussed earlier in Section 8.3.1, eating can best be considered in terms of meals, with daily food intake being based on both meal frequency and meal size. Many types of experiments have revealed how these meal parameters can interact. When constraints are put on meal size such that only small amounts can be consumed at one time, animals eat more meals per day and tend to keep their body weights relatively constant.³⁴ Analogously, when the number of times a day an individual can initiate a meal is limited, meal size increases and weight remains stable.⁵³ This flexibility of meal patterning in the defense of body weight implicates a strong influence of adiposity signals over meals and total food intake.

As discussed in detail earlier in Section 8.3.2, satiation signals such as CCK are generated during meals in response to the quality and quantity of nutrients being consumed. As the various satiation signals accumulate, they eventually cause the meal to stop.¹⁰ The sensitivity of the brain to satiation signals is based in part on adiposity signals, and their levels in turn are a function of how fat or thin the individual is. If an otherwise weight-stable individual goes on a diet (or does not have sufficient food to maintain weight), the levels of insulin and leptin that are secreted decrease in proportion to the loss of body fat. The lower insulin and leptin signal in turn reduces sensitivity to satiation signals. As a result, more food is eaten during a meal before satiation occurs, that is, larger meals are eaten, and this will continue until the weight that was lost is regained. Conversely, when animals or people are experimentally overfed and gain weight, the insulin and leptin signals are increased and there is increased sensitivity to satiation signals. Smaller meals are eaten and weight is lost. Relevant experiments can also be done in other ways. For example, when a normalweight animal is administered a very small amount of insulin⁵⁴ or leptin⁵⁵ into the brain, the ability of CCK to reduce meal size is greatly enhanced. This influence over meal size by satiation signals is one reason why weight tends to remain stable in individuals over long periods and also a reason why body weight often rebounds when one's enthusiasm for a diet regimen wanes.

8.4 SUMMARY AND RECOMMENDATIONS

8.4.1 SUMMARY

Numerous questions have been considered with respect to the consumption of food, including the timing, content, and size of individual meals. Although food intake is often considered a homeostatic behavior, in actuality it is influenced by a combination of both homeostatic and nonhomeostatic factors; thus, evidence does not support the concept that food intake is a reflexive, homeostatic-type response to an acute shortage of energy. Alterations of food intake over long intervals are consistent with a homeostatic control of body fat, with adiposity signals such as insulin and leptin influencing food intake based on energy levels stored in adipocytes. Short-term feedback is provided during meals that is based on nutrient content and is signaled to the brain by GI peptides including CCK, PYY, and GLP-1. These are considered satiation signals because they are released in response to the ingestion of food and act to reduce meal size.

The times at which we eat are largely determined by habit, convenience, or opportunity rather than a depletion of stored energy. A coordinated brain-initiated cephalic secretion of numerous hormones occurs before anticipated meals and primes the digestive system to better prepare for the upcoming caloric load. These cephalic responses are easily entrained to routine feeding times. Superimposed on these metabolic factors are diverse interacting nonhomeostatic factors, including stress, learning, palatability, social influences, and many others.

As discussed earlier in Section 8.3.4, satiation and adiposity signals interact to determine meal size, providing compelling evidence of a homeostatic influence over body weight/adiposity. At the same time, such control makes eating seem automatic or robotic, which, of course, it is not. The reason is that homeostatic influences should be considered as providing a background tone, with other factors intervening and/or overriding much of the time. We are all familiar with feeling full or satiated after a particularly large meal but nonetheless eating more when an appealing dessert is presented. Hedonics, experience with certain foods, opportunity, social situation, habits, stress, and emotions, as well as diverse other factors, can all impact the amount eaten in any given meal.⁵⁶ There is also evidence that the ability of satiation signals to influence meal size can be changed by learning and experience.¹⁰ The plethora of hedonically pleasing and calorically rich foods that is constantly available in many societies is often considered a major causal factor in the general increase of body weight known as the obesity epidemic. The important point is that despite the tightly integrated homeostatic factors that control food intake and body weight, these controls are easily trumped by nonhomeostatic influences. Chronic intervention by nonhomeostatic factors can lead to increased body weight and associated detrimental health outcomes, including diabetes and cardiovascular disease.

8.4.2 RECOMMENDATION

Most people are unaware of how powerfully nonhomeostatic cues can affect eating behavior. Based on the habitual nature of eating, it therefore might be tempting to recommend that dietary interventions should be approached using behavioral therapy. The primary goal for such therapy would be to learn, be conscious of, and permanently change dietary behavior to a pattern consistent with a healthier lifestyle. However, such dietary changes are generally only effective in the short term, with most weight being regained after a number of months or years.^{57,58} Approved pharmacological treatments, including appetite suppressants and lipid absorption inhibitors, are generally effective in producing long-term weight loss so long as treatments are continued. However, current formulations generally produce only modest weight loss. Surgical interventions comprise an attractive alternative for some obese individuals in that besides causing sustained weight loss over many years they also improve glucose homeostasis.^{59,60} Successful bariatric surgery often leads to substantial reductions in food intake and changes in meal patterning and food preferences, including lowered dietary fat intake following surgery.^{61,62} Given the ineffectiveness of alternative treatments, bariatric surgeries provide a viable treatment option for some obese individuals.

ACKNOWLEDGMENT

This work was supported by NIH DK017844 to Stephen C. Woods and by NHMRC Early Career Fellowship 1013264 to Denovan P. Begg.

REFERENCES

- 1. Langhans W, Geary N. Frontiers in Eating and Weight Regulation. Basel: Karger; 2010.
- Elmquist JK, Coppari R, Balthasar N, Ichinose M, Lowell BB. Identifying hypothalamic pathways controlling food intake, body weight, and glucose homeostasis. *J Comp Neurol.* 2005;493:63–71.
- 3. Flier JS. Obesity wars: Molecular progress confronts an expanding epidemic. *Cell*. 2004;116:337–350.
- 4. Schwartz MW, Woods SC, Porte D, Jr., Seeley RJ, Baskin DG. Central nervous system control of food intake. *Nature*. 2000;404:661–671.
- Morton GJ, Cummings DE, Baskin DG, Barsh GS, Schwartz MW. Central nervous system control of food intake and body weight. *Nature*. 2006;443:289–295.
- Seeley RJ, Woods SC. Monitoring of stored and available fuel by the CNS: Implications for obesity. *Nat Rev Neurosci*. 2003;4:901–909.
- Levin BE, Magnan C, Dunn-Meynell A, Le Foll C. Metabolic sensing and the brain: Who, what, where, and how? *Endocrinology*. 2011;152:2552–2557.
- Sisley S, Sandoval D. Hypothalamic control of energy and glucose metabolism. *Rev Endocr Metab Disord*. 2011;12:219–233.
- 9. Woods SC. The eating paradox: How we tolerate food. Psychol Rev. 1991;98:488-505.
- de Kloet AD, Krause EG, Kim DH, Sakai RR, Seeley RJ, Woods SC. The effect of angiotensin-converting enzyme inhibition using *captopril* on energy balance and glucose homeostasis. *Endocrinology*. 2009;150:4114–4123.
- 11. Woods SC, Ramsay DS. Homeostasis: Beyond Curt Richter. Appetite. 2007;49:388–398.
- 12. Mayer J. Glucostatic mechanism of regulation of food intake. N Engl J Med. 1953;249:13–16.
- 13. Mayer J. Regulation of energy intake and the body weight: The glucostatic and lipostatic hypothesis. *Ann NY Acad Sci.* 1955;63:14–42.
- 14. Mayer J, Thomas DW. Regulation of food intake and obesity. Science. 1967;156:328-337.
- 15. Grossman SP. The role of glucose, insulin and glucagon in the regulation of food intake and body weight. *Neurosci Biobehav Rev.* 1986;10:295–315.
- 16. Langhans W. Metabolic and glucostatic control of feeding. *Proc Nutr Soc.* 1996;55:497–515.
- 17. Lotter EC, Woods SC. Injections of insulin and changes of body weight. *Physiol Behav*. 1977;18:293–297.
- 18. Woods SC, Ramsay DS. Food intake, metabolism and homeostasis. *Physiol Behav*. 2011;104:4–7.
- 19. Leibel RL. Molecular physiology of weight regulation in mice and humans. *Int J Obes* (*Lond*). 2008;32 Suppl 7:S98–S108.
- 20. Birch LL, Johnson SL, Andresen G, Peters JC, Schulte MC. The variability of young children's energy intake. *N Engl J Med.* 1991;324:232–235.
- de Castro JM. The control of eating behavior in free living humans. In: Stricker EM, Woods SC, eds. *Handbook of Neurobiology. Neurobiology of Food and Fluid Intake*, Vol. 14, No. 2. New York: Kluwer Academic/Plenum Publishers; 2004:467–502.
- Teff KL. How neural mediation of anticipatory and compensatory insulin release helps us tolerate food. *Physiol Behav.* 2011;103:44–50.
- Behrens M, Meyerhof W. Gustatory and extragustatory functions of mammalian taste receptors. *Physiol Behav*. 2011;105:4–13.
- 24. Treesukosol Y, Smith KR, Spector AC. The functional role of the T1R family of receptors in sweet taste and feeding. *Physiol Behav.* 2011;105:14–26.
- Gibbs J, Young RC, Smith GP. Cholecystokinin elicits satiety in rats with open gastric fistulas. *Nature*. 1973;245:323–325.

- Gibbs J, Young RC, Smith GP. Cholecystokinin decreases food intake in rats. J Comp Physiol Psychol. 1973;84:488–495.
- Steinert RE, Beglinger C. Nutrient sensing in the gut: Interactions between chemosensory cells, visceral afferents and the secretion of satiation peptides. *Physiol Behav*. 2011;105:62–70.
- Moran TH, Kinzig KP. Gastrointestinal satiety signals II. Cholecystokinin. Am J Physiol. 2004;286:G183–G188.
- 29. Dockray GJ. Cholecystokinin. Curr Opin Endocrinol Diabetes Obes. 2012;19:8–12.
- 30. Moran TH, Dailey MJ. Intestinal feedback signaling and satiety. *Physiol Behav*. 2011;105:77–81.
- Moran TH, Ameglio PJ, Schwartz GJ, McHugh PR. Blockade of type A, not type B, CCK receptors attenuates satiety actions of exogenous and endogenous CCK. *Am J Physiol*. 1992;262:R46–R50.
- 32. Matzinger D, Degen L, Drewe J, Meuli J, Duebendorfer R, Ruckstuhl N, D'Amato M, Rovati L, Beglinger C. The role of long chain fatty acids in regulating food intake and cholecystokinin release in humans. *Gut.* 2000;46:688–693.
- Moran TH, Dailey MJ. Minireview: Gut peptides: Targets for antiobesity drug development? *Endocrinology*. 2009;150:2526–2530.
- West DB, Fey D, Woods SC. Cholecystokinin persistently suppresses meal size but not food intake in free-feeding rats. *Am J Physiol*. 1984;246:R776–R787.
- Tschöp M, Smiley DL, Heiman ML. Ghrelin induces adiposity in rodents. *Nature*. 2000;407:908–913.
- Cummings DE. Ghrelin and the short- and long-term regulation of appetite and body weight. *Physiol Behav*. 2006;89:71–84.
- Teff KL, Mattes RD, Engelman K. Cephalic phase insulin release in normal weight males: Verification and reliability. *Am J Physiol*. 1991;261:E430–E436.
- Drazen DL, Vahl TP, D'Alessio DA, Seeley RJ, Woods SC. Effects of a fixed meal pattern on ghrelin secretion: Evidence for a learned response independent of nutrient status. *Endocrinology*. 2006;147:23–30.
- Drazen DL, Wortman MD, Seeley RJ, Woods SC. Neuropeptide Y prepares rats for scheduled feeding. Am J Physiol Regul Integr Comp Physiol. 2005;288:R1606–R1611.
- Vahl TP, Drazen DL, Seeley RJ, D'Alessio DA, Woods SC. Meal-anticipatory glucagonlike peptide-1 secretion in rats. *Endocrinology*. 2010;151:569–575.
- Woods SC, Lotter EC, McKay LD, Porte D, Jr. Chronic intracerebroventricular infusion of insulin reduces food intake and body weight of baboons. *Nature*. 1979;282:503–505.
- 42. Woods SC, Seeley RJ, Porte D, Jr., Schwartz MW. Signals that regulate food intake and energy homeostasis. *Science*. 1998;280:1378–1383.
- 43. Belgardt BF, Bruning JC. CNS leptin and insulin action in the control of energy homeostasis. *Ann NY Acad Sci.* 2010;1212:97–113.
- 44. Begg DP, Woods SC. The central insulin system and energy balance. *Handb Exp Pharmacol*. 2012;209:111–129.
- 45. Lutz TA. The interaction of amylin with other hormones in the control of eating. *Diabetes Obes Metab.* 2012;303 Suppl 4:E496–E503.
- 46. Woods SC, Seeley RJ, Baskin DG, Schwartz MW. Insulin and the blood-brain barrier. *Curr Pharm Des.* 2003;9:795–800.
- 47. Banks WA. Blood-brain barrier as a regulatory interface. Forum Nutr. 2010;63:102-110.
- 48. Biebermann H, Kuhnen P, Kleinau G, Krude H. The neuroendocrine circuitry controlled by POMC, MSH, and AGRP. *Handb Exp Pharmacol*. 2012;209:47–75.
- 49. Hill JW, Elmquist JK, Elias CF. Hypothalamic pathways linking energy balance and reproduction. *Am J Physiol Endocrinol Metab.* 2008;294:E827–E832.
- Woods SC, Figlewicz DP, Madden L, Porte J, D., Sipols AJ, Seeley RJ. NPY and food intake: Discrepancies in the model. *Regul Pept*. 1998;75–76:403–408.

- 51. Williams KW, Elmquist JK. From neuroanatomy to behavior: Central integration of peripheral signals regulating feeding behavior. *Nat Neurosci*. 2012;15:1350–1355.
- 52. Myers MG, Jr., Munzberg H, Leinninger GM, Leshan RL. The geometry of leptin action in the brain: More complicated than a simple arc. *Cell Metab.* 2009;9:117–123.
- 53. Collier G, Johnson DF. The paradox of satiation. Physiol Behav. 2004;82:149–153.
- 54. Riedy CA, Chavez M, Figlewicz DP, Woods SC. Central insulin enhances sensitivity to cholecystokinin. *Physiol Behav.* 1995;58:755–760.
- 55. Matson CA, Reid DF, Cannon TA, Ritter RC. Cholecystokinin and leptin act synergistically to reduce body weight. *Am J Physiol Regul Integr Comp Physiol*. 2000;278:R882–R890.
- 56. Wansink B, Sobal J. Mindless eating: The 200 daily food decisions we overlook. *Environ Behav.* 2007;39:106–123.
- 57. Sumithran P, Proietto J. The defence of body weight: A physiological basis for weight regain after weight loss. *Clin Sci (Lond)*. 2013;124:231–241.
- Curioni CC, Lourenco PM. Long-term weight loss after diet and exercise: A systematic review. *Int J Obes (Lond)*. 2005;29:1168–1174.
- 59. Bradley D, Magkos F, Klein S. Effects of bariatric surgery on glucose homeostasis and type 2 diabetes. *Gastroenterology*. 2012;143:897–912.
- Stefater MA, Wilson-Perez HE, Chambers AP, Sandoval DA, Seeley RJ. All bariatric surgeries are not created equal: Insights from mechanistic comparisons. *Endocr Rev.* 2012;33:595–622.
- 61. Miras AD, Jackson RN, Jackson SN, Goldstone AP, Olbers T, Hackenberg T, Spector AC, le Roux CW. Gastric bypass surgery for obesity decreases the reward value of a sweetfat stimulus as assessed in a progressive ratio task. *Am J Clin Nutr.* 2012;96:467–473.
- 62. Berthoud HR, Zheng H. Modulation of taste responsiveness and food preference by obesity and weight loss. *Physiol Behav*. 2012;107:527–532.

This page intentionally left blank

9 Summary and Recommendations

C. Murray Ardies

CONTENTS

9.1	Overview		405
	9.1.1	Diet and Chronic Disease	406
	9.1.2	Dietary Recommendations	411
	9.1.3	Behavioral Issues	414
	9.1.4	Exercise and Chronic Disease	417
	9.1.5	Exercise Recommendations	419
	9.1.6	Final Analysis	421
Ackr	nowledg	gment	423

9.1 OVERVIEW

Probably the most important concept that has emerged from each of these chapters is the pervasive role that inflammation-associated signaling plays in the etiology of diabetes, osteoporosis, atherosclerosis, cancer, and neurodegenerative diseases (NDs), chronic diseases that are responsible for the vast majority of the many hundreds of billions of dollars spent as annual direct and indirect costs for health care in the United States. The second is that the prevailing inadequate dietary and activity habits of Americans compromise cell functions and enhance proinflammatory signaling to such a degree that they are major contributors to the etiology of these chronic diseases. The detrimental effects of nutritionally insufficient diets, consuming too many calories, and participating in too little physical activity and the impacts of poor diet and activity habits on the progression of chronic diseases are profound. In essence, it is through our lifestyle of mindless ease and plenty that so many of us are in a constant, low-level proinflammatory state. The term *mindless* simply refers to the idea that most people are completely unaware of how detrimental their lifestyle really is to their own health, and even if they are aware they rarely change their behaviors. As a result, a large majority in the United States is highly susceptible to cellular and tissue dysfunctions from any cellular stressors regardless of their source and therefore at high risk for chronic disease.

On a positive note, a tremendous amount of information has been reviewed here to illustrate that many of these risks, risks that we have come to accept as normal, do not really have to exist. With appropriate changes in food choices to ensure a well-balanced and nutritionally adequate diet that also includes the consumption of a wide array of protective phytochemicals, and with the addition of vigorous physical activities to our daily lives, many adverse cellular stresses can be handled with ease and the proinflammatory environment dampened, along with reducing the risks for many chronic diseases. A proper diet will ensure that optimal amounts of nutrients are available so that normal cell functions are not compromised by nutrient inadequacies. Increasing consumption of protective phytochemicals and participating in vigorous activities produce transient changes in the cellular redox microenvironment, which then modifies the activities of various intracellular signaling pathways. These changes result in the enhanced expression of many different proteins that enhance cellular redox control, protect cells from oxidative damage, attenuate proinflammatory signaling, and enhance repair of damage.

On the basis of these excellent reviews of very complex biological processes, the major effects of diet and exercise on the etiology of chronic disease are summarized next and specific recommendations are made for altering lifestyles to optimize prevention.

9.1.1 DIET AND CHRONIC DISEASE

Because the vast majority of Americans consume a diet that is insufficient in one or more nutrients, the ability of cells to synthesize the wide array of essential proteins and enzymes that maintain metabolic and redox control, maintain damage-free DNA and RNA, repair and/or replace damaged proteins and cellular phospholipids, and perform any of the other cellular functions that are necessary for cell survival is certain to be less than optimal. Essentially, immediate-need metabolic activities that contribute to very-short-term cell survival (such as adenosine triphosphate [ATP] synthesis) take precedence over other cellular functions that are not required with such immediacy (such as DNA repair). These immediate-need functions, if they are met by depleting (or significantly diminishing) local nutrient *supplies*, can end up short-changing longer term functions for cell survival that also depend on an optimal supply of the same nutrients.

Cellular survival is dependent, in a large part, on the appropriate regulation of the cellular redox state. This is because the activities of the various signaling pathways that regulate the activation of transcription factors ultimately depend on the local redox microenvironment at different locations throughout the cell. Because a majority of Americans eat less than the estimated average requirements (EAR) for vitamin C (25%), vitamin E (60%), selenium (5%), iron (5%), zinc (8%), and copper (6%), nutrients that either function as antioxidants or are components of antioxidant and redox-control enzymes, antioxidant function and redox control are certain to be less than optimal for most. Dysregulation of antioxidant and redox control leads to a cascade of effects that can result in an enhanced synthesis of a variety of signaling molecules that lead to proinflammatory effects. Although small alterations in the microenvironment at the cellular level due to dietary insufficiencies are rarely if ever fatal to the cell (and they certainly are not noticeable to the individual), they will lead to enhanced inflammatory risks. Production of reactive oxygen species (ROSs) is a common by-product of normal metabolism, and with reduced antioxidant and redox control a more oxidizing microenvironment within the cell is the result. This tends

to elevate ROS-mediated activation of eicosanoid and cytokine synthesis to produce a mildly proinflammatory environment. These proinflammatory signaling responses that are responsible for enhancing the risk for chronic diseases are, however, not often those of a *classic* inflammation response that is often described as following infections or cuts and scrapes. Rather, they are characteristic of a cellular stress response, one that produces many of the same end results. In general, the production of proinflammatory signaling molecules results in the activation of resident inflammatory cells (macrophages, dendritic cells, and mast cells, among others) and the infiltration of the local tissue by circulating inflammatory cells (neutrophils, monocytes, and platelets, among others). This results in additional proinflammatory signaling by the activated inflammatory cells as well as enhanced damage to cellular proteins, DNA, and membrane lipids through various ROS/reactive nitrogen species (RNS)-mediated mechanisms because of an increased production of ROS/RNS by the activated cells. In addition to inflammation-associated damage, alterations in cell functions due to responses to the various inflammation-associated cytokines occur. While the common nutrient insufficiencies are important in exacerbating proinflammatory signaling by compromising both antioxidant function and redox control, overconsumption of calories introduces another form of stress with proinflammatory consequences.

As described in Chapter 4, the metabolic stress of constant fatty acid overload in adipocytes leads to endoplasmic reticulum stress and dysfunction. This stress leads to increased JNK activity as well as the release of excess free fatty acids (FFAs), various eicosanoids, and cytokines, including tumor necrosis factor- α (TNF- α) from the adipocytes. The TNF- α and FFAs that enter the circulation are associated with mitochondrial dysfunction and insulin resistance in both adipocytes and skeletal muscle. The eicosanoids, along with TNF- α , and other cytokines attract circulating monocytes into the adipose tissue as well as act in a paracrine fashion to activate local tissue macrophages and preadipocytes (progenitor cells for adipose tissue) to initiate adipocyte hyperplasia. The accumulation of new cells then leads to hypoxia-mediated activation of a variety of cell activities, including synthesis of growth factors by vascular endothelial cells, macrophages, and fibroblasts to stimulate angiogenesis to support the new tissue cells. Angiogenesis is briefly discussed in Chapter 2 as a component of wound healing and is considered an integral part of damage-associated inflammatory responses that ultimately result in the resolution of the inflammation responses when the damage is repaired. In the case of adipose tissue, it is the same signaling molecules that result in the hypertrophy/ hyperplasia \rightarrow angiogenesis responses, except in this case it is ER stress due to fatty acid overload that initiates the process, not overt cellular damage in the traditional sense. There is also no resolution as long as the calorie overload continues. The process of angiogenesis also happens to be a very important response to hypoxia as part of the etiology of tumorigenesis, as described in more detail for cancer by Cannizzo and Ardies in Chapter 6. Thus, while angiogenesis is an essential component of wound healing following damage-induced inflammation, the same signaling molecules arising from the same types of cells are also components of the tissue hypertrophy/hyperplasia → angiogenesis responses that occur with developing obesity and with tumorigenesis. The same array of eicosanoids and cytokines produced

by stressed adipocytes and by activated macrophages (and other inflammatory cells) are instrumental in stimulating the proliferation of both progenitor cells and stem cells in any tissues and therefore an important component of cancer etiology as well as tissue repair. Metabolic stress of chronic calorie overload thus leads to increased proinflammatory signaling that can have far-reaching effects including aiding in tumorigenesis when circulating cytokines from growing adipose tissues add to the proinflammatory environment of a growing tumor.

The ingestion of compounds from environmental sources provides a common source of cellular stress that also enhances risk for cancer. From constantly breathing airborne pollutants to smoking and using tobacco products; consuming toxins, mutagens, and mitogens from drinking water, food, and food storage containers; and ingesting too much ethanol, exposures to exogenous transition metals, polycyclic aromatic hydrocarbons, various aldehydes, and many other carcinogenic compounds are varied but ubiquitous. Unfortunately, the cytochrome P450–mediated processes that are part of the elimination processes lead to the production of ROSs and in many cases a wide variety of reactive intermediates. In response to the various oxidants, proinflammatory signaling responses are initiated, often adding to the insufficiency-induced proinflammatory environment. The enhanced DNA damage coupled with increased proliferative signaling provides the perfect environment for enhancing risks for tumorigenesis.

As described in detail in Chapter 3, circulating proinflammatory molecules and elevated FFAs that occur as a result of chronic calorie overload are instrumental in the development of type 2 diabetes (T2D). Bourey, Najjar, and others explain that insulin resistance and visceral obesity represent chronic subacute inflammatory states, with TNF- α playing an important role in causing the insulin resistance. In addition, the role of carcinoembryonic antigen-related cell adhesion molecule 1 (CEACAM1)-mediated insulin clearance by the liver should not be overlooked. Impaired insulin clearance leads to increased insulin levels and subsequently the downregulation of peripheral insulin receptors to produce insulin resistance. In the liver, these effects result in enhanced de novo synthesis of lipids, leading to both fatty liver and an increased distribution of fat to adipose tissue to compound the already elevated adipose-mediated proinflammatory signaling. Impaired expression of CEACAM1 concomitant with nonalcoholic fatty liver disease often precedes insulin resistance in the obese, leading one to believe that impaired clearance of insulin by the liver is a primary event in developing T2D along with that of proinflammatory signaling. Inactivity simply compounds the problem by producing a chronic low-level proinflammatory environment in skeletal muscles, which is also a major factor in developing insulin resistance.

In a similar manner, osteoporosis also is now considered to be largely an inflammatory disease, as described by Dodington and Ward in Chapter 5. One reason for the inflammatory nature of osteoporosis etiology is the monocyte lineage of osteoclasts, the bone cells responsible for the degradation of bone. As differentiated monocytes, osteoclasts have many of the same receptors as macrophages and are thus able to produce proinflammatory cytokines and both osteoclasts and osteoblasts are sensitive to proinflammatory signaling as well as cell–cell interactions. In response to the production of proinflammatory cytokines, enhanced expression of

receptor activator of nuclear factor κ - β ligand (RANKL) occurs in osteoblasts, which increases RANKL-mediated activation of receptor activator of nuclear factor κ - β (RANK) on preosteoclasts to induce maturation of the bone-resorbing osteoclasts. Proinflammatory cytokines also increase the expression of RANK, leading to enhanced activation of existing osteoclasts. On activation, osteoclasts greatly increase their bone resorption activities and, in addition, produce macrophage colony-stimulating factor (M-CSF), which subsequently increases the proliferation of the preosteoclasts, further enhancing bone resorption. It is also important to recognize that a majority of Americans consume less than the EAR of vitamin D (60%), calcium (38%), and magnesium (45%), which are major contributing factors to poor mineralization of bone in the first place. Although it is difficult to identify the particular cellular stresses that initiate proinflammatory signaling within bone cells, it is certainly possible that the normal metabolic production of ROSs contributes to the activation process because of the poor antioxidant and redox control due to inadequate nutrient consumption. Any intrinsic production of proinflammatory molecules would certainly be enhanced by the entry of proinflammatory cytokines from the circulation, further increasing resorption. The enhanced bone resorption would be coupled with the impaired regulation of calcium homeostasis and insufficient calcium and vitamin D intake, which impairs bone formation; this is a perfect recipe for osteoporosis.

The nutritional insufficiencies and caloric excesses also exacerbate the development of NDs, as described by Raghavan and Shah in Chapter 7. Similar to the situation in bone, oxidative stress and proinflammatory signaling also appear to be major contributing factors in these diseases. Age-associated declines in mitochondrial function (potentially exacerbated by excess FFAs and TNF- α from weight gain and adipogenesis) and in the ubiquitin–proteasome system, coupled with increased oxidative stress and subsequent oxidative damage, lead to the gradual accumulation of misfolded proteins and protein aggregates over decades. Environmental exposures to transition metals that initiate ROS production also are important in mechanisms of cause, again illustrating the importance of maintaining optimal antioxidant and redox control. Coupled with these processes is an enhanced glutamate release leading to a calcium-associated excitotoxicity. Because mitochondria are important regulators of intracellular calcium homeostasis, the increasing degree of mitochondrial dysfunction that occurs with aging is also implicated as a major factor in ND.

The source of cellular stress that is the proximate cause of atherosclerosis is the intermittent low sheer forces and sometimes turbulent blood flow at curved and bifurcate sections of coronary (and other) arteries. The mechanical stress produced by this type of flow initiates proinflammatory signaling to produce a hypertrophic and proliferative response in the blood vessels at these sites. It is essentially an angiogenesis response to strengthen the vascular walls to compensate for inadequate flow; it also happens to result in the activation and accumulation of macrophages in the arterial wall. Here, the macrophages perform their normal scavenger function of phagocytizing oxidized membrane lipids, advanced glycation end products (AGEs), oxidized cholesterol, and oxidized lipoproteins when they bind to the macrophages' pattern recognition receptors. This enhances proinflammatory signaling further and ultimately results in the production of fatty streaks, plaque, and potentially vascular occlusion and/or thrombi. This process is exacerbated by poor antioxidant function that leads to more oxidation events, poor redox control that enhances proinflammatory signaling, insulin resistance that leads to an enhanced production of proinflammatory AGEs, increased blood pressure that increases mechanical stresses, and increased levels of oxidizable low-density lipoprotein and high-density lipoprotein (HDL). The details of these processes as discussed in Chapter 4 clearly indicate that contrary to popular interpretations of earlier epidemiology research atherosclerosis is essentially a nonresolving inflammatory disease initiated by mechanical stresses and is not a disease caused by cholesterol. In fact, cholesterol appears to be a relatively minor player in the etiology of atherosclerosis in comparison with the pervasive proinflammatory effects of inadequate diets, calorie excess, and inactivity. As described in some of Robert's earlier work, HDL also can be proatherogenic, further illustrating the hazards of making conclusions of cause and of prevention on the basis of epidemiological correlations.

A major role of dietary insufficiencies and calorie excess in risk for chronic diseases is clear. While calorie excess is certainly a major concern in the United States, poor food choices are also exceedingly important. Not only are poor choices responsible for the prevailing dietary insufficiencies but they also result in a very low consumption of phytochemicals. Because approximately 90% of Americans eat less than the recommended amounts of fruits and vegetables, the major sources for phytochemicals, they rarely consume enough to have a consistently beneficial effect. The wide array of phytochemicals in fruits and vegetables provide very important effects that greatly enhance cell survival. The enhanced protective effects include attenuating proinflammatory signaling, enhancing antioxidant function, reducing ROS and oxidant damage, increasing redox control, increasing phase II enzyme activities, enhancing DNA repair, and increasing the sensitivity of proliferating cells to checkpoint control and to apoptosis.

The various phytochemicals appear to modify cell function through modifying redox states at the microenvironment level within cells. Catechol polyphenols autoxidize to form quinones, which then cycle between reduction and oxidation states to produce hydrogen peroxide (H2O2). This then alters the oxidation states of Trx, Prx, and other redox-control enzymes, leading to the transient activation of mitogen-activated protein kinase (MAPK) signal transduction by Trx following the disassociation of oxidized Trx from apoptosis signal-regulating kinase 1. The transcription factors AP-1, Hsf1, and P53 are also activated by Ref1 following its oxidation by Trx. Phosphorylation of Nrf2 by MAPKs and oxidation of Nrf2 by H₂O₂ leads to its disassociation from Keap1, and it subsequently binds to the antioxidant response element (ARE) to induce the expression of Trx, Prx, mitochondrial superoxide dismutase (Sod2), catalase (CAT), glutathione reductase (GSR), glutamate cysteine ligase, glutathione-S-transferase (GST), glutathione peroxidase (GPX), heme-oxygenase-1, UDP-glucuronosyl transferase (UDP-GT), and NADPH:quinone oxidoreductase-1 (NQO1). Following autoxidation, some polyphenols bind covalently to Keap1 to produce the same effects. By inducing Trx, Prx, GST, GSR, Sod2, GSH, GPX, and CAT, intracellular antioxidant and redox control is optimized to reduce oxidant-mediated damage and attenuate proinflammatory signaling, thus producing a global reduction in risks for chronic diseases.

Among the effects of a reduction in ROSs and proinflammatory cytokines is a reduction in RANKL–RANK signaling to reduce osteoclast activation and proliferation. In addition, the phytochemical-associated transient increases in p38 and extracellular signal-related kinase (ERK)–MAPK activities are associated with the activation of Wnt/ β -catenin signaling and an enhanced expression of Runx2 in bone, leading to the differentiation of mesenchymal stem cells to the osteoblast lineage and increasing osteogenesis. Both of these lead to a reduced risk for osteoporosis.

Reduced ROS damage and expression of proinflammatory prostaglandins and cytokines in turn attenuates both nuclear DNA damage and proliferative signaling to reduce risk for cancer. The activation of Ref1 and P53 also increases the sensitivity of proliferating cells to checkpoint controls and enhances DNA repair, whereas the induction of UDP-GT, NQO1, and GSTs enhances the elimination of carcinogens to further reduce risk for tumorigenesis. The reductions in cellular ROSs and in proinflammatory signaling in combination with the induced Sod2 will help sustain mitochondrial function as well as insulin sensitivity, reducing the risk for both diabetes and NDs. In the central nervous system (CNS), polyphenols also appear to induce the expression of brain-derived neurotrophic factor (BDNF) by activating the cyclic adenosine monophosphate (AMP) response element binding (CREB) protein, possibly through the transient activation of ERK–MAPK. These latter effects enhance neural survival and synaptogenesis and thus contribute to the reduction in risk for NDs.

Thus, transient activation of Trx, MAPKs, Ref-1, and Nrf2 (among many others), mostly through oxidant-mediated effects, appears to be a primary event in the induction of the protective benefits of phytochemicals. These effects are in stark contrast to the detrimental effects of a constant increase in the cellular oxidation state that occurs with a chronic low-level increase in the cellular production of oxidants. The detrimental effects are essentially the same whether the increased oxidation state is derived from endogenous ROS/RNS (inflammatory and nutrient insufficiency) origins or exogenous (toxins and carcinogens) origins.

9.1.2 DIETARY RECOMMENDATIONS

With a large majority of Americans consuming a poor diet, fairly substantial changes will need to be made in order for most to obtain an optimal diet, one that is both nutritionally adequate and preventive. The *MyPlate* recommendations provide a very good starting point for ensuring nutrient intake. The minimum recommendations given here are intended for adolescents through 50 years and provide approximately 1900 cal each day on average. For the purposes of obtaining optimal prevention effects, however, with the exception of three servings of dairy, the *MyPlate* recommendations have been modified in order to increase the polyphenol content of the diet to levels observed in traditional Mediterranean diets (Table 9.1). This is accomplished by recommending minimum numbers of servings each week from specific groupings of fruits and vegetables and by separating out nuts/seeds and beans/peas (also good sources of protein) as two separate food categories with their own minimums. In essence, these changes add 0.5 cups of fruit/day and 1.25 cups of vegetables/day to the minimum intake in comparison with the *MyPlate* minimum recommendations for these two categories. In addition, phytochemical-rich extra

Food Group	Minimum Daily Servings (~1900 kcal)	Standard Serving Size	Approximate Kilocalories/ Serving	Additional Servings Each +400 kcal
Fruits	5	0.5 c.	71	1
Vegetables	6	0.5 c.	38	1
Cold-water fish	1 (2X week)	4 oz.	230	
Beans/legumes	2	0.5 c.	110	1
Nuts/seeds	1	0.25 c.	240	
Dairy (skim/low fat)	3	1 c. milk 2 oz. cheese	86	
Breads/cereals	5	1 oz./1 slice	78	
Extra virgin olive oil	2	1 tbsp.	100	
^a Red wine	1	4 oz.	85	

TABLE 9.1 Recommended Daily Food Servings

Note: Fruits: minimum of 2.5 cups each day with at least three different selections from each category each week:

A: elderberries, blueberries, pomegranates, blackberries, raspberries, saskatoon berries, blackcurrants

B: plums, oranges, grapefruit, lemons, cantaloupe, cherries, cranberries, red/black grapes, or apples. C: kiwi, bananas, pineapples, mango, pears, star fruit, guava, strawberries, peaches, or any other fruit.

- *Vegetables:* minimum of 3 cups each day with at least three different selections from each category each week:
 - A: chili peppers, broccoli, water cress, garden cress, spinach, arugula, Brussels sprouts
 - B: cauliflower, tomato, Romaine lettuce, green beans, sweet peppers, red cabbage
 - C: onion, garlic, celery, green cabbage, asparagus, Swiss chard
 - D: squash, zucchini, pumpkin, sweet potatoes, carrots, potato
- *Nuts and seeds:* minimum of one 0.25-cup serving each day with a minimum of two different selections from each category each week:
 - A: chestnuts, pecans, walnuts, pistachios
 - B: hazelnut, almonds, Brazil nuts, cashews, Macadamia nuts, or peanuts
- *Beans and legumes:* minimum of one 1-cup serving each day with a minimum of two different 1-cup selections from each category each week:
 - A: black beans, adzuki beans, kidney beans, peas/pea pods, lima beans
 - B: lentils, pinto beans, chili beans, chick peas, navy beans
- Whole grains and cereals: minimum five 1 oz. servings of whole grains each day with a minimum of one selection from each category each week:
 - A: buckwheat, wheat, bulgur
 - B: oats, corn, barley
 - C: wild rice, rice
- ^a Highly recommended. You should not use alcoholic beverages if you are pregnant or if you have been advised by your physician not to drink.

virgin olive oil is the only recommended oil to use for cooking, salads, and *spreads* (dipping). The number of servings in the meat category is a total of two servings of cold-water fish each week. This is to ensure sufficient consumption of EPA/DHA as well as to slightly reduce the overall protein intake. The consumption of other meats is not prohibited but rather not recommended as a daily or even a weekly habit. The five serving minimum for the grains/cereals category is restricted to servings for whole grains, helping to maximize phytochemical as well as vitamin and mineral intake. When the minimum numbers of servings are adhered to, the diet will on average meet the recommended dietary allowances or adequate intakes for all nutrients as well as providing somewhere between 1730 and 1960 kcal. The range of calorie values is because of the differences in caloric content of different foods within any one category as well as the presence or absence of the recommended servings of fish. The daily average of the minimum recommendations over a week is approximately 1900 kcal. In addition, by following these recommendations a minimum daily consumption of phenolics of approximately 1.3 g will result, well within the low-end range of a traditional Mediterranean diet and much higher than the average of less than 0.5 g/day reported for the "American" diet. Protein intake for this recommendation approximates to an average of 80 g/day (~70-~95 g/day depending on whether fish is eaten or not), somewhat lower than the average intake of ~ 91 g/ day for all adults 19–50 years of age, as reported in Chapter 6. The protein intake of these recommendations is equivalent to the 5 to 6 oz. of protein equivalents for adults recommended by the MyPlate guidelines and consistent with the recommendation for an intake of approximately 80 g/day by Bourey, Najjar, and others and by Raghavan and Shaw. The last recommendation is that the bulk of the calories should be eaten within 2 hours of finishing daily exercise. As discussed in Chapter 3, there is very little evidence to support stressing the importance of breakfast or of multiple small meals. On the other hand, it would be very important to take advantage of an exercise-induced increase in insulin sensitivity, especially for prediabetics and diabetics. Finally, although red wine is the only recommended alcoholic beverage, the one serving minimum is not included in the calorie/nutrient totals because it is not recommended for anyone who, for a variety of reasons, should not drink alcohol.

Randomly selected servings from each of the recommended lists that conformed to the recommended minimums in Table 9.1 were entered into daily food logs (*NutriBase 10 Professional Edition*; CyberSoft, Inc., 3851 E. Thunderhill Place, Phoenix Arizona, 85044) and analyzed for macronutrient, nutrient, and caloric contents in order to calculate the averages for calorie, protein, and nutrient intakes of the aforementioned minimum recommendations. The approximate calorie content for servings in each food group in Table 9.1 was obtained from the 2005 Dietary Guidelines for Americans (Department of Health and Human Services [DHHS]/U.S. Department of Agriculture [USDA]), in which calorie contents of mixed-food servings were reported. This will account for the discrepancy between the values obtained from the nutrient analysis by *NutriBase* for the actual caloric content of daily food menus (minimum servings = ~1900 kcal average, ~1730 kcal without salmon, and ~1960 kcal with salmon) and those obtained (~1700 kcal without salmon and 1830 kcal with salmon) using the USDA values for mixed-food servings. Because there are many foods available that are not listed in the recommended servings categories, the DHHS/USDA values may be more appropriate to use in estimating calorie contents of food servings for calories beyond the ~1900 minimum. It is important to note that as with any calorie analysis there will be differences in values obtained for the same foods simply because of natural variations in macronutrient and water content; therefore, any of the values used here can only be considered to be a reasonable approximation. The minimum phenolic content of the recommendations was based on the total content of a day's minimum recommended servings where the foods were specifically selected from the lists of foods to have the lowest tabled polyphenol content (from Chapter 2, Table 2.1). Therefore, it is assumed that randomly selecting foods from the required categories each day, in combination with any that are in various seasonings, will actually result in a daily polyphenol intake that averages a little more than 1.3 g, similar to a traditional Mediterranean diet. With the average sedentary adult female and male requiring approximately 2000 and 2400 cal each day, respectively, more calories than the 1900 daily average recommended minimum will be required for most people. Certainly, more will be required if they follow the recommended activity recommendations (discussed in Section 9.1.5) that will consume 500 cal or more; this leaves room for food options. For additional calories beyond the minimum recommendations, approximately half are recommended to come from fruit and/or vegetable and/or bean/legume sources, whereas the remainder should be considered optional. There is also nothing wrong with the occasional junk food snack, a chocolate confection or candy cane, indulging in your favorite barbecue ribs, a bowl of butterscotch ripple ice cream, or even overindulging at a party once in a while; there are plenty of optional calories embedded into the recommendations to do so. It just should not be done every day (or even every week).

9.1.3 **BEHAVIORAL ISSUES**

From eating too many calories for mostly the wrong reasons to poor food choices, the dietary behaviors of most Americans are in need of a complete overhaul. A major issue that needs to be dealt with is that the eating behaviors that are responsible for our current high prevalence of overweight and inadequate nutrient intake are behaviors that have been ingrained in us for a lifetime. For the most part we have learned them unquestioningly from our parents, and we continue them unthinkingly as adults. As detailed in Chapter 8 by Begg and Woods, these habitual behaviors not only include the timings and sizes of meals but also the tastes, textures, colors, and aesthetics of the specific foods that are most often chosen (in many cases, the same as those that were originally supplied by our parents) to be eaten, with each of these factors influencing different homeostatic and nonhomeostatic aspects of consumption through largely unconscious means. Because most people seem to be completely unaware that many different environmental cues affect eating behavior, it is apparent that eating is a relatively mindless behavior, that is, most details of the behavior itself and the details surrounding the behavior are simply not paid any attention. Because of the largely mindless aspect of eating behavior and the large array of nonhomeostatic influential factors, hunger and satiety are the least of the many factors that are important in the initiation and cessation of eating. It is also clear from their discussions that the habit of eating at certain times not only conditions the CNS to prime the gastrointestinal (GI) tract for eating at those times but also primes us to be hungry at those same times, even if we had just completed a meal an hour earlier at a nonhabitual time. Further, when eating at nonhabitual times (celebratory dinners, special brunches, out-for-coffee, parties, while on vacation, etc.) our ingrained hunger/satiety mechanisms do not come into play so readily, making it much easier to overeat: I am sure we have all been to parties where we just eat and eat and eat (and maybe even drink a little as well) and never quite feel satiated.

Although a chapter on hunger and satiety might seem out of place in a book on chronic diseases, it actually provides important information that should be considered along with dietary recommendations. It is tempting to think that because of the strong habitual nature of eating integrating some components of cognitive behavioral therapy into the process might be helpful in changing eating behavior, or at least be more helpful than the simplistic "just eat less" or "just eat more fruits and vegetables" give-advice approach. To quote Begg and Woods, "The primary goal for such therapy would be to learn, be conscious of, and permanently change dietary behavior to a pattern consistent with a healthier lifestyle" (Chapter 8, Section 8.4.2). The learning and conscious part of their advice would certainly fit the cognitive component of a cognitive therapy approach. Old behaviors built up over the lifetime of the individual have to be *unlearned*, while completely new permanent ones need to be learned; it is a task that is not to be taken lightly. One only need look to the data on weight loss to get an insight into the difficulties of the problem.

As they noted, most people who successfully lose weight in weight-loss programs ultimately regain the weight within months or years. In the subjects' defense, however, this may have something to do with the actual goal of weight loss itself. Once weight-loss goals are accomplished, the weight-loss diet can be abandoned. The unfortunate consequence of what is clearly a rational thought process is that the old eating habits are easily regained and weight gain ultimately ensues. With a current and very public focus on obesity as a (or more probably the) major health issue today, weight loss becomes the default focus for prevention, as though losing weight is the major relevant factor in regaining optimal health. If it is believed by the general public to be the major health problem then old habits will not be changed, and risks for chronic diseases will remain.

The major focus of the recommendations here is that a permanent change in eating behavior must be made; it must be made in context with learning to be aware of how much food and which specific foods need to be eaten to attain optimal health, which includes appropriate weight control. The ultimate goal is to create new eating behaviors that consciously optimize nutrient and phytochemical intake along with appropriate conscious caloric control to regulate body weight at a healthy level. The implication of this is that learning the new (and *permanent*) diet is one priority and another priority is learning caloric control. At the risk of stepping on the toes of various professionals (I am certainly not trained as a behavioral therapist), the following is a suggestion for one possible approach to learning the new (and permanent) lifestyle diet.

These learning objectives might be met by obtaining cups, containers, and bowls that conform to standardized 0.25, 0.5, and 1.0 c. serving sizes and to volumes that equate to an approximate 4 oz. serving size for fish (about half of a salmon steak).

From the foods and servings in Table 9.1, selections of foods from each recommended category that conform to the minimum recommendations plus sufficient additional servings to meet average caloric needs can be made with one caveat: the standardsize containers *must* be used to measure out the appropriate size and number of servings for the entire day at once. For the first 2 weeks of this procedure, using the average 2000 and 2400 kcal for women and men, respectively, is recommended. To meet the 2000 and 2400 kcal target, calories beyond the minimum 1900 kcal servings should come exclusively from fruits, vegetables, beans/peas, and/or whole grains and the tabled caloric content of the servings should be used as a guide to determine the appropriate number of additional servings. Visualizing an entire day's worth of food (and the inherent calories) is one possible way of learning to get a feel for the different volumes of foods and how they can be variously apportioned throughout the day. (Each day's number of servings and approximate caloric content also should be recorded for later reference as well.) The servings can then be placed on the usual dinnerware and consumed at customary times to minimize homeostatic disturbances. Although this does not necessarily conform to others' recommendations that include using smaller sized dinnerware than the current and customary large sizes, the intent of using serving-size portions is to learn and be conscious of serving sizes, regardless of their ultimate dinnerware environment. Although seemingly trivial, the goal of this initial effort is to continually be conscious of what the different standardized servings for the new diet look like as they are put in different combinations on different serving ware each meal and to be conscious of how much is being eaten. Using the common rule of thumb for behavior modification, it may take at least a month of using the serving-size containers to start to break the old habits of filling the plate and to learn to recognize and use appropriate serving sizes. At the same time, the physical presence (and use) of the serving-size containers might serve as a constant reminder of the learning tasks each day. Although this particular procedure in learning new diets has not been tested rigorously, it is a logical extension of the concept that unconscious interactions between intrinsic and extrinsic factors are an integral part of the preexisting eating behaviors; therefore, we need to focus on and learn to visualize and be conscious of the new foods and their portions as part of learning the new eating behavior.

As new dietary behaviors are being acquired, the second major goal of learning weight control in the context of the new diet can start. Sometime during the first week or two of practicing rigid control of food servings, measuring and recording body weight every day (at the same time and under the same conditions as much as practical) could start. Attempts at changing intake should not be made at this time because the purpose is to monitor weight at the specific food intake being used. Only after at least 2 weeks of recording body weight should attempts be made to relate weight to daily food intake. The reason for this is that a weekly average for body weight should be used as the *reference* body weight and not the weight on any single day. This is to minimize the effects of daily variations in body weight due to the presence or absence of food in the GI tract and fluctuations in hydration state. Once weekly averages are available, adjustments to food intake can be made to see the effect this may have on body weight. For example, one daily serving of grains and one of meat could be removed each day for 2 weeks to see if the reduction of calories

results in a loss or stabilization of body weight. Alternatively, all alcoholic beverages could be removed to the same end. On the other hand, servings of specific foods or beverages could be added and the effects observed. The purpose of such interventions is not necessarily to lose or gain weight but rather to learn to be conscious of the effects that adjusting food intake may have on body weight. Learning these adjustments may also be important in developing an intrinsic sense for what would be a realistic weight-loss goal based on experience with the new dietary behavior. After 4–6 weeks of this initial learning phase, one should have a reasonably good record of food intake and average body weight each week to be able to see what it might take to maintain or lose body weight. After this learning phase has been completed, weight-loss or weight management goals can be discussed and set. The purpose of a paradigm such as this is to suggest that the learning of a new and mindful eating behavior is the first priority of an overall dietary program and that any weight management issues are dealt with only after starting to learn the new dietary behaviors.

Part of this learning phase includes learning to consume unaccustomed quantities of what may be many new foods. Again, this is not a trivial learning experience. Eating the unaccustomed foods in unaccustomed amounts might be in some cases a somewhat unsettling experience. The new foods simply will not conform to the habitual tastes and aesthetics of the usual foods and portions and therefore will not meet the expected nonhomeostatic stimuli, possibly disrupting the satiety and pleasurable feelings normally associated with eating. Further, because of the innumerable pleasant memories of the tastes of various foods and the satiety that accompanies eating them, memories that must have been formed over hundreds and perhaps thousands of repetitions of the same eating behaviors, the old habits might be badly missed as well. Issues relating to shopping for unaccustomed foods (how do you even know they are ripe or ready?) or preparing them also may lead to frustrations. Because of such issues, engaging the services of a professional dietitian, nutritionist, or personal trainer might help to facilitate the process. At first, daily or weekly contacts with such a professional, especially one familiar with using a behavioral-type approach, for discussing and resolving such issues, or for other counseling needs, might be very helpful. Less intensive *update* sessions later on in the program also might be highly useful to help enhance adherence to the new lifestyle. Similar professional assistance is advisable for the exercise component of the recommendations as well.

9.1.4 EXERCISE AND CHRONIC DISEASE

The lack of physical activity for the majority of Americans is a serious issue with respect to risks for chronic diseases. One rather visible consequence of a sedentary lifestyle is a tendency to gain weight simply because of not expending sufficient calories. Sedentary recreational pursuits (computer games, television, social networking, reading, watching movies, etc.) and sedentary occupations make a deadly combination. When one considers the difficulty in changing eating habits, enjoying an increasingly sedentary lifestyle as we age from early adolescence through middle age without decreasing caloric consumption to compensate is certainly understandable. Unfortunately, the end result for a large majority of Americans is a constant gradual increase in adiposity and body weight and ever-worsening fitness, making
it even harder to consider physical activities for recreation (and health). And, the increasing adiposity greatly enhances for chronic diseases through the proinflammatory mechanisms of adipose hyperplasia and angiogenesis.

Another major consequence of inactivity is the development of insulin resistance in skeletal muscle. Inactivity induces a continuous small increase in ROS production by skeletal muscle cells that then produces a mild chronic proinflammatory environment, ultimately leading to TNF- α -mediated inactivation of insulin receptor substrate proteins and impaired insulin function in skeletal muscle. As a result of insulin resistance, levels of glucose in the blood are elevated and the formation of Amadori products is increased. These adducts are eventually modified to form various AGEs, which then activate AGE receptors on macrophages, smooth muscle cells, and vascular endothelial cells, enhancing inflammatory signaling in essentially all tissues. While inactivity-induced insulin resistance in skeletal muscle occurs without weight gain, the progression toward insulin resistance through inactivity can be significantly enhanced by increased circulating FFAs and TNF- α coming from expanding adipose tissue. The combination of inactivity and chronic weight gain is truly a vicious cycle resulting in increasing chronic inflammation and risks for chronic diseases.

An important aspect of physical activity reducing risks for chronic diseases is that its effects go far beyond that of simply avoiding excess risks that are associated with inactivity and calorie excess. As a result of the calcium flooding the muscle cells to stimulate muscle contractions and the metabolic production of ATP, increased production of ROSs and AMP occurs. The increase in ROSs leads to an increase in the oxidation state of muscle cells that lasts at least as long as the activity does, resulting in the oxidation of Trx, Prx, and other redox-control enzymes and subsequent increases in activity of various MAPKs. The Ca²⁺ also activates protein kinase C (PKC), calcineurin A (CnA), and Ca²⁺/calmodulin-dependent protein kinase (CaMK) while the AMP activates AMP-dependent protein kinase (AMPK).

CnA and CaMK activate CREB and MEF2 to enhance the synthesis of PGC-1 α while AMPK activates AS160 and TBC1D1 (Rab-GTPases), silent information regulator T1, and PGC-1 α , with PGC-1 subsequently activating Nrf2, Nrf1, and other transcription factors. These responses lead to enhanced insulin sensitivity in skele-tal muscle as well as mitochondrial biogenesis and enhanced antioxidant and redox control (via Nrf2). These same effects also occur in nonmuscle tissues, resulting in globally enhanced antioxidant and redox control, enhanced DNA repair, reduced ROS and oxidative damage, increased elimination of carcinogens, and increased sensitivity to checkpoint controls to significantly reduce risk for cancers. The activation of various protein kinases by exercise in neural tissues is also associated with activation of CREB and increased expression of BDNF, which when coupled with the enhanced mitochondrial function stabilizes nerve function and reduces risk for NDs.

In addition to the various metabolic stresses placed on cells by exercise are those that result from the physical aspects of weight-bearing movements. The mechanical stresses placed on the bones during exercise produce cellular stress responses via various stretch-activated calcium channels, integrin deformation, and shear-induced activation of various G-proteins. As a result of these effects, increased Wnt/ β -catenin signaling occurs to enhance osteogenesis. It is important to note that all of the mechanisms discussed for exercise-based prevention are very similar to those produced by various phytochemicals and mediated by essentially the same molecular mechanisms. As such, the preventive effects are dependent on the transient nature of exercise stress. Regarding osteoporosis, this concept is illustrated by the fact that in obese individuals where the mechanical stresses are constant bone strength is actually compromised.

The responses of muscle cells to the increased production of oxidants (and to calcium) during muscle contraction also include the activation of PLA2, resulting in the synthesis of proinflammatory prostaglandins, leukotrienes, and thromboxanes. In addition, the increased MAPK activities are associated with a large increase in the expression of interleukin (IL)-6. While these are traditionally considered to be proinflammatory stress responses, in the event that there is no damage and subsequent TNF- α production, the IL-6 response results in a profound anti-inflammatory effect. IL-6 mediates an induction of sIL-1r, sTNFr, sIL-1ra, and IL-10. The sIL-1r and sTNFr bind to circulating IL-1 α , IL-1 β , and TNF- α while the sIL-1ra binds to IL-1 α and IL-1ß receptors, thus attenuating proinflammatory cellular responses. The IL-10 suppresses nuclear factor κ - β and MAPKs, increases suppressor of cytokine signaling, and stimulates nonphlogistic phagocytosis, events that commonly occur during the resolution phase of inflammation. These effects only occur as a result of moderate-intensity large muscle-group exercise that is sustained for about an hour, and they can be enhanced further with even longer exercise periods. They also will not occur with damaging or exhaustive exercise, indicating that too much exercise stress can actually be counterproductive. These IL-6-mediated anti-inflammatory effects occur in addition to those of Nrf2-mediated enhanced antioxidant and redox control and represent an important additional benefit of exercise: a profound systemic anti-inflammatory and proresolution effect. While the non-IL-6-mediated effects discussed earlier will reduce risks for chronic diseases, the inclusion of 60 min/day of moderately stressful exercise is essential to maximize risk reduction.

9.1.5 Exercise Recommendations

Based on the various cellular effects of exercise, in order to obtain optimal reduction in risk for chronic diseases the recommended goal is to perform vigorous activities that are 6 METS or greater in intensity and carried out for a minimum of 1 h/day for 5 days a week when starting an exercise program. These activities include jogging/running, cycling at 10–12 mph or greater, swimming, cross-country skiing, or similar continuous vigorous activities. An example of a suggested protocol to follow is detailed in Table 9.2.

This schedule is actually designed for older or very sedentary adults. Notice that it can take almost a year to progress to the frequency and intensity that will maximize health benefits. After the 48 weeks of progressive training, the final two days of activity can be added in at the same one additional day each 6 weeks to achieve daily activity that should maximize preventive effects. There is simply no easy or fast way to do this. Unaccustomed physical activity performed for the first time results in a lot of physical discomfort, and it will take time to recover from this initial effort. It also takes time for

Training Schedule for a Sedentary Person					
	м	т	W	т	F
Week 1	~20 Minutes ^a walk/jog				
Week 2	~20 Minutes walk/jog			~20 Minutes walk/jog	
Week 3	~20 Minutes walk/jog		~20 Minutes walk/jog		~20 Minutes walk/jog
Week 4	~20 Minutes walk/jog	~20 Minutes walk/jog	~20 Minutes walk/jog		~20 Minutes walk/jog
Week 5–6	~20 Minutes walk/jog	~20 Minutes walk/jog	~20 Minutes walk/jog	~20 Minutes walk/jog	~20 Minutes walk/jog
Week 7–12	Increase the time to 30 minutes, following the same 6-week pattern. (Add 10 minutes to 1 additional day each week.)				
Week 13-18	Increase the intensity to more jogging than walking, following the same 6-week pattern (increasing the intensity on 1 additional day each week).				
Week 19-24	Increase the intensity to jogging only, following the same 6-week pattern.				
Week 25-30	Increase the intensity to running only, following the same 6-week pattern.				
Week 31-36	Increase the time to 40 minutes, following the same 6-week pattern.				
Week 37-42	Increase the tin	ne to 50 minutes	, following the sa	me 6-week patte	ern.
Week 43-48	Increase the tin	ne to 60 minutes	, following the sa	me 6-week patte	ern.
Note: After a	vacation period	, the intensity sh	ould be lowered	when starting ba	ick and gradual

TABLE 9.2 Training Schedule for a Sedentary Person

Note: After a vacation period, the intensity should be lowered when starting back and gradually increased to the previous level of intensity over a period of several weeks.

^a More walking than jogging.

the physical structures of muscles and bones and the various enzyme systems to adapt to the increasing duration and intensity levels of the activity. As with the first time performing any activity, increasing intensity and/or duration also introduces discomfort, hence the 1-week recovery period between the first and second bouts of activity and the gradual increase in the frequency and intensity of the additional exercise periods. The pattern of progressively increasing activity levels that is presented here is consistent with the idea of microprogressions, which is often associated with rehabilitation of sports injuries: small gradual increases in intensity to minimize the risk of reinjury. Although sedentary individuals typically will not have such injuries, the majority will be in a very poor physical condition and sudden large increases in activity could easily lead to injury. For this reason, the microprogression approach is recommended.

Even though meeting these goals for exercise provides optimal benefits that include the profound anti-inflammatory effects of large amounts of IL-6, this does not mean that there are not significant protective benefits to lesser amounts or different types of activity. Simply being active regularly will enhance insulin sensitivity to produce a moderate reduction in risks. Such preventive effects may start to be realized after the first 4 or 5 weeks of following the suggested program, long before the 60-minutes goal is reached. This is an important concept; there are preventive benefits from the lower levels of activity, and sustaining activity at these lower levels will reduce risks. Similar levels of protection might be provided through several short (~10 minutes or so) periods of daily low- to moderate-intensity activities such as fast walking, yard maintenance, leisurely bicycling, playing volleyball, playing baseball, or skating and through resistance-type weight-lifting and muscle endurance–training activities.

Although it does not provide maximal effects, reductions in risk are also obtained from the shorter duration (20–30 minutes) continuous, moderate-intensity to higher intensity running (or cycling, swimming, and cross-country skiing) when performed–three to five times each week. The benefits include maintained insulin sensitivity, enhanced antioxidant protection and redox control, reduced proinflammatory signaling, and possibly enhanced DNA synthesis. These benefits may start to be realized after about the 30-week time frame of the training program described in Table 9.2, and sustaining activity at this level reduces risks more than at the 5-week level.

Developing an exercise habit also entails behavioral issues that are similar to those inherent in learning a new diet. Unaccustomed exercise not only can make one uncomfortable but also is not the first thing a sedentary person will think of when it comes to spending his or her leisure time. There are plenty of sedentary activities that are much more pleasurable to participate in, and more than likely they have been participated in regularly and will be missed if not. This leads to the concept that exercise behaviors need to be learned and developed into a new daily behavior, hopefully one that will be missed if not participated in. It is pretty difficult to have feelings of missing an activity when initially there is little to no positive feedback when it is performed for the first time. For this reason, utilizing the skills of a competent personal trainer who has training in behavioral therapy as well as in exercise (and nutrition) sciences may be highly beneficial. Developing the capacity to learn to enjoy what at first is an unfamiliar and uncomfortable activity with the help of a specialist can be very helpful. The additional encouragement, empathy, expert instruction, and help with learning new behaviors that can be provided by a highly competent personal trainer, one with appropriate advanced training, can often make a difference between successfully maintaining the activity program and reverting back to a sedentary lifestyle.

One last thought on these exercise recommendations: they are made with maximal reduction in risk for chronic diseases in mind, which is after all the focus of this book. Exercise activities for maintaining optimal strength, flexibility, balance, muscle endurance, and all other physical abilities necessary for maintaining an optimal capacity for activities of daily living and for healthy aging are not to be belittled. Alternating days with 60 minutes of continuous activities with days of muscle endurance-type weight-lifting activities may optimize aging while only slightly compromising maximal prevention of chronic diseases. (Due to space limitations, however, details of training progressions for these will not be presented.) Hopefully, with properly designed clinical trials such concepts can be tested.

9.1.6 FINAL ANALYSIS

Exercise clearly has the ability to initiate changes in cellular functions that profoundly reduce risk for chronic diseases, as does a nutritionally adequate and phytochemical-rich diet. These beneficial changes are initiated predominantly through transient increases in the oxidation state of cells that then alters the patterns of activated transcription factors and subsequently the expression of various genes to enhance cell function and survival. How these recommended treatments alter epigenetics is unknown as of the time of this writing. Because hypermethylation processes appear to be initiated in part by ROS-mediated damage, it is certainly possible that the enhanced antioxidant and redox control brought about by following these recommendations will alter the hypermethylation patterns of CpG islands as well. This could conceivably increase the affinity of promoters for their transcription factors and result in greater activation of expression activities, thus further enhancing the preventive effects of phytochemicals and exercise over time. Obviously, this remains to be tested. If confirmed, however, such effects might even be passed on to progeny through inheriting the *new* hypermethylation patterns, producing progeny that may then possess an inherently greater sensitivity to the protective effects of proper diet and exercise. This would certainly be an improvement to what might be the current American Death Spiral of poor diet and inactivity leading to chronic disease and potentially disease-promoting epigenetic effects that might then be passed on to progeny, resulting in ever-worsening risks for disease with each subsequent generation.

Health practitioners have been educating students, patients, and the general public about the health risks associated with poor diets and inactivity and about the health benefits of proper diet and exercise for decades, with very disappointing results. The majority of Americans still do not meet the minimum recommendations for physical activity, and the proportion that does not meet the minimum activity requirements has been continually increasing during the last three decades. At the same time, the majority of the population still eats a nutritionally insufficient diet, a situation that certainly has not been improving either. Something clearly needs to change because the current approach is simply not working very well. In light of this record of failure for the majority of Americans, perhaps a much more intensive and personal approach to wellness is necessary in order to attain the type of behavior modifications that are necessary for achieving optimal health. The aforementioned recommendations appear to include a shameless plug to create more employment opportunities for personal trainers, nutritionists, and dieticians, and it is meant to emphasize the extremely difficult time that most people actually have in successfully learning a healthy and permanent new lifestyle and that most will need professional assistance. Many medical centers throughout the country have already recognized this and established wellness programs that include such services. Unfortunately, these programs are usually on a fee-for-service basis so that they are available predominantly only to those who can afford them.

At the risk of inappropriately entering the sometimes perverse world of American politics, in the final analysis the ability to successfully produce and sustain healthy alterations in behavior probably depends on changes to government policy and public education. There needs to be a universal acceptance that both diet and physical activity are far more important to health than is currently acknowledged and that without comprehensive changes in how health and wellness education and health policy are approached very little will change. The following is one example from the government side of things: the current nationwide farm-subsidy programs in America include subsidies for the production of wheat (and various other grains) as well as feed grains for beef and other livestock, helping to keep meat prices low while little to no support is available for fresh fruit and vegetable produce. Although the subsidies are certainly not the only reason for the relatively low prices for meats and other processed foods, these policies do seem to be the exact opposite of what would be in the best health interests of the population. One result of these policies is that some of the most expensive foods in the grocery store happen to be the fruits and vegetables that Americans eat the least. In some respects, there might be little point to a public health service policy of encouraging increased consumption of fruits and vegetables when so many Americans simply cannot afford to buy them and at the same time our current economic policies ensure they are the most expensive. Perhaps, transferring all of the current subsidies to fruit and vegetables at more reasonable prices, increasing the prices of meat, meat products, and highly processed grain products at the same time. Although not a panacea, such changes in policies might at least create more of a financial incentive for consumers to eat a healthier diet.

Integrating diet, nutrition, and physical activity concepts into daily healthwellness classes to help children develop a conscious focus on health and wellness behaviors from a young age also might help to avoid far greater difficulties in changing their behavior as adults. Unfortunately, to accomplish this it would take a complete overhaul of the public education and teacher education systems; currently, proper nutrition and exercise science education for public school teachers (and for many health practitioners as well for that matter) is essentially nonexistent. Without properly educated health teachers, it is highly doubtful that meaningful health and wellness education could take place anywhere.

Obviously, wholesale changes in lifestyle behaviors are necessary to significantly reduce risks for chronic diseases and reduce the more than \$2.7 trillion (2011) in health spending in the United States. This will probably take a concerted effort on the part of the medical community, community and public school health educators, researchers, employers, and the government. We hope that the material in this book provides some direction on just how far we really have to go to obtain optimal reduction of the biologic-based risks for chronic diseases. The rest is up to the reader.

ACKNOWLEDGMENT

This work was supported in part by a partial salary grant from NEIU to C. Murray Ardies

This page intentionally left blank

Diet, Exercise, and Chronic Disease

The Biological Basis of Prevention

Exercise and diet are key factors in the etiology and prevention of chronic disease. While most books on chronic disease have a decided clinical approach, *Diet, Exercise, and Chronic Disease: The Biological Basis of Prevention* brings together the latest cellular- and molecular-based research on the etiology of chronic diseases and the impact of various aspects of diet and exercise on the causal mechanisms. By focusing on cellular biology, details of the integrative nature of the many different underlying factors are revealed—details that are not evident with the prevailing clinical approach to chronic disease.

This book highlights chronic diseases that are major causes of mortality, and which have sufficient molecular evidence for dietary and activity-related components to their etiology. Individual chapters examine the role of diet and exercise in diabetes, atherosclerosis, osteoporosis, cancer, and neurodegenerative disease. They cover aspects such as disease etiology, effects of diet and exercise, and the cellular and molecular mechanisms of how various dietary components and repeated exercise alter disease etiology to contribute to disease prevention.

Since inflammatory signaling is a fundamental component of the chronic diseases discussed, the book includes a separate chapter on inflammation and innate immune responses. Obesity as a contributing factor is addressed within the specific disease chapters. The book also reviews what is known about the factors that influence food intake in humans. This reference translates molecular-based data on etiology and prevention into a clinical prescription for the prevention of chronic disease.



6000 Broken Sound Parkway, NW Suite 300, Boca Raton, FL 33487 711 Third Avenue New York, NY 10017 2 Park Square, Milton Park Abingdon, Oxon OX14 4RN, UK



K75595

www.crcpress.com